

# **REMEDIAL ACTION WORKPLAN**

For:

Proposed Redevelopment 500 Main Street Laundry New Rochelle, Westchester County, New York

BCP # C360199

**Prepared for:** 

BRP 500 Main LLC c/o BRP Companies 767 3rd Avenue, 33rd Floor New York, New York 10017

Prepared by:

SESI CONSULTING ENGINEERS, D.P.C. 12A Maple Avenue Pine Brook, NJ 07058

> NOVEMBER 2020 REVISED FEBRUARY 2021

I, Fuad Dahan, certify that I am currently a NYS registered professional engineer as defined in 6 NYCRR Part 375 and that this Remedial Action Work Plan was prepared in accordance with all applicable statutes and regulations and in substantial conformance with the DER Technical Guidance for Site Investigation and Remediation (DER-10)

Fuad Dahan	02/05/2021	The New York
NYS Professional Engineer (# 090531)	Date	Signature

It is a violation of Article 130 of New York State Education Law for any person to alter this document in any way without the express written verification of adoption by any New York State licensed engineer in accordance with Section 7209(2), Article 130, New York State Education.

# TABLE OF CONTENTS

I		
LIST OF ACRONYMSVII		
EXECUTIVE SUMMARY1		
1.0 INTRODUCTION		
1.1 Site Location and Description		
1.2 Proposed Redevelopment Plan6		
2.0 DESCRIPTION OF REMEDIAL INVESTIGATION CONCLUSIONS		
2.1 Soil7		
2.2 Groundwater7		
2.3 Soil Vapor		
2.4 Geological Conditions		
2.5 Conceptual Site Model of Contamination Transport9		
2.6 Identification of Standards, Criteria and Guidance10		
2.7 Environmental and Public health assessments13		
2.7.1 Qualitative Human Health Exposure Assessment13		
2.7.2 Fish and Wildlife Impact Analysis14		
2.8 Significant Threat		
2.9 Remedial Action Objectives14		
2.9.1 Groundwater14		
2.9.2 Soil		
2.9.3 Soil Vapor RAOs15		
3.0 DESCRIPTION OF REMEDIAL ACTION PLAN		
3.1 Evaluation of Remedial Alternatives		
3.3 Summary of Selected Remedial Actions23		
4.0 REMEDIAL ACTION PROGRAM		
4.1 Governing Documents 20		
4.1 Governing Documents		
4.1.1 Site Specific freatile Safety Flan ( $(ASF)$		
4.1.2 Quality Associated Hoject Hull (QALL)		
4.1.5 Son/ Water Pollution Prevention Plan (SWPPP)		
4.1.4 Scommunity Air Monitoring Plan (CAMP)		
4 2 General Remedial Construction Information 30		
4 2 1 Project Organization		
4.2.2 Remedial Engineer		
4.2.3 Remedial Action Construction Schedule		
4.2.4 Work Hours		
4.2.5 Site Security		
4.2.6 NYSDEC BCP Signage		

4.3.1 Mobilization	33
4.3.2 Erosion and Sedimentation Controls	
4.3.3 Site Preparation Demolition	
4.3.4 Utility Marker and Easements Layout	34
4.3.5 Sheeting and Shoring	34
4.3.6 Dewatering	34
4.3.7 Equipment and Material Staging	35
4.3.8 Decontamination Area	35
4.3.9 Site Fencing	35
4.3.10 Demobilization	35
4.4 Reporting	
4.4.1 Weekly Reports	
4.4.2 Other Reporting	
4.4.3 Complaint Management Plan	
4.4.4 Deviations from the Remedial Action Work Plan	
5.0 REMEDIAL ACTION	
5.1 Cleanup Objectives	
5.2 Remedial Performance Evaluation	
5.2.1 Soil Removal Verification	
5.2.2 Groundwater Sampling	
5.2.3 Soil Vapor Intrusion Evaluation	
5.3 Methodology	
5.4 Reporting of Results	
5.5 QA/QC	
5.6 DUSR	
5.7 Reporting of Performance Monitoring Data in FER	40
5.8 Estimated Material Removal Quantities	41
5.9 Soil/Materials Management Plan	41
5.9.1 Soil Screening Methods	41
5.9.2 Stockpile Methods for Contaminated Soils	
5.9.3 Materials Excavation and Load Out	
5.9.4 Materials Transport Off-Site	43
5.9.5 Materials Disposal Off-Site	
5.9.6 Fluids Management	
5.9.7 Demarcation Barrier	46
5.9.8 Backfill from Off-Site Sources	
5.9.9 Contingency Plan	
5.9.10 Community Air Monitoring Plan	
5.9.11 Odor, Dust and Nuisance Control Plan	
6.0 ENGINEERING CONTROLS	51
6.1 Monitored Natural Attenuation	
6.1.1 Groundwater Monitoring System	
6.1.2 Criteria for Completion of Remediation/Termination of Groundwater Monitoring	
7.0 INSTITUTIONAL CONTROLS	53

7.1 Environmental Easement	
7.2 Site Management Plan	
8.0 FINAL ENGINEERING REPORT	
8.1 Certifications	57

# LIST OF TABLES

- Table 5.1 Project personnel
- Table 5.2 Remedial Action Schedule
- Table 5.3 Emergency and Contact Numbers

### LIST OF FIGURES

- Figure 1.1 Site Location Map
- Figure 1.2 Site Plan
- Figure 2.1A, B, C, D Soil Boring Sample Results Plans
- Figure 2.2 Groundwater Sampling Location Plan
- Figure 2.2A Groundwater Emerging Contaminant Results
- Figure 2.3 Soil Vapor Results and Location Plan
- Figure 2.4 Groundwater Contour Map
- Figure 3.1 Proposed Excavation Plan

## LIST OF APPENDICES

- Appendix A Site Development Plan
- Appendix B Health and Safety Plan
- Appendix C Quality Assurance Project Plan
- Appendix D Community Air Monitoring Plan
- Appendix E NYSDEC Soil Cleanup Objectives
- Appendix F Citizens Participation Plan
- Appendix G Stormwater Pollution Prevention Plan

# LIST OF ACRONYMS

Acronym	Definition
AWQS	Ambient Water Quality Standards
BCA	Brownfield Cleanup Agreement
BCP	Brownfield Cleanup Program
bsg	Below Street Grade
bbg	Below Basement Grade
CAMP	Community Air Monitoring Plan
COC	Contaminant of Concern
CPP	Citizens Participation Plan
CVOCs	Chlorinated Volatile Organic Compounds
су	Cubic yard
DER	Division of Environmental Remediation
DER-10	NYSDEC Technical Guidance for Site Investigation & Remediation
DMM	Division of Materials Management
DO	Dissolved Oxygen
DUSR	Data Usability Summary Report
ECs	Engineering Controls
ECL	Environmental Conservation Law
EE	Environmental Easement
ESA	Environmental Site Assessment
FER	Final Engineering Report
FWRIA	Fish and Wildlife Resources Impact Analysis
GAC	Granular Activated Carbon
HASP	Health and Safety Plan
HHEA	Human Health Exposure Assessment
ICs	Institutional Controls

Acronym	Definition	
MNA	Monitored Natural Attenuation	
MW	Monitoring Well	
NYSDEC	New York State Department of Environmental Conservation	
NYSDOH	New York State Department of Health	
PAHs	Polyaromatic Hydrocarbons	
РСВ	Polychlorinated Biphenyls	
PFOS	Perfluorooctanesulfonic Acid	
PFHxA	Perfluorohexanoic Acid	
PFOA	Perfluorooctanoic Acid	
PFAS	Per and Polyfluoroalkyl Substances	
PID	Photoionization Detector	
QAPP	Quality Assurance Project Plan	
QA/QC	Quality Assurance/Quality Control	
OSHA	Occupational Safety and Health Administration	
RAOs	Remedial Action Objectives	
RAWP	Remedial Action Work Plan	
RCRA	Resource Conservation and Recovery Act	
RECs	Recognized Environmental Concerns	
RI	Remedial Investigation	
RIR	Remedial Investigation Report	
RIWP	Remedial Investigation Work Plan	
RRSCOs	Restricted Residential Soil Cleanup Objectives	
SESC	Soil Erosion and Sediment Control	
SCG	Standards, Criteria, and Guidance	
SCO	Soil Cleanup Objectives	
SOE	Support of Excavation	

Acronym	Definition
SOP	Standard Operating Procedure
SoMP	Soil/Materials Management Plan
SESI	SESI Consulting Engineers, DPC
SMP	Site Management Plan
SSDS	Sub-Stab Depressurization System
SVOCs	Semi-Volatile Organic Compounds
SWPPP	Storm-Water Pollution Prevention Plan
TAGM	Technical and Administrative Guidance
	Memorandum
TAL	Target Analyte List
TCE	Trichloroethene
TCL	Target Compound List
TOGS	Technical and Operations Guidance Series
USCO	Unrestricted Use Soil Cleanup Objectives
USEPA	United States Environmental Protection Agency
UST	Underground Storage Tank
VOCs	Volatile Organic Compounds
WCDOH	Westchester County Department of Health

# **EXECUTIVE SUMMARY**

#### Site Description/Physical Setting/Site History

The New York State Department of Environmental Conservation (NYSDEC) has entered into a Brownfield Cleanup Agreement (BCA) with BRP 500 Main LLC (the Volunteer) on June 3, 2020 concerning the property located at 500, 506, and 510 Main Street, and 12 Church Street in New Rochelle, Westchester County, New York (the "Site"), designated the 500 Main Street Laundry BCP Site No. C360199. The Site consists of a 0.79-acre property located at 500, 506, and 510 Main Street and 12 Church Street in New Rochelle, Westchester County, New York. The Site is bound to the northwest by Main Street and to the southwest by Church Street and is surrounded by commercial properties to the northeast and southeast.

The 0.79-acre rectangular-shaped Site is located in an urban area in the Downtown District of the City of New Rochelle, New York. The Site comprises four (4) contiguous parcels and is identified on the Westchester County Clerk's map as tax parcels 1-215-0012, 1-215-0011, 1-215-0010, and 1-215-0008.

According to the 1970 Geologic Map of New York – Lower Hudson Sheet published by the University of the State of New York, the bedrock underlying the Site is the Hartland Formation and is comprised primarily of basal amphibolite gneiss overlain by pelitic schists. The stratigraphy of the Site, from the surface down, consists of orangebrown/brown coarse to fine sand and silt fill, some coarse to fine gravel with brick and traces of organics ranging in depth from about two 2 to 16 feet below street grade (ftbsg). Decomposed rock was encountered at depths ranging from about 14 to 38 feet bsg. Bedrock was encountered between 20 and 38 ft-bsg. Groundwater was observed at approximately 7.5 to 12.5 ft-bsg in August 2019.

The Site is improved with four structures including: two churches, retail stores, and professional offices. Based on review of historic Sanborn Fire Insurance maps, the Site has been developed since 1887 and its uses have included: a meat market and sausage shop and a "Chinese Laundry" (dry cleaner) from at least 1887 until 1896 (however, it is possible this dry cleaner stayed on-Site longer due to a gap in Sanborn map coverage). The Site is then identified with uses such as the Huguenot Lodge, a Jewelry facility, and

later the Fire Department Headquarters in 1903. A gas engine is identified on Site from 1887 to 1911, with a 100-gallon gasoline underground storage tank (UST) being identified in 1911. The Site is then depicted as retail stores, an American Legion Post, Fire Department Headquarters, and auto storage in 1942. A fur storage is depicted in 1951. A large building appears at 500 Main Street with smaller structures in the surrounding area, showing similar footprints to the present-day structures, in the 1990 Sanborn map. Based on a review of the City of New Rochelle building department records, the building at 500 Main Street was converted to a skating rink in 1979 and then to retail space in 1987.

This Remedial Action Work Plan (RAWP) includes an analysis of the remedial alternatives available to remediate the nature and extent of contamination as determined from data gathered during the RI, performed from July 27 to August 10, 2020, and then selects a preferred remedy.

#### Summary of the Remedial Investigation

The Remedial Investigation Report (RIR) summarizes the results of prior investigations and the RI performed on the Site. The RI was conducted in accordance with the Remedial Investigation Work Plan (RIWP) for the Site, which was last revised on July 9, 2020 and subsequently approved by the NYSDEC on July 14, 2020. The RI was conducted in accordance with the NYSDEC's Technical Guidance for Site Investigation and Remediation (DER-10).

Twenty-three (23) soil borings, eight (8) permanent groundwater monitoring wells and four (4) soil vapor points were installed and sampled at the Site as part of the RI.

The soil and groundwater samples that were collected were analyzed for target compound list/target analyte list (TCL/TAL+30), including metals (USEPA Methods 6010/7471), semi-volatile organic compounds (SVOCs) (USEPA Method 8270), volatile organic compounds (VOCs) (USEPA Method 8260), PCBs, pesticides (USEPA Methods 8081/8082), Per and Polyfluoroalkyl Substances (PFAS) (USEPA Modified Method 537) and 1,4 dioxane (USEPA Method 8270). Field Blanks were collected each day groundwater sampling was conducted, and trip blanks accompanied all groundwater samples analyzed for VOCs. The soil vapor samples were analyzed for VOCs in

accordance with USEPA Method TO-15.

#### <u>Soil</u>

The results of the RI sampling showed that Poly Aromatic Hydrocarbons (PAHs) impacts exceeding both Unrestricted Use Soil Cleanup Objectives (USCOs) and Restricted Residential Soil Cleanup Objectives (RRSCOs) were identified as deep as 6 ft bsg at one location on the Site, and at 15 ft bsg (2 feet below the basement) at another location, with basement grade approximately 13 feet below street grade. Metals contaminated soils exceeding the RRSCOs were identified in soils down to 19 ft bsg (6 ft beneath the basement), and exceeding the USCOs at a maximum depth of 22 ft bsg in one location (9 ft beneath the basement). Pesticides impacted soils exceeding USCOs were identified in soils down to 16 ft bsg at the Site (3 ft beneath the basement). There were no VOCs or PCBs identified in any of the soil samples.

#### **Groundwater**

The Site groundwater flow direction is to the southeast. The groundwater table ranges from 7.5 to 12.5 ft bsg.

Groundwater investigations during the RI detected contaminant concentrations that exceeded the NYSDEC Technical Operation and Guidance Series 1.1.1 Class GA groundwater Ambient Water Quality Standards (AWQS). Two VOCs were detected in two wells and several SVOCs were detected in three wells at concentrations in excess of the AWQS. There were no metal exceedances of the AWQS with the exception of several naturally occurring metals including aluminum, iron, magnesium, manganese and sodium. No pesticides were detected above the AWQS, and PCBs were not detected in any of the groundwater samples.

Emerging contaminants, which do not yet have established AWQS, were also detected in Site groundwater, including perfluorooctanoic acid (PFOA) and perfluorooctanesulfonic acid (PFOS) exceeding the groundwater action screening levels listed in the NYSDEC Sampling Analysis and Assessment of Per- and Polyfluoralkyl Substances (October 2020). 1,4-dioxane was not detected in any wells.

# Soil Vapor

The results of the RI showed Site soil vapor contains trichloroethene (TCE) at one location. TCE was not detected in groundwater or soil samples. Petroleum Hydrocarbon (PHC) VOCs were detected in soil vapor within the western portion of the Site, but not detected in soil and groundwater samples.

# **Summary of Selected Remedial Actions**

The planned remedy for the Site is to meet Track 1 unrestricted soil cleanup objectives (USCOs) throughout the Site with no engineering or institutional controls.

The remedial actions selected for the Site include the following:

- Site preparation activities including building demolition, lead paint and asbestos abatement and historic foundation removal,
- Installation of a support of excavation (SOE) system
- Conduct two rounds of groundwater samples from the existing wells prior to demolition to confirm the RIR results.
- Excavation of all Site soils exceeding the USCOs and therefore achieving Track 1 for soils for the entire Site,
- Removal and off-Site discharge of contaminated groundwater encountered during dewatering and construction through a filtered carbon dewatering system,
- Installation of a soil vapor barrier/ waterproofing membrane sealing layer as a remedial element,
- Long term monitoring of the Site groundwater if the two additional sampling rounds exhibit contamination that exceeds the AWQS,
- If a Track 1 cleanup cannot be achieved for all or portions of the Site, preparation
  of a Site Management Plan, for post-remediation management of any residual
  contamination as required by the Environmental Easement, particularly as they
  pertain to future phases of construction, including plans for: (1) Institutional and
  Engineering Controls, (2) soil vapor monitoring, and (3) reporting.

# **1.0 INTRODUCTION**

The New York State Department of Environmental Conservation (NYSDEC) has entered into a Brownfield Cleanup Agreement (BCA) with BRP 500 Main LLC (the Volunteer) on June 3, 2020 concerning the property located at 500, 506, and 510 Main Street, and 12 Church Street in New Rochelle, Westchester County, New York (the "Site"), designated 500 Main Street Laundry BCP Site No. C360199. The Site consists of a 0.79-acre property located at 500, 506, and 510 Main Street in New Rochelle, Westchester County, New York. The Site is bounded to the northwest by Main Street and to the southwest by Church Street and is surrounded by commercial properties to the northeast and southeast. **Figure 1.1** presents a Site Location Map. **Figure 1.2** presents a Site Plan.

SESI Consulting Engineers (SESI) has prepared this Remedial Action Work Plan (RAWP) on behalf of the Volunteer. This RAWP includes an analysis of the remedial alternatives available to remediate the nature and extent of contamination as determined from data gathered during the Remedial Investigation (RI), and then selects a preferred remedy. The field work for the RI was performed from July 27 to August 10, 2020.

The RI was completed in accordance with the NYSDEC DER-10, to provide a systematic assessment of environmental conditions on the Site. An RIR was submitted to NYSDEC on September 30, 2020. The RI defines the nature and extent of contamination on-Site, identifies contaminant source areas, migration pathways and producing data of sufficient quantity and quality to complete an on-site exposure assessment and a qualitative off-site exposure assessment for purposes of designing the remedial action for the Site.

The RI for this Site did not identify fish and wildlife resources. Per DER-10 Appendix 3C, no fish and wildlife impact analysis are needed since there are no fish or wildlife resources on or near the Site.

# **1.1 SITE LOCATION AND DESCRIPTION**

The Site consists of a 0.79-acre property located at 500, 506, and 510 Main Street and 12 Church Street in New Rochelle, Westchester County, New York. Four (4) separate buildings comprise the Site located at each of the four addresses. The buildings at 500 Main Street and 12 Church Street are occupied by churches, the building at 506 Main Street is currently vacant, and the building at 510 Main Street contains a uniform store with office space above. The Site is bounded to the northwest by Main Street and to the southwest by Church Street and is surrounded by commercial properties to the northeast and southeast.

# **1.2 PROPOSED REDEVELOPMENT PLAN**

The Remedial Action to be performed under the RAWP is intended to make the Site protective of human health and the environment. The planned redevelopment of the Site entails the construction of a new residential apartment building, with the following levels of uses subsurface and on the first four levels:

Subsurface Fully Enclosed Cellar with Mechanic Room and Parking

Fully Enclosed 1<sup>st</sup> Floor: Church, residential lobby, retail and parking with garage door to enter the parking area from the street as the only opening

Fully Enclosed 2<sup>nd</sup> Floor: Church, school and residential amenities

Open Air 3<sup>rd</sup> Floor: parking

Open Air 4<sup>th</sup> Floor: parking and residential amenities

Fully Enclosed 5th Floor and up: apartments

All existing on-Site structures will be demolished. The proposed re-development plan is included as **Appendix A**.

# 2.0 DESCRIPTION OF REMEDIAL INVESTIGATION CONCLUSIONS

The Site was investigated in accordance with the scope of work presented in the NYSDEC-approved RIWP. The investigation was conducted from July 27 to August 10, 2020. Previous environmental reports were included as an appendix to the RIWP. The RI report (RIR) was submitted to NYSDEC and NYSDOH on September 30, 2020.

# 2.1 SOIL

The following conclusions were made based on the soil results (summarized in Section 1.0 and illustrated on **Figure 2.1A, B, C, and D**:

- PAHs were identified in three soil samples from two soil borings at concentrations that exceed their RRSCOs. The exceedances were identified as deep as 6 ft bsg at one location on the Site, and at 15 ft bsg (2 feet below the basement) at another location in the southern portion of the Site near the firehouse and a former underground storage tank (UST).
- Metals were identified in seven (7) soil samples from six (6) soil borings at concentrations exceeding their USCO and/or RRSCO. The depth of metals impacts extended to a maximum depth of 22 ft-bsg (9 ft below the basement) in the southeast corner of the Site exceeding their USCOs. Levels of mercury and arsenic were detected above the RRSCO in the area of the former firehouse and at the southern boundary of the Site, respectively. The depth of these exceedances ranged from 3 ft-bsg to 19 ft-bsg (6 ft beneath the basement).
- The pesticide 4,4-DDT was detected in three samples from two soil borings at concentrations exceeding its USCO of 0.0033 mg/kg, but below its RRSCO of 7.9 mg/kg. The pesticide exceedances were located in the former firehouse area and extended to a depth of 13 ft-bsg (3 ft beneath the 12 Church basement).
- No VOCs or PCBs were identified exceeding the USCO or the RRSCO.
- PFOA and PFOS were not detected in excess of NYSDEC screening levels.

# **2.2 GROUNDWATER**

Based on the groundwater results (summarized in Section 1.0 and illustrated on **Figure 2.2** and **Figure 2.2A**) it was concluded that:

- Cis-1,2-dichloroethene and hexachlorobenzene were the only VOCs detected above the AWQS. Cis-1,2-dichloroethene was detected in well MW-7 located in the southeast corner of the Site, and hexachlorobenzene was detected in MW-3 in the northwest corner of the Site.
- In wells MW-2, MW-3 and MW-4, several SVOCs were detected at concentrations exceeding their AWQS.
- There were no metal exceedances of the AWQS with the exception of several naturally occurring metals including aluminum, iron, magnesium, manganese and sodium.
- PFOA and PFOS were detected in all onsite wells exceeding the groundwater action level trigger listed in the NYSDEC Sampling Analysis and Assessment of Per- and Polyfluoralkyl Substances (October 2020). No AWQS is currently established for these compounds.

# 2.3 SOIL VAPOR

The following conclusions were made based on the soil vapor results, which are illustrated on **Figure 2.3**:

- TCE was detected in vapor point SVS-7, which is located in the northwestern portion of the Site near the former dry cleaners. There were no exceedances of TCE detected in any of the groundwater or soil samples.
- Other VOCs were detected in soil vapor within the western area of the Site but not detected in soil or and groundwater samples.

# 2.4 GEOLOGICAL CONDITIONS

According to the 1970 Geologic Map of New York – Lower Hudson Sheet published by the University of the State of New York, the bedrock underlying the Site is the Hartland Formation and is comprised primarily of basal amphibolite gneiss overlain by pelitic schists. The stratigraphy of the Site as described in SESI's Geotechnical Report from September 2019, from the surface down, consists of orange-brown/brown coarse to fine sand and silt fill, some coarse to fine gravel with brick and traces of organics ranging in depth from about two 2 to 16 ft-bsg. Decomposed rock was encountered at depths ranging from about 14 to 38 ft-bsg. Bedrock was encountered between 20 and 38 ft-bsg. Groundwater was observed at approximately 7.5 to 12.5 ft-bsg in August 2019.

The groundwater flow direction was determined to be toward the southeast across the Site as shown on the groundwater contour map (**Figure 2.4**).

#### 2.5 CONCEPTUAL SITE MODEL OF CONTAMINATION TRANSPORT

The overall depth of impacted soils ranged from 1 to 22 ft-bsg. PAH impacts were exceeding both USCO and RRSCOs were identified in shallow soils 1 to 6 ft-bsg, and at another location at 15 ft bsg (2 feet below the basement). Metals contaminated soils exceeding the USCOs and the RRSCOs were identified in shallow soils approximately 14-19 ft-bsg(1-6 ft beneath the basement). Metals contaminated soils exceeding the USCO extend down to 22 ft-bsg (9 ft beneath the basement) in one location. Pesticides impacted soils exceeding USCOs were identified in shallow soils in the area of borings SB-14 to a depth of 6 ft-bsg and SB-18 to a depth of approximately 16 ft-bsg (3 ft beneath the basement).

The Site's groundwater has been impacted with VOCs and SVOCs compounds above NYSDEC TOGS AQWS groundwater standards in the northwest portion of the Site as a result of the historical land uses. Exceedances of several metals are naturally occurring and not the result of historical land uses.

The Site groundwater flow direction is to the east-southeast. The groundwater table was measured at 10.5 to 12.5 ft bsg during the RI. The range in groundwater depth is a result of the time of the year and the method (e.g. boring or permanent well) during which the groundwater depth was reported.

The pathway of the contaminated groundwater to human receptors is limited to the ingestion of the groundwater or direct exposure through excavation work. However, groundwater in this area of New Rochelle is not used for drinking. In addition, the impacted Site groundwater is not likely to have an ecological pathway.

TCE was detected in soil vapor in vapor point SVS-7, but there were no exceedances of TCE detected in any of the groundwater or soil samples. Other VOCs were detected within the western area of the Site, but were not detected in soil or groundwater samples. The soil vapor intrusion pathway is not considered a major concern for this Site based on the limited low-level detections during the RI, lack of any corresponding detections in groundwater, and the fact that soils will be removed as part of site development. A ventilated subgrade enclosed parking level is planned that should bring some fresh air into the lower level garage to dissipate any potential vapors. A church and school are planned on the first and second floors respectively as future uses. A soil vapor barrier/ waterproofing membrane sealing layer will be installed as a remedial element to prevent any vapor or moisture from entering the slab and will act as a permanent groundwater infiltration control. The soil vapor intrusion evaluation includes the installation of the sub-slab barrier and the sub-grade vented parking area. Since the barrier will prevent any vapor or moisture from entering the slab, sub-slab vapor and indoor air sampling may not be necessary or applicable. The upper levels of the building from floor five and up will not need vapor mitigation because levels three and four will be an open air garage.

# 2.6 IDENTIFICATION OF STANDARDS, CRITERIA AND GUIDANCE

The following standards and criteria typically will apply to Site Characterizations, Remedial Investigations, remedy selection, remedial actions and Site management activities:

- DER-10 / Technical Guidance for Site Investigation and Remediation
- DER-13 / Strategy for Evaluating Soil Vapor Intrusion at Remedial Sites in New York New York State Department of Environmental Conservation
- 6 NYCRR Part 257 Air Quality Standards
- 29 CFR Part 1910.120 Hazardous Waste Operations and Emergency Response
- TOGS 1.1.1 Ambient Water Quality Standards & Guidance Values and Groundwater Effluent Limitations

- Fish and Wildlife Impact Analysis for Inactive Hazardous Waste Sites (October 1994)
- NYSDOH Guidance for Evaluating Soil Vapor Intrusion in the State of New York (October 2006)
- DER Interim Strategy for Groundwater Remediation at Contaminated Sites in New York State
- 6 NYCRR Part 375 Regulations Subparts 1, 3 and 6 applicable to the Brownfield Cleanup Program
- Citizen Participation in New York's Hazardous Waste Site Remediation Program: A Guidebook (June 1998)
- USEPA Office of Solid Waste and Emergency Response Directive 9355.047FS Presumptive Remedies: Policy and Procedures (September 1993)
- USEPA Office of Solid Waste and Emergency Response Directive 9355.048FS Presumptive Remedies
- Site Characterization and Technology Selection for CERCLA sites with Volatile Organic Compounds in Soils (September 1993)
- 6 NYCRR Part 612 Registration of Petroleum Storage Facilities (February 1992)
- 6 NYCRR Part 613 Handling and Storage of Petroleum (February 1992)
- 6 NYCRR Part 614 Standards for New and Substantially Modified Petroleum Storage Tanks (February 1992)
- 6 NYCRR Part 371 Identification and Listing of Hazardous Wastes (November 1998)
- 6 NYCRR Subpart 374-2 Standards for the Management of Used Oil (November 1998)
- 6 NYCRR 375 Table 375-6.8(a) and Table 375-6.8(b)
- 6 NYCRR Parts 700-706 Water Quality Standards (June 1998)
- 40 CFR Part 280 Technical Standards and Corrective Action Requirements for Owners and Operators of Underground Storage Tanks
- CP-51 Soil Cleanup Guidance

- STARS #2 Biocell and Biopile Designs for Small-Scale Petroleum-Contaminated Soil Projects
- SPOTS #14 Site Assessments at Bulk Storage Facilities (August 1994)
- Spill Response Guidance Manual
- Permanent Closure of Petroleum Storage Tanks (July 1988)
- NYSDOH Environmental Health Manual CSFP-530 "Individual Water Supplies -Activated Carbon Treatment Systems"
- 40 CFR Part 144 Underground Injection Control Program
- 10 NYCRR Part 67 Lead
- 12 NYCRR Part 56 Industrial Code Rule 56 (Asbestos)
- 6 NYCRR Part 175 Special Licenses and Permits--Definitions and Uniform Procedures
- 6 NYCRR Part 371 Identification and Listing of Hazardous Wastes (November 1998)
- 6 NYCRR Part 372 Hazardous Waste Manifest System and Related Standards for Generators, Transporters and Facilities (November 1998)
- 6 NYCRR Subpart 374-1 Standards for the Management of Specific Hazardous Wastes and Specific Types of Hazardous Waste Management Facilities (November 1998)
- 6 NYCRR Subpart 374-3 Standards for Universal Waste (November 1998)
- 6 NYCRR Part 608 Use and Protection of Waters
- TAGM 4013 Emergency Hazardous Waste Drum Removal/ Surficial Cleanup Procedures (March 1996)
- TAGM 4059 Making Changes to Selected Remedies (May 1998)
- Groundwater Effluent Limitations
- TOGS 1.3.8 New Discharges to Publicly Owned Treatment Works
- TOGS 2.1.2 Underground Injection/Recirculation (UIR) at Groundwater Remediation Sites

- OSWER Directive 9200.4-17 Use of Monitored Natural Attenuation at Superfund, RCRA Corrective Action, and Underground Storage Tank Sites (November 1997)
- Groundwater Monitoring Well Decommissioning Procedures (May 1995)
- Sampling, Analysis and Assessment of Per and Polyfluorinated Alkyl Substances (PFAS) under NYSDEC's Part 375 Remedial Programs (October 2020)
- The activity is a component of a program selected by a process complying with the public participation requirements of section 1.10, to the extent applicable.

# 2.7 ENVIRONMENTAL AND PUBLIC HEALTH ASSESSMENTS

#### 2.7.1 Qualitative Human Health Exposure Assessment

There are potential exposure pathways related to the contamination if left unremediated:

The exposure pathway of the contaminated groundwater to human receptors is limited to the direct ingestion of the groundwater or direct exposure through excavation work. As a result, only on-Site construction workers could be exposed. Groundwater was found to contain VOCs including cis-1,2-dichloroethene and hexachlorobenzene and SVOCs (PAHs) at levels that exceed the AWQS. However, groundwater in this area in New Rochelle is not used for drinking water and once the Site is redeveloped, excavation to the depths at which groundwater is present (between 10.5 to 12.5 ft bsg) is unlikely. As a result, groundwater is not anticipated to be a pathway for human heath exposure through ingestion or direct exposure. However, groundwater causing soil vapor exceedances represents an indirect exposure pathway, and will be address through precautionary soil vapor mitigation measures described below.

PAHs, metals and pesticides exceedances of the USCOs and the RRSCOs in the Site soil, which are present in contaminated fill soils, pose a risk to human health. The exposure pathway to humans can be through direct dermal contact with the contaminated soils or incidental ingestion. However, the soil source will be removed through Site remediation activities, which call for the excavation of all on-Site soils above the USCOs. Exposure of on-Site workers during the soil excavation work will once again be prevented

through the implementation of the HASP, CAMP, implementation of dust and vapor suppression measures, direct disposal into trucks to avoid on-Site storage, and proper soil management at the Site.

In Site soil vapor, TCE was found at one location (SVS-7), which may indicate the potential for a vapor intrusion risk depending on the corresponding concentration in indoor air. However, since TCE was not detected in Site groundwater or soil, soil will be removed through Site remediation activities. Moreover, a soil vapor barrier/waterproofing membrane sealing layer as a remedial element will be installed and the sub grade is a vented parking area. The barrier will prevent any vapor or moisture from entering the slab. The vapor barrier/waterproofing membrane and the vented garage will protect the planned presence of a church and school on the first and second floors of the proposed building.

#### 2.7.2 Fish and Wildlife Impact Analysis

The Site does not contain any ecologically sensitive resources and hence the contaminated soils are not expected to have any impacts on any ecological resources.

# 2.8 SIGNIFICANT THREAT

The NYSDEC and NYSDOH have evaluated the RIR to discern if this Site poses a significant threat to human health and the environment. The agencies have determined that the Site does not pose a significant threat to human health and the environment.

# 2.9 REMEDIAL ACTION OBJECTIVES

Based on the results of the Remedial Investigation, the following Remedial Action Objectives (RAOs) have been identified for this Site.

#### 2.9.1 Groundwater

RAOs for Public Health Protection

 Prevent ingestion of groundwater containing contaminant levels exceeding drinking water standards.  Prevent contact with, or inhalation of, volatiles emanating from contaminated groundwater.

RAOs for Environmental Protection

- Restore ground water aquifer, to the extent practicable, to pre-disposal/pre-release conditions.
- Remove the source of ground water contamination.

# 2.9.2 Soil

RAOs for Public Health Protection

- Prevent ingestion/direct contact with contaminated soil.
- Prevent inhalation of, or exposure to, contaminants volatilizing from contaminated soil.

RAOs for Environmental Protection

- Prevent migration of contaminants that would result in groundwater contamination.
- Prevent impacts to biota due to ingestion/direct contact with contaminated soil that would cause toxicity or bioaccumulation through the terrestrial food chain.

# 2.9.3 Soil Vapor RAOs

RAOs for Public Health Protection

• Mitigate impacts to public health resulting from potential present and future soil vapor intrusion into buildings at the Site.

# **3.0 DESCRIPTION OF REMEDIAL ACTION PLAN**

#### **3.1 EVALUATION OF REMEDIAL ALTERNATIVES**

The objective of the remedy for the planned mixed use residential and commercial development, is to achieve at least a conditional Track 1 cleanup, which is most protective of human health and the environment. An unconditional Track 1 soil remedy is expected to be achieved. However, groundwater will require groundwater monitoring that is expected to be shorter than five years, and a vapor barrier will be installed under the basement concrete slab.

#### <u>Track 1</u>

A remedy pursuant to this Track requires compliance with the USCOs for soils set forth in 6 NYCRR Table 375-6.8(a) in the remaining soils on the Site after remedial excavation.

For a conditional Track 1 remedy, institutional and engineering controls are allowed only for periods of less than five years except in the limited instance where a volunteer must conduct more long term remedial groundwater activities to achieve a bulk reduction in groundwater contamination to asymptotic levels. This alternative involves the complete removal and/or remediation of the soil with exceedances to achieve the USCOs, which were encountered at up to 22 ft-bsg across the Site. A feasible remedial technology that may be used to implement this alternative involves the excavation of the contaminated soil and transportation to an approved off-site facility for disposal.

Institutional and engineering controls (ICs and ECs) may be implemented to address contamination in groundwater. Monitored natural attenuation (MNA), which consists of periodic monitoring of the contaminant levels in the Site groundwater monitoring wells, will constitute an engineering control requirement documented in an institutional control environmental easement until the groundwater levels are below the standards or until they reach asymptotic levels that are accepted by the NYSDEC. Given the low levels of groundwater exceedances, it is expected that the groundwater will meet the AWQS or reach asymptotic levels that are accepted by the NYSDEC and NYSDOH in less than

five years after the implementation of excavation, which should result in the removal of contamination sources. However, if the groundwater levels do not stabilize, active groundwater remediation may be required, which will consist of chemical injection remediation to treat the remaining detected organic compounds in groundwater.

Precautionary engineering controls will address potential remaining soil vapor at the Site. The RI results indicated only one (1) low-level detection of TCE across the Site, and no TCE detected in groundwater or soil. In addition, the planned Site development includes removal of all soils to at least 17.5 ft bsg, and a subgrade parking level that will be ventilated with fresh air to dissipate most potential soil vapor intrusion. However, since a church and school will be present directly above this basement level in enclosed spaces, precautionary vapor mitigation controls under the basement slab will be installed. A soil vapor barrier/ waterproofing membrane sealing layer as a remedial element will be installed. The soil vapor intrusion evaluation includes the installation of the sub-slab barrier and the sub grade vented parking area. The barrier will prevent any vapor or moisture from entering the slab.

#### Track 2

A Track 2 remedy consists of achievement of the application restricted use soil cleanup objectives, which for this Site would be the Restricted Residential SCOs (RRSCOs). This Track requires the Volunteer to implement at least a soil cleanup that achieves the lower of the RRSCOs, or the protection of groundwater water SCOs from the tables in 6 NYCRR 375-6.8(b) within the top 15 feet of soil (or bedrock if less than 15 feet). Under a Track 2 remedy, the remedial program may include the use of long-term institutional or engineering controls to address residual contamination related to other media including, but not limited to groundwater and soil vapor. The Site remediation pursuant to Track 2 would still involve excavation and disposal of the contaminated soils to 19 feet bgs to meet the RRSCOs.

Because the soils on the Site will be excavated down to around 17.5 ft bsg in addition to the removal of any deeper exceedances of the USCO, it is anticipated that Track 1 SCOs will be achieved for soil. However, if this is not possible, a Track 2 remedy will be a contingent option for the Site soils.

The same ICs and ECs for Site soil groundwater (i.e. MNA with groundwater monitoring for five years or less) and post-construction soil vapor evaluation will be implemented for a Track 2 remedy in the event that only Track 2 can be achieved on all or portions of the Site. A Site Management Plan (SMP) and Environmental Easement (EE) as institutional controls will be temporarily put in place to ensure that all of the institutional and engineering controls are maintained until no longer required by NYSDEC and NYSDOH.

#### Track 4

A Track 4 remedy for a restricted residential use does not need to meet specific soil cleanup objectives but requires source removal and typically a Site-wide cover system where, as here, there is Site-wide surficial contamination. The Track 4 cleanup does not have specific soil cleanup objectives, but rather the Applicant may solely or in combination use the SCOs in subpart 375-6.8, develop or modify Site specific SCOs, or propose Site specific SCOs which are protective of public health and the environment.

Short and long-term IC and ECs are allowed to achieve protection of public health and the environment. A remedy under a Track 4 alternative for this Site would include a cover system over exposed residual soil contamination left in place. Soils, which are not otherwise covered by structures such as buildings, sidewalks or pavement must be covered with at least 18 inches of soil, which complies with the unrestricted or restricted residential SCOs, and clean Track 1 unrestricted topsoil in the top 6 inches.

Track 4 also includes a Site Management Plan (SMP) and Environmental Easement (EE) as institutional controls to ensure that all of the institutional and engineering controls, including the cover system, are maintained, and any soil removed from the Site post remedial action is managed properly. The SMP will include periodic (annual) monitoring and reporting in relation to the integrity of the cover system to ensure continued protection of the human health and the environment.

#### **No Action Alternative**

The no action alternative would leave existing sources of contamination in soil, groundwater and soil vapor, which would cause potential exposure to anyone present on

the Site. The no action alternative is thus unacceptable and has not been compared to the factors below.

#### Protection of human health and the environment:

Although all Tracks would provide adequate protection of human health and the environment, Track 1 is more protective than the other cleanup tracks because it would remove all soil contamination. Moreover, because a Track 1 remedy requires no long term ongoing institutional or engineering controls to manage contamination indefinitely into the future, the cleanup does not rely on human intervention or mechanical equipment to remain effective in protecting human health and the environment. A Track 2 remedy would also be protective of human health and the environment if the proper long-term engineering and institutional controls are put in place and managed in an SMP. A Track 4 remedy would also be protective of human health and the environment if the proper long-term long-term engineering and institutional controls are put in place and managed in an SMP. However, groundwater may remain contaminated for a longer time.

#### Compliance with standards, criteria, and guidelines (SCGs):

All cleanup Tracks will achieve applicable cleanup standards. However, a Track 1 cleanup achieves a more stringent set of standards than a Track 2 or 4 cleanup.

#### Short-term effectiveness and impacts:

Generally, Track 1 provides the best short-term effectiveness because it promptly removes the most contaminant mass from the Site. Track 2 also accomplishes this, but to a lesser extent. Tracks 1 and 2 are somewhat less favorable in terms of short-term impacts primarily because mass removal of the contaminated soils generates more trucks trips than a Track 4 limited removal remedy. A Track 4 approach also reduces the risk of construction worker exposure by reducing the volume of contaminated soil being managed and has less potential to cause dust and traffic issues. Excavation may result in a greater potential for migration of impacts from the open excavation (e.g. wind erosion, storm water intrusion, etc.). However, best management practices in relation to soil handling, the community air monitoring program (CAMP), and erosion and sediment

controls, and dust control measures will be implemented to minimize and control any migration of dust on-Site and off-Site.

#### Long-term effectiveness and Performance:

Because Tracks 1 & 2 would involve removal of the greatest amount of contaminated soil, which may be impacting grondwater and soil vapor, these remedial alternatives will provide the most long-term effectiveness. As already discussed above, a Track 1 cleanup will allow the Site to be used for any purpose without restriction and without reliance on the long-term employment of ICs or ECs (which can fail and require on-going monitoring and maintenance to remain effective over the long-term). A restricted residential Track 2 clean-up allows the Site to be used for almost all possible uses in an urban setting, but may have longer-term ECs and ICs.

The long-term effectiveness of the Track 4 clean-up will be ensured with adherence to the SMP and recording of an Environmental Easement. Although contaminants are left in on-Site soils, a properly maintained cover system would be effective at eliminating the risk of dermal exposure.

#### Reduction of toxicity, mobility, or volume of contaminated material:

Tracks 1 through 4 will reduce of toxicity and mobility. A Track 1 or 2 would result in more reduction in the volume of contaminated soils than in a Track 4 clean-up. While Track 4 provides a relatively smaller reduction in volume than the other tracks, it relies primarily on the decrease of contaminant mobility.

#### Constructability:

Track 1 and/or Track 2 are implementable given the location and the planned use for the Site.

While there are short term potential impacts from a Track 1 or 2 remedy, the Site is located in the middle of an urban area, and, therefore, off-Site disposal of the contaminated soils in trucks will not be a problem. Moreover, these short-term impacts will be mitigated through implementation of the CAMP and HASP, which will employ truck

washing and odor and dust control measures. Therefore, Track 1 or 2 are implementable remedies for this Site.

#### Cost effectiveness:

The preferred alternative should provide optimal suitability of the eight accompanying evaluation factors with minimal remedial cost. The contaminated soil/fill layer extends from the surface to a maximum depth of 22 ft bsg. Removal of the soil exceeding the USCOs to achieve Track 1 or 2 Site-wide will be the most costly of the remedial alternatives. However, this mass removal results in long term savings by eliminating (or, in the case of Track 2, significantly reducing) the need for indefinite cap monitoring and maintenance. Therefore, a Track 1 or 2 remedy for the Site is cost effective.

The MNA groundwater monitoring remedy for groundwater would be the most cost effective compared to active remediation of groundwater for all the remedial alternatives.

#### Community Acceptance:

A community communications program has been incorporated into all remedial alternatives, per NYSDEC Brownfield Program law and regulations and the Site-specific Citizen's Participation Plan (CPP). The Site development will include an affordable housing component and is part of an area wide transit-oriented redevelopment that includes a mix of modern residences and retail stores in downtown New Rochelle. The community should accept any of the remedies, however, the Track 1 or 2 remedy is likely preferable to the community since it will reduce the most contamination and prevent off-site migration long-term.

#### Land use:

Tracks 1 and 2 would achieve remediation for the planned residential use of the Site, which is consistent with New Rochelle's proposed plans for the area. Developing the property will create short term construction impacts, but the creation of a new downtown housing project on a remediated former brownfield site will provide significant community benefits.

- <u>Zoning</u>: All of the proposed remedies under each track will facilitate the Site to be utilized for a proposed mixed commercial-residential development, which is consistent with applicable zoning laws and anticipated future use of the site.
- <u>Applicable comprehensive community master plans or land use plans:</u> Implementation of all Tracks (with institutional controls) cleanup will facilitate the proposed commercial-residential development, which is consistent with current local land use plan.
- <u>Surrounding property uses:</u> Each cleanup approach is not expected to significantly impact land use of the surrounding properties as the truck traffic and access will be on public roads. There will be short term impacts from the remediation and construction project, but these will result in long-term benefits of converting a defunct, mostly abandoned and contaminated property into new affordable housing and commercial uses.
- <u>Citizen Participation:</u> Citizen Participation during implementation of a remedial program will proceed in accordance with the Citizen Participation Plan included as **Appendix F** of this RAWP, and as noted above, will have minimal community impact. Any short-term impacts will be addressed by the CAMP and HASP.
- <u>Environmental justice concerns:</u> There are no known environmental justice concerns associated with this project.
- Land use designations: A Track 1 remedy will not restrict any current or future land use designations. A restricted residential Track 2 will have very minimal restrictions on the future land use of the property. A Track 2 will have restrictions that will be managed in the SMP. A Track 4 remedy will have additional restrictions, which makes long term site management more challenging.
- <u>Population growth patterns:</u> Any of the proposed remedies will not impact reasonably anticipated population growth patterns in the area other than to better accommodate growth by providing for new downtown, transit-oriented housing.
- <u>Accessibility to existing infrastructure:</u> Access to existing infrastructure is present in the surrounding area, and there is access to mass transit via the Metro North train

station 0.3 miles away. Subsurface infrastructure will be removed during the remedy as part of site preparation activities. However, new infrastructure will be installed subsequent to the remediation as part of the redevelopment.

- <u>Proximity to natural resources:</u> The closest surface water body, the Echo Bay, is located approximately 0.5 miles east of the Site. Storm water drainage patterns are generally consistent with the surrounding topography and primarily flow to the east towards Echo Bay.
- <u>Off-Site groundwater impacts:</u> Based on the RI findings, off-Site groundwater impacts from historical site operations appear to be limited to one VOC (cis-1,2-dichloroethene) and possibly PFOS and PFOA (in excess of background concentrations that exceed guidance values). Measures to prevent any off-site groundwater impacts are proposed in this work plan.

Geography and geology of the Site: See Section 2.3 above.

<u>Current Institutional Controls:</u> There are no current institutional controls associated with the Site. An institutional control may be required to address the long-term management of impacted groundwater remaining at the Site following remediation.

# **3.2 SELECTION OF THE PREFERRED REMEDY**

The remedial alternatives analysis determined that a Track 1 (if achievable) or Track 2 remedy will be the goal for all or portions of the Site.

# 3.3 SUMMARY OF SELECTED REMEDIAL ACTIONS

A summary of the selected Track 1 or contingent Track 2 remedial actions to address the impacts identified are discussed below:

 Demolition of the two buildings currently present on 500 Main Street and 12 Church Strreet is required for the soil removal component of the remedy

- Installation of support of excavation (SOE) system along the side walls of the excavations for structure stability of the remedial excavation pit and to prevent off-Site migration and impacts to off-site structures;
- 3. Excavation to achieve a Track 1 cleanup by removing the remaining contaminated soil/fill source to USCOs if possible, or to achieve the Track 2 RRSCOs if Track 1 USCOs cannot be achieved. If no more contaminated soil and fill are left after the remedial excavation, which is documented by endpoint confirmatory sampling, then the remedy would achieve the Track 1 for soils on the Site. If contaminated soil and/or fill is left un-excavated, which is documented by endpoint confirmatory sampling, the Confirmatory sampling, the volunteer will try to achieve the Track 2 RRSCOs based on the contaminant levels in the remaining fill at the fifteen foot bgs interval or deeper;
- 4. Removal of any inactive USTs will be conducted during the remedial action. The UST removals will be conducted as described in Section 3.4 below;
- Two groundwater sampling rounds will be conducted prior to start of excavation to further assess the detected levels of the VOCs and SVOCs detected during the RI.
- 6. The long term remedy for the groundwater will be MNA. The soil excavation will remove any possible sources on Site of the groundwater contamination. The detected exceedances of the AWQS in groundwater are expected to decrease to below the AWQS or reach asymptotic level that is accepted by the NYSDEC and NYSDOH within five years of the implementation of the remedy. If these levels are not achieved, then an active remedial injection groundwater remediation program will be designed and proposed.
- 7. Installation of soil vapor barrier/ waterproofing membrane sealing layer as a remedial element. The soil vapor intrusion evaluation includes the installation of the sub-slab barrier and the sub grade vented parking area. The barrier will prevent any vapor or moisture from entering the slab.

- Recording of an Environmental Easement for the entire Site if a conditional Track 1 or a contingent Track 2 are accomplished;
- Preparation of a Site Management Plan for long term management of residual groundwater and/or soil vapor contamination as required by the Environmental Easement, particularly as they pertain to future phases of construction, including plans for: (1) Institutional and Engineering Controls, (2) monitoring, (3) soil excavation (only if Track 1 is not achieved and a contingent Track 2 restricted residential has been acheived) and (4) reporting;
- 10. Screening for indications of contamination (by visual means, odor, and monitoring with PID) of all excavated soil during the intrusive remediation Site work;
- 11. Collection and analysis of end-point samples to evaluate the performance of the remedy with respect to attainment of the Track-specific SCOs. Sidewall end-point samples will be collected if not on the property boundary;
- 12. Documentation of all appropriate off-site disposal of all material removed from the Site in accordance with all Federal, State and local rules and regulations for handling, transport, and disposal;
- 13. All responsibilities associated with the Remedial Action, including permitting requirements and pretreatment requirements, will be addressed in accordance with all applicable Federal, State and local rules and regulations and overseen and certified by the SESI Remedial Engineer of Record described below.

# 3.4 UST CLOSURE

The UST removal consists of completing a closure effort by removing any existing inactive USTs. The location of one 5,000-gallon heating oil UST was identified during the previous Phase I investigations. The 5,000-gallon UST is reportedly located beneath the loading dock at the southwestern corner of the Site. The USTs will be removed as part of the remedial action phase at the Site.
The inactive UST(s) will be registered with the Westchester County Department of Health (WCDOH), following the submittal of a Petroleum Bulk Storage Application once their exact size and location are known. In addition, the contractor who is removing the tanks, will obtain a work permit from the WCDOH if required. Following completion of the registration the USTs will be removed as part of the remedial action for the Site.

All UST work will be conducted by a subcontractor licensed by the City of New Rochelle. Soil vapor contamination includes petroleum related compounds which may be from these USTs.

The NYSDEC Division of Environmental Remediation (DER) BCP Project Manager and the WCDOH will be provided a ten (10)-day notice prior to the start of UST removal activities. The petroleum bulk storage (PBS) modification form will be filed and provided as required by 6 NYCRR 612.2(d) subsequent to this UST closure work.

If any liquids have accumulated in the tanks, they will be pumped and disposed of accordingly. The disposal of the liquids will be manifested. The tanks will be cleaned for appropriate scrap metal disposal.

During the UST and piping removal effort, the following field observations will be made and documented:

• A description and photographic documentation of tank and pipeline conditions (e.g., pitting, holes or leak points)

- The excavation floor and sidewalls will be:
- examined for any physical evidence of soil or groundwater contamination;

- field screened with a calibrated PID at transects spaced no more than five (5) feet apart, so that sampling may be biased to the suspected location of greatest contamination.

Immediately after tank removal, if there is no evidence of a discharge in the existing UST excavations, confirmatory soil samples will be collected to demonstrate that the remaining soils meet the SCOs. If no groundwater is present in the excavation, discrete center line soil samples from the bottom of the excavation will be collected at a frequency

equal to the total length of the tank in feet divided by five (5) (minimum of one (1) sample) and one (1) sample will be added for the fraction thereof.

However, if groundwater is present in the excavation, because the USTs are anticipated to have contained gasoline, which has a density that is less than water, soil samples will be collected as follows:

- One sample biased based upon field screening results will be taken near or just above the water table from each excavation sidewall for every 30 linear feet of sidewall (minimum of one (1) sample per sidewall).
- Where seasonal fluctuations in the water table elevation can submerge and smear product over a range of several feet, additional samples will be collected in the smear zone.

If there is evidence of a discharge, excavation will continue until all contaminated soils are removed. All grossly contaminated soils as determined by field screening will be removed. Then excavation will continue until all post-excavation confirmatory samples meet the unrestricted SCOs, or until further excavation is no longer feasible.

Confirmatory post excavation soil samples will be collected to demonstrate that all the contamination has been removed as follows:

- One (1) sidewall sample will be collected for each 30 linear feet of excavation, minimum four (4) samples one on each sidewall, and
- One (1) bottom sample for every 900 square feet of excavation area minimum one sample. Based upon field screening, the samples will be biased toward the suspected location of greatest contamination.

The UST confirmatory soil samples will be sent to an ELAP-certified laboratory for analysis of VOCs and SVOCs via the NYSDEC CP-51 Soil Cleanup Guidance list. If analytical results of soil sampling identify impacts exceeding the unrestricted SCOs or CP-51 SCOs, additional excavation/removals will be conducted to the NYSDEC satisfaction in hotspot areas, and additional confirmatory soil samples will be collected.

Any contaminated groundwater will be addressed as part of the groundwater investigation and remediation. However, if groundwater is encountered in the excavation,

it will be observed for sheen or light non-aqueous phase liquid (LNAPL) and a sample may be collected from the excavation. If any LNAPL is observed, it will be excavated/removed to the NYSDEC's satisfaction.

All excavated soils will be characterized for proper disposal. The characterization samples will be collected in accordance with the disposal facility requirement.

Any groundwater or LNAPL that requires removal from the excavation will be either pumped into a Frac Tank or removed with a Vacuum Truck depending on the quantity and properly disposed of off-site.

SESI will include a tank closure report in the FER that documents the procedures for removal of the USTs in accordance with WCDOH regulations including the following:

- A discussion which details the removal of any residual liquids, purging of vapors, tank inerting, tank and piping removal procedures, tank cleaning and tank disposal;
- A discussion of end point sampling and analysis and results;
- A discussion of soil removed and disposed from the Site;
- A discussion of the excavation water handling and treatment, if applicable;

• A report section that details Westchester County's acceptance of the UST closures.

Remedial activities will be performed at the Site in accordance with this NYSDECapproved RAWP. All deviations from the RAWP will be promptly reported to NYSDEC for approval and fully explained in the FER.

# 4.0 REMEDIAL ACTION PROGRAM

# **4.1 GOVERNING DOCUMENTS**

#### 4.1.1 Site Specific Health & Safety Plan (HASP)

A copy of the SESI HASP is included as **Appendix B**. All remedial work performed under this plan will be in full compliance with governmental requirements, including Site and worker safety requirements mandated by Federal OSHA.

The Volunteer and associated parties preparing the remedial documents submitted to the State and those performing the construction work, are completely responsible for the preparation of an appropriate HASP and for the appropriate performance of work according to that plan and applicable laws.

The HASP and requirements defined in this Remedial Action Work Plan pertain to all remedial and invasive work performed at the Site until the issuance of a Certificate of Completion.

#### 4.1.2 Quality Assurance Project Plan (QAPP)

A copy of the SESI QAPP is included as **Appendix C**. All field sampling procedures and analytical methods will be implemented in accordance with this QAPP.

#### 4.1.3 Soil/Materials Management Plan (SoMP)

The SoMP is included as Section 5.4 and includes detailed plans for managing all soils/materials that are disturbed at the Site, including excavation, handling, storage, transport and disposal. It also includes all of the controls that will be applied to these efforts to assure effective, nuisance-free performance in compliance with all applicable Federal, State and local laws and regulations.

#### 4.1.4 Storm-Water Pollution Prevention Plan (SWPPP)

A SWPPP has been prepared for the Site by others and is included as **Appendix G**. A Soil Erosion and Sediment Control (SESC) Plan has been prepared for the Site and approved by the City of New Rochelle, and is included in the Site Development Plan in Appendix A.

# 4.1.5 Community Air Monitoring Plan (CAMP)

A copy of the CAMP for the Site is included as **Appendix D**.

# 4.2 GENERAL REMEDIAL CONSTRUCTION INFORMATION

# 4.2.1 Project Organization

BRP 500 Main LLC c/o BRP Companies is the BCP Volunteer and developer of the Site. SESI is the environmental consultant for all of these volunteer entities. A table summarizing the various personnel associated with the project is included as **Table 5.1** below.

Name	Company	Project Position	Address	Phone Number
Rashid A. Walker	BRP 500 Main	Volunteer	767 Third Avenue, 33 <sup>rd</sup>	(917) 620-5168
	LLC	Contact	Floor	
			New York, NY 10017	
Jesse Mausner, PG	SESI Consulting	Environmental	12A Maple Avenue	(973) 808-9050
	Engineers, P.C.	Consultant's	Pine Brook, NJ 07058	
		Project Manager		
Fuad Dahan, PE	SESI Consulting	Remedial	12A Maple Avenue	(973) 808-9050
	Engineers, P.C.	Engineer	Pine Brook, NJ 07058	
Kimberly Junkins	NYSDEC	Project Manager	NYSDEC	(845) 633-5457
			21 Putt Corners Rd.	. ,
			New Paltz, NY 12561	
Stephen Lawrence	NYSDOH	Project Manager	NYSDOH	(518) 402-0450
			Empire State Plaza,	. ,
			Corning Tower, Room 1787	
			Albany, New York 12237	

#### TABLE 5.1 – Project Personnel

# 4.2.2 Remedial Engineer

The Remedial Engineer for this project will be Fuad Dahan, PE. The Remedial Engineer is a registered professional engineer licensed by the State of New York. The Remedial Engineer will have primary direct responsibility for implementation of the

remedial program for the 500 Main Street Laundry Site. The Remedial Engineer will certify the Final Engineering Report that the remedial activities were observed by qualified environmental professionals under his supervision and that the remediation requirements set forth in this RAWP and any other relevant provisions of ECL 27-1419 have been achieved in full conformance with that Plan. Other Remedial Engineer certification requirements are listed later in this RAWP.

The Remedial Engineer will review all pre-remedial plans submitted by contractors for compliance with this RAWP and will certify compliance in the Final Engineering Report.

The Remedial Engineer will also provide the certifications listed in the Final Engineering Report.

# 4.2.3 Remedial Action Construction Schedule

A remedial action construction schedule is included as **Table 5.2** below. The schedule includes estimates of time required to complete the activities associated with the remedial action. It is based on elapsed time from receipt of NYSDEC approval. Once NYSDEC approves this RAWP, an updated schedule showing actual dates will be provided to the NYSDEC as an addendum to this plan.

Activity	Date
NYSDEC approves RAWP and issues decision document	January 2021
Start of remedial work (excavation and soil disposal, dewatering)	May 2021
Completion of Soil Excavation	July 2021
Draft final engineering report (FER), submit FER to NYSDEC	October 2021
Certificate of Completion	December 2021

TABLE 5.2 Remedial Action Schedule

#### 4.2.4 Work Hours

The hours for operation of remedial construction will conform to the City of New Rochelle Department of Buildings construction code requirements or according to specific

variances issued by that agency. NYSDEC will be notified by the Applicant of any variances issued by the Department of Buildings.

#### 4.2.5 Site Security

A description of the proposed Site security measures will be included in the Site Operations Plan. The Site is secured with fences and locked gates. Access to Site will be controlled by the local police patrolling the area.

## 4.2.6 NYSDEC BCP Signage

A project sign will be erected at the main entrance to the Site if required by NYSDEC to indicate that the project is being performed under the New York State Brownfield Cleanup Program.

## 4.2.7 Pre-Construction Meeting with NYSDEC

A pre-construction meeting will be held with NYSDEC prior to the start of major remedial construction activities.

# 4.2.8 Emergency Contact Information

An emergency contact sheet with names and phone numbers is included in **Table 5.3** below. That document will define the specific project contacts for use by NYSDEC and NYSDOH in the case of a day or night emergency.

Medical, Fire, and Police:	911
One Call Center:	(800) 272-4480 (3-day notice required for utility markout)
Poison Control Center:	(800) 222-1222
Pollution Toxic Chemical Oil Spills:	(800) 424-8802
NYSDEC Spills Hotline	(800) 457-7362

 TABLE 5.3 - EMERGENCY AND CONTACT NUMBERS

Fuad Dahan – Remedial Engineer (SESI Consulting Engineers)	(973) 808-9050
Director of Construction Rich Wooley	(212) 488-1750

\* Note: Contact numbers subject to change and will be updated as necessary

TBD – To Be Determined

# **4.3 SITE PREPARATION**

# 4.3.1 Mobilization

Mobilization tasks will include:

- Construction of temporary facilities and utilities;
- Set-up of construction equipment and facilities;
- Construction of fencing and barriers;
- Construction of erosion control measures; and
- Construction of decontamination and materials staging areas.

# 4.3.2 Erosion and Sedimentation Controls

Erosion and sediment control measures are outlined in the SESC Plan that has been prepared for the Site and approved by the City of New Rochelle, as included in the redevelopment plans in Appendix A.

# 4.3.3 Site Preparation Demolition

The existing on-Site buildings at 500 Main Street and 12 Chruch Street must be demolished for the soil remediation and the remaining buildings will be demolished in preparation to accomplish the groundwater and soil vapor remediation, including the installation of the soil vapor barrier/ waterproofing membrane required under the planned new slabs for the entire Site. A pre-demolition survey for asbestos containing material (ACM), and lead-based paint (LBP) have been completed. Abatement of the identified

ACM and LBP will be conducted. The on-Site buildings will then be demolished down to foundation. The demolition activities will be reported in the final engineering report (FER). CAMP monitoring will be conducted during the demolition activities.

## 4.3.4 Utility Marker and Easements Layout

The Volunteer and its contractors will be solely responsible for the identification of utilities that might be affected by work under the RAWP and implementation of all required, appropriate, or necessary health and safety measures during performance of work under this RAWP. The Volunteer and its contractors are solely responsible for safe execution of all invasive and other work performed under this RAWP. The Volunteer and its contractors or approvals pertinent to such work that may be required to perform work under this RAWP. Approval of this RAWP by NYSDEC does not constitute satisfaction of these requirements.

## 4.3.5 Sheeting and Shoring

A support of excavation (SOE) system consisting will be installed in the excavation areas as necessary prior to the excavation activities as part of the Site preparation.

The Volunteers and their contractors will be solely responsible for safe execution of all invasive and other work performed under this Plan and the implementation of safety measures (Sheeting and Shoring) as necessary to maintain safe working environment. The Volunteers and their contractors will obtain any local, State or Federal permits or approvals that may be required to perform work under this Plan. Further, the Volunteers and their contractors are solely responsible for the implementation of all required, appropriate, or necessary health and safety measures during performance of work under the approved Plan.

# 4.3.6 Dewatering

Dewatering may be required as part of this remedy because groundwater is present at depths as shallow as 7.5 ft bsg. All groundwater encountered will be treated as contaminated groundwater. The groundwater will be pumped into temporary storage fractanks, treated onsite via bag filters and granular activated carbon (GAC) units, and discharged to the combined or storm sewer. Appropriate discharge permits will be obtained from the City of New Rochelle (Department of Public Works) and County of Westchester.

# 4.3.7 Equipment and Material Staging

Equipment and material staging areas are expected to be relocated throughout the Site during remedial construction.

# 4.3.8 Decontamination Area

The decontamination area construction and operational requirements are provided in the HASP. Truck tires must be washed before exiting the Site.

## 4.3.9 Site Fencing

A construction safety fence is installed around the entire perimeter of the site. Access through gates will be provided at various points as required by the Volunteer and its contractors. These gates will be locked during non-construction hours.

# 4.3.10 Demobilization

Demobilization will include the following:

- Restoration of areas that may have been disturbed to accommodate support areas (e.g., staging areas, decontamination areas, storage areas, temporary water management area[s], and access area);
- Removal of temporary access areas (whether on-Site or off-Site) and restoration of disturbed access areas to pre-remediation conditions;
- Removal of sediment and erosion control measures and disposal of materials in accordance with acceptable rules and regulations;
- Equipment decontamination; and
- General refuse disposal.

# 4.4 REPORTING

#### 4.4.1 Weekly Reports

Weekly reports will be submitted to NYSDEC and NYSDOH Project Managers on Monday following the end of the week of the reporting period and will include:

- Activities relative to the Site during the previous reporting period and those anticipated for the next reporting period, including a quantitative presentation of work performed (i.e. tons of material exported and imported, etc.);
- Description of approved activity modifications, including changes of work scope and/or schedule;
- A summary of any CAMP exceedances from the previous week, including explanations of how exceedances were resolved and if any actions were taken
- Sampling results received following internal data review and validation, as applicable; and,
- An update of the remedial schedule including the percentage of project completion, unresolved delays encountered or anticipated that may affect the future schedule, and efforts made to mitigate such delays.

In the event of a CAMP exceedance, odor issue, or related complaint, the NYSDEC and NYSDOH project managers will be notified within 24 hours.

#### 4.4.2 Other Reporting

Photographs will be taken of all remedial activities and submitted to NYSDEC in digital (JPEG) format. Photos will illustrate all remedial program elements and will be of acceptable quality. Representative photos of the Site prior to any Remedial Actions will be provided. Representative photos will be provided of each contaminant source, source area and Site structures before, during and after remediation. Photos will be submitted to NYSDEC on CD or other acceptable electronic media and will be sent to NYSDEC's Project Manager (2 copies) and to NYSDOH's Project Manager (1 copy). CD's will have

a label and a general file inventory structure that separates photos into directories and sub-directories according to logical Remedial Action components. A photo log keyed to photo file ID numbers will be prepared to provide explanation for all representative photos.

Job-site record keeping for all remedial work will be appropriately documented. These records will be maintained on-site at all times during the project and be available for inspection by NYSDEC and NYSDOH staff.

# 4.4.3 Complaint Management Plan

A public information board will be constructed at the perimeter of the Site. This information board will contain the phone number of the Volunteer where complaints may be directed. General information notices to the public will also be posted on this board for their benefit. The NYSDEC and NYSDOH project managers will be notified of an odor, dust, or health-related public complaint within 24 hours of the occurrence, or sooner, depending on the severity of the complaint.

# 4.4.4 Deviations from the Remedial Action Work Plan

If there are any deviations from the RAWP, the following steps will be taken:

- Reasons for deviating from the approved RAWP will be identified and communicated directly to the NYSDEC Project Manager;
- All deviations will be communicated verbally and in writing (by letter or email) to the NYSDEC Project Manager;
- The deviations will be implemented based on verbal or written approval of the NYSDEC Project Manager. All verbal approvals will be followed-up in writing.
- The effect of the deviations on the overall remedy will be described/addressed in the Final Engineering Report.

# **5.0 REMEDIAL ACTION**

Removal of all contaminated soils under the Remedial Action for the Site will be implemented in accordance with the Soil Management Plan described below (Section 5.10) and the Site-specific (QAPP) (**Appendix C**).

A plan depicting the locations where the remedial excavation activities will be carried out is included as **Figure 3.1**. A table denoting the square footages of remedial excavation areas and estimated quantities to be excavated from each area is included on the figure.

A plan showing the location the groundwater wells is included as Figure 3.2.

# 5.1 CLEANUP OBJECTIVES

The Soil Cleanup Objectives for this Site are the Track 1 Unrestricted SCOs (USCOs) as listed in **Appendix E**.

Soil and materials management on-site will be conducted in accordance with the Soil Management Plan as described below (Section 5.9).

Groundwater cleanup objectives will be the NYSDEC TOGS Ambient Water Quality Standards (AWQS).

Soil vapor mitigation objectives will be the NYSDOH Guideline Values and Decision Matrices for the specific COCs.

# 5.2 REMEDIAL PERFORMANCE EVALUATION

#### 5.2.1 Soil Removal Verification

For all excavations, post-excavation soil samples will be collected in accordance with NYSDEC regulations. The proposed sample locations meet NYSDEC sampling frequency requirements (1 sample per 900 ft<sup>2</sup> of excavation bottom and 1 sample per 30 linear feet of excavation sidewall). Several additional post-excavation soil samples are proposed to be collected outside of the remedial excavation area, where soils will be excavated for construction. Although no exceedances of the USCOs were detected in

soil samples from borings conducted during the RI in these areas (the northern and central portion of the Site), several metals were detected exceeding USCOs in soil samples collected during the 2019 Phase II ESA (described in the RIWP). Proposed post-excavation soil sample locations are shown on **Figure 3.1**.

## 5.2.2 Groundwater Sampling

Groundwater sampling will be conducted in order to further demonstrate groundwater flow direction and to verify that no groundwater remedy is required. The proposed remedial verification sampling locations are in **Figure 3.2.** Two groundwater sampling rounds will be collected from the existing wells prior to start of excavation. The goal of the groundwater remedy is to achieve the TOGS AWQS standards or to reach asymptotic levels since it may not be possible given the surrounding urban environment in the City of New Rochelle to achieve of the TOGS AWQS standards for every contaminant, particularly naturally occurring metals. If five-year short-term MNA is determined to be a necessary component of the groundwater remedy, then the wells will be reconstructed post the soil excavation remedy in the same location and specifications as the RIR. If possible and prior to the certificate of completion (COC), a groundwater sampling round will be conducted to determine the effect of excavation on the groundwater quality of the Site.

#### 5.2.3 Soil Vapor Intrusion Evaluation

A vapor barrier/ waterproofing membrane sealing layer as a remedial element will be installed. The soil vapor intrusion evaluation includes the installation of the sub-slab barrier and the sub grade vented parking area. The barrier will prevent any vapor or moisture from entering the slab. Therefore, sub-slab vapor and indoor air sampling may not be necessary or applicable.

#### 5.3 METHODOLOGY

Soil samples will be collected in accordance with the QAPP using disposable gloves/trowels or dedicated, decontaminated stainless steel spoons. Groundwater

samples will be collected in accordance with the QAPP using the low-flow purging and sampling (LFPS) method and associated decontamination and quality control procedures.

#### **5.4 REPORTING OF RESULTS**

Groundwater and soil samples will be submitted to a NYSDOH ELAP certified laboratory. The results will be reported in accordance with NYSDEC requirements for Category B data deliverables (as outlined in DER-10).

#### 5.5 QA/QC

Collection of QA/QC samples to evaluate potential cross-contamination from sampling equipment and during shipment of samples and repeatability of laboratory analytical practices will be in accordance with the QAPP included as **Appendix C**. Field blanks, trip blanks and duplicate samples associated with daily sampling activities will be collected as a part of the QA/QC practices.

#### **5.6 DUSR**

To ensure that the field sampling and laboratory analytical practices are acceptable, the data associated with all the samples will be validated by a third party (in accordance with requirements of DER-10). The validation approach and results will be presented in a DUSR to be included in the FER.

#### 5.7 REPORTING OF PERFORMANCE MONITORING DATA IN FER

The FER will include a table of end point data with highlights or a summary of exceedances of SCOs, if any. A spider map showing all SCO exceedances will also be presented in the FER.

Chemical labs used for all end-point sample results and contingency sampling will be NYSDOH ELAP certified.

End point sampling, including bottom and side-wall sampling, will be performed in accordance with DER-10 sample frequency requirements. Side-wall samples will be collected a minimum of every 30 linear feet. Bottom samples will be collected at a rate of one for every 900 square feet. The FER will provide a tabular and map summary of all

end-point sample results and exceedances of SCOs.

## **5.8 ESTIMATED MATERIAL REMOVAL QUANTITIES**

Source removal excavation activities will be implemented during the course of the remediation activities for the Site. Based on the RI, the depth of contaminated fill and soil at the Site extends to depths of 22 ft-bsg, and the estimated quantity of contaminated soil and fill is 2,700 CY, as shown on Figure 3.1. [NOTE: The reason that such a small volume of soil requires excavation despite the depth of contamination is because there are already basements present on the Site.]

The actual excavated volume will be reported in the FER as a tally of the manifests and tickets of the soils disposed off-site.

## 5.9 SOIL/MATERIALS MANAGEMENT PLAN

Approximately 2,700 CY of material may be excavated as part of remediation, with additional material removed for construction. Any required fill will consist of imported clean fill that meets the requirements per 6 NYCRR Part 375-6.7(d) and the requirements for emerging contaminants sampling per the October 2020 PFAS Guidance Document.

#### 5.9.1 Soil Screening Methods

Visual, olfactory and PID soil field screening and assessment will be performed by a qualified environmental professional during all remedial excavations into known or potentially contaminated material. Soil screening will be performed regardless of when the invasive work is done and will include all excavation and invasive work performed during the remedy and during development phase, such as excavations for foundations and utility work, prior to issuance of the Certificate of Completion.

All primary contaminant sources identified during Site Characterization, Remedial Investigation, and Remedial Action will be surveyed by a surveyor licensed to practice in the State of New York. This information will be provided on maps in the Final Engineering Report. Screening will be performed by qualified environmental professionals. Resumes will be provided for all personnel responsible for field screening (i.e. those representing the Remedial Engineer) of invasive work for unknown contaminant sources during remediation and development work.

## 5.9.2 Stockpile Methods for Contaminated Soils

Stockpiles of contaminated materials, if needed, will be inspected at a minimum once each week and after every storm event. Results of inspections will be recorded in a logbook and maintained at the Site and available for inspection by NYSDEC.

Stockpiles will be kept covered at all times with appropriately anchored tarps. Stockpiles will be routinely inspected and damaged tarp covers will be promptly replaced.

Soil stockpiles will be encircled with silt fences. Hay bales will be used as needed near catch basins, surface waters and other discharge points.

#### 5.9.3 Materials Excavation and Load Out

The Remediation Engineer or a qualified environmental professional under his/her supervision will oversee all invasive work and the excavation and load-out of all excavated material.

The Volunteers and their contractors are solely responsible for safe execution of all invasive and other work performed under this Plan.

The presence of utilities and easements on the Site has been investigated during the remedial investigation work and a utility plan is included in the Site plan set. It has been determined that no risk or impediment to the planned work under this Remedial Action Work Plan is posed by utilities or easements on the Site.

Loaded vehicles leaving the Site will be appropriately lined, tarped, securely covered, manifested, and placarded in accordance with appropriate Federal, State, local, and NYSDOT requirements (and all other applicable transportation requirements).

A truck tracking pad for soil removal and/or tire washing associated with construction activities will be operational during construction. The Remediation Engineer will be responsible for ensuring that all outbound trucks are not causing any off-site tracking of contaminated soils.

Locations where vehicles enter or exit the Site will be inspected daily for evidence of off-Site sediment tracking.

The Remediation Engineer will ensure that all egress points for truck and equipment transported from the Site will be clean of dirt and other materials derived from the Site during Site remediation and development. Cleaning of the adjacent streets will be performed as needed to maintain a clean condition with respect to site-derived materials.

The Volunteers and associated parties preparing the remedial documents submitted to the State, and parties performing this work, are completely responsible for the safe performance of all invasive work, the structural integrity of excavations, and for structures that may be affected by excavations (such as building foundations and bridge footings).

The Remedial Engineer will ensure that Site development activities will not interfere with, or otherwise impair or compromise, remedial activities proposed in this Remedial Action Work Plan.

#### 5.9.4 Materials Transport Off-Site

All transport of materials will be performed by licensed haulers in accordance with appropriate local, State, and Federal regulations, including 6 NYCRR Part 364. Haulers will be appropriately licensed and trucks properly placarded.

Truck transport routes will be included in the SOP. All trucks loaded with Site materials will exit the vicinity of the Site using only these approved truck routes.

Proposed in-bound and out-bound truck routes to the Site will take into account: (a) limiting transport through residential areas and past sensitive sites; (b) use of city mapped truck routes; (c) prohibiting off-site queuing of trucks entering the facility; (d) limiting total distance to major highways; (e) promoting safety in access to highways; (f) overall safety in transport; and (g) community input through the Citizens Participation Plan (CPP), included in Appendix F.

Trucks will be prohibited from stopping and idling in the neighborhood outside the project Site.

Egress points for truck and equipment transport from the Site will be kept clean of dirt and other materials during Site remediation and development.

Queuing of trucks will be performed in order to minimize off-Site disturbance.

Material transported by trucks exiting the Site will be secured with tight-fitting covers. Loose-fitting canvas-type truck covers will be prohibited. If loads contain wet material capable of producing free liquid, truck liners will be used.

A tracking pad will be installed at the Site egress to ensure clean-up of the soils from the truck tires before the trucks exit the Site. If needed, truck tires will be washed. Truck wash waters will be collected and disposed of off-Site in an appropriate manner.

#### 5.9.5 Materials Disposal Off-Site

Approval from appropriate disposal facilities will be received prior to start of work. The total quantity of soil material expected to be disposed off-site is approximately 2,700 CY.

All soil/fill/solid waste excavated and removed from the Site will be treated as contaminated and regulated material and will be disposed in accordance with all local, State (including 6NYCRR Part 360) and Federal regulations. If disposal of soil/fill from this Site is proposed for unregulated disposal (i.e. clean soil removed for development purposes), a formal request with an associated plan will be made to NYSDEC's Project Manager. Unregulated off-Site management of materials from this Site will not be undertaken without formal NYSDEC approval.

Material that does not meet Track 1 unrestricted SCOs is prohibited from being taken to a New York State recycling facility (6NYCRR Part 360-16 Registration Facility).

The following documentation will be obtained and reported by the Remedial Engineer for each disposal location used in this project to fully demonstrate and document that the disposal of material derived from the Site conforms with all applicable laws: (1) a letter from the Remedial Engineer or Volunteer to the receiving facility describing the material to be disposed and requesting formal written acceptance of the material. This letter will state that material to be disposed is contaminated material generated at an environmental remediation Site in New York State. The letter will provide the project identity and the name and phone number of the Remedial Engineer. The letter will include as an attachment a summary of all chemical data for the material being transported (including Site Characterization data); and (2) a letter from all receiving facilities stating it is in receipt of the correspondence (above) and is approved to accept the material. These documents will be included in the FER.

Non-hazardous historic fill and contaminated soils taken off-Site will be handled, at minimum, as a Municipal Solid Waste per 6NYCRR Part 360-1.2

Historical fill and contaminated soils from the Site are prohibited from being disposed at Part 360-16 Registration Facilities (also known as Soil Recycling Facilities).

The Final Engineering Report will include an accounting of the destination of all material removed from the Site during this Remedial Action, including excavated soil, weathered rock, contaminated soil, historic fill, solid waste, and hazardous waste, non-regulated material, and fluids. Documentation associated with disposal of all material must also include records and approvals for receipt of the material. This information will also be presented in a tabular form in the FER.

Bill of Lading system or equivalent will be used for off-site movement of nonhazardous wastes and contaminated soils. This information will be reported in the FER.

Hazardous wastes, if any, derived from on-site will be stored, transported, and disposed of in full compliance with applicable local, State, and Federal regulations.

Appropriately licensed haulers will be used for material removed from this Site and will be in full compliance with all applicable local, State and Federal regulations.

Waste characterization will be performed for off-site disposal in a manner suitable to the receiving facility and in conformance with applicable permits. Sampling and analytical methods, sampling frequency, analytical results and QA/QC will be reported in the FER. All data available for soil/material to be disposed at a given facility must be submitted to the disposal facility with suitable explanation prior to shipment and receipt.

#### 5.9.6 Fluids Management

All liquids to be removed from the Site, including dewatering fluids, will be handled, treated, and discharged in accordance with applicable local, State, and Federal regulations. If any liquids are needed to be discharged into the sewer system, permits will be obtained from Westchester County and the New Rochelle Department of Public Works, and NYSDEC approval will be sought prior to the discharge. Dewatered fluids will not be recharged back to the land surface or subsurface of the Site without DEC approval.

Water generated during remedial construction will not be discharged to surface waters (i.e. a local pond, stream or river) without a SPDES permit.

#### 5.9.7 Demarcation Barrier

A land survey will be performed by a New York State licensed surveyor, of the Site if a Track 2 clean-up has been selected after the completion of related construction activities. The survey will define the top elevation of residual contaminated soils. This survey will constitute the written record of the upper surface of the 'Residuals Management Zone' in the Site Management Plan. A map showing the survey results will be included in the Final Engineering Report and the Site Management Plan.

#### 5.9.8 Backfill from Off-Site Sources

Backfilling is not expected. However, if necessary, material imported to be used onsite as backfill will be sampled at a frequency of, one composite sample per 500 cubic yards of material from each off-site borrow area. If more than 1,000 cubic yards of soil are needed from the same source area and both samples of the first 1,000 cubic yards meet the USCOs, the sample frequency will be reduced to one composite for every 2,500 cubic yards of additional soils from the same source, up to 5,000 cubic yards. For borrow sources greater than 5,000 cubic yards, sampling frequency may be reduced to one sample per 5,000 cubic yards, provided all earlier samples met the USCOs. The samples will be analyzed for TCL VOCs, TCL SVOCs, pesticides, PCBs, PFAS and TAL metals, including cyanide. The soil may be used as cover material provided that all parameters meet the USCOs, per the NYSDEC regulatory requirements. The imported material, if needed, will be sampled in accordance with DER-10 Section 5.4 (e) and the latest October 2020 PFAS guidance document.

All materials proposed for import onto the Site, will meet the USCOs or PFAS screening levels, will be approved by the Remedial Engineer and will be in compliance with provisions in this RAWP prior to receipt at the Site. A "Soil Reuse/Import" form will be submitted to the NYSDEC for pre-approval prior to importing any soils on -Site. Bills of Lading or equivalent documentation will be obtained to track the amount soil arriving onto the Site and verify the source of soil being imported.

Material from industrial sites, spill sites, other environmental remediation sites or other potentially contaminated sites will not be imported to the Site.

The Final Engineering Report will include the following certification by the Remedial Engineer: "I certify that all import of soils from off-Site, including source evaluation, approval and sampling, has been performed in a manner that is consistent with the methodology defined in the Remedial Action Work Plan".

All imported soils will meet NYSDEC approved backfill or cover soil quality objectives for this Site. Non-compliant soils will not be imported onto the Site without prior approval by NYSDEC. Nothing in the approved Remedial Action Work Plan or its approval by NYSDEC will be construed as an approval for this purpose.

Soils that meet 'exempt' fill requirements under 6 NYCRR Part 360, but do not meet backfill or cover soil objectives for this Site, will not be imported onto the Site without prior approval by NYSDEC. Nothing in this Remedial Action Work Plan will be construed as an approval for this purpose.

Solid waste will not be imported onto the Site.

Trucks entering the Site with imported soils will be securely covered with tight fitting covers.

# 5.9.9 Contingency Plan

If underground tanks or other previously unidentified contaminant sources are found during on-Site remedial excavation or development related construction, sampling will be performed on product, sediment and surrounding soils, etc. Procedures for removal and sampling/closure of underground tanks are included in Section 3.4. Chemical analytical work for other contaminant sources will be for full scan parameters (TAL metals; TCL volatiles and semi-volatiles, TCL pesticides and PCBs). These analyses will not be limited to STARS parameters where tanks are identified without prior approval by NYSDEC. Analyses will not be otherwise limited without NYSDEC approval.

Identification of unknown or unexpected contaminated media identified by screening during invasive Site work will be promptly communicated by phone to NYSDEC's Project Manager. These findings will be also included in daily and periodic electronic media reports.

# 5.9.10 Community Air Monitoring Plan

A copy of the CAMP for the Site is included as **Appendix D.** Exceedances observed in the CAMP will be reported to NYSDEC and NYSDOH Project Managers and included in the Daily Report.

# 5.9.11 Odor, Dust and Nuisance Control Plan

Odor, dust and nuisance control will be in accordance with the site-specific Health and Safety Plan included as **Appendix B**. The NYSDEC and NYSDOH project managers will be notified of an odor, dust, or health-related public complaint within 24 hours of the occurrence, or sooner, depending on the severity of the complaint.

The Final Engineering Report will include the following certification by the Remedial Engineer: "I certify that all invasive work during the remediation and all invasive development work were conducted in accordance with dust and odor suppression methodology defined in the Remedial Action Work Plan."

# Odor Control Plan

This odor control plan is designed to control emissions of nuisance odors off-Site. If nuisance odors are identified, work will be halted and the source of odors will be identified and corrected. Work will not resume until all nuisance odors have been abated. NYSDEC and NYSDOH will be notified of all odor events and of all other complaints about the project. Implementation of all odor controls, including the halt of work, will be the responsibility of the Applicant's Remediation Engineer, who is responsible for certifying the Final Engineering Report.

All necessary means will be employed to prevent on- and off-Site nuisances. At a minimum, procedures will include: (a) limiting the area of open excavations; (b) shrouding open excavations with tarps and other covers; and (c) using foams to cover exposed odorous soils. If odors develop and cannot be otherwise controlled, additional means to eliminate odor nuisances will include: (d) direct load-out of soils to trucks for off-Site disposal; (e) use of chemical deodorants in spray or misting systems; and, (f) use of staff to monitor odors in surrounding neighborhoods.

Where odor nuisances have developed during remedial work and cannot be corrected, or where the release of nuisance odors cannot otherwise be avoided due to on-Site conditions or close proximity to sensitive receptors, odor control will be achieved, as appropriate, by a combination of work stoppages, or sheltering the excavation and handling areas under tented containment structures equipped with appropriate air venting/filtering systems.

# Dust Control Plan

A dust suppression plan that addresses dust management during invasive on-Site work, will include, at a minimum, the items listed below:

- Dust suppression will be achieved through the use of a dedicated on-Site water truck for road wetting. The truck will be equipped with a water cannon capable of spraying water directly onto off-road areas including excavations and stockpiles.
- Clearing and grubbing will be done in stages to limit the area of exposed, unvegetated soils vulnerable to dust production.

- Gravel will be used on roadways to provide a clean and dust-free road surface.
- On-Site roads will be limited in total area to minimize the area required for water truck sprinkling.

## **Other Nuisances**

A plan for rodent control will be developed and utilized by the contractor prior to and during Site clearing and Site grubbing, and during all remedial work.

A plan will be developed and utilized by the contractor for all remedial work and will conform, at a minimum, to local noise control standards.

# **6.0 ENGINEERING CONTROLS**

#### **6.1 MONITORED NATURAL ATTENUATION**

The excavation of the soils will remove any potential sources that may have resulted in groundwater contamination. SESI has evaluated the water quality data for monitored natural attenuation (MNA) such as: dissolved oxygen (DO), oxidation reductive potential (ORP), acidity, (pH). These conditions are conducive for MNA. SESI will continue to monitor ORP, DO, pH CO<sub>2</sub>, and specific conductance, and include in the next round of sampling nitrate, ferrous iron, sulfate, methane, ethene and dissolved organic carbon (DOC) to fully assess if MNA is occurring.

## 6.1.1 Groundwater Monitoring System

A network of groundwater monitoring wells (see Figure 3.2) will be utilized to monitor the groundwater quality and demonstrate the reduction in groundwater contamination to asymptotic levels subsequent to the implementation of the Track 1 soil removal remedy. As the dissolved groundwater impacts detected during the remedial investigation are relatively low, a short-term monitoring program with associated institutional controls will be a cost-effective remedial alternative to address any residual groundwater impacts.

Specifically, the groundwater samples will be collected annually, in accordance with requirement outlined in DER-10. The groundwater samples will be analyzed for VOCs, and PAHs. MNA parameters such as dissolved oxygen (DO), oxidation reductive potential (ORP), acidity, (pH), and other parameters will be collected and evaluated to determine the MNA effectiveness on the Site.

This monitoring protocol will be described in the Site Management Plan.

# 6.1.2 Criteria for Completion of Remediation/Termination of Groundwater Monitoring

Groundwater monitoring activities to assess natural attenuation will continue, as determined by the NYSDEC, until residual groundwater contaminant concentrations are found to be consistently below NYSDEC standards or have become asymptotic at levels accepted by the NYSDEC and NYSDOH over a period of time. Monitoring will continue until permission to discontinue is granted in writing by the NYSDEC. If groundwater contaminant levels become asymptotic at a level that is not acceptable to the NYSDEC, additional treatment and/or control measures will be evaluated. These monitoring activities will be outlined in the Monitoring Plan of the SMP.

If the MNA evaluation does not result in effective destruction of groundwater contamination, in-situ treatment technologies will be designed and proposed to attain the groundwater quality standards within less than a five year period of time post Certificate of Completion.

# 7.0 INSTITUTIONAL CONTROLS

After the remedy is complete, the Site may have residual groundwater and subslab basement level vapor contamination remaining in place.

# 7.1 ENVIRONMENTAL EASEMENT

An institutional control (IC) for Monitored Natural Attenuation will be incorporated into a Site Management Plan and will be enforceable through an Environmental Easement. An Environmental Easement, as defined in Article 71 Title 36 of the Environmental Conservation Law, is required when residual contamination is left on-Site after the Remedial Action is complete. Because groundwater is expected to reach class GA AWQS or asymptotic levels within less than 5-years, this IC is allowed as part of a Track 1 remedy. In addition, as a precautionary measure, a soil vapor barrier sealing layer will be incorporated into the foundation of the proposed building to prevent any potential soil vapor intrusion into the basement and first two levels of the building.

As part of this remedy, if required, an Environmental Easement approved by NYSDEC will be filed and recorded with the Westchester County Clerk. The Environmental Easement will be submitted as part of the FER.

The Environmental Easement renders the Site a temporarily Controlled Property. The environmental Easement will be recorded with the Westchester County Clerk before the Certificate of Completion is issued by NYSDEC. Groundwater monitoring will be performed as defined in the SMP, but there are no other engineering controls anticipated.

# 7.2 SITE MANAGEMENT PLAN

Site Management is the last phase of remediation and begins with the approval of the FER and issuance of the Certificate of Completion for the Remedial Action. If an SMP is needed because of the residual groundwater contamination, it will be submitted as part of the FER but will be written in a manner that allows its removal and use as a complete and independent document. Site Management may continue in perpetuity or until released in writing by NYSDEC. The property owner is responsible to ensure that all Site Management responsibilities defined in the Environmental Easement and the Site Management Plan are performed.

The SMP is intended to provide a detailed description of the procedures required to manage residual groundwater and soil vapor contamination left in place at the Site following completion of the Remedial Action in accordance with the BCA with the NYSDEC, particularly as they pertain to the future phases of development construction proposed for the Site. This includes: (1) development, implementation, and management of all Engineering and Institutional Controls; (2) development and implementation of a Monitoring Plan; (3) submittal of Site Management Reports, performance of inspections and certification of results, and demonstration of proper communication of Site information to NYSDEC; and (4) defining criteria for termination of monitoring. The SMP for this Site, if needed, will not require the development of a plan to operate and maintain treatment, collection, containment, or recovery systems because the only proposed on-going remedy will be groundwater monitoring.

To address these needs, this SMP will include four plans as applicable: (1) an Engineering and Institutional Control Plan for implementation and management of EC/ICs; (2) a Monitoring Plan for implementation of Site Monitoring; (3) an Operation and Maintenance Plan for implementation of remedial collection, containment, treatment, and recovery systems; and (4) a Site Management Reporting Plan for submittal of data, information, recommendations, and certifications to NYSDEC. The SMP will be prepared in accordance with the requirements in NYSDEC DER-10 Technical Guidance for Site Investigation and Remediation, dated May 2010, and the guidelines provided by NYSDEC.

Site management activities, reporting, and EC/IC certification will be scheduled on a certification period basis. The certification period will be annually. The Site Management Plan will be based on a calendar year and will be due for submission to NYSDEC by March 1 of the year following the reporting period.

The SMP and the FER will include a monitoring plan for groundwater at the downgradient Site perimeter to evaluate Site-wide performance of the remedy. If the contaminants in groundwater do not attenuate below the Class GA AWQS concentrations or reach an asymptotic level that is accepted by the NYSDEC, an active treatment (e.g. oxidant injection) will be proposed and designed based on the monitoring data within the 5 year period post receipt of the Certificate of Completion.

No exclusions for handling of residual contaminated soils will be provided in the Site SMP unless Track 1 USCOs are met. All handling of residual contaminated material will be subject to provisions contained in the SMP for soils left in place above the USCOs.

# **8.0 FINAL ENGINEERING REPORT**

A FER and Site Management Plan will be submitted to NYSDEC following implementation of the conditional Track 1 Remedial Action defined in this RAWP. The FER provides the documentation that the conditional Track 1 remedial work required under this RAWP has been completed and has been performed in compliance with this plan. The FER will provide a comprehensive account of the locations and characteristics of the site preparation demolition activities, SOE installation and all material removed from the Site including the surveyed map(s) of all sources. The FER will include as-built drawings for all constructed elements, certifications, manifests, bills of lading as well as the complete Site Management Plan. The FER will provide a description of the changes in the Remedial Action from the elements provided in the RAWP and associated design documents. The FER will provide a tabular summary of all performance evaluation sampling results and all material characterization results and other sampling and chemical analysis performed as part of the Remedial Action. The FER will provide test results demonstrating that all mitigation and remedial systems are functioning properly. The FER will be prepared in conformance with DER-10.

Where determined to be necessary by NYSDEC, a Financial Assurance Plan will be required to ensure the sufficiency of revenue to perform long-term operations, maintenance and monitoring tasks defined in the Site Management Plan and Environmental Easement. This determination will be made by NYSDEC in the context of the FER review.

The FER will include written and photographic documentation of all remedial work performed under this remedy.

The FER will include an itemized tabular description of actual costs incurred during all aspects of the Remedial Action.

The FER will provide a thorough summary of any residual contamination left on the Site after the remedy is complete. Residual contamination includes all contamination that exceeds the Track 1 USCO in 6 NYCRR Part 375-6.8(a). A table that shows exceedances of Track 1 USCOs for any soil/fill remaining at the Site after the Remedial

Action will be included in the FER, if Track 1 is not achieved. A map that shows the location and summarizes exceedances of Track 1 USCOs for any soil/fill remaining at the Site after the Remedial Action will be included in the FER if Track 1 is not achieved.

The FER will provide a thorough summary of any residual contamination that exceeds the SCOs defined for the Site in the RAWP, if present, and must provide an explanation for why the material was not removed as part of the Remedial Action.

The FER will include an accounting of the destination of all material removed from the Site, including excavated contaminated soil, historic fill, solid waste, hazardous waste, non-regulated material and fluids. Documentation associated with disposal of all material must also include records and approvals for receipt of the material. It will provide an accounting of the origin and chemical quality of all material imported onto the Site.

Before approval of a FER and issuance of a Certificate of Completion, all project reports must be submitted in digital form on electronic media (PDF).

## **8.1 CERTIFICATIONS**

The following certification will appear in front of the Executive Summary of the Final Engineering Report. The certification will be signed by the Remedial Engineer Fuad Dahan who is a Professional Engineer registered in New York State. This certification will be appropriately signed and stamped. The certification will include the following statements:

I \_\_\_\_\_\_\_certify that I am currently a NYS registered professional engineer, I had primary direct responsibility for the implementation of the subject construction program, and I certify that the Remedial Work Plan was implemented and that all construction activities were completed in substantial conformance with the DER-approved Remedial Work Plan.

I certify that all use restrictions, institutional controls, engineering controls and/or any operation and maintenance requirements applicable to the site are contained in an environmental easement created and recorded pursuant to ECL 71-3605 and that any

affected local governments, as defined in ECL 71-3603, have been notified that such easement has been recorded.

I certify that a Site Management Plan has been submitted for the continual and proper operation, maintenance, and monitoring of any engineering controls employed at the site including the proper maintenance of any remaining monitoring wells, and that such plan has been approved by DER. **FIGURES** 



LAYOUT: FIG-1.1 aas, 03: 52: 26PM, LOCATION MAP.DWG 09/14/20 SITE USGS I FIG-1.1 I FIGURES\10637 REPORT RIR N: \ACAD\10637'



NOTE: THIS PLAN IS FOR LOCATING SITE PLAN. OTHER SITE WORK SHOWN HERE IS NOT INTENDED FOR CONSTRUCTION.

Unauthorized alterations or additions to this plan are a violation of section 7209 (2) of the New York State Education Law. Copies of this

This drawing and all information contained here on is proprietary information of SESI CONSULTING ENGINEERS D.P.C. and may not be copied or reproduced, either in whole or in part, by any method, without written permission of SESI CONSULTING ENGINEERS D.P.C.




information of SESI CONSULTING ENGINEERS D.P.C. and may not be copied or reproduced, either in whole or in part, by any method,



NOTE:

This drawing and all information contained here on is proprietary information of SESI CONSULTING ENGINEERS D.P.C. and may not be copied or reproduced, either in whole or in part, by any method, without written permission of SESI CONSULTING ENGINEERS D.P.C.

MAIN STREET -O-SB-0 SB-SB-3 SB-4 NYSDEC SB-6 (1-2) Analyte Units GV PFOS 0.00088 0.00041 mg/kg mg/kg 0.00066 ND PFOA SB-6 SB-8 SB-5 SB-7 NYSDEC SB-9 (1-2) Analyte Units GV PFOS 0.00088 0.00038 mg/kg 0.00066 ND PFOA mg/kg **SB-12** NYSDEC SB-9 SB-10 (1-2) Units Analyte GV SB-11 mg/kg 0.00088 SB-10 PFOS 0.00025 RE PFOA mg/kg 0.00066 ND S NYSDEC Units SB-14 (2-3) Analyte GV (RC) SB-15 SB-13 PFOS 0.00088 0.00039 mg/kg SB-14 PFOA mg/kg 0.00066 ND SB NYSDEC G SB-13 (1-2) Analyte Units GV PFOS mg/kg 0.00088 0.00025 mg/kg 0.00066 0.00014 PFOA SB-20 SB-18 SB-17 NYSDEC Analyte Units SB-17 (7-8) GV PFOS mg/kg 0.00088 0.00029 mg/kg 0.00066 ND PFOA NYSDEC SB-21 (1-2) Analyte Units SB-2 GV SB-22 **SB-23** PFOS 0.00088 mg/kg 0.00022 PFOA mg/kg 0.00066 0.00011 1.4 5 m 13

NOTE: 1. THIS

THIS PLAN IS FOR LOCATING BORINGS ONLY.

OTHER SITE WORK SHOWN HERE IS NOT INTENDED FOR CONSTRUCTION. 2. NYSDEC GV: SOIL GUIDANCE VALUES FOR PFOS AND PFOA (OCTOBER 2020)

#### <u>NYS Education Law</u> Unauthorized alterations or additions to this plan are a violation of section 7209 (2) of the New York State Education Law. Copies of this map not having the seal of the engineer shall not be valid.

© SESI CONSULTING ENGINEERS D.P.C. 2021 This drawing and all information contained here on is proprietary information of SESI CONSULTING ENGINEERS D.P.C. and may not be copied or reproduced, either in whole or in part, by any method, without written permission of SESI CONSULTING ENGINEERS D.P.C.

#### LEGEND:

- PROPERTY LINE
- SITE BOUNDARY

- BORING NUMBER & APPROX. LOCATION



	and the second s	-	Ur	nits AV	VQS MW-3							and and	
	North Street	Benzo(a)anthr	cene u	ug/L 0.0	002 0.022			5 1 4 81	Contraction of the	Cold St. Constitutions			-
		Hexachioroben	zene u	1g/L 0.	.04 0.044				1. A. M.	Ar	nalyte	Uni	its
	1150 -	Aluminum	u	1g/L 20	00 6970			and the second	₼ <sup>MW-2</sup>	Ph	henol	uç	g/l
	10	Manganese	u		00 00 000 000		2 >	1	Ψ	Al	luminum	ug	J/L
	and the second	Sodium		ua/l 20	000 89500		1	121 1 1 1 1	The second	Iro	on	ug	/L
	IT	2015	and the				MW-3		No Maria	So	odium	ug	<u>]/l</u>
	Anabda	linite	NYSDEC	MWA			¥	C.C. Maria					
	Benzo(a) anthracene	unit	0.002	0.067		62.8 · 8100	-		A STATE				
	Benzo(b)flouranthre	ne un/l	0.002	0.075	-	Ser in the second		Martha and and	LT		4	17. 21	
	Benzo(k)flouranthre	ne ua/l	0.002	0.032				The section is	The state of the s	Analyt	te	Units	NY
	Benzo(a)pyrene	ug/L	0	0.062		the first of the second		and and the second	MW-5	Alumin	num	ug/L	
1.333		-				A REAL PROPERTY AND INCOME.		the second se		of the second se			
	Indeno(1,2,3-cd)pyr	ene ug/l	0.002	0.052	1				•	Iron		ug/L	
	Aluminum	ene ug/l ug/L	0.002	0.052	EET		_/	Y	•	Iron Mangar	anese	ug/L ug/L	
	Indeno(1.2,3-cd)pyr Aluminum Iron Sodium	ene ug/l ug/L ug/L ug/L	0.002 2000 600 20000	0.052 2100 2830 194000	CH STREET		MW-4			Iron Mangai Sodium	nese n	ug/L ug/L ug/L	
	Aluminum Iron Sodium Analyte	ene ug/l ug/L ug/L ug/L Ug/L	0.002 2000 600 20000 YSDEC AWQS	0.052 2100 2830 194000 MW-8	CHURCH STREET		€ <sup>MW-4</sup>		MW.6	Iron Manga Sodium Analyte	nnese m Units	NYSDB AWQS	C 3
-	Indeno(1.2.3-cd)pyrn Aluminum Iron Sodium Analyte Magnesium Sodium	ene ug/l ug/L ug/L ug/L Units / ug/L ug/L	0.002 2000 600 20000 YSDEC AWQS 5 20000	0.052 2100 2830 194000 MW-8 43400 241000	CHURCH STREET		€ <sup>MW-4</sup>			Iron Manga Sodium Analyte Fon	nnese n Units ug/L	NYSDE AWQS 600	C 3
	Indeno(1.2.3-cd)pyn Aluminum Iron Sodium Analyte Magnesium Sodium	ene ug/l ug/L ug/L ug/L Units ug/L ug/L	0.002 2000 600 20000 20000 YSDEC AWQS 5 20000	0.052 2100 2830 194000 MW-8 43400 241000	CHURCH STREET		MW-4			Iron Mangai Sodium Analyte Iron Manganese	nnese n Units ug/L ug/L	ug/L           ug/L           ug/L           wg/L           NYSDE           AWQS           600           600	
	Indeno(1.2,3-cd)pyn Aluminum Iron Sodium Analyte Magnesium Sodium	ene ug/l ug/L ug/L ug/L Units ug/L ug/L	0.002 2000 600 20000 YSDEC AWQS 5 20000	0.052 2100 2830 194000 MW-8 43400 241000	CHURCH STREET	MW-8	₩W-4			Iron Manga Sodium Analyte Iron Manganese Sodium	nnese n Units ug/L ug/L	ug/L           ug/L           ug/L           ug/L           NYSDE           AWQS           600           600           20000	
and a second	Indeno(1.2.3-cd)pyn Aluminum Iron Sodium Analyte Magnesium Sodium	ene ug/l ug/L ug/L ug/L Units / ug/L	0.002 2000 600 20000 YSDEC AWQS 5 20000 20000	0.052 2100 2830 194000 MW-8 43400 241000	CHURCH STREET	MW-8	€ MW-4		MW-6	Iron Mangai Sodium Analyte Iron Manganese Sodium	nnese n Units ug/L ug/L	ug/L           ug/L           ug/L           ug/L           NYSDE           Awcs           600           600           20000	C 3
	Indeno(1.2.3-cd)pyn Aluminum Iron Sodium Analyte Magnesium Sodium	ene ug/l ug/L ug/L ug/L Units / ug/L ug/L	0.002 2000 600 20000 YSDEC AWQS 5 20000 S NY nits AV Q/L 2 2 2 2 2 2 2 2 2 2 2 2 2	0.052 2100 2830 194000 MW-8 43400 241000 241000 24000 24000 24000 24000 24000 24000 24000 24000 24000 24000 24000 24000 24000 2400 2400 2800 28	CHURCH STREET	MW-8	MW-4		MW-7	Iron Mangai Sodium Analyte Iron Manganese Sodium	nnese n Units ug/L ug/L	ug/L           ug/L           ug/L           ug/L           ug/L           NYSDE           AWQS           600           20000           NYSDE           AWQS	
	Indeno(1.2.3-cd)pyn Aluminum Iron Sodium Analyte Magnesium Sodium Sodium	ene ug/l ug/L ug/L ug/L ug/L ug/L ug/L	0.002 2000 600 20000 YSDEC AWQS 5 20000 NY hits NY A' g/L 2 0	0.052 2100 2830 194000 MW-8 43400 241000 241000 (SDEC WQS M 2000 2 600 5	CHURCH STREET	MW-8	₩W-4			Iron Mangai Sodium Analyte Iron Manganese Sodium Analyte cis-1,2-Dichloroett	nnese n Units ug/L ug/L ug/L ug/L	ug/L           ug/L           ug/L           ug/L           ug/L           NYSDE           Awqs           600           20000           Awqs           Awqs           600           20000           Awqs           5	C S ) ) ; ; ;

#### NOTE:

1

THIS PLAN IS FOR LOCATING GROUNDWATER SAMPLING ONLY. OTHER SITE WORK SHOWN HERE IS NOT INTENDED FOR CONSTRUCTION. THERE ARE CURRENTLY NO GROUND WATER STANDARDS PROMULGATED FOR PFAS COMPOUNDS BY NYSDEC.

#### NYS Education Law

Unauthorized alterations or additions to this plan are a violation of section 7209 (2) of the New York State Education Law. Copies of this map not having the seal of the engineer shall not be valid.

© SESI CONSULTING ENGINEERS D.P.C. 2020 This drawing and all information contained here on is proprietary information of SESI CONSULTING ENGINEERS D.P.C. and may not be copied or reproduced, either in whole or in part, by any method, without written permission of SESI CONSULTING ENGINEERS D.P.C.







## NOTES:

2. NYSDEC GV: GUIDANCE VALUE FORM OCTOBER 2020 NYSDEC PFAS GUIDANCE

Unauthorized alterations or additions to this plan are a violation of section 7209 (2) of the New York State Education Law. Copies of this

This drawing and all information contained here on is proprietary information of SESI CONSULTING ENGINEERS D.P.C. and may not be copied or reproduced, either in whole or in part, by any method, without written permission of SESI CONSULTING ENGINEERS D.P.C.

	Isopropanol	4.42	ALL ALL				MAIN	STREET			Analyte ug/m3	SVS
	Tetrahydrofuran	9.44	State Law								Bromodichloromethane	26
	Heptane	95.5		1157	TELEPINE .	a production of the second	Part Car	and the second state of	and a start of	1 the said	Chloroform	323
Statement of the local division in which the local division in the local division in the local division in the	Ethylbenzene	31.8		2111	HHH						Isopropanol	35
and the second	Carbon disulfide	7.47					SVS-7	1 X 4 8 1 1 2	the second second		Tetrahy drof uran	13
	Toluene	244					303-7	1 4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		112	4-Methyl-2-pentanone	17
*	n-Hexane	145	and the second			li i	i			1.01	2-Hexanone	6.9
Cornel.	Cyclohexane	39.6	1933		1800	*		Contraction of the	Contraction of the local division of the loc	and the second se	Carbon disulfide	19
the second	m,p-Xylene	108	A DESCRIPTION OF A DESC				1.1	1 . Martin Party	The second of	100	Toluene	7.
	o-Xylene	36.4	1 20 10	1					120.67	i i i	Acetone	2
1 1	Acetone	66.5	and showing that				5			and a second	Methyl Bhyl Ketone	2
A 10000	Methyl Ethyl Kotono	16.0	MACTOR ST					and prover	Contraction of the second	Carles and	Tertiary butyl Alcohol	47
-	Trichloroethene	3 56	Analyte ug/m3	SV S-1		A STATE	i	Martin Martin	19.00	-	1.2.4 Transhulkana and	20
ar's Thinks	4-Ethyltoluene	9.73	Dichlorodifluoromethane	2.8		SVS-6	1.50	all a heren	Summer	1	1,2,4 mmeuryibenzene	10
22	2,2,4-Trimethylpentane	222	Chloromethane	2.97	24	1.3	-	the search	The second second second	N. STORE	No. Common	100
	1,3,5-Trimethylbenzene	12.5	Chloroform	3	1200	Longe and	6 / 1	1000	1. 王子子的	inforter - office		
	1,2,4-Trimethylbenzene	37.8	Ethanol	84.2	1000		*	and the second	when we will the second	22.00		
	California	All sound	Isopropanol	6.78	Mar 1	A	1	-		!		
	And a lot		Tetrahydrofuran	9.53	North Con	100 TO 100		1 martin				
and the second second			Heptane	32.8		a state of the sta	and the second		· · · ·	and the second second		
	200		Ethylbenzene	366		SVS-1						
			4-Methyl-2-pentanone	13.3	1000			and the second second second	LEL BL -			
	A naiyte ug/m 3	SV S-2	2-Hexanone	24.9		Construction of the local distance	-	And and a state of the		i		
	Dichlorodifluoromethane	2.53	1,3-Butadiene	63.3				C. State of the		the form	E .	
	Chloromethane	9.09	Carbon disulfide	18.6			and the second	10 Mar			1010	
	Enanoi	293	Toluene	47.1	li			11 m 1 1 1 1		i		
	Tetrahydrofuran	4.66	n-Hexane	53.9			•		Salation of the			
	Heptane	545	mp Xylono	170			1000	The second	1.18.6	C. C. C. C. C.	· · · · · · · · · · · · · · · · · · ·	
	Ethylbenzene	573	o. Xvlene	130					200 . 10	and a manufacture		
	4-Methyl-2-pentanone	7.7	Acetone	134	3/ 10	conners to los	11/1	2.2. 2. 30	RUMER			
	2-Hexanone	32.5	Benzene	13.5	1	and the second division of the second divisio	100 March 100	Service State	a the second			
	1,3-Butadiene	160	Methyl Ethyl Ketone	124	Ser.		3 allering	man	10 10 to 10	Contraction of	1000	19 845
	Carbon disulfide	17.3	4-Ethyltoluene	115	1 12 - 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1					-	1.1.1.1	
	Toluene	59.9	1,3,5-Trimethylbenzene	20.4	li	SVS-2			10 10 10 10 10 10 10 10 10 10 10 10 10 1	ST ST ST		
	n-Hexane	223	1,2,4-Trimethylbenzene	65.9	100	TOTAL MARK				12		
	Cyclohexane	6.8	Carlos Landon	1	1		Martin Landerson				A CONTRACTOR	
	m,p-Xylene	307		/				CALL IN STREET	of Restoration of States	and the second	and the second s	
	o-Xylene	257			-	A REAL PROPERTY AND ADDRESS OF TAXABLE PARTY.	and the second s				0	
	Acetone	106										
	Methyl Ethyl Ketono	39.3										
	4. Bhy Itoluene	298										
	1.3.5-Trimethylbenzene	737	Service Service W	1000	and the second second	and a second descents	a survey by the	the part of the of	and the second second	and the second second	and the second second	1 14 10
	1.2.4-Trimethylbenzene	268										
	.,_, r minouryibenzene	200	1									

- SITE BOUNDARY

g

<u>NYS Education Law</u> Unauthorized alterations or additions to this plan are a violation of section 7209 (2) of the New York State Education Law. Copies of this map not having the seal of the engineer shall not be valid.

© SESI CONSULTING ENGINEERS D.P.C. 2020 This drawing and all information contained here on is proprietary information of SESI CONSULTING ENGINEERS D.P.C. and may not be copied or reproduced, either in whole or in part, by any method, without written permission of SESI CONSULTING ENGINEERS D.P.C.





Unauthorized alterations or additions to this plan are a violation of section 7209 (2) of the New York State Education Law. Copies of this

This drawing and all information contained here on is proprietary information of SESI CONSULTING ENGINEERS D.P.C. and may not be copied or reproduced, either in whole or in part, by any method, without written permission of SESI CONSULTING ENGINEERS D.P.C.



	UnRestricted	Restricted Residential
	mg/kg	mg/kg
ie	1	1
	1	1
ne	1	1
ne	0.8	3.9
	1.0	3.9
ene	0.33	0.33
rene	0.5	0.5
	50	270
	13	16
	63	400
	109	10000
	30	310
	0.18	0.81
	0.0033	7.9

AVATION AREAS	& VOLUMES
AREA (SF)	VOLUME (CY)
384 SF	72 CY
385 SF	100 CY
947 SF	263 CY
1536 SF	228 CY
1650 SF	1039 CY
1200 SF	245 CY
1610 SF	418 CY
905 SF	335 CY
8617 SF	2700 CY
And the owner water and the owner water and the owner owner water and the owner owner owner owner owner owner o	

and a second	
rea to be excavated from street rade	
rea to be excavated beneath 00 Main Street basement tarting at 13 ft bsg	
Trea to be excavated beneath 12 Church Street basement starting	job no drawi
t 10 ft DSG SCALE: 1"=40" 0 40 80	FI

Shine

dwg by: aas	chk by: TTK	scale: AS NOTED	date: 02/05/2021
SOILS / FOUNDATIONS	SITE DESIGN	ENVIRONMENTAL	", N.J. 07058 PH: 973-808-9050
	CCCCC	ENGINEERS D.P.C.	12A MAPLE AVE. PINE BROOK
RAWP 500 MAIN STREET	NEW ROCHELLE, WESTCHESTER COUNTY, NY		PROPOSED EXCAVATION PLAN
job no: drawing <b>FI</b>	<u>1</u> g no G-	<u>063</u> : -3	<sup>z</sup> 5.1

# **APPENDIX A**

Site Development Plan



STEPHEN W. GRESHAM, AIA 277231 PROJECT #: 117013 DRAWN BY: LOW, RJB, JMD, JS CHECKED BY: WJP, SK STEPHEN W. GRESHAM, **AIA ARCHITECT** 300 NORTH LEE STREET SUITE 502 ALEXANDRIA, VA 22314 T 703 836 0915 www.nilesbolton.com No. Description Date SCHEMATIC DESIGN 7/10/19 50% DESIGN 7/26/19 DEVELOPMENT 11/8/19 100% DESIGN DEVELOPMENT 100% REVISED DESIGN 9/10/20 DEVELOPMENT This drawing, as an instrument of service, is and shall remain the property of the Architects and shall not be reproduced, published or used in any way without the permission of the Architect. 10801 500 MAIN STREET 12 CHURCH STREET NEW ROCHELLE, NY | 1080 500 MAIN, LLC BRP SHEET TITLE: BUILDING PLAN-LEVEL U1 SHEET NUMBER: A1.0U10

100% REVISED DESIGN DEVELOPMENT

09/10/2020

2		B B B B B		
		B		
		B =		
	A     B       A     B       A     B       A     B       B     B			
		B		
		B B B B		
2				

1EXTERIOR ELEVATION- NORTHA3.0011/16" = 1'-0"



	 ⇒	 		PENTHOUSE ROOF 414'-2 1/2"
				TOP OF POOL PAVERS 397'-6 1/2" TOP OF PAVERS
43				395'-8 1/2" ROOF 394'-2 1/2"
				LEVEL 26 379'-4"
23				LEVEL 25 369'-1"
23				LEVEL 24 358'-10"
23				LEVEL 23
23			32	
23			32	LEVEL 21
23			32	328'-1"
23			32	317'-10" U
23				307'-7" T
23				
23		*		
23				LEVEL 16 276'-10"
23			32	LEVEL 15 266'-7"
23			32	LEVEL 14 256'-4"
				LEVEL 13 246'-1"
				LEVEL 12 235'-10"
				LEVEL 11 225'-7"
				LEVEL 10 215'-4"
				LEVEL 9 205'-1"
				LEVEL 8 194'-10"
23				LEVEL 7 184'-7"
23				
20		28		
				LEVEL 5 159'-4"
		6		
				141'-10"
				3 LEVEL 3
		50		
				LEVEL 2 111'-0"
				94'-0"
10	<u>    8                                </u>	1		

# PANEL

Key Value



414'-2 1/2"						++		 		
TOP OF POOL PAVERS										
7         397'-6 1/2"           TOP OF PAVERS         395'-8 1/2"						 				
ROOF 394'-2 1/2"							 _		47	
B LEVEL 26 379'-4"	B -	B	B	B	B					
B LEVEL 25	B	B	B	B	B				33	
B LEVEL 24	B	B	B	B	B				33	
358'-10"	B	B	B	B	B				33	
B B B B B B B B B B B B B B B B B B B	B	B	B	B	B				33	
LEVEL 22 338'-4"										
LEVEL 21 328'-1"					B					
B LEVEL 20 317'-10"	B	B	B	B	B					
B LEVEL 19	B	B	B	B	B				33	
B	B	B	B	B	B				33	
297'-4"	B	B	B		B				33	
LEVEL 17 287'-1"									33	
LEVEL 16 276'-10"										
B LEVEL 15 266'-7"	B -	B	B	B	B				33	
B LEVEL 14	B	B	B	B	B				33	
B LEVEL 13	B	B	B	B	B				33	
246'-1" B	B	B	B	B	B				33	
235'-10"	B	B	B		B				33	
LEVEL 11 225'-7"										
LEVEL 10 215'-4"										
B LEVEL 9 205'-1"	B	B	B	B	B				33	
B LEVEL 8	B	B	B	B	B				33	
B I I FYEL 7	B	B	B	B	B				33	
2 184'-7" <b>3</b>	B	B	B	B	B				33	
LEVEL 6 174'-4"										2
		B	B		B					
159'-4"										
LEVEL 4 💣										
141'-10"										
LEVEL 3 🛋										
125'-0"										
LEVEL 2 111'-0"							 	 		

1EXTERIOR ELEVATION- EAST AA3.0021/16" = 1'-0"



2 EXTERIOR ELEVATION- EAST B A3.002 1/16" = 1'-0"

·				PENTHOUSE ROOF 414'-2 1/2"	
			T	OP OF POOL PAVERS 397'-6 1/2"	
				TOP OF PAVERS	¥
		B		394'-2 1/2" 	
				LEVEL 26 379'-4" •	$\mathbf{i}$
	B			LEVEL 25	$\mathbf{i}$
C         B	B B	B		LEVEL 24 358'-10"	$\mathbf{i}$
B	B B	B C C C C C C C C C C C C C C C C C C C			
C B	B B	B		ن معنی 1 معنی LEVEL 22	
	B B	B		338'-4" T	
		B			
				<u>LEVEL 20</u> 317'-10"	$\mathbf{k}$
	BB			LEVEL 19 307'-7"	$\mathbf{i}$
C   B	B B	B		LEVEL 18 297'-4"	
- C B	B B	B		LEVEL 17	
C   B	B B	B		لية بالمعرفة المعرفة الم	
C B	B B	B		276'-10" T	
	B B	B		<u></u>	
				LEVEL 14 256'-4" •	
				LEVEL 13 246'-1"	$\mathbf{i}$
C   B	B B	B		LEVEL 12 235'-10"	
C   B	B B	B		LEVEL 11	$\mathbf{x}$
C   B	B B	B		لية برانية بوني LEVEL 10	
C B	B B	B		215'-4" 🗸	
	B B	B			
				<u>ــــــــــــــــــــــــــــــــــــ</u>	$\mathbf{k}$
				ـــــــــــــــــــــــــــــــــــــ	$\mathbf{i}$
	B B			LEVEL 6 174'-4"	$\mathbf{i}$
				15:-0"	
				LEVEL 5 159'-4"	
				17-6"	
				LEVEL 4 141'-10"	$\mathbf{k}$
				16'-10'	
				LEVEL 3 125'-0"	
				<u>EVEL 2</u>	
				- , , ,	
				RESIDENTIAL LOBBY	

END - KEYNOTES Keynote Text METAL PANEL - NISSAN GRAY METAL PANEL - SATIN MICA METAL PANEL - CADET GRAY WINDOW TRIM (METAL) METAL PANEL REVEAL GLASS RAILING	STEF 2777 PRO DRA CHE ST AI	PHEN V 231 JECT # WN BY CKED F CKED F	V. GR : 1 : LC 3Y: EN CHI	ESHAM, 17013 )W, R. WJP, W. GI TECT	AIA JB, J SK RESI	MD, JS HAM,	
	3000 SUI ALE T 70 WW No. 3 4 5	0 NOR TE 50 EXANI 03 83 w.nil 50% D DEVEL 100% I DEVEL 100% I DEVEL	CTH D2 DRIA 6 09 ESIGN OPME DESIG OPME REVIS OPME	LEE S		T 4 Date 7/26/19 11/8/19 9/10/20 9/10/20	
	be repr	500 MAIN STREET	12 CHURCH STREET	NEW ROCHELLE, NY   10801		BRP 500 MAIN, LLC	
	SHEE EL SHEE		FLE: BUI ATI MBE BED C	LDIN ONS ER: DESIGN I 10/20	NG - E O DEVEL 20	EAST 2 opment	

_	414'-2 1/2" 🖤		5					
	TOP OF POOL PAVERS 397'-6 1/2" TOP OF PAVERS							
	395'-8 1/2" ROOF 394'-2 1/2"							
	LEVEL 26	B	C	B B	B B	G	B	G
	379'-4"	B	C	B B	B B	G	B	G
	<u></u>	B	C	B B	B B	G	B	G
	LEVEL 24 358'-10"					G		G
	LEVEL 23 348'-7"							
	<u>LEVEL 22</u> 338'-4"					G	B	G
	<u>LEVEL 21</u>	B	C	B B	B B	G	B	G
	7LEVEL 20	B	B	B B	B B	G	B	G
	317-10" 🖤	B	C	B B	B B	G	B	G
	307'-7"	B	C	B B	B B	G		G
		B	C	B B	B B	G	B	G
	LEVEL 17 287'-1"							
	6LEVEL 16 276'-10"							
	<u>LEVEL 15</u> 266'-7"							G
		B 	C	B B	BB	G	B	G
		B	B	B B	B B	G	B	G
	LEVEL 12	B	C	B B	B B	G		G
	235'-10" V	B	<u> </u>	B B	B B	) <u>G</u>	B	G
	<u>6</u> 225'-7"	B		B B	BBB	G		G
	<u>LEVEL 10</u> 215'-4"							
	7 LEVEL 9 205'-1"		B			G		G
	LEVEL 8 194'-10"	B B B B C C C C C C C C C C C C C C C C		B B	B B	G	B	G.
		B		B B	BB	G		G
	2 LEVEL 6	B	B	B B	B B	G	B	G
		D B	E			H G	D	H G
	7 LEVEL 5 159'-4"							
	LEVEL 4 141'-10"							
	LEVEL 3 125'-0"							
	111-V Y							
	RESIDENTIAL LOBBY							

1 EXTERIOR ELEVATION- WEST A A3.003 1/16" = 1'-0"

18'-6"												
	$\frac{\text{TOP OF POOL PAVERS}}{397'-6 1/2"} \bigoplus$		22									
	TOP OF PAVERS 395'-8 1/2"											
0 1/2"	ROOF 394'-2 1/2"		42									
14'-1	LEVEL 26											
10'-3"	379'-4"		22									
	LEVEL 25 369'-1"											
10'-	LEVEL 24		22									·
10'-3"			22									
ې ب	<u>LEVEL 23</u> 348'-7"											
10	LEVEL 22 338'-4"											
10'-3"			22									
0'-3"	328'-1"		22									
	LEVEL 20 317'-10"											
10'-3	LEVEL 19		22									
10'-3"	307'-7"		22									
	LEVEL 18 297'-4"											
10'-	LEVEL 17		22									
10'-3"	287-1"		22									
μ	276'-10"											
10	LEVEL 15 266'-7"											
10'-3"			22									
0'-3"	256'-4"											
	LEVEL 13 246'-1"											
10'-3	LEVEL 12		22									
10'-3"	235'-10"		22									
	LEVEL 11 225'-7"											
10'-			22									
10'-3"	215-4 🞔		22									
μ	205'-1"											
10	LEVEL 8 194'-10"											
10'-3"			22									
0'-3"	184'-7"											
	LEVEL 6 174'-4"											
15'-0"												
	7 LEVEL 5 159'-4"											
7-6"						16					]	14
~	2LEVEL 4											
.o	141'-10"								X			
16'-1(											_	
	LEVEL 3 125'-0"						++				]	<u> </u>
14'-0"		40		13				13	13	13		
	LEVEL 2 111'-0"											 [10
17-0"												
		,, <b>  ,, ⊮,                            </b>										

2 EXTERIOR ELEVATION- WEST B A3.003 1/16" = 1'-0"

	LEGEND - KE
Key Value	
1	METAL
2	METAL
3	METAL
4	WINDO
6	METAL
7	GLASS
9	BUILDI
10	ENTRY
14	LOUVE

FYNOTES			
Keynote Text			
L PANEL - NISSAN GRAY			
DW TRIM (METAL)			
S RAILING ING SIGNAGE	STEPHEN W. GRESHAM, AIA		
F PANEL	277231		
	DRAWN BY: LOW, RJB,	JMD, JS	
	CHECKED BY: WJP, SK		
	STEPHEN W. GRES	SHAM,	
	300 NORTH LEE STRE	ET	
	SUITE 502 ALEXANDRIA, VA 223 <sup>2</sup>	14	
	T 703 836 0915		
	www.nilesbolton.con	า	
	No. Description	Date	
	5 100% REVISED DESIGN	9/10/20	
	This drawing, as an instrument of servic remain the property of the Architects a be reproduced, published or used in an the permission of the Architect	ce, is and shall nd shall not y way without	
	801		
	EET == 100	-LC	
	STR STRI NY	IN, L	
	IN S RCH	0 MA	
	MA CHUI	P 50(	
	500 12 ^ RC	BRI	
	L. U.		
	I		
	SHEET TITLE:		
	ELEVATIONS- V	VEST	
	SHEET NUMBER:		
		<b>२</b>	ED FOR
	09/10/2020		NOT R CONST



 	; 			 	
2			36		
 			30		
 			30		
 			30		
 			30		
 			30		
 	<u> </u>		30		
 			30		
 			30		
 	 		30		
 	<u> </u>		30		
 			30		·
 	 		30		
 	<u> </u>		30		·
 	 		30		
 					·
 	<u> </u>				
 					·
 	<u> </u>				
 	<u>=                                     </u>	┉┈┝╋	1		

2 EXTERIOR ELEVATION- NORTH COURTYARD NORTH A3.004 1/16" = 1'-0"

PENTHOUSE ROOF 414'-2 1/2"

	<b>—</b>							
								397'-6 1/2" TOP OF PAVERS
								395'-8 1/2" ROOF
	B	B	G	B	B	B	B	
	B	B	G	B	B	B	B	LEVEL 25
	B	B	G	B	B	B	B	LEVEL 24
	B	B	G	B	B	B	B	<u>LEVEL 23</u> 348'-7"
	B	B	G	B	B	B	B	LEVEL 22
	B	B	G	B	B	B	B	
	B	B	G	B	B	B	B	<u>LEVEL 20</u>
	B	B	G	B	B	B	B	LEVEL 19
	B	B	G	B	B	B	B	LEVEL 18
	B	B	G	B	B	B	B	LEVEL 17
	B	B	G	B	B	B	B	LEVEL 16
	B	B	G	B	B	B	B	LEVEL 15
	B	B	G	B	B	B	B	LEVEL 14
	B	B	G	B	B	B	B	LEVEL 13
	B	B	G	B	B	B	B	LEVEL 12 235'-10"
	B	B	G	B	B	B	B	LEVEL 11
6	B	B	G	B	B	B	B	LEVEL 10
	B	B	G	B	B	B	B	LEVEL 9
4	B	B	G	B	B	B	B	LEVEL 8
	B	B	G	B	B	B	B	LEVEL 7 184'-7"
2	B	B	G	B	B	B	B	
			H					1/4-4 🏸
	B		G			B		LEVEL 5 159'-4"

\_\_\_\_\_



3 EXTERIOR ELEVATION- NORTH COURTYARD EAST A3.004 1/16" = 1'-0"





24	TOP OF POOL PAVERS	
		 ۲. ب
	395' 8 1/2" ROOF 394'-2 1/2"	-10 1/2" 1
	LEVEL 26	<u>7</u>
	LEVEL 25	10'-3"
	₹ 369'-1" ♥ LEVEL 24 ▲	10:-3"
	358'-10" V	10'-3"
	348'-7"	10'-3"
	LEVEL 22 338'-4"	10'-3"
24	LEVEL 21 328'-1"	0'-3"
	LEVEL 20 317'-10"	
	LEVEL 19 307'-7"	
	LEVEL 18 297'-4"	
	LEVEL 17 287'-1"	10'-3
		10:-3"
	LEVEL 15	10-3"
	266'-7" V	10'-3"
		10 <sup>-</sup> .3"
	LEVEL 13 246'-1"	
	LEVEL 12 235'-10"	
	LEVEL 11 225'-7"	
	LEVEL 10 215'-4"	
	LEVEL 9 705'-1"	10'-3
		10'-3"
	₹ 194'-10" \ <b>/</b> I FVFL 7 →	10-3"
		10'-3"
	<u>↓</u> <u>LEVEL 6</u> = 174'-4"	<b>\</b>
21		15-0"
	LEVEL 5 159'-4"	<del>\</del>

1EXTERIOR ELEVATION- SOUTH TERRACE SOUTHA3.0051/16" = 1'-0"

		B	B	B	
2					
	B	B	B	B	
	B	B	B	B	
4	B	B	B	B	
	B	B	B	B	
7	B	B	B	B	
6	B	B	B	B	
	B	B	B	B	
	B	B	B	B	
	B	B	B	B	
	B	B	B	B	
	B	B	B	B	
	B	B	B	B	
	B	B	B	B	
6	B	B	B	B	
	B	B	B	B	
	B	B	B	B	
		B	B	B	
		B	B	B	
			B	B	
		B	B	B	
		B	B	B	
	D B		D B	B	

	OF POOL PAVERS 397'-6 1/2" TOP OF PAVERS 395'-8 1/2"	
<b>-</b>	ROOF 394'-2 1/2"	14'-10 1/2"
<u> </u>	LEVEL 26 379'-4"	10'-3"
Δ	LEVEL 25 369'-1"	10'-3"
<u> </u>	LEVEL 24 358'-10"	10'-3"
Δ	LEVEL 23 348'-7"	10'-3"
<u> </u>	LEVEL 22 338'-4"	10'-3"
<u> </u>	LEVEL 21 328'-1"	10'-3"
<u> </u>	LEVEL 20 317'-10"	10'-3"
<u> </u>	LEVEL 19 307'-7"	10'-3"
<u> </u>	LEVEL 18 297'-4"	10'-3"
<u> </u>	LEVEL 17 287'-1"	10'-3"
<u> </u>	LEVEL 16 276'-10"	10'-3"
Δ	LEVEL 15 266'-7"	10'-3"
<u> </u>	<u>LEVEL 14</u> 256'-4"	10'-3"
<u> </u>	<u>LEVEL 13</u> 246'-1"	10'-3"
<u> </u>	<u>LEVEL 12</u> 235'-10"	10'-3"
	LEVEL 11 225'-7"	10'-3"
<u> </u>	LEVEL 10 215'-4"	10'-3"
<u> </u>	LEVEL 9 205'-1"	10'-3"
<u> </u>	LEVEL 8 194'-10"	10'-3"
<u> </u>	184-7"	10'-3"
Ĵ Ĵ	174'-4"	15'-0"
	LEVEL 5 159'-4"	

2	EXTERIOR ELEVATION- SOUTH TERRACE WEST
A3.005	1/16" = 1'-0"

LEVEL 5 159'-4"

		397'-6 1/2"	¥ۇ—
		ROOF	×
	45	394'-2 1/2"	10 1/
			14'-1
		LEVEL 26 379'-4"	$\rightarrow$
26	31		03"
		LEVEL 25	
		369'-1" Ѱ	ښ ا
26	31		10.
		358'-10"	
26	31		03"
		LEVEL 23	
		348'-7" 🖤	μ,
			10.
		<u></u> <u></u> <u></u> <u></u> <u></u> <u></u> <u></u>	
26	31		03"
		LEVEL 21	
		328'-1" 🖤	<sup>بہ</sup> (
			10
		<u>317'-10"</u>	
26			10'-3'
		LEVEL 19	<b>`</b>
		307-7 🖙	<u>"</u> -
			9
		297'-4"	
26	31		10'-3
		LEVEL 17	<b>\</b>
			3"
		LEVEL 16	<del>2</del>
		276'-10"	
26	31		10-0
		LEVEL 15 266'-7"	<u> </u>
26	31		03"
		256'-4" 🛡	- M
26	31		10'-
		<u>LEVEL 13</u> 246'-1"	
26			03"
		LEVEL 12	
		235-10" 🗸	<sup>ل</sup> م
		I <b>F\/FI 11 →</b>	10
			<u> </u>
26			10'-3
		213-4 🛩	"C-'
		LEVEL 9 🖛	<b>1</b>
		205'-1"	<b>`</b> _
26	31	= <b>7</b>	10'-3
		LEVEL 8 194'-10"	<u> </u>
			)-3"
			¥
		184'-7" 🛡	
26			-10-
		LEVEL 6 174'-4"	$\rightarrow$
		<u>J</u> J	
25	27		15-
		LEVEL 5	

	LEGE	ND - KE
Key Value	I	
		METAL
		METAL
		WINDO
		METAL
		C1 1 CC

EYNOTES	
Keynote Text	
L PANEL - SATIN MICA	
L PANEL - CADET GRAY	
OW TRIM (METAL)	
L PANEL REVEAL	
5 RAILING	

STEPHEN W. G 277231 PROJECT # : 1 DRAWN BY: L CHECKED BY: STEPHEN AIA ARCH	RESHAM, AIA 117013 OW, RJB, JA WJP, SK W. GRESH	MD, JS		
300 NORTH SUITE 502 ALEXANDRI T 703 836 0 www.nilest No. Des 4 100% DESI DEVELOPM 5 100% REVI DEVELOPM	LEE STREE A, VA 22314 0915 Dolton.com Scription GN AENT SED DESIGN AENT	Date 11/8/19 9/10/20		
This drawing, as an ir remain the property be reproduced, publis the permission of the	istrument of service, of the Architects and shed or used in any v Architect.	is and shall I shall not vay without		
500 MAIN STREET 12 CHURCH STREET	NEW ROCHELLE, NY   10801	BRP 500 MAIN, LLC		
SHEET TITLE BU ELEV TE SHEET NUMB 100% REVISED 09.	: ILDING ATIONS RRACE ER: DESIGN DEVELO /10/2020	- 5 DPMENT	NOT RELEASED FOR	CONSTRUCTION



		TOP OF PO	OL PAVERS 397'-6 1/2" OF PAVERS 395'-8 1/2" ROOF 394'-2 1/2"
FLOOR ASSEMBLY TYPE- SEE SHEET G0.014	UNIT B1	. 14'-10 1/2"	1 FVF1 26 🔺
	UNIT B1	10:-3"	379'-4"
FLOOR ASSEMBLY TYPE- FLR SEE SHEET G0.014	UNIT B1		369'-1"
			LEVEL 23 348'-7"

	Ļ	TOP OF POOL PAVER 397'-6 1/
		395'-8 1/
FLOOR ASSEMBLY		
SEE SHEET GO.014		14'-10 1/2"
		LEVEL 2 379'-
		10'-3"
		LEVEL 2 369'-
		10 <sup>'</sup> -3"
FLOOR ASSEMBLY		LEVEL 2 358'-1
SEE SHEET GO.014		10'-3"
	_₩	LEVEL 2





4 WALL SECTION 4 A4.100 1/8" = 1'-0"

3 WALL SECTION 3 A4.100 1/8" = 1'-0"







2 WALL SECTION 2 A4.100 1/8" = 1'-0"



09/10/2020











FLOOR ASSEMBLY TYPE- FLR ------SEE SHEET G0.014

FLOOR ASSEMBLY

SEE SHEET G0.014

TYPE- ROOF -

CORE WALL TYPE- Mae6-----SEE SHEET G0.011

FINISH WALL TYPE- MP2------SEE SHEET G0.011

CORE WALL TYPE- Mae6— SEE SHEET G0.011

FLOOR ASSEMBLY TYPE- FLR SEE SHEET G0.014

FINISH WALL TYPE- MP2------SEE SHEET G0.011

FLOOR ASSEMBLY TYPE- FLR SEE SHEET G0.014

FINISH WALL TYPE- MP2— SEE SHEET G0.011

CORE WALL TYPE- Mae6-----SEE SHEET G0.011 FINISH WALL PE- MP2-ET G0.011



	 	 SEE SI	TYPE HEET
			F
			r t











1 TOWER SECTION 1 A4.103 1/8" = 1'-0"







4 TOWER SECTION 8 A4.104 1/8" = 1'-0"















4 TOWER SECTION 12 A4.105 1/8" = 1'-0"



CORRIDOR LEVEL 13 246'-1" FLOOR ASSEMBLY TYPE- FLR — SEE SHEET G0.014 CORRIDOR LEVEL 12 235'-10" \_\_\_\_\_ CORE WALL TYPE- Mae6-----SEE SHEET G0.011 FINISH WALL TYPE- MP3-----SEE SHEET-G0.011----CORRIDOR LEVEL 11 225'-7" CORRIDOR LEVEL 10 215'-4" CORRIDOR LEVEL 9 205'-1" FLOOR ASSEMBLY TYPE- FLR – SEE SHEET G0.014 CORRIDOR LEVEL 8 194'-10" CORE WALL TYPE- Mae6-----SEE SHEET G0.011 ? CORRIDOR FINISH WALL TYPE- MP3-----SEE SHEET G0.011 LEVEL 7 184'-7" CORRIDOR LEVEL 6 174'-4" FLOOR ASSEMBLY CORR. TYPE- FLR SEE SHEET G0.014

LEVEL 26 379'-4"

3 TOWER SECTION 11 A4.105 1/8" = 1'-0"









2 TOWER SECTION 10 A4.105 1/8" = 1'-0"



# **APPENDIX B**

Health and Safety Plan



# SITE-SPECIFIC HEALTH AND SAFETY PLAN

Proposed Redevelopment 500 Main Street Laundry New Rochelle, Westchester County, New York

**Prepared For:** 

BRP 500 Main LLC 767 3rd Avenue, 33rd Floor New York, New York 10017

**Prepared By:** 

SESI CONSULTING ENGINEERS 12A Maple Avenue Pine Brook, NJ 07058

## Project No.: 10637

## February 2021

**Disclaimer:** This Health and Safety Plan (HASP) is based upon information provided [and, if applicable, conditions discovered during a site visit], and is limited by the project scope.

The HASP should be periodically reviewed and updated based on a number of factors, including but not limited to: (1) changes in applicable governmental requirements; (2) changes in procedures at the site; and (3) site conditions which were unknown to SESI Consulting Engineers (SESI) as of the time the HASP was prepared.

This HASP has been prepared for the sole and exclusive use of BRP Companies., and may not be relied upon by any other person without the express written consent and authorization of SESI.

### SITE-SPECIFIC HEALTH AND SAFETY PLAN

For Proposed Redevelopment 500 Main Street Laundry New Rochelle, Westchester County, New York

one 1 aux

Prepared by:

Date: 02/04/2020

Jesse Mausner, P.G. SESI- Project Manager

Approved by:

Date: 02/04/2020

Fuad Dahan SESI-Principal

# Table of Contents

HEAI	TH AND SAFETY PLAN SUMMARY	
1.0	INTRODUCTION	9
1.1	OBJECTIVE	
1.2	SITE AND FACILITY DESCRIPTION	9
1.3	Policy Statement	9
1.4	References	
1.5	Definitions	
2.0	PROJECT SCOPE OF WORK	
3.0	ROLES AND RESPONSIBILITIES	
21		11
3.1	ALL I ERSONNEL	11
3.2	21  Project  Officer  (PO)	
3	2.2.1 Project Manager (PM)	
3	2.3 Health and Safety Manaoer (HSM)	
3	2.4 Site Safety Officer (SSO)	12
3	2.5 Field Supervisor (FS)	
3	2.6 Field Personnel (FP)	
3.3	SUBCONTRACTORS	
3.4	STOP WORK AUTHORITY	
3.5	ALL ON-SITE PERSONNEL	
3.6	VISITORS	14
4.0	PERSONAL PROTECTIVE EQUIPMENT	15
4.1	LEVELS OF PROTECTION	
4	1.1 Level D Protection	
4	1.2 Modified Level D Protection	
4	1.3 Level C Protection	
4.2	SELECTION OF PPE	
4.3	SITE RESPIRATORY PROTECTION PROGRAM	
4.4	USING PPE	
4	4.1 Donning Procedures	17
4	4.2 Doffing Procedures	17
4.5	SELECTION MATRIX	
5.0	AIR AND NOISE MONITORING	
5.1	AIR MONITORING	
5.2	NOISE MONITORING	
5.3	MONITORING EOUIPMENT MAINTENANCE AND CALIBRATION	
5.4	Action Levels	
6.0	WORK ZONES AND DECONTAMINATION	

6.1	W	ORK ZONES	20
6	.1.1	Authorization to Enter	20
6	.1.2	Site Orientation and Hazard Briefing	20
6	.1.3	Certification Documents	21
6	.1.4	Entry Log	21
6	.1.5	Entry Requirements	21
6	.1.6	Emergency Entry and Exit	21
6	.1.7	Contamination Control Zones	21
6	.1.8	Exclusion Zone (EZ)	21
6	.1.9	Contamination Reduction Zone	21
6	1.10	Support Zone (SZ)	22
6	1.11	Posting	22
6	1.12	Site Inspections	22
6.2	D	ECONTAMINATION	22
6	2.1	Personnel Decontamination	22
6	22	Favinment Decontamination	22
6	23	Personal Protective Fauinment Decontamination	22
	.2.5		
7.0	TRA	AINING AND MEDICAL SURVEILLANCE	23
7.1	T	RAINING	23
7	.1.1	General	23
7	.1.2	Basic 40-Hour Course	23
7	.1.3	Supervisor Course	23
7	.1.4	Site-Specific Training	23
7	.1.5	Daily Safety Meetings	24
7	1.6	First Aid and CPR	24
7.2	M	EDICAL SURVEILLANCE	24
7	21	Medical Examination	24
, 7	22	Pre-nlacement Medical Examination	24
, 7	23	Other Medical Examinations	25
, 7	2.5	Periodic Exam	25
7	2.7	Medical Restriction	25
/	.2.0		20
8.0	GEI	NERAL SAFETY PRACTICES	25
8.1	G	ENERAL SAFETY RULES	25
8.2	B	UDDY SYSTEM	26
8.3	Н	EAT STRESS	27
8.4	Н	EAT STRESS SAFETY PRECAUTIONS	28
T	able 4	4 – Work/Rest Schedule	28
8.5	С	OLD STRESS	29
8.6	S	AFETY PRECAUTIONS FOR COLD STRESS PREVENTION	31
8.7	S	AFE WORK PRACTICES	31
8.8	В	IOLOGICAL HAZARDS	31
8	.8.1	Tick Borne Diseases	32
8	.8.2	Poisonous Plants	32
8	.8.3	Snakes	33
8	8.4	Spiders	33
0		- <i>r</i>	

8.9	) N	OISE	
8.1	0 SI	PILL CONTROL	
8.1	1 SA	ANITATION	
	8.11.1	Break Area	
	8.11.2	Potable Water	
	8.11.3	Sanitary Facilities	
	8.11.4	Lavatory	
8.1	2 Ei	MERGENCY EQUIPMENT	
8.1	3 Lo	OCKOUT/TAGOUT PROCEDURES	
8.1	4 E	LECTRICAL SAFETY	
8.1	5 L	IFTING SAFETY	
8.1	6 L	ADDER SAFETY	
8.1	7 Ti	RAFFIC SAFETY	
9.0	SIT	E-SPECIFIC HAZARDS AND CONTROL MEASURES	
9.1	E	VALUATION OF HAZARDS	
	9.1.1	Hazard Characteristics	
	9.1.2	Potential Health and Safety Hazards	
9.2	2 Fi	ELD ACTIVITIES, HAZARDS, AND CONTROL PROCEDURES	
	9.2.1	Chemical Sample Collection/Analysis	
	9.2.2	Decontamination	
9.3	<b>C</b>	HEMICAL HAZARDS	40
10.0	EM	ERGENCY PROCEDURES	42
10	1 G	ENERAL	42
10	.2 Ei	MERGENCY RESPONSE	
10	1021	Fire	42
	10.2.2	Contaminant Release	42
10	.3 M	EDICAL EMERGENCY	43
10	10.3.1	Emergency Care Steps	
10	.4 Fi	IRST AID - GENERAL	
10	10.4.1	First Aid - Inhalation	44
	10.4.2	First Aid - Ingestion	44
	10.4.3	First Aid - Skin Contact	44
	10.4.4	First Aid - Eve Contact	
10	.5 R	EPORTING INJURIES, ILLNESSES, AND SAFETY INCIDENTS	
10	.6 Ei	mergency Information	
-	10.6.1	Directions to Hospital	
11.0	LOC	GS, REPORTS, AND RECORD KEEPING	
11	1 H	ASP FIELD CHANGE REQUEST	46
11	.2 M	EDICAL AND TRAINING RECORDS.	46
11	3 F	XPOSURE RECORDS	
11	Δ Δ	CCIDENT/INCIDENT REPORT	
11	5 O	SHA FORM 200	
11	.50 60	N-SITE HEALTH AND SAFETY FIELD LOGBOOKS	
11	7 M	A TERIAL SAFETY DATA SHEFTS	
	. /		

12.0	COVID-19 RISK MANAGEMENT	
12.1	BEST PRACTICES TO PREVENT INFECTION	
12.2	RISK IDENTIFICATION	
12.3	RISK MINIMIZATION	
12.4	SAFE WORK PRACTICES	
12.5	PPE TO PREVENT COVID-19 TRANSMISSION	
12.6	NOTIFICATION OF A POTENTIAL OR CONFIRMED INFECTION	

## LIST OF FIGURES

# Figure No. Title

Fig-1	Site	Location	Мар
-------	------	----------	-----

Fig-2 Montefiore New Rochelle Hospital

## LIST OF EMBEDDED TABLES

<u>Table</u>	<u>Title</u>	Page
1.	Key Safety Personnel	8
2.	PPE Selection Matrix	11
3.	Airborne Contaminant Action Levels	12
4.	Work/Rest Schedule	22
5.	Wind Chill Temperature Chart	23
6	List of Primary Contaminants	44
7	Emergency Contacts	51

## LIST OF ATTACHMENTS

Attachment 1	Air Monitoring Log
Attachment 2	OSHA Poster
Attachment 3	HASP Field Change Request Form
Attachment 4	Accident/Incident Report
Attachment 5	Signatory Page
Attachment 6	Material Safety Data Sheets
Attachment 7	COVID-19 Acknowledgement form and Posters

# LIST OF ACRONYMS AND ABBREVIATIONS

ACGIH	American Conference of Governmental Industrial Hygienists
COC	Constituent(s) of Concern
CRZ	Contamination Reduction Zone
EZ	Exclusion Zone
FS	Field Supervisor
GFCI	Ground Fault Circuit Interrupter
HASP	Health and Safety Plan
HSM	Health and Safety Manager
LEL	Lower Explosive Limit
MSDS	Material Safety Data Sheet
OSHA	Occupational Safety and Health Administration
PCB	Polychlorinated Biphenyls
PEL	Permissible Exposure Limit
PID	Photoionization Detector
PM	Project Manager
PO	Project Officer
PPE	Personal Protective Equipment
SESI	SESI Consulting Engineers
SSO	Site Safety Officer
SVOC	Semi-Volatile Organic Compound
SZ	Support Zone
TLV	Threshold Limit Value
USCG	United States Coast Guard
USEPA	United States Environmental Protection Agency
VOC	Volatile Organic Compound

## HEALTH AND SAFETY PLAN SUMMARY

The chemical hazards associated with site operations are related to inhalation, ingestion, and skin exposure to site Chemicals of Concern (COCs). COCs at the site include some volatile organic compounds, semi-volatile organic compounds, and pesticides, in soil and soil vapor. Concentrations of airborne COCs during site tasks may be measurable and air monitoring should be performed.

The potential for inhalation of site COCs is low. The potential for dermal contact with soils containing site COCs during remedial operations is low.

The following table summarizes airborne contaminant action levels that will be used to determine the procedures and protective equipment necessary based on conditions as measured at the site.

Parameter	Reading	Action
Dust	0 to .5 mg/m3	Normal operations
	0.5 to 1 mg/m3	Begin soil wetting procedure (Level C protection would be needed beyond this point)
	> 1 mg/m3	Stop work, fully implement dust control plan
Oxygen	<u>&lt;</u> 19.5%	Stop work, evacuate confined spaces/work area, investigate cause of reading, and ventilate area
	> 19.5% to < 23.5%	Normal operations
	<u>≥</u> 23.5%	Stop work, evacuate confined spaces/work area, investigate cause of reading, and ventilate area
Carbon Monoxide	0 ppm to <u>&lt;</u> 20 ppm	Normal operations
	> 20 ppm	Stop work, evacuate confined spaces/work area, investigate cause of reading, and ventilate area

The level of personal protection selected will be based on air monitoring of the work environment and an assessment by the Field Supervisor and Site Safety Officer. The following table presents a selection matrix to determine appropriate Personal Protective Equipment.

Task	Anticipated Level of Protection
Chemical Sampling	Modified Level D/Level C
Decontamination	Modified Level D

# 1.0 INTRODUCTION

# 1.1 Objective

The objective of this Health and Safety Plan (HASP) is to provide a mechanism for establishing safe working conditions during Remedial Action (RA) activities at the Proposed development at 500, 506, and 510 Main Street, and 12 Church Street in New Rochelle, Westchester County, New York (the "Site"). The safety organization, procedures, and protective equipment have been established based on an analysis of potential physical, chemical, and biological hazards. Specific hazard control methodologies have been evaluated and selected to minimize the potential of injury, illness, or other hazardous incident.

The HASP was written to meet the requirements of all applicable Federal, State, and local health and safety regulations, including 29 CFR 1910.120. The HASP is based on current knowledge regarding the specific chemical and physical hazards that are known or anticipated at the Site. This HASP is a dynamic document, for which changes and/or revisions may be realized as changes in scope and/or site conditions are encountered. Should revised documents be produced, said revised documents will refer to the specific changes and why they were made.

# 1.2 Site and Facility Description

The Site is located in the City of New Rochelle, Westchester County, New York. Figure 1.2 of the Remedial Action Work Plan (RAWP) provides a location of the Site and surrounding properties. The Site consists of an approximately 0.79-acre area property and is located at 500, 506, and 510 Main Street and 12 Church Street, New Rochelle, Westchester County, New York (Site). The Site comprises 4 contiguous parcels and is identified on the Westchester County Clerk's map as tax parcels 1-215-0012, 1-215-0011, 1-215-0010, and 1-215-0008. The Site is improved with four structures. The Site buildings' operations most recently included two churches, retail stores, and professional offices.

# 1.3 Policy Statement

The policy of SESI Consulting Engineers (SESI) is to provide a safe and healthful work environment. No aspect of operations is of greater importance than injury and illness prevention. A fundamental principle of safety management is that all injuries, illnesses, and incidents are preventable. SESI will take every reasonable step to eliminate or control hazards in order to minimize the possibility of injury, illness, or incident.

This HASP prescribes the procedures that must be followed by SESI personnel during activities at the site. Operational changes that could affect the health and safety of personnel, the community, or the environment will not be made without the prior approval of the Project Manager (PM) and the Health and Safety Manager (HSM). This document will be reviewed periodically by the HSM to ensure that it is current and technically correct. Any changes in site conditions and/or the scope of work will require a review and modification to this HASP. Such changes will be completed in the form of an addendum or a revision to the plan.

The provisions of this plan are mandatory for all SESI personnel and are advisory for all contractors, and subcontractors assigned to the project. Subcontractors will be responsible for preparing their own site-specific HASPs that meet the basic

9

*requirements outlined in this HASP.* All visitors to SESI work areas at the site must abide by the requirements of this plan.

# 1.4 References

This HASP complies with applicable Occupational Safety and Health Administration (OSHA) regulations, United States Environmental Protection Agency (USEPA) regulations, and SESI health and safety policies and procedures. This plan follows the guidelines established in the following:

- Standard Operating Safety Guides, USEPA (Publication 9285.1-03, June 1992).
- Occupational Safety and Health Guidance Manual for Hazardous Waste Site Activities, NIOSH, OSHA, USCG, USEPA (86116, October 1985).
- *Title 29 of the Code of Federal Regulations* (CFR), Part 1910.
- *Title 29 of the Code of Federal Regulations* (CFR), Part 1926.
- Pocket Guide to Chemical Hazards, DHHS, PHS, CDC, NIOSH (2004).
- Threshold Limit Values, ACGIH (2005).
- Guide to Occupational Exposure Values, ACGIH (2005).
- Quick Selection Guide to Chemical Protective Clothing, Forsberg, K. and S.Z. Mansdorf, 2nd Ed. (1993).

# 1.5 Definitions

The following definitions (listed alphabetically) are applicable to this HASP:

- Contamination Reduction Zone (CRZ) Area between the exclusion zone and support zone that provides a transition between contaminated and clean areas. Decontamination stations are located in this zone.
- *Exclusion Zone (EZ)* Any portions of the site where hazardous substances are, or are reasonably suspected to be present, and pose an exposure hazard to on-site personnel.
- *Incident* All losses, including first aid cases, injuries, illnesses, spills/leaks, equipment and property damage, motor vehicle accidents, regulatory violations, fires, and business interruptions.
- On-Site Personnel All SESI and subcontractors involved with the project.
- *Project* All on-site work performed under the scope of work.
- *Site* The area described in Section 1.2, Site and Facility Description, where the work is to be performed by SESI personnel and subcontractors.
- Support Zone (SZ) All areas of the site except the EZ and CRZ. The SZ surrounds the CRZ and EZ. Support equipment and break areas are located in this zone.
- Subcontractor Includes contractor personnel hired by SESI.
- *Visitor* All other personnel, except the on-site personnel.
- *Work Area* The portion of the site where work activities are actively being performed. This area may change daily as work progresses and includes the SZ, CRZ, and EZ. If the work area is located in an area on the site that is not contaminated, or suspected of being contaminated, the entire work area may be a SZ.

# 2.0 PROJECT SCOPE OF WORK

This HASP contains information for the following tasks that SESI is anticipated to conduct at the Site. Should additional and/or different tasks be identified, amendments to this HASP will be required to address these changed items.

- Mobilization;
- Excavation of Contaminated Soil;
- End Point Chemical sampling of soil;
- Installation of a Vapor Barrier for Buildings;
- Groundwater MNA Sampling;
- Decontamination and demobilization.

# 3.0 ROLES AND RESPONSIBILITIES

# 3.1 All Personnel

All SESI project personnel must adhere to the procedures outlined in this HASP during the performance of their work. Each person is responsible for completing tasks safely and reporting any unsafe acts or conditions to their supervisor. No person may work in a manner that conflicts with these procedures. After due warnings, the PM will dismiss from the site any SESI employee or subcontractor who violates safety procedures.

All SESI project personnel will receive training in accordance with applicable regulations, and be familiar with the requirements and procedures contained in this HASP prior to initiating site activities. In addition, all SESI personnel will attend an initial hazard briefing prior to beginning work at the site.

The roles of key safety personnel and subcontractors are outlined in the following sections. Key project personnel and contacts are summarized in Table 1.

# 3.2 Key Safety Personnel

# 3.2.1 **Project Officer (PO)**

The PO is responsible for providing resources to assure project activities are completed in accordance with this HASP, and for meeting all regulatory and contractual requirements.

# 3.2.2 Project Manager (PM)

The PM is responsible for verifying that project activities are completed in accordance with the requirements of this HASP. The PM is responsible for confirming that the Field Supervisor (FS) has the equipment, materials, and qualified personnel to fully implement the safety requirements of this HASP, and/or that subcontractors assigned to this project meet the requirements established by SESI. It is also the responsibility of the PM to:

- Consult with the HSM on site health and safety issues;
- Verify that subcontractors meet health and safety requirements prior to commencing work;
- Verify that all incidents are thoroughly investigated;
- Approve, in writing, addenda or modifications of this HASP; and

• Suspend work or modify work practices, as necessary, for personal safety, protection of property, and regulatory compliance.

# 3.2.3 Health and Safety Manager (HSM)

The HSM or his designee, the health and safety manager (HSM), has overall responsibility for the technical health and safety aspects of the project, including review and approval of this HASP. Inquiries regarding health and safety procedures, project procedures, and other technical or regulatory issues should be addressed to this individual. The HSM or his designee must approve changes or addenda to this HASP.

# 3.2.4 Site Safety Officer (SSO)

The SSO is responsible for field health and safety issues, including the execution of this HASP. Questions in the field regarding health and safety procedures, project procedures, and other technical or regulatory issues should be addressed to this individual. The SSO will advise the PM on health and safety issues and will establish and coordinate the project air-monitoring program if one is deemed necessary (see Section 5.1, Air Monitoring). The SSO is the primary site contact on health and safety matters. It is the responsibility of the SSO to:

- Provide on-site technical assistance, if necessary;
- Participate in all accident/incident reports and ensure that they are reported to the HSM, client, and PM within 24 hours;
- Coordinate site and personal air monitoring as required, including equipment maintenance and calibration;
- Conduct site safety orientation training and safety meetings;
- Verify that project personnel have received the required physical examinations and medical certifications;
- Review site activities with respect to compliance with this HASP;
- Maintain required health and safety documents and records; and
- Assist the FS in instructing field personnel on project hazards and protective procedures.

# 3.2.5 Field Supervisor (FS)

The FS is responsible for implementing this HASP, including communicating requirements to on-site personnel and subcontractors. The FS will be responsible for informing the PM of changes in the work plan, procedures, or site conditions so that those changes may be addressed in this HASP. Other responsibilities are to:

- Consult with the SSO on site health and safety issues;
- Stop work, as necessary, for personal safety, protection of property, and regulatory compliance;
- Obtain a site map and determine and post routes to medical facilities and emergency telephone numbers;
- Notify local public emergency representatives (as appropriate) of the nature of the site operations, and post their telephone numbers (i.e., local fire department personnel who would respond for a confined space rescue);
- Observe on-site project personnel for signs of ill health effects;
- Investigate and report any incidents to the SSO;
- Verify that all on-site personnel have had applicable training;
- Verify that on-site personnel are informed of the physical, chemical, and biological hazards associated with the site activities, and the procedures and protective equipment necessary to control the hazards; and
- Issue/obtain any required work permits (hot work, confined space, etc.).

# 3.2.6 Field Personnel (FP)

All SESI field personnel are responsible for following the Health and Safety procedures specified in this HASP and work practices specified in applicable operation procedures. Some specific responsibilities include, but are not limited to:

- Reading and understanding the HASP;
- Reporting all accidents, incidents, injuries, or illnesses to the FS;
- Complying with the requests of the SSO;
- Immediately communicating newly identified hazards or noncompliance issues to the FS or SSO; and
- Stopping work in cases of immediate danger.

## 3.3 Subcontractors

Subcontractors and their personnel must understand and comply with applicable regulations and site requirements established in this HASP. Subcontractors will prepare their own site-specific HASP that must be consistent with the requirements of this HASP.

All subcontractor personnel will receive training in accordance with applicable regulations and be familiar with the requirements and procedures contained in this HASP prior to initiating site activities. All subcontractor personnel will attend an initial hazard briefing prior to beginning work at the site. Additionally, on-site subcontractor personnel must conduct daily site safety meetings.

Subcontractors must designate individuals to function as the PM, HSM, SSO, and FS. In some firms the HSM to be carried out by the PM. This is acceptable provided the PM has the required knowledge, training, and experience to properly address all hazards associated with the work, and to prepare, approve, and oversee the execution of the site-specific HASP. A subcontractor may designate the same person to perform the duties of both the SSO and the FS. However, depending on the level of complexity of a contractor's scope of work, it may be infeasible for one person to perform both functions satisfactorily.

## 3.4 Stop Work Authority

Every SESI employee and subcontractor is empowered, expected, and has the responsibility to stop the work of another co-worker if the working conditions or behaviors are considered unsafe.

# 3.5 All On-Site Personnel

All on-site SESI personnel (including SESI subcontractors) must read and acknowledge their understanding of their respective HASPs before commencing work and abide by the requirements of the plans. All on-site SESI personnel shall sign their HASP Acknowledgement Form following their review of their HASP.

All SESI project personnel will receive training in accordance with applicable regulations and be familiar with the requirements and procedures contained in this HASP prior to initiating site activities. In addition, all on-site personnel will attend an initial hazard briefing provided by the SSO prior to beginning work at the site and conduct daily safety meetings thereafter.

On-site personnel will immediately report the following to the FS or SSO:

- Personal injuries and illnesses no matter how minor;
- Unexpected or uncontrolled release of chemical substances;
- Symptoms of chemical exposure;
- Unsafe or hazardous situations;
- Unsafe or malfunctioning equipment;
- Changes in site conditions that may affect the health and safety of project personnel;
- Damage to equipment or property; and
- Situations or activities for which they are not properly trained.

#### 3.6 Visitors

All SESI personnel and subcontractors visiting the Site must check in with the FS. Visitors will be cautioned to avoid skin contact with surfaces, soils, groundwater, or other materials that may impacted or be suspected to be impacted by constituents of concern (COCs).

Visitors requesting to observe work at the site must don appropriate personal protective equipment (PPE) prior to entry to the work area and must have the appropriate training and medical clearances to do so. If respiratory protective devices are necessary, visitors who wish to enter the work area must have been respirator-trained and fit tested for a respirator within the past 12 months.

#### Table 1 – Key Safety Personnel

SESI Personnel					
Role	Name	Telephone No.			
Project Officer (PO)	Fuad Dahan	973-808-9050 x249			
Project Manager (PM)	Jesse Mausner	973-808-9050 x282			
Senior Project Engineer (SPE)	Fuad Dahan	973-808-9050 x249			
Health and Safety Manager (HSM)	Todd Kelly	973-808-9050 x238			
Site Safety Officer (SSO)	Todd Kelly	973-808-9050 x238			
Field Supervisor (FS)	Jon Stuart	973-600-7630			
Field Personnel	TBD				
Field Personnel	TBD				
Subcontractors					
Company/Role	Name	Telephone No.			
AARCO/Driller	TBD	631-586-5900			

## 4.0 PERSONAL PROTECTIVE EQUIPMENT

#### 4.1 Levels of Protection

PPE is required to safeguard site personnel from various hazards. Varying levels of protection may be required depending on the levels of COCs and the degree of physical hazard. This section presents the various levels of protection and defines the conditions of use for each level. A summary of the levels is presented in Table 2 in this section.

## 4.1.1 Level D Protection

The minimum level of protection that will be required of project personnel at the site will be Level D, which will be worn when site conditions or air monitoring indicates no inhalation hazard exists. The following equipment will be used:

- Work clothing as prescribed by weather;
- Steel toe work boots, meeting American National Standards Institute (ANSI) Z41;
- Safety glasses or goggles, meeting ANSI Z87;
- Leather work gloves and/or nitrile surgical gloves;
- Hard hat, meeting ANSI Z89, when falling object hazards are present;
- Hearing protection (if noise levels exceed 85 dBA, then hearing protection with a USEPA NRR of at least 20 dBA must be used); and
- PFD if working on or near the water.

## 4.1.2 Modified Level D Protection

Modified Level D will be used when airborne contaminants are not present at levels of concern, but site activities present an increased potential for skin contact with contaminated materials. Modified Level D consists of:

- Nitrile gloves worn over nitrile surgical gloves;
- Latex/polyvinyl chloride (PVC) overboots when contact with COC-impacted media is anticipated;
- Steel toe work boots, meeting ANSI Z41;
- Safety glasses or goggles, meeting ANSI Z87;

- Face shield in addition to safety glasses or goggles when projectiles or splash hazards exist (e.g. during Power Washing activities);
- Hard hat, meeting ANSI Z89, when falling object hazards are present;
- Hearing protection (if noise levels exceed 85 dBA, then hearing protection with a USEPA NRR of at least 20 dBA must be used);
- Tyvek<sup>®</sup> suit (polyethylene coated Tyvek<sup>®</sup> suits for handling liquids) when body contact with COC-impacted media is anticipated; and
- PFD if working on or near the water.

## 4.1.3 Level C Protection

Level C protection will be required when the airborne concentration of COC reaches onehalf of the OSHA Permissible Exposure Limit or ACGIH TLV. The following equipment will be used for Level C protection:

- Full-face, air-purifying respirator with combination organic vapor/HEPA cartridges;
- Polyethylene-coated Tyvek<sup>®</sup> suit, with ankles and cuffs taped to boots and gloves;
- Nitrile gloves worn over nitrile surgical gloves;
- Steel toe work boots, meeting ANSI Z41;
- Chemical-resistant boots with steel toes or latex/PVC overboots over steel toe boots;
- Hard hat, meeting ANSI Z89;
- Hearing protection (if noise levels exceed 85 dBA, then hearing protection with a USEPA NRR of at least 20 dBA must be used); and
- PFD if working on or near the water.

# 4.2 Selection of PPE

Equipment for personal protection will be selected based on the potential for contact, site conditions, ambient air quality, and the judgment of supervising site personnel and health and safety professionals. The PPE used will be chosen to be effective against the COCs present on the site.

# 4.3 Site Respiratory Protection Program

Respiratory protection is an integral part of employee health and safety at the site due to potentially hazardous concentrations of airborne COCs. The site respiratory protection program will consist of the following (as a minimum):

- All on-site personnel who may use respiratory protection will have an assigned respirator.
- All on-site personnel who may use respiratory protection will have been fit tested and trained in the use of a full-face air-purifying respirator within the past 12 months. Documentation of the fit test must be provided to the SSO prior to commencement of work.
- All on-site personnel who may use respiratory protection must within the past year have been medically certified as being capable of wearing a respirator. Documentation of the medical certification must be provided to the SSO, prior to commencement of site work.
- Only cleaned, maintained, NIOSH-approved respirators will be used.
- If respirators are used, the respirator cartridge is to be properly disposed of at the end of each work shift, or when load-up or breakthrough occurs.

- Contact lenses are not to be worn when a respirator is worn.
- All on-site personnel who may use respiratory protection must be clean-shaven. Mustaches and sideburns are permitted, but they must not touch the sealing surface of the respirator.
- Respirators will be inspected, and a negative pressure test performed prior to each use.
- After each use, the respirator will be wiped with a disinfectant, cleansing wipe. When used, the respirator will be thoroughly cleaned at the end of the work shift. The respirator will be stored in a clean plastic bag, away from direct sunlight in a clean, dry location, in a manner that will not distort the face piece.

# 4.4 Using PPE

Depending upon the level of protection selected, specific donning and doffing procedures may be required. The procedures presented in this section are mandatory if Modified Level D or Level C PPE is used. All personnel entering the EZ must put on the required PPE in accordance with the requirements of this HASP. When leaving the EZ, PPE will be removed in accordance with the procedures listed, to minimize the spread of COCs.

# 4.4.1 Donning Procedures

These procedures are mandatory only if Modified Level D or Level C PPE is used on the site:

- Remove bulky outerwear. Remove street clothes and store in clean location;
- Put on work clothes or coveralls;
- Put on the required chemical protective coveralls;
- Put on the required chemical protective boots or boot covers;
- Tape the legs of the coveralls to the boots with duct tape;
- Put on the required chemical protective gloves;
- Tape the wrists of the protective coveralls to the gloves;
- Don the required respirator and perform appropriate fit check (Level C);
- Put hood or head covering overhead and respirator straps and tape hood to facepiece (Level C); and
- Don remaining PPE, such as safety glasses or goggles and hard hat.

When these procedures are instituted, one person must remain outside the work area to ensure that each person entering has the proper protective equipment.

## 4.4.2 Doffing Procedures

The following procedures are only mandatory if Modified Level D or Level C PPE is required for the site. Whenever a person leaves the work area, the following decontamination sequence will be followed:

- Upon entering the CRZ, rinse contaminated materials from the boots or remove contaminated boot covers;
- Clean reusable protective equipment;
- Remove protective garments, equipment, and respirator (Level C). All disposable clothing should be placed in plastic bags, which are labeled with contaminated waste labels;
- Wash hands, face, and neck (or shower if necessary);

- Proceed to clean area and dress in clean clothing; and
- Clean and disinfect respirator for next use.

All disposable equipment, garments, and PPE must be bagged in plastic bags, labeled for disposal. See Section 7, Decontamination, for detailed information on decontamination stations.

#### 4.5 Selection Matrix

The level of personal protection selected will be based on air monitoring of the work environment and an assessment by the FS and SSO of the potential for skin contact with COCs. The PPE selection matrix is presented in Table 2. This matrix is based on information available at the time this plan was written. The Airborne Contaminant Action Levels in Table 3, Airborne Contaminant Action Levels, should be used to verify that the PPE prescribed in these matrices is appropriate.

#### Table 2 – PPE Selection Matrix

Task	Anticipated Level of Protection
Soil Excavation	Modified Level D/Level C
Chemical Sampling	Modified Level D/Level C
Decontamination	Modified Level D

## 5.0 AIR AND NOISE MONITORING

#### 5.1 Air Monitoring

Air monitoring, sampling, and testing will be conducted to determine employee exposure to airborne constituents. The monitoring results will dictate work procedures and the selection of PPE. The SESI SSO will be responsible for defining appropriate air monitoring procedures and for utilizing the air monitoring results to determine appropriate procedures and PPE for project personnel. Air monitoring results should be recorded in field notebooks or on an air monitoring log (see Attachment 1 for a copy of the Air Monitoring Log). Any deviations from the procedures listed here should be documented and explained in the Air Monitoring Log.

The monitoring devices to be used are a PDR1000 particulate monitor (or equivalent) and a Rae Systems MultiRAE detector (PID with a 11.7 eV lamp/oxygen/LEL/hydrogen sulfide sensors). Colorimetric detector tubes may be utilized to estimate airborne concentrations of benzene and should be onsite during any activities that may result in elevated PID readings including drilling, excavating, and groundwater sampling.

Air monitoring will be conducted continuously with the LEL/Oxygen meter during drilling in areas where flammable vapors or gases are suspect. All work activity must stop where tests indicate the concentration of flammable vapors exceeds 10% of the LEL at a location with a potential ignition source. Such an area must be ventilated to reduce the concentration to an acceptable level.

## 5.2 Noise Monitoring

Noise monitoring may be conducted as required. Hearing protection is mandatory for all employees in noise hazardous areas, such as around heavy equipment. As a general rule, sound levels that cause speech interference at normal conversation distance should require the use of hearing protection.

## 5.3 Monitoring Equipment Maintenance and Calibration

All direct-reading instrumentation calibrations should be conducted under the approximate environmental conditions the instrument will be used. Instruments must be calibrated before and after use, noting the reading(s) and any adjustments that are necessary. All air monitoring equipment calibrations, including the standard used for calibration, must be documented on a calibration log or in the field notebook. All completed health and safety documentation/forms must be reviewed by the SSO and maintained by the FS.

All air monitoring equipment will be maintained and calibrated in accordance with the specific manufacturer's procedures. Preventive maintenance and repairs will be conducted in accordance with the respective manufacturer's procedures. When applicable, only manufacturer-trained and/or authorized personnel will be allowed to perform instrument repairs or preventive maintenance.

If an instrument is found to be inoperative or suspected of giving erroneous readings, the SSO must be responsible for immediately removing the instrument from service and obtaining a replacement unit. If the instrument is essential for safe operation during a specific activity, that activity must cease until an appropriate replacement unit is obtained. The SSO will be responsible for ensuring a replacement unit is obtained and/or repairs are initiated on the defective equipment.

## 5.4 Action Levels

Table 3 presents airborne contaminant action levels that will be used to determine the procedures and protective equipment necessary based on conditions as measured at the site.

|--|

Parameter	Reading	Action
Total	$0 \text{ ppm to } \leq 1 \text{ ppm}$	Normal operations: continue bourly breathing zone monitoring
Hydrocarbons	$0$ ppin to $\underline{<}$ 1 ppin	
Tryulocarbons	> 1 ppm to 5 ppm	Increase menitoring frequency to every 15 minutes and use
		honzono detector tube to corect for the presence of honzono
	> E nom to $<$ E0 nom	Ungrade to Lovel C DDE, continue coreoning for henzone
	$\geq$ 5 ppm to $\leq$ 50 ppm	Opgrade to Level C PPE, continue screening for benzene
	> 50 nom	Sten work investigate agues of reading
	> 50 ppm	Stop work, investigate cause of reading
Benzene	$\geq$ 1 ppm to 5 ppm	Upgrade to Level C PPE
	> 5 ppm	Stop work; investigate cause of reading
Dust	0 to .5 mg/m3	Normal operations
	0.5 to 1 mg/m3	Begin soil wetting procedure (Level C protection would be needed
		beyond this point)
	> 1 mg/m3	Stop work, fully implement dust control plan
Oxygen	<u>&lt;</u> 19.5%	Stop work, evacuate confined spaces/work area, investigate cause
		of reading, and ventilate area
	> 19.5% to < 23.5%	Normal operations
	<u>&gt;</u> 23.5%	Stop work, evacuate confined spaces/work area, investigate cause
		of reading, and ventilate area
Carbon	0 ppm to <u>&lt;</u> 20 ppm	Normal operations
Monoxide		
	> 20 ppm	Stop work, evacuate confined spaces/work area, investigate cause
		of reading, and ventilate area
Hydrogen	0 ppm to <u>&lt;</u> 5 ppm	Normal operations
Sulfide		
	> 5 ppm	Stop work, evacuate confined spaces/work area, investigate cause
		of reading, and ventilate area
Flammable	< 10% LEL	Normal operations
Vapors (LEL)		
	> 10% LEL	Stop work, ventilate area, investigate source of vapors

## 6.0 WORK ZONES AND DECONTAMINATION

## 6.1 Work Zones

## 6.1.1 Authorization to Enter

Only personnel with the appropriate training and medical certifications (if respirators are required) will be allowed to work at the project site. The FS will maintain a list of authorized persons; only personnel on the authorized persons list will be allowed to enter the site work areas.

## 6.1.2 Site Orientation and Hazard Briefing

No person will be allowed in the work area during site operations without first being given a site orientation and hazard briefing. This orientation will be presented by the FS or SSO and will consist of a review of this HASP. This review must cover the chemical, physical, and biological hazards, protective equipment, safe work procedures, and emergency procedures for the project. Following this initial meeting, daily safety meetings will be held each day before work begins.

All people entering the site work areas, including visitors, must document their attendance at this briefing, as well as the daily safety meetings on the forms included with this plan.

# 6.1.3 Certification Documents

A training and medical file may be established for the project and kept on site during all site operations. Specialty training, such as first aid/cardiopulmonary resuscitation (CPR) certificates, as well as current medical clearances for all project field personnel required to wear respirators, will be maintained within that file. All project personnel must provide their training and medical documentation to the SSO prior to starting work.

# 6.1.4 Entry Log

A log-in/log-out sheet will be maintained at the site by the FS. Personnel must sign in and out on a log sheet as they enter and leave the work area, and the FS may document entry and exit in the field notebook.

# 6.1.5 Entry Requirements

In addition to the authorization, hazard briefing, and certification requirements listed above, no person will be allowed in any SESI work area unless they are wearing the minimum PPE as described in Section 4.0.

# 6.1.6 Emergency Entry and Exit

People who must enter the work area on an emergency basis will be briefed of the hazards by the FS or SSO. All activities will cease in the event of an emergency. People exiting the work area because of an emergency will gather in a designated safe area for a head count. The FS is responsible for ensuring that all people who entered the work area have exited in the event of an emergency.

## 6.1.7 Contamination Control Zones

Contamination control zones are maintained to prevent the spread of contamination and to prevent unauthorized people from entering hazardous areas.

# 6.1.8 Exclusion Zone (EZ)

An EZ may consist of a specific work area or may be the entire area of potential contamination. All employees entering an EZ must use the required PPE and must have the appropriate training and medical clearance for hazardous waste work. The EZ is the defined area where there is a possible respiratory and/or contact health hazard. Cones, caution tape, or a posted site diagram will identify the location of each EZ.

# 6.1.9 Contamination Reduction Zone

The CRZ or transition area will be established, if necessary, to perform decontamination of personnel and equipment. All personnel entering or leaving the EZ will pass through this area to prevent any cross-contamination. Tools, equipment, and machinery will be decontaminated in a specific location. The decontamination of all personnel will be performed on site adjacent to the EZ. Personal protective outer garments and respiratory protection will be removed in the CRZ and prepared for cleaning or disposal. This zone is the only appropriate corridor between the EZ and the support zone (SZ) discussed below.

## 6.1.10 Support Zone (SZ)

The SZ is a clean area outside the CRZ located to prevent employee exposure to hazardous substances. Eating and drinking will be permitted in the support area only after proper decontamination. Smoking may be permitted in the SZ, subject to site requirements.

## 6.1.11 Posting

Work areas will be prominently marked and delineated using cones, caution tape, or a posted site diagram.

## 6.1.12 Site Inspections

The FS will conduct a daily inspection of site activities, equipment, and procedures to verify that the required elements are in place.

## 6.2 Decontamination

## 6.2.1 Personnel Decontamination

All personnel wearing Modified Level D or Level C protective equipment in the EZ must undergo personal decontamination prior to entering the SZ. The personnel decontamination area will consist of the following stations at a minimum:

- *Station 1*: Personnel leaving the contaminated zone will remove the gross contamination from their outer clothing and boots.
- *Station 2*: Personnel will remove their outer garment and gloves and dispose of it in properly labeled containers. Personnel will then decontaminate their hard hats, and boots with an aqueous solution of detergent or other appropriate cleaning solution. These items are then hand carried to the next station.
- *Station 3*: Personnel will thoroughly wash their hands and face before leaving the CRZ. Respirators will be sanitized and then placed in a clean plastic bag.

## 6.2.2 Equipment Decontamination

All vehicles that have entered the EZ will be decontaminated at the decontamination pad prior to leaving the zone. If the level of vehicle contamination is low, decontamination may be limited to rinsing of tires and wheel wells with water. If the vehicle is significantly contaminated, steam cleaning or pressure washing of vehicles and equipment may be required.

## 6.2.3 Personal Protective Equipment Decontamination

Where and whenever possible, single-use, external protective clothing must be used for work within the EZ or CRZ. This protective clothing must be disposed of in properly labeled containers. Reusable protective clothing will be rinsed at the site with detergent and water. The rinsate will be collected for disposal.

When removed from the CRZ, the respirator will be thoroughly cleaned with soap and water. The respirator face piece, straps, valves, and covers must be thoroughly cleaned at the end of each work shift, and ready for use prior to the next shift. Respirator parts may be disinfected with a solution of bleach and water (mixed at 2% bleach by volume), or by using a spray disinfectant.

## 7.0 TRAINING AND MEDICAL SURVEILLANCE

## 7.1 Training

## 7.1.1 General

All on-site project personnel who work in areas where they may be exposed to site contaminants must be trained as required by OSHA Regulation 29 CFR 1910.120 (HAZWOPER). Field employees also must receive a minimum of three days of actual field experience under the direct supervision of a trained, experienced supervisor. Personnel who completed their initial training more than 12 months prior to the start of the project must have completed an eight-hour refresher course within the past 12 months. The FS must have completed an additional eight hours of supervisory training and must have a current first-aid/CPR certificate (See Attachment 2).

## 7.1.2 Basic 40-Hour Course

The following is a list of the topics typically covered in a 40-hour HAZWOPER training course:

- General safety procedures;
- Physical hazards (fall protection, noise, heat stress, cold stress);
- Names and job descriptions of key personnel responsible for site health and safety;
- Safety, health, and other hazards typically present at hazardous waste sites;
- Use, application, and limitations of PPE;
- Work practices by which employees can minimize risks from hazards;
- Safe use of engineering controls and equipment on site;
- Medical surveillance requirements;
- Recognition of symptoms and signs which might indicate overexposure to hazards;
- Worker right-to-know (Hazard Communication OSHA 1910.1200);
- Routes of exposure to contaminants;
- Engineering controls and safe work practices;
- Components of a health and safety program and a site-specific HASP;
- Decontamination practices for personnel and equipment;
- Confined-space entry procedures; and
- General emergency response procedures.

## 7.1.3 Supervisor Course

Management and supervisors must receive an additional eight hours of training, which typically includes:

- General site safety and health procedures;
- PPE programs; and
- Air monitoring techniques.

# 7.1.4 Site-Specific Training

Site-specific training will be accomplished by on-site personnel reading this HASP, and through a thorough site briefing by the PM, FS, or SSO on the contents of this HASP before work begins. The review must include a discussion of the chemical, physical, and

biological hazards; the protective equipment and safety procedures; and emergency procedures.

# 7.1.5 Daily Safety Meetings

Daily safety meetings will be held to cover the work to be accomplished, the hazards anticipated, the PPE and procedures required to minimize site hazards, and emergency procedures. The FS or SSO should present these meetings prior to beginning the day's fieldwork. No work will be performed in an EZ before a daily safety meeting has been held. An additional safety meeting must also be held prior to new tasks, or if new hazards are encountered. The daily safety meetings will be logged in the field notebook.

## 7.1.6 First Aid and CPR

At least one employee current in first aid/CPR will be assigned to the work crew and will be on the site during operations. Site records will document the presence of this individual. Refresher training in first aid (triennially) and CPR (annually) is required to keep the certificate current. These individuals must also receive training regarding the precautions and protective equipment necessary to protect against exposure to bloodborne pathogens.

# 7.2 Medical Surveillance

# 7.2.1 Medical Examination

All personnel who are potentially exposed to site contaminants must participate in a medical surveillance program as defined by OSHA at 29 CFR 1910.120 (f).

## 7.2.2 Pre-placement Medical Examination

All potentially exposed personnel must have completed a comprehensive medical examination prior to assignment, and periodically thereafter as defined by applicable regulations. The pre-placement and periodic medical examinations typically include the following elements:

- Medical and occupational history questionnaire;
- Physical examination;
- Complete blood count, with differential;
- Liver enzyme profile;
- Chest X-ray, at a frequency determined by the physician;
- Pulmonary function test;
- Audiogram;
- Electrocardiogram for persons older than 45 years of age, or if indicated during the physical examination;
- Drug and alcohol screening, as required by job assignment;
- Visual acuity; and
- Follow-up examinations, at the discretion of the examining physician or the corporate medical director.

The examining physician provides the employee with a letter summarizing his findings and recommendations, confirming the worker's fitness for work and ability to wear a respirator. Documentation of medical clearance will be available for each employee during all project site work.

Subcontractors will certify that all their employees have successfully completed a physical examination by a qualified physician. The physical examinations must meet the requirements of 29 CFR 1910.120 and 29 CFR 1910.134. Subcontractors will supply copies of the medical examination certificate for each on-site employee.

## 7.2.3 Other Medical Examinations

In addition to pre-employment, annual, and exit physicals, personnel may be examined:

- At employee request after known or suspected exposure to toxic or hazardous materials; and
- At the discretion of the SSO, HSM, or occupational physician in anticipation of, or after known or suspected exposure to toxic or hazardous materials.

## 7.2.4 Periodic Exam

Following the placement examination, all employees must undergo a periodic examination, similar in scope to the placement examination. For employees potentially exposed over 30 days per year, the frequency of periodic examinations will be annual. For employees potentially exposed less than 30 days per year, the frequency for periodic examinations will be 24 months.

## 7.2.5 Medical Restriction

When the examining physician identifies a need to restrict work activity, the employee's supervisor must communicate the restriction to the employee and the SSO. The terms of the restriction will be discussed with the employee and the supervisor.

## 8.0 GENERAL SAFETY PRACTICES

## 8.1 General Safety Rules

General safety rules for site activities include, but are not limited to, the following:

- At least one copy of this HASP must be in a location at the site that is readily available to personnel, and all project personnel shall review the plan prior to starting work.
- Consume or use food, beverages, chewing gum, and tobacco products only in the SZ or other designated area outside the EZ and CRZ. Cosmetics shall not be applied in the EZ or CRZ.
- Wash hands before eating, drinking, smoking, or using toilet facilities.
- Wear all PPE as required and stop work and replace damaged PPE immediately.
- Secure disposable coveralls, boots, and gloves at the wrists and legs and ensure closure of the suit around the neck.
- Upon skin contact with materials that may be impacted by COCs, remove contaminated clothing and wash the affected area immediately. Contaminated clothing must be changed. Any skin contact with materials potentially impacted by COCs must be reported to the FS or SSO immediately. If needed, medical attention should be sought.
- Practice contamination avoidance. Avoid contact with surfaces either suspected or known to be impacted by COCs, such as standing water, mud, or discolored

soil. Equipment must be stored on elevated or protected surfaces to reduce the potential for incidental contamination.

- Remove PPE as required in the CRZ to limit the spread of COC-containing materials.
- At the end of each shift or as required, dispose of all single-use coveralls, soiled gloves, and respirator cartridges in designated receptacles designated for this purpose.
- Removing soil containing site COCs from protective clothing or equipment with compressed air, shaking, or any other means that disperses contaminants into the air is prohibited.
- Inspect all non-disposable PPE for contamination in the CRZ. Any PPE found to be contaminated must be decontaminated or disposed of appropriately.
- Recognize emergency signals used for evacuation, injury, fire, etc.
- Report all injuries, illnesses, and unsafe conditions or work practices to the FS or SSO.
- Use the "buddy system" during all operations requiring Level C PPE, and when appropriate, during Modified Level D operations.
- Obey all warning signs, tags, and barriers. Do not remove any warnings unless authorized to do so.
- Use, adjust, alter, and repair equipment only if trained and authorized to do so, and in accordance with the manufacturer's directions.
- Personnel are to perform only tasks for which they have been properly trained and will advise their supervisor if they have been assigned a task for which they are not trained.
- The presence or consumption of alcoholic beverages or illicit drugs during the workday, including breaks, is strictly prohibited. Notify your supervisor if you must take prescription or over-the-counter drugs that indicate they may cause drowsiness or, that you should not operate heavy equipment.
- Remain upwind during site activities whenever possible.

## 8.2 Buddy System

On-site personnel must use the buddy system as required by operations. Use of the "buddy system" is required during all operations requiring Level C to Level A PPE, and when appropriate, during Level D operations. Crewmembers must observe each other for signs of chemical exposure, and heat or cold stress. Indications of adverse effects include, but are not limited to:

- Changes in complexion and skin coloration;
- Changes in coordination;
- Changes in demeanor;
- Excessive salivation and pupillary response; and
- Changes in speech pattern.

Crewmembers must also be aware of the potential exposure to possible safety hazards, unsafe acts, or non-compliance with safety procedures.

Field personnel must inform their partners or fellow crewmembers of non-visible effects of exposure to toxic materials that they may be experiencing. The symptoms of such exposure may include, but are not limited to:

- Headaches;
- Dizziness;
- Nausea;
- Blurred vision;
- Cramps; and
- Irritation of eyes, skin, or respiratory tract.

If protective equipment or noise levels impair communications, prearranged hand signals must be used for communication. Personnel must stay within line of sight of another team member.

## 8.3 Heat Stress

Heat stress is caused by a number of interacting factors, including environmental conditions, clothing, workload, etc., as well as the physical and conditioning characteristics of the individual. Since heat stress is one of the most common illnesses associated with heavy outdoor work conducted with direct solar load and, in particular, because wearing PPE can increase the risk of developing heat stress, workers must be capable of recognizing the signs and symptoms of heat-related illnesses. Personnel must be aware of the types and causes of heat-related illnesses and be able to recognize the signs and symptoms of these illnesses in both themselves and their co-workers.

*Heat rashes* are one of the most common problems in hot work environments. Commonly known as prickly heat, a heat rash is manifested as red papules and usually appears in areas where the clothing is restrictive. As sweating increases, these papules give rise to a prickling sensation. Prickly heat occurs in skin that is persistently wetted by unevaporated sweat, and heat rash papules may become infected if they are not treated. In most cases, heat rashes will disappear when the affected individual returns to a cool environment.

*Heat cramps* are usually caused by performing hard physical labor in a hot environment. These cramps have been attributed to an electrolyte imbalance caused by sweating. It is important to understand that cramps can be caused both by too much or too little salt.

Cramps appear to be caused by the lack of water replenishment. Because sweat is a hypotonic solution (plus or minus 0.3% NaCl), excess salt can build up in the body if the water lost through sweating is not replaced. Thirst cannot be relied on as a guide to the need for water; instead, water must be taken every 15 to 20 minutes in hot environments.

Under extreme conditions, such as working for 6 to 8 hours in heavy protective gear, a loss of sodium may occur. Drinking commercially available carbohydrate electrolyte replacement liquids is effective in minimizing physiological disturbances during recovery.

*Heat exhaustion* occurs from increased stress on various body organs due to inadequate blood circulation, cardiovascular insufficiency, or dehydration. Signs and symptoms include pale, cool, moist skin; heavy sweating; dizziness; nausea; headache, vertigo, weakness, thirst, and giddiness. Fortunately, this condition responds readily to prompt treatment.

Heat exhaustion should not be dismissed lightly, however, for several reasons. One is that the fainting associated with heat exhaustion can be dangerous because the victim

may be operating machinery or controlling an operation that should not be left unattended; moreover, the victim may be injured when he or she faints. Also, the signs and symptoms seen in heat exhaustion are similar to those of heat stroke, which is a medical emergency.

Workers suffering from heat exhaustion should be removed from the hot environment, be given fluid replacement, and be encouraged to get adequate rest.

*Heat stroke* is the most serious form of heat stress. Heat stroke occurs when the body's system of temperature regulation fails and the body's temperature rises to critical levels. This condition is caused by a combination of highly variable factors, and its occurrence is difficult to predict. Heat stroke is a medical emergency. The primary signs and symptoms of heat stroke are confusion; irrational behavior; loss of consciousness; convulsions; a lack of sweating (usually); hot, dry skin; and an abnormally high body temperature, e.g., a rectal temperature of 41°C (105.8°F). If body temperature is too high, it causes death. The elevated metabolic temperatures caused by a combination of workload and environmental heat load, both of which contribute to heat stroke, are also highly variable and difficult to predict.

If a worker shows signs of possible heat stroke, professional medical treatment should be obtained immediately. The worker should be placed in a shady area and the outer clothing should be removed. The worker's skin should be wetted and air movement around the worker should be increased to improve evaporative cooling until professional methods of cooling are initiated and the seriousness of the condition can be assessed. Fluids should be replaced as soon as possible. The medical outcome of an episode of heat stroke depends on the victim's physical fitness and the timing and effectiveness of first aid treatment.

Regardless of the worker's protestations, no employee suspected of being ill from heat stroke should be sent home or left unattended unless a physician has specifically approved such an order.

Proper training and preventive measures will help avert serious illness and loss of work productivity. Preventing heat stress is particularly important because once someone suffers from heat stroke or exhaustion, that person may be predisposed to additional heat injuries.

## 8.4 Heat Stress Safety Precautions

Heat stress monitoring and work rest cycle implementation should commence when the ambient adjusted temperature exceeds 72°F. A minimum work rest regimen and procedures for calculating ambient adjusted temperature are described in Table 4.

Adjusted Temperature <sup>b</sup>	Work/Rest Regimen Normal Work Ensemble <sup>c</sup>	Work/Rest Regimen Impermeable Ensemble
90°F (32.2°C) or above	After each 45 minutes of work	After each 15 minutes of work
87.5° - 90°F (30.8°-32.2°C)	After each 60 minutes of work	After each 30 minutes of work

Table 4	- W	ork/Rest	Schedule

Adjusted Temperature <sup>b</sup>	Work/Rest Regimen Normal Work Ensemble <sup>c</sup>	Work/Rest Regimen Impermeable Ensemble
82.5° - 87.5°F (28.1° -	After each 90 minutes of	After each 60 minutes of
30.8°C)	work	work
77.5° - 82.5°F (25.3° -	After each 120 minutes of	After each 90 minutes of
28.1°C)	work	work
72.5° - 77.5°F (30.8° -	After each 150 minutes of	After each 120 minutes of
32.2°C)	work	work

a. For work levels of 250 kilocalories/hour (Light-Moderate Type of Work)

b. Calculate the adjusted air temperature (ta adj) by using this equation: ta adj °F = ta °F + (13 x % sunshine). Measure air temperature (ta) with a standard mercury-in-glass thermometer, with the bulb shielded from radiant heat. Estimate percent sunshine by judging what percent time the sun is not covered by clouds that are thick enough to produce a shadow. (100 percent sunshine = no cloud cover and a sharp, distinct shadow; 0 percent sunshine = no shadows.)

c. A normal work ensemble consists of cotton coveralls or other cotton clothing with long sleeves and pants.

d. The information presented above was generated using the information provided in the American Conference of Governmental Industrial Hygienists (ACGIH) Threshold Limit Values (TLV) Handbook.

In order to determine if the work rest cycles are adequate for the personnel and specific site conditions, additional monitoring of individual heart rates will be conducted during the rest cycle. To check the heart rate, count the radial pulse for 30 seconds at the beginning of the rest period. If the heart rate exceeds 110 beats per minute, shorten the next work period by one third and maintain the same rest period.

Additionally, one or more of the following control measures can be used to help control heat stress and are mandatory if any site worker has a heart rate (measure immediately prior to rest period) exceeding 115 beats per minute:

- Site workers will be encouraged to drink plenty of water and electrolyte replacement fluids throughout the day.
- On-site drinking water will be kept cool (50 to 60°F).
- A work regimen that will provide adequate rest periods for cooling down will be established, as required.
- All personnel will be advised of the dangers and symptoms of heat stroke, heat exhaustion, and heat cramps.
- Cooling devices, such as vortex tubes or cooling vests, should be used when personnel must wear impermeable clothing in conditions of extreme heat.
- Employees should be instructed to monitor themselves and co-workers for signs of heat stress and to take additional breaks as necessary.
- A shaded rest area must be provided. All breaks should take place in the shaded rest area.
- Employees must not be assigned to other tasks during breaks.
- Employees must remove impermeable garments during rest periods. This includes white Tyvek-type garments.

All employees must be informed of the importance of adequate rest, acclimation, and proper diet in the prevention of heat stress disorders.

## 8.5 Cold Stress

Cold stress normally occurs in temperatures at or below freezing, or under certain circumstances, in temperatures of 40°F. Extreme cold for a short time may cause severe injury to exposed body surfaces or result in profound generalized cooling, causing death. Areas of the body that have high surface area-to-volume ratio, such as fingers, toes, and

ears, are the most susceptible. Two factors influence the development of a cold weather injury: ambient temperature and the velocity of the wind. For instance, 10°F with a wind of 15 miles per hour (mph) is equivalent in chilling effect to still air at 18°F. An equivalent chill temperature chart relating the actual dry bulb temperature and wind velocity is presented in Table 5.

	Actua	al Tempe	erature F	Reading	(°F)								
Estimated Wind	50	40	30	20	10	0	-10	-20	-30	-40	-50	-60	
Speed (in mph)													
	Equiv	valent Ch	nill Temj	perature	(°F)								
Calm	50	40	30	20	10	0	-10	-20	-30	-40	-50	-60	
5	48	37	27	16	6	-5	-15	-26	-36	-47	-57	-68	
10	40	28	16	4	-9	-24	-33	-46	-58	-70	-83	-95	
15	36	22	9	-5	-18	-32	-45	-58	-72	-85	-99	-112	
20	32	18	4	-10	-25	-39	-53	-67	-82	-96	-110	-121	
25	30	16	0	-15	-29	-44	-59	-74	-88	-104	-118	-133	
30	28	13	-2	-18	-33	-48	-63	-79	-94	-109	-125	-140	
35	27	11	-4	-20	-35	-51	-67	-82	-98	-113	-129	-145	
40	26	10	-6	-21	-37	-53	-69	-85	-100	-116	-132	-148	
(Wind speeds	LITTL	E DANG	ER		INCR	EASING	DANGER	GREA	T DANG	ER			
greater than 40	Maxi	mum dar	nger of f	alse	Dang	er from f	freezing of	Flesh	may fre	eze witł	nin 30		
mph have little	sens	e of secu	ırity.		expos	sed flesh	n within	secor	ıds.				
additional effect.)					one n	ninute.							
	Trop	ah faat a		aralan fa	at may a		any naint a	n 4h in n	hort				

#### Table 5 – Wind Chill Temperature Chart

Trench foot and immersion foot may occur at any point on this chart.

[This chart was developed by the U.S. Army Research Institute of Environmental Medicine, Natick, MA (Source: ACGIH Threshold Limit Values for Chemical Substances and Physical Agents)].

Local injury resulting from cold is included in the generic term frostbite. There are several degrees of tissue damage associated with frostbite. Frostbite of the extremities can be categorized into:

- *Frost Nip or Incipient Frostbite* characterized by sudden blanching or whitening of skin.
- *Superficial Frostbite* skin has a waxy or white appearance and is firm to the touch, but tissue beneath is resilient.
- Deep Frostbite tissues are cold, pale, and solid; extremely serious injury.

Systemic hypothermia is caused by exposure to freezing or rapidly dropping temperature. It can be fatal. Its symptoms are usually exhibited in five stages: 1) shivering; 2) apathy, listlessness, sleepiness, and (sometimes) rapid cooling of the body to less than 95°F; 3) unconsciousness, glassy stare, slow pulse, and slow respiratory rate; 4) freezing of the extremities; and 5) death. Trauma sustained in freezing or sub-zero conditions requires special attention because an injured worker is predisposed to secondary cold injury. Special provisions must be made to prevent hypothermia and secondary freezing of damaged tissues in addition to providing for first aid treatment. To avoid cold stress, site personnel must wear protective clothing appropriate for the level of cold and physical activity. In addition to protective clothing, preventive safe work practices, additional training, and warming regimens may be utilized to prevent cold stress.

## 8.6 Safety Precautions for Cold Stress Prevention

For air temperature of  $0^{\circ}$ F or less, mittens should be used to protect the hands. For exposed skin, continuous exposure should not be permitted when air speed and temperature results in a wind chill temperature of -25°F.

At air temperatures of 36°F or less, field personnel who become immersed in water or whose clothing becomes wet must be immediately provided with a change of clothing and be treated for hypothermia.

If work is done at normal temperature or in a hot environment before entering the cold, the field personnel must ensure that their clothing is not wet as a consequence of sweating. Wet field personnel must change into dry clothes prior to entering the cold area.

If the available clothing does not give adequate protection to prevent hypothermia or frostbite, work must be modified or suspended until adequate clothing is made available or until weather conditions improve.

Field personnel handling evaporative liquid (e.g., gasoline, alcohol, or cleaning fluids) at air temperatures below 40°F must take special precaution to avoid soaking of clothing or gloves with the liquids because of the added danger of cold injury due to evaporative cooling.

## 8.7 Safe Work Practices

Direct contact between bare skin and cold surfaces (< 20°F) should be avoided. Metal tool handles and/or equipment controls should be covered by thermal insulating material.

For work performed in a wind chill temperature at or below 10°F, workers should be under constant protective observation (buddy system). The work rate should be established to prevent heavy sweating that will result in wet clothing. For heavy work, rest periods must be taken in heated shelters and workers should be provided with an opportunity to change into dry clothing if needed.

Field personnel should be provided the opportunity to become accustomed to coldweather working conditions and required protective clothing. Work should be arranged in such a way that sitting or standing still for long periods is minimized.

During the warming regimen (rest period), field personnel should be encouraged to remove outer clothing to permit sweat evaporation or to change into dry work clothing. Dehydration, or loss of body fluids, occurs insidiously in the cold environment and may increase susceptibility to cold injury due to a significant change in blood flow to the extremities. Fluid replacement with warm, sweet drinks and soups is recommended. The intake of coffee should be limited because of diuretic and circulatory effects.

## 8.8 Biological Hazards

Biological hazards may include poison ivy, snakes, thorny bushes and trees, ticks, mosquitoes, spiders, and other pests.

#### 8.8.1 Tick Borne Diseases

*Lyme Disease* - The disease commonly occurs in summer and is transmitted by the bite of infected ticks. "Hot spots" in the United States include New York, New Jersey, Pennsylvania, Massachusetts, Connecticut, Rhode Island, Minnesota, and Wisconsin.

*Erlichiosis* - The disease also commonly occurs in summer and is transmitted by the bite of infected ticks. "Hot spots" in the United States include New York, Massachusetts, Connecticut, Rhode Island, Minnesota, and Wisconsin.

These diseases are transmitted primarily by the deer tick, which is smaller and redder than the common wood tick. The disease may be transmitted by immature ticks, which are small and hard to see. The tick may be as small as a period on this page.

Symptoms of Lyme disease include a rash or a peculiar red spot, like a bull's eye, which expands outward in a circular manner. The victim may have headache, weakness, fever, a stiff neck, and swelling and pain in the joints, and eventually, arthritis. Symptoms of erlichiosis include muscle and joint aches, flu-like symptoms, but there is typically no skin rash.

*Rocky Mountain Spotted Fever (RMSF)* - This disease is transmitted via the bite of an infected tick. The tick must be attached 4 to 6 hours before the disease-causing organism (Rickettsia rickettsii) becomes reactivated and can infect humans. The primary symptom of RMSF is the sudden appearance of a moderate-to-high fever. The fever may persist for two to three weeks. The victim may also have a headache, deep muscle pain, and chills. A rash appears on the hands and feet on about the third day and eventually spreads to all parts of the body. For this reason, RMSF may be confused with measles or meningitis. The disease may cause death, if untreated, but if identified and treated promptly, death is uncommon.

*Control* - Tick repellant containing diethyltoluamide (DEET) should be used when working in tick-infested areas, and pant legs should be tucked into boots. In addition, workers should search the entire body every three or four hours for attached ticks. Ticks should be removed promptly and carefully without crushing, since crushing can squeeze the disease-causing organism into the skin. A gentle and steady pulling action should be used to avoid leaving the head or mouth parts in the skin. Hands should be protected with surgical gloves when removing ticks.

#### 8.8.2 Poisonous Plants

Poisonous plants may be present in the work area. Personnel should be alerted to its presence, and instructed on methods to prevent exposure.

*Control* - The main control is to avoid contact with the plant, cover arms and hands, and frequently wash potentially exposed skin. Particular attention must be given to avoiding skin contact with objects or protective clothing that have touched the plants. Treat every surface that may have touched the plant as contaminated, and practice contamination avoidance. If skin contact is made, the area should be washed immediately with soap and water, and observed for signs of reddening.

#### 8.8.3 Snakes

The possibility of encountering snakes exists, specifically for personnel working in wooded/vegetated areas. Snake venoms are complex and include proteins, some of which have enzymatic activity. The effects produced by venoms include neurotoxic effects with sensory, motor, cardiac, and respiratory difficulties; cytotoxic effects on red blood cells, blood vessels, heart muscle, kidneys, and lungs; defects in coagulation; and effects from local release of substances by enzymatic actions. Other noticeable effects of venomous snakebites include swelling, edema, and pain around the bite, and the development of ecchymosis (the escape of blood into tissues from ruptured blood vessels).

*Control* - To minimize the threat of snakebites, all personnel walking through vegetated areas must be aware of the potential for encountering snakes, and the need to avoid actions potentiating encounters, such as turning over logs, etc. If a snakebite occurs, an attempt should be made to safely identify the snake via size and markings. The victim must be transported to the nearest hospital within 30 minutes. First aid consists of applying a constriction band and washing the area around the wound to remove any unabsorbed venom.

## 8.8.4 Spiders

Personnel may encounter spiders during work activities.

Two spiders are of concern, the black widow and the brown recluse. Both prefer dark sheltered areas such as basements, equipment sheds and enclosures, and around woodpiles or other scattered debris. The black widow is shiny black, approximately one inch long, and found throughout the United States. There is a distinctive red hourglass marking on the underside of the black widows body. The bite of a black widow is seldom fatal to healthy adults, but effects include respiratory distress, nausea, vomiting, and muscle spasms. The brown recluse is smaller than the black widow and gets its name from its brown coloring and behavior. The brown recluse is more prevalent in the southern United States. The brown recluse has a distinctive violin shape on the top of its body. The bite of the brown recluse is painful and the bite site ulcerates and takes many weeks to heal completely.

*Control* - To minimize the threat of spider bites, all personnel walking through vegetated areas must be aware of the potential for encountering these arachnids. Personnel need to avoid actions that may result in encounters, such as turning over logs, and placing hands in dark places such as behind equipment or in corners of equipment sheds or enclosures. If a spider bite occurs, the victim must be transported to the nearest hospital as soon as possible; first aid consists of applying ice packs and washing the area around the wound to remove any unabsorbed venom.

#### 8.9 Noise

Exposure to noise over the OSHA action level can cause temporary impairment of hearing; prolonged and repeated exposure can cause permanent damage to hearing. The risk and severity of hearing loss increases with the intensity and duration of exposure to noise. In addition to damaging hearing, noise can impair voice communication, thereby increasing the risk of accidents on site.

*Control* - All personnel must wear hearing protection, with a Noise Reduction Rating (NRR) of at least 20, when noise levels exceed 85 dBA. When it is difficult to hear a coworker at normal conversation distance, the noise level is approaching or exceeding 85 dBA, and hearing protection is necessary. All site personnel who may be exposed to noise must also receive baseline and annual audiograms and training as to the causes and prevention of hearing loss. Noise monitoring is discussed in Section 5.2, Noise Monitoring.

Whenever possible, equipment that does not generate excessive noise levels will be selected for this project. If the use of noisy equipment is unavoidable, barriers or increased distance will be used to minimize worker exposure to noise, if feasible.

## 8.10 Spill Control

All personnel must take every precaution to minimize the potential for spills during site operations. All on-site personnel shall immediately report any discharge, no matter how small, to the FS.

Spill control equipment and materials will be located on the site at locations that present the potential for discharge. All sorbent materials used for the cleanup of spills will be containerized and labeled appropriately. In the event of a spill, the FS will follow the provisions in Section 10.0, Emergency Procedures, to contain and control released materials and to prevent their spread to off-site areas.

## 8.11 Sanitation

Site sanitation will be maintained according to OSHA requirements.

## 8.11.1 Break Area

Breaks must be taken in the SZ, away from the active work area after site personnel go through decontamination procedures. There will be no smoking, eating, drinking, or chewing gum or tobacco in any area other than the SZ.

## 8.11.2 Potable Water

The following rules apply to all field operations:

- An adequate supply of potable water will be provided at each project site. Potable water must be kept away from hazardous materials or media, and contaminated clothing or equipment.
- Portable containers used to dispense drinking water must be capable of being tightly closed and must be equipped with a tap dispenser. Water must not be consumed directly from the container (drinking from the tap is prohibited) nor may it be removed from the container by dipping.
- Containers used for drinking water must be clearly marked and shall not be used for any other purpose.
- Disposable drinking cups must be provided. A sanitary container for dispensing cups and a receptacle for disposing of used cups is required.

## 8.11.3 Sanitary Facilities

Access to facilities for washing before eating, drinking, or smoking, or alternate methods such as waterless hand-cleaner and paper towels will be provided.

## 8.11.4 Lavatory

If permanent toilet facilities are not available, an appropriate number of portable chemical toilets will be provided. This requirement does not apply to mobile crews or to normally unattended site locations so long as employees at these locations have transportation immediately available to nearby toilet facilities.

## 8.12 Emergency Equipment

Adequate emergency equipment for the activities being conducted on site and as required by applicable sections of 29 CFR 1910 and 29 CFR 1926 will be on site prior to the commencement of project activities. Personnel will be provided with access to emergency equipment, including, but not limited to, the following:

- Fire extinguishers of adequate size, class, number, and location as required by applicable sections of 29 CFR 1910 and 1926;
- Industrial first aid kits of adequate size for the number of personnel on site; and
- Emergency eyewash and/or shower if required by operations being conducted on site.

## 8.13 Lockout/Tagout Procedures

Only fully qualified and trained personnel will perform maintenance procedures. Before maintenance begins, lockout/tagout procedures per OSHA 29 CFR 1910.147 will be followed.

Lockout is the placement of a device that uses a positive means, such as lock, to hold an energy or material-isolating device such that the equipment cannot be operated until the lockout device is removed. If a device cannot be locked out, a tagout system shall be used. Tagout is the placement of a warning tag on an energy or material isolating device indicating that the equipment controls may not be operated until the personnel who attached the tag remove the tag.

## 8.14 Electrical Safety

Electricity may pose a particular hazard to site workers due to the use of portable electrical equipment. If wiring or other electrical work is needed, a qualified electrician must perform it.

General electrical safety requirements include:

- All electrical wiring and equipment must be a type listed by Underwriters Laboratories (UL), Factory Mutual Engineering Corporation (FM), or other recognized testing or listing agency.
- All installations must comply with the National Electrical Safety Code (NESC), the National Electrical Code (NEC), or USCG regulations.
- Portable and semi-portable tools and equipment must be grounded by a multiconductor cord having an identified grounding conductor and a multi-contact polarized plug-in receptacle.
- Tools protected by an approved system of double insulation, or its equivalent, need not be grounded. Double insulated tools must be distinctly marked and listed by UL or FM.

- Live parts of wiring or equipment must be guarded to prevent persons or objects from touching them.
- Electric wire or flexible cord passing through work areas must be covered or elevated to protect it from damage by foot traffic, vehicles, sharp corners, projections, or pinching.
- All circuits must be protected from overload.
- Temporary power lines, switchboxes, receptacle boxes, metal cabinets, and enclosures around equipment must be marked to indicate the maximum operating voltage.
- Plugs and receptacles must be kept out of water unless of an approved submersible construction.
- All extension cord outlets must be equipped with ground fault circuit interrupters (GFCI).
- Attachment plugs or other connectors must be equipped with a cord grip and be constructed to endure rough treatment.
- Extension cords or cables must be inspected prior to each use, and replaced if worn or damaged. Cords and cables must not be fastened with staples, hung from nails, or suspended by bare wire.
- Flexible cords must be used only in continuous lengths without splice, with the exception of molded or vulcanized splices made by a qualified electrician.

# 8.15 Lifting Safety

Using proper lifting techniques may prevent back strain or injury. The fundamentals of proper lifting include:

- Consider the size, shape, and weight of the object to be lifted. A mechanical lifting device or additional persons must be used to lift an object if it cannot be lifted safely alone.
- The hands and the object should be free of dirt or grease that could prevent a firm grip.
- Gloves must be used, and the object inspected for metal slivers, jagged edges, burrs, or rough or slippery surfaces.
- Fingers must be kept away from points that could crush or pinch them, especially when putting an object down.
- Feet must be placed far enough apart for balance. The footing should be solid and the intended pathway should be clear.
- The load should be kept as low as possible, close to the body with the knees bent.
- To lift the load, grip firmly and lift with the legs, keeping the back as straight as possible.
- A worker should not carry a load that he or she cannot see around or over.
- When putting an object down, the stance and position are identical to that for lifting; the legs are bent at the knees, and the back is straight as the object is lowered.

## 8.16 Ladder Safety

When portable ladders are used for access to an upper landing surface, the ladder side rails shall extend at least 3 feet (9 m) above the upper landing surface to which the ladder is used to gain access; or, when such an extension is not possible because of the ladder's length, then the ladder shall be secured at its top to a rigid support that will not deflect,

and a grasping device, such as a grabrail, shall be provided to assist employees in mounting and dismounting the ladder. In no case shall the extension be such that ladder deflection under a load would, by itself, cause the ladder to slip off its support.

- Ladders shall be maintained free of oil, grease, and other slipping hazards.
- Ladders shall not be loaded beyond the maximum intended load for which they were built, or beyond their manufacturer's rated capacity.
- Ladders shall be used only for the purpose for which they were designed.
- Non-self-supporting ladders shall be used at an angle such that the horizontal distance from the top support to the foot of the ladder is approximately one-quarter of the working length of the ladder (the distance along the ladder between the foot and the top support).
- Wood job-made ladders with spliced side rails shall be used at an angle such that the horizontal distance is one-eighth the working length of the ladder.
- Fixed ladders shall be used at a pitch no greater than 90 degrees from the horizontal, as measured to the back side of the ladder.
- Ladders shall be used only on stable and level surfaces unless secured to prevent accidental displacement.
- Ladders shall not be used on slippery surfaces unless secured or provided with slip-resistant feet to prevent accidental displacement. Slip-resistant feet shall not be used as a substitute for care in placing, lashing, or holding a ladder that is used upon slippery surfaces, including, but not limited to, flat metal or concrete surfaces that are constructed so they cannot be prevented from becoming slippery.
- Ladders placed in any location where they can be displaced by workplace activities or traffic, such as in passageways, doorways, or driveways, shall be secured to prevent accidental displacement, or a barricade shall be used to keep the activities or traffic away from the ladder.
- The area around the top and bottom of ladders shall be kept clear.
- The top of a non-self-supporting ladder shall be placed with the two rails supported equally unless it is equipped with a single support attachment.
- Ladders shall not be moved, shifted, or extended while occupied.
- Ladders shall have non-conductive side rails if they are used where the employee or the ladder could contact exposed energized electrical equipment.
- The top, top step, or the step labeled that it or any step above it should not be used as a step.
- Cross-bracing on the rear section of stepladders shall not be used for climbing unless the ladders are designed and provided with steps for climbing on both front and rear sections.
- Ladders shall be inspected by the HSM for visible defects on a daily basis and after any occurrence that could affect their safe use.
- Portable ladders with structural defects, such as, but not limited to, broken or missing rungs, cleats, or steps; broken or split rails; corroded components; or other faulty or defective components shall either be immediately marked in a manner that readily identifies them as defective, or be tagged with "Do Not Use" or similar language, and shall be withdrawn from service.
- Fixed ladders with structural defects, such as, but not limited to, broken or missing rungs, cleats, or steps; broken or split rails; or corroded components; shall be withdrawn from service.
- Ladder repairs shall restore the ladder to a condition meeting its original design criteria, before the ladder is returned to use.

- Single-rail ladders shall not be used. •
- When ascending or descending a ladder, the user shall face the ladder.
- Each employee shall use at least one hand to grasp the ladder when progressing up and/or down the ladder.
- An employee shall not carry any object or load that could cause the employee to lose balance and fall.

#### 8.17 Traffic Safety

The project site may be located adjacent to a public roadway where exposure to vehicular traffic is likely. Traffic may also be encountered as vehicles enter and exit the area. To minimize the likelihood of project personnel and activities being affected by traffic, the following procedures will be implemented.

Cones must be placed along the shoulder of the roadway starting 100 feet from the work area to alert passing motorists to the presence of personnel and equipment. A "Slow" or "Men Working" sign must be placed at the first cone. Barricades with flashing lights should be placed between the roadway and the work area.

During activities along a roadway, equipment will be aligned parallel to the roadway to the extent feasible, facing into the oncoming traffic so as to place a barrier between the work crew and the oncoming traffic. All crewmembers must remain behind the equipment and the traffic barrier.

All site personnel who are potentially exposed to vehicular traffic must wear an outer layer of orange warning garments, such as vests, jackets, or shirts. If work is performed in hours of dusk or darkness, workers will be outfitted with reflective garments either orange, white (including silver-coated reflective coatings or elements that reflect white light), yellow, fluorescent red-orange, or fluorescent yellow-orange.

The flow of traffic into and out of the adjacent business must be assessed, and precautions taken to warn motorists of the presence of workers and equipment. Where possible, vehicles should be aligned to provide physical protection of people and equipment.

#### 9.0 SITE-SPECIFIC HAZARDS AND CONTROL MEASURES

#### 9.1 **Evaluation of Hazards**

The evaluation of hazards is provided as a quick reference as to the known conditions for the Site, wherein the level of detail for each of the subsections is identified.

9.1.1	Hazard Characteristics		
	Existing information for Site: X Detailed Preliminary	None	
	Hazardous/Contaminated Material F <u>X</u> Solid <u>X</u> Liquid	<sup>5</sup> orm(s): Sludge	Gas X Vapor
	Containment Type(s): Drum <u>X</u> Tank	Pit	Debris
Propose	ed Development	38	Health and Safety

	PondLagoon	Other:
	Hazardous Material Characteristics <u>X</u> Volatile <u>Corrosive</u> Ignitable <u>X</u> Toxic	: Reactive Radioactive X_ Unknown
	Routes of Exposure: <u>X</u> Oral <u>X</u> Dermal	<u>X</u> Eye <u>X</u> Respiratory
9.1.2	Potential Health and Safety Haza	rds
	X       Heat         X       Cold         Confined space entry         Oxygen depletion         Asphyxiation         X       Excavation         X       Cave-ins         X       Falls, slippage	Congested areas X General Construction X Physical injury X Electrical hazards Handling and product transfer X Fire Explosion X Biological Hazards <u>X</u> Plants – Poison Ivy, Poison Oak X Insects – Mosquitoes
etc.)	<u>X</u> Heavy equipment	<u>X</u> Insects – Bees and Wasps <u>X</u> Rats and Mice Non-ionizing Radiation (i.e. UV, IR,

\_\_\_\_\_ Other: Potential Ignition Hazard.

## 9.2 Field Activities, Hazards, and Control Procedures

The following task-specific safety analyses identify potential health, safety, and environmental hazards associated with each type of field activity. Because of the complex and changing nature of field projects, supervisors must continually inspect the site to identify hazards that may affect on-site personnel, the community, or the environment. The FS must be aware of these changing conditions and discuss them with the PM whenever these changes impact employee health, safety, the environment, or performance of the project. The FS will keep on-site personnel informed of the changing conditions, and the PM will write and/or approve addenda or revisions to this HASP as necessary.

## 9.2.1 Chemical Sample Collection/Analysis

## Description of Tasks

This sub-task consists of the collection of soil, groundwater and soil vapor samples for subsequent field and laboratory analysis. The physical hazards of groundwater and soil vapor sampling are primarily associated with the sample collection methods, procedures utilized, and the environment itself.

#### Hazard Identification

Incidental contact with COCs is the primary hazard associated with sampling the stabilized material. This contact may occur through the manipulation of sample media

and equipment, manual transfer of media into sample containers, and proximity of operations to the breathing zone. The primary hazards associated with these sampling procedures are not potentially serious; however, other operations in the area, or the conditions under which samples must be collected, may present chemical and physical hazards. The hazards directly associated with sampling procedures are generally limited to strains/sprains and potential eye hazards. Potential chemical hazards may include contact with media containing site COCs and potential contact with chemicals used for equipment decontamination.

#### **Controls**

*PPE* – To control dermal exposure during sampling activities, a minimum of Level D protection will be worn. If necessary, based on field observations and site conditions, air monitoring may be conducted during sediment sampling activities. If the results of air monitoring indicate the presence of airborne contaminants in a concentration causing concern, personnel will upgrade to Level C protection. Refer to Section 5.1, Air Monitoring, for a description of air monitoring requirements and action levels. A description of each level of personal protection is included in Section 4.0, Personal Protective Equipment.

#### 9.2.2 Decontamination

All sampling equipment will be decontaminated before leaving the site. Personnel involved in decontamination activities may be inadvertently exposed to skin contact with contaminated materials and chemicals brought from the EZ. Personnel involved in decontamination activities must wear PPE that is, at a minimum, one level below the level worn by personnel working in the EZ.

#### 9.3 Chemical Hazards

The chemical hazards associated with site operations are related to inhalation, ingestion, and skin exposure to site COCs. Concentrations of airborne COCs during site tasks may be measurable and will require air monitoring during certain operations. Air monitoring requirements for site tasks are outlined in Section 5.1.

COCs at the site include some VOCs, SVOC and metals.

The potential for inhalation of site COCs is low. The potential for dermal contact with soils containing site COCs during remedial operations is moderate. Table 6 lists the primary contaminants that have been identified at the Site and the media in which they are present.

# Table 6 – List of Primary Contaminants

Media: Soil				
Metals	Highest Detected Concentration (mg/kg)	Applicable Monitoring		
Arsenic	39.7	NA		
Copper	59.5	NA		
Lead	237	NA		
Mercury	2.11	NA		
Nickel	70.2	NA		
Zinc	686	NA		
4,4'-DDT	0.34	NA		
Benzo[a]anthracene	18	NA		
benzo(a)pyrene	16	NA		
benzo(b)fluoranthene	18	NA		
benzo(k)fluoranthene	7	NA		
Chrysene	16	NA		
Dibenz(a,h)anthracene	1.8	NA		
Indeno[1,2,3-				
cd]pyrene	6.7	NA		

Media: Groundwater					
	Highest Detected Concentration (ug/L)	Applicable Monitoring Instrument			
Cis-1,2-					
dichloroethene	6.4	PID			
Benzo(a)anthracene	0.067	NA			
benzo(a)pyrene	0.067	NA			
benzo(b)fluoranthene	0.075	NA			
benzo(k)fluoranthene	0.032	NA			
Indeno(1,2,3-					
cd)pyrene	0.052	NA			
Phenol	2.7	NA			
Hexachlorobenzene	.044	NA			
PFOA/PFAS					
Total PFOA/PFAS	0.44	NA			

Media: Soil Vapor		
VOCs	Highest Detected Concentration (ug/m3)	Applicable Monitoring Instrument
Benzene	39.3	PID
1,3 Butadiene	160	PID
Heptane	545	PID
Hexane	223	PID
Methyl ethyl ketone	124	PID
Toluene	59.9	PID
Trichloroethene	20.8	PID
Xylene	564	PID
Ethylbenzene	573	PID

## 10.0 EMERGENCY PROCEDURES

#### 10.1 General

Prior to the start of operations, the work area will be evaluated for the potential for fire, contaminant release, or other catastrophic event. Unusual conditions or events, activities, chemicals, and conditions will be reported to the FS/SSO immediately.

The FS/SSO will establish evacuation routes and assembly areas for the site. All personnel entering the site will be informed of this route and the assembly area.

#### **10.2 Emergency Response**

If an incident occurs, the following steps will be taken:

- The FS/SSO will evaluate the incident and assess the need for assistance and/or evacuation;
- The FS/SSO will call for outside assistance as needed;
- The FS/SSO will ensure the PM is notified promptly of the incident; and
- The FS/SSO will take appropriate measures to stabilize the incident scene.

#### 10.2.1 Fire

In the case of a fire at the site, the FS/SSO will assess the situation and direct fire-fighting activities. The FS/SSO will ensure that the PM is immediately notified of any fires. Site personnel will attempt to extinguish the fire with available extinguishers, if safe to do so. In the event of a fire that site personnel are unable to safely extinguish with one fire extinguisher, the local fire department will be summoned.

## 10.2.2 Contaminant Release

In the event of a contaminant release, the following steps will be taken:

- Notify FS/SSO immediately;
- Evacuate immediate area of release;
- Conduct air monitoring to determine needed level of PPE; and

• Don required level of PPE and prepare to implement control procedures.

The FS/SSO has the authority to commit resources as needed to contain and control released material and to prevent its spread to off-site areas.

## 10.3 Medical Emergency

All employee injuries must be promptly reported to the SSO/FS, who will:

- Ensure that the injured employee receives prompt first aid and medical attention;
- In emergency situations, the worker is to be transported by appropriate means to the nearest urgent care facility (normally a hospital emergency room); and
- If the injured person is a SESI employee, notify SESI at 973-808-9050.

## 10.3.1 Emergency Care Steps

Survey the scene. Determine if it is safe to proceed. Try to determine if the conditions that caused the incident are still a threat. Protect yourself from exposure before attempting to rescue the victim.

- Do a primary survey of the victim. Check for airway obstruction, breathing, and pulse. Assess likely routes of chemical exposure by examining the eyes, mouth, nose, and skin of the victim for symptoms.
- Phone Emergency Medical Services (EMS). Give the location, telephone number used, caller's name, what happened, number of victims, victim's condition, and help being given.
- Maintain airway and perform rescue breathing as necessary.
- Perform CPR as necessary.
- Do a secondary survey of the victim. Check vital signs and do a head-to-toe exam.

Treat other conditions as necessary. If the victim can be moved, take him/her to a location away from the work area where EMS can gain access.

## 10.4 First Aid - General

All persons must report any injury or illness to their immediate supervisor or the FS. Trained personnel will provide first aid. Injuries and illnesses requiring medical treatment must be documented. The FS and SSO must fill out an accident/incident report as soon as emergency conditions no longer exist and first aid and/or medical treatment has been ensured. The report must be completed and submitted to the PM within 24 hours after the incident.

If first-aid treatment is required, first aid kits are kept at the CRZ. If treatment beyond first aid is required, the injured person(s) should be transported to the medical facility. If the injured person is not ambulatory, or shows any sign of not being in a comfortable and stable condition for transport, then an ambulance/paramedics should be summoned. If there is any doubt as to the injured worker's condition, it is best to let the local paramedic or ambulance service examine and transport the worker.

#### 10.4.1 First Aid - Inhalation

Any employee complaining of symptoms of chemical overexposure as described in Section 4, General Site Safety Procedures, will be removed from the work area and transported to the designated medical facility for examination and treatment.

#### 10.4.2 First Aid - Ingestion

Call EMS and consult a poison control center for advice. If available, refer to the MSDS for treatment information. If the victim is unconscious, keep them on their side and clear the airway if vomiting occurs.

## 10.4.3 First Aid - Skin Contact

Project personnel who have had skin contact with contaminants will, unless the contact is severe, proceed through the CRZ, to the wash area. Personnel will remove any contaminated clothing, and then flush the affected area with water for at least 15 minutes. The worker should be transported to the medical facility if he/she shows any sign of skin reddening, irritation, or if he/she requests a medical examination.

## 10.4.4 First Aid - Eye Contact

Project personnel who have had contaminants splashed in their eyes or who have experienced eye irritation while in the EZ, must immediately proceed to the eyewash station in the CRZ. Do not decontaminate prior to using the eyewash. Remove whatever protective clothing is necessary to use the eyewash. Flush the eye with clean running water for at least 15 minutes. Arrange prompt transport to the designated medical facility.

#### 10.5 Reporting Injuries, Illnesses, and Safety Incidents

Injuries and illnesses, however minor, will be reported to the FS immediately. The FS will complete an injury report and submit it to the HSM, and the PM by end of shift.

## 10.6 Emergency Information

The means to summon local public response agencies such as police, fire, and ambulance will be reviewed in the daily safety meeting. These agencies are identified in Table 7.

Local Emergency Contacts	Telephone No.
EMERGENCY	911
Montefiore New Rochelle Hospital	(914) 632-5000
Police Emergency	911
Fire Emergency	911
Rescue Squad	911
Ambulance	911
Miscellaneous Contacts	Telephone No.
N.Y. Poison Control Center	(800) 222-1222
National Response Center and Terrorist	(800) 424-8802
Hotline	
Center for Disease Control	(800) 311-3435
Utility Mark-Out	(800) 962-7962

#### Table 7 – Emergency Contacts

## 10.6.1 Directions to Hospital

Montefiore New Rochelle Hospital 16 Guion Place New Rochelle, New York

Directions to Hospital:

Head northeast on Main St toward Lawton St Turn left onto Lawton St Turn left onto Huguenot St Turn right onto Memorial Hwy Turn left onto Lockwood Ave Turn left at the 1st cross street onto Guion Pl Destination will be on the right



## 11.0 LOGS, REPORTS, AND RECORD KEEPING

The following is a summary of required health and safety logs, reports, and record keeping for the operations at the subject site.

## 11.1 HASP Field Change Request

To be completed for initiating a change to the HASP. PM approval is required. The original will be kept in the project file (See Attachment 3).

## 11.2 Medical and Training Records

The HSM must obtain and keep a log of personnel meeting appropriate training and medical qualifications for the site work. The log will be kept in the project file. Each company's Human Resources Department will maintain medical records, in accordance with 29 CFR 1910.1020.

## 11.3 Exposure Records

Any personnel monitoring results, laboratory reports, calculations, and air sampling data sheets are part of an employee exposure record. These records will be kept in accordance with 29 CFR 1910.1020. For SESI employees, the originals will be sent to the Human Resources Manager. For subcontractor employees, the original file will be sent to the subcontractor employer with a copy maintained in the SESI project file.

## 11.4 Accident/Incident Report

Any accident/incident reports must be completed following procedures given in Section 10.5 of this HASP. The originals will be sent to the HSM for maintenance. A copy of the forms will be kept in the project file. (See Attachment 4)

# 11.5 OSHA Form 200

An OSHA Form 200 (Log of Occupational Injuries and Illnesses) will be kept at the project site. All recordable injuries or illnesses will be recorded on this form. At the end of the project, the original will be sent to the Human Resources Manager for maintenance. Subcontractor employees must also meet the requirements of maintaining an OSHA 200 Form. The accident/incident report meets the requirements of the OSHA Form 101 (Supplemental Record), which must be maintained with the OSHA Form 200 for all recordable injuries or illnesses.

## 11.6 On-Site Health and Safety Field Logbooks

The HSM or designee will maintain an on-site health and safety log book in which daily Site conditions, activities, personnel, and significant events will be recorded. Calibration records and personnel monitoring results, if available, will also be recorded in the field logbook. The original logbook will be kept in the project file.

Whenever any personnel monitoring is conducted onsite, the monitoring results will be noted in the filed logbook. These will become part of the exposure records file and will be maintained by the HSM.

A signatory page is included (See Attachment 5) and is to be signed by those working on and/or visiting the site.

## 11.7 Material Safety Data Sheets

Material Safety Data Sheets (MSDS) will be obtained and kept on file at the project site for each hazardous chemical brought to, use, or stored at the Site (See Attachment 6).

## 12.0 COVID-19 RISK MANAGEMENT

This Section includes requirements for managing and minimizing the potential for transmission of the Severe Acute Respiratory Syndrome Coronavirus 2 (SARS-CoV-2) virus, which causes the Novel Coronavirus Disease 2019 (COVID-19). COVID-19 typically causes respiratory illness in people. SARS-CoV-2 is currently known to spread via respiratory droplets produced when a person infected with the virus coughs or sneezes, the same way flu and other respiratory illnesses spread. SARS-CoV-2 can also be transmitted if people touch surfaces and objects with the virus on it. COVID-19 can cause mild to severe respiratory illness with symptoms of fever, cough, and difficulty breathing. Preliminary information suggests older adults and people with underlying health conditions or compromised immune systems may be at higher risk of severe illness from this virus. Center for Disease Control (CDC) believes that symptoms of COVID-19 begin between 2 and 14 days after exposure.

# 12.1 Best Practices to Prevent Infection

Currently the best way identified to prevent infection is to minimize the potential of exposure to SARS-CoV-2. CDC recommends everyday actions to help prevent the spread of any respiratory viruses:

- Wash your hands often with soap and water for at least 20 seconds. If soap and water are not available, use an alcohol-based hand sanitizer, containing at least 60% alcohol.
- Avoid touching your eyes, nose, and mouth with unwashed hands.
- Avoid close contact with people who are sick.
- Stay home when you are sick.
- Cover your cough or sneeze with a tissue, then throw the tissue in the trash can and wash hands or use hand sanitizer.
- Clean and disinfect frequently touched objects and surfaces.
- Wear face masks
- Safe social distancing (e.g., maintain a distance of 6 feet between people, limited group meetings)

The above recommendations are illustrated on the poster included in Attachment 7.

# 12.2 Risk Identification

COVID-19 is a new disease; scientists and health agencies are continuously learning about how it spreads. SESI shall adjust site policies based on the most up to date government issued guidance regarding transmission. shall confirm staff that have worked in locations where quarantine orders are in place, have met the minimum quarantine guidance and do not have symptoms prior to mobilizing to site.

# 12.3 Risk Minimization

SESI will implement the following engineering and administrative controls wherever possible to minimize the spread of COVID-19:

## Engineering Controls

1. Increasing ventilation rates of interior workspaces.

- 2. Access controls, including fences and locking gates.
- 3. Maintain 6 feet distances, using distance markers where appropriate in the field.

## Administrative Controls

1. Continuous and effective communication of administrative controls/requirements to all site personnel and visitors, through the posting of site signage, preparation and distribution of site plans, presented during site meetings, and verbal warnings if necessary.

2. Require that all employees exhibiting any COVID-19 symptom do not enter the site and provide sick leave policies to support this requirement.

3. To minimize face-to-face interaction, the Site's Health & Safety Officer's (or other designated employee) phone number shall be prominently posted and disseminated to project staff to be called for the purpose of site sign in and sign out by all visitors to the site upon arrival and exit. The designated employee will receive entry and exit calls each day and will fill out the site entry/exit log (Attachment 7) for each site visitor to reduce traffic in site trailer and/or the number of individuals contacting the site access tracking log.

4. Staffing: only those employees necessary to complete critical path task(s) shall be present on-site at any given time. Work shall be scheduled to minimize the density of personnel in any given area at any given time.

5. Working Remotely; employees shall be encouraged to complete work remotely if possible.

6. Face-to-face meetings shall be replaced with video or phone conferences when practicable.

7. Social distancing shall be exercised for face-to-face meetings e.g. daily Health and Safety tailgate meeting. In addition, SESI shall plan to have multiple meetings (if necessary) to keep the number of participants to a threshold that allows for the practice of social distancing protocol. The Health and Safety officer will keep a record of all present for each meeting on the Health and Safety log.

8. Quarantine staff that have been in contact with anyone that tested positive and notify NYSDEC immediately.

## 12.4 Safe Work Practices

SESI employ social distancing protocol for all onsite activities when able. SESI will provide adequate hand washing stations and hand sanitizer (containing a minimum of 60% alcohol) to allow site personnel and visitors to practice good personal hygiene. SESI shall provide tissues, paper towels, no-touch trash cans, and disinfectants to maintain site cleanliness as needed. Sharing of tools and heavy equipment shall be limited to the extent practicable; handles of shared tools and equipment shall be sanitized regularly.

## 12.5 PPE to Prevent COVID-19 Transmission

Employees shall be provided disposable personal protective equipment (PPE), including gloves, goggles, face shields, face masks, and respiratory protection, as appropriate based on work environment and current recommendations by OSHA and CDC. All PPE must be selected based on hazard to the worker, properly fitted and periodically refitted, consistently and properly worn when required, regularly inspected, maintained, and replaced, as necessary, and properly removed, cleaned, and stored or disposed of, to avoid contamination of self, others, or the environment. **PPE worn to prevent**
# transmission of COVID-19 is not to be confused with PPE for protection against site contaminants.

- Face masks should fit snugly but comfortable against the side of the face and over the nose and be secured with ties or ear loops; cloth masks must include multiple layers of fabric, allow for breathing without restriction, and be able to be laundered and machine dried without damage.
- Face masks should be worn consistently and removed without touching eyes, nose, and mouth. An individual should wash their hands after handling a used face mask.
- Cloth face coverings should be sterilized by machine washing between use; disposable face masks shall be disposed of properly after using.
- Gloves are only effective if changed and disposed of frequently, to avoid crosscontamination.

# 12.6 Notification of a Potential or Confirmed Infection

In the event of a potential or confirmed COVID-19 infection of any personnel working on the Site, SESI will take the following actions:

- Notify the Department immediately upon identification of a suspected or confirmed infection of COVID-19. This notification shall comply with HIPAA regulations.
- Remove an individual suspected to have COVID-19 from the site immediately, as well as those who have worked in close contact with that individual for extended periods of time (an hour at a time or more) over the previous week. The individual with suspected infection shall contact their health care provider and/or follow local health department testing procedures and protocol.
- While in the process of removing an employee exhibiting symptoms, steps should be taken to isolate the individual, place a surgical mask on the individual and inform the local health department and the NYSDEC.
- In the event the individual with suspected infection cannot get home right away, they shall isolate, contact their health care provider, and/or follow local health department testing procedures and protocol.
- In the absence of local health department information, the individual may call the New York State Hotline at 1-888-364-3065.
- SESI shall maintain communication with potentially infected individual(s) and notify the Engineer upon receipt of COVID-19 test results.
- Positively infected individuals may return to work at the site after 72 hours of being symptom-free and 7 days of isolation after the first symptoms appeared, or in accordance with the current federal, state, and local guidelines.
- OSHA recordkeeping requirements at 29 CFR Part 1904 mandate covered employers record certain work-related injuries and illnesses on their OSHA 300 log. COVID-19 can be a recordable illness if a worker is infected as a result of

performing their work-related duties. However, employers are only responsible for recording cases of COVID-19 if all the following are met:

1. The case is a confirmed case of COVID-19 (see CDC information on persons under investigation and presumptive positive and laboratory-confirmed cases of COVID-19).

2. The case is work-related, as defined by 29 CFR 1904.5; and

3. The case involves one or more of the general recording criteria set forth in 29 CFR 1904.7 (e.g. medical treatment beyond first-aid, days away from work).

# ATTACHMENT 1 AIR MONITOR LOG

# Air Monitoring: Sample Collection and Analysis

Date & Time of Monitoring	Task / Operation Being	Substance(s)/ Hazard(s) Being	Monitoring Location	Type/Method of Monitoring	Monitoring Results	Exposure Limits	Required Action

ATTACHMENT 2 OSHA POSTER

# Job Safety and Health It's the law!

#### **EMPLOYEES:**

- You have the right to notify your employer or OSHA about workplace hazards. You may ask OSHA to keep your name confidential.
- You have the right to request an OSHA inspection if you believe that there are unsafe and unhealthful conditions in your workplace. You or your representative may participate in that inspection.
- You can file a complaint with OSHA within 30 days of retaliation or discrimination by your employer for making safety and health complaints or for exercising your rights under the OSH Act.
- You have the right to see OSHA citations issued to your employer. Your employer must post the citations at or near the place of the alleged violations.
- Your employer must correct workplace hazards by the date indicated on the citation and must certify that these hazards have been reduced or eliminated.
- You have the right to copies of your medical records and records of your exposures to toxic and harmful substances or conditions.
- · Your employer must post this notice in your workplace.
- You must comply with all occupational safety and health standards issued under the OSH Act that apply to your own actions and conduct on the job.

#### **EMPLOYERS:**

- You must furnish your employees a place of employment free from recognized hazards.
- You must comply with the occupational safety and health standards issued under the OSH Act.

This free poster available from OSHA – The Best Resource for Safety and Health





# 1-800-321-OSHA (6742)

OSHA-supported consultation programs in each state.

OSHA 3166-02 2012R

www.osha.gov



# ATTACHMENT 3 FILED CHANGE REQUEST FORM

# HEALTH & SAFETY PLAN CHANGE NOTICE

			Pages	of
Proje	ct:		H&S-C	CN
1)	HASP VERSION:	SECTION:	PAGE (s):	
	RE: Change to Addition Other:	o existing HASP to existing HASP	Anticipated Revision Date:	
			CO	NT
2)	PROPOSED CHANGE:			
3)	REASON FOR PROPOSE	D CHANGE(s): by SPEC or Change Order	Other:	
	Dispositi Change i Operation	on of Deficiency n Regulatory or Other Requir nal Experience	ementsC	ONT
4)	EXHIBITS ATTACHED	NOYES (If YES	, describe)CON	Т
5)	PMK APPROVALS	PROJECT MANAGER:	Date:	
		SITE MANAGER: H&S MANAGER:	Date: Date:	
	Client Approval Required:	NOYES (If Y	ES, date submitted)	
6)	CLIENT APPROVAL	APPROVED	REMANDEDREJECTI	ED
			CONT	
	Client Representative:		Date:	
7)	DISTRIBUTION AFTER	APPROVAL		
		LIST OTHER:		
	$\frac{\underline{X}}{\underline{X}} \qquad \text{HASP OFDATE I} \\ \frac{\underline{X}}{\underline{X}} \qquad \text{CLIENT} \\ \frac{\underline{X}}{\underline{X}} \qquad \text{PROJECT FILES} $			

# ATTACHMENT 4 INJURY REPORT FORM

USHA'S Form 301 Injury and Illness This Injury and Illness Incident Report is one of the	Incident Report	employee health and must be used in a manner that protects the confidentiality of employees to the extent possible while the information is being used for occupational safety and health purposes. Information about the case
This <i>Injury and Illness Incident Report</i> is one of the first forms you must fill out when a recordable work-	Information about the employee 1) Full name	Information about the ca 10) Case number from the Log
related injury or illness has occurred. Together with the Log of Work-Related Injurnes and Illnesses and the accompanying Summary, these forms help the	2) Street	11) Date of injury or illness         12) Time employee began work
accompanying Summary, these forms help the employer and OSHA develop a picture of the extent and severity of work-related incidents.	City         State         ZIP           3) Date of birth         //         //	<ol> <li>13) Time of event</li> <li>14) What was the employee doi</li> </ol>
Within 7 calendar days after you receive information that a recordable work-related injury or illness has occurred, you must fill out this form or an equivalent. Some state workers' compensation, insurance, or other reports may be acceptable	4) Date hired / / 5) Male Female	tools, equipment, or material carrying roofing materials";
substitutes. Jo be considered an equivalent lorm, any substitute must contain all the information asked for on this form. According to Public Law 91-596 and 29 CFR 1904, OSHA's recordkeeping rule, you must keep this form on file for 5 years following the year to	Information about the physician or other health ca professional <sup>6)</sup> Name of physician or other health care professional	15) What happened? Tell us how th fell 20 feet"; "Worker was spray developed soreness in wrist ove
which it pertains. If you need additional copies of this form, you may photocopy and use as many as you need.	7) If treatment was given away from the worksite, where was it given? Facility	16) What was the injury or illness? more specific than "hurt," "pain tunnel syndrome."
	Street	-
Completed by	CityStateZIP <sup>8)</sup> Was employee treated in an emergency room? <sup>1</sup> Yes <sup>1</sup> No	17) What object or substance di "radial arm saw." <i>If this quest</i>
Title Phone ()Date//	<ul> <li><sup>9)</sup> Was employee hospitalized overnight as an in-patient?</li> <li>1 Yes</li> <li>1 No</li> </ul>	18) If the employee died, when c

Washington, DC 20210. Do not send the completed forms to this office. II.yo ALL C Departm

OSHA's Form 30	0 (Rev. 01/2004) <b>C-Relat</b>	ed In	juries and	llinesses	employee hea protects the o possible while occupational s	onfidentia the inforr safety and	lity of emp nation is b	inormation relation of in a manner the loyees to the ext leing used for irposes.	ig to nat ent Occupation	Year 20
You must record information about ever days away from work, or medical treatm care professional. You must also record use two lines for a single case if you ne	y work-related death an ent beyond first aid. Yo work-related injuries ai ed to. You must comple	nd about every wor u must also record nd illnesses that m ste an Injury and Ill	k-related injury or illness that in- d significant work-related injuries reet any of the specific recordin fness Incident Report (OSHA Fo	olves loss of consciousness, restricted work and illnesses that are diagnosed by a phys g criteria listed in 29 CFR Part 1904 a throug rm 301) or equivalent form for each injury or	activity or job transfel cian or licensed healt 1 1904.12. Feel free to thness recorded on th	6. 2.			Estable-iment name	Form approved OMB no. 1218-0176
Identify the person		Describe (	the case		C	assify th	e case		- fin	c ang
(A) (B) Case Employee's name	(C) Job title	(D) Date of injury	(E) Where the event occurred	(F) Describe injury or illness, parts of bod	y affected, that	ECK ONLY ied on the t case:	ONE box fo most seriou	r each case s outcome for	Enter the number of days the injured or ill worker was:	Check the "Injury" column or choose one type of illness:
no.	(e.g., Welder)	or onset of illness	(e.g., Loading dock north end)	and object/substance that directly inju or made person ill (e.g., Second degree bu	ed		Rei	nained at Work		(X) order ory n loss
				right forearm from acetylene torch)	Deal	Days a th from w	way Job trai ork or restr	iction able cases	work restriction	Injury Skin disc Respirate condition Poisonin Hearing All other illnesses
		month is			<u>م</u>		<b>D</b> 3	03	(K) (L)	(1) (2) (3) (4) (5) (6)
	l	/ month/day							days days	
I		month day				0	0		days days	
	Ì	month/day	Į			0			days days	с п п п п п
		/ month/day	Ĭ						days days	
Ĩ		month/day							daysdays	
	ĺ	month/day				0			days days	
		month day				0			days days	
Ĩ	Ì	month tay							daysdays	
	Ì	month/da,							days days	
ł		month/day							days days	
	Î	month, Jay						٥	days days	
I		month tay						0	days days	
				P	ige totals>	1	1	Ī	 	
Public reporting burden for this collection of in the instructions, search and gather the data ne to respond to the collection of information that how these structures are set of the second s	nformation is estimated to a eded, and complete and re ess it displays a currently va	worage 14 minutes p view the collection of did OMB control nur	er response, including time to review information. Persons are not require nber. If you have any comments	đ	sure to transfer these tota	its to the Sum	mary page (Fo	rm 300A) before , ou po	st ñ.	Injury in disorder Respiratory condition Poisoning caring lo All other illne
Analysis, Room N-3644, 200 Constitution Aven	iue, NW, Washington, DC 5	20210. Do not send th	neor, Ostra Onice of Statistical completed forms to this office.						Page of	(1) (2) (3) (4) (5) (6)
							-			

Attention: This form contains information relating to

1 )

	For
All establishments covered by Part 1904 must complete this Summary page, even if no work-related injuries or illnesses occurred during the year. Remember to review the to verify that the entries are complete and accurate before completing this summary. Using the Log, count the individual entries you made for each category. Then write the totals below, making sure you've added the entries from every page of the Log. If the processes when the summary is a summary.	Establishment information
Employees, former employees, and their representatives have the right to review the OSHA Form 300 in its entirety. They also have limited access to the OSHA Form 301 its entirety of the second secon	Your establishment name
na equivariant, see co or n' arr rachass, in Oorin's recurring rule, for runner details on the access provisions for these forms.	Street
Number of Cases	City Sate ZIP
Total number of     Total number of       Total number of     Total number of       deaths     cases with days       cases with iob     other recordable	Industry description (e.g., Manufacture of motor truck trailers)
away from work transfer or restriction cases	Standard Industrial Classification (SIC), if known (e.g., 3715)
(c) (H) (I) (J)	OR
Number of Days	North American Industrial Classification (NAICS), if known (e.g., 336212)
Total number of days away     Total number of days of job       from work     transfer or restriction	<b>Employment information</b> (If you don't have these figures, see the Worksheet on the back of this page to estimate.)
	Annual average number of employees
(5)	Total hours worked by all employees last year
Injury and Illness Types	Sign here
Total number of (M)	Knowingly falsifying this document may result in a fine.
(4) Poisonings	
(5) Hearing loss	I certify that I have examined this document and that to the best of n knowledge the entries are true, accurate, and complete.
	Company concurse ( ) Phone // / Date

Labor Materia

ATTACHMENT 5 SIGNATORY PAGE

# Attachment 4 – Site-Specific Health and Safety Orientation Signatory Page HEALTH AND SAFETY PLAN

Title	Name	Signature
Project Manager:	TBD	
Health and Safety Manager:	TBD	

I have read the attached Health and Safety Plan (HASP) and have received site-specific information and orientation regarding the identified physical, chemical, and biological hazards anticipated at this site. My signature certifies that I understand the procedures, equipment, and restrictions applicable to this project site and agree to abide by them.

Signature	Printed Name	Company	Date

# Attachment 4 – Health and Safety Orientation Signatory Page (continued)

Signature	Printed Name	Company	Date
L	Health and Safety Orientation (2 of 2)	Signatory Page	L



# **SAFETY DATA SHEET**

Version 6.0 Revision Date 10/24/2019 Print Date 01/05/2020

# SECTION 1: Identification of the substance/mixture and of the company/undertaking

# 1.1 Product identifiers

Product name:1,3-ButadieneProduct Number:295035Brand:AldrichIndex-No.:601-013-00-X

CAS-No. : 106-99-0

# 1.2 Relevant identified uses of the substance or mixture and uses advised against

Identified uses : Laboratory chemicals, Synthesis of substances

# 1.3 Details of the supplier of the safety data sheet

Company	:	Sigma-Aldrich Inc. 3050 Spruce Street ST. LOUIS MO 63103 UNITED STATES
Telephone Fax	:	+1 314 771-5765 +1 800 325-5052

# **1.4 Emergency telephone number**

Emergency Phone # : +1-703-527-3887

# **SECTION 2: Hazards identification**

# 2.1 Classification of the substance or mixture

# GHS Classification in accordance with 29 CFR 1910 (OSHA HCS)

Flammable gases (Category 1), H220 Gases under pressure (Liquefied gas), H280 Germ cell mutagenicity (Category 1B), H340 Carcinogenicity (Category 1A), H350

For the full text of the H-Statements mentioned in this Section, see Section 16.

# 2.2 GHS Label elements, including precautionary statements

Pictogram



Danger

Signal word Hazard statement(s) H220

Extremely flammable gas.

Aldrich - 295035

Page 1 of 11



H280 H340 H350	Contains gas under pressure; may explode if heated. May cause genetic defects. May cause cancer.
Precautionary statement(s)	
P201	Obtain special instructions before use.
P202	Do not handle until all safety precautions have been read and understood.
P210	Keep away from heat/sparks/open flames/hot surfaces. No smoking.
P280	Wear protective gloves/ protective clothing/ eye protection/ face protection.
P308 + P313	IF exposed or concerned: Get medical advice/ attention.
P377	Leaking gas fire: Do not extinguish, unless leak can be stopped safely.
P381	Eliminate all ignition sources if safe to do so.
P405	Store locked up.
P410 + P403 P501	Protect from sunlight. Store in a well-ventilated place. Dispose of contents/ container to an approved waste disposal plant.

# 2.3 Hazards not otherwise classified (HNOC) or not covered by GHS - none

# **SECTION 3: Composition/information on ingredients**

3.1	<b>Substances</b> Formula	:	C <sub>4</sub> H <sub>6</sub>		
	Molecular weight CAS-No. EC-No. Index-No.		54.09 g/mol 106-99-0 203-450-8 601-013-00-X		
	Component			Classification	Concentration
	1,3-Butadiene				
				Flam. Gas 1; Press. Gas Liquefied gas; Muta. 1B; Carc. 1A; H220, H280, H340, H350	<= 100 %

For the full text of the H-Statements mentioned in this Section, see Section 16.

# **SECTION 4: First aid measures**

# 4.1 Description of first aid measures

# **General advice**

Consult a physician. Show this safety data sheet to the doctor in attendance. Move out of dangerous area.

#### If inhaled

If breathed in, move person into fresh air. If not breathing, give artificial respiration. Consult a physician.

Aldrich - 295035

Page 2 of 11



# In case of skin contact

Wash off with soap and plenty of water. Consult a physician.

# In case of eye contact

Flush eyes with water as a precaution.

# If swallowed

Do NOT induce vomiting. Never give anything by mouth to an unconscious person. Rinse mouth with water. Consult a physician.

4.2 Most important symptoms and effects, both acute and delayed

The most important known symptoms and effects are described in the labelling (see section 2.2) and/or in section 11

**4.3 Indication of any immediate medical attention and special treatment needed** No data available

# **SECTION 5: Firefighting measures**

# 5.1 Extinguishing media

# Suitable extinguishing media

Use water spray, alcohol-resistant foam, dry chemical or carbon dioxide.

**5.2** Special hazards arising from the substance or mixture Carbon oxides

#### **5.3** Advice for firefighters Wear self-contained breathing apparatus for firefighting if necessary.

# 5.4 Further information

Use water spray to cool unopened containers.

# **SECTION 6: Accidental release measures**

# 6.1 Personal precautions, protective equipment and emergency procedures

Use personal protective equipment. Avoid breathing vapours, mist or gas. Ensure adequate ventilation. Remove all sources of ignition. Evacuate personnel to safe areas. Beware of vapours accumulating to form explosive concentrations. Vapours can accumulate in low areas.

For personal protection see section 8.

- **6.2 Environmental precautions** Prevent further leakage or spillage if safe to do so. Do not let product enter drains.
- **6.3 Methods and materials for containment and cleaning up** Clean up promptly by sweeping or vacuum.
- **6.4 Reference to other sections** For disposal see section 13.

# **SECTION 7: Handling and storage**

#### **7.1 Precautions for safe handling** Avoid inhalation of vapour or mist.

Avolu Initialation of vapour of

Aldrich - 295035

Page 3 of 11



Use explosion-proof equipment.Keep away from sources of ignition - No smoking.Take measures to prevent the build up of electrostatic charge. For precautions see section 2.2.

# 7.2 Conditions for safe storage, including any incompatibilities

Keep container tightly closed in a dry and well-ventilated place.

Recommended storage temperature 2 - 8 °C

Contents under pressure. Air sensitive. Light sensitive. Shock or heat may detonate May explode when heated. Handle and store under inert gas. Storage class (TRGS 510): 2A: Gases

# 7.3 Specific end use(s)

Apart from the uses mentioned in section 1.2 no other specific uses are stipulated

# SECTION 8: Exposure controls/personal protection

# 8.1 Control parameters

# Components with workplace control parameters

Component	CAS-No.	Value	Control	Basis			
			parameters				
1,3-Butadiene	106-99-0	TWA	2 ppm	USA. ACGIH Threshold Limit			
				Values (TLV)			
	Remarks	Cancer					
		Suspected	human carcinog	en			
		TWA	1 ppm	USA. Occupational Exposure			
				Limits (OSHA) - Table Z-1			
				Limits for Air Contaminants			
		Substance	listed; for more	information see OSHA document			
		29 CFR 191	10.1051; 29 CFR	1910.19(1)			
		STEL	5 ppm	USA. Occupational Exposure			
				Limits (OSHA) - Table Z-1			
				Limits for Air Contaminants			
		Substance	bstance listed; for more information see OSHA docum CFR 1910.1051; 29 CFR 1910.19(1) e 1910.1051 tential Occupational Carcinogen				
		29 CFR 191					
		See 1910.1					
		Potential O					
		See Append	dix A				
		PEL	1 ppm	OSHA Specifically Regulated			
				Chemicals/Carcinogens			
		1910.1051					
		This section	n applies to all o	ccupational exposures to 1,3-			
		Butadiene	ene (BD), Chemical Abstracts Service Registry No. P-0, except as provided in paragraph (a)(2) of this accept for the recordkeeping provisions in aph (m)(1) of this section, this section does not				
		106-99-0,					
		section. E>					
		paragraph					
		apply to the	the processing, use, or handling of products				
		containing	BD or to other w	vork operations and streams in			
		which BD is	s present where	objective data are reasonably			
		relied upon	that demonstra	te the work operation or the			
		product or	the group of pro	ducts or operations to which it			
		belongs ma	ay not reasonabl	y be foreseen to release BD in			
		airborne co	ncentrations at	or above the action level or in			

Aldrich - 295035

Page 4 of 11



	excess of the STEL under the expected conditions of processing, use, or handling that will cause the greatest possible release or in any plausible accident. This section also does not apply to work operations, products or streams where the only exposure to BD is from liquid mixtures containing 0.1% or less of BD by volume or the vapors released from such liquids, unless objective data become available that show that airborne concentrations generated by such mixtures can exceed the action level or STEL under reasonably predictable conditions of processing, use or handling that will cause the greatest possible release. Except for labeling requirements and requirements for emergency response, this section does not apply to the storage, transportation, distribution or sale of BD or liquid mixtures in intact containers or in transportation pipelines sealed in such a manner as to fully contain BD vapors or liquid. Where products or processes containing BD are exempted under paragraph (a)(2) of this section, the employer shall maintain records of the objective data supporting that exemption and the basis for the employer's reliance on the data, as provided in paragraph (m)(1) of this section 1,3-Butadiene means an organic compound with chemical formula CH2=CH-CH=CH2 that has a molecular weight of approximately 54.15 g/mole OSHA specifically regulated carcinogen
	STEL 5 ppm OSHA Specifically Regulated
	1910.1051 This section applies to all occupational exposures to 1,3- Butadiene (BD), Chemical Abstracts Service Registry No. 106-99-0, except as provided in paragraph (a)(2) of this section. Except for the recordkeeping provisions in paragraph (m)(1) of this section, this section does not apply to the processing, use, or handling of products containing BD or to other work operations and streams in which BD is present where objective data are reasonably relied upon that demonstrate the work operation or the product or the group of products or operations to which it belongs may not reasonably be foreseen to release BD in airborne concentrations at or above the action level or in excess of the STEL under the expected conditions of processing, use, or handling that will cause the greatest possible release or in any plausible accident. This section also does not apply to work operations, products or streams where the only exposure to BD is from liquid mixtures containing 0.1% or less of BD by volume or the vapors released from such liquids, unless objective data become available that show that airborne concentrations generated by such mixtures can exceed the action level or STEL under reasonably predictable conditions of processing, use or handling that will cause the greatest possible release. Except for labeling requirements and requirements for emergency response, this section does not apply to the storage, transportation, distribution or sale of BD or liquid

Aldrich - 295035

Page 5 of 11



mixtures in sealed in su liquid. Whe exempted u employer s supporting reliance on this section 1,3-Butadie formula CH approximat OSHA spec PEL	intact container uch a manner as ere products or p under paragraph hall maintain red that exemption the data, as pro- ene means an or 2=CH-CH=CH2 rely 54.15 g/mol ifically regulated 1 ppm	rs or in transportation pipelines to fully contain BD vapors or processes containing BD are (a)(2) of this section, the cords of the objective data and the basis for the employer's pivded in paragraph (m)(1) of ganic compound with chemical that has a molecular weight of e carcinogen California permissible exposure	
	2.2 mg/m3	contaminants (Title 8, Article 107)	
see section	5201		
STEL	5 ppm 11 mg/m3	California permissible exposure limits for chemical contaminants (Title 8, Article 107)	
see section 5201			

# **Biological occupational exposure limits**

Component	CAS-No.	Parameters	Value	Biological specimen	Basis
1,3-Butadiene	106-99-0	1,2 Dihydroxy- 4-(N- acetylcystei nyl)-butane	2.5 mg/l	Urine	ACGIH - Biological Exposure Indices (BEI)
	Remarks	End of shift (	As soon as	possible after exp	oosure ceases)
		Mixture of N-1 and N- 2(hydroxyb utenyl)valin e	2.5picom oles per gram Hemoglo bin	Hemoglobin (Hb) adducts in blood	ACGIH - Biological Exposure Indices (BEI)
		Not critical			

# 8.2 Exposure controls

# Appropriate engineering controls

Handle in accordance with good industrial hygiene and safety practice. Wash hands before breaks and at the end of workday.

# **Personal protective equipment**

# Eye/face protection

Face shield and safety glasses Use equipment for eye protection tested and approved under appropriate government standards such as NIOSH (US) or EN 166(EU).

# **Skin protection**

Handle with gloves. Gloves must be inspected prior to use. Use proper glove removal technique (without touching glove's outer surface) to avoid skin contact

Aldrich - 295035

Page 6 of 11



with this product. Dispose of contaminated gloves after use in accordance with applicable laws and good laboratory practices. Wash and dry hands.

Full contact Material: Fluorinated rubber Minimum layer thickness: 0.7 mm Break through time: 480 min Material tested: Vitoject® (KCL 890 / Aldrich Z677698, Size M)

Splash contact Material: Fluorinated rubber Minimum layer thickness: 0.7 mm Break through time: 480 min Material tested: Vitoject® (KCL 890 / Aldrich Z677698, Size M)

data source: KCL GmbH, D-36124 Eichenzell, phone +49 (0)6659 87300, e-mail sales@kcl.de, test method: EN374

If used in solution, or mixed with other substances, and under conditions which differ from EN 374, contact the supplier of the CE approved gloves. This recommendation is advisory only and must be evaluated by an industrial hygienist and safety officer familiar with the specific situation of anticipated use by our customers. It should not be construed as offering an approval for any specific use scenario.

# **Body Protection**

Complete suit protecting against chemicals, Flame retardant antistatic protective clothing., The type of protective equipment must be selected according to the concentration and amount of the dangerous substance at the specific workplace.

# **Respiratory protection**

Where risk assessment shows air-purifying respirators are appropriate use a fullface respirator with multi-purpose combination (US) or type AXBEK (EN 14387) respirator cartridges as a backup to engineering controls. If the respirator is the sole means of protection, use a full-face supplied air respirator. Use respirators and components tested and approved under appropriate government standards such as NIOSH (US) or CEN (EU).

# **Control of environmental exposure**

Prevent further leakage or spillage if safe to do so. Do not let product enter drains.

# **SECTION 9: Physical and chemical properties**

# 9.1 Information on basic physical and chemical properties

a)	Appearance	Form: Liquefied gas
b)	Odour	No data available
C)	Odour Threshold	No data available
d)	рН	No data available
e)	Melting point/freezing point	Melting point/range: -109 °C (-164 °F) - lit.
f)	Initial boiling point and boiling range	-4.5 °C 23.9 °F - lit.
g)	Flash point	-76 °C (-105 °F) - closed cup - Tested according to Annex V of Directive 67/548/EEC.
h)	Evaporation rate	No data available

Aldrich - 295035

Page 7 of 11



i)	Flammability (solid, gas)	No data available
j)	Upper/lower flammability or explosive limits	Upper explosion limit: 16.3 %(V) Lower explosion limit: 1.4 %(V)
k)	Vapour pressure	ca.2,400 hPa at 20 °C (68 °F) 3,200 hPa at 30 °C(86 °F) 5,700 hPa at 50 °C(122 °F)
I)	Vapour density	No data available
m)	Relative density	0.62 g/cm3 at 20 °C (68 °F)
n)	Water solubility	0.5 g/l at 20 °C (68 °F) - Tested according to Annex V of Directive 67/548/EEC.
0)	Partition coefficient: n-octanol/water	log Pow: 1.85 at 23 °C (73 °F)
p)	Auto-ignition temperature	No data available
q)	Decomposition temperature	No data available
r)	Viscosity	No data available
S)	Explosive properties	No data available
t)	Oxidizing properties	No data available
Oth	ner safety informatio	n

No data available

# **SECTION 10: Stability and reactivity**

# **10.1 Reactivity**

9.2

No data available

# **10.2** Chemical stability

Test for peroxide formation before using or discard after 3 months. Stable under recommended storage conditions.

#### **10.3 Possibility of hazardous reactions** No data available

10.4 Conditions to avoid

Heat, flames and sparks.

# **10.5** Incompatible materials

Oxidizing agents, Oxygen, Copper, Copper alloys, Carbides, Halogens, Metal oxides, Metals

# **10.6 Hazardous decomposition products**

Hazardous decomposition products formed under fire conditions. - Carbon oxides Other decomposition products - No data available In the event of fire: see section 5

Aldrich - 295035

Page 8 of 11



# SECTION 11: Toxicological information

# 11.1 Information on toxicological effects

# Acute toxicity

LD50 Oral - Rat - 5,480 mg/kg LC50 Inhalation - Rat - 4 h - 285 mg/l Dermal: No data available No data available

# Skin corrosion/irritation

# Serious eye damage/eye irritation

# **Respiratory or skin sensitisation**

# Germ cell mutagenicity

In vivo tests showed mutagenic effects

# Carcinogenicity

This is or contains a component that has been reported to be carcinogenic based on its IARC, OSHA, ACGIH, NTP, or EPA classification. Human carcinogen.

IARC: 1 - Group 1: Carcinogenic to humans (1,3-Butadiene)

NTP: Known - Known to be human carcinogen (1,3-Butadiene)

OSHA: OSHA specifically regulated carcinogen (1,3-Butadiene)

# **Reproductive toxicity**

No data available No data available

# Specific target organ toxicity - single exposure

# Specific target organ toxicity - repeated exposure

# Aspiration hazard

# **Additional Information**

RTECS: Not available

Cholinesterase inhibitors can cause heavy salivation and secretion in the lungs, lachrymation, blurred vision, involuntary defecation, diarrhea, tremor, ataxia, sweating, hypothermia, lowered heart rate, and/or a fall in blood pressure as a result of their action at cholinergic nerve sites., narcosis, Headache, Nausea, Vomiting, Dizziness, Drowsiness, Confusion., Weakness, Muscle cramps/spasms., Change in pupil size., Tremors, Seizures., Incoordination., Convulsions, Coma

Stomach - Irregularities - Based on Human Evidence Stomach - Irregularities - Based on Human Evidence

# SECTION 12: Ecological information

# **12.1 Toxicity**

Toxicity to fish LC50 - other fish - 71.5 mg/l - 24 h

Aldrich - 295035

Page 9 of 11



# 12.2 Persistence and degradability

# 12.3 Bioaccumulative potential

# 12.4 Mobility in soil

# 12.5 Results of PBT and vPvB assessment

PBT/vPvB assessment not available as chemical safety assessment not required/not conducted

# **12.6 Other adverse effects**

# SECTION 13: Disposal considerations

# **13.1 Waste treatment methods**

# Product

Burn in a chemical incinerator equipped with an afterburner and scrubber but exert extra care in igniting as this material is highly flammable. Offer surplus and non-recyclable solutions to a licensed disposal company. Contact a licensed professional waste disposal service to dispose of this material.

# **Contaminated packaging**

Dispose of as unused product.

# **SECTION 14: Transport information**

# DOT (US)

UN number: 1010 Class: 2.1 Proper shipping name: Butadienes, stabilized Reportable Quantity (RQ): 10 lbs Poison Inhalation Hazard: No

# IMDG

UN number: 1010 Class: 2.1 Proper shipping name: BUTADIENES, STABILIZED

# ΙΑΤΑ

UN number: 1010 Class: 2.1 Proper shipping name: Butadienes, stabilized IATA Passenger: Not permitted for transport

# **SECTION 15: Regulatory information**

# SARA 302 Components

No chemicals in this material are subject to the reporting requirements of SARA Title III, Section 302.

# SARA 313 Components

The following components are subject to reporting levels established by SARA Title III, Section 313:

CAS-No.

106-99-0

Aldrich - 295035

EMS-No: F-D, S-U

Page 10 of 11

The life science business of Merck KGaA, Darmstadt, Germany operates as MilliporeSigma in the US and Canada



**Revision Date** 

2007-07-01

# SARA 311/312 Hazards

Fire Hazard, Sudden Release of Pressure Hazard, Chronic Health Hazard

Massachusetts Right To Know Components	CAS No	Dovicion Data
1,3-Butadiene	106-99-0	2007-07-01
Pennsylvania Right To Know Components 1,3-Butadiene	CAS-No. 106-99-0	Revision Date 2007-07-01
New Jersey Right To Know Components 1,3-Butadiene	CAS-No. 106-99-0	Revision Date 2007-07-01
<b>California Prop. 65 Components</b> WARNING! This product contains a chemical known to the State of California to cause cancer.1,3-Butadiene	CAS-No. 106-99-0	Revision Date 2007-09-28
WARNING: This product contains a chemical known to the State of California to cause birth defects or other reproductive harm.1,3-Butadiene	CAS-No. 106-99-0	Revision Date 2007-09-28

# **SECTION 16: Other information**

#### **Further information**

Copyright 2018 Sigma-Aldrich Co. LLC. License granted to make unlimited paper copies for internal use only.

The above information is believed to be correct but does not purport to be all inclusive and shall be used only as a guide. The information in this document is based on the present state of our knowledge and is applicable to the product with regard to appropriate safety precautions. It does not represent any guarantee of the properties of the product. Sigma-Aldrich Corporation and its Affiliates shall not be held liable for any damage resulting from handling or from contact with the above product. See www.sigma-aldrich.com and/or the reverse side of invoice or packing slip for additional terms and conditions of sale.

The branding on the header and/or footer of this document may temporarily not visually match the product purchased as we transition our branding. However, all of the information in the document regarding the product remains unchanged and matches the product ordered. For further information please contact mlsbranding@sial.com.

Version: 6.0

Revision Date: 10/24/2019 Print Date: 01/05/2020

Aldrich - 295035

Page 11 of 11





sigma-aldrich.com

SAFETY DATA SHEET

Version 6.0 Revision Date 01/31/2017 Print Date 06/28/2019

#### 1. PRODUCT AND COMPANY IDENTIFICATION

1.1	Product identifiers Product name	:	Benzene
	Product Number Brand Index-No.	:	270709 SIGALD 601-020-00-8
	CAS-No.	:	71-43-2

#### 1.2 Relevant identified uses of the substance or mixture and uses advised against

Identified uses : Laboratory chemicals, Synthesis of substances

#### 1.3 Details of the supplier of the safety data sheet

Company	:	Sigma-Aldrich Inc. 3050 Spruce Street ST. LOUIS MO 63103 UNITED STATES
Telephone	:	+1 314 771-5765
Fax Emergency telephone nur	nhe	+1 000 323-3032

#### 1.4 Emergency telephone number

Emergency Phone # : +1-703-527-3887

# 2. HAZARDS IDENTIFICATION

#### 2.1 Classification of the substance or mixture

GHS Classification in accordance with 29 CFR 1910 (OSHA HCS)

Flammable liquids (Category 2), H225 Skin irritation (Category 2), H315 Eye irritation (Category 2A), H319 Germ cell mutagenicity (Category 1B), H340 Carcinogenicity (Category 1A), H350 Specific target organ toxicity - repeated exposure (Category 1), H372 Aspiration hazard (Category 1), H304 Acute aquatic toxicity (Category 3), H402 Chronic aquatic toxicity (Category 3), H412

For the full text of the H-Statements mentioned in this Section, see Section 16.

Danger

#### 2.2 GHS Label elements, including precautionary statements

Pictogram

Signal word



Hazard statement(s) H225

Highly flammable liquid and vapour.

H304	May be fatal if swallowed and enters airways.
H315	Causes skin irritation.
H319	Causes serious eye irritation.
H340	May cause genetic defects.
H350	May cause cancer.
H372	Causes damage to organs through prolonged or repeated exposure.
H412	Harmful to aquatic life with long lasting effects.
Precautionary statement(s)	
P201	Obtain special instructions before use.
P202	Do not handle until all safety precautions have been read and understood.
P210	Keep away from heat/sparks/open flames/hot surfaces. No smoking.
P233	Keep container tightly closed.
P240	Ground/bond container and receiving equipment.
P241	Use explosion-proof electrical/ ventilating/ lighting/ equipment.
P242	Use only non-sparking tools.
P243	Take precautionary measures against static discharge.
P260	Do not breathe dust/ fume/ gas/ mist/ vapours/ spray.
P264	Wash skin thoroughly after handling.
P270	Do not eat, drink or smoke when using this product.
P273	Avoid release to the environment.
P280	Wear protective gloves/ protective clothing/ eye protection/ face protection.
P301 + P310	IF SWALLOWED: Immediately call a POISON CENTER/doctor.
P303 + P361 + P353	IF ON SKIN (or hair): Take off immediately all contaminated clothing. Rinse skin with water/shower.
P305 + P351 + P338	IF IN EYES: Rinse cautiously with water for several minutes. Remove contact lenses, if present and easy to do. Continue rinsing.
P308 + P313	IF exposed or concerned: Get medical advice/ attention.
P331	Do NOT induce vomiting.
P332 + P313	If skin irritation occurs: Get medical advice/ attention.
P337 + P313	If eye irritation persists: Get medical advice/ attention.
P362	Take off contaminated clothing and wash before reuse.
P370 + P378	In case of fire: Use dry sand, dry chemical or alcohol-resistant foam to extinguish.
P403 + P235	Store in a well-ventilated place. Keep cool.
P405	Store locked up.
P501	Dispose of contents/ container to an approved waste disposal plant.

# 2.3 Hazards not otherwise classified (HNOC) or not covered by GHS - none

# **3. COMPOSITION/INFORMATION ON INGREDIENTS**

3.1 Substances

Formula	:	С <sub>6</sub> Н <sub>6</sub>
Molecular weight	:	78.11 g/mol
CAS-No.	:	71-43-2
EC-No.	:	200-753-7
Index-No.	:	601-020-00-8

# Hazardous components

Component	Classification	Concentration
Benzene		
	Flam. Liq. 2; Skin Irrit. 2; Eye Irrit. 2A; Muta. 1B; Carc. 1A; STOT RE 1; Asp. Tox. 1; Aquatic Acute 3; Aquatic Chronic 3; H225, H304, H315, H319, H340, H350, H372,	<= 100 %

H412	
------	--

For the full text of the H-Statements mentioned in this Section, see Section 16.

# 4. FIRST AID MEASURES

## 4.1 Description of first aid measures

#### **General advice**

Consult a physician. Show this safety data sheet to the doctor in attendance. Move out of dangerous area.

# If inhaled

If breathed in, move person into fresh air. If not breathing, give artificial respiration. Consult a physician.

#### In case of skin contact

Wash off with soap and plenty of water. Take victim immediately to hospital. Consult a physician.

#### In case of eye contact

Rinse thoroughly with plenty of water for at least 15 minutes and consult a physician.

#### If swallowed

Do NOT induce vomiting. Never give anything by mouth to an unconscious person. Rinse mouth with water. Consult a physician.

# **4.2** Most important symptoms and effects, both acute and delayed The most important known symptoms and effects are described in the labelling (see section 2.2) and/or in section 11

**4.3 Indication of any immediate medical attention and special treatment needed** No data available

# **5. FIREFIGHTING MEASURES**

# 5.1 Extinguishing media

#### **Suitable extinguishing media** Use water spray, alcohol-resistant foam, dry chemical or carbon dioxide.

5.2 Special hazards arising from the substance or mixture Carbon oxides

#### 5.3 Advice for firefighters

Wear self-contained breathing apparatus for firefighting if necessary.

# 5.4 Further information

Use water spray to cool unopened containers.

# 6. ACCIDENTAL RELEASE MEASURES

# 6.1 Personal precautions, protective equipment and emergency procedures

Use personal protective equipment. Avoid breathing vapours, mist or gas. Ensure adequate ventilation. Remove all sources of ignition. Evacuate personnel to safe areas. Beware of vapours accumulating to form explosive concentrations. Vapours can accumulate in low areas. For personal protection see section 8.

#### 6.2 Environmental precautions

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

# 6.3 Methods and materials for containment and cleaning up

Contain spillage, and then collect with an electrically protected vacuum cleaner or by wet-brushing and place in container for disposal according to local regulations (see section 13).

#### 6.4 Reference to other sections

For disposal see section 13.

# 7. HANDLING AND STORAGE

# 7.1 Precautions for safe handling

Avoid contact with skin and eyes. Avoid inhalation of vapour or mist.

Use explosion-proof equipment.Keep away from sources of ignition - No smoking.Take measures to prevent the build up of electrostatic charge.

For precautions see section 2.2.

# 7.2 Conditions for safe storage, including any incompatibilities

Keep container tightly closed in a dry and well-ventilated place. Containers which are opened must be carefully resealed and kept upright to prevent leakage.

# 7.3 Specific end use(s)

Apart from the uses mentioned in section 1.2 no other specific uses are stipulated

# 8. EXPOSURE CONTROLS/PERSONAL PROTECTION

# 8.1 Control parameters

# Components with workplace control parameters

Component	CAS-No.	Value	Control parameters	Basis
Benzene	71-43-2	TWA	0.5 ppm	USA. ACGIH Threshold Limit Values (TLV)
	Remarks	Leukemia Substances for which there is a Biological Exposure Index or Indices (see BEI® section) Confirmed human carcinogen Danger of cutaneous absorption		
		STEL	2.5 ppm	USA. ACGIH Threshold Limit Values (TLV)
		Leukemia Substances for which there is a Biological Exposure Index or Indices (see BEI® section) Confirmed human carcinogen Danger of cutaneous absorption		
		TWĂ	10 ppm	USA. Occupational Exposure Limits (OSHA) - Table Z-2
		Z37.40-1969		
		CEIL	25 ppm	USA. Occupational Exposure Limits (OSHA) - Table Z-2
		Z37.40-1969		
		Peak	50 ppm	USA. Occupational Exposure Limits (OSHA) - Table Z-2
		Z37.40-1969		
		See 1910.1028. See Table Z-2 for the limits applicable in the operations or sectors excluded in 1910.1028 The final benzene standard in 1910.1028 applies to all occupational exposures to benzene except some subsegments of industry where exposures are consistently under the action level (i.e., distribution and sale of fuels, sealed containers and pipelines, coke production, oil and gas drilling and production, natural gas processing, and the percentage exclusion for liquid mixtures); for the excepted subsegments, the benzene limits in Table Z-2 apply.		
		TWA	0.1 ppm	USA. NIOSH Recommended Exposure Limits
		Potential Occupational Carcinogen See Appendix A		
		ST	ST 1 ppm USA. NIOSH Recommended	
		Potential Oco See Appendi	cupational Car x A	inogen
<b>Biological occupation</b>	onal exposure	limits		
Component	CAS-No.	Parameters	Value	Biological Basis

Benzene	71-43-2	S- Phenylmerca pturic acid	0.0300 mg/g	In urine	ACGIH - Biological Exposure Indices (BEI)
	Remarks	End of shift (As soon as possible after exposure ceases)			
		t,t-Muconic	0.5000	In urine	ACGIH - Biological
		acid	mg/g		Exposure Indices
					(BEI)
		End of shift (As soon as possible after exposure ceases)			

# 8.2 Exposure controls

# Appropriate engineering controls

Handle in accordance with good industrial hygiene and safety practice. Wash hands before breaks and at the end of workday.

# Personal protective equipment

# Eye/face protection

Face shield and safety glasses Use equipment for eye protection tested and approved under appropriate government standards such as NIOSH (US) or EN 166(EU).

#### Skin protection

Handle with gloves. Gloves must be inspected prior to use. Use proper glove removal technique (without touching glove's outer surface) to avoid skin contact with this product. Dispose of contaminated gloves after use in accordance with applicable laws and good laboratory practices. Wash and dry hands.

#### Full contact

Material: Fluorinated rubber Minimum layer thickness: 0.7 mm Break through time: 480 min Material tested:Vitoject® (KCL 890 / Aldrich Z677698, Size M)

Splash contact Material: Fluorinated rubber Minimum layer thickness: 0.7 mm Break through time: 480 min Material tested:Vitoject® (KCL 890 / Aldrich Z677698, Size M)

data source: KCL GmbH, D-36124 Eichenzell, phone +49 (0)6659 87300, e-mail sales@kcl.de, test method: EN374

If used in solution, or mixed with other substances, and under conditions which differ from EN 374, contact the supplier of the CE approved gloves. This recommendation is advisory only and must be evaluated by an industria situation of anticipated use by our customers. It should not be construed as offering an approval for any specific use scenario.

Handle with gloves. Gloves must be inspected prior to use. Use proper glove removal technique (without touching glove's outer surface) to avoid skin contact with this product. Dispose of contaminated gloves after use in accordance with applicable laws and good laboratory practices. Wash and dry hands.

#### **Body Protection**

Complete suit protecting against chemicals, Flame retardant antistatic protective clothing., The type of protective equipment must be selected according to the concentration and amount of the dangerous substance at the specific workplace.

# Respiratory protection

Where risk assessment shows air-purifying respirators are appropriate use (US) or type ABEK (EN 14387) respirator cartridges as a backup to enginee protection, use a full-face supplied air respirator. Use respirators and components tested and approved under appropriate government standards such as NIOSH (US) or CEN (EU).

#### Control of environmental exposure

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

# 9. PHYSICAL AND CHEMICAL PROPERTIES

# 9.1 Information on basic physical and chemical properties

a) Appearance Form: liquid Colour: colourless

b)	Odour	No data available
c)	Odour Threshold	No data available
d)	pH	No data available
e)	Melting point/freezing point	5.5 °C (41.9 °F)
f)	Initial boiling point and boiling range	80.0 - 80.2 °C (176.0 - 176.4 °F)
g)	Flash point	-11.0 °C (12.2 °F) - closed cup
h)	Evaporation rate	No data available
i)	Flammability (solid, gas)	No data available
j)	Upper/lower flammability or explosive limits	Upper explosion limit: 8 %(V) Lower explosion limit: 1.3 %(V)
k)	Vapour pressure	221.3 hPa at 37.7 °C (99.9 °F) 99.5 hPa at 20.0 °C(68.0 °F)
I)	Vapour density	No data available
m)	Relative density	0.88 g/cm3
n)	Water solubility	ca.1.88 g/l at 23.5 °C (74.3 °F) - soluble
o)	Partition coefficient: n- octanol/water	log Pow: 2.13 at 25 °C (77 °F)
p)	Auto-ignition temperature	562.0 °C (1043.6 °F)
q)	Decomposition temperature	No data available
r)	Viscosity	No data available
s)	Explosive properties	No data available
t)	Oxidizing properties	No data available
<b>Oth</b> No	<b>her safety information</b> data available	

# **10. STABILITY AND REACTIVITY**

**10.1 Reactivity** No data available

9.2

- **10.2** Chemical stability Stable under recommended storage conditions.
- **10.3 Possibility of hazardous reactions** Vapours may form explosive mixture with air.
- **10.4** Conditions to avoid Heat, flames and sparks.
- **10.5** Incompatible materials acids, Bases, Halogens, Strong oxidizing agents, Metallic salts

# Hazardous decomposition products Hazardous decomposition products formed under fire conditions. - Carbon oxides Other decomposition products - No data available In the event of fire: see section 5

# 11. TOXICOLOGICAL INFORMATION

#### 11.1 Information on toxicological effects

#### Acute toxicity

LD50 Oral - Rat - male - > 5,960 mg/kg(Benzene) (OECD Test Guideline 401) LC50 Inhalation - Rat - female - 4 h - 43.7 mg/l(Benzene) (OECD Test Guideline 403) LD50 Dermal - Rabbit - 8,263 mg/kg(Benzene) No data available(Benzene)

#### Skin corrosion/irritation

Skin - Rabbit(Benzene) Result: Skin irritation - 4 h (OECD Test Guideline 404)

#### Serious eye damage/eye irritation

Eyes - Rabbit(Benzene) Result: Eye irritation

# Respiratory or skin sensitisation

Maximisation Test - Guinea pig(Benzene) Result: Does not cause skin sensitisation.

#### Germ cell mutagenicity

Laboratory experiments have shown mutagenic effects.(Benzene) In vivo tests showed mutagenic effects(Benzene) Chinese hamster lung cells Result: positive OECD Test Guideline 475(Benzene) Mouse - male Result: positive

#### Carcinogenicity

This is or contains a component that has been reported to be carcinogenic classification.(Benzene) Human carcinogen.(Benzene)

IARC: 1 - Group 1: Carcinogenic to humans (Benzene)

NTP: Known to be human carcinogen (Benzene)

OSHA: OSHA specifically regulated carcinogen (Benzene)

#### **Reproductive toxicity**

**Specific target organ toxicity - single exposure** No data available(Benzene)

Specific target organ toxicity - repeated exposure No data available

#### **Aspiration hazard**

May be fatal if swallowed and enters airways.(Benzene)

#### **Additional Information**

Repeated dose toxicity - Rat - male and female - Oral - No observed adverse effect level - 100 mg/kg(Benzene) RTECS: CY1400000

Nausea, Dizziness, Headache, narcosis, Inhalation of high concentrations of benzene may have an initial stimulato exhilaration, nervous excitation and/or giddiness, depression, drowsiness chest, breathlessness, and loss of consciousness. Tremors, convulsions, a collapse can occur in a few minutes to several hours following severe exp causes pulmonary edema and hemorrhage of pulmonary tissue. Direct skin co contact may result in drying, scaling dermatitis, or development of secon hematopoietic system. Bleeding from the nose, gums, or mucous membranes a

leukopenia, thrombocytopenia, aplastic anemia, and leukemia may occur as normal, aplastic or hyperplastic, and may not correlate with peripheral b benzene exposure may be delayed for many months or years after the actual, Blood disorders(Benzene)

Stomach - Irregularities - Based on Human Evidence Stomach - Irregularities - Based on Human Evidence(Benzene)

# **12. ECOLOGICAL INFORMATION**

# 12.1 Toxicity

Toxicity to fish	LC50 - Pimephales promelas (fathead minnow) - 15.00 - 32.00 mg/l - 96 h(Benzene)
Toxicity to daphnia and other aquatic invertebrates	EC50 - Ceriodaphnia dubia (water flea) - 17.2 mg/I - 48 h(Benzene)
Toxicity to algae	Growth inhibition EC50 - Pseudokirchneriella subcapitata (green algae) - 100 mg/l - 72 h(Benzene) (OECD Test Guideline 201)

#### 12.2 Persistence and degradability

Biodegradability aerobic - Exposure time 28 d(Benzene) Result: 96 % - Readily biodegradable. (OECD Test Guideline 301F)

#### 12.3 Bioaccumulative potential

Bioaccumulation Leuciscus idus (Golden orfe) - 3 d - 0.05 mg/l(Benzene)

Bioconcentration factor (BCF): 10

# 12.4 Mobility in soil

No data available(Benzene)

**12.5 Results of PBT and vPvB assessment** PBT/vPvB assessment not available as chemical safety assessment not required/not conducted

10 lbs

#### 12.6 Other adverse effects

An environmental hazard cannot be excluded in the event of unprofessional handling or disposal. Harmful to aquatic life with long lasting effects.

#### **13. DISPOSAL CONSIDERATIONS**

#### 13.1 Waste treatment methods

#### Product

Contact a licensed professional waste disposal service to dispose of this material. Burn in a chemical incinerator equipped with an afterburner and scrubber b highly flammable. Offer surplus and non-recyclable solutions to a licensed disposal company.

#### Contaminated packaging

Dispose of as unused product.

# **14. TRANSPORT INFORMATION**

## DOT (US)

UN number: 1114	Class: 3
Proper shipping name:	Benzene
Reportable Quantity (R	Q) :

Packing group: II

Poison Inhalation Hazard: No

#### IMDG

# IATA

UN number: 1114	Class: 3
Proper shipping name	: Benzene

Packing group: II

# **15. REGULATORY INFORMATION**

# SARA 302 Components No chemicals in this material are subject to the reporting requirements of SARA Title III, Section 302. SARA 313 Components

CARA UTU Componento		
The following components are subject to reporting levels esta	ablished by SARA Title	III, Section 313:
	CAS-No	Revision Dat

	CAS-No.	Revision Date
Benzene	71-43-2	2007-07-01
<b>SARA 311/312 Hazards</b> Fire Hazard, Acute Health Hazard, Chronic Health Hazard		
Massachusetts Right To Know Components		
	CAS-No.	Revision Date
Benzene	71-43-2	2007-07-01
Pennsylvania Right To Know Components		
	CAS-No.	Revision Date
Benzene	71-43-2	2007-07-01
New Jersey Right To Know Components		
	CAS-No.	Revision Date
Benzene	71-43-2	2007-07-01
California Prop. 65 Components		
WARNING! This product contains a chemical known to the	CAS-No.	Revision Date
State of California to cause cancer.	71-43-2	2009-02-01
Benzene		
WARNING: This product contains a chemical known to the	CAS-No.	Revision Date
State of California to cause birth defects or other reproductive	71-43-2	2009-02-01
harm.		
Delizene		

# **16. OTHER INFORMATION**

# Full text of H-Statements referred to under sections 2 and 3.

H225	Highly flammable liquid and vapour.
H304	May be fatal if swallowed and enters airways.
H315	Causes skin irritation.
H319	Causes serious eye irritation.
H340	May cause genetic defects.
H350	May cause cancer.
H372	Causes damage to organs through prolonged or repeated exposure.
H402	Harmful to aquatic life.
H412	Harmful to aquatic life with long lasting effects.

# HMIS Rating

Health hazard:	2
Chronic Health Hazard:	*

Flammability:	3
Physical Hazard	0

# **NFPA** Rating

Health hazard:	2
Fire Hazard:	3
Reactivity Hazard:	0

# Further information

Copyright 2016 Sigma-Aldrich Co. LLC. License granted to make unlimited paper copies for internal use only. The above information is believed to be correct but does not purport to be all inclusive and shall be used only as a guide. The information in this document is based on the present state of our knowledge and is applicable to the product with regard to appropriate safety precautions. It does not represent any guarantee of the properties of the product. Sigma-Aldrich Corporation and its Affiliates shall not be held liable for any damage resulting from handling or from contact with the above product. See www.sigma-aldrich.com and/or the reverse side of invoice or packing slip for additional terms and conditions of sale.

#### **Preparation Information**

Sigma-Aldrich Corporation Product Safety – Americas Region 1-800-521-8956 Version: 6.0 Rev

Revision Date: 01/31/2017

Print Date: 06/28/2019
sigma-aldrich.com

# SAFETY DATA SHEET

Version 6.1 Revision Date 07/17/2018 Print Date 01/21/2019

#### 1. PRODUCT AND COMPANY IDENTIFICATION 1.1 **Product identifiers** Product name Benzo[<l>b</>)fluoranthene Product Number : 48490 Brand Supelco Index-No. 601-034-00-4 CAS-No. ÷ 205-99-2 1.2 Relevant identified uses of the substance or mixture and uses advised against Identified uses : Laboratory chemicals, Synthesis of substances 1.3 Details of the supplier of the safety data sheet Company : Sigma-Aldrich Inc. 3050 Spruce Street ST. LOUIS MO 63103 UNITED STATES Telephone +1 314 771-5765 : Fax +1 800 325-5052

#### 1.4 Emergency telephone number

Emergency Phone # : (314) 776-6555

#### 2. HAZARDS IDENTIFICATION

#### 2.1 Classification of the substance or mixture

#### GHS Classification in accordance with 29 CFR 1910 (OSHA HCS) Carcinogenicity (Category 1B), H350

Acute aquatic toxicity (Category 1), H400

Chronic aquatic toxicity (Category 1), H410

For the full text of the H-Statements mentioned in this Section, see Section 16.

#### 2.2 GHS Label elements, including precautionary statements

Pictogram



Signal word
Hazard statement(s)
H350
H410

Danger

May cause cancer. Very toxic to aquatic life with long lasting effects.

Precautionary statement(s)	
P201	Obtain special instructions before use.
P202	Do not handle until all safety precautions have been read and understood.
P273	Avoid release to the environment.
P281	Use personal protective equipment as required.
P308 + P313	IF exposed or concerned: Get medical advice/ attention.
P391	Collect spillage.
P405	Store locked up.
P501	Dispose of contents/ container to an approved waste disposal plant.

2.3 Hazards not otherwise classified (HNOC) or not covered by GHS - none

3 4-Benzofluoranthene

#### **3. COMPOSITION/INFORMATION ON INGREDIENTS**

# 3.1 Substances

eynenyme	•	o, i Donzonaoranarono
Formula	:	C <sb>20H<sb>12</sb></sb>
Molecular weight	:	252.31 g/mol
CAS-No.	:	205-99-2
EC-No.	:	205-911-9
Index-No.	:	601-034-00-4

#### Hazardous components

Component	Classification	Concentration
Benz[e]acephenanthrylene		
	Carc. 1B; Aquatic Acute 1; Aquatic Chronic 1; H350, H410	<= 100 %

For the full text of the H-Statements mentioned in this Section, see Section 16.

#### 4. FIRST AID MEASURES

#### 4.1 Description of first aid measures

#### General advice

Consult a physician. Show this safety data sheet to the doctor in attendance. Move out of dangerous area.

#### If inhaled

If breathed in, move person into fresh air. If not breathing, give artificial respiration. Consult a physician.

#### In case of skin contact

Wash off with soap and plenty of water. Consult a physician.

#### In case of eye contact

Flush eyes with water as a precaution.

#### If swallowed

Never give anything by mouth to an unconscious person. Rinse mouth with water. Consult a physician.

#### 4.2 Most important symptoms and effects, both acute and delayed

The most important known symptoms and effects are described in the labelling (see section 2.2) and/or in section 11

#### **4.3 Indication of any immediate medical attention and special treatment needed** No data available

#### **5. FIREFIGHTING MEASURES**

#### 5.1 Extinguishing media

#### Suitable extinguishing media

Use water spray, alcohol-resistant foam, dry chemical or carbon dioxide.

# 5.2 Special hazards arising from the substance or mixture Carbon oxides

#### **5.3** Advice for firefighters Wear self-contained breathing apparatus for firefighting if necessary.

5.4 Further information No data available

# 6. ACCIDENTAL RELEASE MEASURES

#### 6.1 Personal precautions, protective equipment and emergency procedures

Use personal protective equipment. Avoid dust formation. Avoid breathing vapours, mist or gas. Ensure adequate ventilation. Evacuate personnel to safe areas. Avoid breathing dust. For personal protection see section 8.

#### 6.2 Environmental precautions

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

#### 6.3 Methods and materials for containment and cleaning up Pick up and arrange disposal without creating dust. Sweep up and shovel. Keep in suitable, closed containers for disposal.

#### 6.4 Reference to other sections

For disposal see section 13.

#### 7. HANDLING AND STORAGE

#### 7.1 Precautions for safe handling

Avoid formation of dust and aerosols. Further processing of solid materials may result in the formation of combustible dusts. The potential for combustible dust formation should be taken into consideration before additional processing occurs.

Provide appropriate exhaust ventilation at places where dust is formed. For precautions see section 2.2.

#### 7.2 Conditions for safe storage, including any incompatibilities

Keep container tightly closed in a dry and well-ventilated place.

Recommended storage temperature 2 - 8 °C

Storage class (TRGS 510): 6.1D: Non-combustible, acute toxic Cat.3 / toxic hazardous materials or hazardous materials causing chronic effects

#### 7.3 Specific end use(s)

Apart from the uses mentioned in section 1.2 no other specific uses are stipulated

#### 8. EXPOSURE CONTROLS/PERSONAL PROTECTION

#### 8.1 Control parameters

#### Components with workplace control parameters

Remarks

	Remarks	Cancer Substances for which there is a Biological Exposure Index or Indices (see BEI® section), see BEI® for Polycyclic Aromatic Hydrocarbons (PAHs) Exposure by all routes should be carefully controlled to levels as low as possible. Suspected human carcinogen			
Biological occupational exposure limits					
Component	CAS-No.	Parameters	Value	Biological specimen	Basis
Benz[e]acephenant hrylene	205-99-2	1- Hydroxypyren e		Urine	ACGIH - Biological Exposure Indices (BEI)

End of shift at end of workweek

#### 8.2 Exposure controls

#### Appropriate engineering controls

Handle in accordance with good industrial hygiene and safety practice. Wash hands before breaks and at the end of workday.

#### Personal protective equipment

#### Eye/face protection

Safety glasses with side-shields conforming to EN166 Use equipment for eye protection tested and approved under appropriate government standards such as NIOSH (US) or EN 166(EU).

#### **Skin protection**

Handle with gloves. Gloves must be inspected prior to use. Use proper glove removal technique (without touching glove's outer surface) to avoid skin contact with this product. Dispose of contaminated gloves after use in accordance with applicable laws and good laboratory practices. Wash and dry hands.

Full contact Material: Nitrile rubber Minimum layer thickness: 0.11 mm Break through time: 480 min Material tested:Dermatril® (KCL 740 / Aldrich Z677272, Size M)

Splash contact Material: Nitrile rubber Minimum layer thickness: 0.11 mm Break through time: 480 min Material tested:Dermatril® (KCL 740 / Aldrich Z677272, Size M)

data source: KCL GmbH, D-36124 Eichenzell, phone +49 (0)6659 87300, e-mail sales@kcl.de, test method: EN374

If used in solution, or mixed with other substances, and under conditions which differ from EN 374, contact the supplier of the CE approved gloves. This recommendation is advisory only and must be evaluated by an industrial hygienist and safety officer familiar with the specific situation of anticipated use by our customers. It should not be construed as offering an approval for any specific use scenario.

#### **Body Protection**

Impervious clothing, The type of protective equipment must be selected according to the concentration and amount of the dangerous substance at the specific workplace.

#### **Respiratory protection**

Where risk assessment shows air-purifying respirators are appropriate use a full-face particle respirator type N100 (US) or type P3 (EN 143) respirator cartridges as a backup to engineering controls. If the respirator is the sole means of protection, use a full-face supplied air respirator. Use respirators and components tested and approved under appropriate government standards such as NIOSH (US) or CEN (EU).

#### Control of environmental exposure

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

#### 9. PHYSICAL AND CHEMICAL PROPERTIES

#### 9.1 Information on basic physical and chemical properties

a)	Appearance	Form: solid
b)	Odour	No data available
c)	Odour Threshold	No data available
d)	рН	No data available
e)	Melting point/freezing point	Melting point/range: 163 - 165 °C (325 - 329 °F) - lit.
f)	Initial boiling point and boiling range	No data available
g)	Flash point	No data available
h)	Evaporation rate	No data available
i)	Flammability (solid, gas)	No data available

j)	Upper/lower flammability or explosive limits	No data available
k)	Vapour pressure	No data available
I)	Vapour density	No data available
m)	Relative density	No data available
n)	Water solubility	No data available
o)	Partition coefficient: n- octanol/water	No data available
p)	Auto-ignition temperature	No data available
q)	Decomposition temperature	No data available
r)	Viscosity	No data available
s)	Explosive properties	No data available
t)	Oxidizing properties	No data available
<b>Oth</b> No	ner safety information data available	

#### **10. STABILITY AND REACTIVITY**

10.1 Reactivity No data available

9.2

- **10.2 Chemical stability** Stable under recommended storage conditions.
- **10.3 Possibility of hazardous reactions** No data available
- **10.4 Conditions to avoid** No data available
- **10.5** Incompatible materials Strong oxidizing agents

# Hazardous decomposition products Hazardous decomposition products formed under fire conditions. - Carbon oxides Other decomposition products - No data available In the event of fire: see section 5

#### **11. TOXICOLOGICAL INFORMATION**

#### 11.1 Information on toxicological effects

#### Acute toxicity

TDLo Oral - Mouse - 7.57 mg/kg Remarks: Liver:Changes in liver weight. Endocrine:Changes in thymus weight. Inhalation: No data available Dermal: No data available No data available

#### Skin corrosion/irritation

No data available

Serious eye damage/eye irritation No data available

#### Respiratory or skin sensitisation

No data available

#### Germ cell mutagenicity

No data available

#### Carcinogenicity

This product is or contains a component that has been reported to be probably carcinogenic based on its IARC, OSHA, ACGIH, NTP, or EPA classification.

Possible human carcinogen

- IARC: 2B Group 2B: Possibly carcinogenic to humans (Benz[e]acephenanthrylene)
- NTP: RAHC Reasonably anticipated to be a human carcinogen (Benz[e]acephenanthrylene)
- OSHA: No component of this product present at levels greater than or equal to 0.1% is on OSHA's list of regulated carcinogens.

#### **Reproductive toxicity**

No data available No data available

Specific target organ toxicity - single exposure No data available

Specific target organ toxicity - repeated exposure No data available

Aspiration hazard No data available

### Additional Information

RTECS: Not available

To the best of our knowledge, the chemical, physical, and toxicological properties have not been thoroughly investigated.

#### **12. ECOLOGICAL INFORMATION**

#### 12.1 Toxicity

Toxicity to daphnia and Immobilization EC50 - Daphnia magna (Water flea) - > 1.024 mg/l - 24 h(Benz[e]acephenanthrylene) invertebrates

#### 12.2 Persistence and degradability

No data available

#### 12.3 Bioaccumulative potential

No data available

#### 12.4 Mobility in soil

No data available(Benz[e]acephenanthrylene)

## 12.5 Results of PBT and vPvB assessment

PBT/vPvB assessment not available as chemical safety assessment not required/not conducted

#### 12.6 Other adverse effects

An environmental hazard cannot be excluded in the event of unprofessional handling or disposal. Very toxic to aquatic life.

#### **13. DISPOSAL CONSIDERATIONS**

#### 13.1 Waste treatment methods

#### Product

Offer surplus and non-recyclable solutions to a licensed disposal company. Contact a licensed professional waste disposal service to dispose of this material. Dissolve or mix the material with a combustible solvent and burn in a chemical incinerator equipped with an afterburner and scrubber.

#### Contaminated packaging

Dispose of as unused product.

#### **14. TRANSPORT INFORMATION**

#### DOT (US)

Not dangerous goods **IMDG** UN number: 3077 Class: 9 Packing group: III EMS-No: F-A, S-F Proper shipping name: ENVIRONMENTALLY HAZARDOUS SUBSTANCE, SOLID, N.O.S. (Benz[e]acephenanthrylene) Marine pollutant : yes

#### IATA

UN number: 3077 Class: 9 Packing group: III Proper shipping name: Environmentally hazardous substance, solid, n.o.s. (Benz[e]acephenanthrylene)

#### **Further information**

EHS-Mark required (ADR 2.2.9.1.10, IMDG code 2.10.3) for single packagings and combination packagings containing inner packagings with Dangerous Goods > 5L for liquids or > 5kg for solids.

#### **15. REGULATORY INFORMATION**

#### SARA 302 Components

No chemicals in this material are subject to the reporting requirements of SARA Title III, Section 302.

#### SARA 313 Components

The following components are subject to reporting levels establis	hed by SARA Title III	, Section 313:
	CAS-No.	Revision Date
Benz[e]acephenanthrylene	205-99-2	2007-03-01
SARA 311/312 Hazards Chronic Health Hazard		
Massachusetts Right To Know Components		
	CAS-No.	Revision Date
Benz[e]acephenanthrylene	205-99-2	2007-03-01
Pennsylvania Right To Know Components		
	CAS-No.	Revision Date
Benz[e]acephenanthrylene	205-99-2	2007-03-01
California Prop. 65 Components		
, which is/are known to the State of California to cause cancer.	CAS-No.	Revision Date
For more information go to www.P65Warnings.ca.gov.	205-99-2	2007-09-28
Benz[e]acephenanthrylene		

#### **16. OTHER INFORMATION**

Full text of H-Statements referred to under sections 2 and 3.

H400	Very toxic to aquatic life.
H410	Very toxic to aquatic life with long lasting effects.

#### Further information

Copyright 2016 Sigma-Aldrich Co. LLC. License granted to make unlimited paper copies for internal use only. The above information is believed to be correct but does not purport to be all inclusive and shall be used only as a guide. The information in this document is based on the present state of our knowledge and is applicable to the product with regard to appropriate safety precautions. It does not represent any guarantee of the properties of the product. Sigma-Aldrich Corporation and its Affiliates shall not be held liable for any damage resulting from handling or from contact with the above product. See www.sigma-aldrich.com and/or the reverse side of invoice or packing slip for additional terms and conditions of sale.

#### **Preparation Information**

Sigma-Aldrich Corporation Product Safety – Americas Region 1-800-521-8956 Version: 6.1

Revision Date: 07/17/2018

Print Date: 01/21/2019

# SIGMA-ALDRICH

# SAFETY DATA SHEET

Version 5.8 Revision Date 02/02/2018 Print Date 10/19/2018

#### **1. PRODUCT AND COMPANY IDENTIFICATION**

1.1	Product identifiers Product name	:	Benzo[ <i>a</i> ]pyrene
	Product Number Brand Index-No.	:	48564 Supelco 601-032-00-3
	CAS-No.	:	50-32-8
1.2	Relevant identified uses	of the	substance or mixture and uses advised against

Identified uses : Laboratory chemicals, Synthesis of substances

#### 1.3 Details of the supplier of the safety data sheet

Company	:	Sigma-Aldrich 3050 Spruce Street SAINT LOUIS MO 63103 USA
Telephone Fax	:	+1 800-325-5832 +1 800-325-5052

#### 1.4 Emergency telephone number

Emergency Phone # : +1-703-527-3887 (CHEMTREC)

#### 2. HAZARDS IDENTIFICATION

#### 2.1 Classification of the substance or mixture

#### GHS Classification in accordance with 29 CFR 1910 (OSHA HCS)

Skin sensitisation (Category 1), H317 Germ cell mutagenicity (Category 1B), H340 Carcinogenicity (Category 1B), H350 Reproductive toxicity (Category 1B), H360 Acute aquatic toxicity (Category 1), H400 Chronic aquatic toxicity (Category 1), H410

For the full text of the H-Statements mentioned in this Section, see Section 16.

Danger

#### 2.2 GHS Label elements, including precautionary statements

Pictogram

Signal word



	5	5
Н	lazard statement(s)	
	H317	May cause an allergic skin reaction.
	H340	May cause genetic defects.
	H350	May cause cancer.
	H360	May damage fertility or the unborn child.
	H410	Very toxic to aquatic life with long lasting effects.
	Precautionary statement(s)	
	P201	Obtain special instructions before use.
	P202	Do not handle until all safety precautions have been read and

	understood.
P261	Avoid breathing dust/ fume/ gas/ mist/ vapours/ spray.
P272	Contaminated work clothing should not be allowed out of the workplace.
P273	Avoid release to the environment.
P280	Wear protective gloves/ protective clothing/ eye protection/ face protection.
P302 + P352	IF ON SKIN: Wash with plenty of soap and water.
P308 + P313	IF exposed or concerned: Get medical advice/ attention.
P333 + P313	If skin irritation or rash occurs: Get medical advice/ attention.
P363	Wash contaminated clothing before reuse.
P391	Collect spillage.
P405	Store locked up.
P501	Dispose of contents/ container to an approved waste disposal plant.

#### 2.3 Hazards not otherwise classified (HNOC) or not covered by GHS - none

understand

#### **3. COMPOSITION/INFORMATION ON INGREDIENTS**

#### 3.1 Substances

Synonyms	:	3,4-Benzpyrene 3,4-Benzopyrene Benzo[def]chrysene benzo[pqr]tetraphene
Formula		CooHee

Torriua	•	<sup>0</sup> 20 <sup>11</sup> 2
Molecular weight	:	252.31 g/mol
CAS-No.	:	50-32-8
EC-No.	:	200-028-5
Index-No.	:	601-032-00-3

#### Hazardous components

Component	Classification	Concentration
Benzo[a]pyrene		
	Skin Sens. 1; Muta. 1B; Carc.	90 - 100 %
	1B; Repr. 1B; Aquatic Acute 1;	
	Aquatic Chronic 1; H317,	
	H340, H350, H360, H410	

For the full text of the H-Statements mentioned in this Section, see Section 16.

#### 4. FIRST AID MEASURES

#### 4.1 Description of first aid measures

#### General advice

Consult a physician. Show this safety data sheet to the doctor in attendance. Move out of dangerous area.

#### If inhaled

If breathed in, move person into fresh air. If not breathing, give artificial respiration. Consult a physician.

#### In case of skin contact

Wash off with soap and plenty of water. Consult a physician.

#### In case of eye contact

Flush eyes with water as a precaution.

#### If swallowed

Never give anything by mouth to an unconscious person. Rinse mouth with water. Consult a physician.

#### 4.2 Most important symptoms and effects, both acute and delayed

The most important known symptoms and effects are described in the labelling (see section 2.2) and/or in section 11

#### **4.3 Indication of any immediate medical attention and special treatment needed** No data available

#### **5. FIREFIGHTING MEASURES**

#### 5.1 Extinguishing media

#### Suitable extinguishing media

Use water spray, alcohol-resistant foam, dry chemical or carbon dioxide.

5.2 Special hazards arising from the substance or mixture No data available

#### 5.3 Advice for firefighters

Wear self-contained breathing apparatus for firefighting if necessary.

5.4 Further information

#### 6. ACCIDENTAL RELEASE MEASURES

#### 6.1 Personal precautions, protective equipment and emergency procedures

Use personal protective equipment. Avoid dust formation. Avoid breathing vapours, mist or gas. Ensure adequate ventilation. Evacuate personnel to safe areas. Avoid breathing dust. For personal protection see section 8.

#### 6.2 Environmental precautions

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

#### 6.3 Methods and materials for containment and cleaning up

Pick up and arrange disposal without creating dust. Sweep up and shovel. Keep in suitable, closed containers for disposal.

#### 6.4 Reference to other sections

For disposal see section 13.

#### 7. HANDLING AND STORAGE

#### 7.1 Precautions for safe handling

Avoid contact with skin and eyes. Avoid formation of dust and aerosols. Further processing of solid materials may result in the formation of combustible dusts. The potential for combustible dust formation should be taken into consideration before additional processing occurs. Provide appropriate exhaust ventilation at places where dust is formed. For precautions see section 2.2.

#### 7.2 Conditions for safe storage, including any incompatibilities

Keep container tightly closed in a dry and well-ventilated place.

#### Store at room temperature.

Storage class (TRGS 510): 6.1D: Non-combustible, acute toxic Cat.3 / toxic hazardous materials or hazardous materials causing chronic effects

#### 7.3 Specific end use(s)

Apart from the uses mentioned in section 1.2 no other specific uses are stipulated

#### 8. EXPOSURE CONTROLS/PERSONAL PROTECTION

#### 8.1 Control parameters

#### Components with workplace control parameters

Component	CAS-No.	Value Control Basis				
			parameters			
	Remarks	Cancer				
		Substances for which there is a Biological Exposure Index or Indice (see BEI® section), see BEI® for Polycyclic Aromatic Hydrocarbons (PAHs)				
		Exposure by all routes should be carefully controlled to levels as low				

		as possible.			
		Substances for which there is a Biological Exposure Index or Indices			
		(see BEI® section) see BEI® for Polycyclic Aromatic Hydrocarbons			
		(PAHs)			
		Exposure by	all routes should l	be carefully controlled to levels as low	
		as possible.			
		Suspected h	uman carcinogen	1	
Benzo[a]pyrene	50-32-8	TWA	0.200000 mg/m3	USA. Occupational Exposure Limits (OSHA) - Table Z-1 Limits for Air Contaminants	
		TWA	0.200000 mg/m3	USA. Occupational Exposure Limits (OSHA) - Table Z-1 Limits for Air Contaminants	
		1910.1002	1910.1002		
		As used in §1910.1000 (Table Z-1), coal tar pitch volatiles include the fused polycyclic hydrocarbons which volatilize from the distillation residues of coal, petroleum (excluding asphalt), wood, and other organic matter. Asphalt (CAS 8052-42-4, and CAS 64742-93-4) is not covered under the 'coal tar pitch volatiles' standard OSHA specifically regulated carcinogen			
		TWA	0.100000	USA, NIOSH Recommended	
			mg/m3	Exposure Limits	
		Potential Occupational Carcinogen			
		NIOSH considers coal tar, coal tar pitch, and creosote to be coal tar			
		products.			
		cyclohexane-extractable fraction			
		See Appendix C			
		TWA	0.2 mg/m3	USA, Occupational Exposure Limits	
		(OSHA) - Table Z-1 Limits for Air Contaminants			
		1910.1002			
		As used in §	1910.1000 (Table	Z-1), coal tar pitch volatiles include	
		the fused po	lycyclic hydrocarbo	ons which volatilize from the	
		distillation re	sidues of coal, pet	roleum (excluding asphalt), wood,	
		64742-93-4)	is not covered und	der the 'coal tar pitch volatiles'	
		standard			
		OSHA specif	fically regulated ca	rcinogen	
		TWA	0.1 mg/m3	USA. NIOSH Recommended Exposure Limits	
		Potential Oc	cupational Carcino	gen	
		NIOSH cons	iders coal tar, coal	tar pitch, and creosote to be coal tar	
		products.			
		cyclohexane-extractable fraction			
		See Appendix C			
			1X A	LISA OSHA TARLE 7.1 limita for	
		IVVA	0.2 mg/m3	Air Contaminants - 1910.1000	
		PEL	0.2 mg/m3	California permissible exposure	
				Title 8 Article 107)	
		PFI	$0.2 \text{ mg/m}^3$	California permissible exposure	
			s.z mg/mo	limits for chemical contaminants (Title 8, Article 107)	

#### **Biological occupational exposure limits**

Component CAS-No. Parameters Value Biological Basis						
	Component	CAS-No.	Parameters	Value	Biological	Basis

			specimen	
-	1- Hydroxypyren e		Urine	ACGIH - Biological Exposure Indices (BEI)
Remarks	End of shift at e	end of workv	veek	
	1- Hydroxypyren e		Urine	ACGIH - Biological Exposure Indices (BEI)
	End of shift at e	end of workv	veek	

#### 8.2 Exposure controls

#### Appropriate engineering controls

Handle in accordance with good industrial hygiene and safety practice. Wash hands before breaks and at the end of workday.

#### Personal protective equipment

#### Eye/face protection

Face shield and safety glasses Use equipment for eye protection tested and approved under appropriate government standards such as NIOSH (US) or EN 166(EU).

#### Skin protection

Handle with gloves. Gloves must be inspected prior to use. Use proper glove removal technique (without touching glove's outer surface) to avoid skin contact with this product. Dispose of contaminated gloves after use in accordance with applicable laws and good laboratory practices. Wash and dry hands.

Full contact Material: Nitrile rubber Minimum layer thickness: 0.11 mm Break through time: 480 min Material tested:Dermatril® (KCL 740 / Aldrich Z677272, Size M)

Splash contact Material: Nitrile rubber Minimum layer thickness: 0.11 mm Break through time: 480 min Material tested:Dermatril® (KCL 740 / Aldrich Z677272, Size M)

data source: KCL GmbH, D-36124 Eichenzell, phone +49 (0)6659 87300, e-mail sales@kcl.de, test method: EN374

If used in solution, or mixed with other substances, and under conditions which differ from EN 374, contact the supplier of the CE approved gloves. This recommendation is advisory only and must be evaluated by an industrial hygienist and safety officer familiar with the specific situation of anticipated use by our customers. It should not be construed as offering an approval for any specific use scenario.

#### **Body Protection**

Complete suit protecting against chemicals, The type of protective equipment must be selected according to the concentration and amount of the dangerous substance at the specific workplace.

#### **Respiratory protection**

Where risk assessment shows air-purifying respirators are appropriate use a full-face particle respirator type N100 (US) or type P3 (EN 143) respirator cartridges as a backup to engineering controls. If the respirator is the sole means of protection, use a full-face supplied air respirator. Use respirators and components tested and approved under appropriate government standards such as NIOSH (US) or CEN (EU).

#### Control of environmental exposure

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

#### 9. PHYSICAL AND CHEMICAL PROPERTIES

#### 9.1 Information on basic physical and chemical properties

- a) Appearance Form: solid
- b) Odour No data available

	c)	Odour Threshold	No data available
	d)	рН	No data available
	e)	Melting point/freezing point	Melting point/range: 177 - 180 °C (351 - 356 °F)
	f)	Initial boiling point and boiling range	495 °C (923 °F)
	g)	Flash point	No data available
	h)	Evaporation rate	No data available
	i)	Flammability (solid, gas)	No data available
	j)	Upper/lower flammability or explosive limits	No data available
	k)	Vapour pressure	No data available
	I)	Vapour density	No data available
	m)	Relative density	1.35 g/cm3
	n)	Water solubility	No data available
	o)	Partition coefficient: n- octanol/water	log Pow: 5.97
	p)	Auto-ignition temperature	No data available
	q)	Decomposition temperature	No data available
	r)	Viscosity	No data available
	s)	Explosive properties	No data available
	t)	Oxidizing properties	No data available
0 N	<b>ther</b> o da	<b>safety information</b> Ita available	

#### **10. STABILITY AND REACTIVITY**

**10.1 Reactivity** No data available

9.2

- **10.2 Chemical stability** Stable under recommended storage conditions.
- **10.3 Possibility of hazardous reactions** No data available
- **10.4 Conditions to avoid** No data available
- **10.5** Incompatible materials Strong oxidizing agents
- 10.6 Hazardous decomposition products Hazardous decomposition products formed under fire conditions. - Carbon oxides Other decomposition products - No data available In the event of fire: see section 5

#### **11. TOXICOLOGICAL INFORMATION**

#### 11.1 Information on toxicological effects

Acute toxicity No data available

#### Inhalation: No data available

#### Dermal: No data available

LD50 Subcutaneous - Rat - 50 mg/kg

#### Skin corrosion/irritation

Skin - Mouse Result: Mild skin irritation

Serious eye damage/eye irritation No data available

**Respiratory or skin sensitisation** Chronic exposure may cause dermatitis.

#### Germ cell mutagenicity May alter genetic material. In vivo tests showed mutagenic effects

#### Carcinogenicity

This product is or contains a component that has been reported to be probably carcinogenic based on its IARC, OSHA, ACGIH, NTP, or EPA classification.

Possible human carcinogen

IARC: 1 - Group 1: Carcinogenic to humans (Benzo[a]pyrene)

NTP: RAHC - Reasonably anticipated to be a human carcinogen (Benzo[a]pyrene)

OSHA: OSHA specifically regulated carcinogen (Benzo[a]pyrene)

#### **Reproductive toxicity**

May cause congenital malformation in the fetus. Presumed human reproductive toxicant

May cause reproductive disorders.

#### Specific target organ toxicity - single exposure No data available

Specific target organ toxicity - repeated exposure No data available

### Aspiration hazard

No data available

#### **Additional Information**

RTECS: Not available

burning sensation, Cough, wheezing, laryngitis, Shortness of breath, Headache, Nausea, Vomiting

#### **12. ECOLOGICAL INFORMATION**

#### 12.1 Toxicity

Toxicity to daphnia and EC50 - Daphnia magna (Water flea) - 0.25 mg/l - 48 h other aquatic invertebrates

Toxicity to algae EC50 - Pseudokirchneriella subcapitata (green algae) - 0.02 mg/l - 72 h

#### 12.2 Persistence and degradability

#### 12.3 Bioaccumulative potential

Bioaccumulation

Lepomis macrochirus (Bluegill) - 48 h - 0.0005 mg/l

Bioconcentration factor (BCF): 3,208

#### 12.4 Mobility in soil

No data available

#### 12.5 Results of PBT and vPvB assessment

PBT/vPvB assessment not available as chemical safety assessment not required/not conducted

#### 12.6 Other adverse effects

An environmental hazard cannot be excluded in the event of unprofessional handling or disposal. Very toxic to aquatic life with long lasting effects.

#### **13. DISPOSAL CONSIDERATIONS**

#### 13.1 Waste treatment methods

#### Product

Offer surplus and non-recyclable solutions to a licensed disposal company. Contact a licensed professional waste disposal service to dispose of this material. Dissolve or mix the material with a combustible solvent and burn in a chemical incinerator equipped with an afterburner and scrubber.

#### **Contaminated packaging**

Dispose of as unused product.

#### **14. TRANSPORT INFORMATION**

#### DOT (US)

UN number: 3077 Class: 9 Packing group: III Proper shipping name: Environmentally hazardous substances, solid, n.o.s. (Benzo[a]pyrene) Reportable Quantity (RQ): 1 lbs Poison Inhalation Hazard: No

#### IMDG

UN number: 3077 Class: 9 Packing group: III EMS-No: F-A, S-F Proper shipping name: ENVIRONMENTALLY HAZARDOUS SUBSTANCE, SOLID, N.O.S. (Benzo[a]pyrene) Marine pollutant:yes

#### ΙΑΤΑ

UN number: 3077 Class: 9 Packing group: III Proper shipping name: Environmentally hazardous substance, solid, n.o.s. (Benzo[a]pyrene)

#### **Further information**

EHS-Mark required (ADR 2.2.9.1.10, IMDG code 2.10.3) for single packagings and combination packagings containing inner packagings with Dangerous Goods > 5L for liquids or > 5kg for solids.

#### **15. REGULATORY INFORMATION**

#### SARA 302 Components

No chemicals in this material are subject to the reporting requirements of SARA Title III, Section 302.

#### SARA 313 Components

The following components are subject to reporting levels establ	lished by SARA Title	III, Section 313:
	CAS No	Povision Data

Benzo[a]pyrene	50-32-8	2007-03-01
SARA 311/312 Hazards Acute Health Hazard, Chronic Health Hazard		
Massachusetts Right To Know Components		
	CAS-No.	Revision Date
Benzo[a]pyrene	50-32-8	2007-03-01
Pennsylvania Right To Know Components		
, , ,	CAS-No.	Revision Date
Benzo[a]pyrene	50-32-8	2007-03-01
	CAS-No.	Revision Date
Benzo[a]pyrene	50-32-8	2007-03-01
New Jersey Right To Know Components		
	CAS-No.	<b>Revision Date</b>

Benzo[a]pyrene	50-32-8	2007-03-01
<b>California Prop. 65 Components</b> WARNING! This product contains a chemical known to the State of California to cause cancer. Benzo[a]pyrene	CAS-No. 50-32-8	Revision Date 1990-01-01

#### **16. OTHER INFORMATION**

#### Full text of H-Statements referred to under sections 2 and 3.

Aquatic Acute	Acute aquatic toxicity
Aquatic Chronic	Chronic aquatic toxicity
Carc.	Carcinogenicity
H317	May cause an allergic skin reaction.
H340	May cause genetic defects.
H350	May cause cancer.
H360	May damage fertility or the unborn child.
H400	Very toxic to aquatic life.
H410	Very toxic to aquatic life with long lasting effects.
Muta.	Germ cell mutagenicity

#### **HMIS Rating**

Health hazard:	3
Chronic Health Hazard:	*
Flammability:	0
Physical Hazard	0
-	

#### **NFPA** Rating

Health hazard:	3
Fire Hazard:	0
Reactivity Hazard:	0

#### **Further information**

Copyright 2016 Sigma-Aldrich Co. LLC. License granted to make unlimited paper copies for internal use only. The above information is believed to be correct but does not purport to be all inclusive and shall be used only as a guide. The information in this document is based on the present state of our knowledge and is applicable to the product with regard to appropriate safety precautions. It does not represent any guarantee of the properties of the product. Sigma-Aldrich Corporation and its Affiliates shall not be held liable for any damage resulting from handling or from contact with the above product. See www.sigma-aldrich.com and/or the reverse side of invoice or packing slip for additional terms and conditions of sale.

#### **Preparation Information**

Sigma-Aldrich Corporation Product Safety – Americas Region 1-800-521-8956

Version: 5.8

Revision Date: 02/02/2018

Print Date: 10/19/2018

# SIGMA-ALDRICH

sigma-aldrich.com

# **SAFETY DATA SHEET**

Version 5.11 Revision Date 08/09/2016 Print Date 06/22/2019

#### **1. PRODUCT AND COMPANY IDENTIFICATION**

1.1	Product identifiers Product name	:	Carbon tetrachloride
	Product Number Brand Index-No.	: : :	571016 Aldrich 602-008-00-5
	CAS-No.	:	56-23-5
1.2	Relevant identified uses of	th	e substance or mixture and uses advised against
	Identified uses	:	Laboratory chemicals, Synthesis of substances

#### 1.3 Details of the supplier of the safety data sheet

Company	:	Sigma-Aldrich 3050 Spruce Street SAINT LOUIS MO 63103 USA
Telephone Fax	:	+1 800-325-5832 +1 800-325-5052

#### 1.4 **Emergency telephone number**

Emergency Phone # : +1-703-527-3887 (CHEMTREC)

#### 2. HAZARDS IDENTIFICATION

#### 2.1 Classification of the substance or mixture

#### GHS Classification in accordance with 29 CFR 1910 (OSHA HCS)

Acute toxicity, Oral (Category 3), H301 Acute toxicity, Inhalation (Category 3), H331 Acute toxicity, Dermal (Category 3), H311 Skin sensitisation (Sub-category 1B), H317 Carcinogenicity (Category 2), H351 Specific target organ toxicity - repeated exposure, Inhalation (Category 1), Liver, Kidney, H372 Acute aquatic toxicity (Category 3), H402 Chronic aquatic toxicity (Category 3), H412 Hazardous to the ozone layer (Category 1), H420

For the full text of the H-Statements mentioned in this Section, see Section 16.

Danger

#### 2.2 GHS Label elements, including precautionary statements

Pictogram



Signal word

Hazard statement(s)	
H301 + H311 + H331	Toxic if swallowed, in contact with skin or if inhaled
H317	May cause an allergic skin reaction.
H351	Suspected of causing cancer.
H372	Causes damage to organs (Liver, Kidney) through prolonged or repeated exposure if inhaled.

H412	Harmful to aquatic life with long lasting effects.
Precautionary statement(s)	
P201	Obtain special instructions before use.
P202	Do not handle until all safety precautions have been read and understood.
P260	Do not breathe dust/ fume/ gas/ mist/ vapours/ spray.
P264	Wash skin thoroughly after handling.
P270	Do not eat, drink or smoke when using this product.
P271	Use only outdoors or in a well-ventilated area.
P272	Contaminated work clothing should not be allowed out of the workplace.
P273	Avoid release to the environment.
P280	Wear protective gloves/ protective clothing/ eye protection/ face
	protection.
P301 + P310 + P330	IF SWALLOWED: Immediately call a POISON CENTER/doctor. Rinse mouth.
P302 + P352 + P312	IF ON SKIN: Wash with plenty of soap and water. Call a POISON CENTER or doctor/ physician if you feel unwell.
P304 + P340 + P311	IF INHALED: Remove person to fresh air and keep comfortable for
	breathing. Call a POISON CENTER/doctor.
P308 + P313	IF exposed or concerned: Get medical advice/ attention.
P333 + P313	If skin irritation or rash occurs: Get medical advice/ attention.
P362	Take off contaminated clothing and wash before reuse.
P403 + P233	Store in a well-ventilated place. Keep container tightly closed.
P405	Store locked up.
P501	Dispose of contents/ container to an approved waste disposal plant.
P502	Refer to manufacturer/ supplier for information on recovery/ recycling.

**2.3 Hazards not otherwise classified (HNOC) or not covered by GHS** Rapidly absorbed through skin.

#### **3. COMPOSITION/INFORMATION ON INGREDIENTS**

#### 3.1 Substances

Synonyms	: Tetrachloromethane
Formula	: CCI <sub>4</sub>
Molecular weight	: 153.82 g/mol
CAS-No.	: 56-23-5
EC-No.	: 200-262-8
Index-No.	: 602-008-00-5

#### Hazardous components

Component	Classification	Concentration
Tetrachloromethane		
	Acute Tox. 3; Skin Sens. 1B; Carc. 2; STOT RE 1; Aquatic Acute 3; Aquatic Chronic 3; Ozone 1; H301 + H311 + H331, H317, H351, H372, H412	<= 100 %

For the full text of the H-Statements mentioned in this Section, see Section 16.

#### 4. FIRST AID MEASURES

#### 4.1 Description of first aid measures

#### General advice

Consult a physician. Show this safety data sheet to the doctor in attendance. Move out of dangerous area.

#### If inhaled

If breathed in, move person into fresh air. If not breathing, give artificial respiration. Consult a physician.

#### In case of skin contact

Wash off with soap and plenty of water. Take victim immediately to hospital. Consult a physician.

#### In case of eye contact

Flush eyes with water as a precaution.

#### If swallowed

Never give anything by mouth to an unconscious person. Rinse mouth with water. Consult a physician.

- **4.2** Most important symptoms and effects, both acute and delayed The most important known symptoms and effects are described in the labelling (see section 2.2) and/or in section 11
- **4.3 Indication of any immediate medical attention and special treatment needed** No data available

#### **5. FIREFIGHTING MEASURES**

#### 5.1 Extinguishing media

#### Suitable extinguishing media

Use water spray, alcohol-resistant foam, dry chemical or carbon dioxide.

5.2 Special hazards arising from the substance or mixture No data available

#### **5.3** Advice for firefighters Wear self-contained breathing apparatus for firefighting if necessary.

# 5.4 Further information No data available

#### 6. ACCIDENTAL RELEASE MEASURES

#### 6.1 Personal precautions, protective equipment and emergency procedures

Wear respiratory protection. Avoid breathing vapours, mist or gas. Ensure adequate ventilation. Evacuate personnel to safe areas.

For personal protection see section 8.

#### 6.2 Environmental precautions

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

#### 6.3 Methods and materials for containment and cleaning up

Soak up with inert absorbent material and dispose of as hazardous waste. Keep in suitable, closed containers for disposal.

6.4 Reference to other sections

For disposal see section 13.

#### 7. HANDLING AND STORAGE

#### 7.1 Precautions for safe handling

Avoid contact with skin and eyes. Avoid inhalation of vapour or mist. For precautions see section 2.2.

- **7.2** Conditions for safe storage, including any incompatibilities Keep container tightly closed in a dry and well-ventilated place. Containers which are opened must be carefully resealed and kept upright to prevent leakage.
- 7.3 Specific end use(s)

Apart from the uses mentioned in section 1.2 no other specific uses are stipulated

#### 8. EXPOSURE CONTROLS/PERSONAL PROTECTION

#### 8.1 Control parameters

#### Components with workplace control parameters

Component	CAS-No.	Value	Control parameters	Basis		
Tetrachloromethane	56-23-5	TWA	5.000000 ppm	USA. ACGIH Threshold Limit Values (TLV)		
	Remarks	Liver damag	e			
		Suspected human carcinogen				
		Danger of cu	utaneous absorptio			
		STEL	10.000000 ppm	USA. ACGIH Threshold Limit Values (TLV)		
		Liver damag	е			
		Suspected h	uman carcinogen			
		Danger of cu	Itaneous absorptio			
		51	2.000000 ppm 12.600000 ma/m3	Exposure Limits		
		Potential Oc	cupational Carcino	gen		
		See Append	ix A	5		
		TWA	10.000000 ppm	USA. Occupational Exposure Limits (OSHA) - Table Z-2		
		Z37.17-1967	7			
		CEIL	25.000000 ppm	USA. Occupational Exposure Limits (OSHA) - Table Z-2		
		Z37.17-1967	7			
		Peak	200.000000	USA. Occupational Exposure Limits		
			ppm	(OSHA) - Table Z-2		
		Z37.17-1967	7			
-		See Table Z	-2			
		TWA	5 ppm	USA. ACGIH Threshold Limit Values (TLV)		
-		Liver damag	Liver damage			
		Suspected h	uman carcinogen utaneous absorptio	n		
		STEL	10 ppm	USA. ACGIH Threshold Limit Values (TLV)		
		Liver damag	е			
		Suspected h	uman carcinogen utaneous absorptio	n		
		ST	2 ppm	USA. NIOSH Recommended		
			12.6 mg/m3	Exposure Limits		
		Potential Oc See Append	cupational Carcinc	gen		
		See Table Z	-2			
		TWA	10 ppm	USA. Occupational Exposure Limits (OSHA) - Table Z-2		
		Z37.17-1967	7			
		CEIL	25 ppm	USA. Occupational Exposure Limits		
		737 17-1967	7	(OSHA) - Table Z-2		
		Peak	200 ppm	USA Occupational Exposure Limite		
				(OSHA) - Table Z-2		
		237.17-1967				
		IWA	2 ppm 12.6 mg/m3	USA. USHA - TABLE 2-1 Limits for Air Contaminants - 1910.1000		
		PEL	2 ppm 12.6 mg/m3	California permissible exposure limits for chemical contaminants (Title 8, Article 107)		
		Skin				

C	200 ppm	California permissible exposure limits for chemical contaminants (Title 8, Article 107)
Skin		
STEL	10 ppm 63 mg/m3	California permissible exposure limits for chemical contaminants (Title 8, Article 107)
Skin		

#### 8.2 Exposure controls

#### Appropriate engineering controls

Avoid contact with skin, eyes and clothing. Wash hands before breaks and immediately after handling the product.

#### Personal protective equipment

#### **Eye/face protection**

Face shield and safety glasses Use equipment for eye protection tested and approved under appropriate government standards such as NIOSH (US) or EN 166(EU).

#### **Skin protection**

Handle with gloves. Gloves must be inspected prior to use. Use proper glove removal technique (without touching glove's outer surface) to avoid skin contact with this product. Dispose of contaminated gloves after use in accordance with applicable laws and good laboratory practices. Wash and dry hands.

Full contact Material: Fluorinated rubber Minimum layer thickness: 0.7 mm Break through time: 480 min Material tested:Vitoject® (KCL 890 / Aldrich Z677698, Size M)

Splash contact Material: Nitrile rubber Minimum layer thickness: 0.4 mm Break through time: 240 min Material tested:Camatril® (KCL 730 / Aldrich Z677442, Size M)

data source: KCL GmbH, D-36124 Eichenzell, phone +49 (0)6659 87300, e-mail sales@kcl.de, test method: EN374

If used in solution, or mixed with other substances, and under conditions which differ from EN 374, contact the supplier of the CE approved gloves. This recommendation is advisory only and must be evaluated by an industrial hygienist and safety officer familiar with the specific situation of anticipated use by our customers. It should not be construed as offering an approval for any specific use scenario.

#### **Body Protection**

Complete suit protecting against chemicals, The type of protective equipment must be selected according to the concentration and amount of the dangerous substance at the specific workplace.

#### **Respiratory protection**

Where risk assessment shows air-purifying respirators are appropriate use a full-face respirator with multipurpose combination (US) or type ABEK (EN 14387) respirator cartridges as a backup to engineering controls. If the respirator is the sole means of protection, use a full-face supplied air respirator. Use respirators and components tested and approved under appropriate government standards such as NIOSH (US) or CEN (EU).

#### Control of environmental exposure

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

#### 9. PHYSICAL AND CHEMICAL PROPERTIES

#### 9.1 Information on basic physical and chemical properties

Appearance	Form: liquid Colour: colourless
Odour	sweet
Odour Threshold	No data available
	Appearance Odour Odour Threshold

d)	рН	No data available
e)	Melting point/freezing point	Melting point/range: -22.99 °C (-9.38 °F)
f)	Initial boiling point and boiling range	77 °C (171 °F) at 1,013 hPa (760 mmHg)
g)	Flash point	does not flash
h)	Evaporation rate	No data available
i)	Flammability (solid, gas)	No data available
j)	Upper/lower flammability or explosive limits	No data available
k)	Vapour pressure	45 hPa (34 mmHg) at 0.3 °C (32.5 °F) 120 hPa (90 mmHg) at 19.8 °C (67.6 °F) 14,549 hPa (10,913 mmHg) at 24 °C (75 °F)
I)	Vapour density	No data available
m)	Relative density	1.59 g/cm3 at 20 °C (68 °F)
n)	Water solubility	0.8461 g/l at 20 °C (68 °F)
o)	Partition coefficient: n- octanol/water	log Pow: 2.83 at 25 °C (77 °F)
p)	Auto-ignition temperature	No data available
q)	Decomposition temperature	No data available
r)	Viscosity	No data available
s)	Explosive properties	No data available
t)	Oxidizing properties	No data available
Oth	ner safety information	
	Surface tension	26.7 mN/m at 20 °C (68 °F) 19.5 mN/m at 80 °C (176 °F)

#### **10. STABILITY AND REACTIVITY**

10.1 Reactivity

9.2

No data available

- **10.2 Chemical stability** Stable under recommended storage conditions.
- **10.3 Possibility of hazardous reactions** No data available
- **10.4 Conditions to avoid** No data available
- **10.5** Incompatible materials Strong oxidizing agents

#### **10.6 Hazardous decomposition products**

Hazardous decomposition products formed under fire conditions. - Carbon oxides, Hydrogen chloride gas Other decomposition products - No data available In the event of fire: see section 5

#### 11. TOXICOLOGICAL INFORMATION

#### 11.1 Information on toxicological effects

#### Acute toxicity

LD50 Oral - Rat - 2,350 mg/kg

LC50 Inhalation - Rat - 4 h - 8000 ppm

LD50 Dermal - Rabbit - > 20,000 mg/kg

No data available

#### Skin corrosion/irritation

Skin - Rabbit Result: Mild skin irritation - 24 h (Draize Test)

#### Serious eye damage/eye irritation

Eyes - Rabbit Result: Mild eye irritation - 24 h (Draize Test)

#### Respiratory or skin sensitisation

- Mouse Result: The product is a skin sensitiser, sub-category 1B. (OECD Test Guideline 429)

#### Germ cell mutagenicity

No data available

#### Carcinogenicity

This product is or contains a component that has been reported to be probably carcinogenic based on its IARC, OSHA, ACGIH, NTP, or EPA classification. Limited evidence of carcinogenicity in animal studies

- IARC: 2B Group 2B: Possibly carcinogenic to humans (Tetrachloromethane)
- NTP: Reasonably anticipated to be a human carcinogen (Tetrachloromethane)
- OSHA: No component of this product present at levels greater than or equal to 0.1% is identified as a carcinogen or potential carcinogen by OSHA.

#### **Reproductive toxicity**

No data available

No data available

Specific target organ toxicity - single exposure No data available

Specific target organ toxicity - repeated exposure Inhalation - Causes damage to organs through prolonged or repeated exposure. - Liver, Kidney

Aspiration hazard No data available

#### **Additional Information**

RTECS: FG4900000

Vomiting, Diarrhoea, Abdominal pain, Nausea, Dizziness, Headache, Damage to the eyes., Liver injury may occur., Kidney injury may occur., Exposure to and/or consumption of alcohol may increase toxic effects., Contact with skin can cause:, Pain, Erythema, hyperemia

#### 12. ECOLOGICAL INFORMATION

12.1 Toxicity

	Toxicity to fish	mortality LC50 - Dan	io rerio (zebra fish) - 24.3	3 mg/l - 96 h	
	Toxicity to daphnia and other aquatic invertebrates	Immobilization EC50 (OECD Test Guidelin	) - Daphnia magna (Wate ne 202)	r flea) - 35 mg/l  - 48 h	
	Toxicity to algae	Growth inhibition EC (OECD Test Guidelin	50 - Algae - 20 mg/l  - 72 ne 201)	h .	
12.2	<b>Persistence and degrac</b> No data available	dability			
12.3	Bioaccumulative poten	tial			
	Bioaccumulation	Lepomis macrochiru - 52.3 µg/l	s (Bluegill) - 21 d		
		Bioconcentration fac	tor (BCF): 30		
12.4	<b>Mobility in soil</b> No data available				
12.5	<b>Results of PBT and vPv</b> PBT/vPvB assessment n	<b>/B assessment</b> ot available as chemic	al safety assessment no	t required/not conducted	
12.6	Other adverse effects An environmental hazard Harmful to aquatic life wir	l cannot be excluded ir th long lasting effects.	n the event of unprofession	onal handling or disposal.	
13. DI	SPOSAL CONSIDERATI	ONS			
13.1	Waste treatment metho	ods			
	<b>Product</b> Offer surplus and non-red disposal service to dispose chemical incinerator equi	cyclable solutions to a se of this material. Dis ipped with an afterburr	licensed disposal compa solve or mix the material her and scrubber.	ny. Contact a licensed profession with a combustible solvent and b	nal waste ourn in a
	<b>Contaminated packagin</b> Dispose of as unused pro	<b>ng</b> oduct.			
14. TF	RANSPORT INFORMATIC	ON			
	DOT (US) UN number: 1846 C Proper shipping name: Ca Reportable Quantity (RQ)	Class: 6.1 arbon tetrachloride ): 10 lbs	Packing group: II		
	Poison Inhalation Hazard	: No			
	IMDG UN number: 1846 C Proper shipping name: C	lass: 6.1 ARBON TETRACHLO	Packing group: II RIDE	EMS-No: F-A, S-A	

Marine pollutant: yes ΙΑΤΑ UN number: 1846 Class: 6.1 Packing group: II Proper shipping name: Carbon tetrachloride

#### **15. REGULATORY INFORMATION**

SARA 302 Components	
No chemicals in this material are subject to the reporting requirements of SARA Title III, Sec	ction 302.

#### SARA 313 Components

SARA 313 Components The following components are subject to reporting levels established by SARA Title III, Section 313: CAS-No. Revision Date

	CAS-No.	Revision Da
Tetrachloromethane	56-23-5	2007-07-01

#### SARA 311/312 Hazards

Acute Health Hazard, Chronic Health Hazard

#### Massachusetts Right To Know Components

	CAS-No.	Revision Date
Tetrachloromethane	56-23-5	2007-07-01
Pennsylvania Right To Know Components		
	CAS-No.	Revision Date
Tetrachloromethane	56-23-5	2007-07-01
New Jersey Right To Know Components		
	CAS-No.	Revision Date
Tetrachloromethane	56-23-5	2007-07-01
California Prop. 65 Components		
WARNING! This product contains a chemical known to the	CAS-No.	Revision Date
State of California to cause cancer.	56-23-5	2007-09-28
Tetrachloromethane		

#### **16. OTHER INFORMATION**

#### Full text of H-Statements referred to under sections 2 and 3.

Acute Tox.	Acute toxicity
Aquatic Acute	Acute aquatic toxicity
Aquatic Chronic	Chronic aquatic toxicity
Carc.	Carcinogenicity
H301	Toxic if swallowed.
H301 + H311 +	Toxic if swallowed, in contact with skin or if inhaled
H331	
H311	Toxic in contact with skin.
H317	May cause an allergic skin reaction.
H331	Toxic if inhaled.
H351	Suspected of causing cancer.
HMIS Rating	

NFPA Rating	
Physical Hazard	0
Flammability:	0
Chronic Health Hazard:	*
Health hazard:	2

Health hazard:	2
Fire Hazard:	0
Reactivity Hazard:	0

#### **Further information**

Copyright 2016 Sigma-Aldrich Co. LLC. License granted to make unlimited paper copies for internal use only. The above information is believed to be correct but does not purport to be all inclusive and shall be used only as a guide. The information in this document is based on the present state of our knowledge and is applicable to the product with regard to appropriate safety precautions. It does not represent any guarantee of the properties of the product. Sigma-Aldrich Corporation and its Affiliates shall not be held liable for any damage resulting from handling or from contact with the above product. See www.sigma-aldrich.com and/or the reverse side of invoice or packing slip for additional terms and conditions of sale.

#### **Preparation Information**

Sigma-Aldrich Corporation Product Safety – Americas Region 1-800-521-8956

sigma-aldrich.com

# SAFETY DATA SHEET

Version 6.1 Revision Date 07/16/2018 Print Date 01/21/2019

#### 1. PRODUCT AND COMPANY IDENTIFICATION 1.1 **Product identifiers** Product name Benzo[<l>k</>]fluoranthene Product Number : 48492 Brand Supelco Index-No. 601-036-00-5 CAS-No. ÷ 207-08-9 1.2 Relevant identified uses of the substance or mixture and uses advised against Identified uses : Laboratory chemicals, Synthesis of substances 1.3 Details of the supplier of the safety data sheet Company : Sigma-Aldrich Inc. 3050 Spruce Street ST. LOUIS MO 63103 UNITED STATES Telephone +1 314 771-5765 : Fax +1 800 325-5052

#### 1.4 Emergency telephone number

Emergency Phone # : (314) 776-6555

#### 2. HAZARDS IDENTIFICATION

#### 2.1 Classification of the substance or mixture

#### GHS Classification in accordance with 29 CFR 1910 (OSHA HCS) Carcinogenicity (Category 1B), H350

Acute aquatic toxicity (Category 1), H400

Chronic aquatic toxicity (Category 1), H410

For the full text of the H-Statements mentioned in this Section, see Section 16.

#### 2.2 GHS Label elements, including precautionary statements

Pictogram



Signal word
Hazard statement(s)
H350
H410

Danger

May cause cancer. Very toxic to aquatic life with long lasting effects.

Obtain special instructions before use.
Do not handle until all safety precautions have been read and understood.
Avoid release to the environment.
Use personal protective equipment as required.
IF exposed or concerned: Get medical advice/ attention.
Collect spillage.
Store locked up.
Dispose of contents/ container to an approved waste disposal plant.

2.3 Hazards not otherwise classified (HNOC) or not covered by GHS - none

#### **3. COMPOSITION/INFORMATION ON INGREDIENTS**

#### 3.1 Substances

:	C <sb>20H<sb>12</sb></sb>
:	252.31 g/mol
:	207-08-9
:	205-916-6
:	601-036-00-5
	:

#### Hazardous components

Component	Classification	Concentration
Benzo[k]fluoranthene		
	Carc. 1B; Aquatic Acute 1; Aquatic Chronic 1; H350, H410	<= 100 %

For the full text of the H-Statements mentioned in this Section, see Section 16.

#### 4. FIRST AID MEASURES

#### 4.1 Description of first aid measures

#### **General advice**

Consult a physician. Show this safety data sheet to the doctor in attendance. Move out of dangerous area.

#### If inhaled

If breathed in, move person into fresh air. If not breathing, give artificial respiration. Consult a physician.

#### In case of skin contact

Wash off with soap and plenty of water. Consult a physician.

#### In case of eye contact

Flush eyes with water as a precaution.

#### If swallowed

Never give anything by mouth to an unconscious person. Rinse mouth with water. Consult a physician.

- **4.2** Most important symptoms and effects, both acute and delayed The most important known symptoms and effects are described in the labelling (see section 2.2) and/or in section 11
- **4.3 Indication of any immediate medical attention and special treatment needed** No data available

#### **5. FIREFIGHTING MEASURES**

#### 5.1 Extinguishing media

#### Suitable extinguishing media

Use water spray, alcohol-resistant foam, dry chemical or carbon dioxide.

# 5.2 Special hazards arising from the substance or mixture Carbon oxides

#### **5.3** Advice for firefighters Wear self-contained breathing apparatus for firefighting if necessary.

5.4 Further information No data available

## 6. ACCIDENTAL RELEASE MEASURES

#### 6.1 Personal precautions, protective equipment and emergency procedures

Use personal protective equipment. Avoid dust formation. Avoid breathing vapours, mist or gas. Ensure adequate ventilation. Evacuate personnel to safe areas. Avoid breathing dust. For personal protection see section 8.

#### 6.2 Environmental precautions

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

#### 6.3 Methods and materials for containment and cleaning up Pick up and arrange disposal without creating dust. Sweep up and shovel. Keep in suitable, closed containers for disposal.

#### 6.4 Reference to other sections

For disposal see section 13.

#### 7. HANDLING AND STORAGE

#### 7.1 Precautions for safe handling

Avoid formation of dust and aerosols.Further processing of solid materials may result in the formation of combustible dusts. The potential for combustible dust formation should be taken into consideration before additional processing occurs.

Provide appropriate exhaust ventilation at places where dust is formed. For precautions see section 2.2.

#### 7.2 Conditions for safe storage, including any incompatibilities

Keep container tightly closed in a dry and well-ventilated place.

Recommended storage temperature 2 - 8 °C

Storage class (TRGS 510): 6.1D: Non-combustible, acute toxic Cat.3 / toxic hazardous materials or hazardous materials causing chronic effects

#### 7.3 Specific end use(s)

Apart from the uses mentioned in section 1.2 no other specific uses are stipulated

#### 8. EXPOSURE CONTROLS/PERSONAL PROTECTION

#### 8.1 Control parameters

# Components with workplace control parameters

Biological occupational exposure limits

Component	CAS-No.	Parameters	Value	Biological	Basis
				specimen	
Benzo[k]fluoranthen	207-08-9	1-		Urine	ACGIH - Biological
е		Hydroxypyren			Exposure Indices
		е			(BEI)
	Remarks	End of shift at end of workweek			

#### 8.2 Exposure controls

#### Appropriate engineering controls

Handle in accordance with good industrial hygiene and safety practice. Wash hands before breaks and at the end of workday.

#### Personal protective equipment

#### Eye/face protection

Safety glasses with side-shields conforming to EN166 Use equipment for eye protection tested and approved under appropriate government standards such as NIOSH (US) or EN 166(EU).

#### **Skin protection**

Handle with gloves. Gloves must be inspected prior to use. Use proper glove removal technique (without touching glove's outer surface) to avoid skin contact with this product. Dispose of contaminated gloves after use in accordance with applicable laws and good laboratory practices. Wash and dry hands.

Full contact Material: Nitrile rubber Minimum layer thickness: 0.11 mm Break through time: 480 min Material tested:Dermatril® (KCL 740 / Aldrich Z677272, Size M)

Splash contact Material: Nitrile rubber Minimum layer thickness: 0.11 mm Break through time: 480 min Material tested:Dermatril® (KCL 740 / Aldrich Z677272, Size M)

data source: KCL GmbH, D-36124 Eichenzell, phone +49 (0)6659 87300, e-mail sales@kcl.de, test method: EN374

If used in solution, or mixed with other substances, and under conditions which differ from EN 374, contact the supplier of the CE approved gloves. This recommendation is advisory only and must be evaluated by an industrial hygienist and safety officer familiar with the specific situation of anticipated use by our customers. It should not be construed as offering an approval for any specific use scenario.

#### **Body Protection**

Impervious clothing, The type of protective equipment must be selected according to the concentration and amount of the dangerous substance at the specific workplace.

#### **Respiratory protection**

Where risk assessment shows air-purifying respirators are appropriate use a full-face particle respirator type N100 (US) or type P3 (EN 143) respirator cartridges as a backup to engineering controls. If the respirator is the sole means of protection, use a full-face supplied air respirator. Use respirators and components tested and approved under appropriate government standards such as NIOSH (US) or CEN (EU).

#### Control of environmental exposure

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

#### 9. PHYSICAL AND CHEMICAL PROPERTIES

#### 9.1 Information on basic physical and chemical properties

a)	Appearance	Form: crystalline Colour: yellow
b)	Odour	No data available
c)	Odour Threshold	No data available
d)	рН	No data available
e)	Melting point/freezing point	Melting point/range: 215 - 217 °C (419 - 423 °F) - lit.
f)	Initial boiling point and boiling range	No data available
g)	Flash point	No data available
h)	Evaporation rate	No data available
i)	Flammability (solid, gas)	No data available
j)	Upper/lower flammability or explosive limits	No data available
k)	Vapour pressure	No data available
I)	Vapour density	No data available

m)	Relative density	No data available	
n)	Water solubility	No data available	
o)	Partition coefficient: n- octanol/water	No data available	
p)	Auto-ignition temperature	No data available	
q)	Decomposition temperature	No data available	
r)	Viscosity	No data available	
s)	Explosive properties	No data available	
t)	Oxidizing properties	No data available	
Oth No	Other safety information No data available		

# 10. STABILITY AND REACTIVITY

#### **10.1 Reactivity** No data available

9.2

#### **10.2 Chemical stability** Stable under recommended storage conditions.

- **10.3 Possibility of hazardous reactions** No data available
- **10.4 Conditions to avoid** No data available
- **10.5 Incompatible materials** Strong oxidizing agents

# Hazardous decomposition products Hazardous decomposition products formed under fire conditions. - Carbon oxides Other decomposition products - No data available In the event of fire: see section 5

#### **11. TOXICOLOGICAL INFORMATION**

#### 11.1 Information on toxicological effects

#### Acute toxicity

No data available Inhalation: No data available Dermal: No data available No data available

## Skin corrosion/irritation

No data available

Serious eye damage/eye irritation No data available

**Respiratory or skin sensitisation** No data available

Germ cell mutagenicity No data available

Carcinogenicity Carcinogenicity- Rat- Implant This product is or contains a component that has been reported to be probably carcinogenic based on its IARC, OSHA, ACGIH, NTP, or EPA classification.

Possible human carcinogen

- IARC: 2B Group 2B: Possibly carcinogenic to humans (Benzo[k]fluoranthene)
- NTP: RAHC Reasonably anticipated to be a human carcinogen (Benzo[k]fluoranthene)
- OSHA: No component of this product present at levels greater than or equal to 0.1% is on OSHA's list of regulated carcinogens.

#### Reproductive toxicity No data available

No data available

Specific target organ toxicity - single exposure No data available

#### Specific target organ toxicity - repeated exposure No data available

#### Aspiration hazard No data available

#### **Additional Information**

RTECS: DF6350000

To the best of our knowledge, the chemical, physical, and toxicological properties have not been thoroughly investigated.

#### **12. ECOLOGICAL INFORMATION**

12.1 Toxicity

No data available

- 12.2 Persistence and degradability No data available
- **12.3 Bioaccumulative potential** No data available
- **12.4 Mobility in soil** No data available(Benzo[k]fluoranthene)

#### **12.5** Results of PBT and vPvB assessment PBT/vPvB assessment not available as chemical safety assessment not required/not conducted

#### 12.6 Other adverse effects

An environmental hazard cannot be excluded in the event of unprofessional handling or disposal. Very toxic to aquatic life with long lasting effects.

#### **13. DISPOSAL CONSIDERATIONS**

#### 13.1 Waste treatment methods

#### Product

Offer surplus and non-recyclable solutions to a licensed disposal company. Contact a licensed professional waste disposal service to dispose of this material. Dissolve or mix the material with a combustible solvent and burn in a chemical incinerator equipped with an afterburner and scrubber.

#### Contaminated packaging

Dispose of as unused product.

#### 14. TRANSPORT INFORMATION

#### DOT (US)

UN number: 3077 Class: 9 Packing group: III Proper shipping name: Environmentally hazardous substance, solid, n.o.s. (Benzo[k]fluoranthene) Supelco- 48492 no

Poison Inhalation Hazard: No

#### IMDG

UN number: 3077 Class: 9 Packing group: III EMS-No: F-A, S-F Proper shipping name: ENVIRONMENTALLY HAZARDOUS SUBSTANCE, SOLID, N.O.S. (Benzo[k]fluoranthene) Marine pollutant : yes

#### ΙΑΤΑ

UN number: 3077 Class: 9 Packing group: III Proper shipping name: Environmentally hazardous substance, solid, n.o.s. (Benzo[k]fluoranthene)

#### **Further information**

EHS-Mark required (ADR 2.2.9.1.10, IMDG code 2.10.3) for single packagings and combination packagings containing inner packagings with Dangerous Goods > 5L for liquids or > 5kg for solids.

#### **15. REGULATORY INFORMATION**

SARA 302 Components

No chemicals in this material are subject to the reporting requirements of SARA Title III, Section 302.

#### SARA 313 Components

This material does not contain any chemical components with known CAS numbers that exceed the threshold (De Minimis) reporting levels established by SARA Title III, Section 313.

#### **Massachusetts Right To Know Components**

	CAS-No.	Revision Date
Benzo[k]fluoranthene	207-08-9	1994-04-01
Pennsylvania Right To Know Components		
	CAS-No.	Revision Date
Benzo[k]fluoranthene	207-08-9	1994-04-01
California Prop. 65 Components		
, which is/are known to the State of California to cause cancer.	CAS-No.	Revision Date
For more information go to www.P65Warnings.ca.gov.	207-08-9	2007-09-28
Benzo[k]fluoranthene		

#### **16. OTHER INFORMATION**

#### Full text of H-Statements referred to under sections 2 and 3.

H350	May cause cancer.
H400	Very toxic to aquatic life.
H410	Very toxic to aquatic life with long lasting effects.

#### Further information

Copyright 2016 Sigma-Aldrich Co. LLC. License granted to make unlimited paper copies for internal use only. The above information is believed to be correct but does not purport to be all inclusive and shall be used only as a guide. The information in this document is based on the present state of our knowledge and is applicable to the product with regard to appropriate safety precautions. It does not represent any guarantee of the properties of the product. Sigma-Aldrich Corporation and its Affiliates shall not be held liable for any damage resulting from handling or from contact with the above product. See www.sigma-aldrich.com and/or the reverse side of invoice or packing slip for additional terms and conditions of sale.

#### **Preparation Information**

Sigma-Aldrich Corporation Product Safety – Americas Region 1-800-521-8956 Version: 6.1

Revision Date: 07/16/2018

Print Date: 01/21/2019

sigma-aldrich.com

# SAFETY DATA SHEET

Version 4.9 Revision Date 06/02/2016 Print Date 06/28/2019

#### **1. PRODUCT AND COMPANY IDENTIFICATION**

1.1	<b>Product identifiers</b> Product name :		Chloromethane
	Product Number Brand Index-No.	:	295507 Aldrich 602-001-00-7
	CAS-No.	:	74-87-3
1.2	Relevant identified use	s of th	e substance or mixture and uses advised against

Identified uses : Laboratory chemicals, Synthesis of substances

#### 1.3 Details of the supplier of the safety data sheet

Company	:	Sigma-Aldrich 3050 Spruce Street SAINT LOUIS MO 63103 USA
Telephone Fax	:	+1 800-325-5832 +1 800-325-5052

#### 1.4 Emergency telephone number

Emergency Phone # : +1-703-527-3887 (CHEMTREC)

#### 2. HAZARDS IDENTIFICATION

#### 2.1 Classification of the substance or mixture

#### GHS Classification in accordance with 29 CFR 1910 (OSHA HCS)

Flammable gases (Category 1), H220 Gases under pressure (Liquefied gas), H280 Carcinogenicity (Category 2), H351 Reproductive toxicity (Category 2), H361 Specific target organ toxicity - repeated exposure, Inhalation (Category 2), Eyes, Nervous system, Testes, H373

For the full text of the H-Statements mentioned in this Section, see Section 16.

#### 2.2 GHS Label elements, including precautionary statements

Pictogram

Signal word



Danger

•	-
Hazard statement(s)	
H220	Extremely flammable gas.
H280	Contains gas under pressure; may explode if heated.
H351	Suspected of causing cancer.
H361	Suspected of damaging fertility or the unborn child.
H373	May cause damage to organs (Eyes, Nervous system, Testes) through prolonged or repeated exposure if inhaled.

Precautionary statement(s) P201

Obtain special instructions before use.

P202	Do not handle until all safety precautions have been read and understood.
P210	Keep away from heat/sparks/open flames/hot surfaces. No smoking.
P260	Do not breathe dust/ fume/ gas/ mist/ vapours/ spray.
P281	Use personal protective equipment as required.
P308 + P313	IF exposed or concerned: Get medical advice/ attention.
P377	Leaking gas fire: Do not extinguish, unless leak can be stopped safely.
P381	Eliminate all ignition sources if safe to do so.
P405	Store locked up.
P410 + P403	Protect from sunlight. Store in a well-ventilated place.
P501	Dispose of contents/ container to an approved waste disposal plant.

#### 2.3 Hazards not otherwise classified (HNOC) or not covered by GHS - none

#### **3. COMPOSITION/INFORMATION ON INGREDIENTS**

#### 3.1 Substances

Synonyms	:	Methyl chloride
Formula	:	CH <sub>3</sub> Cl
Molecular weight	:	50.49 g/mol
CAS-No.	:	74-87-3
EC-No.	:	200-817-4
Index-No.	:	602-001-00-7

#### Hazardous components

Component	Classification	Concentration
Chloromethane		
	Flam. Gas 1; Press. Gas Liquefied gas; Carc. 2; Repr. 2; STOT RE 2; H220, H280, H351, H361, H373	<= 100 %

For the full text of the H-Statements mentioned in this Section, see Section 16.

#### 4. FIRST AID MEASURES

#### 4.1 Description of first aid measures

#### General advice

Consult a physician. Show this safety data sheet to the doctor in attendance. Move out of dangerous area.

#### If inhaled

If breathed in, move person into fresh air. If not breathing, give artificial respiration. Consult a physician.

#### In case of skin contact

Wash off with soap and plenty of water. Consult a physician.

#### In case of eye contact

Flush eyes with water as a precaution.

#### If swallowed

Do NOT induce vomiting. Never give anything by mouth to an unconscious person. Rinse mouth with water. Consult a physician.

#### 4.2 Most important symptoms and effects, both acute and delayed

The most important known symptoms and effects are described in the labelling (see section 2.2) and/or in section 11

## 4.3 Indication of any immediate medical attention and special treatment needed
# **5. FIREFIGHTING MEASURES**

#### 5.1 Extinguishing media

#### Suitable extinguishing media

Use water spray, alcohol-resistant foam, dry chemical or carbon dioxide.

5.2 Special hazards arising from the substance or mixture No data available

#### **5.3** Advice for firefighters Wear self-contained breathing apparatus for firefighting if necessary.

#### 5.4 Further information Use water spray to cool unopened containers.

#### 6. ACCIDENTAL RELEASE MEASURES

#### 6.1 Personal precautions, protective equipment and emergency procedures

Use personal protective equipment. Avoid breathing vapours, mist or gas. Ensure adequate ventilation. Remove all sources of ignition. Evacuate personnel to safe areas. Beware of vapours accumulating to form explosive concentrations. Vapours can accumulate in low areas. For personal protection see section 8.

#### **6.2** Environmental precautions Prevent further leakage or spillage if safe to do so. Do not let product enter drains.

- **6.3** Methods and materials for containment and cleaning up Clean up promptly by sweeping or vacuum.
- 6.4 Reference to other sections

For disposal see section 13.

# 7. HANDLING AND STORAGE

#### 7.1 Precautions for safe handling

Avoid contact with skin and eyes. Avoid inhalation of vapour or mist. Use explosion-proof equipment.Keep away from sources of ignition - No smoking.Take measures to prevent the build up of electrostatic charge. For precautions see section 2.2.

#### 7.2 Conditions for safe storage, including any incompatibilities

Keep container tightly closed in a dry and well-ventilated place.

Contents under pressure. Moisture sensitive. Storage class (TRGS 510): Gases

#### 7.3 Specific end use(s)

Apart from the uses mentioned in section 1.2 no other specific uses are stipulated

#### 8. EXPOSURE CONTROLS/PERSONAL PROTECTION

#### 8.1 Control parameters

#### Components with workplace control parameters

Component	CAS-No.	Value	Control	Basis	
			parameters		
Chloromethane	74-87-3	TWA	50.000000 ppm	USA. ACGIH Threshold Limit Values	
				(TLV)	
	Remarks	Central Nervous System impairment			
		Liver damage			
		Kidney damage			
		Testicular damage			
		Teratogenic effects			
		Not classifiable as a human carcinogen			
		Danger of cutaneous absorption			

STEL	100.000000 ppm	USA. ACGIH Threshold Limit Values (TLV)	
Central Nerv Liver damag Kidney dama Testicular da Teratogenic Not classifia Danger of cu Potential Oc	vous System impai le age amage effects ble as a human ca <u>utaneous absorptic</u> cupational Carcing	rment ircinogen on	
See Append			
TWA	100.000000 ppm	USA. Occupational Exposure Limits (OSHA) - Table Z-2	
Z37,18-1969	)		
CEIL	200.000000 ppm	USA. Occupational Exposure Limits (OSHA) - Table Z-2	
Z37.18-1969	.18-1969		
Peak	300.000000 ppm	USA. Occupational Exposure Limits (OSHA) - Table Z-2	
737 18-1969			
See Table Z-2			
TWA	100 ppm	USA. Occupational Exposure Limits (OSHA) - Table Z-2	
Z37.18-1969	)		
CEIL	200 ppm	USA. Occupational Exposure Limits (OSHA) - Table Z-2	
Z37.18-1969	37.18-1969		
Peak	300 ppm	USA. Occupational Exposure Limits (OSHA) - Table Z-2	
Z37.18-1969	)		
STEL	100 ppm 210 mg/m3	California permissible exposure limits for chemical contaminants (Title 8, Article 107)	
C	300 ppm	California permissible exposure limits for chemical contaminants (Title 8, Article 107)	
PEL	50 ppm 105 mg/m3	California permissible exposure limits for chemical contaminants (Title 8, Article 107)	

#### 8.2 Exposure controls

#### Appropriate engineering controls

Handle in accordance with good industrial hygiene and safety practice. Wash hands before breaks and at the end of workday.

# Personal protective equipment

#### Eye/face protection

Face shield and safety glasses Use equipment for eye protection tested and approved under appropriate government standards such as NIOSH (US) or EN 166(EU).

#### **Skin protection**

Handle with gloves. Gloves must be inspected prior to use. Use proper glove removal technique (without touching glove's outer surface) to avoid skin contact with this product. Dispose of contaminated gloves after use in accordance with applicable laws and good laboratory practices. Wash and dry hands.

Full contact

Material: Fluorinated rubber Minimum layer thickness: 0.7 mm Break through time: 480 min Material tested:Vitoject® (KCL 890 / Aldrich Z677698, Size M)

Splash contact Material: Fluorinated rubber Minimum layer thickness: 0.7 mm Break through time: 480 min Material tested:Vitoject® (KCL 890 / Aldrich Z677698, Size M)

data source: KCL GmbH, D-36124 Eichenzell, phone +49 (0)6659 87300, e-mail sales@kcl.de, test method: EN374

If used in solution, or mixed with other substances, and under conditions which differ from EN 374, contact the supplier of the CE approved gloves. This recommendation is advisory only and must be evaluated by an industrial hygienist and safety officer familiar with the specific situation of anticipated use by our customers. It should not be construed as offering an approval for any specific use scenario.

#### **Body Protection**

Complete suit protecting against chemicals, Flame retardant antistatic protective clothing., The type of protective equipment must be selected according to the concentration and amount of the dangerous substance at the specific workplace.

#### **Respiratory protection**

Where risk assessment shows air-purifying respirators are appropriate use a full-face respirator with multipurpose combination (US) or type AXBEK (EN 14387) respirator cartridges as a backup to engineering controls. If the respirator is the sole means of protection, use a full-face supplied air respirator. Use respirators and components tested and approved under appropriate government standards such as NIOSH (US) or CEN (EU).

#### Control of environmental exposure

Prevent further leakage or spillage if safe to do so. Do not let product enter drains.

#### 9. PHYSICAL AND CHEMICAL PROPERTIES

#### 9.1 Information on basic physical and chemical properties

a)	Appearance	Form: compressed liquefied gas
b)	Odour	No data available
c)	Odour Threshold	No data available
d)	рН	No data available
e)	Melting point/freezing point	Melting point/range: -97 °C (-143 °F) - lit.
f)	Initial boiling point and boiling range	-24.2 °C (-11.6 °F) - lit.
g)	Flash point	No data available
h)	Evaporation rate	No data available
i)	Flammability (solid, gas)	No data available
j)	Upper/lower flammability or explosive limits	Upper explosion limit: 17.4 %(V) Lower explosion limit: 7 %(V)
k)	Vapour pressure	5,060.9 hPa (3,796.0 mmHg) at 20.0 °C (68.0 °F)
I)	Vapour density	No data available
m)	Relative density	0.915 g/cm3 at 25 °C (77 °F)
n)	Water solubility	5.32 g/l at 25 °C (77 °F) - soluble
o)	Partition coefficient: n- octanol/water	log Pow: 0.91

	p)	Auto-ignition temperature	632.0 °C (1,169.6 °F)
	q)	Decomposition temperature	No data available
	r)	Viscosity	No data available
	s)	Explosive properties	No data available
	t)	Oxidizing properties	No data available
9.2	Other safety information		
		Surface tension	0.02 mN/m at 20 °C (68 °F)
10. S	ТАВ	ILITY AND REACTIVITY	
10.1	<b>Reactivity</b> No data available		
10.2	Ch	emical stability	

**0.2 Chemical stability** Stable under recommended storage conditions.

- **10.3 Possibility of hazardous reactions** No data available
- **10.4 Conditions to avoid** Heat, flames and sparks.
- **10.5** Incompatible materials Strong oxidizing agents, Iron

#### 10.6 Hazardous decomposition products Hazardous decomposition products formed under fire conditions. - Carbon oxides, Hydrogen chloride gas Other decomposition products - No data available In the event of fire: see section 5

# **11. TOXICOLOGICAL INFORMATION**

#### 11.1 Information on toxicological effects

# Acute toxicity

LD50 Oral - Rat - 1,800 mg/kg

LC50 Inhalation - Rat - male and female - 4 h - > 21,800 mg/m3 (OECD Test Guideline 403)

Dermal: No data available

No data available

Skin corrosion/irritation No data available

Serious eye damage/eye irritation No data available

**Respiratory or skin sensitisation** No data available

#### Germ cell mutagenicity

Rat - male Result: negative DNA damage DNA repair

#### Carcinogenicity

IARC: No component of this product present at levels greater than or equal to 0.1% is identified as probable, possible or confirmed human carcinogen by IARC.

- NTP: No component of this product present at levels greater than or equal to 0.1% is identified as a known or anticipated carcinogen by NTP.
- OSHA: No component of this product present at levels greater than or equal to 0.1% is identified as a carcinogen or potential carcinogen by OSHA.

#### **Reproductive toxicity**

No data available

No data available

Specific target organ toxicity - single exposure No data available

#### Specific target organ toxicity - repeated exposure

Inhalation - May cause damage to organs through prolonged or repeated exposure. - Central nervous system, Liver, Urogenital tract

#### Aspiration hazard

No data available

#### **Additional Information**

RTECS: PA6300000

Dizziness, Drowsiness, Incoordination., Blurred vision, Headache, Nausea, Vomiting

Stomach - Irregularities - Based on Human Evidence Stomach - Irregularities - Based on Human Evidence

#### **12. ECOLOGICAL INFORMATION**

#### 12.1 Toxicity

Toxicity to fish	LC50 - Lepomis macrochirus (Bluegill) - 550 mg/l - 96 h
Toxicity to daphnia and other aquatic	semi-static test EC50 - Daphnia magna (Water flea) - 200 mg/l  - 48 h (OECD Test Guideline 202)
invertebrates	

#### 12.2 Persistence and degradability

Biodegradability aerobic - Exposure time 28 h Result: 100 % - Readily biodegradable

# 12.3 Bioaccumulative potential

No data available

# 12.4 Mobility in soil

No data available

#### 12.5 Results of PBT and vPvB assessment

PBT/vPvB assessment not available as chemical safety assessment not required/not conducted

#### 12.6 Other adverse effects

No data available

# **13. DISPOSAL CONSIDERATIONS**

#### 13.1 Waste treatment methods

#### Product

Burn in a chemical incinerator equipped with an afterburner and scrubber but exert extra care in igniting as this material is highly flammable. Offer surplus and non-recyclable solutions to a licensed disposal company. Contact a licensed professional waste disposal service to dispose of this material.

#### Contaminated packaging

Dispose of as unused product.

#### **14. TRANSPORT INFORMATION**

DOT (US)

UN number: 1063 Class: 2.1 Proper shipping name: Methyl chloride Reportable Quantity (RQ): 100 lbs

Poison Inhalation Hazard: No

#### IMDG

UN number: 1063 Class: 2.1 Proper shipping name: METHYL CHLORIDE

#### ΙΑΤΑ

UN number: 1063 Class: 2.1 Proper shipping name: Methyl chloride IATA Passenger: Not permitted for transport

# **15. REGULATORY INFORMATION**

# SARA 302 Components

No chemicals in this material are subject to the reporting requirements of SARA Title III, Section 302.

#### SARA 313 Components

The following components are subject to reporting levels established	shed by SARA Title II	I, Section 313:
	CAS-No.	Revision Date
Chloromethane	74-87-3	2007-07-01
SARA 311/312 Hazards Fire Hazard, Sudden Release of Pressure Hazard, Chronic Hea	lth Hazard	
Massachusetts Right To Know Components		
	CAS-No.	<b>Revision Date</b>
Chloromethane	74-87-3	2007-07-01
Pennsylvania Right To Know Components		
	CAS-No.	Revision Date
Chloromethane	74-87-3	2007-07-01
New Jersey Right To Know Components		
	CAS-No.	Revision Date
Chloromethane	74-87-3	2007-07-01
California Prop. 65 Components		
WARNING: This product contains a chemical known to the	CAS-No.	Revision Date
State of California to cause birth defects or other reproductive	74-87-3	2009-09-11
harm.		
Chloromethane		

# **16. OTHER INFORMATION**

# Full text of H-Statements referred to under sections 2 and 3.

Carc.	Carcinogenicity
Flam. Gas	Flammable gases
H220	Extremely flammable gas.
H280	Contains gas under pressure; may explode if heated.
H351	Suspected of causing cancer.
H361	Suspected of damaging fertility or the unborn child.
H373	May cause damage to organs through prolonged or repeated exposure if inhaled.
Press. Gas	Gases under pressure
Repr.	Reproductive toxicity
STOT RE	Specific target organ toxicity - repeated exposure

# **HMIS Rating**

Health hazard:	0
Chronic Health Hazard:	*
Flammability:	4
Physical Hazard	3
NFPA Rating	
Health hazard:	1
Fire Hazard:	4

Reactivity Hazard:

# **Further information**

Copyright 2016 Sigma-Aldrich Co. LLC. License granted to make unlimited paper copies for internal use only. The above information is believed to be correct but does not purport to be all inclusive and shall be used only as a guide. The information in this document is based on the present state of our knowledge and is applicable to the product with regard to appropriate safety precautions. It does not represent any guarantee of the properties of the product. Sigma-Aldrich Corporation and its Affiliates shall not be held liable for any damage resulting from handling or from contact with the above product. See www.sigma-aldrich.com and/or the reverse side of invoice or packing slip for additional terms and conditions of sale.

#### **Preparation Information**

Sigma-Aldrich Corporation Product Safety - Americas Region 1-800-521-8956

Version: 4.9

Revision Date: 06/02/2016

Print Date: 06/28/2019

# SIGMA-ALDRICH

sigma-aldrich.com

# **SAFETY DATA SHEET**

Version 4.4 Revision Date 12/01/2015 Print Date 01/05/2020

# **1. PRODUCT AND COMPANY IDENTIFICATION**

1.1	Product identifiers Product name	:	cis-Dichloroethylene	
	Product Number Brand Index-No.	:	48597 Supelco 602-026-00-3	
	CAS-No.	:	156-59-2	
1.2	Relevant identified uses of the substance or mixture and uses advised against			
	Identified uses	:	Laboratory chemicals, Synthesis of substances	

#### 1.3 Details of the supplier of the safety data sheet

Company	:	Sigma-Aldrich 3050 Spruce Street SAINT LOUIS MO 63103 USA
Telephone Fax	:	+1 800-325-5832 +1 800-325-5052

#### 1.4 Emergency telephone number

Emergency Phone # : +1-703-527-3887 (CHEMTREC)

## 2. HAZARDS IDENTIFICATION

#### 2.1 Classification of the substance or mixture

#### GHS Classification in accordance with 29 CFR 1910 (OSHA HCS)

Flammable liquids (Category 2), H225 Acute toxicity, Inhalation (Category 4), H332 Acute aquatic toxicity (Category 3), H402 Chronic aquatic toxicity (Category 3), H412

For the full text of the H-Statements mentioned in this Section, see Section 16.

# 2.2 GHS Label elements, including precautionary statements

Pictogram



Danger

Hazard statement(s)	
H225	Highly flammable liquid and vapour.
H332	Harmful if inhaled.
H412	Harmful to aquatic life with long lasting effects.
Precautionary statement(s)	
P210	Keep away from heat/sparks/open flames/hot surfaces. No smoking.
P233	Keep container tightly closed.
P240	Ground/bond container and receiving equipment.
P241	Use explosion-proof electrical/ ventilating/ lighting/ equipment.
P242	Use only non-sparking tools.

P243	Take precautionary measures against static discharge.
P261	Avoid breathing dust/ fume/ gas/ mist/ vapours/ spray.
P271	Use only outdoors or in a well-ventilated area.
P273	Avoid release to the environment.
P280	Wear protective gloves/ protective clothing/ eye protection/ face protection.
P303 + P361 + P353	IF ON SKIN (or hair): Remove/ Take off immediately all contaminated clothing. Rinse skin with water/ shower.
P304 + P340	IF INHALED: Remove victim to fresh air and keep at rest in a position comfortable for breathing.
P312	Call a POISON CENTER or doctor/ physician if you feel unwell.
P370 + P378	In case of fire: Use dry sand, dry chemical or alcohol-resistant foam for extinction.
P403 + P235	Store in a well-ventilated place. Keep cool.
P501	Dispose of contents/ container to an approved waste disposal plant.

#### 2.3 Hazards not otherwise classified (HNOC) or not covered by GHS - none

# **3. COMPOSITION/INFORMATION ON INGREDIENTS**

#### 3.1 Substances

Formula	:	C2H2Cl2
Molecular weight	:	96.94 g/mol
CAS-No.	:	156-59-2
EC-No.	:	205-859-7
Index-No.	:	602-026-00-3

# Hazardous components

Component	Classification	Concentration
cis-Dichloroethylene		
	Flam. Liq. 2; Acute Tox. 4; Aquatic Acute 3; Aquatic Chronic 3; H225, H332, H412	<= 100 %

For the full text of the H-Statements mentioned in this Section, see Section 16.

#### 4. FIRST AID MEASURES

#### 4.1 Description of first aid measures

#### **General advice**

Consult a physician. Show this safety data sheet to the doctor in attendance. Move out of dangerous area.

#### If inhaled

If breathed in, move person into fresh air. If not breathing, give artificial respiration. Consult a physician.

#### In case of skin contact

Wash off with soap and plenty of water. Consult a physician.

#### In case of eye contact

Flush eyes with water as a precaution.

#### If swallowed

Do NOT induce vomiting. Never give anything by mouth to an unconscious person. Rinse mouth with water. Consult a physician.

### 4.2 Most important symptoms and effects, both acute and delayed

The most important known symptoms and effects are described in the labelling (see section 2.2) and/or in section 11

#### **4.3 Indication of any immediate medical attention and special treatment needed** No data available

#### **5. FIREFIGHTING MEASURES**

#### 5.1 Extinguishing media

#### Suitable extinguishing media

Use water spray, alcohol-resistant foam, dry chemical or carbon dioxide.

5.2 Special hazards arising from the substance or mixture Carbon oxides, Hydrogen chloride gas

#### **5.3** Advice for firefighters Wear self-contained breathing apparatus for firefighting if necessary.

#### **5.4 Further information** Use water spray to cool unopened containers.

#### 6. ACCIDENTAL RELEASE MEASURES

#### 6.1 Personal precautions, protective equipment and emergency procedures

Use personal protective equipment. Avoid breathing vapours, mist or gas. Ensure adequate ventilation. Remove all sources of ignition. Evacuate personnel to safe areas. Beware of vapours accumulating to form explosive concentrations. Vapours can accumulate in low areas. For personal protection see section 8.

#### 6.2 Environmental precautions

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

#### 6.3 Methods and materials for containment and cleaning up

Contain spillage, and then collect with an electrically protected vacuum cleaner or by wet-brushing and place in container for disposal according to local regulations (see section 13).

#### 6.4 Reference to other sections

For disposal see section 13.

## 7. HANDLING AND STORAGE

#### 7.1 Precautions for safe handling

Avoid contact with skin and eyes. Avoid inhalation of vapour or mist. Use explosion-proof equipment.Keep away from sources of ignition - No smoking.Take measures to prevent the build up of electrostatic charge. For precautions see section 2.2.

T of precaditors see section 2.2.

# 7.2 Conditions for safe storage, including any incompatibilities

Keep container tightly closed in a dry and well-ventilated place. Containers which are opened must be carefully resealed and kept upright to prevent leakage.

Recommended storage temperature 2 - 8 °C

Handle and store under inert gas. Air and moisture sensitive. Light sensitive.

#### 7.3 Specific end use(s)

Apart from the uses mentioned in section 1.2 no other specific uses are stipulated

#### 8. EXPOSURE CONTROLS/PERSONAL PROTECTION

#### 8.1 Control parameters

#### Components with workplace control parameters

Component	CAS-No.	Value	Control parameters	Basis
cis-Dichloroethylene	156-59-2	TWA	200 ppm	USA. ACGIH Threshold Limit Values (TLV)
	Remarks	Central Nervous System impairment Eye irritation		

# 8.2 Exposure controls

#### Appropriate engineering controls

Handle in accordance with good industrial hygiene and safety practice. Wash hands before breaks and at the end of workday.

#### Personal protective equipment

#### Eye/face protection

Face shield and safety glasses Use equipment for eye protection tested and approved under appropriate government standards such as NIOSH (US) or EN 166(EU).

#### **Skin protection**

Handle with gloves. Gloves must be inspected prior to use. Use proper glove removal technique (without touching glove's outer surface) to avoid skin contact with this product. Dispose of contaminated gloves after use in accordance with applicable laws and good laboratory practices. Wash and dry hands.

#### **Body Protection**

Complete suit protecting against chemicals, Flame retardant antistatic protective clothing., The type of protective equipment must be selected according to the concentration and amount of the dangerous substance at the specific workplace.

#### **Respiratory protection**

Where risk assessment shows air-purifying respirators are appropriate use a full-face respirator with multipurpose combination (US) or type AXBEK (EN 14387) respirator cartridges as a backup to engineering controls. If the respirator is the sole means of protection, use a full-face supplied air respirator. Use respirators and components tested and approved under appropriate government standards such as NIOSH (US) or CEN (EU).

#### **Control of environmental exposure**

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

# 9. PHYSICAL AND CHEMICAL PROPERTIES

#### 9.1 Information on basic physical and chemical properties

a)	Appearance	Form: liquid Colour: light yellow
b)	Odour	No data available
c)	Odour Threshold	No data available
d)	рН	No data available
e)	Melting point/freezing point	-80.0 °C (-112.0 °F)
f)	Initial boiling point and boiling range	60.0 - 61.0 °C (140.0 - 141.8 °F)
g)	Flash point	6.0 °C (42.8 °F) - closed cup
h)	Evaporation rate	No data available
i)	Flammability (solid, gas)	No data available
j)	Upper/lower flammability or explosive limits	No data available
k)	Vapour pressure	No data available
I)	Vapour density	No data available
m)	Relative density	1.28 g/cm3
n)	Water solubility	No data available
o)	Partition coefficient: n- octanol/water	No data available

p)	Auto-ignition	No data available
	temperature	

- q) Decomposition No data available temperature
- r) Viscosity No data available
- s) Explosive properties No data available
- t) Oxidizing properties No data available

# 9.2 Other safety information No data available

#### **10. STABILITY AND REACTIVITY**

- 10.1 Reactivity No data available
- **10.2 Chemical stability** Stable under recommended storage conditions.
- **10.3 Possibility of hazardous reactions** Vapours may form explosive mixture with air.

# **10.4 Conditions to avoid** Heat, flames and sparks. Extremes of temperature and direct sunlight.

- **10.5** Incompatible materials Oxidizing agents
- **10.6 Hazardous decomposition products** Other decomposition products - No data available In the event of fire: see section 5

#### **11. TOXICOLOGICAL INFORMATION**

#### 11.1 Information on toxicological effects

#### Acute toxicity

LC50 Inhalation - Rat - 13700 ppm Remarks: Behavioral:Somnolence (general depressed activity). Liver:Fatty liver degeneration.

Dermal: No data available

No data available

#### Skin corrosion/irritation No data available

Serious eye damage/eye irritation No data available

Respiratory or skin sensitisation No data available

#### Germ cell mutagenicity

No data available

#### Carcinogenicity

- IARC: No component of this product present at levels greater than or equal to 0.1% is identified as probable, possible or confirmed human carcinogen by IARC.
- NTP: No component of this product present at levels greater than or equal to 0.1% is identified as a known or anticipated carcinogen by NTP.
- OSHA: No component of this product present at levels greater than or equal to 0.1% is identified as a carcinogen or potential carcinogen by OSHA.

# **Reproductive toxicity**

No data available

No data available

Specific target organ toxicity - single exposure No data available

Specific target organ toxicity - repeated exposure No data available

#### Aspiration hazard No data available

# **Additional Information**

RTECS: KV9420000

narcosis, To the best of our knowledge, the chemical, physical, and toxicological properties have not been thoroughly investigated.

#### **12. ECOLOGICAL INFORMATION**

#### 12.1 Toxicity

No data available

- 12.2 Persistence and degradability No data available
- **12.3 Bioaccumulative potential** No data available
- 12.4 Mobility in soil No data available
- **12.5** Results of PBT and vPvB assessment PBT/vPvB assessment not available as chemical safety assessment not required/not conducted

# 12.6 Other adverse effects

An environmental hazard cannot be excluded in the event of unprofessional handling or disposal. Harmful to aquatic life.

#### **13. DISPOSAL CONSIDERATIONS**

#### 13.1 Waste treatment methods

#### Product

Burn in a chemical incinerator equipped with an afterburner and scrubber but exert extra care in igniting as this material is highly flammable. Offer surplus and non-recyclable solutions to a licensed disposal company. Contact a licensed professional waste disposal service to dispose of this material.

#### Contaminated packaging

Dispose of as unused product.

#### 14. TRANSPORT INFORMATION

# DOT (US)

UN number: 1150 Class: 3 Proper shipping name: 1,2-Dichloroethylene Packing group: II

Poison Inhalation Hazard: No

# IMDG

UN number: 1150 Class: 3 Packing group: II Proper shipping name: 1,2-DICHLOROETHYLENE EMS-No: F-E, S-D

# ΙΑΤΑ

UN number: 1150 Class: 3 Packing group: II Proper shipping name: 1,2-Dichloroethylene

# **15. REGULATORY INFORMATION**

#### SARA 302 Components

No chemicals in this material are subject to the reporting requirements of SARA Title III, Section 302.

#### SARA 313 Components

This material does not contain any chemical components with known CAS numbers that exceed the threshold (De Minimis) reporting levels established by SARA Title III, Section 313.

~ ~ ~ ~

**–** . .

- -

#### SARA 311/312 Hazards

Fire Hazard

#### Massachusetts Right To Know Components

	CAS-No.	Revision Date
cis-Dichloroethylene	156-59-2	1993-04-24
Pennsylvania Right To Know Components		
	CAS-No.	Revision Date
cis-Dichloroethylene	156-59-2	1993-04-24
New Jersey Right To Know Components		
	CAS-No.	Revision Date
cis-Dichloroethylene	156-59-2	1993-04-24

#### California Prop. 65 Components

This product does not contain any chemicals known to State of California to cause cancer, birth defects, or any other reproductive harm.

# **16. OTHER INFORMATION**

#### Full text of H-Statements referred to under sections 2 and 3.

Acute Tox.	Acute toxicity
Aquatic Acute	Acute aquatic toxicity
Aquatic Chronic	Chronic aquatic toxicity
Flam. Liq.	Flammable liquids
H225	Highly flammable liquid and vapour.
H332	Harmful if inhaled.
H402	Harmful to aquatic life.

2 3 0

#### **HMIS** Rating

Health hazard:	1
Chronic Health Hazard:	*
Flammability:	3
Physical Hazard	
NFPA Rating	

9
Health hazard:
Fire Hazard:
Reactivity Hazard:

#### **Further information**

Copyright 2015 Sigma-Aldrich Co. LLC. License granted to make unlimited paper copies for internal use only. The above information is believed to be correct but does not purport to be all inclusive and shall be used only as a guide. The information in this document is based on the present state of our knowledge and is applicable to the product with regard to appropriate safety precautions. It does not represent any guarantee of the properties of the product. Sigma-Aldrich Corporation and its Affiliates shall not be held liable for any damage resulting from handling or from contact with the above product. See www.sigma-aldrich.com and/or the reverse side of invoice or packing slip for additional terms and conditions of sale.

**Preparation Information** Sigma-Aldrich Corporation Product Safety – Americas Region 1-800-521-8956

Version: 4.4

Revision Date: 12/01/2015

Print Date: 01/05/2020



# **SAFETY DATA SHEET**

Version 6.1 Revision Date 03/12/2019 Print Date 06/22/2019

# SECTION 1: Identification of the substance/mixture and of the company/undertaking

#### **1.1** Product identifiers

Product name : Copper Product Number : 31284 Brand : Aldrich CAS-No. : 7440-50-8

# 1.2 Relevant identified uses of the substance or mixture and uses advised against

Identified uses : Laboratory chemicals, Synthesis of substances

# 1.3 Details of the supplier of the safety data sheet

Company	: Sigma-Aldrich Inc.	
	3050 Spruce Street	
	ST. LOUIS MO 6310	03
	UNITED STATES	
Telephone	: +1 314 771-5765	
Fax	: +1 800 325-5052	

#### **1.4 Emergency telephone number**

Emergency Phone # : +1-703-527-3887

# **SECTION 2: Hazards identification**

# 2.1 Classification of the substance or mixture

Not a hazardous substance or mixture.

#### 2.2 GHS Label elements, including precautionary statements

Not a hazardous substance or mixture.

#### 2.3 Hazards not otherwise classified (HNOC) or not covered by GHS - none

# SECTION 3: Composition/information on ingredients

3.1	Substances				
	Formula	:	Cu		
	Molecular weight	:	63.55 g/mol		
	CAS-No.	:	7440-50-8		
	EC-No.	:	231-159-6		
	Component			Classification	Concentration

Aldrich - 31284

Page 1 of 8

ore



Copper,	
	<= 100 %

# SECTION 4: First aid measures

# 4.1 Description of first aid measures

#### If inhaled

If breathed in, move person into fresh air. If not breathing, give artificial respiration.

#### In case of skin contact

Wash off with soap and plenty of water.

#### In case of eye contact

Flush eyes with water as a precaution.

#### If swallowed

Never give anything by mouth to an unconscious person. Rinse mouth with water.

- **4.2 Most important symptoms and effects, both acute and delayed** The most important known symptoms and effects are described in the labelling (see section 2.2) and/or in section 11
- **4.3 Indication of any immediate medical attention and special treatment needed** No data available

#### **SECTION 5: Firefighting measures**

#### 5.1 Extinguishing media

# Suitable extinguishing media

Use water spray, alcohol-resistant foam, dry chemical or carbon dioxide.

- **5.2** Special hazards arising from the substance or mixture Copper oxides
- **5.3** Advice for firefighters Wear self-contained breathing apparatus for firefighting if necessary.
- **5.4 Further information** No data available

#### **SECTION 6:** Accidental release measures

- **6.1 Personal precautions, protective equipment and emergency procedures** Avoid dust formation. Avoid breathing vapours, mist or gas. For personal protection see section 8.
- **6.2 Environmental precautions** No special environmental precautions required.
- **6.3 Methods and materials for containment and cleaning up** Sweep up and shovel. Keep in suitable, closed containers for disposal.

Aldrich - 31284

Page 2 of 8



For disposal see section 13.

# SECTION 7: Handling and storage

# 7.1 Precautions for safe handling

Further processing of solid materials may result in the formation of combustible dusts. The potential for combustible dust formation should be taken into consideration before additional processing occurs. Provide appropriate exhaust ventilation at places where dust is formed.

For precautions see section 2.2.

# 7.2 Conditions for safe storage, including any incompatibilities

Keep container tightly closed in a dry and well-ventilated place.

Store under inert gas. Air sensitive. Storage class (TRGS 510): 13: Non Combustible Solids

# 7.3 Specific end use(s)

Apart from the uses mentioned in section 1.2 no other specific uses are stipulated

# **SECTION 8: Exposure controls/personal protection**

# 8.1 Control parameters

# Components with workplace control parameters

Component	CAS-No.	Value	Control parameters	Basis
Copper,	7440-50-8	TWA	1 mg/m3	USA. ACGIH Threshold Limit Values (TLV)
	Remarks	Irritation Gastrointes metal fume	stinal e fever	
		TWA	0.2 mg/m3	USA. ACGIH Threshold Limit Values (TLV)
		Irritation Gastrointes metal fume	stinal e fever	

Aldrich - 31284

Page 3 of 8



TWA	1 mg/m3	USA. NIOSH Recommended Exposure Limits
TWA	1 mg/m3	USA. NIOSH Recommended Exposure Limits
TWA	1 mg/m3	USA. Occupational Exposure Limits (OSHA) - Table Z-1 Limits for Air Contaminants
TWA	0.1 mg/m3	USA. Occupational Exposure Limits (OSHA) - Table Z-1 Limits for Air Contaminants
PEL	0.1 mg/m3	California permissible exposure limits for chemical contaminants (Title 8, Article 107)

# 8.2 Exposure controls

#### Appropriate engineering controls

General industrial hygiene practice.

# Personal protective equipment

# Eye/face protection

Use equipment for eye protection tested and approved under appropriate government standards such as NIOSH (US) or EN 166(EU).

#### Skin protection

Handle with gloves. Gloves must be inspected prior to use. Use proper glove removal technique (without touching glove's outer surface) to avoid skin contact with this product. Dispose of contaminated gloves after use in accordance with applicable laws and good laboratory practices. Wash and dry hands.

Full contact Material: Nitrile rubber Minimum layer thickness: 0.11 mm Break through time: 480 min Material tested: Dermatril® (KCL 740 / Aldrich Z677272, Size M)

Splash contact Material: Nitrile rubber Minimum layer thickness: 0.11 mm Break through time: 480 min Material tested: Dermatril® (KCL 740 / Aldrich Z677272, Size M)

data source: KCL GmbH, D-36124 Eichenzell, phone +49 (0)6659 87300, e-mail sales@kcl.de, test method: EN374

If used in solution, or mixed with other substances, and under conditions which differ from EN 374, contact the supplier of the CE approved gloves. This recommendation is advisory only and must be evaluated by an industrial hygienist and safety officer familiar with the specific situation of anticipated use by our customers. It should not be construed as offering an approval for any specific use scenario.

# **Body Protection**

Choose body protection in relation to its type, to the concentration and amount of dangerous substances, and to the specific work-place., The type of protective equipment must be selected according to the concentration and amount of the dangerous substance at the specific workplace.

Aldrich - 31284

Page 4 of 8



# **Respiratory protection**

Respiratory protection is not required. Where protection from nuisance levels of dusts are desired, use type N95 (US) or type P1 (EN 143) dust masks. Use respirators and components tested and approved under appropriate government standards such as NIOSH (US) or CEN (EU).

# **Control of environmental exposure**

No special environmental precautions required.

# SECTION 9: Physical and chemical properties

# 9.1 Information on basic physical and chemical properties

	a)	Appearance	Form: Wire Colour: light red
	b)	Odour	No data available
	C)	Odour Threshold	No data available
	d)	рН	No data available
	e)	Melting point/freezing point	Melting point/range: 1,083.4 °C (1,982.1 °F
	f)	Initial boiling point and boiling range	2,567 °C 4,653 °F
	g)	Flash point	()No data available
	h)	Evaporation rate	No data available
	i)	Flammability (solid, gas)	No data available
	j)	Upper/lower flammability or explosive limits	No data available
	k)	Vapour pressure	No data available
	I)	Vapour density	No data available
	m)	Relative density	8.940 g/cm3
	n)	Water solubility	No data available
	o)	Partition coefficient: n-octanol/water	No data available
	p)	Auto-ignition temperature	No data available
	q)	Decomposition temperature	No data available
	r)	Viscosity	No data available
	s)	Explosive properties	No data available
	t)	Oxidizing properties	No data available
9.2	Otł	her safety informatio	n

No data available

Aldrich - 31284

Page 5 of 8



# **SECTION 10: Stability and reactivity**

- **10.1 Reactivity** No data available
- **10.2 Chemical stability** Stable under recommended storage conditions.
- **10.3 Possibility of hazardous reactions** No data available
- **10.4 Conditions to avoid** No data available
- **10.5 Incompatible materials** Strong acids, Strong oxidizing agents, Acid chlorides, Halogens
- **10.6 Hazardous decomposition products** Other decomposition products - No data available Hazardous decomposition products formed under fire conditions. - Copper oxides In the event of fire: see section 5

# SECTION 11: Toxicological information

# **11.1 Information on toxicological effects**

#### Acute toxicity

No data available Inhalation: No data available Dermal: No data available LD50 Intraperitoneal - Mouse - 3.5 mg/kg

#### Skin corrosion/irritation

No data available

# Serious eye damage/eye irritation

No data available

# **Respiratory or skin sensitisation**

No data available

Germ cell mutagenicity

No data available

# Carcinogenicity

- IARC: No component of this product present at levels greater than or equal to 0.1% is identified as probable, possible or confirmed human carcinogen by IARC.
- NTP: No component of this product present at levels greater than or equal to 0.1% is identified as a known or anticipated carcinogen by NTP.
- OSHA: No component of this product present at levels greater than or equal to 0.1% is on OSHA's list of regulated carcinogens.

# **Reproductive toxicity**

No data available No data available

# Specific target organ toxicity - single exposure

No data available

Aldrich - 31284

Page 6 of 8



# Specific target organ toxicity - repeated exposure

No data available

#### **Aspiration hazard**

No data available

# **Additional Information**

RTECS: GL5325000

To the best of our knowledge, the chemical, physical, and toxicological properties have not been thoroughly investigated.

# **SECTION 12: Ecological information**

#### **12.1** Toxicity

No data available

# **12.2 Persistence and degradability** The methods for determining biodegradability are not applicable to inorganic substances.

- 12.3 Bioaccumulative potential No data available
- **12.4 Mobility in soil** No data available
- 12.5 Results of PBT and vPvB assessment PBT/vPvB assessment not available as chemical safety assessment not required/not conducted

# 12.6 Other adverse effects

# **SECTION 13: Disposal considerations**

# **13.1** Waste treatment methods

#### Product

Offer surplus and non-recyclable solutions to a licensed disposal company.

#### **Contaminated packaging**

Dispose of as unused product.

# **SECTION 14: Transport information**

#### DOT (US)

Not dangerous goods

#### IMDG

Not dangerous goods

# ΙΑΤΑ

Not dangerous goods

Aldrich - 31284

Page 7 of 8



# SECTION 15: Regulatory information

# SARA 302 Components

No chemicals in this material are subject to the reporting requirements of SARA Title III, Section 302.

# SARA 313 Components

This material does not contain any chemical components with known CAS numbers that exceed the threshold (De Minimis) reporting levels established by SARA Title III, Section 313.

# SARA 311/312 Hazards

No SARA Hazards

# **Massachusetts Right To Know Components**

No components are subject to the Massachusetts Right to Know Act.

Pennsylvania Right To Know Components	CAS-No.	Revision Date
Copper,	7440-50-8	1993-02-16
Copper,	CAS-No. 7440-50-8	Revision Date 1993-02-16
New Jersey Right To Know Components	CAS-No.	Revision Date
Copper,	7440-50-8	1993-02-16

# California Prop. 65 Components

This product does not contain any chemicals known to State of California to cause cancer, birth defects, or any other reproductive harm.

# **SECTION 16: Other information**

# Further information

Copyright 2018 Sigma-Aldrich Co. LLC. License granted to make unlimited paper copies for internal use only.

The above information is believed to be correct but does not purport to be all inclusive and shall be used only as a guide. The information in this document is based on the present state of our knowledge and is applicable to the product with regard to appropriate safety precautions. It does not represent any guarantee of the properties of the product. Sigma-Aldrich Corporation and its Affiliates shall not be held liable for any damage resulting from handling or from contact with the above product. See www.sigma-aldrich.com and/or the reverse side of invoice or packing slip for additional terms and conditions of sale.

The branding on the header and/or footer of this document may temporarily not visually match the product purchased as we transition our branding. However, all of the information in the document regarding the product remains unchanged and matches the product ordered. For further information please contact mlsbranding@sial.com.

Version: 6.1

Revision Date: 03/12/2019

Print Date: 06/22/2019

Aldrich - 31284

Page 8 of 8





# **SAFETY DATA SHEET**

Version 6.2 Revision Date 01/15/2020 Print Date 02/03/2020

# SECTION 1: Identification of the substance/mixture and of the company/undertaking

#### **1.1 Product identifiers**

Product name : Ethanol Product Number : 24194

Product Number	:	24194
Brand	:	SIGALD
CAS-No.	:	64-17-5

# 1.2 Relevant identified uses of the substance or mixture and uses advised against

Identified uses : Laboratory chemicals, Synthesis of substances

# 1.3 Details of the supplier of the safety data sheet

Company	: Sigma-Aldrich Inc. 3050 Spruce Street ST. LOUIS MO 63103
Telephone	UNITED STATES : +1 314 771-5765
Fax	: +1 800 325-5052

# **1.4 Emergency telephone number**

Emergency Phone # : +1-703-527-3887

# **SECTION 2: Hazards identification**

# 2.1 Classification of the substance or mixture

# GHS Classification in accordance with 29 CFR 1910 (OSHA HCS)

Flammable liquids (Category 2), H225 Eye irritation (Category 2A), H319

For the full text of the H-Statements mentioned in this Section, see Section 16.

# 2.2 GHS Label elements, including precautionary statements

Pictogram



Signal word

Danger

Hazard statement(s) H225 H319

Highly flammable liquid and vapour. Causes serious eye irritation.

Precautionary statement(s) P210

SIGALD - 24194

Keep away from heat/sparks/open flames/hot surfaces. No

Page 1 of 11



	smoking.
P233	Cround/band container and receiving equipment
P240	
P241	Use explosion-proof electrical/ventilating/lighting equipment.
P242	Use only non-sparking tools.
P243	Take precautionary measures against static discharge.
P264	Wash skin thoroughly after handling.
P280	Wear protective gloves/ eye protection/ face protection.
P303 + P361 + P353	IF ON SKIN (or hair): Take off immediately all contaminated
	clothing. Rinse skin with water/shower.
P305 + P351 + P338	IF IN EYES: Rinse cautiously with water for several minutes.
	Remove contact lenses, if present and easy to do. Continue
	rinsing.
P337 + P313	If eye irritation persists: Get medical advice/ attention.
P370 + P378	In case of fire: Use dry sand, dry chemical or alcohol-resistant
	foam to extinguish.
P403 + P235	Store in a well-ventilated place. Keep cool.
P501	Dispose of contents/ container to an approved waste disposal
	plant.

# 2.3 Hazards not otherwise classified (HNOC) or not covered by GHS - none

# SECTION 3: Composition/information on ingredients

.1	<b>Substances</b> Synonyms	:	Ethyl alcohol		
	Formula	:	C <sub>2</sub> H <sub>6</sub> O		
	Molecular weight CAS-No.	:	46.07 g/mol 64-17-5		
	Component			Classification	Concentration
	Ethanol				
				Flam. Liq. 2; Eye Irrit. 2A; H225, H319	<= 100 %

Ethyl methyl ketone		
	Flam. Liq. 2; Eye Irrit. 2A;	>= 1 - < 5 %
	STOT SE 3; H225, H319,	
	H336	
	Concentration limits:	
	20 %: STOT SE 3, H336;	

For the full text of the H-Statements mentioned in this Section, see Section 16.

SIGALD - 24194

Page 2 of 11



# **SECTION 4: First aid measures**

# 4.1 Description of first aid measures

#### **General advice**

Consult a physician. Show this safety data sheet to the doctor in attendance. Move out of dangerous area.

#### If inhaled

If breathed in, move person into fresh air. If not breathing, give artificial respiration. Consult a physician.

#### In case of skin contact

Wash off with soap and plenty of water. Consult a physician.

#### In case of eye contact

Rinse thoroughly with plenty of water for at least 15 minutes and consult a physician.

#### If swallowed

Do NOT induce vomiting. Never give anything by mouth to an unconscious person. Rinse mouth with water. Consult a physician.

## 4.2 Most important symptoms and effects, both acute and delayed

The most important known symptoms and effects are described in the labelling (see section 2.2) and/or in section 11

# **4.3 Indication of any immediate medical attention and special treatment needed** No data available

# **SECTION 5: Firefighting measures**

#### 5.1 Extinguishing media

#### Suitable extinguishing media

Use water spray, alcohol-resistant foam, dry chemical or carbon dioxide.

- **5.2** Special hazards arising from the substance or mixture Carbon oxides
- **5.3** Advice for firefighters Wear self-contained breathing apparatus for firefighting if necessary.

#### 5.4 Further information

Use water spray to cool unopened containers.

### **SECTION 6:** Accidental release measures

# 6.1 Personal precautions, protective equipment and emergency procedures Use personal protective equipment. Avoid breathing vapours, mist or gas. Ensure adequate ventilation. Remove all sources of ignition. Evacuate personnel to safe areas. Beware of vapours accumulating to form explosive concentrations. Vapours can accumulate in low areas. For personal protection see section 8.

6.2 Environmental precautions

Prevent further leakage or spillage if safe to do so. Do not let product enter drains.

SIGALD - 24194

Page 3 of 11



# 6.3 Methods and materials for containment and cleaning up

Contain spillage, and then collect with an electrically protected vacuum cleaner or by wetbrushing and place in container for disposal according to local regulations (see section 13).

# 6.4 Reference to other sections

For disposal see section 13.

# SECTION 7: Handling and storage

# 7.1 Precautions for safe handling

Avoid contact with skin and eyes. Avoid inhalation of vapour or mist. Use explosion-proof equipment.Keep away from sources of ignition - No smoking.Take measures to prevent the build up of electrostatic charge. For precautions see section 2.2.

# 7.2 Conditions for safe storage, including any incompatibilities

Keep container tightly closed in a dry and well-ventilated place. Containers which are opened must be carefully resealed and kept upright to prevent leakage. Storage class (TRGS 510): 3: Flammable liquids

# 7.3 Specific end use(s)

Apart from the uses mentioned in section 1.2 no other specific uses are stipulated

# SECTION 8: Exposure controls/personal protection

# 8.1 Control parameters

# Components with workplace control parameters

Component	CAS-No.	Value	Control parameters	Basis
Ethanol	64-17-5	TWA	1,000 ppm 1,900 mg/m3	USA. OSHA - TABLE Z-1 Limits for Air Contaminants - 1910.1000
		TWA	1,000 ppm 1,900 mg/m3	USA. Occupational Exposure Limits (OSHA) - Table Z-1 Limits for Air Contaminants
	Remarks	The value i	n mg/m3 is appr	roximate.
		STEL	1,000 ppm	USA. ACGIH Threshold Limit Values (TLV)
		Upper Resp Confirmed humans	iratory Tract irri animal carcinoge	tation en with unknown relevance to

Page 4 of 11



		TWA	1,000 ppm 1,900 mg/m3	USA. NIOSH Recommended Exposure Limits
		PEL	1,000 ppm 1,900 mg/m3	California permissible exposure limits for chemical contaminants (Title 8, Article 107)
Ethyl methyl ketone	78-93-3	TWA	200 ppm	USA. ACGIH Threshold Limit Values (TLV)
		Central Net Upper Resp Peripheral Substances or Indices	rvous System im piratory Tract irr Nervous System s for which there (see BEI® sectic	ppairment itation impairment is a Biological Exposure Index on)
		STEL	300 ppm	USA. ACGIH Threshold Limit Values (TLV)
		Central Nei Upper Resp Peripheral Substances or Indices	rvous System im piratory Tract irr Nervous System s for which there (see BEL® section	pairment itation impairment is a Biological Exposure Index on)
		TWA	200 ppm 590 mg/m3	USA. NIOSH Recommended Exposure Limits
		ST	300 ppm 885 mg/m3	USA. NIOSH Recommended Exposure Limits
		TWA	200 ppm 590 mg/m3	USA. Occupational Exposure Limits (OSHA) - Table Z-1 Limits for Air Contaminants
		The value i	n mg/m3 is app	roximate.
		PEL	200 ppm 590 mg/m3	California permissible exposure limits for chemical contaminants (Title 8, Article 107)
		STEL	300 ppm 885 mg/m3	California permissible exposure limits for chemical contaminants (Title 8, Article 107)

# **Biological occupational exposure limits**

Component	CAS-No.	Parameters	Value	Biological specimen	Basis
Ethyl methyl ketone	78-93-3	methyl ethyl ketone	2 mg/l	Urine	ACGIH - Biological Exposure Indices (BEI)
	Remarks	End of shift (As soon as possible after exposure ceases)			

# 8.2 Exposure controls

# Appropriate engineering controls

Handle in accordance with good industrial hygiene and safety practice. Wash hands before breaks and at the end of workday.

SIGALD - 24194

Page 5 of 11



# **Personal protective equipment**

# Eye/face protection

Face shield and safety glasses Use equipment for eye protection tested and approved under appropriate government standards such as NIOSH (US) or EN 166(EU).

# **Skin protection**

Handle with gloves. Gloves must be inspected prior to use. Use proper glove removal technique (without touching glove's outer surface) to avoid skin contact with this product. Dispose of contaminated gloves after use in accordance with applicable laws and good laboratory practices. Wash and dry hands.

Full contact Material: butyl-rubber Minimum layer thickness: 0.3 mm Break through time: > 480 min Material tested: Butoject® (KCL 897 / Aldrich Z677647, Size M)

Splash contact Material: Nitrile rubber Minimum layer thickness: 0.11 mm Break through time: 175 min Material tested: Dermatril® (KCL 740 / Aldrich Z677272, Size M)

data source: KCL GmbH, D-36124 Eichenzell, phone +49 (0)6659 87300, e-mail sales@kcl.de, test method: EN374

If used in solution, or mixed with other substances, and under conditions which differ from EN 374, contact the supplier of the CE approved gloves. This recommendation is advisory only and must be evaluated by an industrial hygienist and safety officer familiar with the specific situation of anticipated use by our customers. It should not be construed as offering an approval for any specific use scenario.

#### **Body Protection**

Impervious clothing, Flame retardant antistatic protective clothing., The type of protective equipment must be selected according to the concentration and amount of the dangerous substance at the specific workplace.

# **Respiratory protection**

Where risk assessment shows air-purifying respirators are appropriate use a fullface respirator with multi-purpose combination (US) or type ABEK (EN 14387) respirator cartridges as a backup to engineering controls. If the respirator is the sole means of protection, use a full-face supplied air respirator. Use respirators and components tested and approved under appropriate government standards such as NIOSH (US) or CEN (EU).

# **Control of environmental exposure**

Prevent further leakage or spillage if safe to do so. Do not let product enter drains.

# **SECTION 9: Physical and chemical properties**

# 9.1 Information on basic physical and chemical properties

- a) Appearance Form: liquid, clear Colour: colourless
- b) Odour No data available
- c) Odour Threshold No data available

SIGALD - 24194

Page 6 of 11



d)	рН	No data available
e)	Melting point/freezing point	Melting point/range: -114 °C (-173 °F) - lit.
f)	Initial boiling point and boiling range	78 °C 172 °F - lit.
g)	Flash point	14.0 °C (57.2 °F) - closed cup
h)	Evaporation rate	No data available
i)	Flammability (solid, gas)	No data available
j)	Upper/lower flammability or explosive limits	Upper explosion limit: 19 %(V) Lower explosion limit: 3.3 %(V)
k)	Vapour pressure	59.5 hPa at 20.0 °C (68.0 °F)
I)	Vapour density	No data available
m)	Relative density	0.789 g/mL at 25 °C (77 °F)
n)	Water solubility	completely soluble
o)	Partition coefficient: n-octanol/water	log Pow: -0.349 at 24 °C (75 °F)
p)	Auto-ignition temperature	363.0 °C (685.4 °F)
q)	Decomposition temperature	No data available
r)	Viscosity	No data available
s)	Explosive properties	No data available
t)	Oxidizing properties	No data available

9.2 Other safety information

No data available

# SECTION 10: Stability and reactivity

# **10.1 Reactivity**

No data available

**10.2 Chemical stability** Stable under recommended storage conditions.

# **10.3 Possibility of hazardous reactions** Vapours may form explosive mixture with air.

# 10.4 Conditions to avoid

Heat, flames and sparks.

- **10.5 Incompatible materials** Alkali metals, Oxidizing agents, Peroxides
- **10.6 Hazardous decomposition products** Hazardous decomposition products formed under fire conditions. - Carbon oxides Other decomposition products - No data available

SIGALD - 24194

Page 7 of 11



# **SECTION 11: Toxicological information**

## **11.1** Information on toxicological effects

#### **Acute toxicity**

LD50 Oral - Rat - male and female - 10,470 mg/kg (OECD Test Guideline 401) LC50 Inhalation - Rat - male and female - 4 h - 124.7 mg/l (OECD Test Guideline 403) Dermal: No data available No data available

# Skin corrosion/irritation

Skin - Rabbit Result: No skin irritation - 24 h (OECD Test Guideline 404)

# Serious eye damage/eye irritation

Eyes - Rabbit Result: Causes serious eye irritation. (OECD Test Guideline 405)

# Respiratory or skin sensitisation

Maximisation Test - Guinea pig Result: negative (OECD Test Guideline 406) Remarks: (in analogy to similar products)

#### Germ cell mutagenicity

Ames test Salmonella typhimurium Result: negative In vitro mammalian cell gene mutation test mouse lymphoma cells Result: negative OECD Test Guideline 478 Mouse - male Result: Positive results were obtained in some in vivo tests.

# Carcinogenicity

- IARC: No component of this product present at levels greater than or equal to 0.1% is identified as probable, possible or confirmed human carcinogen by IARC.
- NTP: No component of this product present at levels greater than or equal to 0.1% is identified as a known or anticipated carcinogen by NTP.
- OSHA: No component of this product present at levels greater than or equal to 0.1% is on OSHA's list of regulated carcinogens.

# **Reproductive toxicity**

No data available

# Specific target organ toxicity - single exposure

No data available

SIGALD - 24194

Page 8 of 11



# Specific target organ toxicity - repeated exposure

No data available

#### **Aspiration hazard**

No data available

#### **Additional Information**

Repeated dose toxicity - Rat - male - Oral - No observed adverse effect level - 1,730 mg/kg - Lowest observed adverse effect level - 3,200 mg/kg RTECS: Not available

irritant effects, respiratory paralysis, Dizziness, narcosis, inebriation, euphoria, Nausea, Vomiting

To the best of our knowledge, the chemical, physical, and toxicological properties have not been thoroughly investigated.

Stomach - Irregularities - Based on Human Evidence Liver - Irregularities - Based on Human Evidence Stomach - Irregularities - Based on Human Evidence

# **SECTION 12: Ecological information**

#### **12.1 Toxicity**

	Toxicity to fish	flow-through test LC50 - Pimephales promelas (fathead minnow) - 15,300 mg/l - 96 h (US-EPA)			
	Toxicity to daphnia and other aquatic invertebrates	static test LC50 - Ceriodaphnia dubia (water flea) - 5,012 mg/l - 48 h Remarks: (ECHA)			
	Toxicity to algae	static test ErC50 - Chlorella vulgaris (Fresh water algae) - 275 mg/l - 72 h (OECD Test Guideline 201)			
	Toxicity to bacteria	static test IC50 - activated sludge - > 1,000 mg/l - 3 h (OECD Test Guideline 209)			
12.2	Persistence and degradability				
	Biodegradability	aerobic - Exposure time 15 d Result: ca.95 % - Readily biodegradable. (OECD Test Guideline 301E)			
	Biochemical Oxygen Demand (BOD)	930 - 1,670 mg/g Remarks: (Lit.)			

Theoretical oxygen2,100 mg/gdemandRemarks: (Lit.)

#### 12.3 Bioaccumulative potential

Due to the distribution coefficient n-octanol/water, accumulation in organisms is not expected.

# 12.4 Mobility in soil

No data available

SIGALD - 24194

Page 9 of 11



# 12.5 Results of PBT and vPvB assessment

PBT/vPvB assessment not available as chemical safety assessment not required/not conducted

# 12.6 Other adverse effects

Additional ecological No data available information

# SECTION 13: Disposal considerations

# 13.1 Waste treatment methods

# Product

Contact a licensed professional waste disposal service to dispose of this material. Burn in a chemical incinerator equipped with an afterburner and scrubber but exert extra care in igniting as this material is highly flammable. Offer surplus and non-recyclable solutions to a licensed disposal company.

# **Contaminated packaging**

Dispose of as unused product.

# **SECTION 14: Transport information**

# DOT (US)

UN number: 1170 Class: 3 Proper shipping name: Ethanol Reportable Quantity (RQ): 5000 lbs Reportable Quantity (RQ): 100 lbs Reportable Quantity (RQ): Poison Inhalation Hazard: No	Packing group: 11	
<b>IMDG</b> UN number: 1170 Class: 3 Proper shipping name: ETHANOL	Packing group: 11	EMS-No: F-E, S-D
<b>IATA</b> UN number: 1170 Class: 3 Proper shipping name: Ethanol	Packing group: 11	

# SECTION 15: Regulatory information

# SARA 302 Components

No chemicals in this material are subject to the reporting requirements of SARA Title III, Section 302.

# SARA 313 Components

This material does not contain any chemical components with known CAS numbers that exceed the threshold (De Minimis) reporting levels established by SARA Title III, Section 313.

# SARA 311/312 Hazards

Fire Hazard, Acute Health Hazard, Chronic Health Hazard

SIGALD - 24194

Page 10 of 11



**Reportable Quantity** D035 lbs

F005 lbs

# **Massachusetts Right To Know Components**

Ethanol	CAS-No. 64-17-5	Revision Date 1993-04-24
Pennsylvania Right To Know Components Ethanol	CAS-No. 64-17-5	Revision Date 1993-04-24
Ethyl methyl ketone	78-93-3	1993-02-16

# California Prop. 65 Components

This product does not contain any chemicals known to State of California to cause cancer, birth defects, or any other reproductive harm.

# **SECTION 16: Other information**

# **Further information**

Copyright 2020 Sigma-Aldrich Co. LLC. License granted to make unlimited paper copies for internal use only.

The above information is believed to be correct but does not purport to be all inclusive and shall be used only as a guide. The information in this document is based on the present state of our knowledge and is applicable to the product with regard to appropriate safety precautions. It does not represent any guarantee of the properties of the product. Sigma-Aldrich Corporation and its Affiliates shall not be held liable for any damage resulting from handling or from contact with the above product. See www.sigma-aldrich.com and/or the reverse side of invoice or packing slip for additional terms and conditions of sale.

The branding on the header and/or footer of this document may temporarily not visually match the product purchased as we transition our branding. However, all of the information in the document regarding the product remains unchanged and matches the product ordered. For further information please contact mlsbranding@sial.com.

Version: 6.2

Revision Date: 01/15/2020

Print Date: 02/03/2020

SIGALD - 24194

Page 11 of 11





# SIGMA-ALDRICH

# SAFETY DATA SHEET

Version 5.12 Revision Date 07/26/2018 Print Date 11/10/2018

#### **1. PRODUCT AND COMPANY IDENTIFICATION**

1.1	Product identifiers Product name	:	Heptane
	Product Number Brand Index-No.	: : :	246654 Sigma-Aldrich 601-008-00-2
	CAS-No.	:	142-82-5

#### 1.2 Relevant identified uses of the substance or mixture and uses advised against

Identified uses : Laboratory chemicals, Synthesis of substances

#### 1.3 Details of the supplier of the safety data sheet

Company	:	Sigma-Aldrich 3050 Spruce Street SAINT LOUIS MO 63103 USA
Telephone Fax	:	+1 800-325-5832 +1 800-325-5052

#### 1.4 Emergency telephone number

Emergency Phone # : +1-703-527-3887 (CHEMTREC)

#### 2. HAZARDS IDENTIFICATION

#### 2.1 Classification of the substance or mixture

#### GHS Classification in accordance with 29 CFR 1910 (OSHA HCS)

Flammable liquids (Category 2), H225 Skin irritation (Category 2), H315 Specific target organ toxicity - single exposure (Category 3), Central nervous system, H336 Aspiration hazard (Category 1), H304 Acute aquatic toxicity (Category 1), H400 Chronic aquatic toxicity (Category 1), H410

For the full text of the H-Statements mentioned in this Section, see Section 16.

#### 2.2 GHS Label elements, including precautionary statements

Pictogram



Signal word

F

Danger

lazard statement(s)	
H225	Highly flammable liquid and vapour.
H304	May be fatal if swallowed and enters airways.
H315	Causes skin irritation.
H336	May cause drowsiness or dizziness.
H410	Very toxic to aquatic life with long lasting effects.
Precautionary statement(s)	
P210	Keep away from heat/sparks/open flames/hot surfaces. No smoking.
P233	Keep container tightly closed.

P240	Ground/bond container and receiving equipment.
P241	Use explosion-proof electrical/ ventilating/ lighting/ equipment.
P242	Use only non-sparking tools.
P243	Take precautionary measures against static discharge.
P261	Avoid breathing dust/ fume/ gas/ mist/ vapours/ spray.
P264	Wash skin thoroughly after handling.
P271	Use only outdoors or in a well-ventilated area.
P273	Avoid release to the environment.
P280	Wear protective gloves/ eye protection/ face protection.
P301 + P310	IF SWALLOWED: Immediately call a POISON CENTER/doctor.
P303 + P361 + P353	IF ON SKIN (or hair): Take off immediately all contaminated clothing. Rinse skin with water/shower.
P304 + P340 + P312	IF INHALED: Remove person to fresh air and keep comfortable for breathing. Call a POISON CENTER/doctor if you feel unwell.
P331	Do NOT induce vomiting.
P332 + P313	If skin irritation occurs: Get medical advice/ attention.
P362	Take off contaminated clothing and wash before reuse.
P370 + P378	In case of fire: Use dry sand, dry chemical or alcohol-resistant foam to extinguish.
P391	Collect spillage.
P403 + P233	Store in a well-ventilated place. Keep container tightly closed.
P403 + P235	Store in a well-ventilated place. Keep cool.
P405	Store locked up.
P501	Dispose of contents/ container to an approved waste disposal plant.

# 2.3 Hazards not otherwise classified (HNOC) or not covered by GHS - none

# **3. COMPOSITION/INFORMATION ON INGREDIENTS**

# 3.1 Substances

Formula	:	С <sub>7</sub> Н <sub>16</sub>
Molecular weight	:	100.20 g/mol
CAS-No.	:	142-82-5
EC-No.	:	205-563-8
Index-No.	:	601-008-00-2
Registration number	:	01-2119457603-38-XXXX

#### Hazardous components

Component	Classification	Concentration
Heptane		
	Flam. Liq. 2; Skin Irrit. 2; STOT SE 3; Asp. Tox. 1; Aquatic Acute 1; Aquatic Chronic 1; H225, H304, H315, H336, H410	90 - 100 %

For the full text of the H-Statements mentioned in this Section, see Section 16.

# 4. FIRST AID MEASURES

# 4.1 Description of first aid measures

# General advice

Consult a physician. Show this safety data sheet to the doctor in attendance. Move out of dangerous area.

# If inhaled

If breathed in, move person into fresh air. If not breathing, give artificial respiration. Consult a physician.

# In case of skin contact

Wash off with soap and plenty of water. Consult a physician.
#### In case of eye contact

Flush eyes with water as a precaution.

#### If swallowed

Do NOT induce vomiting. Never give anything by mouth to an unconscious person. Rinse mouth with water. Consult a physician.

#### 4.2 Most important symptoms and effects, both acute and delayed

The most important known symptoms and effects are described in the labelling (see section 2.2) and/or in section 11

#### **4.3 Indication of any immediate medical attention and special treatment needed** No data available

#### **5. FIREFIGHTING MEASURES**

#### 5.1 Extinguishing media

**Suitable extinguishing media** Dry powder Dry sand

**Unsuitable extinguishing media** Do NOT use water jet.

**5.2** Special hazards arising from the substance or mixture Flash back possible over considerable distance.

#### 5.3 Advice for firefighters

Wear self-contained breathing apparatus for firefighting if necessary.

#### 5.4 Further information

In case of fire: Evacuate area. Fight fire remotely due to the risk of explosion. Use water spray to cool unopened containers.

#### 6. ACCIDENTAL RELEASE MEASURES

#### 6.1 Personal precautions, protective equipment and emergency procedures

Use personal protective equipment. Avoid breathing vapours, mist or gas. Ensure adequate ventilation. Remove all sources of ignition. Evacuate personnel to safe areas. Beware of vapours accumulating to form explosive concentrations. Vapours can accumulate in low areas. For personal protection see section 8.

#### 6.2 Environmental precautions

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

#### 6.3 Methods and materials for containment and cleaning up

Contain spillage, and then collect with non-combustible absorbent material, (e.g. sand, earth, diatomaceous earth, vermiculite) and place in container for disposal according to local / national regulations (see section 13).

#### 6.4 Reference to other sections

For disposal see section 13.

#### 7. HANDLING AND STORAGE

#### 7.1 Precautions for safe handling

Avoid contact with skin and eyes. Avoid inhalation of vapour or mist. Use explosion-proof equipment.Keep away from sources of ignition - No smoking.Take measures to prevent the build up of electrostatic charge.

For precautions see section 2.2.

#### 7.2 Conditions for safe storage, including any incompatibilities

Store under inert gas. Keep container tightly closed in a dry and well-ventilated place. Containers which are opened must be carefully resealed and kept upright to prevent leakage. Storage class (TRGS 510): 3: Flammable liquids

#### 7.3 Specific end use(s)

Apart from the uses mentioned in section 1.2 no other specific uses are stipulated

#### 8. EXPOSURE CONTROLS/PERSONAL PROTECTION

#### 8.1 Control parameters

#### Components with workplace control parameters

Component	CAS-No.	Value	Control parameters	Basis
Heptane	142-82-5	TWA	85 ppm	USA. NIOSH Recommended
			350 mg/m3	Exposure Limits
		С	440 ppm	USA. NIOSH Recommended
			1,800 mg/m3	Exposure Limits
	Remarks	15 minute ce	iling value	
		TWA	500 ppm	USA. Occupational Exposure Limits
			2,000 mg/m3	(OSHA) - Table Z-1 Limits for Air
			-	Contaminants
		The value in	mg/m3 is approxir	nate.
		PEL	400 ppm	California permissible exposure
			1,600 mg/m3	limits for chemical contaminants
				(Title 8, Article 107)
		STEL	500 ppm	California permissible exposure
			2,000 mg/m3	limits for chemical contaminants
			-	(Title 8, Article 107)
		TWA	400 ppm	USA. ACGIH Threshold Limit Values
				(TLV)
		Central Nerv	ous System impair	ment
		Upper Respiratory Tract irritation		
		STEL	500 ppm	USA. ACGIH Threshold Limit Values
				(TLV)
		Central Nervous System impairment		
		Upper Respi	oper Respiratory Tract irritation	

#### 8.2 Exposure controls

#### Appropriate engineering controls

Handle in accordance with good industrial hygiene and safety practice. Wash hands before breaks and at the end of workday.

#### Personal protective equipment

#### **Eye/face protection**

Face shield and safety glasses Use equipment for eye protection tested and approved under appropriate government standards such as NIOSH (US) or EN 166(EU).

#### Skin protection

Handle with gloves. Gloves must be inspected prior to use. Use proper glove removal technique (without touching glove's outer surface) to avoid skin contact with this product. Dispose of contaminated gloves after use in accordance with applicable laws and good laboratory practices. Wash and dry hands.

Full contact Material: Nitrile rubber Minimum layer thickness: 0.4 mm Break through time: 480 min Material tested:Camatril® (KCL 730 / Aldrich Z677442, Size M)

Splash contact Material: Nitrile rubber Minimum layer thickness: 0.2 mm Break through time: 65 min Material tested:Dermatril® P (KCL 743 / Aldrich Z677388, Size M)

data source: KCL GmbH, D-36124 Eichenzell, phone +49 (0)6659 87300, e-mail sales@kcl.de, test method: EN374

If used in solution, or mixed with other substances, and under conditions which differ from EN 374, contact the supplier of the CE approved gloves. This recommendation is advisory only and must be evaluated by an

industrial hygienist and safety officer familiar with the specific situation of anticipated use by our customers. It should not be construed as offering an approval for any specific use scenario.

#### **Body Protection**

Complete suit protecting against chemicals, Flame retardant antistatic protective clothing., The type of protective equipment must be selected according to the concentration and amount of the dangerous substance at the specific workplace.

#### **Respiratory protection**

Where risk assessment shows air-purifying respirators are appropriate use a full-face respirator with multipurpose combination (US) or type ABEK (EN 14387) respirator cartridges as a backup to engineering controls. If the respirator is the sole means of protection, use a full-face supplied air respirator. Use respirators and components tested and approved under appropriate government standards such as NIOSH (US) or CEN (EU).

#### **Control of environmental exposure**

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

#### 9. PHYSICAL AND CHEMICAL PROPERTIES

9.1 Information on basic physical and chemical properties

a)	Appearance	Form: liquid
b)	Odour	No data available
c)	Odour Threshold	No data available
d)	рН	No data available
e)	Melting point/freezing point	Melting point/range: -91 °C (-132 °F)
f)	Initial boiling point and boiling range	98 °C (208 °F)
g)	Flash point	22 °C (72 °F) - closed cup
h)	Evaporation rate	No data available
i)	Flammability (solid, gas)	No data available
j)	Upper/lower flammability or explosive limits	Upper explosion limit: 7 %(V) Lower explosion limit: 1.1 %(V)
k)	Vapour pressure	110.7 hPa (83.0 mmHg) at 37.7 °C (99.9 °F) 53.3 hPa (40.0 mmHg) at 20.0 °C (68.0 °F)
I)	Vapour density	No data available
m)	Relative density	0.684 g/mL at 25 °C (77 °F)
n)	Water solubility	insoluble
o)	Partition coefficient: n- octanol/water	log Pow: > 3.000
p)	Auto-ignition temperature	223.0 °C (433.4 °F)
q)	Decomposition temperature	No data available
r)	Viscosity	No data available
s)	Explosive properties	No data available
t)	Oxidizing properties	No data available
the	safety information	

#### 9.2 Other safety information No data available

#### **10. STABILITY AND REACTIVITY**

- **10.1 Reactivity** No data available
- **10.2 Chemical stability** Stable under recommended storage conditions.
- **10.3 Possibility of hazardous reactions** Vapours may form explosive mixture with air.
- **10.4 Conditions to avoid** Heat, flames and sparks.
- **10.5** Incompatible materials Strong oxidizing agents
- 10.6 Hazardous decomposition products Other decomposition products - No data available Hazardous decomposition products formed under fire conditions. - Carbon oxides In the event of fire: see section 5

#### 11. TOXICOLOGICAL INFORMATION

#### 11.1 Information on toxicological effects

## Acute toxicity

No data available

LC50 Inhalation - Rat - 4 h - 103,000 mg/m3

Inhalation: Irritating to respiratory system.

Dermal: No data available

No data available

#### Skin corrosion/irritation No data available

#### Serious eye damage/eye irritation

Eyes - Rabbit Result: No eye irritation (OECD Test Guideline 405)

#### **Respiratory or skin sensitisation** No data available

Germ cell mutagenicity No data available

#### Carcinogenicity

This product is or contains a component that is not classifiable as to its carcinogenicity based on its IARC, ACGIH, NTP, or EPA classification.

- IARC: No component of this product present at levels greater than or equal to 0.1% is identified as probable, possible or confirmed human carcinogen by IARC.
- NTP: No component of this product present at levels greater than or equal to 0.1% is identified as a known or anticipated carcinogen by NTP.
- OSHA: No component of this product present at levels greater than or equal to 0.1% is on OSHA's list of regulated carcinogens.

#### **Reproductive toxicity**

No data available

No data available

#### Specific target organ toxicity - single exposure

May cause drowsiness or dizziness.

#### Specific target organ toxicity - repeated exposure No data available

#### Aspiration hazard

May be fatal if swallowed and enters airways.

#### Additional Information

RTECS: MI7700000

Prolonged or repeated exposure to skin causes defatting and dermatitis., Central nervous system depression, narcosis, Damage to the lungs.

Stomach - Irregularities - Based on Human Evidence Stomach - Irregularities - Based on Human Evidence

#### **12. ECOLOGICAL INFORMATION**

#### 12.1 Toxicity

Toxicity to fish LC50 - Carassius auratus (goldfish) - 4 mg/l - 24.0 h LC50 - Tilapia mossambica - 375 mg/l - 96.0 h Toxicity to daphnia and EC50 - Daphnia magna (Water flea) - 1.50 mg/l - 48 h other aquatic invertebrates

12.2 Persistence and degradability

Ratio BOD/ThBOD 3.5 %

12.3 Bioaccumulative potential Indication of bioaccumulation.

#### 12.4 Mobility in soil

No data available

Results of PBT and vPvB assessment 12.5 PBT/vPvB assessment not available as chemical safety assessment not required/not conducted

#### 12.6 Other adverse effects

Very toxic to aquatic life with long lasting effects. An environmental hazard cannot be excluded in the event of unprofessional handling or disposal.

Do not empty into drains. Avoid release to the environment.

#### **13. DISPOSAL CONSIDERATIONS**

#### 13.1 Waste treatment methods

#### Product

Burn in a chemical incinerator equipped with an afterburner and scrubber but exert extra care in igniting as this material is highly flammable. Offer surplus and non-recyclable solutions to a licensed disposal company. Contact a licensed professional waste disposal service to dispose of this material.

#### Contaminated packaging

Dispose of as unused product.

#### **14. TRANSPORT INFORMATION**

#### DOT (US)

UN number: 1206 Class: 3 Packing group: II Proper shipping name: Heptanes Reportable Quantity (RQ): Marine pollutant:yes Poison Inhalation Hazard: No

#### IMDG

UN number: 1206 Class: 3

#### ΙΑΤΑ

UN number: 1206 Class: 3 Proper shipping name: Heptanes Packing group: II

#### **15. REGULATORY INFORMATION**

#### SARA 302 Components

No chemicals in this material are subject to the reporting requirements of SARA Title III, Section 302.

#### SARA 313 Components

This material does not contain any chemical components with known CAS numbers that exceed the threshold (De Minimis) reporting levels established by SARA Title III, Section 313.

#### SARA 311/312 Hazards

Fire Hazard, Acute Health Hazard, Chronic Health Hazard

#### Massachusetts Right To Know Components

	CAS-No.	Revision Date
Heptane	142-82-5	1993-02-16
Pennsylvania Right To Know Components		
	CAS-No.	Revision Date
Heptane	142-82-5	1993-02-16
	CAS-No.	Revision Date
Heptane	142-82-5	1993-02-16
New Jersey Right To Know Components		
	CAS-No.	Revision Date
Heptane	142-82-5	1993-02-16

#### California Prop. 65 Components

This product does not contain any chemicals known to State of California to cause cancer, birth defects, or any other reproductive harm.

#### **16. OTHER INFORMATION**

Full text of H-Statements referred to under sections 2 and 3.

Aquatic Acute	Acute aquatic toxicity
Aquatic Chronic	Chronic aquatic toxicity
Asp. Tox.	Aspiration hazard
Flam. Liq.	Flammable liquids
H225	Highly flammable liquid and vapour.
H304	May be fatal if swallowed and enters airways.
H315	Causes skin irritation.
H336	May cause drowsiness or dizziness.
H400	Very toxic to aquatic life.
H410	Very toxic to aquatic life with long lasting effects.
Skin Irrit.	Skin irritation

#### Further information

Copyright 2016 Sigma-Aldrich Co. LLC. License granted to make unlimited paper copies for internal use only. The above information is believed to be correct but does not purport to be all inclusive and shall be used only as a guide. The information in this document is based on the present state of our knowledge and is applicable to the product with regard to appropriate safety precautions. It does not represent any guarantee of the properties of the product. Sigma-Aldrich Corporation and its Affiliates shall not be held liable for any damage resulting from handling or from contact with the above product. See www.sigma-aldrich.com and/or the reverse side of invoice or packing slip for additional terms and conditions of sale.

**Preparation Information** Sigma-Aldrich Corporation Product Safety – Americas Region 1-800-521-8956

Version: 5.12

Revision Date: 07/26/2018

Print Date: 11/10/2018

# SAFETY DATA SHEET

Version 6.1 Revision Date 05/28/2017 Print Date 06/28/2019

#### 1. PRODUCT AND COMPANY IDENTIFICATION 1.1 **Product identifiers** Product name Manganese Product Number 463728 Brand Aldrich CAS-No. : 7439-96-5 1.2 Relevant identified uses of the substance or mixture and uses advised against Identified uses : Laboratory chemicals, Synthesis of substances 1.3 Details of the supplier of the safety data sheet Company : Sigma-Aldrich Inc. 3050 Spruce Street ST. LOUIS MO 63103 UNITED STATES Telephone +1 314 771-5765 +1 800 325-5052

#### 1.4 **Emergency telephone number**

Emergency Phone # : +1-703-527-3887

:

#### 2. HAZARDS IDENTIFICATION

Fax

#### 2.1 Classification of the substance or mixture

#### GHS Classification in accordance with 29 CFR 1910 (OSHA HCS)

Substances and mixtures, which in contact with water, emit flammable gases (Category 1), H260 Acute aquatic toxicity (Category 3), H402 Chronic aquatic toxicity (Category 3), H412

For the full text of the H-Statements mentioned in this Section, see Section 16.

#### 2.2 GHS Label elements, including precautionary statements

Pictogram



Signal word Danger Hazard statement(s) H260 In contact with water releases flammable gases which may ignite spontaneously. H412 Harmful to aquatic life with long lasting effects. Precautionary statement(s) P223 Keep away from any possible contact with water, because of violent reaction and possible flash fire. Handle under inert gas. Protect from moisture. P231 + P232

P273	Avoid release to the environment.
P280	Wear protective gloves/ protective clothing/ eye protection/ face protection.
P335 + P334	Brush off loose particles from skin. Immerse in cool water/ wrap in wet bandages.
P370 + P378	In case of fire: Use dry sand, dry chemical or alcohol-resistant foam for extinction.
P402 + P404 P501	Store in a dry place. Store in a closed container. Dispose of contents/ container to an approved waste disposal plant.

#### 2.3 Hazards not otherwise classified (HNOC) or not covered by GHS - none

#### **3. COMPOSITION/INFORMATION ON INGREDIENTS**

#### 3.1 Substances

Formula	:	Mn
Molecular weight	:	54.94 g/mol
CAS-No.	:	7439-96-5
EC-No.	:	231-105-1

#### Hazardous components

Component	Classification	Concentration
Manganese		
	Water-react. 1; Aquatic Acute 3; Aquatic Chronic 3; H260, H412	<= 100 %

For the full text of the H-Statements mentioned in this Section, see Section 16.

#### **4. FIRST AID MEASURES**

#### 4.1 Description of first aid measures

#### **General advice**

Consult a physician. Show this safety data sheet to the doctor in attendance. Move out of dangerous area.

#### If inhaled

If breathed in, move person into fresh air. If not breathing, give artificial respiration. Consult a physician.

#### In case of skin contact

Wash off with soap and plenty of water. Consult a physician.

#### In case of eye contact

Rinse thoroughly with plenty of water for at least 15 minutes and consult a physician.

#### If swallowed

Never give anything by mouth to an unconscious person. Rinse mouth with water. Consult a physician.

#### 4.2 Most important symptoms and effects, both acute and delayed

The most important known symptoms and effects are described in the labelling (see section 2.2) and/or in section 11

#### **4.3 Indication of any immediate medical attention and special treatment needed** No data available

#### **5. FIREFIGHTING MEASURES**

#### 5.1 Extinguishing media

**Suitable extinguishing media** Dry powder Carbon dioxide (CO2)

#### Unsuitable extinguishing media Water

# 5.2 Special hazards arising from the substance or mixture Manganese/manganese oxides

## **5.3** Advice for firefighters Wear self-contained breathing apparatus for firefighting if necessary.

5.4 Further information No data available

#### 6. ACCIDENTAL RELEASE MEASURES

#### 6.1 Personal precautions, protective equipment and emergency procedures

Use personal protective equipment. Avoid dust formation. Avoid breathing vapours, mist or gas. Ensure adequate ventilation. Evacuate personnel to safe areas. Avoid breathing dust. For personal protection see section 8.

#### 6.2 Environmental precautions

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

#### 6.3 Methods and materials for containment and cleaning up

Sweep up and shovel. Contain spillage, and then collect with an electrically protected vacuum cleaner or by wetbrushing and place in container for disposal according to local regulations (see section 13). Do not flush with water. Keep in suitable, closed containers for disposal.

#### 6.4 Reference to other sections

For disposal see section 13.

#### 7. HANDLING AND STORAGE

#### 7.1 Precautions for safe handling

Avoid contact with skin and eyes. Avoid formation of dust and aerosols.

Provide appropriate exhaust ventilation at places where dust is formed.Keep away from sources of ignition - No smoking.

For precautions see section 2.2.

## 7.2 Conditions for safe storage, including any incompatibilities

Keep container tightly closed in a dry and well-ventilated place. Never allow product to get in contact with water during storage.

Moisture sensitive. Keep in a dry place.

#### 7.3 Specific end use(s)

Apart from the uses mentioned in section 1.2 no other specific uses are stipulated

## 8. EXPOSURE CONTROLS/PERSONAL PROTECTION

#### 8.1 Control parameters

#### Components with workplace control parameters

Component	CAS-No.	Value	Control parameters	Basis
Manganese	7439-96-5	TWA	0.200000 mg/m3	USA. ACGIH Threshold Limit Values (TLV)
	Remarks	Central Nerv Adopted valu are proposed See Notice o	ous System impair les or notations en l in the NIC f Intended Change	ment closed are those for which changes s (NIC)
		С	5.000000 mg/m3	USA. Occupational Exposure Limits (OSHA) - Table Z-1 Limits for Air Contaminants
		Ceiling limit i	s to be determined	from breathing-zone air samples.
		С	5 mg/m3	USA. Occupational Exposure Limits (OSHA) - Table Z-1 Limits for Air Contaminants
		Ceiling limit i	s to be determined	from breathing-zone air samples.

	TWA	1.000000	USA. NIOSH Recommended
		mg/m3	Exposure Limits
	ST	3.000000	USA. NIOSH Recommended
		mg/m3	Exposure Limits
	TWA	1.000000	USA. NIOSH Recommended
		mg/m3	Exposure Limits
	ST	3.000000	USA. NIOSH Recommended
		mg/m3	Exposure Limits
	С	5.000000	USA. Occupational Exposure Limits
		mg/m3	(OSHA) - Table Z-1 Limits for Air
		Ū.	Contaminants
	Ceiling limit	is to be determine	d from breathing-zone air samples.
	TWA	1.000000	USA. NIOSH Recommended
		mg/m3	Exposure Limits
	ST	3.000000	USA. NIOSH Recommended
		mg/m3	Exposure Limits
	TWA	0.200000	USA. ACGIH Threshold Limit Values
		mg/m3	(TLV)
	Central Ner	vous Svstem impai	rment
	Adopted val	ues or notations er	nclosed are those for which changes
	are propose	d in the NIC	5
See Notice of Intended Changes (NIC)			es (NIC)
	varies	Ũ	
	TWA	0.100000	USA. ACGIH Threshold Limit Values
		mg/m3	(TLV)
	Central Ner	vous System impai	rment
	2015 Adopti	ion	
	varies		
	TWA	0.020000	USA. ACGIH Threshold Limit Values
		mg/m3	(TLV)
	Central Nerv	vous System impai	rment
	2015 Adopti	on	
	varies		
	TWA	1 mg/m3	USA. NIOSH Recommended
			Exposure Limits
	ST	3 mg/m3	USA. NIOSH Recommended
			Exposure Limits
	TWA	0.1 mg/m3	USA. ACGIH Threshold Limit Values
			(TLV)
	Central Nerv	vous System impai	rment
	Not classifia	ible as a human ca	rcinogen
	varies		
	TWA	0.02 mg/m3	USA. ACGIH Threshold Limit Values (TLV)
	Central Nervous System impairment		
	Not classifiable as a human carcinogen		
	varies		-

#### 8.2 Exposure controls

#### Appropriate engineering controls

Handle in accordance with good industrial hygiene and safety practice. Wash hands before breaks and at the end of workday.

#### Personal protective equipment

#### Eye/face protection

Safety glasses with side-shields conforming to EN166 Use equipment for eye protection tested and approved under appropriate government standards such as NIOSH (US) or EN 166(EU).

#### Skin protection

Handle with gloves. Gloves must be inspected prior to use. Use proper glove removal technique (without touching glove's outer surface) to avoid skin contact with this product. Dispose of contaminated gloves after use in accordance with applicable laws and good laboratory practices. Wash and dry hands.

#### **Body Protection**

Impervious clothing, Flame retardant protective clothing, The type of protective equipment must be selected according to the concentration and amount of the dangerous substance at the specific workplace. **Respiratory protection** 

Where risk assessment shows air-purifying respirators are appropriate use (EN 143) respirator cartridges as a backup to engineering controls. If th full-face supplied air respirator. Use respirators and components tested and approved under appropriate government standards such as NIOSH (US) or CEN (EU).

#### Control of environmental exposure

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

#### 9. PHYSICAL AND CHEMICAL PROPERTIES

#### 9.1 Information on basic physical and chemical properties

a)	Appearance	Form: powder Colour: grey
b)	Odour	No data available
c)	Odour Threshold	No data available
d)	рН	No data available
e)	Melting point/freezing point	Melting point/range: 1,244 °C (2,271 °F) - lit.
f)	Initial boiling point and boiling range	1,962 °C (3,564 °F) - lit.
g)	Flash point	()Not applicable
h)	Evaporation rate	No data available
i)	Flammability (solid, gas)	No data available
j)	Upper/lower flammability or explosive limits	No data available
k)	Vapour pressure	No data available
I)	Vapour density	No data available
m)	Relative density	7.3 g/mL at 25 °C (77 °F)
n)	Water solubility	No data available
o)	Partition coefficient: n- octanol/water	No data available
p)	Auto-ignition temperature	No data available
q)	Decomposition temperature	No data available
r)	Viscosity	No data available
s)	Explosive properties	No data available
t)	Oxidizing properties	No data available
<b>Otł</b> No	<b>her safety information</b> data available	

#### **10. STABILITY AND REACTIVITY**

#### 10.1 Reactivity

No data available

9.2

#### 10.2 Chemical stability

Stable under recommended storage conditions.

- **10.3 Possibility of hazardous reactions** Reacts violently with water.
- **10.4** Conditions to avoid Exposure to moisture
- **10.5** Incompatible materials acids, Halogens, Bases, Phosphorus, Sulphur oxides, Peroxides
- Hazardous decomposition products
   Hazardous decomposition products formed under fire conditions. Manganese/manganese oxides
   Other decomposition products No data available
   In the event of fire: see section 5

#### **11. TOXICOLOGICAL INFORMATION**

#### 11.1 Information on toxicological effects

#### Acute toxicity

LD50 Oral - Rat - 9,000 mg/kg(Manganese) Inhalation: No data available(Manganese) Dermal: No data available(Manganese) No data available(Manganese)

#### Skin corrosion/irritation

Skin - Rabbit(Manganese) Result: Mild skin irritation - 24 h

Serious eye damage/eye irritation Eyes - Rabbit(Manganese) Result: Mild eye irritation - 24 h

**Respiratory or skin sensitisation** No data available(Manganese)

Germ cell mutagenicity

No data available(Manganese)

#### Carcinogenicity

IARC: No component of this product present at levels greater than or equal to 0.1% is identified as probable, possible or confirmed human carcinogen by IARC.

No component of this product present at levels greater than or equal to 0.1% is identified as probable, possible or confirmed human carcinogen by IARC.

NTP: No component of this product present at levels greater than or equal to 0.1% is identified as a known or anticipated carcinogen by NTP.

No component of this product present at levels greater than or equal to 0.1% is identified as a known or anticipated carcinogen by NTP.

OSHA: No component of this product present at levels greater than or equal to 0.1% is identified as a carcinogen or potential carcinogen by OSHA.

No component of this product present at levels greater than or equal to 0.1% is identified as a carcinogen or potential carcinogen by OSHA.

#### **Reproductive toxicity**

No data available(Manganese)

May cause reproductive disorders.(Manganese)

#### Specific target organ toxicity - single exposure

No data available(Manganese)

#### Specific target organ toxicity - repeated exposure

No data available

#### Aspiration hazard

No data available(Manganese)

#### **Additional Information**

RTECS: 009275000

Men exposed to manganese dusts showed a decrease in fertility. Chronic man system. Early symptoms include languor, sleepiness and weakness in the le disturbances such as uncontrollable laughter and a spastic gait with tend cases. High incidence of pneumonia has been found in workers exposed to t(Manganese)

Stomach - Irregularities - Based on Human Evidence Stomach - Irregularities - Based on Human Evidence(Manganese)

#### **12. ECOLOGICAL INFORMATION**

#### 12.1 Toxicity

Toxicity to daphnia and EC50 - Daphnia magna (Water flea) - 40 mg/l - 48 h(Manganese) other aquatic invertebrates

- **12.2 Persistence and degradability** No data available
- **12.3 Bioaccumulative potential** No data available
- 12.4 Mobility in soil No data available(Manganese)
- **12.5** Results of PBT and vPvB assessment PBT/vPvB assessment not available as chemical safety assessment not required/not conducted

#### 12.6 Other adverse effects

An environmental hazard cannot be excluded in the event of unprofessional handling or disposal. Harmful to aquatic life.

No data available

#### **13. DISPOSAL CONSIDERATIONS**

#### 13.1 Waste treatment methods

#### Product

Burn in a chemical incinerator equipped with an afterburner and scrubber b highly flammable. Offer surplus and nonrecyclable solutions to a licensed disposal company. Contact a licensed professional waste disposal service to dispose of this material.

#### Contaminated packaging

Dispose of as unused product.

#### **14. TRANSPORT INFORMATION**

#### DOT (US)

UN number: 3208 Class: 4.3 Packing group: I Proper shipping name: Metallic substance, water-reactive, n.o.s. (Manganese) Poison Inhalation Hazard: No

#### IMDG

UN number: 3208 Class: 4.3 Packing group: I EMS-No: F-G, S-N Proper shipping name: METALLIC SUBSTANCE, WATER-REACTIVE, N.O.S. (Manganese)

#### **15. REGULATORY INFORMATION**

#### SARA 302 Components

No chemicals in this material are subject to the reporting requirements of SARA Title III, Section 302.

SARA 313 Components		
	CAS-No.	Revision Date
Manganese	7439-96-5	2007-07-01
SARA 311/312 Hazards Reactivity Hazard, Chronic Health Hazard		
Massachusetts Right To Know Components		
	CAS-No.	Revision Date
Manganese	7439-96-5	2007-07-01
Pennsylvania Right To Know Components		
	CAS-No.	Revision Date
Manganese	7439-96-5	2007-07-01
New Jersey Right To Know Components		
	CAS-No.	Revision Date
Manganese	7439-96-5	2007-07-01

#### California Prop. 65 Components

This product does not contain any chemicals known to State of California to cause cancer, birth defects, or any other reproductive harm.

This product does not contain any chemicals known to State of California to cause cancer, birth defects, or any other reproductive harm.

#### **16. OTHER INFORMATION**

#### Full text of H-Statements referred to under sections 2 and 3.

H260	In contact with water releases flammable gases which may ignite spontaneously.
H402	Harmful to aquatic life.
H412	Harmful to aquatic life with long lasting effects.

#### **HMIS Rating**

Health hazard: Chronic Health Hazard	0 *
Flammability:	3
Physical Hazard	2
NFPA Rating	
Health hazard:	0
Fire Hazard:	0
Reactivity Hazard:	2
Special hazard.I:	W

#### **Further information**

Copyright 2016 Sigma-Aldrich Co. LLC. License granted to make unlimited paper copies for internal use only. The above information is believed to be correct but does not purport to be all inclusive and shall be used only as a guide. The information in this document is based on the present state of our knowledge and is applicable to the product with regard to appropriate safety precautions. It does not represent any guarantee of the properties of the product. Sigma-Aldrich Corporation and its Affiliates shall not be held liable for any damage resulting from handling

or from contact with the above product. See www.sigma-aldrich.com and/or the reverse side of invoice or packing slip for additional terms and conditions of sale.

#### **Preparation Information**

Sigma-Aldrich Corporation Product Safety – Americas Region 1-800-521-8956 Version: 6.1

Revision Date: 05/28/2017

Print Date: 06/28/2019

# SIGMA-ALDRICH

## SAFETY DATA SHEET

Version 5.7 Revision Date 06/08/2018 Print Date 11/10/2018

#### **1. PRODUCT AND COMPANY IDENTIFICATION**

1.1	Product identifiers Product name	:	Hexane
	Product Number Brand Index-No.	:	296090 Sigma-Aldrich 601-037-00-0
	CAS-No.	:	110-54-3

#### 1.2 Relevant identified uses of the substance or mixture and uses advised against

Identified uses : Laboratory chemicals, Synthesis of substances

#### 1.3 Details of the supplier of the safety data sheet

Company	:	Sigma-Aldrich 3050 Spruce Street SAINT LOUIS MO 63103 USA
Telephone Fax	:	+1 800-325-5832 +1 800-325-5052

#### 1.4 Emergency telephone number

Emergency Phone # : +1-703-527-3887 (CHEMTREC)

#### 2. HAZARDS IDENTIFICATION

#### 2.1 Classification of the substance or mixture

#### GHS Classification in accordance with 29 CFR 1910 (OSHA HCS)

Flammable liquids (Category 2), H225 Skin irritation (Category 2), H315 Reproductive toxicity (Category 2), H361 Specific target organ toxicity - single exposure (Category 3), Central nervous system, H336 Specific target organ toxicity - repeated exposure, Oral (Category 2), Nervous system, H373 Aspiration hazard (Category 1), H304 Acute aquatic toxicity (Category 2), H401 Chronic aquatic toxicity (Category 2), H411

For the full text of the H-Statements mentioned in this Section, see Section 16.

Danger

#### 2.2 GHS Label elements, including precautionary statements

Pictogram

Signal word

		<b>(!)</b>	
× .	×	×	×

eignaí Mera	Danger
Hazard statement(s)	
H225	Highly flammable liquid and vapour.
H304	May be fatal if swallowed and enters airways.
H315	Causes skin irritation.
H336	May cause drowsiness or dizziness.
H361	Suspected of damaging fertility or the unborn child.
H373	May cause damage to organs (Nervous system) through prolonged or
	repeated exposure if swallowed.

H411	Toxic to aquatic life with long lasting effects.
Precautionary statement(s)	
P201	Obtain special instructions before use.
P202	Do not handle until all safety precautions have been read and
	understood.
P210	Keep away from heat/sparks/open flames/hot surfaces. No smoking.
P233	Keep container tightly closed.
P240	Ground/bond container and receiving equipment.
P241	Use explosion-proof electrical/ ventilating/ lighting/ equipment.
P242	Use only non-sparking tools.
P243	Take precautionary measures against static discharge.
P260	Do not breathe dust/ fume/ gas/ mist/ vapours/ spray.
P264	Wash skin thoroughly after handling.
P271	Use only outdoors or in a well-ventilated area.
P273	Avoid release to the environment.
P280	Wear protective gloves/ protective clothing/ eye protection/ face
	protection.
P301 + P310	IF SWALLOWED: Immediately call a POISON CENTER/doctor.
P303 + P361 + P353	IF ON SKIN (or hair): Take off immediately all contaminated clothing.
	Rinse skin with water/shower.
P304 + P340 + P312	IF INHALED: Remove person to fresh air and keep comfortable for
	breathing. Call a POISON CENTER/doctor if you feel unwell.
P308 + P313	IF exposed or concerned: Get medical advice/ attention.
P331	Do NOT induce vomiting.
P332 + P313	If skin irritation occurs: Get medical advice/ attention.
P362	Take off contaminated clothing and wash before reuse.
P370 + P378	In case of fire: Use dry sand, dry chemical or alcohol-resistant foam to
	extinguish.
P391	Collect spillage.
P403 + P233	Store in a well-ventilated place. Keep container tightly closed.
P403 + P235	Store in a well-ventilated place. Keep cool.
P405	Store locked up.
P501	Dispose of contents/ container to an approved waste disposal plant.

## 2.3 Hazards not otherwise classified (HNOC) or not covered by GHS - none

## **3. COMPOSITION/INFORMATION ON INGREDIENTS**

#### 3.1 Substances

Synonyms	:	n-Hexane
Formula	:	С <sub>6</sub> Н <sub>14</sub>
Molecular weight	:	86.18 g/mol
CAS-No.	:	110-54-3
EC-No.	:	203-777-6
Index-No.	:	601-037-00-0
Registration number	:	01-2119480412-44-XXXX

#### Hazardous components

Component	Classification	Concentration
n-Hexane		
	Flam. Liq. 2; Skin Irrit. 2; Repr.	90 - 100 %
	2; STOT SE 3; STOT RE 2;	
	Asp. Tox. 1; Aquatic Acute 2;	
	Aquatic Chronic 2; H225,	
	H304, H315, H336, H361f,	
	H373, H411	

For the full text of the H-Statements mentioned in this Section, see Section 16.

#### 4. FIRST AID MEASURES

#### 4.1 Description of first aid measures

#### **General advice**

Consult a physician. Show this safety data sheet to the doctor in attendance. Move out of dangerous area.

#### If inhaled

If breathed in, move person into fresh air. If not breathing, give artificial respiration. Consult a physician.

#### In case of skin contact

Wash off with soap and plenty of water. Consult a physician.

#### In case of eye contact

Rinse thoroughly with plenty of water for at least 15 minutes and consult a physician.

#### If swallowed

Do NOT induce vomiting. Never give anything by mouth to an unconscious person. Rinse mouth with water. Consult a physician.

#### 4.2 Most important symptoms and effects, both acute and delayed

The most important known symptoms and effects are described in the labelling (see section 2.2) and/or in section 11

#### **4.3 Indication of any immediate medical attention and special treatment needed** No data available

#### **5. FIREFIGHTING MEASURES**

#### 5.1 Extinguishing media

### Suitable extinguishing media

Use water spray, alcohol-resistant foam, dry chemical or carbon dioxide.

# 5.2 Special hazards arising from the substance or mixture No data available

#### 5.3 Advice for firefighters

Wear self-contained breathing apparatus for firefighting if necessary.

#### 5.4 Further information

Use water spray to cool unopened containers.

#### 6. ACCIDENTAL RELEASE MEASURES

#### 6.1 Personal precautions, protective equipment and emergency procedures

Use personal protective equipment. Avoid breathing vapours, mist or gas. Ensure adequate ventilation. Remove all sources of ignition. Evacuate personnel to safe areas. Beware of vapours accumulating to form explosive concentrations. Vapours can accumulate in low areas. For personal protection see section 8.

#### 6.2 Environmental precautions

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

#### 6.3 Methods and materials for containment and cleaning up

Contain spillage, and then collect with an electrically protected vacuum cleaner or by wet-brushing and place in container for disposal according to local regulations (see section 13).

#### 6.4 Reference to other sections

For disposal see section 13.

#### 7. HANDLING AND STORAGE

#### 7.1 Precautions for safe handling

Avoid contact with skin and eyes. Avoid inhalation of vapour or mist.

Flash back possible over considerable distance.Container explosion may occur under fire conditions.Use explosion-proof equipment.Keep away from sources of ignition - No smoking.Take measures to prevent the build up of electrostatic charge.

For precautions see section 2.2.

#### 7.2 Conditions for safe storage, including any incompatibilities

Keep container tightly closed in a dry and well-ventilated place. Containers which are opened must be carefully resealed and kept upright to prevent leakage.

Storage class (TRGS 510): 3: Flammable liquids

#### 7.3 Specific end use(s)

Apart from the uses mentioned in section 1.2 no other specific uses are stipulated

#### 8. EXPOSURE CONTROLS/PERSONAL PROTECTION

#### 8.1 Control parameters

#### Components with workplace control parameters

Component	CAS-No.	Value	Control parameters	Basis
n-Hexane	110-54-3	TWA	50 ppm	USA. ACGIH Threshold Limit Values (TLV)
	Remarks	Central Nerv Eye irritation Peripheral ne Substances f (see BEI® se Danger of cu	ous System impair europathy for which there is a ection) taneous absorption	ment Biological Exposure Index or Indices n
		TWA	50 ppm 180 mg/m3	USA. NIOSH Recommended Exposure Limits
		TWA	500 ppm 1,800 mg/m3	USA. Occupational Exposure Limits (OSHA) - Table Z-1 Limits for Air Contaminants
		The value in	mg/m3 is approxin	nate.
		PEL	50 ppm 180 mg/m3	California permissible exposure limits for chemical contaminants (Title 8, Article 107)
		Skin		

#### **Biological occupational exposure limits**

U					
Component	CAS-No.	Parameters	Value	Biological	Basis
				specimen	
	-	2,5- Hexanedione	0.4 mg/l	Urine	ACGIH - Biological Exposure Indices (BEI)
	Remarks	End of shift at o	end of workv	veek	

#### 8.2 Exposure controls

#### Appropriate engineering controls

Handle in accordance with good industrial hygiene and safety practice. Wash hands before breaks and at the end of workday.

#### Personal protective equipment

#### Eye/face protection

Face shield and safety glasses Use equipment for eye protection tested and approved under appropriate government standards such as NIOSH (US) or EN 166(EU).

#### Skin protection

Handle with gloves. Gloves must be inspected prior to use. Use proper glove removal technique (without touching glove's outer surface) to avoid skin contact with this product. Dispose of contaminated gloves after use in accordance with applicable laws and good laboratory practices. Wash and dry hands.

Full contact Material: Nitrile rubber Minimum layer thickness: 0.4 mm Break through time: 480 min Material tested:Camatril® (KCL 730 / Aldrich Z677442, Size M) Splash contact Material: Nitrile rubber Minimum layer thickness: 0.2 mm Break through time: 59 min Material tested:Dermatril® P (KCL 743 / Aldrich Z677388, Size M)

data source: KCL GmbH, D-36124 Eichenzell, phone +49 (0)6659 87300, e-mail sales@kcl.de, test method: EN374

If used in solution, or mixed with other substances, and under conditions which differ from EN 374, contact the supplier of the CE approved gloves. This recommendation is advisory only and must be evaluated by an industrial hygienist and safety officer familiar with the specific situation of anticipated use by our customers. It should not be construed as offering an approval for any specific use scenario.

#### **Body Protection**

Complete suit protecting against chemicals, Flame retardant antistatic protective clothing., The type of protective equipment must be selected according to the concentration and amount of the dangerous substance at the specific workplace.

#### **Respiratory protection**

Where risk assessment shows air-purifying respirators are appropriate use a full-face respirator with multipurpose combination (US) or type AXBEK (EN 14387) respirator cartridges as a backup to engineering controls. If the respirator is the sole means of protection, use a full-face supplied air respirator. Use respirators and components tested and approved under appropriate government standards such as NIOSH (US) or CEN (EU).

#### Control of environmental exposure

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

#### 9. PHYSICAL AND CHEMICAL PROPERTIES

9.1 Information on basic physical and chemical properties

a)	Appearance	Form: liquid Colour: colourless
b)	Odour	No data available
c)	Odour Threshold	No data available
d)	рН	7.0
e)	Melting point/freezing point	Melting point/range: -95 °C (-139 °F)
f)	Initial boiling point and boiling range	69 °C (156 °F)
g)	Flash point	-26.0 °C (-14.8 °F) - closed cup
h)	Evaporation rate	15.8
i)	Flammability (solid, gas)	No data available
j)	Upper/lower flammability or explosive limits	Upper explosion limit: 7.7 %(V) Lower explosion limit: 1.2 %(V)
k)	Vapour pressure	341.3 hPa (256.0 mmHg) at 37.7 °C (99.9 °F) 176.0 hPa (132.0 mmHg) at 20.0 °C (68.0 °F)
I)	Vapour density	No data available
m)	Relative density	0.659 g/mL at 25 °C (77 °F)
n)	Water solubility	insoluble
o)	Partition coefficient: n- octanol/water	log Pow: 3.90 - 4.11
p)	Auto-ignition temperature	234.0 °C (453.2 °F)

- q) Decomposition No data available temperature
- r) Viscosity No data available
- s) Explosive properties No data available
- t) Oxidizing properties No data available

# 9.2 Other safety information No data available

#### **10. STABILITY AND REACTIVITY**

#### 10.1 Reactivity

No data available

- **10.2 Chemical stability** Stable under recommended storage conditions.
- **10.3 Possibility of hazardous reactions** Vapours may form explosive mixture with air.

#### 10.4 Conditions to avoid

Exposure to moisture may affect product quality. Heat, flames and sparks.

**10.5** Incompatible materials Oxidizing agents

#### 10.6 Hazardous decomposition products

Other decomposition products - No data available Hazardous decomposition products formed under fire conditions. - Carbon oxides In the event of fire: see section 5

#### **11. TOXICOLOGICAL INFORMATION**

#### 11.1 Information on toxicological effects

#### Acute toxicity

LD50 Oral - Rat - male and female - 16,000 mg/kg (OECD Test Guideline 401)

LC50 Inhalation - Rat - 4 h - 172 mg/l Remarks: (RTECS)

LD50 Dermal - Rabbit - > 2,000 mg/kg Remarks: (ECHA)

Skin corrosion/irritation Serious eye damage/eye irritation Respiratory or skin sensitisation Germ cell mutagenicity

In vitro mammalian cell gene mutation test Mouse lymphoma test Result: Positive results were obtained in some in vitro tests.

Ames test Salmonella typhimurium Result: negative

Result: negative (National Toxicology Program)

#### Carcinogenicity

IARC: No component of this product present at levels greater than or equal to 0.1% is identified as probable, possible or confirmed human carcinogen by IARC.

- NTP: No component of this product present at levels greater than or equal to 0.1% is identified as a known or anticipated carcinogen by NTP.
- OSHA: No component of this product present at levels greater than or equal to 0.1% is on OSHA's list of regulated carcinogens.

#### **Reproductive toxicity**

Suspected of damaging the unborn child.

Suspected of damaging fertility.

### Specific target organ toxicity - single exposure

May cause drowsiness or dizziness. - Central nervous system

#### Specific target organ toxicity - repeated exposure

Inhalation - May cause damage to organs through prolonged or repeated exposure. - Nervous system

#### Aspiration hazard

Aspiration hazard, Aspiration may cause pulmonary oedema and pneumonitis.

#### **Additional Information**

RTECS: MN9275000

To the best of our knowledge, the chemical, physical, and toxicological properties have not been thoroughly investigated. Drowsiness, irritant effects, somnolence

narcosis, Nausea, Tiredness, CNS disorders, paralysis symptoms

Risk of corneal clouding.

It generally applies for aliphatic hydrocarbons with 6 - 18 carbon atoms that they may cause pneumonia, in some cases also pulmonary oedema, upon direct inhalation, i.e. in conditions that can occur only in very special circumstances (nebulizations, spraying, inhalation of aerosols and similar). After absorption of very large quantities: narcosis.

Testes. - Irregularities - Based on Human Evidence

#### **12. ECOLOGICAL INFORMATION**

#### 12.1 Toxicity

Toxicity to fish	LC50 - Pimephales promelas (fathead minnow) - 2.5 mg/l - 96 h Remarks: (ECOTOX Database)
Toxicity to daphnia and other aquatic invertebrates	EC50 - Daphnia magna (Water flea) - 2.1 mg/l  - 48 h Remarks: (Lit.)

#### 12.2 Persistence and degradability

#### 12.3 Bioaccumulative potential

#### 12.4 Mobility in soil

#### 12.5 Results of PBT and vPvB assessment

PBT/vPvB assessment not available as chemical safety assessment not required/not conducted

#### 12.6 Other adverse effects

An environmental hazard cannot be excluded in the event of unprofessional handling or disposal. Toxic to aquatic life with long lasting effects.

#### **13. DISPOSAL CONSIDERATIONS**

#### 13.1 Waste treatment methods

#### Product

Burn in a chemical incinerator equipped with an afterburner and scrubber but exert extra care in igniting as this material is highly flammable. Offer surplus and non-recyclable solutions to a licensed disposal company. Contact a licensed professional waste disposal service to dispose of this material.

#### Contaminated packaging

Dispose of as unused product.

#### **14. TRANSPORT INFORMATION**

### DOT (US)

UN number: 1208 Class: 3 Proper shipping name: Hexanes Reportable Quantity (RQ): 5000 lbs Poison Inhalation Hazard: No	Packing group: II	
IMDG UN number: 1208 Class: 3 Proper shipping name: HEXANES Marine pollutant:yes	Packing group: II	EMS-No: F-E, S-D
IATA UN number: 1208 Class: 3 Proper shipping name: Hexanes	Packing group: II	

#### **15. REGULATORY INFORMATION**

#### SARA 302 Components

No chemicals in this material are subject to the reporting requirements of SARA Title III, Section 302.

#### SARA 313 Components

The following components are subject to reporting levels established by SARA Title III, Section 313:

	CAS-No.	Revision Date
n-Hexane	110-54-3	2007-07-01
Massachusetts Right To Know Components		
	CAS-No.	Revision Date
n-Hexane	110-54-3	2007-07-01
Pennsylvania Right To Know Components		
	CAS-No.	Revision Date
n-Hexane	110-54-3	2007-07-01
New Jersey Right To Know Components		
	CAS-No.	Revision Date
n-Hexane	110-54-3	2007-07-01

#### California Prop. 65 Components

This product does not contain any chemicals known to State of California to cause cancer, birth defects, or any other reproductive harm.

## **16. OTHER INFORMATION**

Full text of H-Statements referred to under sections 2 and 3.

2

Aquatic Acute	Acute aquatic toxicity
Aquatic Chronic	Chronic aquatic toxicity
Asp. Tox.	Aspiration hazard
Flam. Liq.	Flammable liquids
H225	Highly flammable liquid and vapour.
H304	May be fatal if swallowed and enters airways.
H315	Causes skin irritation.
H336	May cause drowsiness or dizziness.
H361	Suspected of damaging fertility or the unborn child.
H361f	Suspected of damaging fertility.
H373	May cause damage to organs (/\$/*_2ORG_REP_ORA/\$/) through prolonged or
	repeated exposure if swallowed.
H401	Toxic to aquatic life.
H411	Toxic to aquatic life with long lasting effects.
HMIS Rating	

Chronic Health Hazard:	*
Flammability:	3
Physical Hazard	
NFPA Rating	

Health hazard:	2
Fire Hazard:	3
Reactivity Hazard:	0

#### Further information

Copyright 2016 Sigma-Aldrich Co. LLC. License granted to make unlimited paper copies for internal use only. The above information is believed to be correct but does not purport to be all inclusive and shall be used only as a guide. The information in this document is based on the present state of our knowledge and is applicable to the product with regard to appropriate safety precautions. It does not represent any guarantee of the properties of the product. Sigma-Aldrich Corporation and its Affiliates shall not be held liable for any damage resulting from handling or from contact with the above product. See www.sigma-aldrich.com and/or the reverse side of invoice or packing slip for additional terms and conditions of sale.

#### **Preparation Information**

Sigma-Aldrich Corporation Product Safety – Americas Region 1-800-521-8956

Version: 5.7

Revision Date: 06/08/2018

Print Date: 11/10/2018

# SIGMA-ALDRICH

## SAFETY DATA SHEET

Version 4.11 Revision Date 10/12/2018 Print Date 06/28/2019

1. F	RODUCT AND COMPAN	Y IDENT	TFICATION	
1.1	<b>Product identifiers</b> Product name	:	Lead	
	Product Number Brand	:	391352 Aldrich	
	CAS-No.	:	7439-92-1	
1.2	Relevant identified uses of the substance or mixture and uses advised against			
	Identified uses	:	Laboratory chemicals, Synthesis of substances	
1.3	Details of the supplier of	of the sa	fety data sheet	
	Company	:	Sigma-Aldrich 3050 Spruce Street SAINT LOUIS MO 63103	

## 1.4 Emergency telephone number

Emergency Phone # : +1-703-527-38	87 (CHEMTREC)
-----------------------------------	---------------

:

USA

+1 800-325-5832

+1 800-325-5052

#### 2. HAZARDS IDENTIFICATION

Telephone

Fax

#### 2.1 Classification of the substance or mixture

#### GHS Classification in accordance with 29 CFR 1910 (OSHA HCS)

Acute toxicity, Oral (Category 4), H302 Carcinogenicity (Category 2), H351 Reproductive toxicity (Category 2), H361 Specific target organ toxicity - repeated exposure (Category 2), H373 Acute aquatic toxicity (Category 1), H400 Chronic aquatic toxicity (Category 1), H410

For the full text of the H-Statements mentioned in this Section, see Section 16.

#### 2.2 GHS Label elements, including precautionary statements

Pictogram

Signal word



Warning

Hazard statement(s)	
H302	Harmful if swallowed.
H351	Suspected of causing cancer.
H361	Suspected of damaging fertility or the unborn child.
H373	May cause damage to organs through prolonged or repeated exposure.
H410	Very toxic to aquatic life with long lasting effects.
Precautionary statement(s)	
P201	Obtain special instructions before use.
P202	Do not handle until all safety precautions have been read and understood.

P260	Do not breathe dust/ fume/ gas/ mist/ vapours/ spray.
P264	Wash skin thoroughly after handling.
P270	Do not eat, drink or smoke when using this product.
P273	Avoid release to the environment.
P280	Wear protective gloves/ protective clothing/ eye protection/ face protection.
P301 + P312 + P330	IF SWALLOWED: Call a POISON CENTER/doctor if you feel unwell. Rinse mouth.
P308 + P313	IF exposed or concerned: Get medical advice/ attention.
P391	Collect spillage.
P405	Store locked up.
P501	Dispose of contents/ container to an approved waste disposal plant.

#### 2.3 Hazards not otherwise classified (HNOC) or not covered by GHS - none

#### **3. COMPOSITION/INFORMATION ON INGREDIENTS**

#### 3.1 Substances

Formula	:	Pb
Molecular weight	:	207.20 g/mol
CAS-No.	:	7439-92-1
EC-No.	:	231-100-4

#### Hazardous components

Component	Classification	Concentration
Lead		
	Acute Tox. 4; Carc. 2; STOT 9 RE 1; Aquatic Acute 1; Aquatic Chronic 1; H302, H351, H372,	90 - 100 %
	11410	

For the full text of the H-Statements mentioned in this Section, see Section 16.

#### 4. FIRST AID MEASURES

#### 4.1 Description of first aid measures

#### **General advice**

Consult a physician. Show this safety data sheet to the doctor in attendance. Move out of dangerous area.

#### If inhaled

If breathed in, move person into fresh air. If not breathing, give artificial respiration. Consult a physician.

#### In case of skin contact

Wash off with soap and plenty of water. Consult a physician.

#### In case of eye contact

Flush eyes with water as a precaution.

#### If swallowed

Never give anything by mouth to an unconscious person. Rinse mouth with water. Consult a physician.

#### 4.2 Most important symptoms and effects, both acute and delayed

The most important known symptoms and effects are described in the labelling (see section 2.2) and/or in section 11

#### **4.3 Indication of any immediate medical attention and special treatment needed** No data available

#### **5. FIREFIGHTING MEASURES**

#### 5.1 Extinguishing media

#### Suitable extinguishing media

Use water spray, alcohol-resistant foam, dry chemical or carbon dioxide.

# 5.2 Special hazards arising from the substance or mixture No data available

#### **5.3** Advice for firefighters Wear self-contained breathing apparatus for firefighting if necessary.

5.4 Further information No data available

#### 6. ACCIDENTAL RELEASE MEASURES

#### 6.1 Personal precautions, protective equipment and emergency procedures

Use personal protective equipment. Avoid dust formation. Avoid breathing vapours, mist or gas. Ensure adequate ventilation. Evacuate personnel to safe areas. Avoid breathing dust. For personal protection see section 8.

#### 6.2 Environmental precautions

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

#### 6.3 Methods and materials for containment and cleaning up

Pick up and arrange disposal without creating dust. Sweep up and shovel. Keep in suitable, closed containers for disposal.

#### 6.4 Reference to other sections

For disposal see section 13.

#### 7. HANDLING AND STORAGE

#### 7.1 Precautions for safe handling

Avoid contact with skin and eyes. Avoid formation of dust and aerosols. Further processing of solid materials may result in the formation of combustible dusts. The potential for combustible dust formation should be taken into consideration before additional processing occurs.

Provide appropriate exhaust ventilation at places where dust is formed. For precautions see section 2.2.

#### 7.2 Conditions for safe storage, including any incompatibilities

Keep container tightly closed in a dry and well-ventilated place.

Keep in a dry place.

Storage class (TRGS 510): 6.1D: Non-combustible, acute toxic Cat.3 / toxic hazardous materials or hazardous materials causing chronic effects

#### 7.3 Specific end use(s)

Apart from the uses mentioned in section 1.2 no other specific uses are stipulated

#### 8. EXPOSURE CONTROLS/PERSONAL PROTECTION

#### 8.1 Control parameters

#### Components with workplace control parameters

Component	CAS-No.	Value	Control	Basis
			parameters	
	Remarks	See 1910.10	25	
Lead	7439-92-1	TWA	0.05 mg/m3	USA. ACGIH Threshold Limit Values
			0	(TLV)
		Confirmed an	nimal carcinogen w	vith unknown relevance to humans
		TWA	0.05 mg/m3	USA. ACGIH Threshold Limit Values
				(TLV)
		Central Nerv	ous System impair	ment
		Hematologic	effects	
		Peripheral Nervous System impairment		
		Substances for which there is a Biological Exposure Index or Indices		
		(see BEI® section)		
		Confirmed animal carcinogen with unknown relevance to humans		

	TWA	0.05 mg/m3	USA. NIOSH Recommended Exposure Limits
	See Appendi	x C	

#### **Biological occupational exposure limits**

Component	CAS-No.	Parameters	Value	Biological specimen	Basis
	-	Lead	200 µg/l	In blood	ACGIH - Biological Exposure Indices (BEI)
	Remarks	Not critical			

#### 8.2 Exposure controls

#### Appropriate engineering controls

Handle in accordance with good industrial hygiene and safety practice. Wash hands before breaks and at the end of workday.

#### Personal protective equipment

#### Eye/face protection

Safety glasses with side-shields conforming to EN166 Use equipment for eye protection tested and approved under appropriate government standards such as NIOSH (US) or EN 166(EU).

#### Skin protection

Handle with gloves. Gloves must be inspected prior to use. Use proper glove removal technique (without touching glove's outer surface) to avoid skin contact with this product. Dispose of contaminated gloves after use in accordance with applicable laws and good laboratory practices. Wash and dry hands.

Full contact Material: Nitrile rubber Minimum layer thickness: 0.11 mm Break through time: 480 min Material tested:Dermatril® (KCL 740 / Aldrich Z677272, Size M)

Splash contact Material: Nitrile rubber Minimum layer thickness: 0.11 mm Break through time: 480 min Material tested:Dermatril® (KCL 740 / Aldrich Z677272, Size M)

data source: KCL GmbH, D-36124 Eichenzell, phone +49 (0)6659 87300, e-mail sales@kcl.de, test method: EN374

If used in solution, or mixed with other substances, and under conditions which differ from EN 374, contact the supplier of the CE approved gloves. This recommendation is advisory only and must be evaluated by an industrial hygienist and safety officer familiar with the specific situation of anticipated use by our customers. It should not be construed as offering an approval for any specific use scenario.

#### **Body Protection**

Complete suit protecting against chemicals, The type of protective equipment must be selected according to the concentration and amount of the dangerous substance at the specific workplace.

#### **Respiratory protection**

Where risk assessment shows air-purifying respirators are appropriate use a full-face particle respirator type N100 (US) or type P3 (EN 143) respirator cartridges as a backup to engineering controls. If the respirator is the sole means of protection, use a full-face supplied air respirator. Use respirators and components tested and approved under appropriate government standards such as NIOSH (US) or CEN (EU).

#### Control of environmental exposure

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

### 9. PHYSICAL AND CHEMICAL PROPERTIES

- 9.1 Information on basic physical and chemical properties
  - a) Appearance Form: powder
  - b) Odour No data available

c)	Odour Threshold	No data available
d)	рН	No data available
e)	Melting point/freezing point	Melting point/range: 327.4 °C (621.3 °F) - lit.
f)	Initial boiling point and boiling range	1,740 °C (3,164 °F) - lit.
g)	Flash point	Not applicable
h)	Evaporation rate	No data available
i)	Flammability (solid, gas)	No data available
j)	Upper/lower flammability or explosive limits	No data available
k)	Vapour pressure	No data available
I)	Vapour density	No data available
m)	Relative density	No data available
n)	Water solubility	No data available
o)	Partition coefficient: n- octanol/water	No data available
p)	Auto-ignition temperature	No data available
q)	Decomposition temperature	No data available
r)	Viscosity	No data available
s)	Explosive properties	No data available
t)	Oxidizing properties	No data available
<b>Othe</b> No da	r <b>safety information</b> ata available	

#### **10. STABILITY AND REACTIVITY**

**10.1 Reactivity** No data available

9.2

- **10.2 Chemical stability** Stable under recommended storage conditions.
- **10.3 Possibility of hazardous reactions** No data available
- **10.4 Conditions to avoid** No data available
- **10.5** Incompatible materials Strong acids
- Hazardous decomposition products
   Hazardous decomposition products formed under fire conditions. Lead oxides
   Other decomposition products No data available
   In the event of fire: see section 5

## **11. TOXICOLOGICAL INFORMATION**

#### 11.1 Information on toxicological effects

Acute toxicity No data available Inhalation: No data available

Dermal: No data available

No data available

Skin corrosion/irritation No data available

Serious eye damage/eye irritation No data available

**Respiratory or skin sensitisation** No data available

#### Germ cell mutagenicity

Rat Cytogenetic analysis

#### Carcinogenicity

Limited evidence of carcinogenicity in animal studies

- IARC: 2B Group 2B: Possibly carcinogenic to humans (Lead)
- NTP: RAHC Reasonably anticipated to be a human carcinogenThe reference note has been added by TD based on the background information of the NTP. (Lead)
- OSHA: OSHA specifically regulated carcinogen (Lead)

#### **Reproductive toxicity**

Reproductive toxicity - Rat - Inhalation Effects on Newborn: Biochemical and metabolic.

Reproductive toxicity - Rat - Oral Effects on Newborn: Behavioral.

Reproductive toxicity - Mouse - Oral Effects on Fertility: Female fertility index (e.g., # females pregnant per females mated ). Effects on Fertility: Preimplantation mortality (e.g., reduction in numbe corpora lutea).

May damage fertility. May damage the unborn child.

Developmental Toxicity - Rat - Inhalation Effects on Embryo or Fetus: Fetotoxicity (except death, e.g., stunted fetus). Specific Developmental Abnormalities: Blood and lymphatic system (including spleen and marrow).

Developmental Toxicity - Rat - Oral Specific Developmental Abnormalities: Blood and lymphatic system (including spleen and marrow). Effects on Newborn: Growth statistics (e.g., reduced weight gain).

Developmental Toxicity - Rat - Oral Effects on Embryo or Fetus: Fetotoxicity (except death, e.g., stunted fetus). Effects on Embryo or Fetus: Fetal death.

Developmental Toxicity - Mouse - Oral Effects on Embryo or Fetus: Fetotoxicity (except death, e.g., stunted fetus). Effects on Embryo or Fetus: Fetal death.

# Specific target organ toxicity - single exposure

No data available

Specific target organ toxicity - repeated exposure Causes damage to organs through prolonged or repeated exposure.

Aspiration hazard No data available

Additional Information RTECS: OF7525000

anemia

Stomach - Irregularities - Based on Human Evidence

#### **12. ECOLOGICAL INFORMATION**

12.1 Toxicity

	Toxicity to fish	mortality LOEC - Oncorhynchus mykiss (rainbow trout) - 1.19 mg/l - 96.0 h
		LC50 - Micropterus dolomieui - 2.2 mg/l - 96.0 h
		mortality NOEC - Salvelinus fontinalis - 1.7 mg/l - 10.0 d
	Toxicity to daphnia and other aquatic invertebrates	mortality LOEC - Daphnia (water flea) - 0.17 mg/l  - 24 h
		mortality NOEC - Daphnia (water flea) - 0.099 mg/l  - 24 h
	Toxicity to algae	mortality EC50 - Skeletonema costatum - 7.94 mg/l - 10 d
12.2	Persistence and degradat No data available	bility
12.3	Bioaccumulative potential Bioaccumulation	Oncorhynchus kisutch - 2 Weeks

- 150 µg/l

Bioconcentration factor (BCF): 12

#### 12.4 Mobility in soil

No data available

#### Results of PBT and vPvB assessment 12.5

PBT/vPvB assessment not available as chemical safety assessment not required/not conducted

#### 12.6 Other adverse effects

An environmental hazard cannot be excluded in the event of unprofessional handling or disposal. Very toxic to aquatic life with long lasting effects.

### **13. DISPOSAL CONSIDERATIONS**

#### 13.1 Waste treatment methods

#### Product

Offer surplus and non-recyclable solutions to a licensed disposal company. Contact a licensed professional waste disposal service to dispose of this material. Dissolve or mix the material with a combustible solvent and burn in a chemical incinerator equipped with an afterburner and scrubber.

#### Contaminated packaging

Dispose of as unused product.

#### 14. TRANSPORT INFORMATION

#### DOT (US)

UN number: 3077 Class: 9 Packing group: III Proper shipping name: Environmentally hazardous substance, solid, n.o.s. (Lead) Reportable Quantity (RQ): 10 lbs Poison Inhalation Hazard: No

#### IMDG

UN number: 3077 Class: 9 Packing group: III EMS-No: F-A, S-F Proper shipping name: ENVIRONMENTALLY HAZARDOUS SUBSTANCE, SOLID, N.O.S. (Lead) Marine pollutant:yes

#### ΙΑΤΑ

UN number: 3077 Class: 9 Packing group: III Proper shipping name: Environmentally hazardous substance, solid, n.o.s. (Lead)

#### **Further information**

EHS-Mark required (ADR 2.2.9.1.10, IMDG code 2.10.3) for single packagings and combination packagings containing inner packagings with Dangerous Goods > 5L for liquids or > 5kg for solids.

SARA 302 Components No chemicals in this material are subject to the reporting requirem	ents of SARA Title	e III, Section 302.
SARA 313 Components The following components are subject to reporting levels establish Lead	ed by SARA Title CAS-No. 7439-92-1	III, Section 313: Revision Date 2015-11-23
SARA 311/312 Hazards Acute Health Hazard, Chronic Health Hazard		
Massachusetts Right To Know Components		
Lead	CAS-No. 7439-92-1	Revision Date 2015-11-23
Pennsylvania Right To Know Components		
	CAS-No.	<b>Revision Date</b>
Lead	7439-92-1	2015-11-23
l ead	CAS-No. 7/30-02-1	Revision Date
	7409-92-1	2013-11-23
New Jersey Right To Know Components		Povision Data
Lead	7439-92-1	2015-11-23
California Prop. 65 Components WARNING! This product contains a chemical known to the State of California to cause cancer. Lead	CAS-No. 7439-92-1	Revision Date 2009-02-01
WARNING: This product contains a chemical known to the State of California to cause birth defects or other reproductive harm. Lead	CAS-No. 7439-92-1	Revision Date 2009-02-01

#### **16. OTHER INFORMATION**

**15. REGULATORY INFORMATION** 

Full text of H-Statements referred to under sections 2 and 3.

Acute toxicity
Acute aquatic toxicity
Chronic aquatic toxicity
Carcinogenicity
Harmful if swallowed.
Suspected of causing cancer.
Suspected of damaging fertility or the unborn child.
Causes damage to organs through prolonged or repeated exposure.
May cause damage to organs through prolonged or repeated exposure.

#### Further information

Copyright 2016 Sigma-Aldrich Co. LLC. License granted to make unlimited paper copies for internal use only. The above information is believed to be correct but does not purport to be all inclusive and shall be used only as a guide. The information in this document is based on the present state of our knowledge and is applicable to the product with regard to appropriate safety precautions. It does not represent any guarantee of the properties of the product. Sigma-Aldrich Corporation and its Affiliates shall not be held liable for any damage resulting from handling or from contact with the above product. See www.sigma-aldrich.com and/or the reverse side of invoice or packing slip for additional terms and conditions of sale.

**Preparation Information** Sigma-Aldrich Corporation Product Safety – Americas Region 1-800-521-8956

Version: 4.11

Revision Date: 10/12/2018

Print Date: 06/28/2019

SAFETY DATA SHEET

Version 6.0 Revision Date 03/14/2018 Print Date 01/21/2019

#### **1. PRODUCT AND COMPANY IDENTIFICATION**

1.1	<b>Product identifiers</b> Product name	:	Methyl Ethyl Ketone, United States Pharmacopeia (USP) Reference Standard
	Product Number Brand	:	1430101 US Pharmacopeia
	CAS-No.	:	78-93-3
1.2	2 Relevant identified uses of the substance or mixture and uses advised against		
	Identified uses	:	Laboratory chemicals, Synthesis of substances
1.3	Details of the supplier of t	he s	safety data sheet

# 1.3 Details of the supplier of the safety data sheet

5052
5765
10 63103 ATES
n inc. Street

#### 1.4 Emergency telephone number

Emergency Phone # : (314) 776-6555

## 2. HAZARDS IDENTIFICATION

#### 2.1 Classification of the substance or mixture

GHS Classification in accordance with 29 CFR 1910 (OSHA HCS) Flammable liquids (Category 2), H225

Eye irritation (Category 2A), H319

Specific target organ toxicity - single exposure (Category 3), Central nervous system, H336

For the full text of the H-Statements mentioned in this Section, see Section 16.

#### 2.2 GHS Label elements, including precautionary statements

Pictogram

0:----



Signal word	Danger
Hazard statement(s)	
H225	Highly flammable liquid and vapour.
H319	Causes serious eye irritation.
H336	May cause drowsiness or dizziness.

Precautionary statement(s)	
P210	Keep away from heat/sparks/open flames/hot surfaces. No smoking.
P233	Keep container tightly closed.
P240	Ground/bond container and receiving equipment.
P241	Use explosion-proof electrical/ ventilating/ lighting/ equipment.
P242	Use only non-sparking tools.
P243	Take precautionary measures against static discharge.
P261	Avoid breathing dust/ fume/ gas/ mist/ vapours/ spray.
P264	Wash skin thoroughly after handling.
P271	Use only outdoors or in a well-ventilated area.
P280	Wear protective gloves/ eye protection/ face protection.
P303 + P361 + P353	IF ON SKIN (or hair): Take off immediately all contaminated clothing.
	Rinse skin with water/shower.
P304 + P340 + P312	IF INHALED: Remove person to fresh air and keep comfortable for
	breathing. Call a POISON CENTER/doctor if you feel unwell.
P305 + P351 + P338	IF IN EYES: Rinse cautiously with water for several minutes. Remove
	contact lenses, if present and easy to do. Continue rinsing.
P337 + P313	If eye irritation persists: Get medical advice/ attention.
P370 + P378	In case of fire: Use dry sand, dry chemical or alcohol-resistant foam to
	extinguish.
P403 + P233	Store in a well-ventilated place. Keep container tightly closed.
P403 + P235	Store in a well-ventilated place. Keep cool.
P405	Store locked up.
P501	Dispose of contents/ container to an approved waste disposal plant.

#### 2.3 Hazards not otherwise classified (HNOC) or not covered by GHS - none

#### 3. COMPOSITION/INFORMATION ON INGREDIENTS

#### 3.1 Substances

Molecular weight	:	72.11 g/mol
CAS-No.	:	78-93-3

Hazardous components		
Component	Classification	Concentration
Ethyl methyl ketone		
	Flam. Liq. 2; Eye Irrit. 2A; STOT SE 3; H225, H319, H336	<= 100 %

For the full text of the H-Statements mentioned in this Section, see Section 16.

#### 4. FIRST AID MEASURES

#### 4.1 Description of first aid measures

#### **General advice**

Consult a physician. Show this safety data sheet to the doctor in attendance. Move out of dangerous area.

#### If inhaled

If breathed in, move person into fresh air. If not breathing, give artificial respiration. Consult a physician.

#### In case of skin contact

Wash off with soap and plenty of water. Consult a physician.

#### In case of eye contact

Rinse thoroughly with plenty of water for at least 15 minutes and consult a physician.

#### If swallowed

Do NOT induce vomiting. Never give anything by mouth to an unconscious person. Rinse mouth with water. Consult a physician.
#### 4.2 Most important symptoms and effects, both acute and delayed The most important known symptoms and effects are described in the labelling (see section 2.2) and/or in section 11

#### Indication of any immediate medical attention and special treatment needed 4.3 No data available

#### 5. FIREFIGHTING MEASURES

#### 5.1 Extinguishing media

Suitable extinguishing media Dry powder Dry sand

Unsuitable extinguishing media Do NOT use water iet.

- 5.2 Special hazards arising from the substance or mixture Carbon oxides
- 5.3 Advice for firefighters Wear self-contained breathing apparatus for firefighting if necessary.

#### 5.4 **Further information**

Use water spray to cool unopened containers.

#### 6. ACCIDENTAL RELEASE MEASURES

#### 6.1 Personal precautions, protective equipment and emergency procedures

Use personal protective equipment. Avoid breathing vapours, mist or gas. Ensure adequate ventilation. Remove all sources of ignition. Evacuate personnel to safe areas. Beware of vapours accumulating to form explosive concentrations. Vapours can accumulate in low areas. For personal protection see section 8.

#### 6.2 **Environmental precautions**

Prevent further leakage or spillage if safe to do so. Do not let product enter drains.

- 6.3 Methods and materials for containment and cleaning up Contain spillage, and then collect with non-combustible absorbent material. (e.g. sand, earth, diatomaceous earth, vermiculite) and place in container for disposal according to local / national regulations (see section 13).
- **Reference to other sections** 6.4 For disposal see section 13.

#### 7. HANDLING AND STORAGE

#### 7.1 Precautions for safe handling

Avoid contact with skin and eyes. Avoid inhalation of vapour or mist. Use explosion-proof equipment.Keep away from sources of ignition - No smoking.Take measures to prevent the build up of electrostatic charge. For precautions see section 2.2.

#### 7.2 Conditions for safe storage, including any incompatibilities

Keep container tightly closed in a dry and well-ventilated place. Containers which are opened must be carefully resealed and kept upright to prevent leakage. Storage class (TRGS 510): 3: Flammable liquids

#### 7.3 Specific end use(s)

Apart from the uses mentioned in section 1.2 no other specific uses are stipulated

#### 8. EXPOSURE CONTROLS/PERSONAL PROTECTION

#### 8.1 **Control parameters**

- Components with workplace control parameters
- **Exposure controls** 8.2

#### Appropriate engineering controls

Handle in accordance with good industrial hygiene and safety practice. Wash hands before breaks and at the end of workday.

#### Personal protective equipment

#### Eye/face protection

Face shield and safety glasses Use equipment for eye protection tested and approved under appropriate government standards such as NIOSH (US) or EN 166(EU).

#### Skin protection

Handle with gloves. Gloves must be inspected prior to use. Use proper glove removal technique (without touching glove's outer surface) to avoid skin contact with this product. Dispose of contaminated gloves after use in accordance with applicable laws and good laboratory practices. Wash and dry hands.

#### **Body Protection**

Impervious clothing, Flame retardant antistatic protective clothing., The type of protective equipment must be selected according to the concentration and amount of the dangerous substance at the specific workplace. Respiratory protection

Where risk assessment shows air-purifying respirators are appropriate use a full-face respirator with multipurpose combination (US) or type ABEK (EN 14387) respirator cartridges as a backup to engineering controls. If the respirator is the sole means of protection, use a full-face supplied air respirator. Use respirators and components tested and approved under appropriate government standards such as NIOSH (US) or CEN (EU). Control of environmental exposure

Prevent further leakage or spillage if safe to do so. Do not let product enter drains.

#### 9. PHYSICAL AND CHEMICAL PROPERTIES

#### 9.1 Information on basic physical and chemical properties

a)	Appearance	Form: liquid, clear Colour: colourless
b)	Odour	No data available
c)	Odour Threshold	No data available
d)	рН	No data available
e)	Melting point/freezing point	-87 °C (-125 °F)
f)	Initial boiling point and boiling range	79 - 80 °C (174 - 176 °F)
g)	Flash point	-3 °C (27 °F) - closed cup
h)	Evaporation rate	No data available
i)	Flammability (solid, gas)	No data available
j)	Upper/lower flammability or explosive limits	Upper explosion limit: 10.1 %(V) Lower explosion limit: 1.8 %(V)
k)	Vapour pressure	95 hPa at 20 °C (68 °F)
I)	Vapour density	2.49 - (Air = 1.0)
m)	Relative density	0.805 g/cm3
n)	Water solubility	soluble
o)	Partition coefficient: n- octanol/water	log Pow: 0.29
p)	Auto-ignition temperature	No data available
q)	Decomposition temperature	No data available
r)	Viscosity	No data available
s)	Explosive properties	No data available
t)	Oxidizing properties	No data available

#### 9.2 Other safety information

Surface tension 24.6 mN/m at 20 °C (68 °F)

Relative vapour density 2.49 - (Air = 1.0)

#### **10. STABILITY AND REACTIVITY**

#### 10.1 Reactivity

No data available

**10.2 Chemical stability** Stable under recommended storage conditions.

#### **10.3 Possibility of hazardous reactions** Vapours may form explosive mixture with air.

- **10.4** Conditions to avoid Heat, flames and sparks.
- **10.5** Incompatible materials Oxidizing agents, Strong reducing agents

#### 10.6 Hazardous decomposition products Hazardous decomposition products formed under fire conditions. - Carbon oxides In the event of fire: see section 5

#### **11. TOXICOLOGICAL INFORMATION**

#### 11.1 Information on toxicological effects

#### Acute toxicity

LD50 Oral - Rat - 2,737 mg/kg LC50 Inhalation - Mouse - 4 h - 32,000 mg/m3 LC50 Inhalation - Mammal - 38,000 mg/m3 LD50 Dermal - Rabbit - 6,480 mg/kg No data available

#### Skin corrosion/irritation

Skin - Rabbit Result: No skin irritation (OECD Test Guideline 404)

#### Serious eye damage/eye irritation

Eyes - Rabbit Result: Irritating to eyes. (OECD Test Guideline 405)

**Respiratory or skin sensitisation** No data available

#### Germ cell mutagenicity

No data available

#### Carcinogenicity

- IARC: No component of this product present at levels greater than or equal to 0.1% is identified as probable, possible or confirmed human carcinogen by IARC.
- NTP: No component of this product present at levels greater than or equal to 0.1% is identified as a known or anticipated carcinogen by NTP.
- OSHA: No component of this product present at levels greater than or equal to 0.1% is identified as a carcinogen or potential carcinogen by OSHA.

#### **Reproductive toxicity**

No data available No data available **Specific target organ toxicity - single exposure** May cause drowsiness or dizziness.

Specific target organ toxicity - repeated exposure No data available

Aspiration hazard No data available

#### **Additional Information**

RTECS: Not available

Central nervous system depression, Gastrointestinal disturbance, narcosis

Liver - Irregularities - Based on Human Evidence Liver - Irregularities - Based on Human Evidence

#### **12. ECOLOGICAL INFORMATION**

#### 12.1 Toxicity

Toxicity to fish	mortality NOEC - Cyprinodon variegatus (sheepshead minnow) - 400 mg/l - 96 h(Ethyl methyl ketone)
	LC50 - Pimephales promelas (fathead minnow) - 3,130 - 3,320 mg/l  - 96 h(Ethyl methyl ketone)
Toxicity to daphnia and other aquatic invertebrates	LC50 - Daphnia magna (Water flea) - > 520 mg/l  - 48 h(Ethyl methyl ketone)
	EC50 - Daphnia magna (Water flea) - 7,060 mg/l  - 24 h(Ethyl methyl ketone)

#### 12.2 Persistence and degradability No data available

**12.3 Bioaccumulative potential** No data available

#### 12.4 Mobility in soil No data available(Ethyl methyl ketone)

**12.5 Results of PBT and vPvB assessment** PBT/vPvB assessment not available as chemical safety assessment not required/not conducted

#### 12.6 Other adverse effects

No data available

#### **13. DISPOSAL CONSIDERATIONS**

#### 13.1 Waste treatment methods

#### Product

Contact a licensed professional waste disposal service to dispose of this material. Offer surplus and non-recyclable solutions to a licensed disposal company. Burn in a chemical incinerator equipped with an afterburner and scrubber but exert extra care in igniting as this material is highly flammable.

#### Contaminated packaging

Dispose of as unused product.

#### 14. TRANSPORT INFORMATION

#### DOT (US)

UN number: 1193 Class: 3 Packing group: II Proper shipping name: Ethyl methyl ketone Reportable Quantity (RQ) : 5000 lbs Poison Inhalation Hazard: No

#### IMDG

UN number: 1193 Class: 3 Packing group: II Proper shipping name: ETHYL METHYL KETONE

#### EMS-No: F-E, S-D

#### ΙΑΤΑ

UN number: 1193 Class: 3 F Proper shipping name: Ethyl methyl ketone

Packing group: II

#### **15. REGULATORY INFORMATION**

#### SARA 302 Components

No chemicals in this material are subject to the reporting requirements of SARA Title III, Section 302.

#### SARA 313 Components

This material does not contain any chemical components with known CAS numbers that exceed the threshold (De Minimis) reporting levels established by SARA Title III, Section 313.

#### SARA 311/312 Hazards

Fire Hazard, Acute Health Hazard, Chronic Health Hazard

#### Massachusetts Right To Know Components

Ethyl methyl ketone	CAS-No. 78-93-3	Revision Date
Pennsylvania Right To Know Components		
	CAS-No.	Revision Date
Ethyl methyl ketone	78-93-3	
	CAS-No.	Revision Date
Ethyl methyl ketone	78-93-3	
New Jersey Right To Know Components		
	CAS-No.	Revision Date
Ethyl methyl ketone	78-93-3	

#### California Prop. 65 Components

This product does not contain any chemicals known to State of California to cause cancer, birth defects, or any other reproductive harm.

#### **16. OTHER INFORMATION**

#### Full text of H-Statements referred to under sections 2 and 3.

H225 H H319 ( H336 N	Highly flammable liquid and vapour. Causes serious eye irritation. May cause drowsiness or dizziness.
HMIS Rating	
Health hazard:	2
Chronic Health Hazar	rd: *
	0

Flammability:	3
Physical Hazard	0
NFPA Rating	
Health hazard:	2
Fire Hazard:	3
Reactivity Hazard:	0

#### **Further information**

Copyright 2016 Sigma-Aldrich Co. LLC. License granted to make unlimited paper copies for internal use only. The above information is believed to be correct but does not purport to be all inclusive and shall be used only as a guide. The information in this document is based on the present state of our knowledge and is applicable to the product with regard to appropriate safety precautions. It does not represent any guarantee of the properties of the product. Sigma-Aldrich Corporation and its Affiliates shall not be held liable for any damage resulting from handling or from contact with the above product. See www.sigma-aldrich.com and/or the reverse side of invoice or packing slip for additional terms and conditions of sale.

#### **Preparation Information**

Sigma-Aldrich Corporation Product Safety – Americas Region 1-800-521-8956 Version: 6.0 Re

Revision Date: 03/14/2018

Print Date: 01/21/2019

# SIGMA-ALDRICH

# SAFETY DATA SHEET

Version 3.15 Revision Date 03/05/2018 Print Date 06/28/2019

#### **1. PRODUCT AND COMPANY IDENTIFICATION**

1.1	Product identifiers Product name	:	Mercury
	Product Number Brand Index-No.	:	215457 Sigma-Aldrich 080-001-00-0
	CAS-No.	:	7439-97-6

#### 1.2 Relevant identified uses of the substance or mixture and uses advised against

Identified uses : Laboratory chemicals, Synthesis of substances

#### 1.3 Details of the supplier of the safety data sheet

Company	:	Sigma-Aldrich 3050 Spruce Street SAINT LOUIS MO 63103 USA
Telephone Fax	:	+1 800-325-5832 +1 800-325-5052

#### 1.4 Emergency telephone number

Emergency Phone # : +1-703-527-3887 (CHEMTREC)

#### 2. HAZARDS IDENTIFICATION

#### 2.1 Classification of the substance or mixture

#### GHS Classification in accordance with 29 CFR 1910 (OSHA HCS)

Acute toxicity, Inhalation (Category 2), H330 Reproductive toxicity (Category 1B), H360 Specific target organ toxicity - repeated exposure (Category 1), H372 Acute aquatic toxicity (Category 1), H400 Chronic aquatic toxicity (Category 1), H410

For the full text of the H-Statements mentioned in this Section, see Section 16.

#### 2.2 GHS Label elements, including precautionary statements

Pictogram



Signal word Danger Hazard statement(s) Fatal if inhaled. H330 H360 May damage fertility or the unborn child. Causes damage to organs through prolonged or repeated exposure. H372 Very toxic to aquatic life with long lasting effects. H410 Precautionary statement(s) P201 Obtain special instructions before use. P202 Do not handle until all safety precautions have been read and understood. P260 Do not breathe dust/ fume/ gas/ mist/ vapours/ spray.

P264	Wash skin thoroughly after handling.
P270	Do not eat, drink or smoke when using this product.
P271	Use only outdoors or in a well-ventilated area.
P273	Avoid release to the environment.
P280	Wear protective gloves/ protective clothing/ eye protection/ face protection.
P284	Wear respiratory protection.
P304 + P340 + P310	IF INHALED: Remove person to fresh air and keep comfortable for breathing. Immediately call a POISON CENTER/doctor.
P308 + P313	IF exposed or concerned: Get medical advice/ attention.
P391	Collect spillage.
P403 + P233 P405	Store in a well-ventilated place. Keep container tightly closed. Store locked up.
P501	Dispose of contents/ container to an approved waste disposal plant.

#### 2.3 Hazards not otherwise classified (HNOC) or not covered by GHS - none

# **3. COMPOSITION/INFORMATION ON INGREDIENTS**

#### 3.1 Substances

Formula	:	Hg
Molecular weight	:	200.59 g/mol
CAS-No.	:	7439-97-6
EC-No.	:	231-106-7
Index-No.	:	080-001-00-0

#### Hazardous components

Component	Classification	Concentration
Mercury		
	Acute Tox. 2; Repr. 1B; STOT RE 1; Aquatic Acute 1; Aquatic Chronic 1; H330, H360, H372, H410	90 - 100 %

For the full text of the H-Statements mentioned in this Section, see Section 16.

#### 4. FIRST AID MEASURES

#### 4.1 Description of first aid measures

#### General advice

Consult a physician. Show this safety data sheet to the doctor in attendance. Move out of dangerous area.

#### If inhaled

If breathed in, move person into fresh air. If not breathing, give artificial respiration. Consult a physician.

#### In case of skin contact

Wash off with soap and plenty of water. Take victim immediately to hospital. Consult a physician.

#### In case of eye contact

Flush eyes with water as a precaution.

#### If swallowed

Never give anything by mouth to an unconscious person. Rinse mouth with water. Consult a physician.

#### **4.2 Most important symptoms and effects, both acute and delayed** The most important known symptoms and effects are described in the labelling (see section 2.2) and/or in section 11

#### **4.3 Indication of any immediate medical attention and special treatment needed** No data available

#### **5. FIREFIGHTING MEASURES**

#### 5.1 Extinguishing media

#### Suitable extinguishing media

Use water spray, alcohol-resistant foam, dry chemical or carbon dioxide.

5.2 Special hazards arising from the substance or mixture No data available

#### 5.3 Advice for firefighters Wear self-contained breathin

Wear self-contained breathing apparatus for firefighting if necessary.

5.4 Further information No data available

#### 6. ACCIDENTAL RELEASE MEASURES

#### 6.1 Personal precautions, protective equipment and emergency procedures

Wear respiratory protection. Avoid breathing vapours, mist or gas. Ensure adequate ventilation. Evacuate personnel to safe areas.

For personal protection see section 8.

#### 6.2 Environmental precautions

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

#### 6.3 Methods and materials for containment and cleaning up

Soak up with inert absorbent material and dispose of as hazardous waste. Keep in suitable, closed containers for disposal. In some instances, a mercury spill kit may be used. Please consult with your site EHS representative to determine the most appropriate clean up method. Soak up with inert absorbent material and dispose of as hazardous waste. Keep in suitable, closed containers for disposal.

#### 6.4 Reference to other sections

For disposal see section 13.

#### 7. HANDLING AND STORAGE

#### 7.1 Precautions for safe handling

Avoid contact with skin and eyes. Avoid inhalation of vapour or mist. For precautions see section 2.2.

#### 7.2 Conditions for safe storage, including any incompatibilities

Keep container tightly closed in a dry and well-ventilated place. Containers which are opened must be carefully resealed and kept upright to prevent leakage.

# Store under inert gas.

Storage class (TRGS 510): 6.1B: Non-combustible, acute toxic Cat. 1 and 2 / very toxic hazardous materials

#### 7.3 Specific end use(s)

Apart from the uses mentioned in section 1.2 no other specific uses are stipulated

#### 8. EXPOSURE CONTROLS/PERSONAL PROTECTION

#### 8.1 Control parameters

#### Components with workplace control parameters

Component	CAS-No.	Value	Control	Basis
			parameters	
Mercury	7439-97-6	С	0.1 mg/m3	USA. NIOSH Recommended
				Exposure Limits
	Remarks	Potential for dermal absorption		
		CEIL	1.0mg/10m3	USA. Occupational Exposure Limits
				(OSHA) - Table Z-2
		TWA	0.05 mg/m3	USA. OSHA - TABLE Z-1 Limits for
				Air Contaminants - 1910.1000
		Skin notation	1	

TWA	0.025 mg/m3	USA. ACGIH Threshold Limit Values (TLV)	
Central Nervous System impairment			
Substances for which there is a Biological Exposure Index or Indices (see BEI® section)			
Not classifiable as a human carcinogen			
Danger of cutaneous absorption			
TWA	0.05 mg/m3	USA. NIOSH Recommended	
		Exposure Limits	
Potential for dermal absorption			

#### 8.2 Exposure controls

#### Appropriate engineering controls

Avoid contact with skin, eyes and clothing. Wash hands before breaks and immediately after handling the product.

#### Personal protective equipment

#### Eye/face protection

Face shield and safety glasses Use equipment for eye protection tested and approved under appropriate government standards such as NIOSH (US) or EN 166(EU).

#### Skin protection

Handle with gloves. Gloves must be inspected prior to use. Use proper glove removal technique (without touching glove's outer surface) to avoid skin contact with this product. Dispose of contaminated gloves after use in accordance with applicable laws and good laboratory practices. Wash and dry hands.

Full contact Material: Nitrile rubber Minimum layer thickness: 0.11 mm Break through time: 480 min Material tested:Dermatril® (KCL 740 / Aldrich Z677272, Size M)

Splash contact Material: Nitrile rubber Minimum layer thickness: 0.11 mm Break through time: 480 min Material tested:Dermatril® (KCL 740 / Aldrich Z677272, Size M)

data source: KCL GmbH, D-36124 Eichenzell, phone +49 (0)6659 87300, e-mail sales@kcl.de, test method: EN374

If used in solution, or mixed with other substances, and under conditions which differ from EN 374, contact the supplier of the CE approved gloves. This recommendation is advisory only and must be evaluated by an industrial hygienist and safety officer familiar with the specific situation of anticipated use by our customers. It should not be construed as offering an approval for any specific use scenario.

#### **Body Protection**

Complete suit protecting against chemicals, The type of protective equipment must be selected according to the concentration and amount of the dangerous substance at the specific workplace.

#### **Respiratory protection**

Where risk assessment shows air-purifying respirators are appropriate use a full-face respirator with multipurpose combination (US) or type ABEK (EN 14387) respirator cartridges as a backup to engineering controls. If the respirator is the sole means of protection, use a full-face supplied air respirator. Use respirators and components tested and approved under appropriate government standards such as NIOSH (US) or CEN (EU).

#### Control of environmental exposure

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

#### 9. PHYSICAL AND CHEMICAL PROPERTIES

#### 9.1 Information on basic physical and chemical properties

a) Appearance

Form: liquid Colour: silver, white

	b)	Odour	odourless
	c)	Odour Threshold	No data available
	d)	рН	No data available
	e)	Melting point/freezing point	Melting point/range: -38.87 °C (-37.97 °F) - lit.
	f)	Initial boiling point and boiling range	356.6 °C (673.9 °F) - lit.
	g)	Flash point	Not applicable
	h)	Evaporation rate	No data available
	i)	Flammability (solid, gas)	No data available
	j)	Upper/lower flammability or explosive limits	No data available
	k)	Vapour pressure	< 0.01 hPa (< 0.01 mmHg) at 20 °C (68 °F) 1 hPa (1 mmHg) at 126 °C (259 °F)
	I)	Vapour density	6.93 - (Air = 1.0)
	m)	Relative density	13.55 g/cm3 at 25 °C (77 °F)
	n)	Water solubility	0.00006 g/l at 25 °C (77 °F)
	o)	Partition coefficient: n- octanol/water	No data available
	p)	Auto-ignition temperature	No data available
	q)	Decomposition temperature	No data available
	r)	Viscosity	No data available
	s)	Explosive properties	No data available
	t)	Oxidizing properties	No data available
.2	Othe	r safety information	
		Relative vapour density	6.93 - (Air = 1.0)
10	STAR	ILITY AND REACTIVITY	
0.1	Reac	tivity	
	No da	ata available	

# 10.2 Chemical stability

9.2

10.1

Stable under recommended storage conditions.

- 10.3 Possibility of hazardous reactions No data available
- 10.4 Conditions to avoid No data available

#### 10.5 Incompatible materials Strong oxidizing agents, Ammonia, Azides, Nitrates, Chlorates, Copper

# **10.6 Hazardous decomposition products** Hazardous decomposition products formed under fire conditions. - Mercury/mercury oxides. Other decomposition products - No data available In the event of fire: see section 5

#### **11. TOXICOLOGICAL INFORMATION**

#### 11.1 Information on toxicological effects

#### Acute toxicity No data available

LC50 Inhalation - Rat - male - 2 h - < 27 mg/m3

Dermal: No data available

No data available

Skin corrosion/irritation No data available

Serious eye damage/eye irritation No data available

Respiratory or skin sensitisation No data available

#### Germ cell mutagenicity No data available

#### Carcinogenicity

This product is or contains a component that is not classifiable as to its carcinogenicity based on its IARC, ACGIH, NTP, or EPA classification.

- IARC: No component of this product present at levels greater than or equal to 0.1% is identified as probable, possible or confirmed human carcinogen by IARC.
- NTP: No component of this product present at levels greater than or equal to 0.1% is identified as a known or anticipated carcinogen by NTP.
- OSHA: No component of this product present at levels greater than or equal to 0.1% is on OSHA's list of regulated carcinogens.

#### Reproductive toxicity

Presumed human reproductive toxicant

Specific target organ toxicity - single exposure No data available

#### Specific target organ toxicity - repeated exposure

Causes damage to organs through prolonged or repeated exposure.

#### Aspiration hazard No data available

# Additional Information

RTECS: OV4550000

To the best of our knowledge, the chemical, physical, and toxicological properties have not been thoroughly investigated.

Stomach - Irregularities - Based on Human Evidence Stomach - Irregularities - Based on Human Evidence

#### **12. ECOLOGICAL INFORMATION**

#### 12.1 Toxicity

Toxicity to fish mortality LC50 - Cyprinus carpio (Carp) - 0.160 mg/l - 96 h

#### **12.2 Persistence and degradability** No data available

#### 12.3 Bioaccumulative potential

Bioaccumulation

Carassius auratus (goldfish) - 1,789 d - 0.25 µg/l

#### 12.4 Mobility in soil

No data available

#### 12.5 Results of PBT and vPvB assessment

PBT/vPvB assessment not available as chemical safety assessment not required/not conducted

#### 12.6 Other adverse effects

An environmental hazard cannot be excluded in the event of unprofessional handling or disposal. Very toxic to aquatic life with long lasting effects.

#### **13. DISPOSAL CONSIDERATIONS**

#### 13.1 Waste treatment methods

#### Product

Offer surplus and non-recyclable solutions to a licensed disposal company. Contact a licensed professional waste disposal service to dispose of this material.

#### Contaminated packaging

Dispose of as unused product.

#### **14. TRANSPORT INFORMATION**

#### DOT (US)

UN number: 2809 Class: 8 (6.1) Proper shipping name: A. W. Mercury Reportable Quantity (RQ): 1 lbs Poison Inhalation Hazard: No Packing group: III

#### IMDG

#### ΙΑΤΑ

UN number: 2809 Class: 8 (6.1) Proper shipping name: Mercury Packing group: III

#### **15. REGULATORY INFORMATION**

#### SARA 302 Components

No chemicals in this material are subject to the reporting requirements of SARA Title III, Section 302.

#### SARA 313 Components

This material does not contain any chemical components with known CAS numbers that exceed the threshold (De Minimis) reporting levels established by SARA Title III, Section 313.

#### SARA 311/312 Hazards

Acute Health Hazard, Chronic Health Hazard

#### Massachusetts Right To Know Components

Mercury	CAS-No. 7439-97-6	Revision Date 2015-11-23
Pennsylvania Right To Know Components		
Mercury	CAS-No. 7439-97-6	Revision Date 2015-11-23
Mercury	CAS-No. 7439-97-6	Revision Date 2015-11-23
New Jersey Right To Know Components		
Mercury	CAS-No. 7439-97-6	Revision Date 2015-11-23
California Prop. 65 Components		

WARNING: This product contains a chemical known to the CAS-No. Revision Date State of California to cause birth defects or other reproductive 7439-97-6 2013-12-20 harm. Mercury

## **16. OTHER INFORMATION**

#### Full text of H-Statements referred to under sections 2 and 3.

0 0

Acute toxicity
Acute aquatic toxicity
Chronic aquatic toxicity
Fatal if inhaled.
May damage fertility or the unborn child.
Causes damage to organs through prolonged or repeated exposure.
Very toxic to aquatic life.
Very toxic to aquatic life with long lasting effects.
Reproductive toxicity

#### HMIS Rating

Health hazard:	2
Chronic Health Hazard:	*
Flammability:	0
Physical Hazard	0
NFPA Rating	
Health hazard:	2

Health hazard:	
Fire Hazard:	
Reactivity Hazard:	

#### Further information

Copyright 2016 Sigma-Aldrich Co. LLC. License granted to make unlimited paper copies for internal use only. The above information is believed to be correct but does not purport to be all inclusive and shall be used only as a guide. The information in this document is based on the present state of our knowledge and is applicable to the product with regard to appropriate safety precautions. It does not represent any guarantee of the properties of the product. Sigma-Aldrich Corporation and its Affiliates shall not be held liable for any damage resulting from handling or from contact with the above product. See www.sigma-aldrich.com and/or the reverse side of invoice or packing slip for additional terms and conditions of sale.

#### **Preparation Information**

Sigma-Aldrich Corporation Product Safety – Americas Region 1-800-521-8956

Version: 3.15

Revision Date: 03/05/2018

Print Date: 06/28/2019

sigma-aldrich.com

# SAFETY DATA SHEET

Version 6.0 Revision Date 05/28/2017 Print Date 06/28/2019

#### 1. PRODUCT AND COMPANY IDENTIFICATION

1.1	Product identifiers Product name		Zinc	
	Product Number Brand Index-No.	:	324930 Aldrich 030-001-00-1	
	CAS-No.	:	7440-66-6	

#### 1.2 Relevant identified uses of the substance or mixture and uses advised against

Identified uses : Laboratory chemicals, Synthesis of substances

#### 1.3 Details of the supplier of the safety data sheet

Company	:	Sigma-Aldrich Inc. 3050 Spruce Street ST. LOUIS MO 63103 UNITED STATES
Telephone Fax	:	+1 314 771-5765 +1 800 325-5052
Emergency telephone nu	mhe	r

#### 1.4 Emergency telephone number

: +1-703-527-3887 Emergency Phone #

#### 2. HAZARDS IDENTIFICATION

#### 2.1 Classification of the substance or mixture

#### GHS Classification in accordance with 29 CFR 1910 (OSHA HCS)

Pyrophoric solids (Category 1), H250 Self-heating substances and mixtures (Category 1), H251 Substances and mixtures, which in contact with water, emit flammable gases (Category 1), H260 Acute aquatic toxicity (Category 1), H400 Chronic aquatic toxicity (Category 1), H410

For the full text of the H-Statements mentioned in this Section, see Section 16.

#### 2.2 GHS Label elements, including precautionary statements

Pictogram

Ł

Signal word	Danger
Hazard statement(s)	
H250	Catches fire spontaneously if exposed to air.
H251	Self-heating: may catch fire.
H260	In contact with water releases flammable gases which may ignite spontaneously.
H410	Very toxic to aquatic life with long lasting effects.

Precautionary statement(s)	
P210	Keep away from heat/sparks/open flames/hot surfaces. No smoking.
P222	Do not allow contact with air.
P223	Do not allow contact with water.
P231 + P232	Handle under inert gas. Protect from moisture.
P235 + P410	Keep cool. Protect from sunlight.
P273	Avoid release to the environment.
P280	Wear protective gloves/ eye protection/ face protection.
P335 + P334	Brush off loose particles from skin. Immerse in cool water/ wrap in wet bandages.
P370 + P378	In case of fire: Use dry sand, dry chemical or alcohol-resistant foam to extinguish.
P391	Collect spillage.
P402 + P404	Store in a dry place. Store in a closed container.
P407	Maintain air gap between stacks/ pallets.
P413	Store bulk masses greater than .? kg/ .? lbs at temperatures not exceeding .? °C/ .? °F.
P420	Store away from other materials.
P422	Store contents under inert gas.
P501	Dispose of contents/ container to an approved waste disposal plant.

#### 2.3 Hazards not otherwise classified (HNOC) or not covered by GHS Combustible dust

#### **3. COMPOSITION/INFORMATION ON INGREDIENTS**

#### 3.1 Substances

Formula	:	Zn
Molecular weight	:	65.39 g/mol
CAS-No.	:	7440-66-6
EC-No.	:	231-175-3
Index-No.	:	030-001-00-1

#### Hazardous components

Component	Classification	Concentration
Zinc powder (pyrophoric)		
	Pyr. Sol. 1; Self-heat. 1; Water-react. 1; Aquatic Acute 1; Aquatic Chronic 1; H250, H251, H260, H410	<= 100 %

For the full text of the H-Statements mentioned in this Section, see Section 16.

#### 4. FIRST AID MEASURES

#### 4.1 Description of first aid measures

#### **General advice**

Consult a physician. Show this safety data sheet to the doctor in attendance. Move out of dangerous area.

#### If inhaled

If breathed in, move person into fresh air. If not breathing, give artificial respiration. Consult a physician.

#### In case of skin contact

Wash off with soap and plenty of water. Consult a physician.

#### In case of eye contact

Flush eyes with water as a precaution.

#### If swallowed

Never give anything by mouth to an unconscious person. Rinse mouth with water. Consult a physician.

#### **4.2 Most important symptoms and effects, both acute and delayed** The most important known symptoms and effects are described in the labelling (see section 2.2) and/or in section 11

**4.3 Indication of any immediate medical attention and special treatment needed** No data available

#### **5. FIREFIGHTING MEASURES**

#### 5.1 Extinguishing media

Suitable extinguishing media Dry powder

5.2 Special hazards arising from the substance or mixture Zinc/zinc oxides

#### **5.3** Advice for firefighters Wear self-contained breathing apparatus for firefighting if necessary.

5.4 Further information No data available

#### 6. ACCIDENTAL RELEASE MEASURES

#### 6.1 Personal precautions, protective equipment and emergency procedures

Avoid dust formation. Avoid breathing vapours, mist or gas. Ensure adequate ventilation. Evacuate personnel to safe areas.

For personal protection see section 8.

#### 6.2 Environmental precautions

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

#### 6.3 Methods and materials for containment and cleaning up

Sweep up and shovel. Contain spillage, and then collect with an electrically protected vacuum cleaner or by wetbrushing and place in container for disposal according to local regulations (see section 13). Do not flush with water. Keep in suitable, closed containers for disposal. Contain spillage, pick up with an electrically protected vacuum cleaner or by wet-brushing and transfer to a container for disposal according to local regulations (see section 13).

#### 6.4 Reference to other sections

For disposal see section 13.

#### 7. HANDLING AND STORAGE

#### 7.1 Precautions for safe handling

Further processing of solid materials may result in the formation of combu formation should be taken into consideration before additional processing

Provide appropriate exhaust ventilation at places where dust is formed.Keep away from sources of ignition - No smoking.

For precautions see section 2.2.

#### 7.2 Conditions for safe storage, including any incompatibilities

Keep container tightly closed in a dry and well-ventilated place. Never allow product to get in contact with water during storage.

Keep in a dry place.

#### 7.3 Specific end use(s)

Apart from the uses mentioned in section 1.2 no other specific uses are stipulated

## 8. EXPOSURE CONTROLS/PERSONAL PROTECTION

#### 8.1 Control parameters

#### Components with workplace control parameters

Contains no substances with occupational exposure limit values. Hazardous components without workplace control parameters

#### 8.2 Exposure controls

#### Appropriate engineering controls

Handle in accordance with good industrial hygiene and safety practice. Wash hands before breaks and at the end of workday.

#### Personal protective equipment

#### Eye/face protection

Face shield and safety glasses Use equipment for eye protection tested and approved under appropriate government standards such as NIOSH (US) or EN 166(EU).

#### Skin protection

Handle with gloves. Gloves must be inspected prior to use. Use proper glove removal technique (without touching glove's outer surface) to avoid skin contact with this product. Dispose of contaminated gloves after use in accordance with applicable laws and good laboratory practices. Wash and dry hands.

Protective gloves against thermal risks

Full contact Material: Nitrile rubber Minimum layer thickness: 0.11 mm Break through time: 480 min Material tested:Dermatril® (KCL 740 / Aldrich Z677272, Size M)

Splash contact Material: Nitrile rubber Minimum layer thickness: 0.11 mm Break through time: 480 min Material tested:Dermatril® (KCL 740 / Aldrich Z677272, Size M)

data source: KCL GmbH, D-36124 Eichenzell, phone +49 (0)6659 87300, e-mail sales@kcl.de, test method: EN374

If used in solution, or mixed with other substances, and under conditions which differ from EN 374, contact the supplier of the CE approved gloves. This recommendation is advisory only and must be evaluated by an industria situation of anticipated use by our customers. It should not be construed as offering an approval for any specific use scenario.

#### **Body Protection**

Flame retardant protective clothing, The type of protective equipment must be selected according to the concentration and amount of the dangerous substance at the specific workplace.

#### **Respiratory protection**

Where risk assessment shows air-purifying respirators are appropriate use (EN 143) respirator cartridges as a backup to engineering controls. If th full-face supplied air respirator. Use respirators and components tested and approved under appropriate government standards such as NIOSH (US) or CEN (EU).

#### Control of environmental exposure

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

#### 9. PHYSICAL AND CHEMICAL PROPERTIES

#### 9.1 Information on basic physical and chemical properties

a)	Appearance	Form: powder Colour: grey
b)	Odour	No data available
c)	Odour Threshold	No data available
d)	рН	No data available
e)	Melting point/freezing point	Melting point/range: 420 °C (788 °F) - lit.
f)	Initial boiling point and boiling range	907 °C (1665 °F) - lit.
g)	Flash point	()No data available
h)	Evaporation rate	No data available

i)	Flammability (solid, gas)	May form combustible dust concentrations in air.
j)	Upper/lower flammability or explosive limits	No data available
k)	Vapour pressure	1 hPa at 487 °C (909 °F)
I)	Vapour density	No data available
m)	Relative density	7.133 g/mL at 25 °C (77 °F)
n)	Water solubility	No data available
o)	Partition coefficient: n- octanol/water	log Pow: 5
p)	Auto-ignition temperature	The substance or mixture is classified as self heating with the category 1., The substance or mixture is pyrophoric with the category 1.
q)	Decomposition temperature	No data available
r)	Viscosity	No data available
s)	Explosive properties	No data available
t)	Oxidizing properties	No data available
Oth	ner safety information	

No data available

## **10. STABILITY AND REACTIVITY**

10.1 Reactivity No data available

9.2

- **10.2 Chemical stability** Stable under recommended storage conditions.
- **10.3 Possibility of hazardous reactions** Reacts violently with water.
- **10.4** Conditions to avoid Exposure to moisture
- **10.5** Incompatible materials Strong acids and oxidizing agents

# Hazardous decomposition products Hazardous decomposition products formed under fire conditions. - Zinc/zinc oxides Other decomposition products - No data available In the event of fire: see section 5

#### 11. TOXICOLOGICAL INFORMATION

#### 11.1 Information on toxicological effects

#### Acute toxicity

No data availableZinc powder (pyrophoric) Inhalation: No data available(Zinc powder (pyrophoric)) Dermal: No data available(Zinc powder (pyrophoric)) No data available(Zinc powder (pyrophoric))

#### Skin corrosion/irritation

No data available(Zinc powder (pyrophoric))

#### Serious eye damage/eye irritation No data available(Zinc powder (pyrophoric))

#### Respiratory or skin sensitisation

Did not cause sensitisation on laboratory animals.(Zinc powder (pyrophoric))

#### Germ cell mutagenicity

No data available(Zinc powder (pyrophoric))

#### Carcinogenicity

- IARC: No component of this product present at levels greater than or equal to 0.1% is identified as probable, possible or confirmed human carcinogen by IARC.
- NTP: No component of this product present at levels greater than or equal to 0.1% is identified as a known or anticipated carcinogen by NTP.
- OSHA: No component of this product present at levels greater than or equal to 0.1% is identified as a carcinogen or potential carcinogen by OSHA.

#### **Reproductive toxicity**

No data available(Zinc powder (pyrophoric))

No data available(Zinc powder (pyrophoric))

**Specific target organ toxicity - single exposure** No data available(Zinc powder (pyrophoric))

Specific target organ toxicity - repeated exposure No data available

#### Aspiration hazard

No data available(Zinc powder (pyrophoric))

#### **Additional Information**

RTECS: ZG8600000

chills, dry throat, sweet taste, Fever, Cough, Nausea, Vomiting, Weakness(Zinc powder (pyrophoric)) To the best of our knowledge, the chemical, physical, and toxicological properties have not been thoroughly investigated.(Zinc powder (pyrophoric))

#### **12. ECOLOGICAL INFORMATION**

#### 12.1 Toxicity

Toxicity to fish	LC50 - Cyprinus carpio (Carp) - 450.0 µg/l - 96.0 h(Zinc powder (pyrophoric))
Toxicity to daphnia and other aquatic invertebrates	LC50 - Daphnia magna (Water flea) - 0.068 mg/I  - 48 h(Zinc powder (pyrophoric))
	mortality NOEC - Daphnia (water flea) - 0.101 - 0.14 mg/l  - 7 d(Zinc powder (pyrophoric))

#### 12.2 Persistence and degradability

#### 12.3 Bioaccumulative potential

Bioaccumulation Algae - 7 d

at 16 °C - 5 µg/l(Zinc powder (pyrophoric))

Bioconcentration factor (BCF): 466

#### 12.4 Mobility in soil

No data available(Zinc powder (pyrophoric))

#### 12.5 Results of PBT and vPvB assessment

PBT/vPvB assessment not available as chemical safety assessment not required/not conducted

#### 12.6 Other adverse effects

An environmental hazard cannot be excluded in the event of unprofessional handling or disposal. Very toxic to aquatic life.

#### **13. DISPOSAL CONSIDERATIONS**

#### 13.1 Waste treatment methods

#### Product

Burn in a chemical incinerator equipped with an afterburner and scrubber b highly flammable. Offer surplus and non-recyclable solutions to a licensed disposal company. Contact a licensed professional waste disposal service to dispose of this material.

#### Contaminated packaging

Dispose of as unused product.

#### **14. TRANSPORT INFORMATION**

# DOT (US)

DOT (03)				
UN number: 1436	Class: 4.3 (4.2)	Packing	group: II	
Proper shipping name:	Zinc powder			
Reportable Quantity (R	Q) :	1000 lbs		
Poison Inhalation Haza	rd: No			
IMDG				

UN number: 1436 Class: 4.3 (4.2) Proper shipping name: ZINC POWDER Marine pollutant : yes

Packing group: II

EMS-No: F-G, S-O

#### ΙΑΤΑ

UN number: 1436 Class: 4.3 (4.2) Proper shipping name: Zinc powder

Packing group: II

#### **15. REGULATORY INFORMATION**

#### SARA 302 Components

No chemicals in this material are subject to the reporting requirements of SARA Title III, Section 302.

#### SARA 313 Components

The following components are subject to reporting levels establish	ed by SARA Title III,	Section 313:
Zinc nowder (nyronhoric)	CAS-No. 7440-66-6	Revision Date
	7440-00-0	1333-04-24
SARA 311/312 Hazards Reactivity Hazard		
Massachusetts Right To Know Components		
	CAS-No.	<b>Revision Date</b>
Zinc powder (pyrophoric)	7440-66-6	1993-04-24
Pennsylvania Right To Know Components		
	CAS-No.	<b>Revision Date</b>
Zinc powder (pyrophoric)	7440-66-6	1993-04-24
New Jersey Right To Know Components		
	CAS-No.	<b>Revision Date</b>
Zinc powder (pyrophoric)	7440-66-6	1993-04-24

#### California Prop. 65 Components

This product does not contain any chemicals known to State of California to cause cancer, birth defects, or any other reproductive harm.

#### **16. OTHER INFORMATION**

#### Full text of H-Statements referred to under sections 2 and 3.

H250	Catches fire spontaneously if exposed to air.
H251	Self-heating: may catch fire.
H260	In contact with water releases flammable gases which may ignite spontaneously.
H400	Very toxic to aquatic life.
H410	Very toxic to aquatic life with long lasting effects.

#### **HMIS** Rating

Health hazard:	0
Chronic Health Hazard:	
Flammability:	3
Physical Hazard	1
NFPA Rating	
Health hazard:	0
Fire Hazard:	3
Reactivity Hazard:	1
Special hazard.I:	W

#### Further information

Copyright 2016 Sigma-Aldrich Co. LLC. License granted to make unlimited paper copies for internal use only. The above information is believed to be correct but does not purport to be all inclusive and shall be used only as a guide. The information in this document is based on the present state of our knowledge and is applicable to the product with regard to appropriate safety precautions. It does not represent any guarantee of the properties of the product. Sigma-Aldrich Corporation and its Affiliates shall not be held liable for any damage resulting from handling or from contact with the above product. See www.sigma-aldrich.com and/or the reverse side of invoice or packing slip for additional terms and conditions of sale.

#### **Preparation Information**

Sigma-Aldrich Corporation Product Safety – Americas Region 1-800-521-8956 Version: 6.0

Revision Date: 05/28/2017

Print Date: 06/28/2019

# SIGMA-ALDRICH

# SAFETY DATA SHEET

Version 3.15 Revision Date 03/05/2018 Print Date 06/28/2019

#### **1. PRODUCT AND COMPANY IDENTIFICATION**

1.1	Product identifiers Product name	:	Mercury
	Product Number Brand Index-No.	:	215457 Sigma-Aldrich 080-001-00-0
	CAS-No.	:	7439-97-6

#### 1.2 Relevant identified uses of the substance or mixture and uses advised against

Identified uses : Laboratory chemicals, Synthesis of substances

#### 1.3 Details of the supplier of the safety data sheet

Company	:	Sigma-Aldrich 3050 Spruce Street SAINT LOUIS MO 6310 USA	
Telephone Fax	:	+1 800-325-5832 +1 800-325-5052	

#### 1.4 Emergency telephone number

Emergency Phone # : +1-703-527-3887 (CHEMTREC)

#### 2. HAZARDS IDENTIFICATION

#### 2.1 Classification of the substance or mixture

#### GHS Classification in accordance with 29 CFR 1910 (OSHA HCS)

Acute toxicity, Inhalation (Category 2), H330 Reproductive toxicity (Category 1B), H360 Specific target organ toxicity - repeated exposure (Category 1), H372 Acute aquatic toxicity (Category 1), H400 Chronic aquatic toxicity (Category 1), H410

For the full text of the H-Statements mentioned in this Section, see Section 16.

#### 2.2 GHS Label elements, including precautionary statements

Pictogram



Signal word Danger Hazard statement(s) Fatal if inhaled. H330 H360 May damage fertility or the unborn child. Causes damage to organs through prolonged or repeated exposure. H372 Very toxic to aquatic life with long lasting effects. H410 Precautionary statement(s) P201 Obtain special instructions before use. P202 Do not handle until all safety precautions have been read and understood. P260 Do not breathe dust/ fume/ gas/ mist/ vapours/ spray.

P264	Wash skin thoroughly after handling.
P270	Do not eat, drink or smoke when using this product.
P271	Use only outdoors or in a well-ventilated area.
P273	Avoid release to the environment.
P280	Wear protective gloves/ protective clothing/ eye protection/ face protection.
P284	Wear respiratory protection.
P304 + P340 + P310	IF INHALED: Remove person to fresh air and keep comfortable for breathing. Immediately call a POISON CENTER/doctor.
P308 + P313	IF exposed or concerned: Get medical advice/ attention.
P391	Collect spillage.
P403 + P233 P405	Store in a well-ventilated place. Keep container tightly closed. Store locked up.
P501	Dispose of contents/ container to an approved waste disposal plant.

#### 2.3 Hazards not otherwise classified (HNOC) or not covered by GHS - none

# **3. COMPOSITION/INFORMATION ON INGREDIENTS**

#### 3.1 Substances

Formula	:	Hg
Molecular weight	:	200.59 g/mol
CAS-No.	:	7439-97-6
EC-No.	:	231-106-7
Index-No.	:	080-001-00-0

#### Hazardous components

Component	Classification	Concentration
Mercury		
	Acute Tox. 2; Repr. 1B; STOT RE 1; Aquatic Acute 1; Aquatic Chronic 1; H330, H360, H372, H410	90 - 100 %

For the full text of the H-Statements mentioned in this Section, see Section 16.

#### 4. FIRST AID MEASURES

#### 4.1 Description of first aid measures

#### General advice

Consult a physician. Show this safety data sheet to the doctor in attendance. Move out of dangerous area.

#### If inhaled

If breathed in, move person into fresh air. If not breathing, give artificial respiration. Consult a physician.

#### In case of skin contact

Wash off with soap and plenty of water. Take victim immediately to hospital. Consult a physician.

#### In case of eye contact

Flush eyes with water as a precaution.

#### If swallowed

Never give anything by mouth to an unconscious person. Rinse mouth with water. Consult a physician.

#### **4.2 Most important symptoms and effects, both acute and delayed** The most important known symptoms and effects are described in the labelling (see section 2.2) and/or in section 11

#### **4.3 Indication of any immediate medical attention and special treatment needed** No data available

#### **5. FIREFIGHTING MEASURES**

#### 5.1 Extinguishing media

#### Suitable extinguishing media

Use water spray, alcohol-resistant foam, dry chemical or carbon dioxide.

5.2 Special hazards arising from the substance or mixture No data available

#### 5.3 Advice for firefighters Wear self-contained breathin

Wear self-contained breathing apparatus for firefighting if necessary.

5.4 Further information No data available

#### 6. ACCIDENTAL RELEASE MEASURES

#### 6.1 Personal precautions, protective equipment and emergency procedures

Wear respiratory protection. Avoid breathing vapours, mist or gas. Ensure adequate ventilation. Evacuate personnel to safe areas.

For personal protection see section 8.

#### 6.2 Environmental precautions

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

#### 6.3 Methods and materials for containment and cleaning up

Soak up with inert absorbent material and dispose of as hazardous waste. Keep in suitable, closed containers for disposal. In some instances, a mercury spill kit may be used. Please consult with your site EHS representative to determine the most appropriate clean up method. Soak up with inert absorbent material and dispose of as hazardous waste. Keep in suitable, closed containers for disposal.

#### 6.4 Reference to other sections

For disposal see section 13.

#### 7. HANDLING AND STORAGE

#### 7.1 Precautions for safe handling

Avoid contact with skin and eyes. Avoid inhalation of vapour or mist. For precautions see section 2.2.

#### 7.2 Conditions for safe storage, including any incompatibilities

Keep container tightly closed in a dry and well-ventilated place. Containers which are opened must be carefully resealed and kept upright to prevent leakage.

# Store under inert gas.

Storage class (TRGS 510): 6.1B: Non-combustible, acute toxic Cat. 1 and 2 / very toxic hazardous materials

#### 7.3 Specific end use(s)

Apart from the uses mentioned in section 1.2 no other specific uses are stipulated

#### 8. EXPOSURE CONTROLS/PERSONAL PROTECTION

#### 8.1 Control parameters

#### Components with workplace control parameters

Component	CAS-No.	Value	Control	Basis
			parameters	
Mercury	7439-97-6	С	0.1 mg/m3	USA. NIOSH Recommended
				Exposure Limits
	Remarks	Potential for dermal absorption		
		CEIL	1.0mg/10m3	USA. Occupational Exposure Limits
				(OSHA) - Table Z-2
		TWA	0.05 mg/m3	USA. OSHA - TABLE Z-1 Limits for
				Air Contaminants - 1910.1000
		Skin notation	1	

TWA	0.025 mg/m3	USA. ACGIH Threshold Limit Values (TLV)		
Central Nerv	ous System impair	ment		
Substances for which there is a Biological Exposure Index or Indices (see BEI® section)				
Not classifiable as a human carcinogen				
Danger of cutaneous absorption				
TWA	0.05 mg/m3	USA. NIOSH Recommended		
		Exposure Limits		
Potential for dermal absorption				

#### 8.2 Exposure controls

#### Appropriate engineering controls

Avoid contact with skin, eyes and clothing. Wash hands before breaks and immediately after handling the product.

#### Personal protective equipment

#### Eye/face protection

Face shield and safety glasses Use equipment for eye protection tested and approved under appropriate government standards such as NIOSH (US) or EN 166(EU).

#### Skin protection

Handle with gloves. Gloves must be inspected prior to use. Use proper glove removal technique (without touching glove's outer surface) to avoid skin contact with this product. Dispose of contaminated gloves after use in accordance with applicable laws and good laboratory practices. Wash and dry hands.

Full contact Material: Nitrile rubber Minimum layer thickness: 0.11 mm Break through time: 480 min Material tested:Dermatril® (KCL 740 / Aldrich Z677272, Size M)

Splash contact Material: Nitrile rubber Minimum layer thickness: 0.11 mm Break through time: 480 min Material tested:Dermatril® (KCL 740 / Aldrich Z677272, Size M)

data source: KCL GmbH, D-36124 Eichenzell, phone +49 (0)6659 87300, e-mail sales@kcl.de, test method: EN374

If used in solution, or mixed with other substances, and under conditions which differ from EN 374, contact the supplier of the CE approved gloves. This recommendation is advisory only and must be evaluated by an industrial hygienist and safety officer familiar with the specific situation of anticipated use by our customers. It should not be construed as offering an approval for any specific use scenario.

#### **Body Protection**

Complete suit protecting against chemicals, The type of protective equipment must be selected according to the concentration and amount of the dangerous substance at the specific workplace.

#### **Respiratory protection**

Where risk assessment shows air-purifying respirators are appropriate use a full-face respirator with multipurpose combination (US) or type ABEK (EN 14387) respirator cartridges as a backup to engineering controls. If the respirator is the sole means of protection, use a full-face supplied air respirator. Use respirators and components tested and approved under appropriate government standards such as NIOSH (US) or CEN (EU).

#### Control of environmental exposure

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

#### 9. PHYSICAL AND CHEMICAL PROPERTIES

#### 9.1 Information on basic physical and chemical properties

a) Appearance

Form: liquid Colour: silver, white

	b)	Odour	odourless		
	c)	Odour Threshold	No data available		
	d)	рН	No data available		
	e)	Melting point/freezing point	Melting point/range: -38.87 °C (-37.97 °F) - lit.		
	f)	Initial boiling point and boiling range	356.6 °C (673.9 °F) - lit.		
	g)	Flash point	Not applicable		
	h)	Evaporation rate	No data available		
	i)	Flammability (solid, gas)	No data available		
	j)	Upper/lower flammability or explosive limits	No data available		
	k)	Vapour pressure	< 0.01 hPa (< 0.01 mmHg) at 20 °C (68 °F) 1 hPa (1 mmHg) at 126 °C (259 °F)		
	I)	Vapour density	6.93 - (Air = 1.0)		
	m)	Relative density	13.55 g/cm3 at 25 °C (77 °F)		
	n)	Water solubility	0.00006 g/l at 25 °C (77 °F)		
	o)	Partition coefficient: n- octanol/water	No data available		
	p)	Auto-ignition temperature	No data available		
	q)	Decomposition temperature	No data available		
	r)	Viscosity	No data available		
	s)	Explosive properties	No data available		
	t)	Oxidizing properties	No data available		
.2	Othe	r safety information			
		Relative vapour density	6.93 - (Air = 1.0)		
10. STABILITY AND REACTIVITY					
	No data available				

# 10.2 Chemical stability

9.2

10.1

Stable under recommended storage conditions.

- 10.3 Possibility of hazardous reactions No data available
- 10.4 Conditions to avoid No data available

#### 10.5 Incompatible materials Strong oxidizing agents, Ammonia, Azides, Nitrates, Chlorates, Copper

# **10.6 Hazardous decomposition products** Hazardous decomposition products formed under fire conditions. - Mercury/mercury oxides. Other decomposition products - No data available In the event of fire: see section 5

#### **11. TOXICOLOGICAL INFORMATION**

#### 11.1 Information on toxicological effects

#### Acute toxicity No data available

LC50 Inhalation - Rat - male - 2 h - < 27 mg/m3

Dermal: No data available

No data available

Skin corrosion/irritation No data available

Serious eye damage/eye irritation No data available

Respiratory or skin sensitisation No data available

#### Germ cell mutagenicity No data available

#### Carcinogenicity

This product is or contains a component that is not classifiable as to its carcinogenicity based on its IARC, ACGIH, NTP, or EPA classification.

- IARC: No component of this product present at levels greater than or equal to 0.1% is identified as probable, possible or confirmed human carcinogen by IARC.
- NTP: No component of this product present at levels greater than or equal to 0.1% is identified as a known or anticipated carcinogen by NTP.
- OSHA: No component of this product present at levels greater than or equal to 0.1% is on OSHA's list of regulated carcinogens.

#### Reproductive toxicity

Presumed human reproductive toxicant

Specific target organ toxicity - single exposure No data available

#### Specific target organ toxicity - repeated exposure

Causes damage to organs through prolonged or repeated exposure.

#### Aspiration hazard No data available

# Additional Information

RTECS: OV4550000

To the best of our knowledge, the chemical, physical, and toxicological properties have not been thoroughly investigated.

Stomach - Irregularities - Based on Human Evidence Stomach - Irregularities - Based on Human Evidence

#### **12. ECOLOGICAL INFORMATION**

#### 12.1 Toxicity

Toxicity to fish mortality LC50 - Cyprinus carpio (Carp) - 0.160 mg/l - 96 h

#### **12.2 Persistence and degradability** No data available

#### 12.3 Bioaccumulative potential

Bioaccumulation

Carassius auratus (goldfish) - 1,789 d - 0.25 µg/l

#### 12.4 Mobility in soil

No data available

#### 12.5 Results of PBT and vPvB assessment

PBT/vPvB assessment not available as chemical safety assessment not required/not conducted

#### 12.6 Other adverse effects

An environmental hazard cannot be excluded in the event of unprofessional handling or disposal. Very toxic to aquatic life with long lasting effects.

#### **13. DISPOSAL CONSIDERATIONS**

#### 13.1 Waste treatment methods

#### Product

Offer surplus and non-recyclable solutions to a licensed disposal company. Contact a licensed professional waste disposal service to dispose of this material.

#### Contaminated packaging

Dispose of as unused product.

#### **14. TRANSPORT INFORMATION**

#### DOT (US)

UN number: 2809 Class: 8 (6.1) Proper shipping name: A. W. Mercury Reportable Quantity (RQ): 1 lbs Poison Inhalation Hazard: No Packing group: III

#### IMDG

#### ΙΑΤΑ

UN number: 2809 Class: 8 (6.1) Proper shipping name: Mercury Packing group: III

#### **15. REGULATORY INFORMATION**

#### SARA 302 Components

No chemicals in this material are subject to the reporting requirements of SARA Title III, Section 302.

#### SARA 313 Components

This material does not contain any chemical components with known CAS numbers that exceed the threshold (De Minimis) reporting levels established by SARA Title III, Section 313.

#### SARA 311/312 Hazards

Acute Health Hazard, Chronic Health Hazard

#### Massachusetts Right To Know Components

Mercury	CAS-No. 7439-97-6	Revision Date 2015-11-23
Pennsylvania Right To Know Components		
Mercury	CAS-No. 7439-97-6	Revision Date 2015-11-23
Mercury	CAS-No. 7439-97-6	Revision Date 2015-11-23
New Jersey Right To Know Components		
Mercury	CAS-No. 7439-97-6	Revision Date 2015-11-23
California Prop. 65 Components		

WARNING: This product contains a chemical known to the CAS-No. Revision Date State of California to cause birth defects or other reproductive 7439-97-6 2013-12-20 harm. Mercury

## **16. OTHER INFORMATION**

#### Full text of H-Statements referred to under sections 2 and 3.

0 0

Acute toxicity
Acute aquatic toxicity
Chronic aquatic toxicity
Fatal if inhaled.
May damage fertility or the unborn child.
Causes damage to organs through prolonged or repeated exposure.
Very toxic to aquatic life.
Very toxic to aquatic life with long lasting effects.
Reproductive toxicity

#### HMIS Rating

Health hazard:	2
Chronic Health Hazard:	*
Flammability:	0
Physical Hazard	0
NFPA Rating	
Health hazard:	2

Health hazard:	
Fire Hazard:	
Reactivity Hazard:	

#### Further information

Copyright 2016 Sigma-Aldrich Co. LLC. License granted to make unlimited paper copies for internal use only. The above information is believed to be correct but does not purport to be all inclusive and shall be used only as a guide. The information in this document is based on the present state of our knowledge and is applicable to the product with regard to appropriate safety precautions. It does not represent any guarantee of the properties of the product. Sigma-Aldrich Corporation and its Affiliates shall not be held liable for any damage resulting from handling or from contact with the above product. See www.sigma-aldrich.com and/or the reverse side of invoice or packing slip for additional terms and conditions of sale.

#### **Preparation Information**

Sigma-Aldrich Corporation Product Safety – Americas Region 1-800-521-8956

Version: 3.15

Revision Date: 03/05/2018

Print Date: 06/28/2019

# SIGMA-ALDRICH

# SAFETY DATA SHEET

Version 5.8 Revision Date 07/28/2018 Print Date 06/28/2019

#### **1. PRODUCT AND COMPANY IDENTIFICATION**

1.1	Product identifiers Product name	:	Phenol
	Product Number Brand Index-No.	:	P1037 Sigma-Aldrich 604-001-00-2
	CAS-No.	:	108-95-2

#### 1.2 Relevant identified uses of the substance or mixture and uses advised against

Identified uses : Laboratory chemicals, Synthesis of substances

#### 1.3 Details of the supplier of the safety data sheet

Company	:	Sigma-Aldrich 3050 Spruce Street SAINT LOUIS MO 63103 USA
Telephone Fax	:	+1 800-325-5832 +1 800-325-5052

#### 1.4 Emergency telephone number

Emergency Phone # : +1-703-527-3887 (CHEMTREC)

#### 2. HAZARDS IDENTIFICATION

#### 2.1 Classification of the substance or mixture

#### GHS Classification in accordance with 29 CFR 1910 (OSHA HCS)

Acute toxicity, Oral (Category 3), H301 Acute toxicity, Inhalation (Category 3), H331 Acute toxicity, Dermal (Category 3), H311 Skin corrosion (Category 1B), H314 Serious eye damage (Category 1), H318 Germ cell mutagenicity (Category 2), H341 Specific target organ toxicity - repeated exposure (Category 2), H373 Acute aquatic toxicity (Category 3), H402 Chronic aquatic toxicity (Category 2), H411

For the full text of the H-Statements mentioned in this Section, see Section 16.

Danger

#### 2.2 GHS Label elements, including precautionary statements

Pictogram



Signal word

Hazard statement(s)	
H301 + H311 + H331	Toxic if swallowed, in contact with skin or if inhaled.
H314	Causes severe skin burns and eye damage.
H341	Suspected of causing genetic defects.
H373	May cause damage to organs through prolonged or repeated exposure
H402	Harmful to aquatic life.
H411	Toxic to aquatic life with long lasting effects.

Precautionary statement(s)	
P201	Obtain special instructions before use.
P202	Do not handle until all safety precautions have been read and
	understood.
P260	Do not breathe dust/ fume/ gas/ mist/ vapours/ spray.
P264	Wash skin thoroughly after handling.
P270	Do not eat, drink or smoke when using this product.
P271	Use only outdoors or in a well-ventilated area.
P273	Avoid release to the environment.
P280	Wear protective gloves/ protective clothing/ eye protection/ face protection.
P301 + P310 + P330	IF SWALLOWED: Immediately call a POISON CENTER/doctor. Rinse mouth.
P301 + P330 + P331	IF SWALLOWED: Rinse mouth. Do NOT induce vomiting.
P303 + P361 + P353	IF ON SKIN (or hair): Take off immediately all contaminated clothing. Rinse skin with water/shower.
P304 + P340 + P310	IF INHALED: Remove person to fresh air and keep comfortable for breathing. Immediately call a POISON CENTER/doctor.
P305 + P351 + P338 + P310	IF IN EYES: Rinse cautiously with water for several minutes. Remove contact lenses, if present and easy to do. Continue rinsing. Immediately call a POISON CENTER/doctor.
P308 + P313	IF exposed or concerned: Get medical advice/ attention.
P362	Take off contaminated clothing and wash before reuse.
P391	Collect spillage.
P403 + P233	Store in a well-ventilated place. Keep container tightly closed.
P405	Store locked up.
P501	Dispose of contents/ container to an approved waste disposal plant.

## Hazards not otherwise classified (HNOC) or not covered by GHS Vesicant., Rapidly absorbed through skin. 2.3

#### **3. COMPOSITION/INFORMATION ON INGREDIENTS**

#### 3.1 Substances Synonyms

Synonyms	:	Hydroxybenzene
Formula	:	C <sub>6</sub> H <sub>6</sub> O
Molecular weight	:	94.11 g/mol
CAS-No.	:	108-95-2
EC-No.	:	203-632-7
Index-No.	:	604-001-00-2
Registration number	:	01-2119471329-32-XXXX

#### Hazardous components

omponent Classification		Concentration
Phenol		
	Acute Tox. 3; Skin Corr. 1B;	90 - 100 %
	Eye Dam. 1; Muta. 2; STOT	
	RE 2; Aquatic Acute 3; Aquatic	
	Chronic 2; H301 + H311 +	
	H331, H314, H341, H373,	
	H402, H411	

For the full text of the H-Statements mentioned in this Section, see Section 16.

#### 4. FIRST AID MEASURES

#### 4.1 Description of first aid measures

#### **General advice**

Consult a physician. Show this safety data sheet to the doctor in attendance. Move out of dangerous area.

#### If inhaled

If breathed in, move person into fresh air. If not breathing, give artificial respiration. Consult a physician.

#### In case of skin contact

Take off contaminated clothing and shoes immediately. Wash off with soap and plenty of water. Take victim immediately to hospital. Consult a physician.

#### In case of eye contact

Rinse thoroughly with plenty of water for at least 15 minutes and consult a physician.Continue rinsing eyes during transport to hospital.

#### If swallowed

Do NOT induce vomiting. Never give anything by mouth to an unconscious person. Rinse mouth with water. Consult a physician.

#### **4.2 Most important symptoms and effects, both acute and delayed** The most important known symptoms and effects are described in the labelling (see section 2.2) and/or in section 11

#### **4.3 Indication of any immediate medical attention and special treatment needed** No data available

#### **5. FIREFIGHTING MEASURES**

#### 5.1 Extinguishing media

#### Suitable extinguishing media

Use water spray, alcohol-resistant foam, dry chemical or carbon dioxide.

5.2 Special hazards arising from the substance or mixture No data available

# 5.3 Advice for firefighters

Wear self-contained breathing apparatus for firefighting if necessary.

#### 5.4 Further information No data available

#### 6. ACCIDENTAL RELEASE MEASURES

#### 6.1 Personal precautions, protective equipment and emergency procedures

Wear respiratory protection. Avoid dust formation. Avoid breathing vapours, mist or gas. Ensure adequate ventilation. Evacuate personnel to safe areas. Avoid breathing dust. For personal protection see section 8.

#### 6.2 Environmental precautions

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

#### 6.3 Methods and materials for containment and cleaning up

Pick up and arrange disposal without creating dust. Sweep up and shovel. Keep in suitable, closed containers for disposal.

#### 6.4 Reference to other sections

For disposal see section 13.

#### 7. HANDLING AND STORAGE

#### 7.1 Precautions for safe handling

Avoid contact with skin and eyes. Avoid formation of dust and aerosols. Further processing of solid materials may result in the formation of combustible dusts. The potential for combustible dust formation should be taken into consideration before additional processing occurs.

Provide appropriate exhaust ventilation at places where dust is formed. For precautions see section 2.2.

#### 7.2 Conditions for safe storage, including any incompatibilities

Keep container tightly closed in a dry and well-ventilated place.

Recommended storage temperature 2 - 8 °C

Light sensitive. Handle and store under inert gas. Storage class (TRGS 510): 6.1B: Non-combustible, acute toxic Cat. 1 and 2 / very toxic hazardous materials

#### 7.3 Specific end use(s)

Apart from the uses mentioned in section 1.2 no other specific uses are stipulated

#### 8. EXPOSURE CONTROLS/PERSONAL PROTECTION

#### 8.1 Control parameters

#### Components with workplace control parameters

Component	CÁS-No.	Value	Control parameters	Basis
Phenol	108-95-2	TWA	5 ppm	USA. ACGIH Threshold Limit Values (TLV)
	Remarks	Central Nerv Upper Respir Lung damag Substances f (see BEI® se Not classifiat	ous System impair ratory Tract irritatic e for which there is a ection) ole as a human car	ment n Biological Exposure Index or Indices rcinogen
		Danger of cu	taneous absorptio	n
		TWA	5 ppm 19 mg/m3	USA. NIOSH Recommended Exposure Limits
		Potential for dermal absorption		
		С	15.6 ppm 60 mg/m3	USA. NIOSH Recommended Exposure Limits
		Potential for dermal absorption 15 minute ceiling value		·
		TWA	5 ppm 19 mg/m3	USA. Occupational Exposure Limits (OSHA) - Table Z-1 Limits for Air Contaminants
		Skin designation The value in mg/m3 is approximate.		
		PEL	5 ppm 19 mg/m3	California permissible exposure limits for chemical contaminants (Title 8, Article 107)
		Skin		

#### **Biological occupational exposure limits**

Component	CAS-No.	Parameters	Value	Biological specimen	Basis
Aromatic compound	-	Phenol	250mg/g Creatinine	Urine	ACGIH - Biological Exposure Indices (BEI)
	Remarks	End of shift (As soon as possible after exposure ceases)			

#### 8.2 Exposure controls

#### Appropriate engineering controls

Avoid contact with skin, eyes and clothing. Wash hands before breaks and immediately after handling the product.

#### Personal protective equipment

#### **Eye/face protection**

Face shield and safety glasses Use equipment for eye protection tested and approved under appropriate government standards such as NIOSH (US) or EN 166(EU).

#### **Skin protection**

Handle with gloves. Gloves must be inspected prior to use. Use proper glove removal technique (without touching glove's outer surface) to avoid skin contact with this product. Dispose of contaminated gloves after use in accordance with applicable laws and good laboratory practices. Wash and dry hands.

Full contact Material: butyl-rubber Minimum layer thickness: 0.3 mm Break through time: 480 min Material tested:Butoject® (KCL 897 / Aldrich Z677647, Size M)

Splash contact Material: Nitrile rubber Minimum layer thickness: 0.2 mm Break through time: 30 min Material tested:Dermatril® P (KCL 743 / Aldrich Z677388, Size M)

data source: KCL GmbH, D-36124 Eichenzell, phone +49 (0)6659 87300, e-mail sales@kcl.de, test method: EN374

If used in solution, or mixed with other substances, and under conditions which differ from EN 374, contact the supplier of the CE approved gloves. This recommendation is advisory only and must be evaluated by an industrial hygienist and safety officer familiar with the specific situation of anticipated use by our customers. It should not be construed as offering an approval for any specific use scenario.

#### **Body Protection**

Complete suit protecting against chemicals, The type of protective equipment must be selected according to the concentration and amount of the dangerous substance at the specific workplace.

#### **Respiratory protection**

Where risk assessment shows air-purifying respirators are appropriate use a full-face particle respirator type N100 (US) or type P3 (EN 143) respirator cartridges as a backup to engineering controls. If the respirator is the sole means of protection, use a full-face supplied air respirator. Use respirators and components tested and approved under appropriate government standards such as NIOSH (US) or CEN (EU).

#### **Control of environmental exposure**

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

#### 9. PHYSICAL AND CHEMICAL PROPERTIES

9.1 Information on basic physical and chemical properties

a)	Appearance	Form: solid
b)	Odour	No data available
c)	Odour Threshold	No data available
d)	рН	6.0
e)	Melting point/freezing point	Melting point/range: 38 - 43 °C (100 - 109 °F)
f)	Initial boiling point and boiling range	182.0 °C (359.6 °F)
g)	Flash point	79.0 °C (174.2 °F) - closed cup
h)	Evaporation rate	No data available
i)	Flammability (solid, gas)	No data available
j)	Upper/lower flammability or explosive limits	Upper explosion limit: 8.6 %(V) Lower explosion limit: 1.7 %(V)
k)	Vapour pressure	6.3 hPa (4.7 mmHg) at 55.0 °C (131.0 °F) 0.5 hPa (0.4 mmHg) at 20.0 °C (68.0 °F)
I)	Vapour density	No data available
m)	Relative density	1.07 g/cm3

n)	Water solubility	84 g/l at 20 °C (68 °F)
o)	Partition coefficient: n- octanol/water	log Pow: 1.46
p)	Auto-ignition temperature	715.0 °C (1,319.0 °F)
q)	Decomposition temperature	No data available
r)	Viscosity	No data available
s)	Explosive properties	No data available
t)	Oxidizing properties	No data available
Othe	r safety information	
	Surface tension	38.2 mN/m at 50.0 °C (122.0 °F)

#### **10. STABILITY AND REACTIVITY**

# 10.1 Reactivity

9.2

No data available

#### **10.2 Chemical stability** Stable under recommended storage conditions.

#### **10.3 Possibility of hazardous reactions** No data available

**10.4 Conditions to avoid** No data available

#### **10.5 Incompatible materials** Strong oxidizing agents, Strong bases, Strong acids

# 10.6 Hazardous decomposition products

Hazardous decomposition products formed under fire conditions. - Carbon oxides Other decomposition products - No data available In the event of fire: see section 5

## **11. TOXICOLOGICAL INFORMATION**

#### 11.1 Information on toxicological effects

#### Acute toxicity LD50 Oral - Rat - 410.0 - 650.0 mg/kg

LD50 Oral - Rat - 317.0 mg/kg Remarks: Behavioral:Convulsions or effect on seizure threshold.

LC50 Inhalation - Rat - 8 h - 900 mg/m3

LD50 Dermal - Rabbit - 630.0 mg/kg

No data available

## Skin corrosion/irritation

Skin - Rabbit Result: Severe skin irritation - 24 h

#### Serious eye damage/eye irritation Eyes - Rabbit

Result: Corrosive (OECD Test Guideline 405)

**Respiratory or skin sensitisation** No data available

Germ cell mutagenicity In vitro tests showed mutagenic effects
#### Carcinogenicity

This product is or contains a component that is not classifiable as to its carcinogenicity based on its IARC, ACGIH, NTP, or EPA classification.

- IARC: No component of this product present at levels greater than or equal to 0.1% is identified as probable, possible or confirmed human carcinogen by IARC.
- No component of this product present at levels greater than or equal to 0.1% is identified as a NTP: known or anticipated carcinogen by NTP.
- OSHA: No component of this product present at levels greater than or equal to 0.1% is on OSHA's list of regulated carcinogens.

#### **Reproductive toxicity**

No data available

No data available

Specific target organ toxicity - single exposure No data available

#### Specific target organ toxicity - repeated exposure

May cause damage to organs through prolonged or repeated exposure.

#### Aspiration hazard No data available

#### Additional Information

RTECS: SJ3325000

Material is extremely destructive to tissue of the mucous membranes and upper respiratory tract, eyes, and skin., spasm, inflammation and edema of the larynx, spasm, inflammation and edema of the bronchi, pneumonitis, pulmonary edema, burning sensation, Cough, wheezing, laryngitis, Shortness of breath, Headache, Nausea, Vomiting, Circulatory collapse, tachypnea, paralysis, Convulsions, Coma., necrosis of mouth and G.I. Tract, Jaundice, respiratory failure, cardiac arrest To the best of our knowledge, the chemical, physical, and toxicological properties have not been thoroughly investigated.

Stomach - Irregularities - Based on Human Evidence Stomach - Irregularities - Based on Human Evidence

#### **12. ECOLOGICAL INFORMATION**

#### 12.1 Toxicity

12.3

	Toxicity to fish	LC50 - Leuciscus idus (Golden orfe) - 14.00 - 25.00 mg/l - 48 h
		LC50 - Carassius auratus (goldfish) - 36.10 - 68.80 mg/l  - 96 h
	Toxicity to daphnia and other aquatic invertebrates	EC50 - Daphnia magna (Water flea) - 56 mg/l  - 48 h
	Toxicity to algae	EC50 - Chlorella vulgaris (Fresh water algae) - 370.00 mg/l - 96 h
12.2	Persistence and degrada Biodegradability	bility Result: - Readily biodegradable.
2.3 E	Bioaccumulative potentia Bioaccumulation	l Danio rerio (zebra fish) - 5 h - 2 mg/l
		Bioconcentration factor (BCF): 17.5 Remarks: Does not bioaccumulate.

12.4 Mobility in soil

No data available

#### 12.5 Results of PBT and vPvB assessment

PBT/vPvB assessment not available as chemical safety assessment not required/not conducted

#### 12.6 Other adverse effects

An environmental hazard cannot be excluded in the event of unprofessional handling or disposal. Toxic to aquatic life with long lasting effects.

### **13. DISPOSAL CONSIDERATIONS**

#### 13.1 Waste treatment methods

#### Product

Contact a licensed professional waste disposal service to dispose of this material. Dissolve or mix the material with a combustible solvent and burn in a chemical incinerator equipped with an afterburner and scrubber. Offer surplus and non-recyclable solutions to a licensed disposal company.

#### **Contaminated packaging**

Dispose of as unused product.

#### **14. TRANSPORT INFORMATION**

<b>DOT (US)</b> UN number: 1671 ( Proper shipping name: P Reportable Quantity (RQ Poison Inhalation Hazard	Class: 6.1 Yhenol, solid \): 1000 lbs d: No	Packing group: II		
IMDG UN number: 1671 ( Proper shipping name: P Marine pollutant:yes	Class: 6.1 'HENOL, SOLID	Packing group: II	EMS-N	o: F-A, S-A
<b>IATA</b> UN number: 1671 ( Proper shipping name: P	Class: 6.1 'henol, solid	Packing group: II		
15. REGULATORY INFORMA	TION			
SARA 302 Components				
The following components	are subject to reporting	levels established b	y SARA Title III, S	Section 302:
Phenol			AS-No. )8-95-2	Revision Date 2007-07-01
SARA 313 Components				
The following components	are subject to reporting	levels established b	y SARA Title III, S	Section 313:
		C	AS-No.	Revision Date
Phenol		10	08-95-2	2007-07-01
SARA 311/312 Hazards Acute Health Hazard, Chro	onic Health Hazard			
Massachusetts Right To	Know Components			
5	•	C	AS-No.	Revision Date
Phenol		10	08-95-2	2007-07-01
Pennsylvania Right To K	(now Components			
Phenol	·	C. 10	AS-No. 08-95-2	Revision Date 2007-07-01
California Prop. 65 Com This product does not con	p <b>onents</b> tain any chemicals know	n to State of Califorr	nia to cause canc	er, birth defects, or any other

reproductive harm.

### **16. OTHER INFORMATION**

Full text of H-Statements referred to under sections 2 and 3.

Acute Tox.	Acute toxicity
Aquatic Acute	Acute aquatic toxicity
Aquatic Chronic	Chronic aquatic toxicity
Eye Dam.	Serious eye damage
H301	Toxic if swallowed.
H301 + H311 +	Toxic if swallowed, in contact with skin or if inhaled.
H331	
H311	Toxic in contact with skin.
H314	Causes severe skin burns and eye damage.
H318	Causes serious eye damage.
H331	Toxic if inhaled.
H341	Suspected of causing genetic defects.
H373	May cause damage to organs through prolonged or repeated exposure.

Further information

Copyright 2016 Sigma-Aldrich Co. LLC. License granted to make unlimited paper copies for internal use only. The above information is believed to be correct but does not purport to be all inclusive and shall be used only as a guide. The information in this document is based on the present state of our knowledge and is applicable to the product with regard to appropriate safety precautions. It does not represent any guarantee of the properties of the product. Sigma-Aldrich Corporation and its Affiliates shall not be held liable for any damage resulting from handling or from contact with the above product. See www.sigma-aldrich.com and/or the reverse side of invoice or packing slip for additional terms and conditions of sale.

#### **Preparation Information**

Sigma-Aldrich Corporation Product Safety – Americas Region 1-800-521-8956

Version: 5.8

Revision Date: 07/28/2018

Print Date: 06/28/2019

# SIGMA-ALDRICH

## SAFETY DATA SHEET

Version 4.10 Revision Date 01/04/2018 Print Date 06/28/2019

#### **1. PRODUCT AND COMPANY IDENTIFICATION**

1.1	<b>Product identifiers</b> Product name	:	Trichloroethylene
	Product Number Brand Index-No.	:	251402 Sigma-Aldrich 602-027-00-9
	CAS-No.	:	79-01-6
1.2	Relevant identified uses	of the	substance or mixture and uses adv

## Relevant identified uses of the substance or mixture and uses advised against

Identified uses : Laboratory chemicals, Synthesis of substances

#### 1.3 Details of the supplier of the safety data sheet

Company	:	Sigma-Aldrich 3050 Spruce Street SAINT LOUIS MO 63103 USA
Telephone Fax	:	+1 800-325-5832 +1 800-325-5052

#### 1.4 Emergency telephone number

Emergency Phone # : +1-703-527-3887 (CHEMTREC)

#### 2. HAZARDS IDENTIFICATION

#### 2.1 Classification of the substance or mixture

#### GHS Classification in accordance with 29 CFR 1910 (OSHA HCS)

Skin irritation (Category 2), H315 Eye irritation (Category 2A), H319 Germ cell mutagenicity (Category 2), H341 Carcinogenicity (Category 1B), H350 Specific target organ toxicity - single exposure (Category 3), Central nervous system, H336 Acute aquatic toxicity (Category 3), H402 Chronic aquatic toxicity (Category 3), H412

For the full text of the H-Statements mentioned in this Section, see Section 16.

#### 2.2 GHS Label elements, including precautionary statements

Pictogram



Signal word

Danger

Hazard statement(s)	
H315	Causes skin irritation.
H319	Causes serious eye irritation.
H336	May cause drowsiness or dizziness.
H341	Suspected of causing genetic defects.
H350	May cause cancer.
H412	Harmful to aquatic life with long lasting effects.

Precautionary statement(s)	
P201	Obtain special instructions before use.
P202	Do not handle until all safety precautions have been read and understood.
P261	Avoid breathing dust/ fume/ gas/ mist/ vapours/ spray.
P264	Wash skin thoroughly after handling.
P271	Use only outdoors or in a well-ventilated area.
P273	Avoid release to the environment.
P280	Wear eye protection/ face protection.
P280	Wear protective gloves.
P281	Use personal protective equipment as required.
P302 + P352	IF ON SKIN: Wash with plenty of soap and water.
P304 + P340 + P312	IF INHALED: Remove victim to fresh air and keep at rest in a position comfortable for breathing. Call a POISON CENTER or doctor/ physician if you feel unwell.
P305 + P351 + P338	IF IN EYES: Rinse cautiously with water for several minutes. Remove contact lenses, if present and easy to do. Continue rinsing.
P308 + P313	IF exposed or concerned: Get medical advice/ attention.
P332 + P313	If skin irritation occurs: Get medical advice/ attention.
P337 + P313	If eye irritation persists: Get medical advice/ attention.
P362	Take off contaminated clothing and wash before reuse.
P403 + P233	Store in a well-ventilated place. Keep container tightly closed.
P405	Store locked up.
P501	Dispose of contents/ container to an approved waste disposal plant.

2.3 Hazards not otherwise classified (HNOC) or not covered by GHS - none

### **3. COMPOSITION/INFORMATION ON INGREDIENTS**

#### 3.1 Substances

Synonyms	:	TCE Trichloroethene	
Formula	:	C <sub>2</sub> HCl <sub>3</sub>	
Molecular weight	:	131.39 g/mol	
CAS-No.	:	79-01-6	
EC-No.	:	201-167-4	
Index-No.	:	602-027-00-9	

#### Hazardous components

Component	Classification	Concentration
Trichloroethylene		
	Skin Irrit. 2; Eye Irrit. 2A; Muta. 2; Carc. 1B; STOT SE 3; Aquatic Acute 3; Aquatic Chronic 3; H315, H319, H336, H341, H350, H412	90 - 100 %

For the full text of the H-Statements mentioned in this Section, see Section 16.

### 4. FIRST AID MEASURES

#### 4.1 Description of first aid measures

#### General advice

Consult a physician. Show this safety data sheet to the doctor in attendance. Move out of dangerous area.

#### If inhaled

If breathed in, move person into fresh air. If not breathing, give artificial respiration. Consult a physician.

#### In case of skin contact

Wash off with soap and plenty of water. Consult a physician.

#### In case of eye contact

Rinse thoroughly with plenty of water for at least 15 minutes and consult a physician.

#### If swallowed

Never give anything by mouth to an unconscious person. Rinse mouth with water. Consult a physician.

### **4.2** Most important symptoms and effects, both acute and delayed The most important known symptoms and effects are described in the labelling (see section 2.2) and/or in section 11

**4.3** Indication of any immediate medical attention and special treatment needed No data available

#### **5. FIREFIGHTING MEASURES**

#### 5.1 Extinguishing media

#### Suitable extinguishing media

Use water spray, alcohol-resistant foam, dry chemical or carbon dioxide.

5.2 Special hazards arising from the substance or mixture No data available

#### 5.3 Advice for firefighters

Wear self-contained breathing apparatus for firefighting if necessary.

5.4 Further information No data available

### 6. ACCIDENTAL RELEASE MEASURES

#### 6.1 Personal precautions, protective equipment and emergency procedures

Use personal protective equipment. Avoid breathing vapours, mist or gas. Ensure adequate ventilation. Evacuate personnel to safe areas.

For personal protection see section 8.

#### 6.2 Environmental precautions

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

6.3 Methods and materials for containment and cleaning up

Soak up with inert absorbent material and dispose of as hazardous waste. Keep in suitable, closed containers for disposal.

6.4 Reference to other sections

For disposal see section 13.

#### 7. HANDLING AND STORAGE

#### 7.1 Precautions for safe handling

Avoid contact with skin and eyes. Avoid inhalation of vapour or mist. For precautions see section 2.2.

#### 7.2 Conditions for safe storage, including any incompatibilities

Keep container tightly closed in a dry and well-ventilated place. Containers which are opened must be carefully resealed and kept upright to prevent leakage.

Light sensitive. Handle and store under inert gas. Storage class (TRGS 510): 6.1D: Non-combustible, acute toxic Cat.3 / toxic hazardous materials or hazardous materials causing chronic effects

#### 7.3 Specific end use(s)

Apart from the uses mentioned in section 1.2 no other specific uses are stipulated

## 8. EXPOSURE CONTROLS/PERSONAL PROTECTION

### 8.1 Control parameters

### Components with workplace control parameters

Component	CAS-No.	Value	Control parameters	Basis		
Trichloroethylene	79-01-6	TWA	10.000000 ppm	USA. ACGIH Threshold Limit Values (TLV)		
	Remarks	Central Nervous System impairment cognitive decrement Renal toxicity				
		(see BEI® section)				
		STEL	25.000000 ppm	USA. ACGIH Threshold Limit Values (TLV)		
		Central Nervous System impairment cognitive decrement Renal toxicity Substances for which there is a Biological Exposure Index or Indice (see BEI® section)				
		Potential Occupational Carcinogen See Appendix C See Appendix A				
		See Table Z	-2			
		TWA	100.000000 ppm	USA. Occupational Exposure Limits (OSHA) - Table Z-2		
		Z37.19-1967	7			
		CEIL	200.000000 ppm	USA. Occupational Exposure Limits (OSHA) - Table Z-2		
		Z37,19-1967	7			
		Peak	300.000000 ppm	USA. Occupational Exposure Limits (OSHA) - Table Z-2		
		Z37.19-1967	7			
		TWA	100 ppm	USA. Occupational Exposure Limits (OSHA) - Table Z-2		
		Z37.19-1967				
		CEIL	200 ppm	USA. Occupational Exposure Limits (OSHA) - Table Z-2		
		Z37.19-1967	7	••••••		
		Peak	300 ppm	USA. Occupational Exposure Limits (OSHA) - Table Z-2		
		Z37.19-1967	7			

STEL	100 ppm 537 mg/m3	California permissible exposure limits for chemical contaminants (Title 8, Article 107)
C	300 ppm	California permissible exposure limits for chemical contaminants (Title 8, Article 107)
PEL	25 ppm 135 mg/m3	California permissible exposure limits for chemical contaminants (Title 8, Article 107)

#### **Biological occupational exposure limits**

Component	CAS-No.	Parameters	Value	Biological specimen	Basis
	-	Trichloroaceti c acid	15.0000 mg/l	Urine	ACGIH - Biological Exposure Indices (BEI)
	Remarks	End of shift at end of workweek			
		Trichloroetha nol	0.5000 mg/l	In blood	ACGIH - Biological Exposure Indices (BEI)
		End of shift at e	end of workv	veek	
		Trichloroethyl ene		In blood	ACGIH - Biological Exposure Indices (BEI)
		End of shift at e	end of workv	veek	
		Trichloroethyl ene		In end-exhaled air	ACGIH - Biological Exposure Indices (BEI)
		End of shift at e	end of workv	veek	

#### 8.2 Exposure controls

#### Appropriate engineering controls

Handle in accordance with good industrial hygiene and safety practice. Wash hands before breaks and at the end of workday.

#### Personal protective equipment

#### **Eye/face protection**

Safety glasses with side-shields conforming to EN166 Use equipment for eye protection tested and approved under appropriate government standards such as NIOSH (US) or EN 166(EU).

#### Skin protection

Handle with gloves. Gloves must be inspected prior to use. Use proper glove removal technique (without touching glove's outer surface) to avoid skin contact with this product. Dispose of contaminated gloves after use in accordance with applicable laws and good laboratory practices. Wash and dry hands.

Full contact Material: Fluorinated rubber Minimum layer thickness: 0.7 mm Break through time: 480 min Material tested:Vitoject® (KCL 890 / Aldrich Z677698, Size M)

Splash contact Material: Fluorinated rubber Minimum layer thickness: 0.7 mm Break through time: 480 min Material tested:Vitoject® (KCL 890 / Aldrich Z677698, Size M)

data source: KCL GmbH, D-36124 Eichenzell, phone +49 (0)6659 87300, e-mail sales@kcl.de, test method: EN374

If used in solution, or mixed with other substances, and under conditions which differ from EN 374, contact the supplier of the CE approved gloves. This recommendation is advisory only and must be evaluated by an industrial hygienist and safety officer familiar with the specific situation of anticipated use by our customers. It should not be construed as offering an approval for any specific use scenario.

#### **Body Protection**

Complete suit protecting against chemicals, The type of protective equipment must be selected according to the concentration and amount of the dangerous substance at the specific workplace.

#### **Respiratory protection**

Where risk assessment shows air-purifying respirators are appropriate use a full-face respirator with multipurpose combination (US) or type ABEK (EN 14387) respirator cartridges as a backup to engineering controls. If the respirator is the sole means of protection, use a full-face supplied air respirator. Use respirators and components tested and approved under appropriate government standards such as NIOSH (US) or CEN (EU).

#### **Control of environmental exposure**

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

#### 9. PHYSICAL AND CHEMICAL PROPERTIES

#### 9.1 Information on basic physical and chemical properties

a)	Appearance	Form: liquid, clear Colour: colourless
b)	Odour	No data available
c)	Odour Threshold	No data available
d)	рН	No data available
e)	Melting point/freezing point	Melting point/range: -84.8 °C (-120.6 °F) - lit.
f)	Initial boiling point and boiling range	86.7 °C (188.1 °F) - lit.
g)	Flash point	No data available
h)	Evaporation rate	No data available
i)	Flammability (solid, gas)	No data available
j)	Upper/lower flammability or explosive limits	Upper explosion limit: 10.5 %(V) Lower explosion limit: 8 %(V)
k)	Vapour pressure	81.3 hPa (61.0 mmHg) at 20.0 °C (68.0 °F)
I)	Vapour density	No data available
m)	Relative density	1.463 g/mL at 25 °C (77 °F)
n)	Water solubility	No data available
o)	Partition coefficient: n- octanol/water	log Pow: 2.29log Pow: 5
p)	Auto-ignition temperature	410.0 °C (770.0 °F)
q)	Decomposition temperature	No data available
r)	Viscosity	No data available
s)	Explosive properties	No data available
t)	Oxidizing properties	No data available
Othe	r safety information	

### No data available

#### **10. STABILITY AND REACTIVITY**

#### 10.1 Reactivity No data available

9.2

#### 10.2 Chemical stability

Stable under recommended storage conditions.

- **10.3 Possibility of hazardous reactions** No data available
- **10.4 Conditions to avoid** No data available

**10.5** Incompatible materials Oxidizing agents, Strong bases, Magnesium

### 10.6 Hazardous decomposition products Hazardous decomposition products formed under fire conditions. - Carbon oxides, Hydrogen chloride gas Other decomposition products - No data available In the event of fire: see section 5

### **11. TOXICOLOGICAL INFORMATION**

### 11.1 Information on toxicological effects

### Acute toxicity

LD50 Oral - Rat - 4,920 mg/kg

LC50 Inhalation - Mouse - 4 h - 8450 ppm

LD50 Dermal - Rabbit - > 20,000 mg/kg

No data available

### Skin corrosion/irritation

Skin - Rabbit Result: Severe skin irritation - 24 h

Serious eye damage/eye irritation Eyes - Rabbit Result: Eye irritation - 24 h

Respiratory or skin sensitisation No data available

### Germ cell mutagenicity

Laboratory experiments have shown mutagenic effects. In vitro tests showed mutagenic effects

### Carcinogenicity

This product is or contains a component that has been reported to be probably carcinogenic based on its IARC, OSHA, ACGIH, NTP, or EPA classification.

Possible human carcinogen

IARC:	1 - Group 1: Carcinogenic to humans (Trichloroethylene)	
-------	---	--

- NTP: RAHC Reasonably anticipated to be a human carcinogen (Trichloroethylene)
- OSHA: No component of this product present at levels greater than or equal to 0.1% is on OSHA's list of regulated carcinogens.

### Reproductive toxicity

No data available

No data available

Specific target organ toxicity - single exposure No data available

Specific target organ toxicity - repeated exposure No data available

Aspiration hazard No data available

### Additional Information

RTECS: KX4550000

burning sensation, Cough, wheezing, laryngitis, Shortness of breath, Headache, Nausea, Vomiting, Exposure to and/or consumption of alcohol may increase toxic effects., Gastrointestinal disturbance, Kidney injury may occur., narcosis To the best of our knowledge, the chemical, physical, and toxicological properties have not been thoroughly investigated.

### **12. ECOLOGICAL INFORMATION**

#### 12.1 Toxicity

Toxicity to fish	LC50 - Pimephales promelas (fathead minnow) - 41 mg/l - 96.0 h			
	LOEC - other fish - 11 mg/l - 10.0 d			
	NOEC - Oryzias latipes - 40 mg/l  - 10.0 d			
Toxicity to daphnia and other aquatic invertebrates	EC50 - Daphnia magna (Water flea) - 18.00 mg/l  - 48 h			

Toxicity to algae IC50 - Pseudokirchneriella subcapitata (green algae) - 175.00 mg/l - 96 h

#### **12.2 Persistence and degradability** No data available

#### **12.3 Bioaccumulative potential** Does not bioaccumulate.

12.4 Mobility in soil

No data available

### 12.5 Results of PBT and vPvB assessment

PBT/vPvB assessment not available as chemical safety assessment not required/not conducted

#### 12.6 Other adverse effects

An environmental hazard cannot be excluded in the event of unprofessional handling or disposal. Harmful to aquatic life with long lasting effects.

An environmental hazard cannot be excluded in the event of unprofessional handling or disposal.

Harmful to aquatic organisms, may cause long-term adverse effects in the aquatic environment.

#### **13. DISPOSAL CONSIDERATIONS**

#### 13.1 Waste treatment methods

#### Product

Offer surplus and non-recyclable solutions to a licensed disposal company. Contact a licensed professional waste disposal service to dispose of this material. Dissolve or mix the material with a combustible solvent and burn in a chemical incinerator equipped with an afterburner and scrubber.

Packing group: III

#### **Contaminated packaging**

Dispose of as unused product.

### **14. TRANSPORT INFORMATION**

#### DOT (US)

UN number: 1710 Class: 6.1 Proper shipping name: Trichloroethylene Reportable Quantity (RQ): 100 lbs Poison Inhalation Hazard: No

IMDG

UN number: 1710 Class: 6.1 Packing group: III Proper shipping name: TRICHLOROETHYLENE

p: III EM

EMS-No: F-A, S-A

### ΙΑΤΑ

UN number: 1710 Class: 6.1

Sigma-Aldrich - 251402

Packing group: III

#### **15. REGULATORY INFORMATION**

#### SARA 302 Components No chemicals in this material are subject to the reporting requirements of SARA Title III, Section 302. SARA 313 Components The following components are subject to reporting levels established by SARA Title III, Section 313: CAS-No. **Revision Date** Trichloroethylene 79-01-6 2007-07-01 SARA 311/312 Hazards Acute Health Hazard, Chronic Health Hazard Massachusetts Right To Know Components CAS-No. Revision Date Trichloroethylene 79-01-6 2007-07-01 Pennsylvania Right To Know Components CAS-No. **Revision Date** Trichloroethylene 79-01-6 2007-07-01 New Jersey Right To Know Components CAS-No. **Revision Date** Trichloroethylene 79-01-6 2007-07-01 California Prop. 65 Components WARNING! This product contains a chemical known to the CAS-No. Revision Date State of California to cause cancer. 79-01-6 2011-09-01 Trichloroethylene CAS-No. WARNING: This product contains a chemical known to the **Revision Date** State of California to cause birth defects or other reproductive 79-01-6 2011-09-01 harm. Trichloroethylene

### **16. OTHER INFORMATION**

### Full text of H-Statements referred to under sections 2 and 3.

Aquatic Acute	Acute aquatic toxicity
Aquatic Chronic	Chronic aquatic toxicity
Carc.	Carcinogenicity
Eye Irrit.	Eye irritation
H315	Causes skin irritation.
H319	Causes serious eye irritation.
H336	May cause drowsiness or dizziness.
H341	Suspected of causing genetic defects
H350	May cause cancer.
H402	Harmful to aquatic life.

0

#### HMIS Rating

Health hazard:	2
Chronic Health Hazard:	*
Flammability:	0
Physical Hazard	0
NFPA Rating	
Health hazard:	2
Fire Hazard:	0

**Reactivity Hazard:** 

#### **Further information**

Copyright 2016 Sigma-Aldrich Co. LLC. License granted to make unlimited paper copies for internal use only. The above information is believed to be correct but does not purport to be all inclusive and shall be used only as a guide. The information in this document is based on the present state of our knowledge and is applicable to the product with regard to appropriate safety precautions. It does not represent any guarantee of the properties of the product. Sigma-Aldrich Corporation and its Affiliates shall not be held liable for any damage resulting from handling or from contact with the above product. See www.sigma-aldrich.com and/or the reverse side of invoice or packing slip for additional terms and conditions of sale.

#### **Preparation Information**

Sigma-Aldrich Corporation Product Safety – Americas Region 1-800-521-8956

Version: 4.10

Revision Date: 01/04/2018

Print Date: 06/28/2019

# SIGMA-ALDRICH

## SAFETY DATA SHEET

Version 4.11 Revision Date 06/28/2017 Print Date 06/22/2019

#### **1. PRODUCT AND COMPANY IDENTIFICATION**

1.1	<b>Product identifiers</b> Product name	:	Tetrachloroethylene
	Product Number Brand Index-No.	:	371696 Sigma-Aldrich 602-028-00-4
	CAS-No.	:	127-18-4
1.2	Relevant identified uses	of the	substance or mixture and uses advised against

Identified uses : Laboratory chemicals, Synthesis of substances

#### 1.3 Details of the supplier of the safety data sheet

Company	:	Sigma-Aldrich 3050 Spruce Street SAINT LOUIS MO 63103 USA
Telephone Fax	:	+1 800-325-5832 +1 800-325-5052

#### 1.4 Emergency telephone number

Emergency Phone # : +1-703-527-3887 (CHEMTREC)

#### 2. HAZARDS IDENTIFICATION

#### 2.1 Classification of the substance or mixture

#### GHS Classification in accordance with 29 CFR 1910 (OSHA HCS)

Skin irritation (Category 2), H315 Eye irritation (Category 2A), H319 Skin sensitisation (Category 1), H317 Carcinogenicity (Category 2), H351 Specific target organ toxicity - single exposure (Category 3), Central nervous system, H336 Acute aquatic toxicity (Category 2), H401 Chronic aquatic toxicity (Category 2), H411

For the full text of the H-Statements mentioned in this Section, see Section 16.

#### 2.2 GHS Label elements, including precautionary statements

Pictogram



Signal word

Warning

Hazard statement(s)	
H315	Causes skin irritation.
H317	May cause an allergic skin reaction.
H319	Causes serious eye irritation.
H336	May cause drowsiness or dizziness.
H351	Suspected of causing cancer.
H411	Toxic to aquatic life with long lasting effects.

Precautionary statement(s)	
P201	Obtain special instructions before use.
P202	Do not handle until all safety precautions have been read and understood.
P261	Avoid breathing dust/ fume/ gas/ mist/ vapours/ spray.
P264	Wash skin thoroughly after handling.
P271	Use only outdoors or in a well-ventilated area.
P272	Contaminated work clothing should not be allowed out of the workplace.
P273	Avoid release to the environment.
P280	Wear protective gloves/ protective clothing/ eye protection/ face protection.
P302 + P352	IF ON SKIN: Wash with plenty of soap and water.
P304 + P340 + P312	IF INHALED: Remove person to fresh air and keep comfortable for breathing. Call a POISON CENTER/doctor if you feel unwell.
P305 + P351 + P338	IF IN EYES: Rinse cautiously with water for several minutes. Remove contact lenses, if present and easy to do. Continue rinsing.
P308 + P313	IF exposed or concerned: Get medical advice/ attention.
P333 + P313	If skin irritation or rash occurs: Get medical advice/ attention.
P337 + P313	If eye irritation persists: Get medical advice/ attention.
P362	Take off contaminated clothing and wash before reuse.
P391	Collect spillage.
P403 + P233	Store in a well-ventilated place. Keep container tightly closed.
P405	Store locked up.
P501	Dispose of contents/ container to an approved waste disposal plant.

2.3 Hazards not otherwise classified (HNOC) or not covered by GHS - none

### **3. COMPOSITION/INFORMATION ON INGREDIENTS**

#### 3.1 Substances

Synonyms	:	Perchloroethylene PCE
Formula	:	C <sub>2</sub> Cl <sub>4</sub>
Molecular weight	:	165.83 g/mol
CAS-No.	:	127-18-4
EC-No.	:	204-825-9

#### Hazardous components

Index-No.

Component	Classification	Concentration	
Tetrachloroethylene			
	Skin Irrit. 2; Eye Irrit. 2A; Skin	90 - 100 %	
	Sens. 1; Carc. 2; STOT SE 3;		
	Aquatic Acute 2; Aquatic		
	Chronic 2; H315, H317, H319,		
	H336, H351, H411		

For the full text of the H-Statements mentioned in this Section, see Section 16.

: 602-028-00-4

### 4. FIRST AID MEASURES

### 4.1 Description of first aid measures

#### **General advice**

Consult a physician. Show this safety data sheet to the doctor in attendance. Move out of dangerous area.

#### If inhaled

If breathed in, move person into fresh air. If not breathing, give artificial respiration. Consult a physician.

#### In case of skin contact

Wash off with soap and plenty of water. Consult a physician.

#### In case of eye contact

Rinse thoroughly with plenty of water for at least 15 minutes and consult a physician.

#### If swallowed

Never give anything by mouth to an unconscious person. Rinse mouth with water. Consult a physician.

### **4.2 Most important symptoms and effects, both acute and delayed** The most important known symptoms and effects are described in the labelling (see section 2.2) and/or in section 11

**4.3** Indication of any immediate medical attention and special treatment needed No data available

#### **5. FIREFIGHTING MEASURES**

#### 5.1 Extinguishing media

#### Suitable extinguishing media

Use water spray, alcohol-resistant foam, dry chemical or carbon dioxide.

5.2 Special hazards arising from the substance or mixture No data available

#### 5.3 Advice for firefighters

Wear self-contained breathing apparatus for firefighting if necessary.

5.4 Further information No data available

#### 6. ACCIDENTAL RELEASE MEASURES

#### 6.1 Personal precautions, protective equipment and emergency procedures

Use personal protective equipment. Avoid breathing vapours, mist or gas. Ensure adequate ventilation. Evacuate personnel to safe areas.

For personal protection see section 8.

#### 6.2 Environmental precautions

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

- 6.3 Methods and materials for containment and cleaning up Soak up with inert absorbent material and dispose of as hazardous waste. Keep in suitable, closed containers for disposal.
- 6.4 Reference to other sections

For disposal see section 13.

#### 7. HANDLING AND STORAGE

#### 7.1 Precautions for safe handling Avoid contact with skin and eyes. Avoid inhalation of vapour or mist. For precautions see section 2.2.

7.2 Conditions for safe storage, including any incompatibilities Keep container tightly closed in a dry and well-ventilated place. Containers which are opened must be carefully resealed and kept upright to prevent leakage.

#### 7.3 Specific end use(s)

Apart from the uses mentioned in section 1.2 no other specific uses are stipulated

#### 8. EXPOSURE CONTROLS/PERSONAL PROTECTION

#### 8.1 Control parameters

#### Components with workplace control parameters

Component	CAS-No.	Value	Control	Basis		
			parameters			
Tetrachloroethylene	127-18-4	TWA	25.000000 ppm	USA. ACGIH Threshold Limit Values (TLV)		
	Remarks	Central Nerv	ous System impair	rment		
		Substances	for which there is a	a Biological Exposure Index or Indices		
		(see BEI® section)				
		Confirmed animal carcinogen with unknown relevance to humans				
		STEL	100.000000	USA. ACGIH Threshold Limit Values		
			ppm	(TLV)		
	1	Central Nerv	ous System impair	rment		
		Substances	for which there is a	a Biological Exposure Index or Indices		
		(see BEI® se	ection)	0		
		Confirmed a	nimal carcinogen v	vith unknown relevance to humans		
		Potential Oc	cupational Carcino	gen		
		Minimize wo	rkplace exposure of	concentrations.		
		See Appendix A				
		See Table Z	-2			
		TWA	100.000000	USA. Occupational Exposure Limits		
			ppm	(OSHA) - Table Z-2		
		CEIL	200.000000	USA. Occupational Exposure Limits		
			ppm	(OSHA) - Table Z-2		
		Peak	300.000000	USA. Occupational Exposure Limits		
			ppm	(OSHA) - Table Z-2		
		TWA	25 ppm	USA. ACGIH Threshold Limit Values		
				[(ILV)		
		Central Nerv	Central Nervous System impairment			
		Substances for which there is a Biological Exposure Index or Indices				
		(see BEI® section)				
			nimai carcinogen v	VIIIn unknown relevance to numans		
		SIEL	100 ppm			
	-	Control Nor	 			
		Central Nerv	ous System impair	ment Biological Exposure Index or Indiana		
		(see BEI® section)				
		(See DENS Security)				
	+	Potential Oc	cupational Carcino			
		Minimize workplace exposure concentrations				
		See Annend	ix A			
	+	See Table 7	_2	<u>`</u>		

TWA	100 ppm	USA. Occupational Exposure Limits (OSHA) - Table Z-2
CEIL	200 ppm	USA. Occupational Exposure Limits (OSHA) - Table Z-2
Peak	300 ppm	USA. Occupational Exposure Limits (OSHA) - Table Z-2
TWA	25 ppm 170 mg/m3	USA. OSHA - TABLE Z-1 Limits for Air Contaminants - 1910.1000
STEL	100 ppm 685 mg/m3	California permissible exposure limits for chemical contaminants (Title 8, Article 107)
С	300 ppm	California permissible exposure limits for chemical contaminants (Title 8, Article 107)
PEL	25 ppm 170 mg/m3	California permissible exposure limits for chemical contaminants (Title 8, Article 107)

#### **Biological occupational exposure limits**

		<b>_</b>		<b>B 1 1 1</b>	
Component	CAS-No.	Parameters	Value	Biological	Basis
				specimen	
Tetrachloroethylene	127-18-4	Tetrachloroet	3ppm	In end-exhaled air	ACGIH - Biological
		hylono	• • • • • • • • • • • • • • • • • • • •		Exposuro Indicos
		пуюте			Exposure mulces
					(BEI)
	Remarks	Prior to shift (1	6 hours after	r exposure ceases)	
		Tetrachloroet	0.5000	In blood	ACGIH - Biological
		hvlene	ma/l		Exposure Indices
		nyiono			
		Prior to shift (1	6 hours after	r exposure ceases)	
		Tetrachloroet	3ppm	In end-exhaled air	ACGIH - Biological
		hylene			Exposure Indices
		Ingione			
					(DEI)
		Prior to shift (1	6 hours after	r exposure ceases)	
		Tetrachloroet	0.5 mg/l	In blood	ACGIH - Biological
		hvlene	, i i i i i i i i i i i i i i i i i i i		Exposure Indices
				L	
		Prior to shift (16 hours after exposure ceases)			

#### 8.2 Exposure controls

#### Appropriate engineering controls

Handle in accordance with good industrial hygiene and safety practice. Wash hands before breaks and at the end of workday.

#### Personal protective equipment

#### **Eye/face protection**

Face shield and safety glasses Use equipment for eye protection tested and approved under appropriate government standards such as NIOSH (US) or EN 166(EU).

#### Skin protection

Handle with gloves. Gloves must be inspected prior to use. Use proper glove removal technique (without touching glove's outer surface) to avoid skin contact with this product. Dispose of contaminated gloves after use in accordance with applicable laws and good laboratory practices. Wash and dry hands.

Full contact Material: Fluorinated rubber Minimum layer thickness: 0.7 mm Break through time: 480 min Material tested:Vitoject® (KCL 890 / Aldrich Z677698, Size M)

Splash contact Material: Nitrile rubber Minimum layer thickness: 0.2 mm Break through time: 49 min Material tested:Dermatril® P (KCL 743 / Aldrich Z677388, Size M)

data source: KCL GmbH, D-36124 Eichenzell, phone +49 (0)6659 87300, e-mail sales@kcl.de, test method: EN374

If used in solution, or mixed with other substances, and under conditions which differ from EN 374, contact the supplier of the CE approved gloves. This recommendation is advisory only and must be evaluated by an industrial hygienist and safety officer familiar with the specific situation of anticipated use by our customers. It should not be construed as offering an approval for any specific use scenario.

#### **Body Protection**

Complete suit protecting against chemicals, The type of protective equipment must be selected according to the concentration and amount of the dangerous substance at the specific workplace.

#### **Respiratory protection**

Where risk assessment shows air-purifying respirators are appropriate use a full-face respirator with multipurpose combination (US) or type ABEK (EN 14387) respirator cartridges as a backup to engineering controls. If the respirator is the sole means of protection, use a full-face supplied air respirator. Use respirators and components tested and approved under appropriate government standards such as NIOSH (US) or CEN (EU).

#### **Control of environmental exposure**

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

### 9. PHYSICAL AND CHEMICAL PROPERTIES

#### 9.1 Information on basic physical and chemical properties

a)	Appearance	Form: liquid, clear Colour: colourless
b)	Odour	No data available
c)	Odour Threshold	No data available
d)	рН	No data available
e)	Melting point/freezing point	Melting point/range: -22 °C (-8 °F) - lit.
f)	Initial boiling point and boiling range	121 °C (250 °F) - lit.
g)	Flash point	No data available
h)	Evaporation rate	No data available
i)	Flammability (solid, gas)	No data available
j)	Upper/lower flammability or explosive limits	No data available
k)	Vapour pressure	25.3 hPa (19.0 mmHg) at 25.0 °C (77.0 °F) 17.3 hPa (13.0 mmHg) at 20.0 °C (68.0 °F)
I)	Vapour density	No data available
m)	Relative density	1.623 g/cm3 at 25 °C (77 °F)
n)	Water solubility	0.15 g/l at 25 °C (77 °F)
o)	Partition coefficient: n- octanol/water	log Pow: 2.53 at 23 °C (73 °F)
p)	Auto-ignition temperature	No data available
q)	Decomposition temperature	No data available
r)	Viscosity	No data available
s)	Explosive properties	No data available

- t) Oxidizing properties
- No data available

#### 9.2 Other safety information

Surface tension

32.1 mN/m at 20 °C (68 °F)

#### **10. STABILITY AND REACTIVITY**

### 10.1 Reactivity No data available

## 10.2 Chemical stability

Stable under recommended storage conditions.

- **10.3 Possibility of hazardous reactions** No data available
- **10.4 Conditions to avoid** No data available
- **10.5** Incompatible materials Strong oxidizing agents, Strong bases

#### 10.6 Hazardous decomposition products Hazardous decomposition products formed under fire conditions. - Carbon oxides, Hydrogen chloride gas Other decomposition products - No data available In the event of fire: see section 5

### **11. TOXICOLOGICAL INFORMATION**

#### 11.1 Information on toxicological effects

#### Acute toxicity

LD50 Oral - Rat - female - 3,385 mg/kg (OECD Test Guideline 401)

Inhalation: No data available

Dermal: No data available

No data available

#### Skin corrosion/irritation

Skin - Rabbit Result: Skin irritation - 4 h (OECD Test Guideline 404)

#### Serious eye damage/eye irritation

Eyes - Rabbit Result: Mild eye irritation - 24 h

#### Respiratory or skin sensitisation

- Mouse Result: May cause sensitisation by skin contact. (OECD Test Guideline 429)

### Germ cell mutagenicity

Hamster ovary Result: negative

OECD Test Guideline 474 Mouse - male Result: negative

#### Carcinogenicity

Limited evidence of carcinogenicity in animal studies

#### NTP: RAHC - Reasonably anticipated to be a human carcinogen (Tetrachloroethylene)

No component of this product present at levels greater than or equal to 0.1% is identified as a OSHA: carcinogen or potential carcinogen by OSHA.

#### **Reproductive toxicity**

No data available

No data available

Specific target organ toxicity - single exposure May cause drowsiness or dizziness.

Specific target organ toxicity - repeated exposure No data available

Aspiration hazard No data available

#### Additional Information

Repeated dose Mouse - female - Oral - LOAEL : 390 mg/kg toxicity RTECS: KX3850000

narcosis, Liver injury may occur., Kidney injury may occur.

#### **12. ECOLOGICAL INFORMATION**

#### 12.1 Toxicity

12.3

12.4

	Toxicity to fish	flow-through test LC50 - Oncorhynchus mykiss (rainbow trout) - 5 mg/l $$ - 96 h
	Toxicity to daphnia and other aquatic invertebrates	EC50 - Daphnia magna (Water flea) - 7.50 mg/l  - 48 h
	Toxicity to algae	static test EC50 - Skeletonema costatum - > 16 mg/l  - 7 h
12.2	<b>Persistence and degrad</b> Biodegradability	<b>lability</b> aerobic - Exposure time 28 d Result: 11 % - Not readily biodegradable. (OECD Test Guideline 301C)
2.3 E	Bioaccumulative potentia Bioaccumulation	I Lepomis macrochirus (Bluegill) - 21 d - 0.00343 mg/l
2.4	Mobility in soil	BIOCONCENTRATION TACTOR (BUF): 49

### No data available

#### 12.5 Results of PBT and vPvB assessment

PBT/vPvB assessment not available as chemical safety assessment not required/not conducted

#### 12.6 Other adverse effects

An environmental hazard cannot be excluded in the event of unprofessional handling or disposal. Toxic to aquatic life with long lasting effects.

### 13. DISPOSAL CONSIDERATIONS

#### 13.1 Waste treatment methods

### Product

Offer surplus and non-recyclable solutions to a licensed disposal company. Contact a licensed professional waste disposal service to dispose of this material. Dissolve or mix the material with a combustible solvent and burn in a chemical incinerator equipped with an afterburner and scrubber.

**14. TRANSPORT INFORMATION** 

#### DOT (US)

UN number: 1897 Class: 6.1 Packing group: III Proper shipping name: Tetrachloroethylene Reportable Quantity (RQ): 100 lbsReportable Quantity (RQ): 100 lbs Poison Inhalation Hazard: No

### IMDG

UN number: 1897 Class: 6.1 Packing group: III EMS-No: F-A, S-A Proper shipping name: TETRACHLOROETHYLENE Marine pollutant: yes

### IATA

UN number: 1897	Class: 6.1	Packing group: III
Proper shipping name:	Tetrachloroethylene	

#### 15. REGULATORY INFORMATION

#### SARA 302 Components

No chemicals in this material are subject to the reporting requirements of SARA Title III, Section 302.

#### SARA 313 Components

The following components are subject to reporting levels es	tablished by SARA Title I	II, Section 313:
	CAS-No.	Revision Date
Tetrachloroethylene	127-18-4	2007-07-01

#### SARA 311/312 Hazards

Acute Health Hazard, Chronic Health Hazard

Reportable Quantity	D039 lbs		
Massachusetts Right To Kno	ow Components		
		CAS-No.	Revision Date
Tetrachloroethylene		127-18-4	2007-07-01
Pennsylvania Right To Know	/ Components		
	-	CAS-No.	Revision Date
Tetrachloroethylene		127-18-4	2007-07-01
		CAS-No.	Revision Date
Tetrachloroethylene		127-18-4	2007-07-01
New Jersey Right To Know (	Components		
	-	CAS-No.	Revision Date
Tetrachloroethylene		127-18-4	2007-07-01
California Prop. 65 Compone	ents		
WARNING! This product cor	tains a chemical known to the	CAS-No.	Revision Date
State of California to cause of	cancer.	127-18-4	2007-09-28
Tetrachloroethylene			

#### **16. OTHER INFORMATION**

Full text of H-Statements referred to under sections 2 and 3.

Aquatic Acute	Acute aquatic toxicity
Aquatic Chronic	Chronic aquatic toxicity
Carc.	Carcinogenicity

Eye Irrit.	Eye irritation
H315	Causes skin irritation.
H317	May cause an allergic skin reaction.
H319	Causes serious eye irritation.
H336	May cause drowsiness or dizziness.
H351	Suspected of causing cancer.
H401	Toxic to aquatic life.
H411	Toxic to aquatic life with long lasting effects

#### **HMIS Rating**

Health hazard:	3
Chronic Health Hazard:	*
Flammability:	0
Physical Hazard	0
NFPA Rating	
Health hazard:	2

	_
Fire Hazard:	0
Reactivity Hazard:	0

#### Further information

Copyright 2016 Sigma-Aldrich Co. LLC. License granted to make unlimited paper copies for internal use only. The above information is believed to be correct but does not purport to be all inclusive and shall be used only as a guide. The information in this document is based on the present state of our knowledge and is applicable to the product with regard to appropriate safety precautions. It does not represent any guarantee of the properties of the product. Sigma-Aldrich Corporation and its Affiliates shall not be held liable for any damage resulting from handling or from contact with the above product. See www.sigma-aldrich.com and/or the reverse side of invoice or packing slip for additional terms and conditions of sale.

#### **Preparation Information**

Sigma-Aldrich Corporation Product Safety – Americas Region 1-800-521-8956

Version: 4.11

Revision Date: 06/28/2017

Print Date: 06/22/2019



## **SAFETY DATA SHEET**

Version 6.3 Revision Date 03/06/2019 Print Date 06/22/2019

### SECTION 1: Identification of the substance/mixture and of the company/undertaking

#### **1.1 Product identifiers**

Product name : Toluene Product Number : 244511 Brand : Sigma-Aldrich Index-No. : 601-021-00-3 CAS-No. : 108-88-3

#### **1.2** Relevant identified uses of the substance or mixture and uses advised against

Identified uses : Laboratory chemicals, Synthesis of substances

### 1.3 Details of the supplier of the safety data sheet

Company	:	Sigma-Aldrich Inc. 3050 Spruce Street ST. LOUIS MO 63103 UNITED STATES
Telephone Fax	:	+1 314 771-5765 +1 800 325-5052

#### **1.4 Emergency telephone number**

Emergency Phone # : +1-703-527-3887

### **SECTION 2: Hazards identification**

#### 2.1 Classification of the substance or mixture

### GHS Classification in accordance with 29 CFR 1910 (OSHA HCS)

Flammable liquids (Category 2), H225 Skin irritation (Category 2), H315 Reproductive toxicity (Category 2), H361 Specific target organ toxicity - single exposure (Category 3), Central nervous system, H336 Specific target organ toxicity - repeated exposure (Category 2), H373 Aspiration hazard (Category 1), H304 Short-term (acute) aquatic hazard (Category 2), H401

For the full text of the H-Statements mentioned in this Section, see Section 16.

#### 2.2 GHS Label elements, including precautionary statements

Danger

Pictogram



Signal word

Sigma-Aldrich - 244511

Page 1 of 11



Hazard statement(s) H225 H304 H315 H336 H361 H373 H401	Highly flammable liquid and vapour. May be fatal if swallowed and enters airways. Causes skin irritation. May cause drowsiness or dizziness. Suspected of damaging fertility or the unborn child. May cause damage to organs through prolonged or repeated exposure. Toxic to aquatic life.
Precautionary statement(s)	
P201 P202	Obtain special instructions before use. Do not handle until all safety precautions have been read and understood.
P210	Keep away from heat/sparks/open flames/hot surfaces. No smoking.
P233	Keep container tightly closed.
P240	Ground/bond container and receiving equipment.
P241	Use explosion-proof electrical/ventilating/lighting equipment.
P242	Use only non-sparking tools.
P243	Take precautionary measures against static discharge.
P20U D264	Wash skip thoroughly after bandling
P204 D271	Wash skill thoroughly alter handling.
	Avoid release to the environment
P280	Wear protective gloves/ protective clothing/ eve protection/ face
1200	notection
P301 + P310	IF SWALLOWED: Immediately call a POISON CENTER/doctor
P303 + P361 + P353	IF ON SKIN (or hair): Take off immediately all contaminated clothing. Rinse skin with water/shower
P304 + P340 + P312	IF INHALED: Remove person to fresh air and keep comfortable
	for breathing. Call a POISON CENTER/doctor if you feel unwell.
P308 + P313	IF exposed or concerned: Get medical advice/ attention.
P331	Do NOT induce vomiting.
P332 + P313	If skin irritation occurs: Get medical advice/ attention.
P362	Take off contaminated clothing and wash before reuse.
P370 + P378	In case of fire: Use dry sand, dry chemical or alcohol-resistant foam to extinguish.
P403 + P233	Store in a well-ventilated place. Keep container tightly closed.
P403 + P235	Store in a well-ventilated place. Keep cool.
P405	Store locked up.
P501	Dispose of contents/ container to an approved waste disposal plant.

## 2.3 Hazards not otherwise classified (HNOC) or not covered by GHS - none

## **SECTION 3:** Composition/information on ingredients

## 3.1 Substances

Formula	:	C7H8
Molecular weight	:	92.14 g/mol
CAS-No.	:	108-88-3
EC-No.	:	203-625-9

Sigma-Aldrich - 244511

Page 2 of 11



Index-No.

: 601-021-00-3

Component	Classification	Concentration
Toluene		
	Flam. Liq. 2; Skin Irrit. 2; Repr. 2; STOT SE 3; STOT RE 2; Asp. Tox. 1; Aquatic Acute 2; H225, H315, H361d, H336, H373,	< = 100 %
	H304, H401 Concentration limits: 20 %: STOT SE 3, H336;	

For the full text of the H-Statements mentioned in this Section, see Section 16.

### **SECTION 4: First aid measures**

### 4.1 Description of first aid measures

#### **General advice**

Consult a physician. Show this safety data sheet to the doctor in attendance. Move out of dangerous area.

#### If inhaled

If breathed in, move person into fresh air. If not breathing, give artificial respiration. Consult a physician.

#### In case of skin contact

Wash off with soap and plenty of water. Consult a physician.

#### In case of eye contact

Flush eyes with water as a precaution.

#### If swallowed

Do NOT induce vomiting. Never give anything by mouth to an unconscious person. Rinse mouth with water. Consult a physician.

#### 4.2 Most important symptoms and effects, both acute and delayed

The most important known symptoms and effects are described in the labelling (see section 2.2) and/or in section 11

#### **4.3 Indication of any immediate medical attention and special treatment needed** No data available

#### **SECTION 5: Firefighting measures**

#### 5.1 Extinguishing media

Suitable extinguishing media Use water spray, alcohol-resistant foam, dry chemical or carbon dioxide.

**5.2** Special hazards arising from the substance or mixture Carbon oxides

### **5.3** Advice for firefighters Wear self-contained breathing apparatus for firefighting if necessary.

Sigma-Aldrich - 244511

Page 3 of 11



### 5.4 Further information

Use water spray to cool unopened containers.

### **SECTION 6:** Accidental release measures

#### 6.1 Personal precautions, protective equipment and emergency procedures

Use personal protective equipment. Avoid breathing vapours, mist or gas. Ensure adequate ventilation. Remove all sources of ignition. Evacuate personnel to safe areas. Beware of vapours accumulating to form explosive concentrations. Vapours can accumulate in low areas.

For personal protection see section 8.

#### 6.2 Environmental precautions

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

#### 6.3 Methods and materials for containment and cleaning up

Contain spillage, and then collect with an electrically protected vacuum cleaner or by wetbrushing and place in container for disposal according to local regulations (see section 13).

### 6.4 Reference to other sections

For disposal see section 13.

#### **SECTION 7: Handling and storage**

### 7.1 Precautions for safe handling

Avoid contact with skin and eyes. Avoid inhalation of vapour or mist. Use explosion-proof equipment.Keep away from sources of ignition - No smoking.Take measures to prevent the build up of electrostatic charge. For precautions see section 2.2.

#### 7.2 Conditions for safe storage, including any incompatibilities

Keep container tightly closed in a dry and well-ventilated place. Containers which are opened must be carefully resealed and kept upright to prevent leakage.

Handle and store under inert gas. Storage class (TRGS 510): 3: Flammable liquids

#### 7.3 Specific end use(s)

Apart from the uses mentioned in section 1.2 no other specific uses are stipulated

#### SECTION 8: Exposure controls/personal protection

#### 8.1 Control parameters

#### **Components with workplace control parameters**

Sigma-Aldrich - 244511

Page 4 of 11



Component	CAS-No.	Value	Control parameters	Basis
Toluene	108-88-3	TWA	100 ppm 375 mg/m3	USA. OSHA - TABLE Z-1 Limits for Air Contaminants - 1910.1000
		STEL	150 ppm 560 mg/m3	USA. OSHA - TABLE Z-1 Limits for Air Contaminants - 1910.1000
		TWA	200 ppm	USA. Occupational Exposure Limits (OSHA) - Table Z-2
	Remarks	Z37.12-19	67	
		CEIL	300 ppm	USA. Occupational Exposure Limits (OSHA) - Table Z-2
		Z37.12-1967		
		Peak	500 ppm	USA. Occupational Exposure Limits (OSHA) - Table Z-2
		Z37.12-1967		
		TWA	20 ppm	USA. ACGIH Threshold Limit Values (TLV)
		Visual impairment Female reproductive Pregnancy loss 2018 Adoption Substances for which there is a Biological Exposure Index or Indices (see BEI® section) Not classifiable as a human carcinogen		
		TWA	100 ppm 375 mg/m3	USA. NIOSH Recommended Exposure Limits
		ST	150 ppm 560 mg/m3	USA. NIOSH Recommended Exposure Limits

### Biological occupational exposure limits

Component	CAS-No.	Parameters	Value	Biological specimen	Basis
Toluene	108-88-3	Toluene	0.02 mg/l	In blood	ACGIH - Biological Exposure Indices (BEI)
	Remarks	Prior to last s	shift of wor	kweek	
		Toluene	0.03 mg/l	Urine	ACGIH - Biological Exposure Indices (BEI)
		End of shift (	As soon as	possible after exp	oosure ceases)
		o-Cresol	0.3mg/g Creatinin e	Urine	ACGIH - Biological Exposure Indices (BEI)
		End of shift (	As soon as	possible after exp	osure ceases)

### **Predicted No Effect Concentration (PNEC)**

Compartment	Value
Soil	2.89 mg/kg

Sigma-Aldrich - 244511

Page 5 of 11



Marine water	0.68 mg/l
Fresh water	0.68 mg/l
Marine sediment	16.39 mg/kg
Fresh water sediment	16.39 mg/kg
Sewage treatment plant	13.61 mg/l
Aquatic intermittent release	0.68 mg/l

### 8.2 Exposure controls

### Appropriate engineering controls

Handle in accordance with good industrial hygiene and safety practice. Wash hands before breaks and at the end of workday.

### **Personal protective equipment**

### Eye/face protection

Face shield and safety glasses Use equipment for eye protection tested and approved under appropriate government standards such as NIOSH (US) or EN 166(EU).

### **Skin protection**

Handle with gloves. Gloves must be inspected prior to use. Use proper glove removal technique (without touching glove's outer surface) to avoid skin contact with this product. Dispose of contaminated gloves after use in accordance with applicable laws and good laboratory practices. Wash and dry hands.

Full contact Material: Fluorinated rubber Minimum layer thickness: 0.7 mm Break through time: 480 min Material tested: Vitoject® (KCL 890 / Aldrich Z677698, Size M)

Splash contact Material: Fluorinated rubber Minimum layer thickness: 0.7 mm Break through time: 480 min Material tested: Vitoject® (KCL 890 / Aldrich Z677698, Size M)

data source: KCL GmbH, D-36124 Eichenzell, phone +49 (0)6659 87300, e-mail sales@kcl.de, test method: EN374

If used in solution, or mixed with other substances, and under conditions which differ from EN 374, contact the supplier of the CE approved gloves. This recommendation is advisory only and must be evaluated by an industrial hygienist and safety officer familiar with the specific situation of anticipated use by our customers. It should not be construed as offering an approval for any specific use scenario.

#### **Body Protection**

Complete suit protecting against chemicals, Flame retardant antistatic protective clothing., The type of protective equipment must be selected according to the concentration and amount of the dangerous substance at the specific workplace.

#### **Respiratory protection**

Where risk assessment shows air-purifying respirators are appropriate use a fullface respirator with multi-purpose combination (US) or type ABEK (EN 14387) respirator cartridges as a backup to engineering controls. If the respirator is the sole means of protection, use a full-face supplied air respirator. Use respirators and components tested and approved under appropriate government standards such as NIOSH (US) or CEN (EU).

Sigma-Aldrich - 244511

Page 6 of 11



### **Control of environmental exposure**

Prevent further leakage or spillage if safe to do so. Do not let product enter drains. Discharge into the environment must be avoided.

### SECTION 9: Physical and chemical properties

#### 9.1 Information on basic physical and chemical properties

a)	Appearance	Form: liquid Colour: colourless
b)	Odour	aromatic
C)	Odour Threshold	No data available
d)	рН	No data available
e)	Melting point/freezing point	Melting point/range: -93 °C (-135 °F)
f)	Initial boiling point and boiling range	110 - 111 °C 230 - 232 °F
g)	Flash point	4.0 °C (39.2 °F) - closed cup
h)	Evaporation rate	No data available
i)	Flammability (solid, gas)	No data available
j)	Upper/lower flammability or explosive limits	Upper explosion limit: 7 %(V) Lower explosion limit: 1.2 %(V)
k)	Vapour pressure	29.1 hPa at 20.0 °C (68.0 °F)
I)	Vapour density	No data available
m)	Relative density	0.865 g/mL at 25 °C (77 °F)
n)	Water solubility	0.5 g/l at 15 °C (59 °F)
o)	Partition coefficient: n-octanol/water	No data available
p)	Auto-ignition temperature	535.0 °C (995.0 °F)
q)	Decomposition temperature	No data available
r)	Viscosity	No data available
S)	Explosive properties	No data available
t)	Oxidizing properties	No data available

# 9.2 Other safety information No data available

### SECTION 10: Stability and reactivity

#### **10.1 Reactivity**

No data available

Sigma-Aldrich - 244511

Page 7 of 11



### **10.2 Chemical stability**

Stable under recommended storage conditions.

### **10.3 Possibility of hazardous reactions** Vapours may form explosive mixture with air.

## **10.4 Conditions to avoid**

Heat, flames and sparks.

### **10.5** Incompatible materials

Strong oxidizing agents

### 10.6 Hazardous decomposition products

Hazardous decomposition products formed under fire conditions. - Carbon oxides Other decomposition products - No data available In the event of fire: see section 5

#### **SECTION 11: Toxicological information**

#### **11.1 Information on toxicological effects**

#### **Acute toxicity**

LD50 Oral - Rat - male - 5,580 mg/kg (Tested according to Directive 92/69/EEC.) LC50 Inhalation - Rat - male and female - 4 h - 25.7 mg/l (OECD Test Guideline 403) LD50 Dermal - Rabbit - 12,124 mg/kg Remarks: (ECHA) No data available

#### Skin corrosion/irritation

Skin - Rabbit Result: Irritating to skin. - 4 h Remarks: (ECHA)

#### Serious eye damage/eye irritation

Eyes - Rabbit Result: No eye irritation (OECD Test Guideline 405)

#### **Respiratory or skin sensitisation**

No data available

#### Germ cell mutagenicity

In vitro mammalian cell gene mutation test Mouse lymphoma test Result: negative Ames test S. typhimurium Result: negative

Rat - Bone marrow Result: negative (ECHA)

#### Carcinogenicity

IARC: No component of this product present at levels greater than or equal to 0.1% is identified as probable, possible or confirmed human carcinogen by IARC.

Sigma-Aldrich - 244511

Page 8 of 11



- NTP: No component of this product present at levels greater than or equal to 0.1% is identified as a known or anticipated carcinogen by NTP.
- No component of this product present at levels greater than or equal to 0.1% is OSHA: on OSHA's list of regulated carcinogens.

#### **Reproductive toxicity**

Suspected of damaging the unborn child.

#### Specific target organ toxicity - single exposure

May cause drowsiness or dizziness. - Central nervous system

### Specific target organ toxicity - repeated exposure

May cause damage to organs through prolonged or repeated exposure. - Central nervous system

#### Aspiration hazard

Aspiration hazard, Aspiration may cause pulmonary oedema and pneumonitis.

#### **Additional Information**

RTECS: XS5250000

Drowsiness, irritant effects, Dizziness, Convulsions, Headache, Nausea, Vomiting, Circulatory collapse, somnolence, inebriation, Unconsciousness, respiratory arrest, CNS disorders, respiratory paralysis, death To the best of our knowledge, the chemical, physical, and toxicological properties have not been thoroughly investigated.

Stomach - Irregularities - Based on Human Evidence Stomach - Irregularities - Based on Human Evidence

#### **SECTION 12: Ecological information**

#### 12.1 Toxicity

Toxicity to fish	LC50 - Oncorhynchus mykiss (rainbow trout) - 5.8 mg/l - 96 h Remarks: (ECOTOX Database)	
	NOEC - Pimephales promelas (fathead minnow) - 5.44 mg/l - 7 d	
Toxicity to daphnia and other aquatic invertebrates	Immobilization EC50 - Daphnia magna (Water flea) - 6 mg/l - 48 h Remarks: (ECOTOX Database)	
Toxicity to algae	EC50 - Chlorella vulgaris (Fresh water algae) - 245.00 mg/l - 24 h Remarks: (ECOTOX Database)	
	EC50 - Pseudokirchneriella subcapitata (green algae) - 10.00 mg/l - 24 h Remarks: (ECOTOX Database)	
Persistence and deg	radability	

### 12.2

Biodegradability aerobic - Exposure time 20 d Result: 86 % - Readily biodegradable. Remarks: (IUCLID)

#### 12.3 Bioaccumulative potential

Bioaccumulation

Leuciscus idus (Golden orfe) - 3 d - 0.05 mg/I(Toluene)

Sigma-Aldrich - 244511

Page 9 of 11



### 12.4 Mobility in soil

No data available

#### 12.5 Results of PBT and vPvB assessment

PBT/vPvB assessment not available as chemical safety assessment not required/not conducted

### **12.6 Other adverse effects**

An environmental hazard cannot be excluded in the event of unprofessional handling or disposal. Toxic to aquatic life. No data available

### SECTION 13: Disposal considerations

#### 13.1 Waste treatment methods

### Product

Burn in a chemical incinerator equipped with an afterburner and scrubber but exert extra care in igniting as this material is highly flammable. Offer surplus and non-recyclable solutions to a licensed disposal company. Contact a licensed professional waste disposal service to dispose of this material.

#### **Contaminated packaging**

Dispose of as unused product.

SECTION 14: Transport information		
<b>DOT (US)</b> UN number: 1294 Class: 3 Proper shipping name: Toluene Reportable Quantity (RQ): 1000 lbs Poison Inhalation Hazard: No	Packing group: 11	
<b>IMDG</b> UN number: 1294 Class: 3 Proper shipping name: TOLUENE	Packing group: 11	EMS-No: F-E, S-D
<b>IATA</b> UN number: 1294 Class: 3 Proper shipping name: Toluene	Packing group: 11	

### SECTION 15: Regulatory information

#### SARA 302 Components

No chemicals in this material are subject to the reporting requirements of SARA Title III, Section 302.

#### SARA 313 Components

The following components are subject to reporting levels established by SARA Title III, Section 313:

CAS-No. Revision Date

Sigma-Aldrich - 244511

Page 10 of 11



Toluene	108-88-3	2007-07-01
<b>SARA 311/312 Hazards</b> Fire Hazard, Acute Health Hazard, Chronic Health Ha	zard	
Massachusetts Right To Know Components		
Toluene	CAS-No. 108-88-3	Revision Date 2007-07-01
Pennsylvania Right To Know Components		
loluene	CAS-No. 108-88-3	Revision Date 2007-07-01
California Prop. 65 Components		
, which is/are known to the State of California to	CAS-No.	Revision Date
cause birth defects or other reproductive harm. For more information go to	108-88-3	2009-02-01
www.P65Warnings.ca.gov.Toluene		

### **SECTION 16: Other information**

#### **Further information**

Copyright 2018 Sigma-Aldrich Co. LLC. License granted to make unlimited paper copies for internal use only.

The above information is believed to be correct but does not purport to be all inclusive and shall be used only as a guide. The information in this document is based on the present state of our knowledge and is applicable to the product with regard to appropriate safety precautions. It does not represent any guarantee of the properties of the product. Sigma-Aldrich Corporation and its Affiliates shall not be held liable for any damage resulting from handling or from contact with the above product. See www.sigma-aldrich.com and/or the reverse side of invoice or packing slip for additional terms and conditions of sale.

The branding on the header and/or footer of this document may temporarily not visually match the product purchased as we transition our branding. However, all of the information in the document regarding the product remains unchanged and matches the product ordered. For further information please contact mlsbranding@sial.com.

Version: 6.3

Revision Date: 03/06/2019

Print Date: 06/22/2019

Sigma-Aldrich - 244511

Page 11 of 11



# APPENDIX C

**Quality Assurance Project Plan** 

### **Quality Assurance Project Plan**

### Proposed Redevelopment 500 Main Street Laundry New Rochelle, Westchester County, New York

### 1.0 **PROJECT DESCRIPTION**

This document presents the Quality Assurance Project Plan (QAPP) for the remedial action (RA) for the proposed development at 500, 506, and 510 Main Street and 12 Church Street in New Rochelle, Westchester County, New York (the "Site"). The Site consists of an approximately 0.79-acre area property and is located at 500, 506, and 510 Main Street and 12 Church Street, New Rochelle, Westchester County, New York (Site). The Site comprises 4 parcels and is identified on the Westchester County Clerk's map as tax parcels 1-215-0012, 1-215-0011, 1-215-0010, and 1-215-0008. The Site is improved with four structures. The site building operations most recently included two churches, retail stores, and professional offices.

The Site is located in a residential and commercial area and is bounded to the northwest by Main Street and to the southwest by Church Street and is surrounded by commercial properties to the northeast and southeast. The Site has been developed since the 1887 and was historically occupied by with a meat market and sausage shop, and a "Chinese Laundry" (dry cleaner), Huguenot Lodge, Jewelry, and later the Fire Department Headquarters, retail stores, an American Legion Post, auto storage, fur storage, ands skating rink.

#### 2.0 PROJECT ORGANIZATION

The RIWP will be conducted by SESI Consulting Engineers DPC (SESI), on behalf of BRP Companies (BRP) (the "Volunteer"). The organization of SESI's key project management and field staff, and respective areas of responsibility, is presented below.

SESI Personnel				
Role	Name	Telephone No.		
Project Principal	Fuad Dahan, P.E., PhD	973-808-9050 x249		
Project Manager (PM)	Jesse Mausner, P.G.	973-808-9050 x282		
Principal Engineer	Fuad Dahan, P.E., PhD	973-808-9050 x249		
Field Operations Manager	Todd Kelly	973-808-9050 x238		
Field Team Leader	Jon Stuart	973-600-7630		
Quality Assurance Officer	Todd Kelly	973-808-9050 x238		
Field Personnel	TBD			
-----------------	-----	--		

#### 2.1 **Project Principal**

Provide technical and administrative oversight and guidance throughout the project, assist in securing company resources, participate in technical review of deliverables, and attend key meetings as needed.

#### 2.2 Principal Engineer

Provide technical guidance and review of reports, analytical data. Will have key involvement in screening and development of remedial alternatives.

#### 2.3 Project Manager

Responsible for maintaining the day-to-day schedule for completing the fieldwork and deliverables according to BCP program requirements and client expectations.

#### 2.4 Field Program Manager

Responsible for coordinating and directing field efforts of SESI staff and subcontractors, and for maintaining that work is done according to QAPP specifications.

#### 2.5 Field Team Leader

Responsible for overseeing field work during the RA, including observing subcontractors, maintaining field notes, and collecting samples of various environmental media, in accordance with the NYSDEC-approved Work Plan.

#### 2.6 Quality Assurance Officer

Responsible for will reviewing sampling procedures and certify that the data was collected and analyzed using the appropriate procedures.

#### 3.0 QA/QC OBJECTIVES FOR MEASUREMENT OF DATA

In cases where NYSDOH ELAP Certification exists for a specific group or category of parameters, the laboratories performing analysis in connection with this project will have appropriate NYSDOH ELAP Certification. Analytical Service Protocol (ASP, June 2000) Category B deliverables are required for all samples.

Detection limits set by NYSDEC-ASP will be used for all sample analyses unless otherwise noted. If NYSDEC-ASP-dictated detection limits prove insufficient to assess project goals (i.e., comparison to drinking water standards or attainment of Applicable or Relevant and Appropriate Requirements [ARARs]), then ASP Special Analytical Services (SAS) or other appropriate methods will be utilized. The quality assurance/quality control objectives for all measurement data include completeness, representativeness, comparability, precision and accuracy.

#### 3.1 COMPLETENESS

The analyses performed must be appropriate and inclusive. The parameters selected for analysis are chosen to meet the objectives of the study.

Completeness of the analyses will be assessed by comparing the number of parameters intended to be analyzed with the number of parameters successfully determined and validated. Data must meet QC acceptance criteria for 100 percent or more of requested determinations.

#### 3.2 REPRESENTATIVENESS

Samples must be taken of the population and, where appropriate, the population will be characterized statistically to express the degree to which the data accurately and precisely represent a characteristic of a population, parameter variations at a sampling point, a process, or environmental condition.

Non-dedicated sampling devices will be cleaned between sampling points by washing and rinsing with pesticide-grade methanol, followed by a thorough rinse with distilled water. Specific cleaning techniques are described in the Field Sampling Procedure. Two types of blank samples will accompany each sample set where Target Compound List (TCL) volatiles are to be analyzed (water matrix only). A trip blank, consisting of a 40 ml VOA vial of organic-free water prepared by the laboratory, will accompany each set of sample bottles from the laboratory to the field and back. This bottle will remain sealed throughout the shipment and sampling process. This blank will be analyzed for TCL volatile organic compounds along with the groundwater samples to ensure that contamination with TCL volatile compounds has not occurred during the bottle preparation, shipment and sampling phase of the project. In order to check for contaminant carryover when non-dedicated sampling equipment is used, a rinsate blank will be submitted to the laboratory. This blank will also be analyzed for TCL volatile organic compounds. The TCL compounds are identified in the United States Environmental Protection Agency (USEPA) Contract Laboratory Program dated 10/2016 or as periodically updated.

The analysis results obtained from the determination of identical parameters in field duplicate samples can be used to further assess the representativeness of the sample data.

#### 3.3 COMPARABILITY

Consistency in the acquisition, preparation, handling and analysis of samples is necessary in order for the results to be compared where appropriate. Additionally, the results obtained from analyses of the samples will be compared with the results obtained in previous studies, if available.

To ensure the comparability of analytical results with those obtained in previous or future testing, all samples will be analyzed by NYSDEC-approved methods. The NYSDEC-ASP mandated holding times for various analyses will be strictly adhered to.

#### 3.4 PRECISION AND ACCURACY

The validity of the data produced will be assessed for precision and accuracy. Analytical methods which will be used include gas chromatography/mass spectrometry (GC/MS), gas chromatography (GC), colorimetry, atomic spectroscopy, gravimetric and titrametric techniques. The following outlines the procedures for evaluating precision and accuracy, routine monitoring procedures, and corrective actions to maintain analytical quality control. All data evaluations will be consistent with NYSDEC-ASP procedures (June 2000). Data will be 100 percent compliant with NYSDEC-ASP requirements.

The number of duplicate, spiked and blank samples analyzed will a minimum of 1 duplicate for every 20 samples per each medium of groundwater and soil. The inclusion and frequency of analysis of field blanks will be on the order of one per every 20 samples (soil) for the aqueous matrix field blanks will be collected at a frequency of one per day. Samples to be analyzed for volatile organic compounds will be accompanied by a trip blank for each shipment and field blanks (water matrix) or field blanks (soil).

Quality assurance audit samples will be prepared and submitted by the laboratory QA manager for each analytical procedure used. The degree of accuracy and the recovery of analyte to be expected for the analysis of QA samples and spiked samples is dependent upon the matrix, method of analysis, and compound or element being determined. The concentration of the analyte relative to the detection limit is also a major factor in determining the accuracy of the measurement. The lower end of the

analytical range for most analyses is generally accepted to be five times the detection limit. At or above this level, the determination and spike recoveries for metals in water samples will be expected to range from 75 to 125 percent. The recovery of organic surrogate compounds and matrix spiking compounds determined by GC/MS will be compared to the guidelines for recovery of individual compounds as established by the United States Environmental Protection Agency Contract Laboratory Program dated 7/85 or as periodically updated.

The quality of results obtained for inorganic ion and demand parameters will be assessed by comparison of QC data with laboratory control charts for each test.

#### 4.0 SAMPLING PROCEDURES

#### 4.1 SAMPLING PROGRAM

The sampling program for this project will include soil and groundwater. Soil samples will be collected from split spoon sampling or macrocore devices retrieved from soil borings. Groundwater samples will be collected from groundwater monitoring wells using low flow purging techniques.

#### 4.1.1 Drilling/Sampling Procedures

Soil and groundwater samples will be collected by means of a soil boring program. Soil borings shall be completed using the hollow stem auger drilling methods, direct push methods, or rotary drilling methods, whichever methods are determined to be best suited to site conditions by the SESI project manager and SESI field team leader.

Soil samples will be collected from soil borings and analyzed in accordance with the NYSDEC-approved Work Plan. Monitoring wells for groundwater sample collection will be installed in select completed soil borings. Either hollow stem auger (HSA) or direct push drilling methods may be utilized for monitoring well completion.

Samples of the encountered overburden materials shall be collected continuously during drilling so that a complete soil profile is examined and described by the SESI field geologist. The sampling method employed shall be ASTM D-1586/Split Barrel Sampling using a standard 2-foot long, 2-inch outside diameter split- spoon sampler with a 140-pound hammer, in cases where HSA methods are used. Upon retrieval of the sampling barrel, the collected sample shall be placed in glass jars and labeled, stored on site (on ice in a cooler if necessary), and transmitted to the appropriate testing laboratory or

storage facility. Chain-of-custody procedures will be practiced following Section 15, EPA-600/4-82-029, Handbook for Sampling and Sample Preservation of Water and Waste Waters.

A geologist or engineer will be on site during the drilling operations to fully describe each soil sample, following the New York State Soil Description Procedure, and to retain representative portions of each sample.

The drilling contractor will be responsible for obtaining accurate and representative samples, informing the geologist of changes in drilling pressure, keeping a separate general log of soils encountered including blow counts [i.e., the number of blows from a soil sampling drive weight (140 pounds)] required to drive the split-spoon sampler in 6-inch increments and installing monitoring wells to levels directed by the supervising geologist following specifications further outlined in this protocol.

#### 4.1.2 Monitoring Well Completion

Monitoring wells will be constructed of 0.010-inch slot size PVC well screen and riser casing. Other materials utilized for completion will be washed silica sand (Q-Rock No. 4 or approved equivalent) bentonite grout, Portland cement, and a protective steel locking well casing and cap with locks. The depth of the wells will be determined based on the depth to water, type of contaminant and field conditions encountered.

The monitoring well installation method for wells installed within unconsolidated sediments shall be to place the screen and riser assembly into the casing once the screen interval has been selected. At that time, a washed silica sand pack will be placed around the well screen if required to prevent screen plugging. If a sand pack is not warranted, the auger string will be pulled back to allow the native aquifer material to collapse 2 to 3 feet above the top of the screen. Bentonite pellets will then be added to the annulus between the casing and the inside auger to insure proper sealing. Cement/bentonite grout will continue to be added during the extraction of the augers until the entire aquifer thickness has been sufficiently sealed off from horizontal and/or vertical flow above the screened interval. During placement of sand and bentonite pellets, frequent measurements will be made to check the height of the sand pack and thickness of bentonite layers by a weighted drop tape measure.

A bolt-down protective curb box will be installed, flush with the ground, or steel "stick-up" protective casing and secured by a Portland cement seal. The cement seal shall extend laterally at least 1 foot in all directions from the protective casing and shall slope gently away to drain water away from the well.

#### 4.1.3 Well Development

All monitoring wells will be developed or cleared of all fine-grained materials and sediments that have settled in or around the well during installation so that the screen is transmitting representative portions of the groundwater. The development will be by one of two methods, pumping or bailing groundwater from the well until it yields relatively sediment-free water.

A decontaminated pump or bailer will be used and subsequently decontaminated after each use following procedures outlined in the Decontamination Protocol. Pumping or bailing will cease when the turbidity falls below 50 NTUs or until specific conductivity, pH, and temperature are stable (i.e., consecutive readings are within 10 percent with no overall upward or downward trends in measurements). Well development water will be disposed of on the ground surface at each well location or contained in drums for any wells installed inside buildings, or if odorous or stained groundwater is encountered.

#### 4.1.4 Decontamination

All drilling equipment and associated tools including augers, drill rods, sampling equipment, wrenches and any other equipment or tools that have come in contact with contaminated materials will be decontaminated before any drilling on site begins, between each well, and prior to removing any equipment from the site. The preferred decontamination procedure will be to scrape the equipment from any residual soils and then rinse with water and Alconox®. Every effort will be made to minimize the generation of contaminated water. Any contaminated water generated will be drummed. The contaminated water drums will be disposed of at an appropriate facility after approval and sampling in accordance with the specific facility requirements.

#### 4.1.5 PFAS Sampling Considerations

This section contains the materials limitations for Per- and polyfluoroalkyl substances (PFAS) sampling in accordance with the Sampling, Analysis and Assessment of Per and Polyfluorinated Alkyl Substances (PFAS) under NYSDEC's Part 375 Remedial Programs (October 2020).

The groundwater samples will be analyzed for PFAS using Modified USEPA Method 537. Reporting limits for PFOA and PFOS will not exceed 2 nanogram per liter (ng/L). Category B deliverables and an electronic data deliverable will be completed.

PFAS are very persistent in the environment and in the human body. Due to their presence in a variety of products, persistence in the environment and very low drinking water standards, care must be used when groundwater sampling for PFAS to avoid cross contamination from the sampling equipment and personal protective equipment (PPE).

No fabric softener will be used on clothing to be worn in field. Cosmetics, moisturizers, hand cream, unauthorized sunscreen, insect repellant or other related products will not be used the morning of sampling. The field samplers will wear powder-free nitrile gloves while filling and sealing the sample bottles. The sampling equipment components and sample containers will not come in contact with material that may potentially contain PFAS such as aluminum foil, low density polyethylene (LDPE), glass or polytetrafluoroethylene (PTFE, Teflon<sup>™</sup>) materials including sample bottle cap liners with a PTFE layer. Clothing that contains PTFE material (including GORE-TEX®) or that have been waterproofed with PFAS materials will be avoided. Food and drink packaging materials will be avoided, as well.

Sampling will be performed using certified PFAS-free sampling materials such as stainless steel, high density polyethylene (HDPE), PVC, silicone, acetate or polypropylene pump and tubing. Rinse water must be laboratory provided certified PFAS-free distilled or de-ionized water. Standard two step decontamination using Alconox® detergent and clean certified PFAS-free water rinse will be performed for equipment that does come in contact with PFAS materials.

No waterproof field books, plastic clipboards, binders, or spiral hard cover will be used for PFAS containers. No adhesives (i.e. Post-It® Notes), sharpies, or permanent markers will be used for PFAS containers. The PFAS containers will be labeled with ball point pens. PFAS samples will be stored in separate cooler filled with regular ice only with no chemical (blue) ice packs.

Pre-cleaned sample bottles with closures, coolers, sample labels and a chain of custody form will be provided by the laboratory.

#### 4.2 Groundwater Sampling Program.

#### 4.2.1 Well Evacuation

Prior to sampling a monitoring well, the static water level will be recorded. All well data will be recorded on a field sampling record. The wells will be sampled in accordance with the USEPA guidelines for the Low Flow Purging Sampling (LFPS). The purpose of LFPS is to collect groundwater samples from monitoring wells that are representative of ambient groundwater conditions in the aquifer. The LFPS method reduces turbidity which is needed particularly when sampling for metals.

#### 4.2.2 Sampling Procedure

The wells will be sampled using the LFPS technique. A flow rate of 100 ml to 250 ml per minute is used to purge the wells. Drawdown should not exceed 0.3 feet. QED bladder pumps are used for this method. The pump intake is lowered to the midpoint of the water column or as subsurface features such as bedrock fractures or more permeable zones warrant. At the initiation of low flow purging a water level is recorded as well as field parameters. Field parameters are then monitored every five minutes during low flow purging using a flow through cell. When three consecutive measurements of pH differ by 0.1 units or less, with ORP within 10 mv or less, turbidity varies 10 percent or less, conductivity differs by 3 percent or less and dissolved oxygen by 10 percent or less, sampling may begin. Flow through cells are used so continuous real time readings are made. When the parameters stabilize the flow through cell is disconnected and sample bottles are filled directly from the tubing. Sampling procedures are summarized on Table 4.2.

#### 4.3 Soil Vapor Sampling

Soil vapor sampling, if necessary, will be conducted in accordance with NYSDOH Guidance for Evaluating Soil Vapor Intrusion in New York State (October 2006). Soil vapor samples will be collected in the vadose zone from shallow (5 feet) vapor points. Each vapor point will be installed in a shallow boring drilled either by hand-operated equipment (e.g. hand auger or percussion hammer drill), or by a small truck-mounted drill rig. Drilling equipment used shall be based on soil conditions, and the method that provides the most practical approach.

Each vapor point will consist of an inert sampling tube (polyethylene, stainless steel, or Teflon®) with a 6-inch screened section at the bottom through which soil vapors

can be sampled. The screen slot size will be 0.0075 inches. A sampling zone will be created around the screened section by backfilling with 1 to 2 feet of porous coarse sand or glass beads, and at least three feet of bentonite will be placed above the porous sampling zone to form a seal from the surface. Native clean soil will be packed around the remaining annulus to the ground surface.

The regulator will be set to collect a soil vapor sample at a flow rate of less than 0.2 liters per minute. After the summa canister is filled, the valve will be closed.

Each canister will be listed according to a specific sample I.D. on a chain of custody form. Sample canisters will be delivered to the laboratory within 24 hours and analyzed for VOCs by method TO-15. The detection limit for VOCs will be 1  $\mu$ g/m3 or less.

The soil vapor sampling effort will include the use of inert helium tracer gas to verify that the soil vapor samples are not diluted by ambient air. The atmosphere around the sampling tube will be enriched with the tracer gas, and the soil vapor sample will be collected in the presence of the enriched tracer atmosphere. This will be accomplished by placing an inverted plastic pail over the sampling point, and filling the pail with the tracer gas via a small tube penetrating the site of the pail. Refer to NYSDOH Guidance for Evaluating Indoor Air Intrusion in New York State (October 2006).

Weather conditions in the 48 hours prior to the test, and during the test, will be noted, including average wind speed, precipitation, temperature, and barometric pressure.

#### 4.4 SAMPLE PRESERVATION AND SHIPMENT

Since all bottles will contain the necessary preservatives as shown in Table 4.1, they need only be filled. The 40 ml VOA vials must be filled brim full with no air bubbles. The other bottles should be filled to within about 1 inch from the top.

The bottles will be sent from the laboratory in coolers which will be organized on a per site basis. Following sample collection, the bottles should be placed on ice in the shipping cooler. The samples will be cooled to 4°C, but not frozen.

Final packing and shipment of coolers will be performed in accordance with guidelines outlined in the ASP.

#### 5.0 SAMPLE CUSTODY

The program for sample custody and sample transfer is in compliance with the NYSDEC-ASP, as periodically updated. If samples may be needed for legal purposes, chain-of-custody procedures, as defined by NEIC Policies and Procedures (USEPA-330/9-78-001-R, Revised June 1988) will be used. Sample chain-of-custody is initiated by the laboratory with selection and preparation of the sample containers. To reduce the chance for error, the number of personnel handling the samples should be minimized.

#### 5.1 FIELD SAMPLE CUSTODY

A chain-of-custody record accompanies the samples from initial sample container selection and preparation at the laboratory, shipment to the field for sample containment and preservation, and return to the laboratory. Two copies of this record follow the samples to the laboratory. The laboratory maintains one file copy and the completed original is returned to the site inspection team. Individual sample containers provided by the laboratory are used for shipping samples. The shipping containers are insulated and ice is used to maintain samples at approximately 4°C until samples are returned and in the custody of the laboratory. All sample bottles within each shipping container are individually labeled and controlled. Samples are to be shipped to the laboratory within 24-48 hours of the day of collection depending on parameter holding times.

Each sample shipping container is assigned a unique identification number by the laboratory. This number is recorded on the chain-of-custody record and is marked with indelible ink on the outside of the shipping container. The field sampler will indicate the sample designation/location number in the space provided on the appropriate chainof-custody form for each sample collected. The shipping container is closed and a seal provided by the laboratory is affixed to the latch. This seal must be broken to open the container, and this indicates possible tampering if the seal is broken before receipt at the laboratory. The laboratory will contact the project manager and the sample will not be analyzed if tampering is apparent.

#### 5.2 LABORATORY SAMPLE CUSTODY

The Project Quality Assurance Officer notifies the laboratory of upcoming field sampling activities and the subsequent transfer of samples to the laboratory. This notification will include information concerning the number and type of samples to be shipped as well as the anticipated date of arrival. The laboratory sample program meets the following criteria:

The laboratory has designated a sample custodian who is responsible for maintaining custody of the samples and for maintaining all associated records documenting that custody.

Upon receipt of the samples, the custodian will check the original chain-ofcustody documents and compare them with the labeled contents of each sample container for correctness and traceability. The sample custodian signs the chain-ofcustody record and records the date and time received.

Care is exercised to annotate any labeling or descriptive errors. In the event of discrepant documentation, the laboratory will immediately contact the project manager as part of the corrective action process. A qualitative assessment of each sample container is performed to note any anomalies, such as broken or leaking bottles. This assessment is recorded as part of the incoming chain-of-custody procedure.

- 1. The samples are stored in a secured area at a temperature of approximately 4°C until analyses are to commence.
- 2. A laboratory chain-of-custody record accompanies the sample or sample fraction through final analysis for control.
- 3. A copy of the chain-of-custody form will accompany the laboratory report and will become a permanent part of the project records.

#### 5.3 FINAL EVIDENCE FILES

Final evidence files include all originals of laboratory reports and are maintained under documented control in a secure area.

A sample or an evidence file is under custody if:

- It is in your possession; it is in your view, after being in your possession.
- It was in your possession and you placed it in a secure area.
- It is in a designated secure area.

#### 6.0 CALIBRATION PROCEDURES

Instruments and equipment used to gather, generate or measure environmental data will be calibrated with sufficient frequency and in such a manner that accuracy and reproducibility of results are consistent with the appropriate manufacturer's specifications or project specific requirements. The procedures for instrument calibration, calibration verification, and the frequency of calibrations are described in the ASP. The calibration

of instruments used for the determination of metals will be as described in the appropriate CLP standard operating procedures.

Calibration of other instruments required for measurements associated with these analyses will be in accordance with the manufacturer's recommendations and the standard operating procedures of the laboratory.

#### 7.0 ANALYTICAL PROCEDURES

Analytical procedures shall conform to the most recent revision of the NYSDEC-ASP (June 2000) and are summarized on Table 7.1. In the absence of USEPA or NYSDEC guidelines, appropriate procedures shall be submitted for approval by NYSDEC prior to use.

The procedures for the sample preparation and analysis for organic compounds are as specified in the NYSDEC-ASP. Analytical cleanups are mandatory where matrix interferences are noted. No sample shall be diluted any more than a factor of five. The sample shall be either re-extracted, re-sonicated, re-stream distilled, etc. or be subjected to any one analytical cleanup noted in SW846 or a combination thereof. The analytical laboratory shall expend such effort and discretion to demonstrate good laboratory practice and demonstrate an attempt to best achieve the method detection limit.

#### 7.1 VOLATILE ORGANICS (VOA)

For the analysis of water samples for Target Compound List (TCL), volatile organic compounds (VOCs), no sample preparation is required. The analytical procedure for volatiles is detailed in NYSDEC-ASP (Volume I, Section D-I). A measured portion of the sample is placed in the purge and trap apparatus and the sample analysis is performed by gas chromatography/mass spectrometry for the first round. USEPA Method 8260 will be used, plus tentatively identified compounds (TICs). USEPA Methods 8010 or 8020 (gas chromatography with different detectors) will be used if subsequent rounds with lower limits of detection are warranted.

#### 7.2 SEMI-VOLATILE ORGANIC COMPOUNDS

The extraction and analytical procedures used for preparation of water, soil and sediment samples for the analysis of the TCL semi-volatile organic compounds are described in NYSDEC-ASP Volume I, Section D-III. USEPA Method 8270 will be used, plus tentatively identified compounds (TICs).

Instrument calibration, compound identification, and quantitation are performed as described in Section 6 of this document and in the NYSDEC-ASP.

#### 7.3 PESTICIDE AND PCB COMPOUNDS

The sample preservation procedures for gas chromatography for pesticides and PCB's will be as described in the NYSDEC-ASP methods (Section D-IV). The analysis of standard mixes, blanks and spiked samples will be performed at the prescribed frequency with adherence to the 72-hour requirement described in the method.

#### 7.4 METALS

Water, soil and waste samples will be analyzed for the metals listed in Table 7.1. The detection limits for these metals are as specified in the NYSDEC-ASP, Section D-V. The instrument detection limits will be determined using calibration standards and procedures specified in the NYSDEC-ASP. The detection limits for individual samples may be higher due to the sample matrix. The procedures for these analyses will be as described in the NYSDEC-ASP.

The analyses for metals will be performed by atomic absorption spectroscopy (AAS) or inductively-coupled plasma emission spectroscopy (ICPES), as specified in the ASP with regard to AAS flame analysis.

#### 7.5 PER- AND POLYFLUOROALKYL SUBSTANCES (PFAS)

The NYSDEC has developed a list of 21 PFAS Analytes List on table 7.1 for remedial programs. Currently, ELAP does not offer certification for Per- and polyfluoroalkyl substances (PFAS) compounds in matrices other than finished drinking water. Per the Sampling, Analysis and Assessment of Per and Polyfluorinated Alkyl Substances (PFAS) under NYSDEC's Part 375 Remedial Programs (October 2020), the analytical procedure for soil and groundwater sampling of PFAS is Modified EPA Method 537. Reporting limits for PFOA and PFOS in groundwater should not exceed 2 ng/L.

#### 7.6 SITE SPECIFICITY OF ANALYSES

Work plans prepared for remedial action waste sites contain recommendations for the chemical parameters to be determined for each site. Thus, some or all of the referenced methods will apply to the analysis of samples collected at the individual waste sites. Analyses of Target Compound List (TCL) analytes will be performed on all samples. TABLES

#### TABLE 4.1 – SAMPLE CONTAINERIZATION

PARAMETER & ANALYTICAL METHOD	NO.	BOTTLE TYPE	PRESERVATIVE <sup>(1)</sup>	HOLDING TIME
Aqueous Samples	•	•	•	·
VOCs – USEPA 8260C	2	40 mL, glass vial with septum cap	Hydrochloric Acid to pH <2	14 days
SVOCs (BNAs) – USEPA 8270D	2	1-liter amber glass bottle	None	7 days (until extraction) 40 days (extracted)
Pesticides – USEPA 8081B	2	1-liter amber glass bottle	None	7 days (until extraction) 40 days (extracted)
PCBs – USEPA 8082A	2	1-liter amber glass bottle	None	7 days (until extraction) 40 days (extracted)
Metals <sup>(2)</sup>	1	1-liter, plastic bottle	Nitric acid to pH <2 NaOH for cyanide	180 days Cyanide: 14 days Mercury: 28 days
Soil, Sediment, Solid Wa	aste Sample	es		
VOCs – USEPA 8260C	3	15-gram EnCore samplers	Chilled to 0 - 6°C	14 days
SVOCs (BNAs) – USEPA 8270D or E	1	4-oz. glass jar with Teflon lid	Chilled to 0 - 6°C	14 days (until extraction, 40 days extracted)
Pesticides – USEPA 8081B	1	4-oz. glass jar with Teflon lid	Chilled to 0 - 6°C	14 days (until extraction) 40 days (extracted)
PCBs – USEPA 8082A	1	4-oz. glass jar with Teflon lid	Chilled to 0 - 6°C	None
Metals <sup>(2)</sup>	1	4-oz. glass jar with Teflon lid	Chilled to 0 - 6°C	180 days Cyanide: 14 days Mercury: 28 days
Soil Vapor / Indoor Air S	Samples			
VOCs – USEPA TO-15	1	Summa Canister	None	30 days

(1) All samples will be preserved with ice during collection and shipment.

(2) Metals refers to the 24 metals and cyanide in the Target Compound List (NYSDEC-CLP

11/87). Metals will be analyzed by Method 6010D, 7470A for mercury, and 9012B for cyanide

(3) A complete list of compounds is provided on Table 7.1.

# TABLE 4.2 – SAMPLING PROCEDURE FOR MONITORING WELLS USING VOLUME AVERAGED PURGING

- 1. Initial static water level recorded with an electric contact probe accurate to the nearest 0.1 foot.
- 2. Calculate water column volume from the gauged water depth and well depth
- 3. Sampling device lowered into well.
  - a. Bailer lowered by dedicated PVC or polypropylene line.
  - b. Empty 3-5 well volumes based on the calculated water volume from above.
- 4. Sample taken.
  - a. Sample is poured slowly from the open end of the bailer with the sample bottle tilted so that aeration and turbulence are minimized.
  - b. Duplicate sample is collected when appropriate.
- 5. Samples are capped, labeled and placed in laboratory coolers with ice packs or bagged ice.
- 6. Equipment/wash blanks are collected when non-dedicated sampling equipment is used.
- 7. Chain-of-custody forms are completed in triplicate.
  - a. The original and one carbon copy are put into a zip-lock bag and placed into the cooler.
- 8. The original will be returned following sample analysis.
  - a. A second carbon copy is kept on file.
- 9. Cooler is sealed with strapping tape and chain-of-custody seals to assure integrity and to prevent tampering of sample.

#### TABLE 4.3 – SAMPLING PROCEDURE FOR MONITORING WELLS USING LOW-STESS (LOW-FLOW) METHODS

- 1. Initial static water level recorded with an electric contact probe accurate to the nearest 0.1 foot.
- 2. Sampling device is lowered into well. Slowly lower the pump, safety cable, tubing and electrical lines into the well to the depth specified for that well. Pump intake must be no less than 2 feet from the bottom of the well to prevent disturbance and resuspension of sediments which may be at the bottom of the well.
- 3. Measure water level again: Before starting the pump, measure the water level again with the pump in the well. Leave the water level measuring device in the well.
- 4. Purge Well: Start pumping the well at 200 to 500 milliliters per minute (ml/min). The water level should be monitored approximately every five minutes. Ideally, a steady flow rate should be maintained that results in a stabilized water level (drawdown of 0.3 ft or less). Pumping rates should, if needed, be reduced to the minimum capabilities of the pump to ensure stabilization of the water level. As noted above, care should be taken to maintain pump suction and to avoid entrainment of air in the tubing. Record each adjustment made to the pumping rate and the water level measured immediately after each adjustment.
- 5. Monitor Indicator Parameters: During purging of the well, monitor and record the field indicator parameters (turbidity, temperature, specific conductance, pH, Eh, and DO) approximately every five minutes. The well is considered stabilized and ready for sample collection when the indicator parameters have stabilized for three consecutive readings as follows (Puls and Barcelona, 1996):
  - a. 0.1 for pH
  - b. 3% for specific conductance (conductivity)
  - c. 10 mv for redox potential
  - d. 10% for DO and turbidity
- 6. Dissolved oxygen and turbidity usually require the longest time to achieve stabilization. The pump must not be removed from the well between purging and sampling.
- 7. Collect Samples: Collect samples at a flow rate between 100 and 250 ml/min and such that drawdown of the water level within the well does not exceed the maximum allowable drawdown of 0.3 ft. VOC samples must be collected first and directly into sample containers. All sample containers should be filled with minimal turbulence by allowing the ground water to flow from the tubing gently down the inside of the container.
- 8. Ground water samples to be analyzed for volatile organic compounds (VOCs) require pH adjustment. The appropriate EPA Program Guidance should be consulted to determine whether pH adjustment is necessary. If pH adjustment is necessary for VOC sample preservation, the amount of acid to be added to each sample vial prior to sampling should be determined, drop by drop, on a separate and

equal volume of water (e.g., 40 ml). Groundwater purged from the well prior to sampling can be used for this purpose.

- 9. Remove Pump and Tubing: After collection of the samples, the tubing, unless permanently installed, must be properly discarded or dedicated to the well for resampling by hanging the tubing inside the well.
- 10. Measure and record well depth.
- 11. Close and lock the well.
- 12. Samples are capped, labeled and placed in laboratory coolers with ice packs or bagged ice.
- 13. All equipment is cleaned with successive rinses of pesticide-grade methanol and distilled water.
  - a. Dedicated line is disposed of or left at well site.
- 14. Equipment/wash blanks are collected when non-dedicated sampling equipment is used.
- 15. Chain-of-custody forms are completed in triplicate.
  - a. The original and one carbon copy are put into a zip-lock bag and placed into the cooler. The original will be returned following sample analysis.
  - b. A second carbon copy is kept on file.
- 16. Cooler is sealed with strapping tape and chain-of-custody seals to assure integrity and to prevent tampering of sample.

# TABLE 7-1 – CONTRACT-REQUIRED QUANTITATION LEVELS AND ANALYTICAL METHODS FOR ASP INORGANICS, ASP VOLATILES, ASP SEMI-VOLATILES, ASP PESTICIDES, AND PCBS

	SECTION	1 - ASP INORGANIC	CS Met	hod: NYSDEC-ASP-91-4	
PARAMETER		CONTRACT- REQUIRED DETECTION LEVEL* (µg/L)		PARAMETER	CONTRACT- REQUIRED DETECTION LEVEL* (µg/L)
1.	Aluminum	200	13.	Magnesium	5,000
2.	Antimony	60	14.	Manganese	15
3.	Arsenic	15	15.	Mercury	0.2
4.	Barium	200	16.	Nickel	40
5.	Beryllium	5	17.	Potassium	5,000
6.	Cadmium	5	18.	Selenium	35
7.	Calcium	5,000	19.	Silver	10
8.	Chromium	10	20.	Sodium	5,000
9.	Cobalt	50	21.	Thallium	25
10.	Copper	25	22.	Vanadium	50
11.	Iron	100	23.	Zinc	60
12.	Lead	10	24.	Cyanide	10

Target Compound List (TCL) and Contract-Required Quantitation Limit

	SECTION 2 – ASP ORGANICS (VOLATILES) Method: NYSDEC-ASP-91-1							
	VOLATILE	CONTRACT- REQUIRED QUANTITATION LIMIT** (µg/L)	VOLATILE		CONTRACT- REQUIRED QUANTITATION LIMIT** (µg/L)			
1.	Chloromethane	5.0	18.	1,2-Dichloropropane	5.0			
2.	Bromomethane	5.0	19.	cis-1,3- Dichloropropene	5.0			
3.	Vinyl Chloride	5.0	20.	Trichloroethene	5.0			
4.	Chloroethane	5.0	21.	Dibromochloromethane	5.0			
5.	Methylene Chloride	5.0	22.	1,1,2-Trichloroethane	5.0			
6.	Acetone	10.0	23.	Benzene	5.0			
7.	Carbon Disulfide	5.0	24.	Trans-1.3- Dichloropropene	5.0			
8.	1,1-Dichloroethylene	5.0	25.	Bromoform	5.0			
9.	1,1-Dichloroethane	5.0	26.	2-Hexanone	10.0			
10.	1,2-Dichloroethylene (total)	5.0	27.	4-Methyl, 1,2- Pentanone	10.0			
11.	Chloroform	5.0	28.	Tetrachloroethylene	5.0			
12.	1,2-Dichloroethane	5.0	29.	Toluene	5.0			
13.	2-Butanone	10.0	30.	Chlorobenzene	5.0			
14.	1,1,1-Trichloroethane	5.0	31.	Ethylbenzene	5.0			
15.	Carbon Tetrachloride	5.0	32.	Styrene	5.0			
16.	Bromodichloromethane	5.0	33.	Total Xylenes	5.0			
17.	1,1,2,2- Tetrachloroethane	5.0						

	SECTION 3 - ASP ORGANICS (SEMI-VOLATILES) Method: NYSDEC-ASP-91-2							
	SEMI-VOLATILE	CONTRACT- REQUIRED QUANTITATION LIMIT (µg/I)		SEMI-VOLATILE	CONTRACT- REQUIRED QUANTITATION LIMIT (µg/I)			
1.	Phenol	5.0	33.	Acenaphthene	5.0			
2.	Bis(2-chloroethyl)ether	5.0	34.	2,4-Dinitrophenol	10.0			
3.	2-Chlorophenol	5.0	35.	4-Nitrophenol	10.0			
4.	1,3-Dichlorobenzene	5.0	36.	Dibenzofuran	5.0			
5.	1,4-Dichlorobenzene	5.0	37.	Dinitrotoluene	5.0			
6.	1,2-Dichlorobenzene	5.0	38.	Diethylphthalate	5.0			
7.	2-Methylphenol	5.0	39.	4-Chlorophenyl phenyl ether	5.0			
8.	2,2'oxybis(1- Chloropropane)	5.0	40.	Fluorene	5.0			
9.	4-Methylphenol	5.0	41.	4-Nitroanile	10.0			
10.	N-Nitroso-dipropylamine	5.0	42.	4,6-Dinitro-2- methylphenol	10.0			
11.	Hexachloroethane	5.0	43.	N-nitrosodiphenyl amine	5.0			
12.	Nitrobenzene	5.0	44.	4-Bromophenyl phenyl ether	5.0			
13.	Isophorone	5.0	45.	Hexachlorobenzene	5.0			
14.	2-Nitrophenol	5.0	46.	Pentachlorophenol	10.0			
15.	2,4-Dimethylphenol	5.0	47.	Phenanthrene	5.0			
16.	Bis(2-Chloroethoxy) methane	5.0	48.	Anthracene	5.0			
17.	2,4-Dichlorophenol	5.0	49.	Carbazole	5.0			
18.	1,2,4-Trichlorobenzene	5.0	50.	Di-n-butyl phthalate	5.0			
19.	Naphthalene	5.0	51.	Fluoranthene	5.0			
20.	4-Chloroaniline	5.0	52.	Pyrene	5.0			
21.	Hexachlorobutadiene	5.0	53.	Butyl benzyl phthalate	5.0			
22.	4-Chloro-3-methylphenol	5.0	54.	3,3'-Dichloro benzidine	5.0			
23.	2-Methylnaphthalene	5.0	55.	Benz(a)anthracene	5.0			
24.	Hexachlorocyclopentadiene	5.0	56.	Chrysene	5.0			
25.	2,4,6-Trichlorophenol	5.0	57.	Bis(2-ethylhexyl) phthalate	5.0			
26.	2,4,5-Trichlorophenol	10.0	58.	Di-n-octyl phthalate	5.0			
27.	2-Chloronapthalene	5.0	59.	Benzo(b)fluoranthene	5.0			
28.	2-Nitroananiline	10.0	60.	Benzo(k)fluoranthene	5.0			
29.	Dimethyl phthalate	5.0	61.	Benzo(a)pyrene	5.0			
30.	Acenaphthylene	5.0	62.	Indeno(1,2,3-cd) pyrene	5.0			
31.	2,6-Dinitrotoluene	5.0	63.	Dibenz(a,h) anthracene	5.0			
32.	3-Nitroaniline	10.0	64.	Benzo(q,h,i)perylene	5.0			

	SECTION 3 - ASP ORGANICS (PESTICIDES/PCBS) Method: NYSDEC-ASP-91-3							
	PESTICIDE/PCB	CONTRACT- REQUIRED QUANTITATION LIMIT (µg/I)		PESTICIDE/PCB	CONTRACT- REQUIRED QUANTITATION LIMIT (µg/I)			
1.	Alpha-BHC	0.05	15.	4,4'-DDT	0.10			
2.	Beta-BHC	0.05	16.	Methoxychlor	0.5			
3.	Delta-BHC	0.05	17.	Endrin ketone	0.10			
4.	Gamma-BHC (lindane)	0.05	18.	Endrin aldehyde	0.10			
5.	Heptachlor	0.05	19.	Alpha-Chlordane	0.05			
6.	Aldrin	0.05	20.	Gamma-Chlordane	0.05			
7.	Heptachlor epoxide	0.05	21.	Toxaphene	5.0			
8.	Endosulfan I	0.05	22.	AROCHLOR-1016	1.0			
9.	Dieldrin	0.10	23.	AROCHLOR-1221	1.0			
10.	4,4'-DDE	0.10	24.	AROCHLOR-1232	1.0			
11.	Endrin	0.10	25.	AROCHLOR-1242	1.0			
12.	Endosulfan II	0.10	26.	AROCHLOR-1248	1.0			
13.	4,4'-DDD	0.10	27.	AROCHLOR-1254	1.0			
14.	Endosulfan sulfate	0.10	28.	AROCHLOR-1260	1.0			

\*Matrix: groundwater. For soil matrix, multiply CRDL by 100. \*\*Quantitation limit for medium-level soil is 1,200 µg/kg (wet weight basis).

## APPENDIX D

Community Air Monitoring Plan

## **Community Air Monitoring Plan**

Proposed Development 500 Main Street Laundry New Rochelle, New York

#### 1.0 INTRODUCTION

This document presents a Community Air Monitoring Plan (CAMP) for the remedial action (RA) for the proposed development at 500, 506, and 510 Main Street and 12 Church Street in New Rochelle, Westchester County, New York (the "Site").

The Site consists of an approximately 0.79-acre area property and is located at 500, 506, and 510 Main Street and 12 Church Street, New Rochelle, Westchester County, New York (Site). The Site comprises 4 parcels and is identified on the Westchester County Clerk's map as tax parcels 1-215-0012, 1-215-0011, 1-215-0010, and 1-215-0008. The Site is improved with four structures. The site building operations most recently included two churches, retail stores, and professional offices.

The Site is located in a residential and commercial area and is bounded to the northwest by Main Street and to the southwest by Church Street and is surrounded by commercial properties to the northeast and southeast. The Site has been developed since the 1887 and was historically occupied by with a meat market and sausage shop, and a "Chinese Laundry" (dry cleaner), Huguenot Lodge, Jewelry, and later the Fire Department Headquarters, retail stores, an American Legion Post, auto storage, fur storage, ands skating rink.

#### 1.1 OBJECTIVES

The objective of this CAMP is to provide a measure of protection for the downwind community from potential airborne contaminant releases that may arise as a result of the planned remedial action activities, which will include soil excavation.

#### 1.2 METHODS

The CAMP will include monitoring for particulate matter (e.g., airborne "dust") and volatile organic compounds (VOCs) during the remedial action activities. Readings will be recorded and will be submitted for State (DEC and DOH) personnel to review in the form of weekly CAMP data summaries. The DEC and DOH will be notified of any CAMP exceedances within 24 hours.

#### 1.3 PARTICULATE MONITORING

During all ground-intrusive activities, particulate (e.g. "dust") emissions will be measured continuously at the upwind and downwind work zone boundaries. Real time monitoring equipment (e.g. Trak TSI Dust monitors or equivalent), with audible alarms and capable of measuring particulate matter less than 10 micrometers in size (PM-10), will be used. If the wind is calm, the monitors should be placed between each work area and the nearest sensitive receptors. If the wind is variable, the monitors must be placed accordingly to ensure there is a monitor downwind of each work area at all times. Air monitoring locations will be selected daily based on prevailing wind conditions and specific locations where field-work is to be conducted on a daily basis.

- If the downwind particulate level is 100 micrograms per cubic meter (ug/m<sup>3</sup>) greater than background (upwind) for a 15-minute period or if airborne dust is observed leaving the work area, then the following dust suppression techniques will be employed.
  - a) Applying water on haul roads (water supplied from hydrant);
  - b) Wetting equipment and excavation faces;
  - c) Spraying water on buckets during excavation and dumping;
  - d) Hauling materials in properly tarped or watertight containers;
  - e) Restricting vehicle speeds to 10 mph;
  - f) Covering excavated areas and material after excavation activity ceases; and
  - g) Reducing the excavation size and/or number of excavations.

Work will continue with dust suppression provided that downwind particulate levels do not exceed 150 ug/m<sup>3</sup> above upwind levels and provided that no visible dust is migrating from the work area.

- If, after dust suppression techniques, downwind particulate levels are greater than 150 ug/m<sup>3</sup> above upwind levels, work will be stopped and a re-evaluation of activities will be initiated. Work will resume, provided that dust suppression measures and other controls are successful in reducing downwind particulate concentrations to within 150 ug/m<sup>3</sup> of the upwind level and in preventing visible dust migration.
- All readings must be recorded and be available for State (NYSDEC and NYSDOH) and County Health personnel to review.

#### 1.4 VOC MONITORING

Volatile organic compounds (VOCs) must be monitored at the downwind perimeter of the immediate work area (i.e., the exclusion zone) on a continuous basis or as otherwise specified. Upwind concentrations should be measured at the start of each workday and periodically thereafter to establish background conditions, particularly if wind direction changes. The monitoring work should be performed using equipment appropriate to measure the types of contaminants known or suspected to be present. The equipment should be calibrated at least daily for the contaminant(s) of concern or for an appropriate surrogate. The equipment should be capable of calculating 15-minute running average concentrations, which will be compared to the levels specified below.

1. If the ambient air concentration of total organic vapors at the downwind perimeter of the work area or exclusion zone exceeds 5 parts per million (ppm) above background for the 15-minute average, work activities must be temporarily halted and monitoring continued. If the total organic vapor level readily decreases (per instantaneous readings) below 5 ppm over background, work activities can resume with continued monitoring.

2. If total organic vapor levels at the downwind perimeter of the work area or exclusion zone persist at levels in excess of 5 ppm over background but less than 25 ppm, work activities must be halted, the source of vapors identified, corrective actions taken to abate emissions, and monitoring continued. After these steps, work activities can resume provided that the total organic vapor level 200 feet downwind of the exclusion zone or half the distance to the nearest potential receptor or residential/commercial structure, whichever is less - but in no case less than 20 feet, is below 5 ppm over background for the 15-minute average.

3. If the organic vapor level is above 25 ppm at the perimeter of the work area, activities must be shutdown.

4. All 15-minute readings must be recorded and be available for State (DEC and NYSDOH) personnel to review. Instantaneous readings, if any, used for decision purposes should also be recorded.

## **APPENDIX E**

NYSDEC Soil Cleanup Objectives

#### 375-6.8

**Soil cleanup objective tables.** Unrestricted use soil cleanup objectives. (a)

Contaminant	CAS Number	Unrestricted Use		
	Metals			
Arsenic	7440-38-2	13 °		
Barium	7440-39-3	350 °		
Beryllium	7440-41-7	7.2		
Cadmium	7440-43-9	2.5 °		
Chromium, hexavalent <sup>e</sup>	18540-29-9	1 <sup>b</sup>		
Chromium, trivalent <sup>e</sup>	16065-83-1	30 °		
Copper	7440-50-8	50		
Total Cyanide <sup>e, f</sup>		27		
Lead	7439-92-1	63 °		
Manganese	7439-96-5	1600 <sup>c</sup>		
Total Mercury		0.18 <sup>c</sup>		
Nickel	7440-02-0	30		
Selenium	7782-49-2	3.9 <sup>c</sup>		
Silver	7440-22-4	2		
Zinc	7440-66-6	109 °		
	PCBs/Pesticides			
2,4,5-TP Acid (Silvex) <sup>f</sup>	93-72-1	3.8		
4,4'-DDE	72-55-9	0.0033 <sup>b</sup>		
4,4'-DDT	50-29-3	0.0033 <sup>b</sup>		
4,4'-DDD	72-54-8	0.0033 <sup>b</sup>		
Aldrin	309-00-2	0.005 °		
alpha-BHC	319-84-6	0.02		
beta-BHC	319-85-7	0.036		
Chlordane (alpha)	5103-71-9	0.094		

## Table 375-6.8(a):Unrestricted Use Soil Cleanup Objectives

Contaminant	CAS Number	<b>Unrestricted</b> Use				
delta-BHC <sup>g</sup>	319-86-8	0.04				
Dibenzofuran <sup>f</sup>	132-64-9	7				
Dieldrin	60-57-1	0.005 °				
Endosulfan I <sup>d, f</sup>	959-98-8	2.4				
Endosulfan II <sup>d, f</sup>	33213-65-9	2.4				
Endosulfan sulfate <sup>d, f</sup>	1031-07-8	2.4				
Endrin	72-20-8	0.014				
Heptachlor	76-44-8	0.042				
Lindane	58-89-9	0.1				
Polychlorinated biphenyls	1336-36-3	0.1				
Semivolatile organic compounds						
Acenaphthene	83-32-9	20				
Acenapthylene <sup>f</sup>	208-96-8	100 <sup>a</sup>				
Anthracene <sup>f</sup>	120-12-7	100 <sup>a</sup>				
Benz(a)anthracene <sup>f</sup>	56-55-3	$1^{c}$				
Benzo(a)pyrene	50-32-8	1°				
Benzo(b)fluoranthene <sup>f</sup>	205-99-2	1°				
Benzo(g,h,i)perylene <sup>f</sup>	191-24-2	100				
Benzo(k)fluoranthene <sup>f</sup>	207-08-9	0.8 °				
Chrysene <sup>f</sup>	218-01-9	1°				
Dibenz(a,h)anthracene <sup>f</sup>	53-70-3	0.33 <sup>b</sup>				
Fluoranthene <sup>f</sup>	206-44-0	100 <sup>a</sup>				
Fluorene	86-73-7	30				
Indeno(1,2,3-cd)pyrene <sup>f</sup>	193-39-5	0.5 °				
m-Cresol <sup>f</sup>	108-39-4	0.33 <sup>b</sup>				
Naphthalene <sup>f</sup>	91-20-3	12				
o-Cresol <sup>f</sup>	95-48-7	0.33 <sup>b</sup>				

Table 375-6.8(a):Unrestricted Use Soil Cleanup Objectives

Contaminant	CAS Number	Unrestricted Use					
p-Cresol <sup>f</sup>	106-44-5	0.33 <sup>b</sup>					
Pentachlorophenol	87-86-5	0.8 <sup>b</sup>					
Phenanthrene <sup>f</sup>	85-01-8	100					
Phenol	108-95-2	0.33 <sup>b</sup>					
Pyrene <sup>f</sup>	129-00-0	100					
Volatile organic compounds							
1,1,1-Trichloroethane <sup>f</sup>	71-55-6	0.68					
1,1-Dichloroethane <sup>f</sup>	75-34-3	0.27					
1,1-Dichloroethene <sup>f</sup>	75-35-4	0.33					
1,2-Dichlorobenzene <sup>f</sup>	95-50-1	1.1					
1,2-Dichloroethane	107-06-2	0.02 °					
cis -1,2-Dichloroethene <sup>f</sup>	156-59-2	0.25					
trans-1,2-Dichloroethene <sup>f</sup>	156-60-5	0.19					
1,3-Dichlorobenzene <sup>f</sup>	541-73-1	2.4					
1,4-Dichlorobenzene	106-46-7	1.8					
1,4-Dioxane	123-91-1	0.1 <sup>b</sup>					
Acetone	67-64-1	0.05					
Benzene	71-43-2	0.06					
n-Butylbenzene <sup>f</sup>	104-51-8	12					
Carbon tetrachloride <sup>f</sup>	56-23-5	0.76					
Chlorobenzene	108-90-7	1.1					
Chloroform	67-66-3	0.37					
Ethylbenzene <sup>f</sup>	100-41-4	1					
Hexachlorobenzene	118-74-1	0.33 <sup>b</sup>					
Methyl ethyl ketone	78-93-3	0.12					
Methyl tert-butyl ether <sup>f</sup>	1634-04-4	0.93					
Methylene chloride	75-09-2	0.05					

Table 375-6.8(a):Unrestricted Use Soil Cleanup Objectives

Contaminant	CAS Number	Unrestricted Use
n - Propylbenzene <sup>f</sup>	103-65-1	3.9
sec-Butylbenzene <sup>f</sup>	135-98-8	11
tert-Butylbenzene <sup>f</sup>	98-06-6	5.9
Tetrachloroethene	127-18-4	1.3
Toluene	108-88-3	0.7
Trichloroethene	79-01-6	0.47
1,2,4-Trimethylbenzene <sup>f</sup>	95-63-6	3.6
1,3,5-Trimethylbenzene <sup>f</sup>	108-67-8	8.4
Vinyl chloride <sup>f</sup>	75-01-4	0.02
Xylene (mixed)	1330-20-7	0.26

 Table 375-6.8(a):Unrestricted Use Soil Cleanup Objectives

All soil cleanup objectives (SCOs) are in parts per million (ppm).

#### Footnotes

<sup>a</sup> The SCOs for unrestricted use were capped at a maximum value of 100 ppm. See Technical Support Document (TSD), section 9.3.

<sup>b</sup> For constituents where the calculated SCO was lower than the contract required quantitation limit (CRQL), the CRQL is used as the Track 1 SCO value.

<sup>c</sup> For constituents where the calculated SCO was lower than the rural soil background concentration, as determined by the Department and Department of Health rural soil survey, the rural soil background concentration is used as the Track 1 SCO value for this use of the site.

<sup>d</sup> SCO is the sum of endosulfan I, endosulfan II and endosulfan sulfate.

<sup>e</sup> The SCO for this specific compound (or family of compounds) is considered to be met if the analysis for the total species of this contaminant is below the specific SCO.

<sup>f</sup> Protection of ecological resources SCOs were not developed for contaminants identified in Table 375-6.8(b) with "NS". Where such contaminants appear in Table 375-6.8(a), the applicant may be required by the Department to calculate a protection of ecological resources SCO according to the TSD.

## (b) Restricted use soil cleanup objectives.

	CAS	Protection of Public Health				Protection	Protection
Contaminant	Number	Residential	Restricted- Residential	Commercial	Industrial	Ecological Resources	Ground- water
Metals							
Arsenic	7440-38-2	16 <sup>f</sup>	16 <sup>f</sup>	16 <sup>f</sup>	16 <sup>f</sup>	13 <sup>f</sup>	16 <sup>f</sup>
Barium	7440-39-3	350 <sup>f</sup>	400	400	10,000 <sup>d</sup>	433	820
Beryllium	7440-41-7	14	72	590	2,700	10	47
Cadmium	7440-43-9	2.5 <sup>f</sup>	4.3	9.3	60	4	7.5
Chromium, hexavalent h	18540-29-9	22	110	400	800	1 <sup>e</sup>	19
Chromium, trivalent <sup>h</sup>	16065-83-1	36	180	1,500	6,800	41	NS
Copper	7440-50-8	270	270	270	10,000 <sup>d</sup>	50	1,720
Total Cyanide <sup>h</sup>		27	27	27	10,000 <sup>d</sup>	NS	40
Lead	7439-92-1	400	400	1,000	3,900	63 <sup>f</sup>	450
Manganese	7439-96-5	2,000 <sup>f</sup>	2,000 <sup>f</sup>	10,000 <sup>d</sup>	10,000 <sup>d</sup>	1600 <sup>f</sup>	2,000 <sup>f</sup>
Total Mercury		0.81 <sup>j</sup>	0.81 <sup>j</sup>	2.8 <sup>j</sup>	5.7 <sup>j</sup>	0.18 <sup>f</sup>	0.73
Nickel	7440-02-0	140	310	310	10,000 <sup>d</sup>	30	130
Selenium	7782-49-2	36	180	1,500	6,800	3.9 <sup>f</sup>	4 <sup>f</sup>
Silver	7440-22-4	36	180	1,500	6,800	2	8.3
Zinc	7440-66-6	2200	10,000 <sup>d</sup>	10,000 <sup>d</sup>	10,000 <sup>d</sup>	109 <sup>f</sup>	2,480
PCBs/Pesticides							
2,4,5-TP Acid (Silvex)	93-72-1	58	100 <sup>a</sup>	500 <sup>b</sup>	1,000°	NS	3.8
4,4'-DDE	72-55-9	1.8	8.9	62	120	0.0033 <sup>e</sup>	17
4,4'-DDT	50-29-3	1.7	7.9	47	94	0.0033 <sup>e</sup>	136
4,4'- DDD	72-54-8	2.6	13	92	180	0.0033 <sup>e</sup>	14
Aldrin	309-00-2	0.019	0.097	0.68	1.4	0.14	0.19
alpha-BHC	319-84-6	0.097	0.48	3.4	6.8	0.04 <sup>g</sup>	0.02
beta-BHC	319-85-7	0.072	0.36	3	14	0.6	0.09
Chlordane (alpha)	5103-71-9	0.91	4.2	24	47	1.3	2.9

## Table 375-6.8(b): Restricted Use Soil Cleanup Objectives

	CAS	Protection of Public Health				Protection	Protection
Contaminant	Number	Residential	Restricted- Residential	Commercial	Industrial	Ecological Resources	Ground- water
delta-BHC	319-86-8	100 <sup>a</sup>	100 <sup>a</sup>	500 <sup>b</sup>	1,000 <sup>c</sup>	0.04 <sup>g</sup>	0.25
Dibenzofuran	132-64-9	14	59	350	1,000 <sup>c</sup>	NS	210
Dieldrin	60-57-1	0.039	0.2	1.4	2.8	0.006	0.1
Endosulfan I	959-98-8	4.8 <sup>i</sup>	24 <sup>i</sup>	200 <sup>i</sup>	920 <sup>i</sup>	NS	102
Endosulfan II	33213-65-9	4.8 <sup>i</sup>	24 <sup>i</sup>	200 <sup>i</sup>	920 <sup>i</sup>	NS	102
Endosulfan sulfate	1031-07-8	4.8 <sup>i</sup>	24 <sup>i</sup>	200 <sup>i</sup>	920 <sup>i</sup>	NS	1,000 <sup>c</sup>
Endrin	72-20-8	2.2	11	89	410	0.014	0.06
Heptachlor	76-44-8	0.42	2.1	15	29	0.14	0.38
Lindane	58-89-9	0.28	1.3	9.2	23	6	0.1
Polychlorinated biphenyls	1336-36-3	1	1	1	25	1	3.2
Semivolatiles							
Acenaphthene	83-32-9	100 <sup>a</sup>	100 <sup>a</sup>	500 <sup>b</sup>	1,000 <sup>c</sup>	20	98
Acenapthylene	208-96-8	100ª	100ª	500 <sup>b</sup>	1,000 <sup>c</sup>	NS	107
Anthracene	120-12-7	100 <sup>a</sup>	100 <sup>a</sup>	500 <sup>b</sup>	1,000 <sup>c</sup>	NS	1,000 <sup>c</sup>
Benz(a)anthracene	56-55-3	1 <sup>f</sup>	1 <sup>f</sup>	5.6	11	NS	$1^{\mathrm{f}}$
Benzo(a)pyrene	50-32-8	$1^{\mathrm{f}}$	$1^{\mathrm{f}}$	$1^{\mathrm{f}}$	1.1	2.6	22
Benzo(b)fluoranthene	205-99-2	$1^{\mathrm{f}}$	$1^{\mathrm{f}}$	5.6	11	NS	1.7
Benzo(g,h,i)perylene	191-24-2	100 <sup>a</sup>	100 <sup>a</sup>	500 <sup>b</sup>	1,000 <sup>c</sup>	NS	1,000 <sup>c</sup>
Benzo(k)fluoranthene	207-08-9	1	3.9	56	110	NS	1.7
Chrysene	218-01-9	$1^{\mathrm{f}}$	3.9	56	110	NS	$1^{\mathrm{f}}$
Dibenz(a,h)anthracene	53-70-3	0.33 <sup>e</sup>	0.33 <sup>e</sup>	0.56	1.1	NS	1,000 <sup>c</sup>
Fluoranthene	206-44-0	100 <sup>a</sup>	100 <sup>a</sup>	500 <sup>b</sup>	1,000 <sup>c</sup>	NS	1,000 <sup>c</sup>
Fluorene	86-73-7	100 <sup>a</sup>	100 <sup>a</sup>	500 <sup>b</sup>	1,000 <sup>c</sup>	30	386
Indeno(1,2,3-cd)pyrene	193-39-5	0.5 <sup>f</sup>	0.5 <sup>f</sup>	5.6	11	NS	8.2
m-Cresol	108-39-4	100 <sup>a</sup>	100 <sup>a</sup>	500 <sup>b</sup>	1,000 <sup>c</sup>	NS	0.33 <sup>e</sup>
Naphthalene	91-20-3	100 <sup>a</sup>	100 <sup>a</sup>	500 <sup>b</sup>	1,000 <sup>c</sup>	NS	12

## Table 375-6.8(b): Restricted Use Soil Cleanup Objectives

Contaminant	CAS Number	Protection of Public Health				Protection	Protection
		Residential	Restricted- Residential	Commercial	Industrial	Ecological Resources	Ground- water
o-Cresol	95-48-7	100 <sup>a</sup>	100 <sup>a</sup>	500 <sup>b</sup>	1,000 <sup>c</sup>	NS	0.33 <sup>e</sup>
p-Cresol	106-44-5	34	100 <sup>a</sup>	500 <sup>b</sup>	1,000 <sup>c</sup>	NS	0.33 <sup>e</sup>
Pentachlorophenol	87-86-5	2.4	6.7	6.7	55	0.8 <sup>e</sup>	0.8 <sup>e</sup>
Phenanthrene	85-01-8	100 <sup>a</sup>	100 <sup>a</sup>	500 <sup>b</sup>	1,000 <sup>c</sup>	NS	1,000 <sup>c</sup>
Phenol	108-95-2	100 <sup>a</sup>	100ª	500 <sup>b</sup>	1,000 <sup>c</sup>	30	0.33 <sup>e</sup>
Pyrene	129-00-0	100ª	100 <sup>a</sup>	500 <sup>b</sup>	1,000 <sup>c</sup>	NS	1,000 <sup>c</sup>
Volatiles							
1,1,1-Trichloroethane	71-55-6	100 <sup>a</sup>	100 <sup>a</sup>	500 <sup>b</sup>	1,000 <sup>c</sup>	NS	0.68
1,1-Dichloroethane	75-34-3	19	26	240	480	NS	0.27
1,1-Dichloroethene	75-35-4	100 <sup>a</sup>	100 <sup>a</sup>	500 <sup>b</sup>	1,000 <sup>c</sup>	NS	0.33
1,2-Dichlorobenzene	95-50-1	100 <sup>a</sup>	100 <sup>a</sup>	500 <sup>b</sup>	1,000 <sup>c</sup>	NS	1.1
1,2-Dichloroethane	107-06-2	2.3	3.1	30	60	10	$0.02^{\mathrm{f}}$
cis-1,2-Dichloroethene	156-59-2	59	100 <sup>a</sup>	500 <sup>b</sup>	1,000 <sup>c</sup>	NS	0.25
trans-1,2-Dichloroethene	156-60-5	100 <sup>a</sup>	100 <sup>a</sup>	500 <sup>b</sup>	1,000 <sup>c</sup>	NS	0.19
1,3-Dichlorobenzene	541-73-1	17	49	280	560	NS	2.4
1,4-Dichlorobenzene	106-46-7	9.8	13	130	250	20	1.8
1,4-Dioxane	123-91-1	9.8	13	130	250	0.1 <sup>e</sup>	0.1 <sup>e</sup>
Acetone	67-64-1	100 <sup>a</sup>	100 <sup>b</sup>	500 <sup>b</sup>	1,000 <sup>c</sup>	2.2	0.05
Benzene	71-43-2	2.9	4.8	44	89	70	0.06
Butylbenzene	104-51-8	100 <sup>a</sup>	100 <sup>a</sup>	500 <sup>b</sup>	1,000 <sup>c</sup>	NS	12
Carbon tetrachloride	56-23-5	1.4	2.4	22	44	NS	0.76
Chlorobenzene	108-90-7	100 <sup>a</sup>	100 <sup>a</sup>	500 <sup>b</sup>	1,000°	40	1.1
Chloroform	67-66-3	10	49	350	700	12	0.37
Ethylbenzene	100-41-4	30	41	390	780	NS	1
Hexachlorobenzene	118-74-1	0.33 <sup>e</sup>	1.2	6	12	NS	3.2
Methyl ethyl ketone	78-93-3	100 <sup>a</sup>	100 <sup>a</sup>	500 <sup>b</sup>	1,000 <sup>c</sup>	100 <sup>a</sup>	0.12

## Table 375-6.8(b): Restricted Use Soil Cleanup Objectives

Contaminant	CAS Number	Protection of Public Health				Protection of	Protection of
		Residential	Restricted- Residential	Commercial	Industrial	Ecological Resources	Ground- water
Methyl tert-butyl ether	1634-04-4	62	100 <sup>a</sup>	500 <sup>b</sup>	1,000 <sup>c</sup>	NS	0.93
Methylene chloride	75-09-2	51	100 <sup>a</sup>	500 <sup>b</sup>	1,000 <sup>c</sup>	12	0.05
n-Propylbenzene	103-65-1	100 <sup>a</sup>	100 <sup>a</sup>	500 <sup>b</sup>	1,000 <sup>c</sup>	NS	3.9
sec-Butylbenzene	135-98-8	100 <sup>a</sup>	100 <sup>a</sup>	500 <sup>b</sup>	1,000 <sup>c</sup>	NS	11
tert-Butylbenzene	98-06-6	100 <sup>a</sup>	100 <sup>a</sup>	500 <sup>b</sup>	1,000 <sup>c</sup>	NS	5.9
Tetrachloroethene	127-18-4	5.5	19	150	300	2	1.3
Toluene	108-88-3	100 <sup>a</sup>	100 <sup>a</sup>	500 <sup>b</sup>	1,000 <sup>c</sup>	36	0.7
Trichloroethene	79-01-6	10	21	200	400	2	0.47
1,2,4-Trimethylbenzene	95-63-6	47	52	190	380	NS	3.6
1,3,5- Trimethylbenzene	108-67-8	47	52	190	380	NS	8.4
Vinyl chloride	75-01-4	0.21	0.9	13	27	NS	0.02
Xylene (mixed)	1330-20-7	100 <sup>a</sup>	100 <sup>a</sup>	500 <sup>b</sup>	1,000 <sup>c</sup>	0.26	1.6

 Table 375-6.8(b): Restricted Use Soil Cleanup Objectives

All soil cleanup objectives (SCOs) are in parts per million (ppm).

NS=Not specified. See Technical Support Document (TSD).

### Footnotes

<sup>a</sup> The SCOs for residential, restricted-residential and ecological resources use were capped at a maximum value of 100 ppm. See TSD section 9.3.

<sup>b</sup> The SCOs for commercial use were capped at a maximum value of 500 ppm. See TSD section 9.3.

<sup>c</sup> The SCOs for industrial use and the protection of groundwater were capped at a maximum value of 1000 ppm. See TSD section 9.3.

<sup>d</sup> The SCOs for metals were capped at a maximum value of 10,000 ppm. See TSD section 9.3.

<sup>e</sup> For constituents where the calculated SCO was lower than the contract required quantitation limit (CRQL), the CRQL is used as the SCO value.

<sup>f</sup> For constituents where the calculated SCO was lower than the rural soil background concentration as determined by the Department and Department of Health rural soil survey, the rural soil background concentration is used as the Track 2 SCO value for this use of the site.

<sup>g</sup> This SCO is derived from data on mixed isomers of BHC.

<sup>h</sup> The SCO for this specific compound (or family of compounds) is considered to be met if the analysis for the total species of this contaminant is below the specific SCO.

<sup>i</sup> This SCO is for the sum of endosulfan I, endosulfan II, and endosulfan sulfate.

<sup>j</sup> This SCO is the lower of the values for mercury (elemental) or mercury (inorganic salts). See TSD Table 5.6-1.

## **APPENDIX F**

**Citizens Participation Plan** 



Department of Environmental Conservation

## **Brownfield Cleanup Program**

## Citizen Participation Plan for 500 Main Street Laundry Site

November 2020

C360199 500, 506 and 510 Main Street and 12 Church Street New Rochelle Westchester County, New York

www.dec.ny.gov
## Contents

Section	Page Number
1. What is New York's Brownfield Cleanup Program?	3
2. Citizen Participation Activities	3
3. Major Issues of Public Concern	9
4. Site Information	9
5. Investigation and Cleanup Process	10
Appendix A - Project Contacts and Locations of Reports and Information	15
Appendix B - Site Contact List	17
Appendix C - Site Location Map	
Appendix D - Brownfield Cleanup Program Process	19

\* \* \* \* \*

**Note:** The information presented in this Citizen Participation Plan was current as of the date of its approval by the New York State Department of Environmental Conservation. Portions of this Citizen Participation Plan may be revised during the site's investigation and cleanup process.

Applicant: **BRP 500 Main LLC** Site Name: **500 Main Street Laundry Site ("Site")** Site Address: **500, 506 and 510 Main Street and 12 Church Street** Site County: **Westchester County** Site Number: **C360199** 

### 1. What is New York's Brownfield Cleanup Program?

New York's Brownfield Cleanup Program (BCP) works with private developers to encourage the voluntary cleanup of contaminated properties known as "brownfields" so that they can be reused and developed. These uses include recreation, housing, and business.

A *brownfield* is any real property that is difficult to reuse or redevelop because of the presence or potential presence of contamination. A brownfield typically is a former industrial or commercial property where operations may have resulted in environmental contamination. A brownfield can pose environmental, legal, and financial burdens on a community. If a brownfield is not addressed, it can reduce property values in the area and affect economic development of nearby properties.

The BCP is administered by the New York State Department of Environmental Conservation (NYSDEC) which oversees Applicants who conduct brownfield site investigation and cleanup activities. An Applicant is a person who has requested to participate in the BCP and has been accepted by NYSDEC. The BCP contains investigation and cleanup requirements, ensuring that cleanups protect public health and the environment. When NYSDEC certifies that these requirements have been met, the property can be reused or redeveloped for the intended use.

For more information about the BCP, go online at: <u>http://www.dec.ny.gov/chemical/8450.html</u>.

### 2. Citizen Participation Activities

### Why NYSDEC Involves the Public and Why It Is Important

NYSDEC involves the public to improve the process of investigating and cleaning up contaminated sites, and to enable citizens to participate more fully in decisions that affect their health, environment, and social well-being. NYSDEC provides opportunities for citizen involvement and encourages early two-way communication with citizens before decision makers form or adopt final positions.

Involving citizens affected and interested in site investigation and cleanup programs is important for many reasons. These include:

- Promoting the development of timely, effective site investigation and cleanup programs that protect public health and the environment
- Improving public access to, and understanding of, issues and information related to a particular site and that site's investigation and cleanup process
- Providing citizens with early and continuing opportunities to participate in NYSDEC's site investigation and cleanup process
- Ensuring that NYSDEC makes site investigation and cleanup decisions that benefit from input that reflects the interests and perspectives found within the affected community
- Encouraging dialogue to promote the exchange of information among the affected/interested public, State agencies, and other interested parties that strengthens trust among the parties, increases understanding of site and community issues and concerns, and improves decision making.

This Citizen Participation (CP) Plan provides information about how NYSDEC will inform and involve the public during the investigation and cleanup of the site identified above. The public information and involvement program will be carried out with assistance, as appropriate, from the Applicant.

## **Project Contacts**

Appendix A identifies NYSDEC project contact(s) to whom the public should address questions or request information about the site's investigation and cleanup program. The public's suggestions about this CP Plan and the CP program for the site are always welcome. Interested people are encouraged to share their ideas and suggestions with the project contacts at any time.

## Locations of Reports and Information

The locations of the reports and information related to the site's investigation and cleanup program also are identified in Appendix A. These locations provide convenient access to important project documents for public review and comment. Some documents may be placed on the NYSDEC web site. If this occurs, NYSDEC will inform the public in fact sheets distributed about the site and by other means, as appropriate.

### Site Contact List

Appendix B contains the site contact list. This list has been developed to keep the community informed about, and involved in, the site's investigation and cleanup process. The site contact list will be used periodically to distribute fact sheets that provide updates about the status of the project. These will include notifications of upcoming activities at the site (such as fieldwork), as well as availability of project documents and announcements about public comment periods. The site contact list includes, at a minimum:

- Chief executive officer and planning board chairperson of each county, city, town and village in which the site is located;
- Residents, owners, and occupants of the site and properties adjacent to the site;
- The public water supplier which services the area in which the site is located;
- Any person who has requested to be placed on the site contact list;
- The administrator of any school or day care facility located on or near the site for purposes of posting and/or dissemination of information at the facility;
- Location(s) of reports and information.

The site contact list will be reviewed periodically and updated as appropriate. Individuals and organizations will be added to the site contact list upon request. Such requests should be submitted to the NYSDEC project contact(s) identified in Appendix A. Other additions to the site contact list may be made at the discretion of the NYSDEC project manager, in consultation with other NYSDEC staff as appropriate.

**Note:** The first site fact sheet (usually related to the draft Remedial Investigation Work Plan) is distributed both by paper mailing through the postal service and through DEC Delivers, its email listserv service. The fact sheet includes instructions for signing up with the appropriate county listserv to receive future notifications about the site. See <a href="http://www.dec.ny.gov/chemical/61092.html">http://www.dec.ny.gov/chemical/61092.html</a>.

Subsequent fact sheets about the site will be distributed exclusively through the listserv, except for households without internet access that have indicated the need to continue to receive site information in paper form. Please advise the NYSDEC site project manager identified in Appendix A if that is the case. Paper mailings may continue during the investigation and cleanup process for some sites, based on public interest and need.

## **CP** Activities

The table at the end of this section identifies the CP activities, at a minimum, that have been and will be conducted during the site's investigation and cleanup program. The

flowchart in Appendix D shows how these CP activities integrate with the site investigation and cleanup process. The public is informed about these CP activities through fact sheets and notices distributed at significant points during the program. Elements of the investigation and cleanup process that match up with the CP activities are explained briefly in Section 5.

- **Notices and fact sheets** help the interested and affected public to understand contamination issues related to a site, and the nature and progress of efforts to investigate and clean up a site.
- **Public forums, comment periods and contact with project managers** provide opportunities for the public to contribute information, opinions and perspectives that have potential to influence decisions about a site's investigation and cleanup.

The public is encouraged to contact project staff at any time during the site's investigation and cleanup process with questions, comments, or requests for information.

This CP Plan may be revised due to changes in major issues of public concern identified in Section 3 or in the nature and scope of investigation and cleanup activities. Modifications may include additions to the site contact list and changes in planned citizen participation activities.

### Technical Assistance Grant

NYSDEC must determine if the site poses a significant threat to public health or the environment. This determination generally is made using information developed during the investigation of the site, as described in Section 5.

If the site is determined to be a significant threat, a qualifying community group may apply for a Technical Assistance Grant (TAG). The purpose of a TAG is to provide funds to the qualifying group to obtain independent technical assistance. This assistance helps the TAG recipient to interpret and understand existing environmental information about the nature and extent of contamination related to the site and the development/implementation of a remedy.

An eligible community group must certify that its membership represents the interests of the community affected by the site, and that its members' health, economic well-being or enjoyment of the environment may be affected by a release or threatened release of contamination at the site.

As of the date the declaration (page 2) was signed by the NYSDEC project manager, the significant threat determination for the site had not yet been made.

To verify the significant threat status of the site, the interested public may contact the NYSDEC project manager identified in Appendix A.

For more information about TAGs, go online at <u>http://www.dec.ny.gov/regulations/2590.html</u>

Note: The table identifying the citizen participation activities related to the site's investigation and cleanup program follows on the next page:

Citizen Participation Activities	Timing of CP Activity(ies)			
Application Process:				
<ul> <li>Prepare site contact list</li> <li>Establish document repository(ies)</li> </ul>	At time of preparation of application to participate in the BCP.			
<ul> <li>Publish notice in Environmental Notice Bulletin (ENB) announcing receipt of application and 30-day public comment period</li> <li>Publish above ENB content in local newspaper</li> <li>Mail above ENB content to site contact list</li> <li>Conduct 30-day public comment period</li> </ul>	When NYSDEC determines that BCP application is complete. The 30-day public comment period begins on date of publication of notice in ENB. End date of public comment period is as stated in ENB notice. Therefore, ENB notice, newspaper notice, and notice to the site contact list should be provided to the public at the same time.			
After Execution of Brownfield Site Cleanup Agreement (BCA):				
Prepare Citizen Participation (CP) Plan	Before start of Remedial Investigation <b>Note:</b> Applicant must submit CP Plan to NYSDEC for review and approval within 20 days of the effective date of the BCA.			
Before NYSDEC Approves Remedial Investigation (RI) Work Plan:				
<ul> <li>Distribute fact sheet to site contact list about proposed RI activities and announcing 30-day public comment period about draft RI Work Plan</li> <li>Conduct 30-day public comment period</li> </ul>	Before NYSDEC approves RI Work Plan. If RI Work Plan is submitted with application, public comment periods will be combined and public notice will include fact sheet. Thirty-day public comment period begins/ends as per dates identified in fact sheet.			
After Applicant Completes Remedial Investigation:				
<ul> <li>Distribute fact sheet to site contact list that describes RI results</li> </ul>	Before NYSDEC approves RI Report			
Before NYSDEC Approves Remedial Work Plan (RWP):				
<ul> <li>Distribute fact sheet to site contact list about draft RWP and announcing 45-day public comment period</li> <li>Public meeting by NYSDEC about proposed RWP (if requested by affected community or at discretion of NYSDEC project manager)</li> <li>Conduct 45-day public comment period</li> </ul>	Before NYSDEC approves RWP. Forty-five day public comment period begins/ends as per dates identified in fact sheet. Public meeting would be held within the 45- day public comment period.			

Citizen Participation Activities	Timing of CP Activity(ies)	
Before Applicant Sta	rts Cleanup Action:	
<ul> <li>Distribute fact sheet to site contact list that describes upcoming cleanup action</li> </ul>	Before the start of cleanup action.	
After Applicant Completes Cleanup Action:		
<ul> <li>Distribute fact sheet to site contact list that announces that cleanup action has been completed and that NYSDEC is reviewing the Final Engineering Report</li> <li>Distribute fact sheet to site contact list announcing NYSDEC approval of Final Engineering Report and</li> </ul>	At the time the cleanup action has been completed. <b>Note:</b> The two fact sheets are combined when possible if there is not a delay in issuing the COC.	
issuance of Certificate of Completion (COC)		

## 3. Major Issues of Public Concern

This section of the CP Plan identifies major issues of public concern that relate to the site. Additional major issues of public concern may be identified during the course of the site's investigation and cleanup process.

There will be demolition of existing structures and foundations and soil excavation occurring at the Site. Therefore, once this activity commences, there may be concerns regarding dust, odors, noise or truck traffic coming from the Site. However, these impacts will be mitigated through implementation of a Health and Safety Plan (HASP) and Soil Management Plan approved by the Department, which will be designed to minimize these impacts. A Community Air Monitoring Plan (CAMP) will also be implemented to monitor dust and vapors to ensure the community is not impacted.

## 4. Site Information

Appendix C contains a map identifying the location of the site.

## Site Description

- Location 500, 506 and 510 Main Street and 12 Church Street, New Rochelle, New York
- Setting Urban
- Site size .79 acres
- Adjacent properties residential, commercial

## History of Site Use, Investigation, and Cleanup

The 500 Main Street portion of the site appears to have started as a Meat Market and Sausage Shop in the late 1890's, early 1900's, and was then converted into the Masonic Huguenot Lodge. A gas engine was identified on Site from 1887 to 1911, with a 100-gallon gasoline underground storage tank (UST) identified on Site in 1911 according to Fire Department records. The gas engine was located in the southern portion of 500 Main Street parcel and the underground tank was located beneath the asphalt driveway associated with the 12 Church Street parcel. The large commercial structure on the northwestern portion of the Subject Property was labeled containing two lard kettles, a brick furnace, a wood house, and a gasoline fired engine according to historic maps. The building at 500 Main Street was converted to a skating rink in 1979, then into retail space in 1987, and thereafter into the New Covenant Church, which was historically heated via an oil-fired boiler, still located in the basement. There is also a vaulted 5,000-gallon No. 2 fuel oil UST located below the loading dock, which may still contain fuel oil. The UST historically fueled the basement-level boiler, which has been

out of service for approximately five years; however, all components of the boiler system are still intact. A vent pipe for the UST was observed on the southwestern exterior of the building on this parcel.

The 506 Main Street portion of the site was originally developed with a meat market and sausage shop, and a "Chinese Laundry" according to Fire Department Maps from at least 1887 until 1896. By 1903, the property is listed as a "market" but it is not clear if the laundry ceased to exist or remained present for some time since until 1941 the use is just listed as a "store" until the use changes to fur storage in 1941 through 1951 in the southern portion of 506 Main Street. Its most recent use was a Dollar Best store.

The building at 510 Main Street is a three-story building that has been occupied by Liebman's Children Clothing (a children's uniform retailer) since 1929.

The 1903 Sanborn map shows 12 Church Street as a Fire Department building, which was present until 1966, and then acquired by Cuisenaire Company of American, Inc. Most recently, this building was converted into the French Speaking Baptist Church. Former Fire stations have been associated with the use of fire-fighting foam discharges which may leave residual Per- and polyfluoroalkyl substances or "PFAS" contamination.

Based on the investigations conducted to date, the primary contaminants of concern remaining at the site in soil are heavy metals including lead, nickel, copper, mercury, zinc, and arsenic; polycyclic aromatic hydrocarbons (PAHs); and pesticides. The contaminants present in groundwater include volatile organic compounds (VOCs), semi-volatile organic compounds (SVOCs,) heavy metals and PFAS. PFAS exceedances of the NYSDEC Guidance Value of 10 ng/L (0.01 ug/L) were noted across the Site, with a more elevated concentration detected in the vicinity of the former Fire Department facility. The VOC exceedance included cis-1,2-dichloroethene (the degradation product of a common dry cleaning solvent). The SVOC exceedances included benzo(a)pyrene, benzo(b)fluoranthene, benzo(k)fluoranthene, hexachlorobenzene, and phenol. Heavy metals were also identified in groundwater, but these exceedances were of common naturally occurring metals including aluminum, manganese, calcium, and sodium, and not due to any known onsite source. Trichloroethene in soil vapor was detected in one sample slightly exceeding the NYSDOH Sub-Slab Vapor Matrix Concentration.

## 5. Investigation and Cleanup Process

#### Application

The Applicant has applied for and been accepted into New York's Brownfield Cleanup Program as a Volunteer. This means that the Applicant was not responsible for the disposal or discharge of the contaminants or whose ownership or operation of the site took place after the discharge or disposal of contaminants. The Volunteer must fully characterize the nature and extent of contamination onsite, and must conduct a "qualitative exposure assessment," a process that characterizes the actual or potential exposures of people, fish and wildlife to contaminants on the site and to contamination that has migrated from the site.

The planned redevelopment of the Site entails the relocation and redevelopment of an existing on-Site church called the New Covenant Church, the off-site relocation of a second current on-Site church called the French Speaking Baptist Church, which is departing from the site, and construction of a new residential apartment building. All existing on-Site structures will be demolished. Under the built-over area of approximately 32,000 square feet (sq. ft.), there will be at least a partial basement of approximate 4,500 sq. ft. in the southern corner of the proposed building. The basement area will consist of utility rooms. The first floor of the building will consist of a church, retail space, lobby area and the entrance to a vehicle lift area, providing access to parking facilities on the third, fourth and fifth floors. The second floor will be utilized for church classrooms for Sunday School, other children related activities, office space for the church and amenity space for the residential portion on the building. The remaining floors will provide residential housing units and the internal parking garage.

As a result of the proposed mixed residential, school and commercial use, the Applicant in its Application proposes that the site will be remediated to unrestricted use, or if this is not possible, to restricted residential use cleanup standards.

To achieve this goal, the Applicant will conduct Investigation activities at the site with oversight provided by NYSDEC. The Brownfield Cleanup Agreement executed by NYSDEC and the Applicant sets forth the responsibilities of each party in conducting these activities at the site.

### Investigation

The Applicant will conduct an investigation of the site officially called a "remedial investigation" (RI). This investigation will be performed with NYSDEC oversight. The Applicant has developed a remedial investigation workplan, which was subject to public comment along with the application.

The site investigation has several goals:

- 1) Define the nature and extent of contamination in soil, surface water, groundwater and any other parts of the environment that may be affected;
- 2) Identify the source(s) of the contamination;
- Assess the impact of the contamination on public health and the environment; and
- 4) Provide information to support the development of a proposed remedy to address the contamination or the determination that cleanup is not necessary.

The Applicant submitted a draft "Remedial Investigation Work Plan" to NYSDEC for review and approval. NYSDEC made the draft plan available to the public review during a 30-day public comment period along with the application.

When the investigation is complete, the Applicant will prepare and submit a report that summarizes the results. This report also will recommend whether cleanup action is needed to address site-related contamination. The investigation report is subject to review and approval by NYSDEC.

NYSDEC will use the information in the investigation report to determine if the site poses a significant threat to public health or the environment. If the site is a "significant threat," it must be cleaned up using a remedy selected by NYSDEC from an analysis of alternatives prepared by the Applicant and approved by NYSDEC. If the site does not pose a significant threat, the Applicant may select the remedy from the approved analysis of alternatives.

## Interim Remedial Measures

An Interim Remedial Measure (IRM) is an action that can be undertaken at a site when a source of contamination or exposure pathway can be effectively addressed before the site investigation and analysis of alternatives are completed. If an IRM is likely to represent all or a significant part of the final remedy, NYSDEC will require a 30-day public comment period.

### **Remedy Selection**

When the investigation of the site has been determined to be complete, the project likely would proceed in one of two directions:

1. The Applicant may recommend in its investigation report that no action is necessary at the site. In this case, NYSDEC would make the investigation report available for public comment for 45 days. NYSDEC then would complete its review, make any necessary revisions, and, if appropriate, approve the investigation report. NYSDEC would then issue a "Certificate of Completion" (described below) to the Applicant.

### or

2. The Applicant may recommend in its investigation report that action needs to be taken to address site contamination. After NYSDEC approves the investigation report, the Applicant may then develop a cleanup plan, officially called a "Remedial Action Work Plan". The Remedial Action Work Plan describes the Applicant's proposed remedy for addressing contamination related to the site.

When the Applicant submits a draft Remedial Work Plan for approval, NYSDEC would announce the availability of the draft plan for public review during a 45-day public comment period.

## **Cleanup Action**

NYSDEC will consider public comments, and revise the draft cleanup plan if necessary, before approving the proposed remedy. The New York State Department of Health (NYSDOH) must concur with the proposed remedy. After approval, the proposed remedy becomes the selected remedy. The selected remedy is formalized in the site Decision Document.

The Applicant may then design and perform the cleanup action to address the site contamination. NYSDEC and NYSDOH oversee the activities. When the Applicant completes cleanup activities, it will prepare a final engineering report that certifies that cleanup requirements have been achieved or will be achieved within a specific time frame. NYSDEC will review the report to be certain that the cleanup is protective of public health and the environment for the intended use of the site.

## Certificate of Completion

When NYSDEC is satisfied that cleanup requirements have been achieved or will be achieved for the site, it will approve the final engineering report. NYSDEC then will issue a Certificate of Completion (COC) to the Applicant. The COC states that cleanup goals have been achieved, and relieves the Applicant from future liability for site-related contamination, subject to certain conditions. The Applicant would be eligible to redevelop the site after it receives a COC.

## Site Management

The purpose of site management is to ensure the safe reuse of the property if contamination will remain in place. Site management is the last phase of the site cleanup program. This phase begins when the COC is issued. Site management incorporates any institutional and engineering controls required to ensure that the remedy implemented for the site remains protective of public health and the environment. All significant activities are detailed in a Site Management Plan.

An *institutional control* is a non-physical restriction on use of the site, such as a deed restriction that would prevent or restrict certain uses of the property. An institutional control may be used when the cleanup action leaves some contamination that makes the site suitable for some, but not all uses.

An *engineering control* is a physical barrier or method to manage contamination. Examples include: caps, covers, barriers, fences, and treatment of water supplies.

Site management also may include the operation and maintenance of a component of the remedy, such as a system that pumps and treats groundwater. Site management continues until NYSDEC determines that it is no longer needed.

## Appendix A -Project Contacts and Locations of Reports and Information

## **Project Contacts**

For information about the site's investigation and cleanup program, the public may contact any of the following project staff:

## New York State Department of Environmental Conservation (NYSDEC):

Kimberly Junkins Project Manager NYSDEC Division of Environmental Conservation Region 3 Headquarters 21 South Putt Corners New Paltz, NY 12561 (845) 633- 5457 kimberly.junkins@dec.ny.gov

## New York State Department of Health (NYSDOH):

Stephen Lawrence Project Manager NYSDOH Empire State Plaza Corning Tower Room 1787 Albany, NY 12237 (518) 402-7860 <u>stephen.lawrence@health.ny.gov</u>

## Locations of Reports and Information

The facilities identified below are being used to provide the public with convenient access to important project documents:

New Rochelle Public Library Attn: Tom Geoffino 1 Library Plaza New Rochelle, NY 10801 Phone: (914) 632-7878 If repositories are temporarily unavailable due to COVID-19 precautions, you can access the online repository at: <u>https://gisservices.dec.ny.gov/gis/dil/</u>

## Appendix B - Site Contact List

Federal and State Officials		
Chuck E. Schumer U.S. Senate 780 Third Ave, Suite 2301 New York, NY 10017	Kirsten Gillibrand U.S. Senate 780 Third Ave, Suite 2601 New York, NY 10017 <u>casework@gillibrand.senate.gov</u>	Eliot L. Engel U.S. House of Representatives- 16th Congressional District 6 Gramatan Ave, Suite 205 Mt. Vernon, NY 10550 https://engel.house.gov/email/
Pete Harckham New York State Senator- NY Senate District 40 1 Park Place, Suite 302 Peekskill, NY 10566 <u>harckham@nysenate.gov</u>	George Latimer Westchester County Executive 148 Martine Ave, Suite 900 White Plains, NY 10601 <u>CE@westchestergov.com</u>	Richard Hyman Westchester County Planning Board, Chairperson 148 Martine Ave, Room 420 White Plains, NY 10601 <u>nvv1@westchestergov.com</u>
Timothy C. Idoni Westchester County Clerk 111 Dr. Martin Luther King Jr. Blvd White Plains, NY 10601 <u>tci2@westchestergov.com</u>		
Local Officials		
Noam Bramson Mayor of New Rochelle 515 North Avenue New Rochelle, NY 10801 <u>nbramson@newrochelleny.com</u>	Sarah C. Dobbs-Brown New Rochelle Planning Board, Chairperson 90 Beaufort Place (City Hall)- Room B-1 New Rochelle, NY 10801 <u>kkain@newrochelleny.com</u>	
Public Water Supplier		
Katie Marino Mount Kisco Water Bureau, Public Water Supplier Village Hall (1st Floor) 104 Main Street		

Mount Kisco, NY 10549		
Media Outlet		
The Journal News 1133 Westchester Avenue, Suite N110 White Plains, NY 10604		
Sc	hools and Other Organization	ons
Jennifer DiCosimo Executive Director of Hallen School 97 Ce+D19:G39ntre Avenue New Rochelle, NY 10801	Camille Edwards Thomas House Principal of Campus Alternative High School 50 Washington Avenue New Rochelle, NY 10801	Takashi Katayama Principal of Japanese Weekend School of New York 56 Harrison Street, Suite 503 New Rochelle, NY 10801
Anthony Bongo Principal of Isaac E. Young Middle School 270 Centre Avenue New Rochelle, NY 10805	Michael Hilderbrand Principal of Trinity Elementary School 180 Pelham Road New Rochelle, NY 10805	Jennifer E. Jones Owners/ Director of the Growing Minds of New York 466 Main Street, Suite LL20 New Rochelle, NY 10801
Carmen M. Youngs Owner of Little Rascals Daycare 18 Badeau Place New Rochelle, NY 10801	Lynn Ann Zazzali Owner of the Learning Experience 1 Bally Place New Rochelle, NY 10801	
Adjacent Property Owners		
Rev. Dr. David Holder New York Covenant Church, Property Owner 500 Main Street New Rochelle, NY 10801	Jean Luzincourt, Pastor French Speaking Baptist Church of New Rochelle, Property Owner 12 Church Street New Rochelle, NY 10801	Jonathon Newman Owners of Liebman's Children's Clothing, Property Operator 510 Main Street New Rochelle, NY 10801
The City of Rochelle Adjacent Property Owner of Clinton Place 515 North Avenue New Rochelle, NY 10801	Mipor Associates LLC Adjacent Property Owner of 514 Main Street 72-29 137th Street Flushing, NY 11367	Palace Prime Realty LLC Adjacent Property Owner of 518 Main Street 11 Glen Oaks Drive Rye, NY 10580

Miele Realty LLC	507 MS LLC	505 Main Street Associates,
Adjacent Property Owner of	Adjacent Property Owner	LLC
519 Main Street	of 1 Memorial Highway	Adjacent Property Owner of
45 Pintard Avenue	28 Liberty Street	505 Main Street
New Rochelle, NY 10801	New York, NY 10005	455 Central Park Avenue,
		Suite 205
		Scarsdale, NY 10583
John Ashley Kim LLC	499 Main St., LLC	2-4 Lawton LLC
Adjacent Property	Adjacent Property	Adjacent Property
Owner/Operator of 503	Owner/Operator of 499	Owner/Operator of 497
Main Street	Main Street	Main Street
503 Main Street	499 Main Street	497 Main Street
New Rochelle, NY 10801	New Rochelle, NY 10801	New Rochelle, NY 10801
·····	,	,
JP Morgan Chase Bank	Woodworth Realty LLC	The Salvation Army
Adjacent Property Owner of	Adjacent Property Owner	Adjacent Property
491 Main Street	of 490 Main Street	Owner/Operator of 22
270 Park Avenue	11 Echo Bay Drive	Church Street
New York. NY 10017	New Rochelle, NY 10805	22 Church Street
,		New Rochelle, NY 10801
Cricket Wireless Authorized	Colony Taxi Corp	Starlite Luncheonette
Retailer	Adjacent Property	Adjacent Property Operator
Adjacent Property Operator	Operator of 514 Main	of 519 Main Street
of 514 Main Street	Street	519 Main Street
514 Main Street	514 Main Street	New Rochelle, NY 10801
New Rochelle, NY 10801	New Rochelle, NY 10801	,
Lupitas Tattoos	Queen's Wig	Exotic Nails
Adjacent Property Operator	Adjacent Property	Adjacent Property Operator
of 519 Main Street	Operator of 505 Main	of 505 Main Street
519 Main Street	Street	505 Main Street. Suite 2
New Rochelle, NY 10801	505 Main Street	New Rochelle, NY 10801
,	New Rochelle, NY 10801	,
Embajada Latina	Elegant Beauty Supply	Mexican Corner
Adjacent Property Operator	Adjacent Property	Adjacent Property Operator
of 503 Main Street	Operator of 503 Main	of 497 Main Street
503 Main Street	Street	497 Main Street
New Rochelle, NY 10801	503 Main Street	New Rochelle, NY 10801
	New Rochelle, NY 10801	,
Mexican Corner		
Adjacent Property Operator		
of 497 Main Street 497 ain		
New Rochelle, NY 10801		

Appendix C - Site Location Map



## **Appendix D– Brownfield Cleanup Program Process**





Department of Environmental Conservation

**Division of Environmental Remediation** 

## Remedial Programs Scoping Sheet for Major Issues of Public Concern (see instructions)

Site Name: 500 Main Street Laundry Site

Site Number: C360199

**Site Address and County:** 500, 506 and 510 Main Street and 12 Church Street, New Rochelle, Westchester, NY

Remedial Party(ies): BRP 500 Main LLC

Note: For Parts 1. – 3. the individuals, groups, organizations, businesses and units of government identified should be added to the site contact list as appropriate.

**Part 1.** List major issues of public concern and information the community wants. Identify individuals, groups, organizations, businesses and/or units of government related to the issue(s) and information needs. Use this information as an aid to prepare or update the Major Issues of Public Concern section of the site Citizen Participation Plan.

The list of potential impacts contained in the CPP are typical impacts of remediation on brownfield sites.

How were these issues and/or information needs identified? See response above.

**Part 2.** List important information needed **from** the community, if applicable. Identify individuals, groups, organizations, businesses and/or units of government related to the information needed. Nothing is needed from the community at this time

How were these information needs identified? NA

**Part 3.** List major issues and information that need to be communicated **to** the community. Identify individuals, groups, organizations, businesses and/or units of government related to the issue(s) and/or information.

Communication of each step in the BCP process must be communicated in Fact Sheets and public hearings if required.

How were these issues and/or information needs identified? This is part of the CPP process.

**Part 4.** Identify the following characteristics of the affected/interested community. This knowledge will help to identify and understand issues and information important to the community, and ways to effectively develop and implement the site citizen participation plan (mark all that apply):

**a.** Land use/zoning at and around site:

$\boxtimes$ F	Residential	Agricultural	Recreational	Commercial	Industrial

**b.** Residential type around site:

🛛 Urban 🗌 Suburban 🗆 Rural

**c.** Population density around site:

 $\boxtimes$  High  $\Box$  Medium  $\Box$  Low

d.	Water su	pply	of nearby resid	dences:
$\boxtimes$	Public		<b>Private Wells</b>	Mixed

**e.** Is part or all of the water supply of the affected/interested community currently impacted by the site?  $\Box$  Yes  $\boxtimes$  No

Provide details if appropriate: Click here to enter text.

f. Other environmental issues significantly impacted/impacting the affected community?  $\boxtimes$  Yes  $\ \Box$  No

There are a number of brownfield sites in downtown New Rochelle which appear to be collectively causing area-wide groundwater contamination. Fortunately, there are a number of BCP projects ongoing, which should collectively reduce the area-wide groundwater contamination.

**g.** Is the site and/or the affected/interested community wholly or partly in an Environmental Justice Area? ⊠ Yes □ No

h. Special considerations: ⊠ Language □ Age □ Transportation □ Other

Explain any marked categories in **h**: Large Hispanic Population

**Part 5.** The site contact list must include, at a minimum, the individuals, groups, and organizations identified in Part 2. of the Citizen Participation Plan under 'Site Contact List'. Are *other* individuals, groups, organizations, and units of government affected by, or interested in, the site, or its remedial program? (Mark and identify all that apply, then adjust the site contact list as appropriate.)

□ Non-Adjacent Residents/Property Owners: Click here to enter text.

- □ Local Officials: Click here to enter text.
- □ **Media:** Click here to enter text.
- **Business/Commercial Interests:** Click here to enter text.
- □ Labor Group(s)/Employees: Click here to enter text.
- □ Indian Nation: Click here to enter text.
- □ Citizens/Community Group(s): Click here to enter text.
- **Environmental Justice Group(s):** Click here to enter text.
- **Environmental Group(s):** Click here to enter text.
- **Civic Group(s):** Click here to enter text.
- **Recreational Group(s):** Click here to enter text.
- **Other(s):** Click here to enter text.

Prepared/Updated By: Linda R. Shaw, Esq.

Date: 6/11/2020

Reviewed Approved By: Click here to enter text.

**Date:** Click here to enter text.

## **APPENDIX G**

## **Stormwater Pollution Prevention Plan**

## STORMWATER POLLUTION PREVENTION PLAN (SWPPP)

Proposed 500 Main Street

500 Main Street, New Rochelle, NY 10801

June 2018

**Revised: September 2020** 

Prepared for:

BRP 500 Main LLC 767 Third Avenue, 33<sup>rd</sup> Floor New York, NY 10017

Prepared by:



Michael W. Junghans, P.E.

New York License # 072072



## **Table of Contents**

1. Introduction	2
2. Project Description	
2.1 - Overview	3
2.2 - Drainage	
2.2 - Hydrologic Analysis	
2.3 – Water Quality	5
3 - Erosion and Sediment Control	10
4 - Soil Restoration	12
5 - General Construction Stages	14
6 - Construction Inspection and Maintenance	15
7- Spill Prevention/Control and Usage of Fertilizers	16
8- Appendices	17
Appendix 1 – NYSDEC SPDES Permit No. GP-0-20-001 Appendix 2 – Certificate Statement Appendix 3 – Existing and Proposed Drainage Plans Appendix 4 – Existing Hydrologic Analysis	
Appendix 4 – Existing Hydrologic Analysis Appendix 5 – Proposed Hydrologic Analysis Appendix 6 – Inspection and Maintenance Schedule Appendix 7 – Sample of Construction Site Log Book Appendix 8 – NYS Dishwasher Detergent and Nutrient Runoff Law	
· · · · · · · · · · · · · · · · · · ·	

Appendix 9 – Water Quality Calculations

## 1. Introduction

This Stormwater Pollution Prevention Plan (SWPPP) is prepared and shall be maintained by the owner/operator according to the general guidelines from the New York State Department of Environmental Conservation (NYSDEC) SPDES General Permit for Stormwater Discharges from Construction Activities (Permit Number GP-0-20-001).

The owner/operator shall perform the following activities:

- Owner/Operator shall read and understand the general terms of the SPDES Permit No GP-0-20-001. A copy of the NYSDEC SPDES Permit is included in Appendix 1 of the SWPPP.
- Prior to the commencement of construction activities, the owner/operator must identify the contractor(s) and subcontractor(s) that will be responsible for implementing the practices in the SWPPP.
- Owner/Operator shall have each contractor(s) and subcontractor(s) sign a certificate statement prior to commencement of construction activity. The certificate statement is included in Appendix 2 of the SWPPP. The owner/operator shall attach the signed certificate statement to the copy of the SWPPP that is maintained at the construction site.
- > Perform inspections and maintenance as designated in the SWPPP.
- > Update plans, as necessary, to document major site changes.
- Document any spills.
- Owner/Operator shall retain a copy of the SWPPP at the site during the construction period. The SWPPP is a dynamic document, and must be continually updated and/or revised as necessary by the owner/operator throughout the construction.

## 2. Project Description

## 2.1 - Overview

The proposed site is located at 500 Main Street / 12 Church Street in the City of New Rochelle, NY (the "Site") as shown in Figure 1. The site is bounded by Main Street to the north, Church Street to the west and existing commercial areas to the east and south. Based on Federal Emergency Management Agency Flood Insurance Rate Maps (36119C0341F) as shown in Figure 2, the project site is not located within the 100-year floodplain limit.

The proposed project is an approximately 0.785-acre redevelopment project with no increase of impervious coverage. The proposed development will replace two existing churches and commercial areas with a proposed church, commercial and residential areas.

Since the entire existing site is covered with impervious area, the hydrologic soil type modeled will be considered Type D.

## 2.2 - Drainage

The drainage patterns for the project are relatively the same under existing and proposed conditions. All the onsite runoff will drain to the existing offsite city drainage pipe system.

Under existing conditions, the entire site area of 0.785 acre is covered with impervious area. Under proposed conditions, there will be no increase in impervious coverage and as such there will be no increase in stormwater runoff as compared to the existing condition. The existing and proposed drainage area plans are provided in Appendix 3.

Due to the limit of disturbance is less than one acre and no increase in impervious coverage, NYSDEC requirements of water quality, runoff reduction and water quantity are all waived. However, to comply with City of New Rochelle requirements, a Cascade water quality treatment device, manufactured by Contech Stormwater Solutions, will be provided. Please refer to Section 2.3 "Water Quality" and Appendix 9 for water quality calculations.

Since the project site is located within the Downtown Overlay Zone (DOZ) in City of New Rochelle, the City requires the proposed development to temporarily detain 3.5" of rainfall. A subsurface detention system has been provided in the proposed conditions to satisfy this requirement.

## 2.2 - Hydrologic Analysis

HydroCAD software is utilized to perform hydrologic analysis for the project. Times of concentration and CN values were calculated under existing and proposed conditions by following TR-55 Handbook Guideline for each drainage area. Minimum times of concentration of 6 minutes is utilized for impervious area and 10 minutes for pervious area.

Flows were established for existing and developed conditions utilizing the SCS Method. The hydrograph calculations and summations were prepared using the HydroCAD software. To establish these flows, the standard 1-, 10-, and 100-year, 24-hour storm precipitation values were extracted from the NYSDEC Stormwater Management Design Manual (January 2015) and utilized. A three and half (3.5) inches precipitation also added to HydroCAD for City of New Rochelle requirements. The proposed underground detention system is sized to temporarily detain the 3.5-inches runoff as required by City of New Rochelle.

From HydroCAD analysis, all the proposed conditions peak flows are less than the existing peak flows for all storms ranging from the 1-year to the 100-year design frequencies. The proposed detention system is designed to temporarily store the 3.5-inches runoff volume of 0.22 ac-ft generated from the proposed impervious roof area as required by City of New Rochelle.

Summary Tables #1 and #2 below show the comparison of flows produced under existing and developed conditions.

Refer to Appendix 4 and 5 for supporting calculations for the hydrologic analysis for both existing and proposed conditions.

Return Frequency	Peak Flow Rate, Q (ft <sup>3</sup> /s)
1-year	2.08
3.5-inches	2.61
10-year	3.94
100-year	6.78

### Table 1: Existing Conditions Peak Flow Rate for Various Storm

Table 2. I Toposeu Conultons I eak Tiow Nate for Various Storn
--

Return Frequency	Peak Flow Rate, Q (ft <sup>3</sup> /s)
1-year	0.40
3.5-inches	0.46
10-year	0.60
100-year	1.94

Note: All proposed peak flow rates are less than existing peak flow rates.

## 2.3 – Water Quality

The NYS Stormwater Management Design Manual (SMDM) provides a six-step process that integrates site-planning, usage of green infrastructure practices and standard stormwater management practices to treat stormwater. The six steps process are:

- 1. Site Planning to preserve natural area and reduce impervious cover,
- 2. Calculate initial required Water Quality Volume for the site,
- 3. Provide Runoff Reduction by incorporating green infrastructure technique and standard stormwater management practice (SMP) with Runoff Reduction Volume (RRv) capacity,
- 4. Calculate minimum RRv required,
- 5. Provide standard SMP's to treat remaining portion of water quality volume (WQv) not addressed by green infrastructure and standard SMP's with RRv capacity, and
- 6. Provide volume and peak rate control practices where required.

Following further discusses each of the six steps process in detail.

## Step 1: Site Planning

During site planning process, the designer tries to conserve natural resources and reduce proposed impervious coverage to reduce the impact of water quality from proposed development.

Due to the nature of the project, redevelopment of the site will maintain the existing amount of impervious coverage, Step 1 is not applicable.

## Step 2: Determine the required Water Quality Volume (WQv)

Required WQv was calculated for the site based on 90% rule as per Chapter 4 of New York State Stormwater Management Design Manual (NYS-SMDM). Following equation is used to calculate the water quality volume:

WQv (ac-ft) = (P)(R)(A) / 12

Where: P = 90% Rainfall Event = 1.5 inches Rv = 0.05 + 0.009(I) I = Percentage of impervious cover A = Drainage area in acres.

The required WQv is 0.093 ac-ft. Please refer to Appendix 9 for water quality calculations.

## Step 3: Runoff Reduction Volume (RRv)

RRv requirement can be achieved through application of green infrastructure and standard SWM with runoff reduction capacity.

Since this is a redevelopment project with no increase in impervious area, the runoff reduction requirement is waived.

## **Step 4: Determine the minimum Runoff Reduction Volume (RRv)**

The percent runoff reduction of the impervious area to be constructed on site is based on the Hydrologic Soil Groups present on the site and is determined by the Specific Reduction Factor (S). The following equation is used to determine the minimum runoff reduction volume:

 $RRv (ac-ft) = (P)(Rv^*)(Aic)(S) / 12$ 

Where: P = 90% Rainfall Event = 1.5 inches  $Rv^* = 0.05 + 0.009(I) = 0.95$  where I is 100% impervious Aic = total area of new impervious cover S = hydrologic soil ground (HSG) specific reduction factor (S)

Since this is a redevelopment project with no increase of impervious area, the runoff reduction requirement is waived.

### Step 5: Water Quality Volume by Standard or NYSDEC Certified Stormwater Management Practice

Cascade water quality treatment devices manufactured by Contech Stormwater Solutions is proposed to treat the required water quality as required. The Cascade treatment device is a proprietary water quality device that approved by NYSDEC to be used for redevelopment sites. Since it is a flow rate-based device, the proposed Cascade unit(s) is sized based upon 100% water quality flow rate with internal by-pass capable of by-passing the 100-year storm.

Refer to Appendix 9 for supporting calculations.

## **Step 6: Apply Volume and Peak Rate Control Practices**

This step includes satisfying the requirements for the channel protection volume (CPv), overbank flood control (Qp) and extreme flood control (Qf).

Since this is a redevelopment project with reduction of impervious area, channel protection requirement is not required.

The no increase of impervious area under proposed conditions will result in no increase of peak rates for both 10-year and 100-year storm events. Therefore, no detention system for extreme flood control is required by the NYSDEC. However, since the project site is located within the Downtown Overlay Zone

(DOZ) in City of New Rochelle, the City requires the proposed development to temporarily detain 3.5" of rainfall. A subsurface detention system has been provided in the proposed conditions to satisfy this City requirement.





FIGURE 1 - SITE LOCATION MAP JUNE 2018 500 Main Street New Rochelle, New York





## **3 - Erosion and Sediment Control**

The purposes of providing erosion and sediment control is to minimize temporary impacts to downgradient open water during any construction activities by controlling runoff and retaining sediment as much as possible within the site. Refer to site plan for proposed erosion control practices and details. The erosion and sediment control practices shall base on the guidelines from the latest NYSDEC SPDES General Permit GP-0-20-001 and New York Standards and Specifications for Erosion and Sediment Control (November 2016).

Erosion and sediment control practices includes, but not limiting to, providing the following activities by the owner/operator:

A) Silt Fence

A temporary barrier of geotextile fabric installed on the contours across a slope used to intercept sediment laden runoff from drainage area of disturbed soil by temporary ponding the sediment laden runoff allowing settling to occur.

B) Straw Bale

A temporary barrier of straw used to intercept sediment laden runoff from drainage area of disturbed soil to reduce runoff velocity and effect deposition of the transported sediment load.

C) Inlet Protection

A temporary barrier with low permeability, installed around inlets in the form of fence, berm or excavation around an opening, detaining water and thereby reducing the sediment content of sediment laden water by settling thus preventing heavily sediment laden water from entering a storm drainage system.

D) Dust Control

Water shall be strayed from water truck during construction activity to prevent dust from forming and minimize sediment transport that may cause off-site damage, health hazards or traffic safety problem.

E) Pavement Sweeping

Pavement sweeping will remove sediments from the paved surfaces directly thus preventing sediment from stormwater runoff.

F) Catch Basin Cleaning

Sediments that are not removed by pavement sweeping or inlet protection practices will be drained by stormwater runoff into the site's catch basin system. Catch basin shall be cleaned on a regular basis to make sure the catch basin system function as intended.
#### G) Stabilized Construction Entrance

A stabilized pad of aggregate underlain with geotextile located at any point where traffic will be entering or leaving a construction site to or from a public right-of-way, street, or parking area. The purpose of stabilized construction entrance is to reduce or eliminate the tracking of sediment onto public right-of way or streets.

#### H) Mulching

Applying coarse plant residue or chips, or other suitable materials, to cover the soil surface to provide initial erosion control while a seeding or shrub planting is establishing. Mulch will conserve moisture and modify the surface soil temperature and reduce fluctuation of both. Mulch will prevent soil surface crusting and aid in weed control.

#### I) Concrete Truck Washout

A temporary excavated or above ground lined constructed pit where concrete truck mixers and equipment can be washed after their loads have been discharged, to prevent highly alkaline runoff from entering the storm drainage systems or leaching into soil.

kimley-horn.com

# 4 - Soil Restoration

Soil restoration is required practice by NYSDEC applied across area of a development site where soils have been disturbed and will be vegetated in order to recover the original properties and porosity of the soil. Healthy soil is vital to a sustainable environment and landscape.

Soil restoration is applied in the cleanup, restoration and landscaping phase of construction followed by the permanent establishment of an appropriate, deep-rooted groundcover to help maintain the restored soil structure. Table below summarized the soil restoration requirements by NYSDEC

Soil Restoration Requirements						
Type of Soil Disturbance	Soil Restoration Requirement		Comments/Examples			
No soil disturbance	Restoration not permitted		Preservation of natural features			
Minimal soil disturbance	Restoration not required		Clearing and grubbing			
Areas where topsoil is stripped	HSG A & B	HSG C & D	Protect area from any ongoing			
omy no enange in grade	Apply 6 inches of topsoil	Aerate* and apply 6 inches of topsoil				
Areas of cut or fill	HSG A & B	HSG C & D				
	Aerate and apply 6 inches of topsoil	Apply full soil restoration**				
Heavy traffic areas on site (especially in a zone 5-25 feet around buildings but not within a 5 foot perimeter around foundation walls	Apply full soil restora and compost enhancer	tion (de-compaction nent)				
Areas where runoff reduction and/or infiltration practices are applied	Restoration not required, but may be applied to enhance the reduction specified for appropriate practices.		Keep construction equipment from crossing these areas. To protect newly installed practice from any ongoing construction activities construct a single phase operation fence area			
Redevelopment projects	Soil restoration is required on redevelopment projects in areas where existing impervious area will be converted to pervious area.					

\*Aeration includes the use of machines such as tractor-drawn implements with coulters making a narrow slit in the soil, a rolloer with many spikes making indentations in the soil, or prongs which function like a mini-subsoiler.

\*\*Per "Deep Ripping and De-compaction, DEC 2008"

During periods of relatively low to moderate subsoil moisture, the disturbed subsoils are returned to rough grade and the following soil restoration steps applied:

- 1. Apply 3 inches of compost over subsoil
- 2. Till compost into subsoil to a depth of at least 12 inches using a cat-mounted ripper, tractormounted disc. Or tiller, mixing, and circulating are and compost into subsoils
- 3. Rock-pick until uplifted stone/rock materials of four inches and larger site are cleaned off the site
- 4. Apply topsoil to a depth of 6 inches
- 5. Vegetate as required by approved plan

At the end of the project an inspector should be able to push a 3/8" metal bar 12 inches into the soil just with body weight.

# **5 - General Construction Stages**

Site development in general will occur in three overlapping stages:

- 1. Site Preparation and Erosion Control
- 2. Construction
- 3. Final grading and Stabilization

#### 1) Site Preparation and Erosion Control

Prior to any construction activities, erosion control measures shall be implement to minimize or control erosion on site. These include but not limited to silt fence, straw bale, inlet protection, stabilized construction entrance, concrete truck wash-out area and stock pile area. Fencing will shall be placed around trees to be protected. Other site preparation including setting up of staging area, construction fence, temporary access road.

Duration for site preparation and erosion control work = 2 weeks

#### 2) Construction

After the site preparation and erosion control works, contractor can begin site demolition such as pavement removal, building demolition, utilities removal and other site demolition. Contractor can begin proposed utilities installation, building and road constructions, onsite rough grading and other proposed improvement works such as lighting and landscaping.

Duration for construction work = 12 months

#### 3) Final Grading and Stabilization

Final site grading and stabilization shall be completed as soon as practicable to eliminate exposed soil and minimize erosion. Proposed pavement area shall be covered by bituminous pavement after final subgrades are set.

Contractor shall plan a permanent top soil, seed, mulch and install other stabilization practices in the disturbed areas as appropriate. Stabilization shall be undertaken no later than 14 days after construction activities has ceased except as noted in the GP-0-20-001.

All temporary control measures shall be removed once the site has been stabilized. All litter shall be removed from site.

Duration for final grading and stabilization work = 1 months

# **6** - Construction Inspection and Maintenance

The SPDES General Permit GP-0-20-001 requires that the owner/operator be responsible for inspecting and maintaining the erosion control practices implementing on site. The owner/operator must document compliance with the permit throughout the entire construction process.

#### A) Inspection

- The owner/operator shall have a qualified inspector inspect all erosion and sediment control practices to ensure their integrity and effectiveness throughout the entire construction process.
- The qualified inspector shall perform inspection at least once every seven (7) calendar days.
- Within one business day of the completion of an inspection, the qualified inspector shall notify the owner/operator and appropriate contractor or subcontractor of any corrective actions shall be taken.
- The qualified inspector shall prepare an inspection report in accordance with the permit subsequent to each and every inspection. The owner/operator shall maintain a record of all inspection reports in a site log book as part of the updated SWPPP and shall be make available upon request by permitting authority.
- B) Maintenance
  - Sediment shall be removed from behind silt fence or straw bale if accumulation of greater than 6-inches deep or as needed.
  - Sediment that is collected in inlet protection practice shall be removed on a regular basis to ensure the integrity of the drainage inlet system.
  - The underside of straw bale shall be kept in close contact with the ground surface.
  - Straw bale and silt fence that are damaged shall be replaced or as necessary.
  - On site's paved areas shall be swept on an as needed basis during the construction process.
  - The contractor or subcontractor shall begin implementing the corrective actions within one business day of the notification from qualified inspector and shall complete the corrective actions within a reasonable time frame.

Refer to Appendix 6 of the SWPPP for inspection and maintenance schedule; and refer to Appendix 7 for sample of construction site log book.

# 7- Spill Prevention/Control and Usage of Fertilizers

#### Spill Prevention

The following material management practices shall be implemented to minimize the risk of spills of material or substances to stormwater runoff:

- All materials stored onsite will be stored in an organized and proper manner in their appropriate containers, and (if possible) in a building or other enclosure.
- Products will be kept in their original containers and their original label.
- Un-used, remaining products will be stored in an appropriate manner to prevent leakage.
- No petroleum products or fertilizers shall be stored or handled within 100 feet of a wetland or waterway.
- Substance will not be mixed with one another unless recommended by the manufacturer and in a safe environment.
- Concrete trucks will not be allowed to wash out or discharge surplus concrete within 100 feet of wetland and waterway or into existing catch basins.
- Disposal of the products shall follow the manufacturer's recommendation.
- The contractor shall inspect the storage area daily to ensure proper use and disposal of the material onsite.

#### Spill Control

The following management practices shall be implemented for spill control, notification and cleanup:

- Manufacturer's recommended methods for spill cleanup shall be posted onsite and personnel shall be informed of the cleanup procedure.
- Cleanup material and supplies shall be adequately provided onsite at all time. These includes, but not limited to, shovels, brooms, dustpans, rags, mops, goggles, speedy-dry sand, metal trash containers.
- All spills shall be cleaned up immediately after discovery.
- Personnel cleaning up the spills shall wear proper protective clothing to avoid injury.
- Spill of hazardous material that cannot be cleaned up properly shall be reported to the NYSDEC Spill Hotline: 1-800-457-7362 or others appropriate agency.

#### Usage of Fertilizers

Fertilizer onsite shall only be used in the minimum amounts recommended by the manufacturer and in strict conformance with the 2012 NYS Dishwasher Detergent and Nutrient Runoff Law. No fertilizer storage shall occur within 100 feet of a wetland or waterway. Refer to Appendix 8 for NYS Dishwasher Detergent and Nutrient Runoff Law.

# 8- Appendices

# Appendix 1

NYSDEC SPDES Permit No. GP-0-20-001



Department of Environmental Conservation

### NEW YORK STATE DEPARTMENT OF ENVIRONMENTAL CONSERVATION

#### SPDES GENERAL PERMIT FOR STORMWATER DISCHARGES

From

#### CONSTRUCTION ACTIVITY

Permit No. GP- 0-20-001

Issued Pursuant to Article 17, Titles 7, 8 and Article 70

of the Environmental Conservation Law

Effective Date: January 29, 2020

Expiration Date: January 28, 2025

John J. Ferguson

Chief Permit Administrator

Authorized Signature

1-23-20

Date

Address: NYS DEC Division of Environmental Permits 625 Broadway, 4th Floor Albany, N.Y. 12233-1750

#### PREFACE

Pursuant to Section 402 of the Clean Water Act ("CWA"), stormwater *discharges* from certain *construction activities* are unlawful unless they are authorized by a *National Pollutant Discharge Elimination System ("NPDES")* permit or by a state permit program. New York administers the approved State Pollutant Discharge Elimination System (SPDES) program with permits issued in accordance with the New York State Environmental Conservation Law (ECL) Article 17, Titles 7, 8 and Article 70.

An owner or operator of a construction activity that is eligible for coverage under this permit must obtain coverage prior to the *commencement of construction activity*. Activities that fit the definition of "*construction activity*", as defined under 40 CFR 122.26(b)(14)(x), (15)(i), and (15)(ii), constitute construction of a *point source* and therefore, pursuant to ECL section 17-0505 and 17-0701, the *owner or operator* must have coverage under a SPDES permit prior to *commencing construction activity*. The *owner or operator* cannot wait until there is an actual *discharge* from the *construction site* to obtain permit coverage.

#### \*Note: The italicized words/phrases within this permit are defined in Appendix A.

## NEW YORK STATE DEPARTMENT OF ENVIRONMENTAL CONSERVATION SPDES GENERAL PERMIT FOR STORMWATER DISCHARGES FROM CONSTRUCTION ACTIVITIES

# Table of Contents

Part 1.	PERMIT COVERAGE AND LIMITATIONS	1
Α.	Permit Application	1
В.	Effluent Limitations Applicable to Discharges from Construction Activities	1
C.	Post-construction Stormwater Management Practice Requirements	4
D.	Maintaining Water Quality	8
E.	Eligibility Under This General Permit	9
F.	Activities Which Are Ineligible for Coverage Under This General Permit	9
Part II.	PERMIT COVERAGE	12
Α.	How to Obtain Coverage	12
В.	Notice of Intent (NOI) Submittal	13
C.	Permit Authorization	13
D.	General Requirements For Owners or Operators With Permit Coverage	15
E.	Permit Coverage for Discharges Authorized Under GP-0-15-002	17
F.	Change of Owner or Operator	17
Part III.	STORMWATER POLLUTION PREVENTION PLAN (SWPPP)	18
Α.	General SWPPP Requirements	18
В.	Required SWPPP Contents	20
C.	Required SWPPP Components by Project Type	24
Part IV.	INSPECTION AND MAINTENANCE REQUIREMENTS	24
Α.	General Construction Site Inspection and Maintenance Requirements	24
В.	Contractor Maintenance Inspection Requirements	24
C.	Qualified Inspector Inspection Requirements	25
Part V.	TERMINATION OF PERMIT COVERAGE	29
Α.	Termination of Permit Coverage	29
Part VI.	REPORTING AND RETENTION RECORDS	31
Α.	Record Retention	31
В.	Addresses	31
Part VII	. STANDARD PERMIT CONDITIONS	31
Α.	Duty to Comply	31
В.	Continuation of the Expired General Permit	32
C.	Enforcement	32
D.	Need to Halt or Reduce Activity Not a Defense	32
E.	Duty to Mitigate	33
F.	Duty to Provide Information	33
G.	Other Information	33
Н.	Signatory Requirements	33
I.	Property Rights	35
J.	Severability	35

K.	Requirement to Obtain Coverage Under an Alternative Permit	35
L.	Proper Operation and Maintenance	
M.	Inspection and Entry	
N.	Permit Actions	37
О.	Definitions	37
Ρ.	Re-Opener Clause	37
Q.	Penalties for Falsification of Forms and Reports	37
R.	Other Permits	
APPEN	IDIX A – Acronyms and Definitions	
Acro	nyms	
Defin	nitions	40
APPEN	IDIX B – Required SWPPP Components by Project Type	48
Table	e 1	48
Table	e 2	50
APPEN	IDIX C – Watersheds Requiring Enhanced Phosphorus Removal	52
APPEN	IDIX D – Watersheds with Lower Disturbance Threshold	58
APPEN	IDIX E – 303(d) Segments Impaired by Construction Related Pollutant(s)	59
APPEN	IDIX F – List of NYS DEC Regional Offices	65

## Part 1. PERMIT COVERAGE AND LIMITATIONS

#### A. Permit Application

This permit authorizes stormwater *discharges* to *surface waters of the State* from the following *construction activities* identified within 40 CFR Parts 122.26(b)(14)(x), 122.26(b)(15)(i) and 122.26(b)(15)(ii), provided all of the eligibility provisions of this permit are met:

- 1. Construction activities involving soil disturbances of one (1) or more acres; including disturbances of less than one acre that are part of a *larger common plan of development or sale* that will ultimately disturb one or more acres of land; excluding *routine maintenance activity* that is performed to maintain the original line and grade, hydraulic capacity or original purpose of a facility;
- 2. Construction activities involving soil disturbances of less than one (1) acre where the Department has determined that a *SPDES* permit is required for stormwater *discharges* based on the potential for contribution to a violation of a *water quality standard* or for significant contribution of *pollutants* to *surface waters of the State.*
- 3. *Construction activities* located in the watershed(s) identified in Appendix D that involve soil disturbances between five thousand (5,000) square feet and one (1) acre of land.

#### **B. Effluent Limitations Applicable to Discharges from Construction Activities**

*Discharges* authorized by this permit must achieve, at a minimum, the effluent limitations in Part I.B.1. (a) – (f) of this permit. These limitations represent the degree of effluent reduction attainable by the application of best practicable technology currently available.

 Erosion and Sediment Control Requirements - The owner or operator must select, design, install, implement and maintain control measures to minimize the discharge of pollutants and prevent a violation of the water quality standards. The selection, design, installation, implementation, and maintenance of these control measures must meet the non-numeric effluent limitations in Part I.B.1.(a) – (f) of this permit and be in accordance with the New York State Standards and Specifications for Erosion and Sediment Control, dated November 2016, using sound engineering judgment. Where control measures are not designed in conformance with the design criteria included in the technical standard, the owner or operator must include in the Stormwater Pollution Prevention Plan ("SWPPP") the reason(s) for the deviation or alternative design and provide information which demonstrates that the deviation or alternative design is *equivalent* to the technical standard.

- a. **Erosion and Sediment Controls.** Design, install and maintain effective erosion and sediment controls to *minimize* the *discharge* of *pollutants* and prevent a violation of the *water quality standards*. At a minimum, such controls must be designed, installed and maintained to:
  - (i) *Minimize* soil erosion through application of runoff control and soil stabilization control measure to *minimize pollutant discharges*;
  - (ii) Control stormwater *discharges*, including both peak flowrates and total stormwater volume, to *minimize* channel and *streambank* erosion and scour in the immediate vicinity of the *discharge* points;
  - (iii) *Minimize* the amount of soil exposed during *construction activity*;
  - (iv) *Minimize* the disturbance of *steep slopes*;
  - (v) *Minimize* sediment *discharges* from the site;
  - (vi) Provide and maintain *natural buffers* around surface waters, direct stormwater to vegetated areas and maximize stormwater infiltration to reduce *pollutant discharges*, unless *infeasible*;
  - (vii) *Minimize* soil compaction. Minimizing soil compaction is not required where the intended function of a specific area of the site dictates that it be compacted;
  - (viii) Unless *infeasible*, preserve a sufficient amount of topsoil to complete soil restoration and establish a uniform, dense vegetative cover; and
  - (ix) *Minimize* dust. On areas of exposed soil, *minimize* dust through the appropriate application of water or other dust suppression techniques to control the generation of pollutants that could be discharged from the site.
- b. Soil Stabilization. In areas where soil disturbance activity has temporarily or permanently ceased, the application of soil stabilization measures must be initiated by the end of the next business day and completed within fourteen (14) days from the date the current soil disturbance activity ceased. For construction sites that *directly discharge* to one of the 303(d) segments

listed in Appendix E or is located in one of the watersheds listed in Appendix C, the application of soil stabilization measures must be initiated by the end of the next business day and completed within seven (7) days from the date the current soil disturbance activity ceased. See Appendix A for definition of *Temporarily Ceased*.

- c. **Dewatering**. *Discharges* from *dewatering* activities, including *discharges* from *dewatering* of trenches and excavations, must be managed by appropriate control measures.
- d. **Pollution Prevention Measures**. Design, install, implement, and maintain effective pollution prevention measures to *minimize* the *discharge* of *pollutants* and prevent a violation of the *water quality standards*. At a minimum, such measures must be designed, installed, implemented and maintained to:
  - (i) Minimize the discharge of pollutants from equipment and vehicle washing, wheel wash water, and other wash waters. This applies to washing operations that use clean water only. Soaps, detergents and solvents cannot be used;
  - (ii) Minimize the exposure of building materials, building products, construction wastes, trash, landscape materials, fertilizers, pesticides, herbicides, detergents, sanitary waste, hazardous and toxic waste, and other materials present on the site to precipitation and to stormwater. Minimization of exposure is not required in cases where the exposure to precipitation and to stormwater will not result in a *discharge* of *pollutants*, or where exposure of a specific material or product poses little risk of stormwater contamination (such as final products and materials intended for outdoor use); and
  - (iii) Prevent the *discharge* of *pollutants* from spills and leaks and implement chemical spill and leak prevention and response procedures.
- e. Prohibited Discharges. The following discharges are prohibited:
  - (i) Wastewater from washout of concrete;
  - (ii) Wastewater from washout and cleanout of stucco, paint, form release oils, curing compounds and other construction materials;

- (iii) Fuels, oils, or other *pollutants* used in vehicle and equipment operation and maintenance;
- (iv) Soaps or solvents used in vehicle and equipment washing; and
- (v) Toxic or hazardous substances from a spill or other release.
- f. Surface Outlets. When discharging from basins and impoundments, the outlets shall be designed, constructed and maintained in such a manner that sediment does not leave the basin or impoundment and that erosion at or below the outlet does not occur.

# C. Post-construction Stormwater Management Practice Requirements

- The owner or operator of a construction activity that requires post-construction stormwater management practices pursuant to Part III.C. of this permit must select, design, install, and maintain the practices to meet the *performance criteria* in the New York State Stormwater Management Design Manual ("Design Manual"), dated January 2015, using sound engineering judgment. Where post-construction stormwater management practices ("SMPs") are not designed in conformance with the *performance criteria* in the Design Manual, the owner or operator must include in the SWPPP the reason(s) for the deviation or alternative design and provide information which demonstrates that the deviation or alternative design is *equivalent* to the technical standard.
- 2. The *owner or operator* of a *construction activity* that requires post-construction stormwater management practices pursuant to Part III.C. of this permit must design the practices to meet the applicable *sizing criteria* in Part I.C.2.a., b., c. or d. of this permit.

# a. Sizing Criteria for New Development

- (i) Runoff Reduction Volume ("RRv"): Reduce the total Water Quality Volume ("WQv") by application of RR techniques and standard SMPs with RRv capacity. The total WQv shall be calculated in accordance with the criteria in Section 4.2 of the Design Manual.
- (ii) Minimum RRv and Treatment of Remaining Total WQv: Construction activities that cannot meet the criteria in Part I.C.2.a.(i) of this permit due to site limitations shall direct runoff from all newly constructed impervious areas to a RR technique or standard SMP with RRv capacity unless infeasible. The specific site limitations that prevent the reduction of 100% of the WQv shall be documented in the SWPPP.

For each impervious area that is not directed to a RR technique or standard SMP with RRv capacity, the SWPPP must include documentation which demonstrates that all options were considered and for each option explains why it is considered infeasible.

In no case shall the runoff reduction achieved from the newly constructed impervious areas be less than the Minimum RRv as calculated using the criteria in Section 4.3 of the Design Manual. The remaining portion of the total WQv that cannot be reduced shall be treated by application of standard SMPs.

- (iii) Channel Protection Volume ("Cpv"): Provide 24 hour extended detention of the post-developed 1-year, 24-hour storm event; remaining after runoff reduction. The Cpv requirement does not apply when:
  - (1) Reduction of the entire Cpv is achieved by application of runoff reduction techniques or infiltration systems, or
  - (2) The site discharges directly to tidal waters, or fifth order or larger streams.
- (iv) Overbank Flood Control Criteria ("Qp"): Requires storage to attenuate the post-development 10-year, 24-hour peak discharge rate (Qp) to predevelopment rates. The Qp requirement does not apply when:
  - (1) the site discharges directly to tidal waters or fifth order or larger streams, or
  - (2) A downstream analysis reveals that *overbank* control is not required.
- (v) Extreme Flood Control Criteria ("Qf"): Requires storage to attenuate the post-development 100-year, 24-hour peak discharge rate (Qf) to predevelopment rates. The Qf requirement does not apply when:
  - (1) the site discharges directly to tidal waters or fifth order or larger streams, or
  - (2) A downstream analysis reveals that *overbank* control is not required.

# b. *Sizing Criteria* for *New Development* in Enhanced Phosphorus Removal Watershed

Runoff Reduction Volume (RRv): Reduce the total Water Quality
Volume (WQv) by application of RR techniques and standard SMPs
with RRv capacity. The total WQv is the runoff volume from the 1-year,
24 hour design storm over the post-developed watershed and shall be

calculated in accordance with the criteria in Section 10.3 of the Design Manual.

(ii) Minimum RRv and Treatment of Remaining Total WQv: Construction activities that cannot meet the criteria in Part I.C.2.b.(i) of this permit due to site limitations shall direct runoff from all newly constructed impervious areas to a RR technique or standard SMP with RRv capacity unless infeasible. The specific site limitations that prevent the reduction of 100% of the WQv shall be documented in the SWPPP. For each impervious area that is not directed to a RR technique or standard SMP with RRv capacity, the SWPPP must include documentation which demonstrates that all options were considered and for each option explains why it is considered infeasible.

In no case shall the runoff reduction achieved from the newly constructed *impervious areas* be less than the Minimum RRv as calculated using the criteria in Section 10.3 of the Design Manual. The remaining portion of the total WQv that cannot be reduced shall be treated by application of standard SMPs.

- (iii) Channel Protection Volume (Cpv): Provide 24 hour extended detention of the post-developed 1-year, 24-hour storm event; remaining after runoff reduction. The Cpv requirement does not apply when:
  - (1) Reduction of the entire Cpv is achieved by application of runoff reduction techniques or infiltration systems, or
  - (2) The site *discharges* directly to tidal waters, or fifth order or larger streams.
- (iv) Overbank Flood Control Criteria (Qp): Requires storage to attenuate the post-development 10-year, 24-hour peak discharge rate (Qp) to predevelopment rates. The Qp requirement does not apply when:
  - (1) the site *discharges* directly to tidal waters or fifth order or larger streams, or
  - (2) A downstream analysis reveals that *overbank* control is not required.
- (v) Extreme Flood Control Criteria (Qf): Requires storage to attenuate the post-development 100-year, 24-hour peak *discharge* rate (Qf) to predevelopment rates. The Qf requirement does not apply when:
  - (1) the site *discharges* directly to tidal waters or fifth order or larger streams, or
  - (2) A downstream analysis reveals that *overbank* control is not required.

### c. Sizing Criteria for Redevelopment Activity

- (i) Water Quality Volume (WQv): The WQv treatment objective for redevelopment activity shall be addressed by one of the following options. Redevelopment activities located in an Enhanced Phosphorus Removal Watershed (see Part III.B.3. and Appendix C of this permit) shall calculate the WQv in accordance with Section 10.3 of the Design Manual. All other redevelopment activities shall calculate the WQv in accordance with Section 4.2 of the Design Manual.
  - (1) Reduce the existing *impervious cover* by a minimum of 25% of the total disturbed, *impervious area*. The Soil Restoration criteria in Section 5.1.6 of the Design Manual must be applied to all newly created pervious areas, or
  - (2) Capture and treat a minimum of 25% of the WQv from the disturbed, *impervious area* by the application of standard SMPs; or reduce 25% of the WQv from the disturbed, *impervious area* by the application of RR techniques or standard SMPs with RRv capacity., or
  - (3) Capture and treat a minimum of 75% of the WQv from the disturbed, *impervious area* as well as any additional runoff from tributary areas by application of the alternative practices discussed in Sections 9.3 and 9.4 of the Design Manual., or
  - (4) Application of a combination of 1, 2 and 3 above that provide a weighted average of at least two of the above methods. Application of this method shall be in accordance with the criteria in Section 9.2.1(B) (IV) of the Design Manual.

If there is an existing post-construction stormwater management practice located on the site that captures and treats runoff from the *impervious area* that is being disturbed, the WQv treatment option selected must, at a minimum, provide treatment equal to the treatment that was being provided by the existing practice(s) if that treatment is greater than the treatment required by options 1 - 4 above.

- (ii) Channel Protection Volume (Cpv): Not required if there are no changes to hydrology that increase the *discharge* rate from the project site.
- (iii) Overbank Flood Control Criteria (Qp): Not required if there are no changes to hydrology that increase the *discharge* rate from the project site.
- (iv) Extreme Flood Control Criteria (Qf): Not required if there are no changes to hydrology that increase the *discharge* rate from the project site

# d. Sizing Criteria for Combination of Redevelopment Activity and New Development

Construction projects that include both New Development and Redevelopment Activity shall provide post-construction stormwater management controls that meet the sizing criteria calculated as an aggregate of the Sizing Criteria in Part I.C.2.a. or b. of this permit for the New Development portion of the project and Part I.C.2.c of this permit for Redevelopment Activity portion of the project.

# D. Maintaining Water Quality

The Department expects that compliance with the conditions of this permit will control *discharges* necessary to meet applicable *water quality standards*. It shall be a violation of the *ECL* for any discharge to either cause or contribute to a violation of *water quality standards* as contained in Parts 700 through 705 of Title 6 of the Official Compilation of Codes, Rules and Regulations of the State of New York, such as:

- 1. There shall be no increase in turbidity that will cause a substantial visible contrast to natural conditions;
- 2. There shall be no increase in suspended, colloidal or settleable solids that will cause deposition or impair the waters for their best usages; and
- 3. There shall be no residue from oil and floating substances, nor visible oil film, nor globules of grease.

If there is evidence indicating that the stormwater *discharges* authorized by this permit are causing, have the reasonable potential to cause, or are contributing to a violation of the *water quality standards*; the *owner or operator* must take appropriate corrective action in accordance with Part IV.C.5. of this general permit and document in accordance with Part IV.C.4. of this general permit. To address the *water quality standard* violation the *owner or operator* may need to provide additional information, include and implement appropriate controls in the SWPPP to correct the problem, or obtain an individual SPDES permit.

If there is evidence indicating that despite compliance with the terms and conditions of this general permit it is demonstrated that the stormwater *discharges* authorized by this permit are causing or contributing to a violation of *water quality standards*, or if the Department determines that a modification of the permit is necessary to prevent a violation of *water quality standards*, the authorized *discharges* will no longer be eligible for coverage under this permit. The Department may require the *owner or operator* to obtain an individual SPDES permit to continue discharging.

## E. Eligibility Under This General Permit

- 1. This permit may authorize all *discharges* of stormwater from *construction activity* to *surface waters of the State* and *groundwaters* except for ineligible *discharges* identified under subparagraph F. of this Part.
- 2. Except for non-stormwater *discharges* explicitly listed in the next paragraph, this permit only authorizes stormwater *discharges*; including stormwater runoff, snowmelt runoff, and surface runoff and drainage, from *construction activities*.
- 3. Notwithstanding paragraphs E.1 and E.2 above, the following non-stormwater discharges are authorized by this permit: those listed in 6 NYCRR 750-1.2(a)(29)(vi), with the following exception: "Discharges from firefighting activities are authorized only when the firefighting activities are emergencies/unplanned"; waters to which other components have not been added that are used to control dust in accordance with the SWPPP; and uncontaminated *discharges* from *construction site* de-watering operations. All non-stormwater discharges must be identified in the SWPPP. Under all circumstances, the *owner or operator* must still comply with *water quality standards* in Part I.D of this permit.
- 4. The *owner or operator* must maintain permit eligibility to *discharge* under this permit. Any *discharges* that are not compliant with the eligibility conditions of this permit are not authorized by the permit and the *owner or operator* must either apply for a separate permit to cover those ineligible *discharges* or take steps necessary to make the *discharge* eligible for coverage.

#### F. Activities Which Are Ineligible for Coverage Under This General Permit

All of the following are **<u>not</u>** authorized by this permit:

- 1. *Discharges* after *construction activities* have been completed and the site has undergone *final stabilization*;
- 2. *Discharges* that are mixed with sources of non-stormwater other than those expressly authorized under subsection E.3. of this Part and identified in the SWPPP required by this permit;
- 3. *Discharges* that are required to obtain an individual SPDES permit or another SPDES general permit pursuant to Part VII.K. of this permit;
- 4. Construction activities or discharges from construction activities that may adversely affect an endangered or threatened species unless the owner or

*operator* has obtained a permit issued pursuant to 6 NYCRR Part 182 for the project or the Department has issued a letter of non-jurisdiction for the project. All documentation necessary to demonstrate eligibility shall be maintained on site in accordance with Part II.D.2 of this permit;

- 5. *Discharges* which either cause or contribute to a violation of *water quality standards* adopted pursuant to the *ECL* and its accompanying regulations;
- 6. Construction activities for residential, commercial and institutional projects:
  - a. Where the *discharges* from the *construction activities* are tributary to waters of the state classified as AA or AA-s; and
  - b. Which are undertaken on land with no existing *impervious cover*; and
  - c. Which disturb one (1) or more acres of land designated on the current United States Department of Agriculture ("USDA") Soil Survey as Soil Slope Phase "D", (provided the map unit name is inclusive of slopes greater than 25%), or Soil Slope Phase "E" or "F" (regardless of the map unit name), or a combination of the three designations.
- 7. *Construction activities* for linear transportation projects and linear utility projects:
  - a. Where the *discharges* from the *construction activities* are tributary to waters of the state classified as AA or AA-s; and
  - b. Which are undertaken on land with no existing impervious cover; and

c. Which disturb two (2) or more acres of land designated on the current USDA Soil Survey as Soil Slope Phase "D" (provided the map unit name is inclusive of slopes greater than 25%), or Soil Slope Phase "E" or "F" (regardless of the map unit name), or a combination of the three designations.

- 8. *Construction activities* that have the potential to affect an *historic property*, unless there is documentation that such impacts have been resolved. The following documentation necessary to demonstrate eligibility with this requirement shall be maintained on site in accordance with Part II.D.2 of this permit and made available to the Department in accordance with Part VII.F of this permit:
  - a. Documentation that the *construction activity* is not within an archeologically sensitive area indicated on the sensitivity map, and that the *construction activity* is not located on or immediately adjacent to a property listed or determined to be eligible for listing on the National or State Registers of Historic Places, and that there is no new permanent building on the *construction site* within the following distances from a building, structure, or object that is more than 50 years old, or if there is such a new permanent building on the *construction site* within those parameters that NYS Office of Parks, Recreation and Historic Preservation (OPRHP), a Historic Preservation Commission of a Certified Local Government, or a qualified preservation professional has determined that the building, structure, or object more than 50 years old is not historically/archeologically significant.
    - 1-5 acres of disturbance 20 feet
    - 5-20 acres of disturbance 50 feet
    - 20+ acres of disturbance 100 feet, or
  - b. DEC consultation form sent to OPRHP, and copied to the NYS DEC Agency Historic Preservation Officer (APO), and
    - the State Environmental Quality Review (SEQR) Environmental Assessment Form (EAF) with a negative declaration or the Findings Statement, with documentation of OPRHP's agreement with the resolution; or
    - (ii) documentation from OPRHP that the *construction activity* will result in No Impact; or
    - (iii) documentation from OPRHP providing a determination of No Adverse Impact; or
    - (iv) a Letter of Resolution signed by the owner/operator, OPRHP and the DEC APO which allows for this *construction activity* to be eligible for coverage under the general permit in terms of the State Historic Preservation Act (SHPA); or
  - c. Documentation of satisfactory compliance with Section 106 of the National Historic Preservation Act for a coterminous project area:

- (i) No Affect
- (ii) No Adverse Affect
- (iii) Executed Memorandum of Agreement, or
- d. Documentation that:
- (i) SHPA Section 14.09 has been completed by NYS DEC or another state agency.
- 9. *Discharges* from *construction activities* that are subject to an existing SPDES individual or general permit where a SPDES permit for *construction activity* has been terminated or denied; or where the *owner or operator* has failed to renew an expired individual permit.

#### Part II. PERMIT COVERAGE

#### A. How to Obtain Coverage

- An owner or operator of a construction activity that is not subject to the requirements of a regulated, traditional land use control MS4 must first prepare a SWPPP in accordance with all applicable requirements of this permit and then submit a completed Notice of Intent (NOI) to the Department to be authorized to discharge under this permit.
- 2. An owner or operator of a construction activity that is subject to the requirements of a regulated, traditional land use control MS4 must first prepare a SWPPP in accordance with all applicable requirements of this permit and then have the SWPPP reviewed and accepted by the regulated, traditional land use control MS4 prior to submitting the NOI to the Department. The owner or operator shall have the "MS4 SWPPP Acceptance" form signed in accordance with Part VII.H., and then submit that form along with a completed NOI to the Department.
- 3. The requirement for an owner or operator to have its SWPPP reviewed and accepted by the regulated, traditional land use control MS4 prior to submitting the NOI to the Department does not apply to an owner or operator that is obtaining permit coverage in accordance with the requirements in Part II.F. (Change of Owner or Operator) or where the owner or operator of the construction activity is the regulated, traditional land use control MS4. This exemption does not apply to construction activities subject to the New York City Administrative Code.

#### B. Notice of Intent (NOI) Submittal

 Prior to December 21, 2020, an owner or operator shall use either the electronic (eNOI) or paper version of the NOI that the Department prepared. Both versions of the NOI are located on the Department's website (http://www.dec.ny.gov/). The paper version of the NOI shall be signed in accordance with Part VII.H. of this permit and submitted to the following address:

#### NOTICE OF INTENT NYS DEC, Bureau of Water Permits 625 Broadway, 4<sup>th</sup> Floor Albany, New York 12233-3505

- 2. Beginning December 21, 2020 and in accordance with EPA's 2015 NPDES Electronic Reporting Rule (40 CFR Part 127), the *owner or operator* must submit the NOI electronically using the *Department's* online NOI.
- 3. The *owner or operator* shall have the SWPPP preparer sign the "SWPPP Preparer Certification" statement on the NOI prior to submitting the form to the Department.
- 4. As of the date the NOI is submitted to the Department, the *owner or operator* shall make the NOI and SWPPP available for review and copying in accordance with the requirements in Part VII.F. of this permit.

#### C. Permit Authorization

- 1. An owner or operator shall not commence construction activity until their authorization to discharge under this permit goes into effect.
- 2. Authorization to *discharge* under this permit will be effective when the *owner or operator* has satisfied <u>all</u> of the following criteria:
  - a. project review pursuant to the State Environmental Quality Review Act ("SEQRA") have been satisfied, when SEQRA is applicable. See the Department's website (<u>http://www.dec.ny.gov/</u>) for more information,
  - b. where required, all necessary Department permits subject to the Uniform Procedures Act ("UPA") (see 6 NYCRR Part 621), or the equivalent from another New York State agency, have been obtained, unless otherwise notified by the Department pursuant to 6 NYCRR 621.3(a)(4). Owners or operators of construction activities that are required to obtain UPA permits

must submit a preliminary SWPPP to the appropriate DEC Permit Administrator at the Regional Office listed in Appendix F at the time all other necessary *UPA* permit applications are submitted. The preliminary SWPPP must include sufficient information to demonstrate that the *construction activity* qualifies for authorization under this permit,

- c. the final SWPPP has been prepared, and
- d. a complete NOI has been submitted to the Department in accordance with the requirements of this permit.
- 3. An *owner or operator* that has satisfied the requirements of Part II.C.2 above will be authorized to *discharge* stormwater from their *construction activity* in accordance with the following schedule:
  - a. For *construction activities* that are <u>not</u> subject to the requirements of a *regulated, traditional land use control MS4*:
    - (i) Five (5) business days from the date the Department receives a complete electronic version of the NOI (eNOI) for *construction activities* with a SWPPP that has been prepared in conformance with the design criteria in the technical standard referenced in Part III.B.1 and the *performance criteria* in the technical standard referenced in Parts III.B., 2 or 3, for *construction activities* that require post-construction stormwater management practices pursuant to Part III.C.; or
    - (ii) Sixty (60) business days from the date the Department receives a complete NOI (electronic or paper version) for *construction activities* with a SWPPP that has <u>not</u> been prepared in conformance with the design criteria in technical standard referenced in Part III.B.1. or, for *construction activities* that require post-construction stormwater management practices pursuant to Part III.C., the *performance criteria* in the technical standard referenced in Parts III.B., 2 or 3, or;
    - (iii) Ten (10) business days from the date the Department receives a complete paper version of the NOI for *construction activities* with a SWPPP that has been prepared in conformance with the design criteria in the technical standard referenced in Part III.B.1 and the *performance criteria* in the technical standard referenced in Parts III.B., 2 or 3, for *construction activities* that require post-construction stormwater management practices pursuant to Part III.C.

- b. For *construction activities* that are subject to the requirements of a *regulated, traditional land use control MS4*:
  - Five (5) business days from the date the Department receives both a complete electronic version of the NOI (eNOI) and signed "*MS4* SWPPP Acceptance" form, or
  - (ii) Ten (10) business days from the date the Department receives both a complete paper version of the NOI and signed "MS4 SWPPP Acceptance" form.
- 4. Coverage under this permit authorizes stormwater *discharges* from only those areas of disturbance that are identified in the NOI. If an *owner or operator* wishes to have stormwater *discharges* from future or additional areas of disturbance authorized, they must submit a new NOI that addresses that phase of the development, unless otherwise notified by the Department. The *owner or operator* shall not *commence construction activity* on the future or additional areas until their authorization to *discharge* under this permit goes into effect in accordance with Part II.C. of this permit.

## D. General Requirements For Owners or Operators With Permit Coverage

- The owner or operator shall ensure that the provisions of the SWPPP are implemented from the commencement of construction activity until all areas of disturbance have achieved final stabilization and the Notice of Termination ("NOT") has been submitted to the Department in accordance with Part V. of this permit. This includes any changes made to the SWPPP pursuant to Part III.A.4. of this permit.
- 2. The owner or operator shall maintain a copy of the General Permit (GP-0-20-001), NOI, NOI Acknowledgment Letter, SWPPP, MS4 SWPPP Acceptance form, inspection reports, responsible contractor's or subcontractor's certification statement (see Part III.A.6.), and all documentation necessary to demonstrate eligibility with this permit at the construction site until all disturbed areas have achieved final stabilization and the NOT has been submitted to the Department. The documents must be maintained in a secure location, such as a job trailer, on-site construction office, or mailbox with lock. The secure location must be accessible during normal business hours to an individual performing a compliance inspection.
- 3. The owner or operator of a construction activity shall not disturb greater than five (5) acres of soil at any one time without prior written authorization from the Department or, in areas under the jurisdiction of a *regulated, traditional land*

*use control MS4,* the *regulated, traditional land use control MS4* (provided the *regulated, traditional land use control MS4* is not the *owner or operator* of the *construction activity*). At a minimum, the *owner or operator* must comply with the following requirements in order to be authorized to disturb greater than five (5) acres of soil at any one time:

- a. The owner or operator shall have a qualified inspector conduct at least two (2) site inspections in accordance with Part IV.C. of this permit every seven (7) calendar days, for as long as greater than five (5) acres of soil remain disturbed. The two (2) inspections shall be separated by a minimum of two (2) full calendar days.
- b. In areas where soil disturbance activity has temporarily or permanently ceased, the application of soil stabilization measures must be initiated by the end of the next business day and completed within seven (7) days from the date the current soil disturbance activity ceased. The soil stabilization measures selected shall be in conformance with the technical standard, New York State Standards and Specifications for Erosion and Sediment Control, dated November 2016.
- c. The *owner or operator* shall prepare a phasing plan that defines maximum disturbed area per phase and shows required cuts and fills.
- d. The *owner or operator* shall install any additional site-specific practices needed to protect water quality.
- e. The *owner or operator* shall include the requirements above in their SWPPP.
- 4. In accordance with statute, regulations, and the terms and conditions of this permit, the Department may suspend or revoke an *owner's or operator's* coverage under this permit at any time if the Department determines that the SWPPP does not meet the permit requirements or consistent with Part VII.K..
- 5. Upon a finding of significant non-compliance with the practices described in the SWPPP or violation of this permit, the Department may order an immediate stop to all activity at the site until the non-compliance is remedied. The stop work order shall be in writing, describe the non-compliance in detail, and be sent to the *owner or operator*.
- 6. For *construction activities* that are subject to the requirements of a *regulated, traditional land use control MS4*, the *owner or operator* shall notify the

regulated, traditional land use control MS4 in writing of any planned amendments or modifications to the post-construction stormwater management practice component of the SWPPP required by Part III.A. 4. and 5. of this permit. Unless otherwise notified by the *regulated, traditional land use control MS4*, the owner or operator shall have the SWPPP amendments or modifications reviewed and accepted by the *regulated, traditional land use control MS4* prior to commencing construction of the post-construction stormwater management practice.

# E. Permit Coverage for Discharges Authorized Under GP-0-15-002

 Upon renewal of SPDES General Permit for Stormwater Discharges from *Construction Activity* (Permit No. GP-0-15-002), an *owner or operator* of *a construction activity* with coverage under GP-0-15-002, as of the effective date of GP- 0-20-001, shall be authorized to *discharge* in accordance with GP- 0-20-001, unless otherwise notified by the Department.

An *owner or operator* may continue to implement the technical/design components of the post-construction stormwater management controls provided that such design was done in conformance with the technical standards in place at the time of initial project authorization. However, they must comply with the other, non-design provisions of GP-0-20-001.

## F. Change of Owner or Operator

- When property ownership changes or when there is a change in operational control over the construction plans and specifications, the original owner or operator must notify the new owner or operator, in writing, of the requirement to obtain permit coverage by submitting a NOI with the Department. For construction activities subject to the requirements of a regulated, traditional land use control MS4, the original owner or operator must also notify the MS4, in writing, of the change in ownership at least 30 calendar days prior to the change in ownership.
- 2. Once the new *owner or operator* obtains permit coverage, the original *owner or operator* shall then submit a completed NOT with the name and permit identification number of the new *owner or operator* to the Department at the address in Part II.B.1. of this permit. If the original *owner or operator* maintains ownership of a portion of the *construction activity* and will disturb soil, they must maintain their coverage under the permit.
- 3. Permit coverage for the new *owner or operator* will be effective as of the date the Department receives a complete NOI, provided the original *owner or*

*operator* was not subject to a sixty (60) business day authorization period that has not expired as of the date the Department receives the NOI from the new *owner or operator*.

### Part III. STORMWATER POLLUTION PREVENTION PLAN (SWPPP)

#### A. General SWPPP Requirements

- 1. A SWPPP shall be prepared and implemented by the owner or operator of each construction activity covered by this permit. The SWPPP must document the selection, design, installation, implementation and maintenance of the control measures and practices that will be used to meet the effluent limitations in Part I.B. of this permit and where applicable, the post-construction stormwater management practice requirements in Part I.C. of this permit. The SWPPP shall be prepared prior to the submittal of the NOI. The NOI shall be submitted to the Department prior to the commencement of construction activity. A copy of the completed, final NOI shall be included in the SWPPP.
- 2. The SWPPP shall describe the erosion and sediment control practices and where required, post-construction stormwater management practices that will be used and/or constructed to reduce the *pollutants* in stormwater *discharges* and to assure compliance with the terms and conditions of this permit. In addition, the SWPPP shall identify potential sources of pollution which may reasonably be expected to affect the quality of stormwater *discharges*.
- 3. All SWPPPs that require the post-construction stormwater management practice component shall be prepared by a *qualified professional* that is knowledgeable in the principles and practices of stormwater management and treatment.
- 4. The *owner or operator* must keep the SWPPP current so that it at all times accurately documents the erosion and sediment controls practices that are being used or will be used during construction, and all post-construction stormwater management practices that will be constructed on the site. At a minimum, the *owner or operator* shall amend the SWPPP, including construction drawings:
  - a. whenever the current provisions prove to be ineffective in minimizing *pollutants* in stormwater *discharges* from the site;

- b. whenever there is a change in design, construction, or operation at the *construction site* that has or could have an effect on the *discharge* of *pollutants*;
- c. to address issues or deficiencies identified during an inspection by the *qualified inspector,* the Department or other regulatory authority; and
- d. to document the final construction conditions.
- 5. The Department may notify the *owner or operator* at any time that the SWPPP does not meet one or more of the minimum requirements of this permit. The notification shall be in writing and identify the provisions of the SWPPP that require modification. Within fourteen (14) calendar days of such notification, or as otherwise indicated by the Department, the *owner or operator* shall make the required changes to the SWPPP and submit written notification to the Department that the changes have been made. If the *owner or operator* does not respond to the Department's comments in the specified time frame, the Department may suspend the *owner's or operator's* coverage under this permit or require the *owner or operator* to obtain coverage under an individual SPDES permit in accordance with Part II.D.4. of this permit.
- 6. Prior to the commencement of construction activity, the owner or operator must identify the contractor(s) and subcontractor(s) that will be responsible for installing, constructing, repairing, replacing, inspecting and maintaining the erosion and sediment control practices included in the SWPPP; and the contractor(s) and subcontractor(s) that will be responsible for constructing the post-construction stormwater management practices included in the SWPPP. The owner or operator shall have each of the contractors and subcontractors identify at least one person from their company that will be responsible for implementation of the SWPPP. This person shall be known as the *trained contractor*. The owner or operator shall ensure that at least one *trained contractor* is on site on a daily basis when soil disturbance activities are being performed.

The *owner or operator* shall have each of the contractors and subcontractors identified above sign a copy of the following certification statement below before they commence any *construction activity*:

"I hereby certify under penalty of law that I understand and agree to comply with the terms and conditions of the SWPPP and agree to implement any corrective actions identified by the *qualified inspector* during a site inspection. I also understand that the *owner or operator* must comply with the terms and conditions of the most current version of the New York State Pollutant Discharge Elimination System ("SPDES") general permit for stormwater *discharges* from *construction activities* and that it is unlawful for any person to cause or contribute to a violation of *water quality standards*. Furthermore, I am aware that there are significant penalties for submitting false information, that I do not believe to be true, including the possibility of fine and imprisonment for knowing violations"

In addition to providing the certification statement above, the certification page must also identify the specific elements of the SWPPP that each contractor and subcontractor will be responsible for and include the name and title of the person providing the signature; the name and title of the *trained contractor* responsible for SWPPP implementation; the name, address and telephone number of the contracting firm; the address (or other identifying description) of the site; and the date the certification statement is signed. The *owner or operator* shall attach the certification statement(s) to the copy of the SWPPP that is maintained at the *construction site*. If new or additional contractors are hired to implement measures identified in the SWPPP after construction has commenced, they must also sign the certification statement and provide the information listed above.

7. For projects where the Department requests a copy of the SWPPP or inspection reports, the *owner or operator* shall submit the documents in both electronic (PDF only) and paper format within five (5) business days, unless otherwise notified by the Department.

# **B. Required SWPPP Contents**

- Erosion and sediment control component All SWPPPs prepared pursuant to this permit shall include erosion and sediment control practices designed in conformance with the technical standard, New York State Standards and Specifications for Erosion and Sediment Control, dated November 2016. Where erosion and sediment control practices are not designed in conformance with the design criteria included in the technical standard, the *owner or operator* must demonstrate *equivalence* to the technical standard. At a minimum, the erosion and sediment control component of the SWPPP shall include the following:
  - a. Background information about the scope of the project, including the location, type and size of project

- b. A site map/construction drawing(s) for the project, including a general location map. At a minimum, the site map shall show the total site area; all improvements; areas of disturbance; areas that will not be disturbed; existing vegetation; on-site and adjacent off-site surface water(s); floodplain/floodway boundaries; wetlands and drainage patterns that could be affected by the *construction activity*; existing and final contours; locations of different soil types with boundaries; material, waste, borrow or equipment storage areas located on adjacent properties; and location(s) of the stormwater *discharge*(s);
- c. A description of the soil(s) present at the site, including an identification of the Hydrologic Soil Group (HSG);
- d. A construction phasing plan and sequence of operations describing the intended order of *construction activities*, including clearing and grubbing, excavation and grading, utility and infrastructure installation and any other activity at the site that results in soil disturbance;
- e. A description of the minimum erosion and sediment control practices to be installed or implemented for each *construction activity* that will result in soil disturbance. Include a schedule that identifies the timing of initial placement or implementation of each erosion and sediment control practice and the minimum time frames that each practice should remain in place or be implemented;
- f. A temporary and permanent soil stabilization plan that meets the requirements of this general permit and the technical standard, New York State Standards and Specifications for Erosion and Sediment Control, dated November 2016, for each stage of the project, including initial land clearing and grubbing to project completion and achievement of *final stabilization*;
- g. A site map/construction drawing(s) showing the specific location(s), size(s), and length(s) of each erosion and sediment control practice;
- The dimensions, material specifications, installation details, and operation and maintenance requirements for all erosion and sediment control practices. Include the location and sizing of any temporary sediment basins and structural practices that will be used to divert flows from exposed soils;
- i. A maintenance inspection schedule for the contractor(s) identified in Part III.A.6. of this permit, to ensure continuous and effective operation of the erosion and sediment control practices. The maintenance inspection

schedule shall be in accordance with the requirements in the technical standard, New York State Standards and Specifications for Erosion and Sediment Control, dated November 2016;

- j. A description of the pollution prevention measures that will be used to control litter, construction chemicals and construction debris from becoming a *pollutant* source in the stormwater *discharges*;
- k. A description and location of any stormwater *discharges* associated with industrial activity other than construction at the site, including, but not limited to, stormwater *discharges* from asphalt plants and concrete plants located on the *construction site*; and
- I. Identification of any elements of the design that are not in conformance with the design criteria in the technical standard, New York State Standards and Specifications for Erosion and Sediment Control, dated November 2016. Include the reason for the deviation or alternative design and provide information which demonstrates that the deviation or alternative design is *equivalent* to the technical standard.
- Post-construction stormwater management practice component The owner or operator of any construction project identified in Table 2 of Appendix B as needing post-construction stormwater management practices shall prepare a SWPPP that includes practices designed in conformance with the applicable sizing criteria in Part I.C.2.a., c. or d. of this permit and the performance criteria in the technical standard, New York State Stormwater Management Design Manual dated January 2015

Where post-construction stormwater management practices are not designed in conformance with the *performance criteria* in the technical standard, the *owner or operator* must include in the SWPPP the reason(s) for the deviation or alternative design and provide information which demonstrates that the deviation or alternative design is *equivalent* to the technical standard.

The post-construction stormwater management practice component of the SWPPP shall include the following:

 a. Identification of all post-construction stormwater management practices to be constructed as part of the project. Include the dimensions, material specifications and installation details for each post-construction stormwater management practice;

- b. A site map/construction drawing(s) showing the specific location and size of each post-construction stormwater management practice;
- c. A Stormwater Modeling and Analysis Report that includes:
  - Map(s) showing pre-development conditions, including watershed/subcatchments boundaries, flow paths/routing, and design points;
  - Map(s) showing post-development conditions, including watershed/subcatchments boundaries, flow paths/routing, design points and post-construction stormwater management practices;
  - (iii) Results of stormwater modeling (i.e. hydrology and hydraulic analysis) for the required storm events. Include supporting calculations (model runs), methodology, and a summary table that compares pre and postdevelopment runoff rates and volumes for the different storm events;
  - (iv) Summary table, with supporting calculations, which demonstrates that each post-construction stormwater management practice has been designed in conformance with the *sizing criteria* included in the Design Manual;
  - (v) Identification of any *sizing criteria* that is not required based on the requirements included in Part I.C. of this permit; and
  - (vi) Identification of any elements of the design that are not in conformance with the *performance criteria* in the Design Manual. Include the reason(s) for the deviation or alternative design and provide information which demonstrates that the deviation or alternative design is *equivalent* to the Design Manual;
- d. Soil testing results and locations (test pits, borings);
- e. Infiltration test results, when required; and
- f. An operations and maintenance plan that includes inspection and maintenance schedules and actions to ensure continuous and effective operation of each post-construction stormwater management practice. The plan shall identify the entity that will be responsible for the long term operation and maintenance of each practice.

3. Enhanced Phosphorus Removal Standards - All construction projects identified in Table 2 of Appendix B that are located in the watersheds identified in Appendix C shall prepare a SWPPP that includes post-construction stormwater management practices designed in conformance with the applicable *sizing criteria* in Part I.C.2. b., c. or d. of this permit and the *performance criteria*, Enhanced Phosphorus Removal Standards included in the Design Manual. At a minimum, the post-construction stormwater management practice component of the SWPPP shall include items 2.a - 2.f. above.

# C. Required SWPPP Components by Project Type

Unless otherwise notified by the Department, *owners or operators* of *construction activities* identified in Table 1 of Appendix B are required to prepare a SWPPP that only includes erosion and sediment control practices designed in conformance with Part III.B.1 of this permit. *Owners or operators* of the *construction activities* identified in Table 2 of Appendix B shall prepare a SWPPP that also includes post-construction stormwater management practices designed in conformance with Part III.B.2 or 3 of this permit.

# Part IV. INSPECTION AND MAINTENANCE REQUIREMENTS

### A. General Construction Site Inspection and Maintenance Requirements

- 1. The *owner or operator* must ensure that all erosion and sediment control practices (including pollution prevention measures) and all post-construction stormwater management practices identified in the SWPPP are inspected and maintained in accordance with Part IV.B. and C. of this permit.
- 2. The terms of this permit shall not be construed to prohibit the State of New York from exercising any authority pursuant to the ECL, common law or federal law, or prohibit New York State from taking any measures, whether civil or criminal, to prevent violations of the laws of the State of New York or protect the public health and safety and/or the environment.

## **B.** Contractor Maintenance Inspection Requirements

1. The owner or operator of each construction activity identified in Tables 1 and 2 of Appendix B shall have a *trained contractor* inspect the erosion and sediment control practices and pollution prevention measures being implemented within the active work area daily to ensure that they are being maintained in effective operating condition at all times. If deficiencies are identified, the contractor shall
begin implementing corrective actions within one business day and shall complete the corrective actions in a reasonable time frame.

- 2. For construction sites where soil disturbance activities have been temporarily suspended (e.g. winter shutdown) and *temporary stabilization* measures have been applied to all disturbed areas, the *trained contractor* can stop conducting the maintenance inspections. The *trained contractor* shall begin conducting the maintenance inspections in accordance with Part IV.B.1. of this permit as soon as soil disturbance activities resume.
- 3. For construction sites where soil disturbance activities have been shut down with partial project completion, the *trained contractor* can stop conducting the maintenance inspections if all areas disturbed as of the project shutdown date have achieved *final stabilization* and all post-construction stormwater management practices required for the completed portion of the project have been constructed in conformance with the SWPPP and are operational.

#### C. Qualified Inspector Inspection Requirements

The *owner or operator* shall have a *qualified inspector* conduct site inspections in conformance with the following requirements:

[Note: The *trained contractor* identified in Part III.A.6. and IV.B. of this permit **cannot** conduct the *qualified inspector* site inspections unless they meet the *qualified inspector* qualifications included in Appendix A. In order to perform these inspections, the *trained contractor* would have to be a:

- licensed Professional Engineer,
- Certified Professional in Erosion and Sediment Control (CPESC),
- New York State Erosion and Sediment Control Certificate Program holder
- Registered Landscape Architect, or
- someone working under the direct supervision of, and at the same company as, the licensed Professional Engineer or Registered Landscape Architect, provided they have received four (4) hours of Department endorsed training in proper erosion and sediment control principles from a Soil and Water Conservation District, or other Department endorsed entity].
- 1. A *qualified inspector* shall conduct site inspections for all *construction activities* identified in Tables 1 and 2 of Appendix B, <u>with the exception of</u>:
  - a. the construction of a single family residential subdivision with 25% or less *impervious cover* at total site build-out that involves a soil disturbance of one (1) or more acres of land but less than five (5) acres and is <u>not</u> located

in one of the watersheds listed in Appendix C and <u>not</u> directly discharging to one of the 303(d) segments listed in Appendix E;

- b. the construction of a single family home that involves a soil disturbance of one (1) or more acres of land but less than five (5) acres and is <u>not</u> located in one of the watersheds listed in Appendix C and <u>not</u> directly discharging to one of the 303(d) segments listed in Appendix E;
- c. construction on agricultural property that involves a soil disturbance of one
   (1) or more acres of land but less than five (5) acres; and
- d. *construction activities* located in the watersheds identified in Appendix D that involve soil disturbances between five thousand (5,000) square feet and one (1) acre of land.
- 2. Unless otherwise notified by the Department, the *qualified inspector* shall conduct site inspections in accordance with the following timetable:
  - a. For construction sites where soil disturbance activities are on-going, the *qualified inspector* shall conduct a site inspection at least once every seven (7) calendar days.
  - b. For construction sites where soil disturbance activities are on-going and the owner or operator has received authorization in accordance with Part II.D.3 to disturb greater than five (5) acres of soil at any one time, the *qualified inspector* shall conduct at least two (2) site inspections every seven (7) calendar days. The two (2) inspections shall be separated by a minimum of two (2) full calendar days.
  - c. For construction sites where soil disturbance activities have been temporarily suspended (e.g. winter shutdown) and *temporary stabilization* measures have been applied to all disturbed areas, the *qualified inspector* shall conduct a site inspection at least once every thirty (30) calendar days. The *owner or operator* shall notify the DOW Water (SPDES) Program contact at the Regional Office (see contact information in Appendix F) or, in areas under the jurisdiction of a *regulated, traditional land use control MS4*, the *regulated, traditional land use control MS4* (provided the *regulated, traditional land use control MS4* is not the *owner or operator* of the *construction activity*) in writing prior to reducing the frequency of inspections.

- d. For construction sites where soil disturbance activities have been shut down with partial project completion, the qualified inspector can stop conducting inspections if all areas disturbed as of the project shutdown date have achieved *final stabilization* and all post-construction stormwater management practices required for the completed portion of the project have been constructed in conformance with the SWPPP and are operational. The owner or operator shall notify the DOW Water (SPDES) Program contact at the Regional Office (see contact information in Appendix F) or, in areas under the jurisdiction of a regulated, traditional land use control MS4, the regulated, traditional land use control MS4 (provided the regulated, traditional land use control MS4 is not the owner or operator of the *construction activity*) in writing prior to the shutdown. If soil disturbance activities are not resumed within 2 years from the date of shutdown, the owner or operator shall have the qualified inspector perform a final inspection and certify that all disturbed areas have achieved final stabilization, and all temporary, structural erosion and sediment control measures have been removed; and that all post-construction stormwater management practices have been constructed in conformance with the SWPPP by signing the "Final Stabilization" and "Post-Construction Stormwater Management Practice" certification statements on the NOT. The owner or operator shall then submit the completed NOT form to the address in Part II.B.1 of this permit.
- e. For construction sites that directly *discharge* to one of the 303(d) segments listed in Appendix E or is located in one of the watersheds listed in Appendix C, the *qualified inspector* shall conduct at least two (2) site inspections every seven (7) calendar days. The two (2) inspections shall be separated by a minimum of two (2) full calendar days.
- 3. At a minimum, the *qualified inspector* shall inspect all erosion and sediment control practices and pollution prevention measures to ensure integrity and effectiveness, all post-construction stormwater management practices under construction to ensure that they are constructed in conformance with the SWPPP, all areas of disturbance that have not achieved *final stabilization,* all points of *discharge* to natural surface waterbodies located within, or immediately adjacent to, the property boundaries of the *construction site*, and all points of *discharge* from the *construction site*.
- 4. The *qualified inspector* shall prepare an inspection report subsequent to each and every inspection. At a minimum, the inspection report shall include and/or address the following:

- a. Date and time of inspection;
- b. Name and title of person(s) performing inspection;
- c. A description of the weather and soil conditions (e.g. dry, wet, saturated) at the time of the inspection;
- d. A description of the condition of the runoff at all points of *discharge* from the *construction site*. This shall include identification of any *discharges* of sediment from the *construction site*. Include *discharges* from conveyance systems (i.e. pipes, culverts, ditches, etc.) and overland flow;
- e. A description of the condition of all natural surface waterbodies located within, or immediately adjacent to, the property boundaries of the *construction site* which receive runoff from disturbed areas. This shall include identification of any *discharges* of sediment to the surface waterbody;
- f. Identification of all erosion and sediment control practices and pollution prevention measures that need repair or maintenance;
- g. Identification of all erosion and sediment control practices and pollution prevention measures that were not installed properly or are not functioning as designed and need to be reinstalled or replaced;
- Description and sketch of areas with active soil disturbance activity, areas that have been disturbed but are inactive at the time of the inspection, and areas that have been stabilized (temporary and/or final) since the last inspection;
- i. Current phase of construction of all post-construction stormwater management practices and identification of all construction that is not in conformance with the SWPPP and technical standards;
- j. Corrective action(s) that must be taken to install, repair, replace or maintain erosion and sediment control practices and pollution prevention measures; and to correct deficiencies identified with the construction of the postconstruction stormwater management practice(s);
- k. Identification and status of all corrective actions that were required by previous inspection; and

- I. Digital photographs, with date stamp, that clearly show the condition of all practices that have been identified as needing corrective actions. The *qualified inspector* shall attach paper color copies of the digital photographs to the inspection report being maintained onsite within seven (7) calendar days of the date of the inspection. The *qualified inspector* shall also take digital photographs, with date stamp, that clearly show the condition of the practice(s) after the corrective action has been completed. The *qualified inspector* shall attach paper color copies of the digital photographs to the inspection report that documents the completion of the corrective action work within seven (7) calendar days of that inspection.
- 5. Within one business day of the completion of an inspection, the *qualified inspector* shall notify the *owner or operator* and appropriate contractor or subcontractor identified in Part III.A.6. of this permit of any corrective actions that need to be taken. The contractor or subcontractor shall begin implementing the corrective actions within one business day of this notification and shall complete the corrective actions in a reasonable time frame.
- 6. All inspection reports shall be signed by the *qualified inspector*. Pursuant to Part II.D.2. of this permit, the inspection reports shall be maintained on site with the SWPPP.

#### Part V. TERMINATION OF PERMIT COVERAGE

#### A. Termination of Permit Coverage

- An owner or operator that is eligible to terminate coverage under this permit must submit a completed NOT form to the address in Part II.B.1 of this permit. The NOT form shall be one which is associated with this permit, signed in accordance with Part VII.H of this permit.
- 2. An *owner or operator* may terminate coverage when one or more the following conditions have been met:
  - a. Total project completion All *construction activity* identified in the SWPPP has been completed; <u>and</u> all areas of disturbance have achieved *final stabilization*; <u>and</u> all temporary, structural erosion and sediment control measures have been removed; <u>and</u> all post-construction stormwater management practices have been constructed in conformance with the SWPPP and are operational;

- b. Planned shutdown with partial project completion All soil disturbance activities have ceased; <u>and</u> all areas disturbed as of the project shutdown date have achieved *final stabilization*; <u>and</u> all temporary, structural erosion and sediment control measures have been removed; <u>and</u> all postconstruction stormwater management practices required for the completed portion of the project have been constructed in conformance with the SWPPP and are operational;
- c. A new *owner or operator* has obtained coverage under this permit in accordance with Part II.F. of this permit.
- d. The *owner or operator* obtains coverage under an alternative SPDES general permit or an individual SPDES permit.
- 3. For *construction activities* meeting subdivision 2a. or 2b. of this Part, the *owner or operator* shall have the *qualified inspector* perform a final site inspection prior to submitting the NOT. The *qualified inspector* shall, by signing the "*Final Stabilization*" and "Post-Construction Stormwater Management Practice certification statements on the NOT, certify that all the requirements in Part V.A.2.a. or b. of this permit have been achieved.
- 4. For construction activities that are subject to the requirements of a regulated, traditional land use control MS4 and meet subdivision 2a. or 2b. of this Part, the owner or operator shall have the regulated, traditional land use control MS4 sign the "MS4 Acceptance" statement on the NOT in accordance with the requirements in Part VII.H. of this permit. The regulated, traditional land use control MS4 official, by signing this statement, has determined that it is acceptable for the owner or operator to submit the NOT in accordance with the requirements of this Part. The regulated, traditional land use control MS4 can make this determination by performing a final site inspection themselves or by accepting the qualified inspector's final site inspection certification(s) required in Part V.A.3. of this permit.
- 5. For *construction activities* that require post-construction stormwater management practices and meet subdivision 2a. of this Part, the *owner or operator* must, prior to submitting the NOT, ensure one of the following:
  - a. the post-construction stormwater management practice(s) and any right-ofway(s) needed to maintain such practice(s) have been deeded to the municipality in which the practice(s) is located,

- b. an executed maintenance agreement is in place with the municipality that will maintain the post-construction stormwater management practice(s),
- c. for post-construction stormwater management practices that are privately owned, the *owner or operator* has a mechanism in place that requires operation and maintenance of the practice(s) in accordance with the operation and maintenance plan, such as a deed covenant in the *owner or operator's* deed of record,
- d. for post-construction stormwater management practices that are owned by a public or private institution (e.g. school, university, hospital), government agency or authority, or public utility; the *owner or operator* has policy and procedures in place that ensures operation and maintenance of the practices in accordance with the operation and maintenance plan.

# Part VI. REPORTING AND RETENTION RECORDS

# A. Record Retention

The owner or operator shall retain a copy of the NOI, NOI

Acknowledgment Letter, SWPPP, MS4 SWPPP Acceptance form and any inspection reports that were prepared in conjunction with this permit for a period of at least five (5) years from the date that the Department receives a complete NOT submitted in accordance with Part V. of this general permit.

#### B. Addresses

With the exception of the NOI, NOT, and MS4 SWPPP Acceptance form (which must be submitted to the address referenced in Part II.B.1 of this permit), all written correspondence requested by the Department, including individual permit applications, shall be sent to the address of the appropriate DOW Water (SPDES) Program contact at the Regional Office listed in Appendix F.

# Part VII. STANDARD PERMIT CONDITIONS

# A. Duty to Comply

The *owner or operator* must comply with all conditions of this permit. All contractors and subcontractors associated with the project must comply with the terms of the SWPPP. Any non-compliance with this permit constitutes a violation of the Clean Water

(Part VII.A)

Act (CWA) and the ECL and is grounds for an enforcement action against the *owner or operator* and/or the contractor/subcontractor; permit revocation, suspension or modification; or denial of a permit renewal application. Upon a finding of significant non-compliance with this permit or the applicable SWPPP, the Department may order an immediate stop to all *construction activity* at the site until the non-compliance is remedied. The stop work order shall be in writing, shall describe the non-compliance in detail, and shall be sent to the *owner or operator*.

If any human remains or archaeological remains are encountered during excavation, the *owner or operator* must immediately cease, or cause to cease, all *construction activity* in the area of the remains and notify the appropriate Regional Water Engineer (RWE). *Construction activity* shall not resume until written permission to do so has been received from the RWE.

# B. Continuation of the Expired General Permit

This permit expires five (5) years from the effective date. If a new general permit is not issued prior to the expiration of this general permit, an *owner or operator* with coverage under this permit may continue to operate and *discharge* in accordance with the terms and conditions of this general permit, if it is extended pursuant to the State Administrative Procedure Act and 6 NYCRR Part 621, until a new general permit is issued.

#### C. Enforcement

Failure of the *owner or operator*, its contractors, subcontractors, agents and/or assigns to strictly adhere to any of the permit requirements contained herein shall constitute a violation of this permit. There are substantial criminal, civil, and administrative penalties associated with violating the provisions of this permit. Fines of up to \$37,500 per day for each violation and imprisonment for up to fifteen (15) years may be assessed depending upon the nature and degree of the offense.

#### D. Need to Halt or Reduce Activity Not a Defense

It shall not be a defense for an *owner or operator* in an enforcement action that it would have been necessary to halt or reduce the *construction activity* in order to maintain compliance with the conditions of this permit.

### E. Duty to Mitigate

The *owner or operator* and its contractors and subcontractors shall take all reasonable steps to *minimize* or prevent any *discharge* in violation of this permit which has a reasonable likelihood of adversely affecting human health or the environment.

## F. Duty to Provide Information

The owner or operator shall furnish to the Department, within a reasonable specified time period of a written request, all documentation necessary to demonstrate eligibility and any information to determine compliance with this permit or to determine whether cause exists for modifying or revoking this permit, or suspending or denying coverage under this permit, in accordance with the terms and conditions of this permit. The NOI, SWPPP and inspection reports required by this permit are public documents that the owner or operator must make available for review and copying by any person within five (5) business days of the owner or operator receiving a written request by any such person to review these documents. Copying of documents will be done at the requester's expense.

#### G. Other Information

When the *owner or operator* becomes aware that they failed to submit any relevant facts, or submitted incorrect information in the NOI or in any of the documents required by this permit, or have made substantive revisions to the SWPPP (e.g. the scope of the project changes significantly, the type of post-construction stormwater management practice(s) changes, there is a reduction in the sizing of the post-construction stormwater management practice, or there is an increase in the disturbance area or *impervious area*), which were not reflected in the original NOI submitted to the Department, they shall promptly submit such facts or information to the Department using the contact information in Part II.A. of this permit. Failure of the *owner or operator* to correct or supplement any relevant facts within five (5) business days of becoming aware of the deficiency shall constitute a violation of this permit.

#### H. Signatory Requirements

- 1. All NOIs and NOTs shall be signed as follows:
  - a. For a corporation these forms shall be signed by a responsible corporate officer. For the purpose of this section, a responsible corporate officer means:

- a president, secretary, treasurer, or vice-president of the corporation in charge of a principal business function, or any other person who performs similar policy or decision-making functions for the corporation; or
- (ii) the manager of one or more manufacturing, production or operating facilities, provided the manager is authorized to make management decisions which govern the operation of the regulated facility including having the explicit or implicit duty of making major capital investment recommendations, and initiating and directing other comprehensive measures to assure long term environmental compliance with environmental laws and regulations; the manager can ensure that the necessary systems are established or actions taken to gather complete and accurate information for permit application requirements; and where authority to sign documents has been assigned or delegated to the manager in accordance with corporate procedures;
- b. For a partnership or sole proprietorship these forms shall be signed by a general partner or the proprietor, respectively; or
- c. For a municipality, State, Federal, or other public agency these forms shall be signed by either a principal executive officer or ranking elected official. For purposes of this section, a principal executive officer of a Federal agency includes:
  - (i) the chief executive officer of the agency, or
  - (ii) a senior executive officer having responsibility for the overall operations of a principal geographic unit of the agency (e.g., Regional Administrators of EPA).
- 2. The SWPPP and other information requested by the Department shall be signed by a person described in Part VII.H.1. of this permit or by a duly authorized representative of that person. A person is a duly authorized representative only if:
  - a. The authorization is made in writing by a person described in Part VII.H.1. of this permit;
  - b. The authorization specifies either an individual or a position having responsibility for the overall operation of the regulated facility or activity, such as the position of plant manager, operator of a well or a well field,

superintendent, position of *equivalent* responsibility, or an individual or position having overall responsibility for environmental matters for the company. (A duly authorized representative may thus be either a named individual or any individual occupying a named position) and,

- c. The written authorization shall include the name, title and signature of the authorized representative and be attached to the SWPPP.
- 3. All inspection reports shall be signed by the *qualified inspector* that performs the inspection.
- 4. The MS4 SWPPP Acceptance form shall be signed by the principal executive officer or ranking elected official from the *regulated, traditional land use control MS4,* or by a duly authorized representative of that person.

It shall constitute a permit violation if an incorrect and/or improper signatory authorizes any required forms, SWPPP and/or inspection reports.

# I. Property Rights

The issuance of this permit does not convey any property rights of any sort, nor any exclusive privileges, nor does it authorize any injury to private property nor any invasion of personal rights, nor any infringement of Federal, State or local laws or regulations. *Owners or operators* must obtain any applicable conveyances, easements, licenses and/or access to real property prior to *commencing construction activity*.

# J. Severability

The provisions of this permit are severable, and if any provision of this permit, or the application of any provision of this permit to any circumstance, is held invalid, the application of such provision to other circumstances, and the remainder of this permit shall not be affected thereby.

# K. Requirement to Obtain Coverage Under an Alternative Permit

1. The Department may require any owner or operator authorized by this permit to apply for and/or obtain either an individual SPDES permit or another SPDES general permit. When the Department requires any discharger authorized by a general permit to apply for an individual SPDES permit, it shall notify the discharger in writing that a permit application is required. This notice shall

include a brief statement of the reasons for this decision, an application form, a statement setting a time frame for the owner or operator to file the application for an individual SPDES permit, and a deadline, not sooner than 180 days from owner or operator receipt of the notification letter, whereby the authorization to discharge under this general permit shall be terminated. Applications must be submitted to the appropriate Permit Administrator at the Regional Office. The Department may grant additional time upon demonstration, to the satisfaction of the Department, that additional time to apply for an alternative authorization is necessary or where the Department has not provided a permit determination in accordance with Part 621 of this Title.

2. When an individual SPDES permit is issued to a discharger authorized to *discharge* under a general SPDES permit for the same *discharge*(s), the general permit authorization for outfalls authorized under the individual SPDES permit is automatically terminated on the effective date of the individual permit unless termination is earlier in accordance with 6 NYCRR Part 750.

#### L. Proper Operation and Maintenance

The *owner or operator* shall at all times properly operate and maintain all facilities and systems of treatment and control (and related appurtenances) which are installed or used by the *owner or operator* to achieve compliance with the conditions of this permit and with the requirements of the SWPPP.

#### M. Inspection and Entry

The *owner or operator* shall allow an authorized representative of the Department, EPA, applicable county health department, or, in the case of a *construction site* which *discharges* through an *MS4*, an authorized representative of the *MS4* receiving the discharge, upon the presentation of credentials and other documents as may be required by law, to:

- 1. Enter upon the owner's or operator's premises where a regulated facility or activity is located or conducted or where records must be kept under the conditions of this permit;
- 2. Have access to and copy at reasonable times, any records that must be kept under the conditions of this permit; and

- 3. Inspect at reasonable times any facilities or equipment (including monitoring and control equipment), practices or operations regulated or required by this permit.
- 4. Sample or monitor at reasonable times, for purposes of assuring permit compliance or as otherwise authorized by the Act or ECL, any substances or parameters at any location.

#### N. Permit Actions

This permit may, at any time, be modified, suspended, revoked, or renewed by the Department in accordance with 6 NYCRR Part 621. The filing of a request by the *owner or operator* for a permit modification, revocation and reissuance, termination, a notification of planned changes or anticipated noncompliance does not limit, diminish and/or stay compliance with any terms of this permit.

#### O. Definitions

Definitions of key terms are included in Appendix A of this permit.

#### P. Re-Opener Clause

- If there is evidence indicating potential or realized impacts on water quality due to any stormwater discharge associated with construction activity covered by this permit, the owner or operator of such discharge may be required to obtain an individual permit or alternative general permit in accordance with Part VII.K. of this permit or the permit may be modified to include different limitations and/or requirements.
- 2. Any Department initiated permit modification, suspension or revocation will be conducted in accordance with 6 NYCRR Part 621, 6 NYCRR 750-1.18, and 6 NYCRR 750-1.20.

#### **Q.** Penalties for Falsification of Forms and Reports

In accordance with 6NYCRR Part 750-2.4 and 750-2.5, any person who knowingly makes any false material statement, representation, or certification in any application, record, report or other document filed or required to be maintained under this permit, including reports of compliance or noncompliance shall, upon conviction, be punished in accordance with ECL §71-1933 and or Articles 175 and 210 of the New York State Penal Law.

# **R. Other Permits**

Nothing in this permit relieves the *owner or operator* from a requirement to obtain any other permits required by law.

# **APPENDIX A – Acronyms and Definitions**

# Acronyms

APO – Agency Preservation Officer

BMP – Best Management Practice

CPESC – Certified Professional in Erosion and Sediment Control

Cpv – Channel Protection Volume

CWA – Clean Water Act (or the Federal Water Pollution Control Act, 33 U.S.C. §1251 et seq)

DOW – Division of Water

EAF – Environmental Assessment Form

ECL - Environmental Conservation Law

EPA – U. S. Environmental Protection Agency

HSG – Hydrologic Soil Group

MS4 – Municipal Separate Storm Sewer System

NOI – Notice of Intent

NOT – Notice of Termination

NPDES – National Pollutant Discharge Elimination System

OPRHP – Office of Parks, Recreation and Historic Places

Qf – Extreme Flood

Qp – Overbank Flood

RRv – Runoff Reduction Volume

RWE - Regional Water Engineer

SEQR – State Environmental Quality Review

SEQRA - State Environmental Quality Review Act

SHPA – State Historic Preservation Act

SPDES – State Pollutant Discharge Elimination System

SWPPP – Stormwater Pollution Prevention Plan

TMDL – Total Maximum Daily Load

UPA – Uniform Procedures Act

USDA – United States Department of Agriculture

WQv – Water Quality Volume

#### Definitions

<u>All definitions in this section are solely for the purposes of this permit.</u> **Agricultural Building –** a structure designed and constructed to house farm implements, hay, grain, poultry, livestock or other horticultural products; excluding any structure designed, constructed or used, in whole or in part, for human habitation, as a place of employment where agricultural products are processed, treated or packaged, or as a place used by the public.

**Agricultural Property** –means the land for construction of a barn, *agricultural building*, silo, stockyard, pen or other structural practices identified in Table II in the "Agricultural Management Practices Catalog for Nonpoint Source Pollution in New York State" prepared by the Department in cooperation with agencies of New York Nonpoint Source Coordinating Committee (dated June 2007).

Alter Hydrology from Pre to Post-Development Conditions - means the postdevelopment peak flow rate(s) has increased by more than 5% of the pre-developed condition for the design storm of interest (e.g. 10 yr and 100 yr).

**Combined Sewer** - means a sewer that is designed to collect and convey both "sewage" and "stormwater".

**Commence (Commencement of) Construction Activities -** means the initial disturbance of soils associated with clearing, grading or excavation activities; or other construction related activities that disturb or expose soils such as demolition, stockpiling of fill material, and the initial installation of erosion and sediment control practices required in the SWPPP. See definition for "*Construction Activity(ies)*" also.

**Construction Activity(ies)** - means any clearing, grading, excavation, filling, demolition or stockpiling activities that result in soil disturbance. Clearing activities can include, but are not limited to, logging equipment operation, the cutting and skidding of trees, stump removal and/or brush root removal. Construction activity does not include routine maintenance that is performed to maintain the original line and grade, hydraulic capacity, or original purpose of a facility.

**Construction Site** – means the land area where *construction activity(ies)* will occur. See definition for "*Commence (Commencement of) Construction Activities*" and "*Larger Common Plan of Development or Sale*" also.

**Dewatering** – means the act of draining rainwater and/or groundwater from building foundations, vaults or excavations/trenches.

**Direct Discharge (to a specific surface waterbody) -** means that runoff flows from a *construction site* by overland flow and the first point of discharge is the specific surface waterbody, or runoff flows from a *construction site* to a separate storm sewer system

and the first point of discharge from the separate storm sewer system is the specific surface waterbody.

**Discharge(s)** - means any addition of any pollutant to waters of the State through an outlet or *point source*.

Embankment – means an earthen or rock slope that supports a road/highway.

**Endangered or Threatened Species** – see 6 NYCRR Part 182 of the Department's rules and regulations for definition of terms and requirements.

**Environmental Conservation Law (ECL)** - means chapter 43-B of the Consolidated Laws of the State of New York, entitled the Environmental Conservation Law.

**Equivalent (Equivalence)** – means that the practice or measure meets all the performance, longevity, maintenance, and safety objectives of the technical standard and will provide an equal or greater degree of water quality protection.

**Final Stabilization** - means that all soil disturbance activities have ceased and a uniform, perennial vegetative cover with a density of eighty (80) percent over the entire pervious surface has been established; or other equivalent stabilization measures, such as permanent landscape mulches, rock rip-rap or washed/crushed stone have been applied on all disturbed areas that are not covered by permanent structures, concrete or pavement.

**General SPDES permit** - means a SPDES permit issued pursuant to 6 NYCRR Part 750-1.21 and Section 70-0117 of the ECL authorizing a category of discharges.

**Groundwater(s)** - means waters in the saturated zone. The saturated zone is a subsurface zone in which all the interstices are filled with water under pressure greater than that of the atmosphere. Although the zone may contain gas-filled interstices or interstices filled with fluids other than water, it is still considered saturated.

**Historic Property** – means any building, structure, site, object or district that is listed on the State or National Registers of Historic Places or is determined to be eligible for listing on the State or National Registers of Historic Places.

**Impervious Area (Cover) -** means all impermeable surfaces that cannot effectively infiltrate rainfall. This includes paved, concrete and gravel surfaces (i.e. parking lots, driveways, roads, runways and sidewalks); building rooftops and miscellaneous impermeable structures such as patios, pools, and sheds.

**Infeasible** – means not technologically possible, or not economically practicable and achievable in light of best industry practices.

Larger Common Plan of Development or Sale - means a contiguous area where multiple separate and distinct *construction activities* are occurring, or will occur, under one plan. The term "plan" in "larger common plan of development or sale" is broadly defined as any announcement or piece of documentation (including a sign, public notice or hearing, marketing plan, advertisement, drawing, permit application, State Environmental Quality Review Act (SEQRA) environmental assessment form or other documents, zoning request, computer design, etc.) or physical demarcation (including boundary signs, lot stakes, surveyor markings, etc.) indicating that *construction activities* may occur on a specific plot.

For discrete construction projects that are located within a larger common plan of development or sale that are at least 1/4 mile apart, each project can be treated as a separate plan of development or sale provided any interconnecting road, pipeline or utility project that is part of the same "common plan" is not concurrently being disturbed.

**Minimize** – means reduce and/or eliminate to the extent achievable using control measures (including best management practices) that are technologically available and economically practicable and achievable in light of best industry practices.

**Municipal Separate Storm Sewer (MS4)** - a conveyance or system of conveyances (including roads with drainage systems, municipal streets, catch basins, curbs, gutters, ditches, man-made channels, or storm drains):

- (i) Owned or operated by a State, city, town, borough, county, parish, district, association, or other public body (created by or pursuant to State law) having jurisdiction over disposal of sewage, industrial wastes, stormwater, or other wastes, including special districts under State law such as a sewer district, flood control district or drainage district, or similar entity, or an Indian tribe or an authorized Indian tribal organization, or a designated and approved management agency under section 208 of the CWA that discharges to surface waters of the State;
- (ii) Designed or used for collecting or conveying stormwater;
- (iii) Which is not a combined sewer; and
- (iv) Which is not part of a Publicly Owned Treatment Works (POTW) as defined at 40 CFR 122.2.

**National Pollutant Discharge Elimination System (NPDES)** - means the national system for the issuance of wastewater and stormwater permits under the Federal Water Pollution Control Act (Clean Water Act).

**Natural Buffer** – means an undisturbed area with natural cover running along a surface water (e.g. wetland, stream, river, lake, etc.).

**New Development** – means any land disturbance that does not meet the definition of Redevelopment Activity included in this appendix.

**New York State Erosion and Sediment Control Certificate Program** – a certificate program that establishes and maintains a process to identify and recognize individuals who are capable of developing, designing, inspecting and maintaining erosion and sediment control plans on projects that disturb soils in New York State. The certificate program is administered by the New York State Conservation District Employees Association.

**NOI Acknowledgment Letter** - means the letter that the Department sends to an owner or operator to acknowledge the Department's receipt and acceptance of a complete Notice of Intent. This letter documents the owner's or operator's authorization to discharge in accordance with the general permit for stormwater discharges from *construction activity*.

**Nonpoint Source** - means any source of water pollution or pollutants which is not a discrete conveyance or *point source* permitted pursuant to Title 7 or 8 of Article 17 of the Environmental Conservation Law (see ECL Section 17-1403).

**Overbank** –means flow events that exceed the capacity of the stream channel and spill out into the adjacent floodplain.

**Owner or Operator** - means the person, persons or legal entity which owns or leases the property on which the *construction activity* is occurring; an entity that has operational control over the construction plans and specifications, including the ability to make modifications to the plans and specifications; and/or an entity that has day-to-day operational control of those activities at a project that are necessary to ensure compliance with the permit conditions.

**Performance Criteria** – means the design criteria listed under the "Required Elements" sections in Chapters 5, 6 and 10 of the technical standard, New York State Stormwater Management Design Manual, dated January 2015. It does not include the Sizing Criteria (i.e. WQv, RRv, Cpv, Qp and Qf) in Part I.C.2. of the permit.

**Point Source** - means any discernible, confined and discrete conveyance, including but not limited to any pipe, ditch, channel, tunnel, conduit, well, discrete fissure, container, rolling stock, concentrated animal feeding operation, vessel or other floating craft, or landfill leachate collection system from which *pollutants* are or may be discharged.

**Pollutant** - means dredged spoil, filter backwash, solid waste, incinerator residue, sewage, garbage, sewage sludge, munitions, chemical wastes, biological materials, radioactive materials, heat, wrecked or discarded equipment, rock, sand and industrial, municipal, agricultural waste and ballast discharged into water; which may cause or might reasonably be expected to cause pollution of the waters of the state in contravention of the standards or guidance values adopted as provided in 6 NYCRR Parts 700 et seq.

**Qualified Inspector** - means a person that is knowledgeable in the principles and practices of erosion and sediment control, such as a licensed Professional Engineer, Certified Professional in Erosion and Sediment Control (CPESC), Registered Landscape Architect, New York State Erosion and Sediment Control Certificate Program holder or other Department endorsed individual(s).

It can also mean someone working under the direct supervision of, and at the same company as, the licensed Professional Engineer or Registered Landscape Architect, provided that person has training in the principles and practices of erosion and sediment control. Training in the principles and practices of erosion and sediment control means that the individual working under the direct supervision of the licensed Professional Engineer or Registered Landscape Architect has received four (4) hours of Department endorsed training in proper erosion and sediment control principles from a Soil and Water Conservation District, or other Department endorsed entity. After receiving the initial training, the individual working under the direct supervision of the licensed Professional Engineer or Registered Landscape Architect has received four (4) hours of the licensed water Conservation District, or other Department endorsed entity. After receiving the initial training, the individual working under the direct supervision of the licensed Professional Engineer or Registered Landscape Architect supervision of the licensed Professional Engineer or Registered Landscape Architect supervision of the licensed Professional Engineer or Registered Landscape Architect shall receive four (4) hours of training every three (3) years.

It can also mean a person that meets the *Qualified Professional* qualifications in addition to the *Qualified Inspector* qualifications.

Note: Inspections of any post-construction stormwater management practices that include structural components, such as a dam for an impoundment, shall be performed by a licensed Professional Engineer.

**Qualified Professional -** means a person that is knowledgeable in the principles and practices of stormwater management and treatment, such as a licensed Professional Engineer, Registered Landscape Architect or other Department endorsed individual(s). Individuals preparing SWPPPs that require the post-construction stormwater management practice component must have an understanding of the principles of hydrology, water quality management practice design, water quantity control design, and, in many cases, the principles of hydraulics. All components of the SWPPP that involve the practice of engineering, as defined by the NYS Education Law (see Article 145), shall be prepared by, or under the direct supervision of, a professional engineer licensed to practice in the State of New York.

**Redevelopment Activity(ies)** – means the disturbance and reconstruction of existing impervious area, including impervious areas that were removed from a project site within five (5) years of preliminary project plan submission to the local government (i.e. site plan, subdivision, etc.).

**Regulated, Traditional Land Use Control MS4 -** means a city, town or village with land use control authority that is authorized to discharge under New York State DEC's

SPDES General Permit For Stormwater Discharges from Municipal Separate Stormwater Sewer Systems (MS4s) or the City of New York's Individual SPDES Permit for their Municipal Separate Storm Sewer Systems (NY-0287890).

**Routine Maintenance Activity -** means *construction activity* that is performed to maintain the original line and grade, hydraulic capacity, or original purpose of a facility, including, but not limited to:

- Re-grading of gravel roads or parking lots,
- Cleaning and shaping of existing roadside ditches and culverts that maintains the approximate original line and grade, and hydraulic capacity of the ditch,
- Cleaning and shaping of existing roadside ditches that does not maintain the approximate original grade, hydraulic capacity and purpose of the ditch if the changes to the line and grade, hydraulic capacity or purpose of the ditch are installed to improve water quality and quantity controls (e.g. installing grass lined ditch),
- Placement of aggregate shoulder backing that stabilizes the transition between the road shoulder and the ditch or *embankment*,
- Full depth milling and filling of existing asphalt pavements, replacement of concrete pavement slabs, and similar work that does not expose soil or disturb the bottom six (6) inches of subbase material,
- Long-term use of equipment storage areas at or near highway maintenance facilities,
- Removal of sediment from the edge of the highway to restore a previously existing sheet-flow drainage connection from the highway surface to the highway ditch or *embankment*,
- Existing use of Canal Corp owned upland disposal sites for the canal, and
- Replacement of curbs, gutters, sidewalks and guide rail posts.

**Site limitations –** means site conditions that prevent the use of an infiltration technique and or infiltration of the total WQv. Typical site limitations include: seasonal high groundwater, shallow depth to bedrock, and soils with an infiltration rate less than 0.5 inches/hour. The existence of site limitations shall be confirmed and documented using actual field testing (i.e. test pits, soil borings, and infiltration test) or using information from the most current United States Department of Agriculture (USDA) Soil Survey for the County where the project is located.

**Sizing Criteria** – means the criteria included in Part I.C.2 of the permit that are used to size post-construction stormwater management control practices. The criteria include; Water Quality Volume (WQv), Runoff Reduction Volume (RRv), Channel Protection Volume (Cpv), *Overbank* Flood (Qp), and Extreme Flood (Qf).

**State Pollutant Discharge Elimination System (SPDES)** - means the system established pursuant to Article 17 of the ECL and 6 NYCRR Part 750 for issuance of permits authorizing discharges to the waters of the state.

**Steep Slope** – means land area designated on the current United States Department of Agriculture ("USDA") Soil Survey as Soil Slope Phase "D", (provided the map unit name is inclusive of slopes greater than 25%), or Soil Slope Phase E or F, (regardless of the map unit name), or a combination of the three designations.

**Streambank** – as used in this permit, means the terrain alongside the bed of a creek or stream. The bank consists of the sides of the channel, between which the flow is confined.

**Stormwater Pollution Prevention Plan (SWPPP)** – means a project specific report, including construction drawings, that among other things: describes the construction activity(ies), identifies the potential sources of pollution at the *construction site*; describes and shows the stormwater controls that will be used to control the pollutants (i.e. erosion and sediment controls; for many projects, includes post-construction stormwater management controls); and identifies procedures the *owner or operator* will implement to comply with the terms and conditions of the permit. See Part III of the permit for a complete description of the information that must be included in the SWPPP.

**Surface Waters of the State** - shall be construed to include lakes, bays, sounds, ponds, impounding reservoirs, springs, rivers, streams, creeks, estuaries, marshes, inlets, canals, the Atlantic ocean within the territorial seas of the state of New York and all other bodies of surface water, natural or artificial, inland or coastal, fresh or salt, public or private (except those private waters that do not combine or effect a junction with natural surface waters), which are wholly or partially within or bordering the state or within its jurisdiction. Waters of the state are further defined in 6 NYCRR Parts 800 to 941.

**Temporarily Ceased** – means that an existing disturbed area will not be disturbed again within 14 calendar days of the previous soil disturbance.

**Temporary Stabilization** - means that exposed soil has been covered with material(s) as set forth in the technical standard, New York Standards and Specifications for Erosion and Sediment Control, to prevent the exposed soil from eroding. The materials can include, but are not limited to, mulch, seed and mulch, and erosion control mats (e.g. jute twisted yarn, excelsior wood fiber mats).

**Total Maximum Daily Loads** (TMDLs) - A TMDL is the sum of the allowable loads of a single pollutant from all contributing point and *nonpoint sources*. It is a calculation of the maximum amount of a pollutant that a waterbody can receive on a daily basis and still meet *water quality standards*, and an allocation of that amount to the pollutant's sources. A TMDL stipulates wasteload allocations (WLAs) for *point source* discharges, load allocations (LAs) for *nonpoint sources*, and a margin of safety (MOS).

**Trained Contractor** - means an employee from the contracting (construction) company, identified in Part III.A.6., that has received four (4) hours of Department endorsed

Appendix A

training in proper erosion and sediment control principles from a Soil and Water Conservation District, or other Department endorsed entity. After receiving the initial training, the *trained contractor* shall receive four (4) hours of training every three (3) years.

It can also mean an employee from the contracting (construction) company, identified in Part III.A.6., that meets the *qualified inspector* qualifications (e.g. licensed Professional Engineer, Certified Professional in Erosion and Sediment Control (CPESC), Registered Landscape Architect, New York State Erosion and Sediment Control Certificate Program holder, or someone working under the direct supervision of, and at the same company as, the licensed Professional Engineer or Registered Landscape Architect, provided they have received four (4) hours of Department endorsed training in proper erosion and sediment control principles from a Soil and Water Conservation District, or other Department endorsed entity).

The *trained contractor* is responsible for the day to day implementation of the SWPPP.

**Uniform Procedures Act (UPA) Permit** - means a permit required under 6 NYCRR Part 621 of the Environmental Conservation Law (ECL), Article 70.

**Water Quality Standard** - means such measures of purity or quality for any waters in relation to their reasonable and necessary use as promulgated in 6 NYCRR Part 700 et seq.

# **APPENDIX B – Required SWPPP Components by Project Type**

#### Table 1

# Construction Activities that Require the Preparation of a SWPPP That Only Includes Erosion and Sediment Controls

The following construction activities that involve soil disturbances of one (1) or more acres of land, but less than five (5) acres:

- Single family home <u>not</u> located in one of the watersheds listed in Appendix C or <u>not</u> *directly discharging* to one of the 303(d) segments listed in Appendix E
- Single family residential subdivisions with 25% or less impervious cover at total site build-out and not located in one of the watersheds listed in Appendix C and not directly discharging to one of the 303(d) segments listed in Appendix E
- Construction of a barn or other *agricultural building*, silo, stock yard or pen.

The following construction activities that involve soil disturbances between five thousand (5000) square feet and one (1) acre of land:

All construction activities located in the watersheds identified in Appendix D that involve soil disturbances between five thousand (5,000) square feet and one (1) acre of land.

- Installation of underground, linear utilities; such as gas lines, fiber-optic cable, cable TV, electric, telephone, sewer mains, and water mains
- Environmental enhancement projects, such as wetland mitigation projects, stormwater retrofits and stream restoration projects
- Pond construction
- Linear bike paths running through areas with vegetative cover, including bike paths surfaced with an impervious cover
- Cross-country ski trails and walking/hiking trails
- Sidewalk, bike path or walking path projects, surfaced with an impervious cover, that are not part of residential, commercial or institutional development;
- Sidewalk, bike path or walking path projects, surfaced with an impervious cover, that include incidental shoulder or curb work along an existing highway to support construction of the sidewalk, bike path or walking path.
- Slope stabilization projects
- Slope flattening that changes the grade of the site, but does not significantly change the runoff characteristics

Appendix B

# Table 1 (Continued) CONSTRUCTION ACTIVITIES THAT REQUIRE THE PREPARATION OF A SWPPP

#### THAT ONLY INCLUDES EROSION AND SEDIMENT CONTROLS

- Spoil areas that will be covered with vegetation
- Vegetated open space projects (i.e. recreational parks, lawns, meadows, fields, downhill ski trails) excluding projects that *alter hydrology from pre to post development* conditions,
- Athletic fields (natural grass) that do not include the construction or reconstruction of *impervious* area and do not alter hydrology from pre to post development conditions
- · Demolition project where vegetation will be established, and no redevelopment is planned
- Overhead electric transmission line project that does not include the construction of permanent access roads or parking areas surfaced with *impervious cover*
- Structural practices as identified in Table II in the "Agricultural Management Practices Catalog for Nonpoint Source Pollution in New York State", excluding projects that involve soil disturbances of greater than five acres and construction activities that include the construction or reconstruction of impervious area
- Temporary access roads, median crossovers, detour roads, lanes, or other temporary impervious areas that will be restored to pre-construction conditions once the construction activity is complete

#### Table 2

### CONSTRUCTION ACTIVITIES THAT REQUIRE THE PREPARATION OF A SWPPP THAT INCLUDES POST-CONSTRUCTION STORMWATER MANAGEMENT PRACTICES

- Single family home located in one of the watersheds listed in Appendix C or *directly discharging* to one of the 303(d) segments listed in Appendix E
- Single family home that disturbs five (5) or more acres of land
- Single family residential subdivisions located in one of the watersheds listed in Appendix C or *directly discharging* to one of the 303(d) segments listed in Appendix E
- Single family residential subdivisions that involve soil disturbances of between one (1) and five (5) acres of land with greater than 25% impervious cover at total site build-out
- Single family residential subdivisions that involve soil disturbances of five (5) or more acres of land, and single family residential subdivisions that involve soil disturbances of less than five (5) acres that are part of a larger common plan of development or sale that will ultimately disturb five or more acres of land
- Multi-family residential developments; includes duplexes, townhomes, condominiums, senior housing complexes, apartment complexes, and mobile home parks
- Airports
- Amusement parks
- · Breweries, cideries, and wineries, including establishments constructed on agricultural land
- Campgrounds
- Cemeteries that include the construction or reconstruction of impervious area (>5% of disturbed area) or *alter the hydrology from pre to post development* conditions
- Commercial developments
- Churches and other places of worship
- Construction of a barn or other *agricultural building* (e.g. silo) and structural practices as identified in Table II in the "Agricultural Management Practices Catalog for Nonpoint Source Pollution in New York State" that include the construction or reconstruction of *impervious area*, excluding projects that involve soil disturbances of less than five acres.
- Golf courses
- Institutional development; includes hospitals, prisons, schools and colleges
- Industrial facilities; includes industrial parks
- Landfills
- Municipal facilities; includes highway garages, transfer stations, office buildings, POTW's, water treatment plants, and water storage tanks
- Office complexes
- · Playgrounds that include the construction or reconstruction of impervious area
- Sports complexes
- Racetracks; includes racetracks with earthen (dirt) surface
- Road construction or reconstruction, including roads constructed as part of the construction activities listed in Table 1

# Table 2 (Continued)

#### CONSTRUCTION ACTIVITIES THAT REQUIRE THE PREPARATION OF A SWPPP THAT INCLUDES POST-CONSTRUCTION STORMWATER MANAGEMENT PRACTICES

- Parking lot construction or reconstruction, including parking lots constructed as part of the construction activities listed in Table 1
- Athletic fields (natural grass) that include the construction or reconstruction of impervious area (>5% of disturbed area) or *alter the hydrology from pre to post development* conditions
- Athletic fields with artificial turf
- Permanent access roads, parking areas, substations, compressor stations and well drilling pads, surfaced with *impervious cover*, and constructed as part of an over-head electric transmission line project, wind-power project, cell tower project, oil or gas well drilling project, sewer or water main project or other linear utility project
- Sidewalk, bike path or walking path projects, surfaced with an impervious cover, that are part of a residential, commercial or institutional development
- Sidewalk, bike path or walking path projects, surfaced with an impervious cover, that are part of a highway construction or reconstruction project
- All other construction activities that include the construction or reconstruction of *impervious area* or *alter the hydrology from pre to post development* conditions, and are not listed in Table 1

# **APPENDIX C – Watersheds Requiring Enhanced Phosphorus Removal**

Watersheds where *owners or operators* of construction activities identified in Table 2 of Appendix B must prepare a SWPPP that includes post-construction stormwater management practices designed in conformance with the Enhanced Phosphorus Removal Standards included in the technical standard, New York State Stormwater Management Design Manual ("Design Manual").

- Entire New York City Watershed located east of the Hudson River Figure 1
- Onondaga Lake Watershed Figure 2
- Greenwood Lake Watershed -Figure 3
- Oscawana Lake Watershed Figure 4
- Kinderhook Lake Watershed Figure 5

## Figure 1 - New York City Watershed East of the Hudson







Appendix C

# Figure 3 - Greenwood Lake Watershed



# Figure 4 - Oscawana Lake Watershed



# Figure 5 - Kinderhook Lake Watershed



# **APPENDIX D – Watersheds with Lower Disturbance Threshold**

Watersheds where *owners or operators* of construction activities that involve soil disturbances between five thousand (5000) square feet and one (1) acre of land must obtain coverage under this permit.

Entire New York City Watershed that is located east of the Hudson River - See Figure 1 in Appendix C

# **APPENDIX E – 303(d) Segments Impaired by Construction Related Pollutant(s)**

List of 303(d) segments impaired by pollutants related to *construction activity* (e.g. silt, sediment or nutrients). The list was developed using "The Final New York State 2016 Section 303(d) List of Impaired Waters Requiring a TMDL/Other Strategy" dated November 2016. *Owners or operators* of single family home and single family residential subdivisions with 25% or less total impervious cover at total site build-out that involve soil disturbances of one or more acres of land, but less than 5 acres, and *directly discharge* to one of the listed segments below shall prepare a SWPPP that includes post-construction stormwater management practices designed in conformance with the New York State Stormwater Management Design Manual ("Design Manual"), dated January 2015.

COUNTY	WATERBODY	POLLUTANT
Albany	Ann Lee (Shakers) Pond, Stump Pond	Nutrients
Albany	Basic Creek Reservoir	Nutrients
Allegany	Amity Lake, Saunders Pond	Nutrients
Bronx	Long Island Sound, Bronx	Nutrients
Bronx	Van Cortlandt Lake	Nutrients
Broome	Fly Pond, Deer Lake, Sky Lake	Nutrients
Broome	Minor Tribs to Lower Susquehanna (north)	Nutrients
Broome	Whitney Point Lake/Reservoir	Nutrients
Cattaraugus	Allegheny River/Reservoir	Nutrients
Cattaraugus	Beaver (Alma) Lake	Nutrients
Cattaraugus	Case Lake	Nutrients
Cattaraugus	Linlyco/Club Pond	Nutrients
Сауида	Duck Lake	Nutrients
Cayuga	Little Sodus Bay	Nutrients
Chautauqua	Bear Lake	Nutrients
Chautauqua	Chadakoin River and tribs	Nutrients
Chautauqua	Chautauqua Lake, North	Nutrients
Chautauqua	Chautauqua Lake, South	Nutrients
Chautauqua	Findley Lake	Nutrients
Chautauqua	Hulburt/Clymer Pond	Nutrients
Clinton	Great Chazy River, Lower, Main Stem	Silt/Sediment
Clinton	Lake Champlain, Main Lake, Middle	Nutrients
Clinton	Lake Champlain, Main Lake, North	Nutrients
Columbia	Kinderhook Lake	Nutrients
Columbia	Robinson Pond	Nutrients
Cortland	Dean Pond	Nutrients

Dutchess	Fall Kill and tribs	Nutrients
Dutchess	Hillside Lake	Nutrients
Dutchess	Wappingers Lake	Nutrients
Dutchess	Wappingers Lake	Silt/Sediment
Erie	Beeman Creek and tribs	Nutrients
Erie	Ellicott Creek, Lower, and tribs	Silt/Sediment
Erie	Ellicott Creek, Lower, and tribs	Nutrients
Erie	Green Lake	Nutrients
Erie	Little Sister Creek, Lower, and tribs	Nutrients
Erie	Murder Creek, Lower, and tribs	Nutrients
Erie	Rush Creek and tribs	Nutrients
Erie	Scajaquada Creek, Lower, and tribs	Nutrients
Erie	Scajaquada Creek, Middle, and tribs	Nutrients
Erie	Scajaquada Creek, Upper, and tribs	Nutrients
Erie	South Branch Smoke Cr, Lower, and tribs	Silt/Sediment
Erie	South Branch Smoke Cr, Lower, and tribs	Nutrients
Essex	Lake Champlain, Main Lake, South	Nutrients
Essex	Lake Champlain, South Lake	Nutrients
Essex	Willsboro Bay	Nutrients
Genesee	Bigelow Creek and tribs	Nutrients
Genesee	Black Creek, Middle, and minor tribs	Nutrients
Genesee	Black Creek, Upper, and minor tribs	Nutrients
Genesee	Bowen Brook and tribs	Nutrients
Genesee	LeRoy Reservoir	Nutrients
Genesee	Oak Orchard Cr, Upper, and tribs	Nutrients
Genesee	Tonawanda Creek, Middle, Main Stem	Nutrients
Greene	Schoharie Reservoir	Silt/Sediment
Greene	Sleepy Hollow Lake	Silt/Sediment
Herkimer	Steele Creek tribs	Silt/Sediment
Herkimer	Steele Creek tribs	Nutrients
Jefferson	Moon Lake	Nutrients
Kings	Hendrix Creek	Nutrients
Kings	Prospect Park Lake	Nutrients
Lewis	Mill Creek/South Branch, and tribs	Nutrients
Livingston	Christie Creek and tribs	Nutrients
Livingston	Conesus Lake	Nutrients
Livingston	Mill Creek and minor tribs	Silt/Sediment
Monroe	Black Creek, Lower, and minor tribs	Nutrients
Monroe	Buck Pond	Nutrients
Monroe	Cranberry Pond	Nutrients

# 303(d) Segments Impaired by Construction Related Pollutant(s)
Monroe	Lake Ontario Shoreline, Western	Nutrients
Monroe	Long Pond	Nutrients
Monroe	Mill Creek and tribs	Nutrients
Monroe	Mill Creek/Blue Pond Outlet and tribs	Nutrients
Monroe	Minor Tribs to Irondequoit Bay	Nutrients
Monroe	Rochester Embayment - East	Nutrients
Monroe	Rochester Embayment - West	Nutrients
Monroe	Shipbuilders Creek and tribs	Nutrients
Monroe	Thomas Creek/White Brook and tribs	Nutrients
Nassau	Beaver Lake	Nutrients
Nassau	Camaans Pond	Nutrients
Nassau	East Meadow Brook, Upper, and tribs	Silt/Sediment
Nassau	East Rockaway Channel	Nutrients
Nassau	Grant Park Pond	Nutrients
Nassau	Hempstead Bay	Nutrients
Nassau	Hempstead Lake	Nutrients
Nassau	Hewlett Bay	Nutrients
Nassau	Hog Island Channel	Nutrients
Nassau	Long Island Sound, Nassau County Waters	Nutrients
Nassau	Massapequa Creek and tribs	Nutrients
Nassau	Milburn/Parsonage Creeks, Upp, and tribs	Nutrients
Nassau	Reynolds Channel, west	Nutrients
Nassau	Tidal Tribs to Hempstead Bay	Nutrients
Nassau	Tribs (fresh) to East Bay	Nutrients
Nassau	Tribs (fresh) to East Bay	Silt/Sediment
Nassau	Tribs to Smith/Halls Ponds	Nutrients
Nassau	Woodmere Channel	Nutrients
New York	Harlem Meer	Nutrients
New York	The Lake in Central Park	Nutrients
Niagara	Bergholtz Creek and tribs	Nutrients
Niagara	Hyde Park Lake	Nutrients
Niagara	Lake Ontario Shoreline, Western	Nutrients
Niagara	Lake Ontario Shoreline, Western	Nutrients
Oneida	Ballou, Nail Creeks and tribs	Nutrients
Onondaga	Harbor Brook, Lower, and tribs	Nutrients
Onondaga	Ley Creek and tribs	Nutrients
Onondaga	Minor Tribs to Onondaga Lake	Nutrients
Onondaga	Ninemile Creek, Lower, and tribs	Nutrients
Onondaga	Onondaga Creek, Lower, and tribs	Nutrients
Onondaga	Onondaga Creek, Middle, and tribs	Nutrients

Onondaga	Onondaga Lake, northern end	Nutrients
Onondaga	Onondaga Lake, southern end	Nutrients
Ontario	Great Brook and minor tribs	Silt/Sediment
Ontario	Great Brook and minor tribs	Nutrients
Ontario	Hemlock Lake Outlet and minor tribs	Nutrients
Ontario	Honeoye Lake	Nutrients
Orange	Greenwood Lake	Nutrients
Orange	Monhagen Brook and tribs	Nutrients
Orange	Orange Lake	Nutrients
Orleans	Lake Ontario Shoreline, Western	Nutrients
Orleans	Lake Ontario Shoreline, Western	Nutrients
Oswego	Lake Neatahwanta	Nutrients
Oswego	Pleasant Lake	Nutrients
Putnam	Bog Brook Reservoir	Nutrients
Putnam	Boyd Corners Reservoir	Nutrients
Putnam	Croton Falls Reservoir	Nutrients
Putnam	Diverting Reservoir	Nutrients
Putnam	East Branch Reservoir	Nutrients
Putnam	Lake Carmel	Nutrients
Putnam	Middle Branch Reservoir	Nutrients
Putnam	Oscawana Lake	Nutrients
Putnam	Palmer Lake	Nutrients
Putnam	West Branch Reservoir	Nutrients
Queens	Bergen Basin	Nutrients
Queens	Flushing Creek/Bay	Nutrients
Queens	Jamaica Bay, Eastern, and tribs (Queens)	Nutrients
Queens	Kissena Lake	Nutrients
Queens	Meadow Lake	Nutrients
Queens	Willow Lake	Nutrients
Rensselaer	Nassau Lake	Nutrients
Rensselaer	Snyders Lake	Nutrients
Richmond	Grasmere Lake/Bradys Pond	Nutrients
Rockland	Congers Lake, Swartout Lake	Nutrients
Rockland	Rockland Lake	Nutrients
Saratoga	Ballston Lake	Nutrients
Saratoga	Dwaas Kill and tribs	Silt/Sediment
Saratoga	Dwaas Kill and tribs	Nutrients
Saratoga	Lake Lonely	Nutrients
Saratoga	Round Lake	Nutrients
Saratoga	Tribs to Lake Lonely	Nutrients

Schenectady	Collins Lake	Nutrients
Schenectady	Duane Lake	Nutrients
Schenectady	Mariaville Lake	Nutrients
Schoharie	Engleville Pond	Nutrients
Schoharie	Summit Lake	Nutrients
Seneca	Reeder Creek and tribs	Nutrients
St.Lawrence	Black Lake Outlet/Black Lake	Nutrients
St.Lawrence	Fish Creek and minor tribs	Nutrients
Steuben	Smith Pond	Nutrients
Suffolk	Agawam Lake	Nutrients
Suffolk	Big/Little Fresh Ponds	Nutrients
Suffolk	Canaan Lake	Silt/Sediment
Suffolk	Canaan Lake	Nutrients
Suffolk	Flanders Bay, West/Lower Sawmill Creek	Nutrients
Suffolk	Fresh Pond	Nutrients
Suffolk	Great South Bay, East	Nutrients
Suffolk	Great South Bay, Middle	Nutrients
Suffolk	Great South Bay, West	Nutrients
Suffolk	Lake Ronkonkoma	Nutrients
Suffolk	Long Island Sound, Suffolk County, West	Nutrients
Suffolk	Mattituck (Marratooka) Pond	Nutrients
Suffolk	Meetinghouse/Terrys Creeks and tribs	Nutrients
Suffolk	Mill and Seven Ponds	Nutrients
Suffolk	Millers Pond	Nutrients
Suffolk	Moriches Bay, East	Nutrients
Suffolk	Moriches Bay, West	Nutrients
Suffolk	Peconic River, Lower, and tidal tribs	Nutrients
Suffolk	Quantuck Bay	Nutrients
Suffolk	Shinnecock Bay and Inlet	Nutrients
Suffolk	Tidal tribs to West Moriches Bay	Nutrients
Sullivan	Bodine, Montgomery Lakes	Nutrients
Sullivan	Davies Lake	Nutrients
Sullivan	Evens Lake	Nutrients
Sullivan	Pleasure Lake	Nutrients
Tompkins	Cayuga Lake, Southern End	Nutrients
Tompkins	Cayuga Lake, Southern End	Silt/Sediment
Tompkins	Owasco Inlet, Upper, and tribs	Nutrients
Ulster	Ashokan Reservoir	Silt/Sediment
Ulster	Esopus Creek, Upper, and minor tribs	Silt/Sediment
Warren	Hague Brook and tribs	Silt/Sediment

Warren	Huddle/Finkle Brooks and tribs	Silt/Sediment
Warren	Indian Brook and tribs	Silt/Sediment
Warren	Lake George	Silt/Sediment
Warren	Tribs to L.George, Village of L George	Silt/Sediment
Washington	Cossayuna Lake	Nutrients
Washington	Lake Champlain, South Bay	Nutrients
Washington	Tribs to L.George, East Shore	Silt/Sediment
Washington	Wood Cr/Champlain Canal and minor tribs	Nutrients
Wayne	Port Bay	Nutrients
Westchester	Amawalk Reservoir	Nutrients
Westchester	Blind Brook, Upper, and tribs	Silt/Sediment
Westchester	Cross River Reservoir	Nutrients
Westchester	Lake Katonah	Nutrients
Westchester	Lake Lincolndale	Nutrients
Westchester	Lake Meahagh	Nutrients
Westchester	Lake Mohegan	Nutrients
Westchester	Lake Shenorock	Nutrients
Westchester	Long Island Sound, Westchester (East)	Nutrients
Westchester	Mamaroneck River, Lower	Silt/Sediment
Westchester	Mamaroneck River, Upper, and minor tribs	Silt/Sediment
Westchester	Muscoot/Upper New Croton Reservoir	Nutrients
Westchester	New Croton Reservoir	Nutrients
Westchester	Peach Lake	Nutrients
Westchester	Reservoir No.1 (Lake Isle)	Nutrients
Westchester	Saw Mill River, Lower, and tribs	Nutrients
Westchester	Saw Mill River, Middle, and tribs	Nutrients
Westchester	Sheldrake River and tribs	Silt/Sediment
Westchester	Sheldrake River and tribs	Nutrients
Westchester	Silver Lake	Nutrients
Westchester	Teatown Lake	Nutrients
Westchester	Titicus Reservoir	Nutrients
Westchester	Truesdale Lake	Nutrients
Westchester	Wallace Pond	Nutrients
Wyoming	Java Lake	Nutrients
Wyoming	Silver Lake	Nutrients

<b>APPENDIX</b>	F – List	of NYS	DEC	Regional	Offices

<u>Region</u>	<u>Covering the</u> Following counties:	DIVISION OF ENVIRONMENTAL PERMITS (DEP) <u>PERMIT ADMINISTRATORS</u>	DIVISION OF WATER (DOW) <u>Water (SPDES) Program</u>
1	NASSAU AND SUFFOLK	50 CIRCLE ROAD STONY BROOK, NY 11790 TEL. (631) 444-0365	50 CIRCLE ROAD STONY BROOK, NY 11790-3409 TEL. (631) 444-0405
2	BRONX, KINGS, NEW YORK, QUEENS AND RICHMOND	1 HUNTERS POINT PLAZA, 47-40 21st St. Long Island City, Ny 11101-5407 Tel. (718) 482-4997	1 Hunters Point Plaza, 47-40 21st St. Long Island City, Ny 11101-5407 Tel. (718) 482-4933
3	DUTCHESS, ORANGE, PUTNAM, Rockland, Sullivan, Ulster and Westchester	21 South Putt Corners Road New Paltz, Ny 12561-1696 Tel. (845) 256-3059	100 HILLSIDE AVENUE, SUITE 1W WHITE PLAINS, NY 10603 TEL. (914) 428 - 2505
4	Albany, Columbia, Delaware, Greene, Montgomery, Otsego, Rensselaer, Schenectady and Schoharie	1150 North Westcott Road Schenectady, Ny 12306-2014 Tel. (518) 357-2069	1130 North Westcott Road Schenectady, Ny 12306-2014 Tel. (518) 357-2045
5	Clinton, Essex, Franklin, Fulton, Hamilton, Saratoga, Warren and Washington	1115 State Route 86, Ро Вох 296 Ray Brook, Ny 12977-0296 Tel. (518) 897-1234	232 GOLF COURSE ROAD WARRENSBURG, NY 12885-1172 TEL. (518) 623-1200
6	HERKIMER, JEFFERSON, LEWIS, ONEIDA AND ST. LAWRENCE	STATE OFFICE BUILDING 317 WASHINGTON STREET WATERTOWN, NY 13601-3787 TEL. (315) 785-2245	STATE OFFICE BUILDING 207 GENESEE STREET UTICA, NY 13501-2885 TEL. (315) 793-2554
7	BROOME, CAYUGA, CHENANGO, CORTLAND, MADISON, ONONDAGA, OSWEGO, TIOGA AND TOMPKINS	615 ERIE BLVD. WEST SYRACUSE, NY 13204-2400 TEL. (315) 426-7438	615 ERIE BLVD. WEST SYRACUSE, NY 13204-2400 TEL. (315) 426-7500
8	CHEMUNG, GENESEE, LIVINGSTON, MONROE, ONTARIO, ORLEANS, SCHUYLER, SENECA, STEUBEN, WAYNE AND YATES	6274 EAST AVON-LIMA ROADAVON, NY 14414-9519 TEL. (585) 226-2466	6274 EAST AVON-LIMA RD. AVON, NY 14414-9519 TEL. (585) 226-2466
9	ALLEGANY, CATTARAUGUS, CHAUTAUQUA, ERIE, NIAGARA AND WYOMING	270 MICHIGAN AVENUE BUFFALO, NY 14203-2999 TEL. (716) 851-7165	270 MICHIGAN AVENUE BUFFALO, NY 14203-2999 TEL. (716) 851-7070



# FACT SHEET

For

## NEW YORK STATE DEPARTMENT OF ENVIRONMENTAL CONSERVATION SPDES GENERAL PERMIT FOR STORMWATER DISCHARGES from CONSTRUCTION ACTIVITY

Permit No. GP-0-20-001

Issued Pursuant to Article 17, Titles 7, 8 and Article 70 of the Environmental Conservation Law

January 2020

Page 1 of 8

Page left intentionally blank

## INTRODUCTION

The New York State Department of Environmental Conservation (NYSDEC) has issued the State Pollutant Discharge Elimination System (SPDES) General Permit for Stormwater Discharges from Construction Activity (GP-0-20-001). Upon its effective date of January 29, 2020, GP-0-20-001 replaces the previous general permit, GP-0-15-002, which expires on January 28, 2020.

GP-0-20-001 is a five (5) year general permit for discharges of stormwater to surface waters of the State from construction activities as defined in 40 CFR 122.26(b)(14)(x) and (b)(15)(i - ii). This general permit may also authorize discharges of stormwater to groundwater in cases where the NYSDEC has determined that a permit is necessary.

Pursuant to Section 402 of the Clean Water Act (CWA), stormwater discharges from certain construction activities (including discharges through a municipal separate storm sewer system) are unlawful unless they are authorized by a National Pollutant Discharge Elimination System (NPDES) permit or by a state permit program. New York administers the approved SPDES program with permits issued in accordance with the New York State Environmental Conservation Law (ECL) Article 17, Titles 7, 8 and Article 70. An owner or operator of a construction activity must operate under an effective individual SPDES permit, which addresses the stormwater discharges, or obtain coverage under GP-0-20-001.

### SUMMARY OF CHANGES

The following is a summary of the changes made from GP-0-15-001 to GP-0-20-001, as well as minor changes made from the draft of GP-0-20-001 to the issued GP-0-20-001 in response to comments received during the public notice period.

# Effluent Limitation Guidelines Applicable to Discharges from Construction Activities:

As required by 40 CFR 450.21, GP-0-20-001 includes additional criteria under "Erosion and Sediment Controls" (Part I.B.1.a.) to comply with Effluent Limitation Guidelines (ELGs) promulgated in 2015. The changes are consistent with recent updates the United States Environmental Protection Agency (EPA) made to its 2017 General Permit for Stormwater Associated with Construction Activities, modified in June 2019. Peak flow rates and total stormwater volume are included in the "Control stormwater discharge" requirement (Part I.B.1.a.ii). The second change is the requirement on "Minimizing Dust" (Part I.B.1.a.ix).

The ELGs apply primarily to the selection, design, and implementation of the erosion and sediment controls (i.e. during construction controls) to be used on the site. These non-numeric effluent limits require an owner or operator to minimize the discharge of pollutants through the selection, design and implementation of erosion and sediment control measures. As unchanged in GP-0-20-001, the term "minimize" means to reduce and/or eliminate to the extent achievable using control measures that are technologically available and economically achievable and practicable in light of best industry practice.

## Eligibility Requirements - Non-stormwater Discharges (Part I.E.3.)

GP-0-20-001 clarifies the non-stormwater discharges that may be authorized the by general permit.

## Activities Which Are Ineligible for Coverage Under This General Permit

The following slope designation: "land designated on the current United States Department of Agriculture ("USDA") Soil Survey as Soil Slope Phase of "D", (provided the map unit name is inclusive of slopes greater than 25%) is included in Part I.F.6. and 7. of the ineligibility criteria in GP-0-20-001. This designation addresses additional "steep slope" areas (i.e. slopes greater than or equal to 25%) that are in some of the mountain areas across New York State. This criterion is necessary because of the increased potential for erosion from construction on the steep slopes. The higher level of oversite will reduce the potential for an erosion problem, and therefore, is more protective of water quality.

### Notice of Intent (NOI) Submittal

GP-0-20-001 allows for the use of either the electronic Notice of Intent (eNOI) or paper version until December 21, 2020. Beginning December 21, 2020, in accordance with 40 CFR Part 127 (EPA's 2015 NPDES Electronic Reporting Rule), the owner or operator must submit the NOI electronically using the Department's online NOI.

### **Permit Authorization**

GP-0-20-001 clarifies the requirement for the owner or operator to obtain any "Department certifications," regardless of the issuing New York State agency, prior to their authorization under the general permit being in effect. This is consistent with how the Department has implemented this condition in the past.

### Change of Owner or Operator

GP-0-20-001 includes a requirement for the owner or operator to notify the regulated, traditional land use control Municipal Separate Storm Sewer System (MS4), in writing, of change in ownership or operation of a construction activity. This requirement is added to ensure the MS4 is aware of the responsible party for a construction activity that is under its jurisdiction. The timeframe in GP-0-20-001 is consistent with transfer procedures for individual SPDES permits in 6 NYCRR 750-1.17.

### **General SWPPP Requirements**

GP-0-20-001 clarifies that the owner or operator must amend the SWPPP and construction drawings to document the final construction conditions. These amendments/updates can be used by the owner or operator to document compliance with GP-0-20-001 and the SWPPP and for long term operation and maintenance of any post-construction control practices that were constructed as part of the construction activity. This is consistent with how the Department has implemented this condition in the past.

## Definitions – Appendix A

The Department made the following changes to assist project owners and their design professionals with SWPPP development and implementation. These changes make GP-0-20-001 more protective than the current permit.

Agricultural Building – GP-0-20-001 includes a definition for "Agricultural Building" to address confusion over EPA's agricultural exemption. This definition will also help clarify SWPPP and permitting requirements for the construction of breweries, cideries and wineries on agriculture land.

Qualified Inspector – The definition includes individuals that hold a current certificate under the "New York State Erosion and Sediment Control Certificate Program ". This certification program was recently developed by New York State Agriculture and Markets with assistance from members of the New York State Conservation District Employees Association.

Regulated, Traditional Land Use Control MS4 – The definition includes the City of New York's Municipal Separate Storm Sewer System. The City of New York is considered a "large MS4" that is authorized to discharge under an individual SPDES Permit (NY-0287890). The owner or operator of a construction activity that discharges to NYC's MS4 system will be required to follow the SWPPP review and acceptance process in order to gain coverage under GP-0-20-001.

Steep Slope – The definition includes *land designated on the current United States Department of Agriculture ("USDA") Soil Survey as Soil Slope Phase "D"(provided the map unit name is inclusive of slopes greater than 25%).* This slope designation in Parts I.F.6. and 7. of the ineligibility criteria addresses additional "steep slope" areas (i.e. slopes greater than or equal to 25%) that are in some of the mountain areas across New York State.

Trained Contractor - The definition was updated to include individuals that hold a current certificate under the "New York State Erosion and Sediment Control Certificate Program", for the same reason as "Qualified Inspector" above.

GP-0-20-001 includes definitions of the following terms for clarification and consistency with EPA's 2017 Construction General Permit, modified in June 2019:

- Agricultural Property
- Construction Site
- Dewatering
- Embankment
- Endangered of Threatened Species
- Natural Buffer
- Nonpoint Source
- Overbank
- Point Source
- Streambank
- Stormwater Pollution Prevention Plan

### Required SWPPP Components by Project Type – Appendix B

Tables 1 and 2 of Appendix B include additional types of construction activity and clarify required SWPPP components for several types of construction activity. The Department made these changes to add clarity on expectations for compliance/implementation with GP-0-20-001.

The clarifications were based on questions the Department received over the last permit term from construction activity owners, design professionals and SWPPP reviewers and comments received during the public notice period.

Table 1 (Erosion and Sediment Controls Only) - Updated table as follows:

- Added "Linear bike paths running through areas with vegetative cover, including bike paths surfaced with impervious cover"
- Added cross-country ski trails and walking/hiking paths
- Added Pond construction
- Added "bike path or walking path", "surfaced with an impervious cover" and "not part of a residential, commercial or institutional development" to the "Sidewalk construction projects" activity
- Deleted "Land clearing and grading for the purpose of creating..." under the vegetated open space activity.
- Added "Sidewalk, bike path or walking path projects, surfaced with an impervious cover, that include incidental shoulder or curb work along an existing highway to support construction of the sidewalk, bike path or walking path"
- Added "Temporary access roads, median crossovers, detour roads, lanes or other temporary impervious areas that will be restored to pre-construction conditions once the construction activity is complete"

Table 2 (Post-Construction Stormwater Management Practices also) – Updated table as follows:

- Added "duplexes" under the "Multi-family" construction activity
- Added "Breweries, cideries and wineries, including establishments constructed on agricultural land"
- Added "water storage tanks" under "Municipal facilities" construction activity
- Added "Playgrounds that include the construction or reconstruction of impervious area"
- Added "including roads constructed as part of the construction activities listed in Table 1" to the "Road construction or reconstruction" activity
- Added "including parking lots constructed as part of the construction activities listed in Table 1" to the "Parking lot construction or reconstruction" activity
- Added "Sidewalk, bike path or walking path projects, surfaced with an impervious cover, that are part of a residential, commercial or institutional development,
- Added "Sidewalk, bike path or walking path projects, surfaced with an impervious cover, that are part of a highway construction or reconstruction project

### 303(d) Segments for Construction Activity – Appendix E

- Appendix E of GP-0-20-001 includes the 2016 NYS Section 303(d) list of Impaired/TMDL Waters (2016 List) for waterbody segments impaired by silt, sediment or nutrients. The list of impaired waterbodies identifies the impairment for each of the waterbodies and includes all appropriate waterbodies.

# Kimley »Horn

Appendix 2

**Certificate Statement** 

#### CONTRACTORS CERTIFICATION STATEMENT

#### Proposed 500 Main Street 500 Main Street New Rochelle, NY 10801

The following certification shall be signed and completed by each contractor and subcontractor responsible for the on-site construction activities.

"I certify under penalty of law that I understand and agree to comply with terms and conditions of the Stormwater Pollution Prevention Plan. I also understand that it is unlawful for any person to cause or contribute to a violation of water quality standards."

Owner/Operator	Contractor	Subcontractor
Signature and Date	Signature and Date	Signature and Date
Name and Title	Name and Title	Name and Title
Company and Address	Company and Address	Company and Address
Subcontractor	Subcontractor	Subcontractor
Signature and Date	Signature and Date	Signature and Date
Name and Title	Name and Title	Name and Title
Company and Address	Company and Address	Company and Addres

# Kimley »Horn

# Appendix 3

Existing and Proposed Drainage Plans





# Kimley »Horn

Appendix 4 Existing Hydrologic Analysis



### Area Listing (all nodes)

Area	CN	Description
(acres)		(subcatchment-numbers)
0.785	98	Roofs, pavemet (EX Imp)
0.785	98	TOTAL AREA

### Soil Listing (all nodes)

Area	Soil	Subcatchment
(acres)	Group	Numbers
0.000	HSG A	
0.000	HSG B	
0.000	HSG C	
0.000	HSG D	
0.785	Other	EX Imp
0.785		TOTAL AREA

### Ground Covers (all nodes)

HSG-A	HSG-B	HSG-C	HSG-D	Other	Total	Ground	Subcatchment
(acres)	(acres)	(acres)	(acres)	(acres)	(acres)	Cover	Numbers
0.000 <b>0.000</b>	0.000 <b>0.000</b>	0.000 <b>0.000</b>	0.000 <b>0.000</b>	0.785 <b>0.785</b>	0.785 <b>0.785</b>	Roofs, pavemet <b>TOTAL AREA</b>	EX Imp

 Type III 24-hr
 1 Year Rainfall=2.80"

 Printed
 8/20/2020

 \_LC
 Page 5

Time span=0.00-72.00 hrs, dt=0.05 hrs, 1441 points Runoff by SCS TR-20 method, UH=SCS, Weighted-CN Reach routing by Stor-Ind+Trans method - Pond routing by Stor-Ind method

SubcatchmentEX Imp: EX Imp Runoff Area=34,197 sf 100.00% Impervious Runoff Depth=2.57" Tc=6.0 min CN=98 Runoff=2.08 cfs 0.168 af

Link EX OUT: EX TOTAL

Inflow=2.08 cfs 0.168 af Primary=2.08 cfs 0.168 af

Total Runoff Area = 0.785 ac Runoff Volume = 0.168 af Average Runoff Depth = 2.57" 0.00% Pervious = 0.000 ac 100.00% Impervious = 0.785 ac

#### Summary for Subcatchment EX Imp: EX Imp

Runoff = 2.08 cfs @ 12.09 hrs, Volume= 0.168 af, Depth= 2.57"

Runoff by SCS TR-20 method, UH=SCS, Weighted-CN, Time Span= 0.00-72.00 hrs, dt= 0.05 hrs Type III 24-hr 1 Year Rainfall=2.80"



### Summary for Link EX OUT: EX TOTAL

Inflow Are	ea =	0.785 ac,10	0.00% Impervious,	Inflow Depth = 2.	57" for 1 Year event
Inflow	=	2.08 cfs @	12.09 hrs, Volume	e= 0.168 af	
Primary	=	2.08 cfs @	12.09 hrs, Volume	e= 0.168 af,	Atten= 0%, Lag= 0.0 min

Primary outflow = Inflow, Time Span= 0.00-72.00 hrs, dt= 0.05 hrs

### Link EX OUT: EX TOTAL



Time span=0.00-72.00 hrs, dt=0.05 hrs, 1441 points Runoff by SCS TR-20 method, UH=SCS, Weighted-CN Reach routing by Stor-Ind+Trans method - Pond routing by Stor-Ind method

SubcatchmentEX Imp: EX Imp

Runoff Area=34,197 sf 100.00% Impervious Runoff Depth=3.27" Tc=6.0 min CN=98 Runoff=2.61 cfs 0.214 af

Link EX OUT: EX TOTAL

Inflow=2.61 cfs 0.214 af Primary=2.61 cfs 0.214 af

Total Runoff Area = 0.785 ac Runoff Volume = 0.214 af Average Runoff Depth = 3.27" 0.00% Pervious = 0.000 ac 100.00% Impervious = 0.785 ac

#### Summary for Subcatchment EX Imp: EX Imp

Runoff = 2.61 cfs @ 12.09 hrs, Volume= 0.214 af, Depth= 3.27"

Runoff by SCS TR-20 method, UH=SCS, Weighted-CN, Time Span= 0.00-72.00 hrs, dt= 0.05 hrs Type III 24-hr 3.5" Rainfall=3.50"



### Summary for Link EX OUT: EX TOTAL

Inflow Ar	ea =	0.785 ac,10	0.00% Impervious,	Inflow Depth = 3.2	27" for 3.5" event
Inflow	=	2.61 cfs @	12.09 hrs, Volume	= 0.214 af	
Primary	=	2.61 cfs @	12.09 hrs, Volume	= 0.214 af,	Atten= 0%, Lag= 0.0 min

Primary outflow = Inflow, Time Span= 0.00-72.00 hrs, dt= 0.05 hrs

### Link EX OUT: EX TOTAL



 Type III 24-hr
 10 Year Rainfall=5.25"

 Printed
 8/20/2020

 \$ LLC
 Page 11

Time span=0.00-72.00 hrs, dt=0.05 hrs, 1441 points Runoff by SCS TR-20 method, UH=SCS, Weighted-CN Reach routing by Stor-Ind+Trans method - Pond routing by Stor-Ind method

> Runoff Area=34,197 sf 100.00% Impervious Runoff Depth=5.01" Tc=6.0 min CN=98 Runoff=3.94 cfs 0.328 af

Link EX OUT: EX TOTAL

SubcatchmentEX Imp: EX Imp

Inflow=3.94 cfs 0.328 af Primary=3.94 cfs 0.328 af

Total Runoff Area = 0.785 ac Runoff Volume = 0.328 af Average Runoff Depth = 5.01" 0.00% Pervious = 0.000 ac 100.00% Impervious = 0.785 ac

#### Summary for Subcatchment EX Imp: EX Imp

Runoff = 3.94 cfs @ 12.09 hrs, Volume= 0.328 af, Depth= 5.01"

Runoff by SCS TR-20 method, UH=SCS, Weighted-CN, Time Span= 0.00-72.00 hrs, dt= 0.05 hrs Type III 24-hr 10 Year Rainfall=5.25"



### Summary for Link EX OUT: EX TOTAL

Inflow A	rea =	0.785 ac,10	0.00% Impervious,	Inflow Depth = 5.	01" for 10 Year event
Inflow	=	3.94 cfs @	12.09 hrs, Volume	e= 0.328 af	
Primary	=	3.94 cfs @	12.09 hrs, Volume	e= 0.328 af,	Atten= 0%, Lag= 0.0 min

Primary outflow = Inflow, Time Span= 0.00-72.00 hrs, dt= 0.05 hrs

### Link EX OUT: EX TOTAL



 Type III 24-hr
 100 Year Rainfall=9.00"

 Printed
 8/20/2020

 ns LLC
 Page 14

Time span=0.00-72.00 hrs, dt=0.05 hrs, 1441 points Runoff by SCS TR-20 method, UH=SCS, Weighted-CN Reach routing by Stor-Ind+Trans method - Pond routing by Stor-Ind method

SubcatchmentEX Imp: EX Imp Runoff Area=34,197 sf 100.00% Impervious Runoff Depth=8.76" Tc=6.0 min CN=98 Runoff=6.78 cfs 0.573 af

Link EX OUT: EX TOTAL

Inflow=6.78 cfs 0.573 af Primary=6.78 cfs 0.573 af

Total Runoff Area = 0.785 ac Runoff Volume = 0.573 af Average Runoff Depth = 8.76" 0.00% Pervious = 0.000 ac 100.00% Impervious = 0.785 ac

#### Summary for Subcatchment EX Imp: EX Imp

Runoff = 6.78 cfs @ 12.09 hrs, Volume= 0.573 af, Depth= 8.76"

Runoff by SCS TR-20 method, UH=SCS, Weighted-CN, Time Span= 0.00-72.00 hrs, dt= 0.05 hrs Type III 24-hr 100 Year Rainfall=9.00"



### Summary for Link EX OUT: EX TOTAL

Inflow Ar	rea =	0.785 ac,10	0.00% Impervious,	Inflow Depth = 8.	76" for 100 Year event
Inflow	=	6.78 cfs @	12.09 hrs, Volume	e= 0.573 af	
Primary	=	6.78 cfs @	12.09 hrs, Volume	e= 0.573 af,	Atten= 0%, Lag= 0.0 min

Primary outflow = Inflow, Time Span= 0.00-72.00 hrs, dt= 0.05 hrs

### Link EX OUT: EX TOTAL



# Kimley »Horn

Appendix 5

Proposed Hydrologic Analysis


#### Area Listing (all nodes)

Area	CN	Description
(acres)		(subcatchment-numbers)
0.031	98	Paved parking, HSG D (PR-IMP)
0.754	98	Roofs, pavemet (PR-ROOF)
0.785	98	TOTAL AREA

#### Soil Listing (all nodes)

Area	Soil	Subcatchment
(acres)	Group	Numbers
0.000	HSG A	
0.000	HSG B	
0.000	HSG C	
0.031	HSG D	PR-IMP
0.754	Other	PR-ROOF
0.785		TOTAL AREA

				(-	<b>,</b>		
HSG-A (acres)	HSG-B (acres)	HSG-C (acres)	HSG-D (acres)	Other (acres)	Total (acres)	Ground Cover	Subcatchment Numbers
 0.000	0.000	0.000	0.031	0.000	0.031	Paved parking	PR-IMP
0.000	0.000	0.000	0.000	0.754	0.754	Roofs, pavemet	PR-ROOF
0.000	0.000	0.000	0.031	0.754	0.785	TOTAL AREA	

#### Ground Covers (all nodes)

500 Main_PR HydroCAD_connection to main st_2020-08-20	
Prepared by Kimley-Horn and Associates	Printed 9/8/2020
HydroCAD® 10.00-22 s/n 09843 © 2018 HydroCAD Software Solutions LLC	Page 5
	-

			•	0	·	,			
Line#	Node Number	In-Invert (feet)	Out-Invert (feet)	Length (feet)	Slope (ft/ft)	n	Diam/Width (inches)	Height (inches)	Inside-Fill (inches)
1	DET	88.00	87.90	10.0	0.0100	0.013	18.0	0.0	0.0
2	DET	91.00	90.85	10.0	0.0150	0.013	15.0	0.0	0.0

#### Pipe Listing (all nodes)

Time span=0.00-72.00 hrs, dt=0.05 hrs, 1441 points Runoff by SCS TR-20 method, UH=SCS, Weighted-CN Reach routing by Stor-Ind+Trans method - Pond routing by Stor-Ind method

SubcatchmentPR-IMP: PR-IMP	Runoff Area=1,346 sf 100.00% Impervious Runoff Depth=2.57" Tc=6.0 min CN=98 Runoff=0.08 cfs 0.007 af
SubcatchmentPR-ROOF: PR-ROOF	Runoff Area=32,851 sf  100.00% Impervious  Runoff Depth=2.57" Tc=6.0 min  CN=98  Runoff=1.99 cfs  0.161 af
Pond DET: DET	Peak Elev=88.97' Storage=2,829 cf Inflow=1.99 cfs 0.161 af Outflow=0.38 cfs 0.161 af
Link OUT: PR TOTAL	Inflow=0.40 cfs 0.168 af Primary=0.40 cfs 0.168 af

Total Runoff Area = 0.785 ac Runoff Volume = 0.168 af Average Runoff Depth = 2.57" 0.00% Pervious = 0.000 ac 100.00% Impervious = 0.785 ac

#### Summary for Subcatchment PR-IMP: PR-IMP

Runoff = 0.08 cfs @ 12.09 hrs, Volume= 0.007 af, Depth= 2.57"

Runoff by SCS TR-20 method, UH=SCS, Weighted-CN, Time Span= 0.00-72.00 hrs, dt= 0.05 hrs Type III 24-hr 1 Year Rainfall=2.80"

	Area (s	f) CN	Desc	ription			
	1,34	16 98	Pave	d parki	ing, HSG D	D	
	1,34	16	100.0	)0% Im	pervious A	Area	
- (mi	Tc Leng in) (fe	gth Slo et) (ft/	pe Ve ⁄ft) (f	locity t/sec)	Capacity (cfs)	Description	
6	6.0					Direct Entry,	
				Su	ıbcatchm	ment PR-IMP: PR-IMP	
	~				Hydro	ograph	
	0.09	·¦¦¦¦ - ·		¦¦¦ ¦¦		,,,,,,,,,,	off
	0.085	-     <mark> 0.0</mark>		¦¦¦ 	$\begin{vmatrix} - \frac{1}{1} - $	Type III 24-hr	
	0.075					1 Year Rainfall=2.80"	
	0.065			!!! !!!		Runoff Area=1,346 sf	
	0.06	-		   _  _  !!!		Runoff Volume=0.007 af	
(cfs)	0.05			!!! !!!	$\begin{vmatrix} 1 & 1 & 1 & 1 \\ -\frac{1}{1} - \frac{1}{1} - \frac{1}{1} - \frac{1}{1} - \frac{1}{1} \end{vmatrix}$	Runoff Depth=2.57"	
Flow	0.045	·¦¦¦¦ ·		¦¦¦		Tc=6.0 min	
	0.035			iii	   - + - + - + -		
	0.03	-		!!!			
	0.025	-   - + 		 			
	0.015	-lllll 		 			
	0.01	·'= -'= -' = -' = -' =               -'''		iii iii			
	0.005						

0 - 4 6 8 10 12 14 16 18 20 22 24 26 28 30 32 34 36 38 40 42 44 46 48 50 52 54 56 58 60 62 64 66 68 70 72 Time (hours)

#### Summary for Subcatchment PR-ROOF: PR-ROOF

Runoff = 1.99 cfs @ 12.09 hrs, Volume= 0.161 af, Depth= 2.57"

Runoff by SCS TR-20 method, UH=SCS, Weighted-CN, Time Span= 0.00-72.00 hrs, dt= 0.05 hrs Type III 24-hr 1 Year Rainfall=2.80"



#### Summary for Pond DET: DET

Inflow Area	ı =	0.754 ac,10	0.00% Impervious,	Inflow Depth =	2.57" f	or 1 Yea	ir event
Inflow	=	1.99 cfs @	12.09 hrs, Volume	e= 0.161	af		
Outflow	=	0.38 cfs @	12.53 hrs, Volume	e= 0.161	af, Atten	i= 81%, I	_ag= 26.4 min
Primary	=	0.38 cfs @	12.53 hrs, Volume	e= 0.161	af		-

Routing by Stor-Ind method, Time Span= 0.00-72.00 hrs, dt= 0.05 hrs Peak Elev= 88.97' @ 12.53 hrs Surf.Area= 2,908 sf Storage= 2,829 cf

Plug-Flow detention time= 130.0 min calculated for 0.161 af (100% of inflow) Center-of-Mass det. time= 131.3 min (890.6 - 759.3)

Volume	Inver	t Avail.Sto	rage Storage	Description	
#1	88.00	)' 11,63	32 cf Custom	Stage Data (Pr	<b>ismatic)</b> Listed below (Recalc)
Elevatio	on S	Surf.Area	Inc.Store	Cum.Store	
(fee	et)	(sq-ft)	(cubic-feet)	(cubic-feet)	
88.0	00	2,908	0	0	
89.0	00	2,908	2,908	2,908	
90.0	00	2,908	2,908	5,816	
91.0	00	2,908	2,908	8,724	
92.0	00	2,908	2,908	11,632	
Device	Routing	Invert	Outlet Devices	8	
#1	Primary	88.00'	18.0" Round Inlet / Outlet In	RCP_Round 1 nvert= 88.00' / 8	<b>18"</b> L= 10.0' Ke= 0.500 7.90' S= 0.0100'/' Cc= 0.900
#2	Device 1	88.00'	n= 0.013 Con 4.0" Vert. Ori	fice/Grate C=	ds & connections, Flow Area= 1.77 st 0.600
#3	Device 1	91.20'	<b>8.0' long x 0.</b> Head (feet) 0 Coef. (English	<b>5' breadth Broa</b> .20 0.40 0.60 ( ) 2.80 2.92 3.0	ad-Crested Weir 0.80 1.00 08 3.30 3.32
#4	Primary	91.00'	<b>15.0" Round</b> Inlet / Outlet In n= 0.013 Con	<b>RCP_Round</b> 1 nvert= 91.00' / 9 icrete pipe, bend	I <b>5"</b> L= 10.0' Ke= 0.500 0.85' S= 0.0150 '/' Cc= 0.900 ds & connections, Flow Area= 1.23 sf
Primary	OutFlow	Max=0.38 cfs (	@ 12.53 hrs HV	V=88.97' (Free	Discharge)

**Target Difference The Second Secon** 

2=Orifice/Grate (Orifice Controls 0.38 cfs @ 4.32 fps)

-3=Broad-Crested Weir (Controls 0.00 cfs)

4=RCP\_Round 15" (Controls 0.00 cfs)

Pond DET: DET



#### Summary for Link OUT: PR TOTAL

Inflow Ar	ea =	0.785 ac,10	0.00% Impervious,	Inflow Depth = $2.5$	57" for 1 Year event
Inflow	=	0.40 cfs @	12.39 hrs, Volume	= 0.168 af	
Primary	=	0.40 cfs @	12.39 hrs, Volume	= 0.168 af,	Atten= 0%, Lag= 0.0 min

Primary outflow = Inflow, Time Span= 0.00-72.00 hrs, dt= 0.05 hrs

#### Link OUT: PR TOTAL



Time span=0.00-72.00 hrs, dt=0.05 hrs, 1441 points Runoff by SCS TR-20 method, UH=SCS, Weighted-CN Reach routing by Stor-Ind+Trans method - Pond routing by Stor-Ind method

SubcatchmentPR-IMP: PR-IMP	Runoff Area=1,346 sf 100.00% Impervious Runoff Depth=3.27" Tc=6.0 min CN=98 Runoff=0.10 cfs 0.008 af
SubcatchmentPR-ROOF: PR-ROOF	Runoff Area=32,851 sf 100.00% Impervious Runoff Depth=3.27" Tc=6.0 min CN=98 Runoff=2.51 cfs 0.205 af
Pond DET: DET	Peak Elev=89.24' Storage=3,602 cf Inflow=2.51 cfs 0.205 af Outflow=0.44 cfs 0.205 af
Link OUT: PR TOTAL	Inflow=0.46 cfs 0.214 af Primary=0.46 cfs 0.214 af

Total Runoff Area = 0.785 ac Runoff Volume = 0.214 af Average Runoff Depth = 3.27" 0.00% Pervious = 0.000 ac 100.00% Impervious = 0.785 ac

#### Summary for Subcatchment PR-IMP: PR-IMP

Runoff = 0.10 cfs @ 12.09 hrs, Volume= 0.008 af, Depth= 3.27"

Runoff by SCS TR-20 method, UH=SCS, Weighted-CN, Time Span= 0.00-72.00 hrs, dt= 0.05 hrs Type III 24-hr 3.5" Rainfall=3.50"

Area (sf) CN Description	
1,346 98 Paved parking, HSG D	
1,346 100.00% Impervious Area	
(min) (feet) (ft/ft) (ft/sec) (cfs)	
6.0 Direct Entry,	
Subcatchment PR-IMP: PR-IMP	
Hydrograph	
	Runoff
	_
0.095 0.09	_
0.085 0.08 Runoff Area=1,346 sf	_
0.075	-
<b>E</b> <sup>0.065</sup> <b>Runoff Depth=3.27</b>	_
	-
0.045	_
	_
	-
	_

0 2 4 6 8 10 12 14 16 18 20 22 24 26 28 30 32 34 36 38 40 42 44 46 48 50 52 54 56 58 60 62 64 66 68 70 72 Time (hours)

#### Summary for Subcatchment PR-ROOF: PR-ROOF

Runoff = 2.51 cfs @ 12.09 hrs, Volume= 0.205 af, Depth= 3.27"

Runoff by SCS TR-20 method, UH=SCS, Weighted-CN, Time Span= 0.00-72.00 hrs, dt= 0.05 hrs Type III 24-hr 3.5" Rainfall=3.50"



#### Summary for Pond DET: DET

Inflow Area	ı =	0.754 ac,10	0.00% Impervious,	Inflow Depth =	3.27" for	3.5" ev	vent
Inflow	=	2.51 cfs @	12.09 hrs, Volume	= 0.205	af		
Outflow	=	0.44 cfs @	12.55 hrs, Volume	= 0.205	af, Atten=	83%, L	.ag= 27.6 min
Primary	=	0.44 cfs @	12.55 hrs, Volume	= 0.205	af		-

Routing by Stor-Ind method, Time Span= 0.00-72.00 hrs, dt= 0.05 hrs Peak Elev= 89.24' @ 12.55 hrs Surf.Area= 2,908 sf Storage= 3,602 cf

Plug-Flow detention time= 129.5 min calculated for 0.205 af (100% of inflow) Center-of-Mass det. time= 130.7 min (885.3 - 754.6)

Volume	Inve	rt Avail.Sto	rage Storage	Description	
#1	88.0	0' 11,63	32 cf Custom	i Stage Data (Pr	ismatic)Listed below (Recalc)
Elevati	on s	Surf.Area	Inc.Store	Cum.Store	
(fee	et)	(sq-ft)	(cubic-feet)	(cubic-feet)	
88.	00	2,908	0	0	
89.	00	2,908	2,908	2,908	
90.	00	2,908	2,908	5,816	
91.	00	2,908	2,908	8,724	
92.	00	2,908	2,908	11,632	
Device	Routing	Invert	Outlet Device	S	
#1	Primary	88.00'	<b>18.0" Round</b> Inlet / Outlet I n= 0.013 Cor	I RCP_Round 1 nvert= 88.00' / 8 ncrete pipe, bend	I <b>8"</b> L= 10.0' Ke= 0.500 7.90' S= 0.0100 '/' Cc= 0.900 ds & connections, Flow Area= 1.77 sf
#2	Device 1	88.00'	4.0" Vert. Ori	ifice/Grate C=	0.600
#3	Device 1	91.20'	<b>8.0' long x 0</b> Head (feet) 0 Coef. (English	<b>.5' breadth Bro</b> 0.20 0.40 0.60 ( 1) 2.80 2.92 3.0	ad-Crested Weir 0.80 1.00 08 3.30 3.32
#4	Primary	91.00'	<b>15.0" Round</b> Inlet / Outlet I n= 0.013 Cor	I RCP_Round 1 nvert= 91.00' / 9 ncrete pipe, bend	I <b>5"</b> L= 10.0' Ke= 0.500 0.85' S= 0.0150 '/' Cc= 0.900 ds & connections, Flow Area= 1.23 sf
Drimon		Mov-0 44 of a	12 55 bra ⊔\	N-90 21' (Eroo	

**Primary OutFlow** Max=0.44 cfs @ 12.55 hrs HW=89.24' (Free Discharge)

-1=RCP\_Round 18" (Passes 0.44 cfs of 4.59 cfs potential flow)

2=Orifice/Grate (Orifice Controls 0.44 cfs @ 4.99 fps)

-3=Broad-Crested Weir (Controls 0.00 cfs)

4=RCP Round 15" (Controls 0.00 cfs)

Pond DET: DET



### Summary for Link OUT: PR TOTAL

Inflow Ar	ea =	0.785 ac,100.00% Impervious,	Inflow Depth = 3.2	7" for 3.5" event
Inflow	=	0.46 cfs @ 12.38 hrs, Volume	= 0.214 af	
Primary	=	0.46 cfs @ 12.38 hrs, Volume	= 0.214 af, .	Atten= 0%, Lag= 0.0 min

Primary outflow = Inflow, Time Span= 0.00-72.00 hrs, dt= 0.05 hrs

#### Link OUT: PR TOTAL



Time span=0.00-72.00 hrs, dt=0.05 hrs, 1441 points Runoff by SCS TR-20 method, UH=SCS, Weighted-CN Reach routing by Stor-Ind+Trans method - Pond routing by Stor-Ind method

SubcatchmentPR-IMP: PR-IMP	Runoff Area=1,346 sf  100.00% Impervious  Runoff Depth=5.01" Tc=6.0 min  CN=98  Runoff=0.16 cfs  0.013 af
SubcatchmentPR-ROOF: PR-ROOF	Runoff Area=32,851 sf 100.00% Impervious Runoff Depth=5.01" Tc=6.0 min CN=98 Runoff=3.78 cfs 0.315 af
Pond DET: DET	Peak Elev=89.94' Storage=5,627 cf Inflow=3.78 cfs 0.315 af Outflow=0.56 cfs 0.315 af
Link OUT: PR TOTAL	Inflow=0.60 cfs 0.328 af Primary=0.60 cfs 0.328 af

Total Runoff Area = 0.785 ac Runoff Volume = 0.328 af Average Runoff Depth = 5.01" 0.00% Pervious = 0.000 ac 100.00% Impervious = 0.785 ac

#### Summary for Subcatchment PR-IMP: PR-IMP

Runoff = 0.16 cfs @ 12.09 hrs, Volume= 0.013 af, Depth= 5.01"

Runoff by SCS TR-20 method, UH=SCS, Weighted-CN, Time Span= 0.00-72.00 hrs, dt= 0.05 hrs Type III 24-hr 10 Year Rainfall=5.25"



#### Summary for Subcatchment PR-ROOF: PR-ROOF

Runoff = 3.78 cfs @ 12.09 hrs, Volume= 0.315 af, Depth= 5.01"

Runoff by SCS TR-20 method, UH=SCS, Weighted-CN, Time Span= 0.00-72.00 hrs, dt= 0.05 hrs Type III 24-hr 10 Year Rainfall=5.25"



#### Summary for Pond DET: DET

Inflow Area	=	0.754 ac,10	0.00% Impervi	ous, Inflow [	Depth = 5.0	)1" for 1	0 Year event
Inflow	=	3.78 cfs @	12.09 hrs, Vo	lume=	0.315 af		
Outflow	=	0.56 cfs @	12.58 hrs, Vo	lume=	0.315 af,	Atten= 85	%, Lag= 29.9 min
Primary	=	0.56 cfs @	12.58 hrs, Vo	lume=	0.315 af		-

Routing by Stor-Ind method, Time Span= 0.00-72.00 hrs, dt= 0.05 hrs Peak Elev= 89.94' @ 12.58 hrs Surf.Area= 2,908 sf Storage= 5,627 cf

Plug-Flow detention time= 137.1 min calculated for 0.315 af (100% of inflow) Center-of-Mass det. time= 136.9 min (884.1 - 747.2)

Volume	Inver	t Avail.Stor	rage Storage	Description	
#1	88.00	' 11,63	32 cf Custom	Stage Data (Pri	<b>smatic)</b> Listed below (Recalc)
Elevati	on S	urf.Area	Inc.Store	Cum.Store	
(fee	et)	(sq-ft)	(cubic-feet)	(cubic-feet)	
88.	00	2,908	0	0	
89.	00	2,908	2,908	2,908	
90.	00	2,908	2,908	5,816	
91.	00	2,908	2,908	8,724	
92.	00	2,908	2,908	11,632	
Device	Routing	Invert	Outlet Device	S	
#1	Primary	88.00'	18.0" Round Inlet / Outlet In n= 0.013 Cor	<b>RCP_Round 1</b> nvert= 88.00' / 87 ncrete pipe, bend	<b>8"</b> L= 10.0' Ke= 0.500 '.90' S= 0.0100 '/' Cc= 0.900 s & connections, Flow Area= 1.77 sf
#2	Device 1	88.00'	4.0" Vert. Ori	fice/Grate C= C	0.600
#3	Device 1	91.20'	<b>8.0' long x 0</b> Head (feet) 0 Coef. (English	<b>.5' breadth Broa</b> .20 0.40 0.60 0 1) 2.80 2.92 3.0	d-Crested Weir .80 1.00 8 3.30 3.32
#4	Primary	91.00'	<b>15.0" Round</b> Inlet / Outlet In n= 0.013 Cor	<b>RCP_Round 1</b> nvert= 91.00' / 90 ncrete pipe, bend	<b>5"</b> L= 10.0' Ke= 0.500 0.85' S= 0.0150 '/' Cc= 0.900 s & connections, Flow Area= 1.23 sf
Driman		Jay-0 56 cfs (	⊕ 12.58 bre H\	N-80 03' (Free	Discharge)

r**imary OutFlow** Max=0.56 cfs @ 12.58 hrs HW=89.93' (Free Discharge)

-1=RCP\_Round 18" (Passes 0.56 cfs of 8.33 cfs potential flow) 2=Orifice/Grate (Orifice Controls 0.56 cfs @ 6.40 fps)

-3=Broad-Crested Weir (Controls 0.00 cfs)

4=RCP Round 15" (Controls 0.00 cfs)

Pond DET: DET



#### Summary for Link OUT: PR TOTAL

Inflow Ar	rea =	0.785 ac,100.	00% Impervious,	Inflow Depth = 5.	01" for 10 Year event
Inflow	=	0.60 cfs @ 1	2.13 hrs, Volume	= 0.328 af	
Primary	=	0.60 cfs @ 1	2.13 hrs, Volume	= 0.328 af,	Atten= 0%, Lag= 0.0 min

Primary outflow = Inflow, Time Span= 0.00-72.00 hrs, dt= 0.05 hrs



#### Link OUT: PR TOTAL

Time span=0.00-72.00 hrs, dt=0.05 hrs, 1441 points Runoff by SCS TR-20 method, UH=SCS, Weighted-CN Reach routing by Stor-Ind+Trans method - Pond routing by Stor-Ind method

SubcatchmentPR-IMP: PR-IMP	Runoff Area=1,346 sf 100.00% Impervious Runoff Depth=8.76" Tc=6.0 min CN=98 Runoff=0.27 cfs 0.023 af
SubcatchmentPR-ROOF: PR-ROOF	Runoff Area=32,851 sf 100.00% Impervious Runoff Depth=8.76" Tc=6.0 min CN=98 Runoff=6.51 cfs 0.551 af
Pond DET: DET	Peak Elev=91.30' Storage=9,595 cf Inflow=6.51 cfs 0.551 af Outflow=1.84 cfs 0.551 af
Link OUT: PR TOTAL	Inflow=1.94 cfs 0.573 af Primary=1.94 cfs 0.573 af

Total Runoff Area = 0.785 ac Runoff Volume = 0.573 af Average Runoff Depth = 8.76" 0.00% Pervious = 0.000 ac 100.00% Impervious = 0.785 ac

#### Summary for Subcatchment PR-IMP: PR-IMP

Runoff = 0.27 cfs @ 12.09 hrs, Volume= 0.023 af, Depth= 8.76"

Runoff by SCS TR-20 method, UH=SCS, Weighted-CN, Time Span= 0.00-72.00 hrs, dt= 0.05 hrs Type III 24-hr 100 Year Rainfall=9.00"

	Area (sf)	CN De	scription						
	1,346	98 Pa	ved parki	ing, HSG D					
	1,346	10	0.00% Im	pervious A	rea				
(n	Tc Length nin) (feet)	c Length Slope Velocity Capacity Description ) (feet) (ft/ft) (ft/sec) (cfs)							
	6.0				Direct E	ntry,			
			Sı	ıbcatchm	nent PR-	IMP: PR-I	IMP		
				Hydro	graph				_
	0.28	+ - + - + - + - + - + - + -						$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Runoff
	0.26					·	Туре	III 24-hr	
	0.24				1	00 Year	Rainfa	all=9.00"	
	0.22					Dunof	f Aroa=	1-246 cf	
	0.2				L - L - L <b>-</b> - L				
	0.18			             	<b>K</b>	unott vo	oiume=	=0.023 at	
(cfs	0.16			i i i i i 	<u> </u> _      _	Runc	off Dep	th=8.76"	
-low	0.14			                 - + - + - + - +	           + -    -	- + - + - + - + - + - + - + - + - + - +	Tc	=6.0 min	
-	0.12			  - - - - - - -	 				
	0.1				L _ L				
	0.08		-+	+ - + - +	           -	· · · · · · · · · · · · · · · · · · ·			
	0.06				<u> </u> -    -			$\begin{array}{cccccccccccccccccccccccccccccccccccc$	
	0.04				<del> </del> -    -				
	0.02					·		· · · · · · · · · · · · · · · · · · ·	J
	0-1111111111111111111111111111111111111	<del>í m fan fan fan fan fan fan d</del>	m fi m fi m fi m fi						

0 2 4 6 8 10 12 14 16 18 20 22 24 26 28 30 32 34 36 38 40 42 44 46 48 50 52 54 56 58 60 62 64 66 68 70 72 Time (hours)

#### Summary for Subcatchment PR-ROOF: PR-ROOF

Runoff = 6.51 cfs @ 12.09 hrs, Volume= 0.551 af, Depth= 8.76"

Runoff by SCS TR-20 method, UH=SCS, Weighted-CN, Time Span= 0.00-72.00 hrs, dt= 0.05 hrs Type III 24-hr 100 Year Rainfall=9.00"



#### Summary for Pond DET: DET

Inflow Area	ı =	0.754 ac,10	0.00% Impervious	Inflow Depth =	8.76" for	100	Year event
Inflow	=	6.51 cfs @	12.09 hrs, Volum	e= 0.551	af		
Outflow	=	1.84 cfs @	12.43 hrs, Volum	e= 0.551	af, Atten=	72%,	Lag= 20.4 min
Primary	=	1.84 cfs @	12.43 hrs, Volum	e= 0.551	af		-

Routing by Stor-Ind method, Time Span= 0.00-72.00 hrs, dt= 0.05 hrs Peak Elev= 91.30' @ 12.43 hrs Surf.Area= 2,908 sf Storage= 9,595 cf

Plug-Flow detention time= 148.2 min calculated for 0.551 af (100% of inflow) Center-of-Mass det. time= 147.9 min (887.7 - 739.8)

Volume	١n	/ert Avai	I.Storage	Storage De	escription	
#1	88	00'	11,632 cf	Custom S	tage Data (Pi	rismatic)Listed below (Recalc)
Elevati	on	Surf.Area	Inc	.Store	Cum.Store	
(fee	et)	(sq-ft)	(cubi	c-feet)	(cubic-feet)	
88.	00	2,908		0	0	
89.	00	2,908		2,908	2,908	
90.	00	2,908		2,908	5,816	
91.	00	2,908		2,908	8,724	
92.	00	2,908		2,908	11,632	
Device	Routing	In	vert Outl	et Devices		
#1	Primary	, 88	.00' 18.0	" Round R	CP_Round	<b>18"</b> L= 10.0' Ke= 0.500
			Inlet	/ Outlet Inv	ert= 88.00' / 8	7.90' S= 0.0100 '/' Cc= 0.900
			n= C	.013 Concr	ete pipe, ben	ds & connections, Flow Area= 1.77 sf
#2	Device	1 88	.00' <b>4.0''</b>	Vert. Orific	ce/Grate C=	0.600
#3	Device	1 91	.20' <b>8.0'</b>	long x 0.5'	breadth Bro	ad-Crested Weir
			Hea	d (feet) 0.2	0 0.40 0.60	0.80 1.00
			Coe	f. (English)	2.80 2.92 3.	08 3.30 3.32
#4	Primary	91	.00' <b>15.0</b>	Round R	CP_Round	<b>15"</b> L= 10.0' Ke= 0.500
			Inlet	/ Outlet Inv	ert= 91.00' / 9	0.85' S = 0.0150 % Cc = 0.900
			n= 0	0.013 Concr	ete pipe, ben	ds & connections, Flow Area= 1.23 st
Primary	y OutFlow	<b>v</b> Max=1.81	cfs @ 12.4	43 hrs HW=	=91.30' (Free	e Discharge)

-1=RCP\_Round 18" (Passes 1.43 cfs of 13.58 cfs potential flow)

2=Orifice/Grate (Orifice Controls 0.74 cfs @ 8.52 fps)

-3=Broad-Crested Weir (Weir Controls 0.68 cfs @ 0.88 fps)

4=RCP Round 15" (Barrel Controls 0.38 cfs @ 2.56 fps)

Pond DET: DET



#### Summary for Link OUT: PR TOTAL

Inflow Are	ea =	0.785 ac,10	0.00% Impervious,	Inflow Depth = 8	76" for 100 Year event
Inflow	=	1.94 cfs @	12.42 hrs, Volume	= 0.573 af	
Primary	=	1.94 cfs @	12.42 hrs, Volume	= 0.573 af	Atten= 0%, Lag= 0.0 min

Primary outflow = Inflow, Time Span= 0.00-72.00 hrs, dt= 0.05 hrs

#### Link OUT: PR TOTAL



# Kimley »Horn

## Appendix 6

Inspection and Maintenance Schedule

#### Project: 500 Main Street

#### Location: New Rochelle, New York

#### Inspection and Maintenance Schedule

Description	Inspection Frequency	Maintenance Requirement		
Silt Fence	Weekly and after every major rainfall event	Sediment shall be removed once accumulated up to 6 inches of sediment.		
		Damaged silt fence shall be replaced immediately.		
Inlet Protection	Weekly and after every major rainfall event	Inspect integrity of the inlet protection.		
		Sediment shall be removed once accumulated up to 6 inches of sediment or affecting the		
		functionality of inlet protection.		
		Damaged inlet protection shall be replaced immediately.		
Pavement Sweeping	Bi-weekly (Twice a month) and as directed by City	Remove sediment on pavement as much as possible.		
Catch Basin Cleaning	Semi-annually	Remove all sediment within the catch basin.		
Concrete Truck Washout	Daily	Accumulated hardened material shall be removed when 75% of the storage capacity of the structure is filled.		
		Any excess wash water shall be pumped into a containment vessel and properly disposed of offsite.		
		Damaged or leaking facilities shall be repaired or replaced immediately.		
Dust Control	Daily	Maintain dust control measures through dry weather periods until all disturbed areas are stabilized.		
Tree Protection	Weekly	Damaged tree protection fence shall be replaced immediately.		
Stabilized Construction Entrance	Weekly	Inspect and perform maintenance including washing, top-dressing with additional stone, reworking, and compaction.		
		Remove sediment by shoveling or sweeping and transport to a suitable disposal area where it can be stabilized.		



## Cascade Separator<sup>™</sup> Inspection and Maintenance Guide





#### Maintenance

The Cascade Separator<sup>™</sup> system should be inspected at regular intervals and maintained when necessary to ensure optimum performance. The rate at which the system collects sediment and debris will depend upon on-site activities and site pollutant characteristics. For example, unstable soils or heavy winter sanding will cause the sediment storage sump to fill more quickly but regular sweeping of paved surfaces will slow accumulation.

#### Inspection

Inspection is the key to effective maintenance and is easily performed. Pollutant transport and deposition may vary from year to year and regular inspections will help ensure that the system is cleaned out at the appropriate time. At a minimum, inspections should be performed twice per year (i.e. spring and fall). However, more frequent inspections may be necessary in climates where winter sanding operations may lead to rapid accumulations, or in equipment wash-down areas. Installations should also be inspected more frequently where excessive amounts of trash are expected.

A visual inspection should ascertain that the system components are in working order and that there are no blockages or obstructions in the inlet chamber, flumes or outlet channel. The inspection should also quantify the accumulation of hydrocarbons, trash and sediment in the system. Measuring pollutant accumulation can be done with a calibrated dipstick, tape measure or other measuring instrument. If absorbent material is used for enhanced removal of hydrocarbons, the level of discoloration of the sorbent material should also be identified during inspection. It is useful and often required as part of an operating permit to keep a record of each inspection. A simple form for doing so is provided in this Inspection and Maintenance Guide.

Access to the Cascade Separator unit is typically achieved through one manhole access cover. The opening allows for inspection and cleanout of the center chamber (cylinder) and sediment storage sump, as well as inspection of the inlet chamber and slanted skirt. For large units, multiple manhole covers allow access to the chambers and sump.

The Cascade Separator system should be cleaned before the level of sediment in the sump reaches the maximum sediment depth and/or when an appreciable level of hydrocarbons and trash has accumulated. If sorbent material is used, it must be replaced when significant discoloration has occurred. Performance may be impacted when maximum sediment storage capacity is exceeded. Contech recommends maintaining the system when sediment level reaches the 50% storage volume. The level of sediment is easily determined by measuring from finished grade down to the top of the sediment pile. To avoid underestimating the level of sediment in the chamber, the measuring device must be lowered to the top of the sediment pile carefully. Finer, silty particles at the top of the pile typically offer less resistance to the end of the rod than larger particles toward the bottom of the pile. Once this measurement is recorded, it should be compared to the as-built drawing for the unit to determine if the height of the sediment pile off the bottom of the sump floor exceeds 50% of the total height of sediment storage sump.

### Cleaning

Cleaning of a Cascade Separator system should be done during dry weather conditions when no flow is entering the system. The use of a vacuum truck is generally the most effective and convenient method of removing pollutants from the system. Simply remove the manhole cover and insert the vacuum hose down through the center chamber and into the sump. The system should be completely drained down and the sump fully evacuated of sediment. The areas outside the center chamber and the slanted skirt should also be washed off if pollutant build-up exists in these areas.

In installations where the risk of petroleum spills is small, liquid contaminants may not accumulate as quickly as sediment. However, the system should be cleaned out immediately in the event of an oil or gasoline spill. Motor oil and other hydrocarbons that accumulate on a more routine basis should be removed when an appreciable layer has been captured. To remove these pollutants, it may be preferable to use absorbent pads since they are usually less expensive to dispose than the oil/water emulsion that may be created by vacuuming the oily layer. Trash and debris can be netted out to separate it from the other pollutants. Then the system should be power washed to ensure it is free of trash and debris.

Manhole covers should be securely seated following cleaning activities to prevent leakage of runoff into the system from above and to ensure proper safety precautions. Confined space entry procedures need to be followed if physical access is required. Disposal of all material removed from the Cascade Separator system must be done is accordance with local regulations. In many locations, disposal of evacuated sediments may be handled in the same manner as disposal of sediments removed from catch basins or deep sump manholes. Check your local regulations for specific requirements on disposal. If any components are damaged, replacement parts can be ordered from the manufacturer.



Cascade Separator Inspection & Maintenance Log							
Cascade Model:			Location:				
Date	Water Depth to Sediment <sup>1</sup>	Floatable Layer Thickness <sup>2</sup>	Describe Maintenance Performed	Maintenance Personnel	Comments		

1. The depth to sediment is determined by taking a measurement from the manhole opening to the top of the sediment pile. Once this measurement is recorded, it should be compared to the as-built drawing for the unit to determine if the height of the sediment pile off the bottom of the sump floor exceeds 50% of the total height of sediment storage sump. Note: to avoid underestimating the volume of sediment in the chamber, the measuring device must be carefully lowered to the top of the sediment pile.

2. For optimum performance, the system should be cleaned out when the floating hydrocarbon layer accumulates to an appreciable thickness. In the event of an oil spill, the system should be cleaned immediately.



A Cascade Separator unit can be easily cleaned in less than 30 minutes.



A vacuum truck excavates pollutants from the systems.

#### SUPPORT

- Drawings and specifications are available at www.ContechES.com.
- Site-specific design support is available from our engineers.
- ©2018 Contech Engineered Solutions LLC, a QUIKRETE Company

Contech Engineered Solutions LLC provides site solutions for the civil engineering industry. Contech's portfolio includes bridges, drainage, sanitary sewer, stormwater, and earth stabilization products. For information, visit www.ContechES.com or call 800.338.1122

NOTHING IN THIS CATALOG SHOULD BE CONSTRUED AS A WARRANTY. APPLICATIONS SUGGESTED HEREIN ARE DESCRIBED ONLY TO HELP READERS MAKE THEIR OWN EVALUATIONS AND DECISIONS, AND ARE NEITHER GUARANTEES NOR WARRANTIES OF SUITABILITY FOR ANY APPLICATION. CONTECH MAKES NO WARRANTY WHATSOEVER, EXPRESS OR IMPLIED, RELATED TO THE APPLICATIONS, MATERIALS, COATINGS, OR PRODUCTS DISCUSSED HEREIN. ALL IMPLIED WARRANTIES OF MERCHANTABILITY AND ALL IMPLIED WARRANTIES OF FITNESS FOR ANY PARTICULAR PURPOSE ARE DISCLAIMED BY CONTECH. SEE CONTECH'S CONDITIONS OF SALE (AVAILABLE AT WWW.CONTECHES.COM/COS) FOR MORE INFORMATION.



# Kimley »Horn

## Appendix 7

Sample of Construction Site Log Book
### APPENDIX F CONSTRUCTION SITE INSPECTION AND MAINTENANCE LOG BOOK

## STATE POLLUTANT DISCHARGE ELIMINATION SYSTEM FOR CONSTRUCTION ACTIVITIES

## SAMPLE CONSTRUCTION SITE LOG BOOK

### Table of Contents

- I. Pre-Construction Meeting Documents
  - a. Preamble to Site Assessment and Inspections
  - b. Pre-Construction Site Assessment Checklist

### **II.** Construction Duration Inspections

- a. Directions
- b. Modification to the SWPPP

#### I. PRE-CONSTRUCTION MEETING DOCUMENTS

Project Name	
Permit No.	Date of Authorization
Name of Operator	
Prime Contractor	

#### a. Preamble to Site Assessment and Inspections

The Following Information To Be Read By All Person's Involved in The Construction of Stormwater Related Activities:

The Operator agrees to have a qualified inspector<sup>1</sup> conduct an assessment of the site prior to the commencement of construction<sup>2</sup> and certify in this inspection report that the appropriate erosion and sediment controls described in the SWPPP have been adequately installed or implemented to ensure overall preparedness of the site for the commencement of construction.

Prior to the commencement of construction, the Operator shall certify in this site logbook that the SWPPP has been prepared in accordance with the State's standards and meets all Federal, State and local erosion and sediment control requirements. A preconstruction meeting should be held to review all of the SWPPP requirements with construction personnel.

When construction starts, site inspections shall be conducted by the qualified inspector at least every 7 calendar days. The Operator shall maintain a record of all inspection reports in this site logbook. The site logbook shall be maintained on site and be made available to the permitting authorities upon request.

Prior to filing the Notice of Termination or the end of permit term, the Operator shall have a qualified inspector perform a final site inspection. The qualified inspector shall certify that the site has undergone final stabilization<sup>3</sup> using either vegetative or structural stabilization methods and that all temporary erosion and sediment controls (such as silt fencing) not needed for long-term erosion control have been removed. In addition, the Operator must identify and certify that all permanent structures described in the SWPPP have been constructed and provide the owner(s) with an operation and maintenance plan that ensures the structure(s) continuously functions as designed.

1 Refer to "Qualified Inspector" inspection requirements in the current SPDES General Permit for Stormwater Discharges from Construction Activity for complete list of inspection requirements.

3 "Final stabilization" means that all soil-disturbing activities at the site have been completed and a uniform, perennial vegetative cover with a density of eighty (80) percent has been established or equivalent stabilization measures (such as the use of mulches or geotextiles) have been employed on all unpaved areas and areas not covered by permanent structures.

<sup>2 &</sup>quot;Commencement of construction" means the initial removal of vegetation and disturbance of soils associated with clearing, grading or excavating activities or other construction activities.

#### b. Pre-construction Site Assessment Checklist (NOTE: Provide comments below as necessary)

1. Notice of Intent, SWPPP, and Contractors Certification:

#### Yes No NA

- [] [] Has a Notice of Intent been filed with the NYS Department of Conservation?
- [] [] [] Is the SWPPP on-site? Where?
- [] [] [] Is the Plan current? What is the latest revision date?\_\_\_\_\_
- [] [] Is a copy of the NOI (with brief description) onsite? Where?
- [] [] Have all contractors involved with stormwater related activities signed a contractor's certification?

#### 2. Resource Protection

#### Yes No NA

- [] [] Are construction limits clearly flagged or fenced?
- [] [] Important trees and associated rooting zones, on-site septic system absorption fields, existing vegetated areas suitable for filter strips, especially in perimeter areas, have been flagged for protection.
- [] [] Creek crossings installed prior to land-disturbing activity, including clearing and blasting.
- 3. Surface Water Protection

#### Yes No NA

- [] [] Clean stormwater runoff has been diverted from areas to be disturbed.
- [] [] Bodies of water located either on site or in the vicinity of the site have been identified and protected.
- [] [] Appropriate practices to protect on-site or downstream surface water are installed.
- [] [] Are clearing and grading operations divided into areas <5 acres?

#### 4. Stabilized Construction Access

Yes No NA

- [] [] A temporary construction entrance to capture mud and debris from construction vehicles before they enter the public highway has been installed.
- [] [] Other access areas (entrances, construction routes, equipment parking areas) are stabilized immediately as work takes place with gravel or other cover.
- [] [] Sediment tracked onto public streets is removed or cleaned on a regular basis.
- 5. Sediment Controls

#### Yes No NA

- [] [] Silt fence material and installation comply with the standard drawing and specifications.
- [] [] Silt fences are installed at appropriate spacing intervals
- [] [] Sediment/detention basin was installed as first land disturbing activity.
- [] [] [] Sediment traps and barriers are installed.

#### 6. Pollution Prevention for Waste and Hazardous Materials

#### Yes No NA

- [] [] The Operator or designated representative has been assigned to implement the spill prevention avoidance and response plan.
- [] [] The plan is contained in the SWPPP on page
- [] [] Appropriate materials to control spills are onsite. Where?

#### **II. CONSTRUCTION DURATION INSPECTIONS**

#### a. Directions:

#### Inspection Forms will be filled out during the entire construction phase of the project.

Required Elements:

- 1) On a site map, indicate the extent of all disturbed site areas and drainage pathways. Indicate site areas that are expected to undergo initial disturbance or significant site work within the next 14-day period;
- 2) Indicate on a site map all areas of the site that have undergone temporary or permanent stabilization;
- 3) Indicate all disturbed site areas that have not undergone active site work during the previous 14-day period;
- 4) Inspect all sediment control practices and record the approximate degree of sediment accumulation as a percentage of sediment storage volume (for example, 10 percent, 20 percent, 50 percent);
- 5) Inspect all erosion and sediment control practices and record all maintenance requirements such as verifying the integrity of barrier or diversion systems (earthen berms or silt fencing) and containment systems (sediment basins and sediment traps). Identify any evidence of rill or gully erosion occurring on slopes and any loss of stabilizing vegetation or seeding/mulching. Document any excessive deposition of sediment or ponding water along barrier or diversion systems. Record the depth of sediment within containment structures, any erosion near outlet and overflow structures, and verify the ability of rock filters around perforated riser pipes to pass water; and
- 6) Immediately report to the Operator any deficiencies that are identified with the implementation of the SWPPP.

#### SITE PLAN/SKETCH

Inspector (print name)Date of Inspection

**Qualified Inspector (print name)** 

Qualified Inspector Signature

The above signed acknowledges that, to the best of his/her knowledge, all information provided on the forms is accurate and complete.

#### **CONSTRUCTION DURATION INSPECTIONS**

#### **Maintaining Water Quality**

#### Yes No NA

- [] [] Is there an increase in turbidity causing a substantial visible contrast to natural conditions at the outfalls?
- [] [] Is there residue from oil and floating substances, visible oil film, or globules or grease at the outfalls?
- [] [] All disturbance is within the limits of the approved plans.
- [] [] Have receiving lake/bay, stream, and/or wetland been impacted by silt from project?

#### Housekeeping

1. General Site Conditions

#### Yes No NA

- [] [] [] Is construction site litter, debris and spoils appropriately managed?
- [] [] Are facilities and equipment necessary for implementation of erosion and sediment control in working order and/or properly maintained?
- [] [] [] Is construction impacting the adjacent property?
- [] [] [] Is dust adequately controlled?

#### 2. Temporary Stream Crossing

#### Yes No NA

- [] [] Maximum diameter pipes necessary to span creek without dredging are installed.
- [] [] Installed non-woven geotextile fabric beneath approaches.
- [] [] Is fill composed of aggregate (no earth or soil)?
- [] [] Rock on approaches is clean enough to remove mud from vehicles & prevent sediment from entering stream during high flow.
- 3. Stabilized Construction Access

#### Yes No NA

- [] [] [] Stone is clean enough to effectively remove mud from vehicles.
- [] [] Installed per standards and specifications?
- [] [] Does all traffic use the stabilized entrance to enter and leave site?
- [] [] [] Is adequate drainage provided to prevent ponding at entrance?

#### **Runoff Control Practices**

1. Excavation Dewatering

#### Yes No NA

- [] [] Upstream and downstream berms (sandbags, inflatable dams, etc.) are installed per plan.
- [] [] Clean water from upstream pool is being pumped to the downstream pool.
- [] [] Sediment laden water from work area is being discharged to a silt-trapping device.
- [] [] Constructed upstream berm with one-foot minimum freeboard.

#### **Runoff Control Practices (continued)**

2. Flow Spreader

#### Yes No NA

- [] [] [] Installed per plan.
- [] [] Constructed on undisturbed soil, not on fill, receiving only clear, non-sediment laden flow.
- [] [] Flow sheets out of level spreader without erosion on downstream edge.

#### 3. Interceptor Dikes and Swales

#### Yes No NA

- [] [] [] Installed per plan with minimum side slopes 2H:1V or flatter.
- [] [] Stabilized by geotextile fabric, seed, or mulch with no erosion occurring.
- [] [] Sediment-laden runoff directed to sediment trapping structure

#### 4. Stone Check Dam

#### Yes No NA

- [] [] [] Is channel stable? (flow is not eroding soil underneath or around the structure).
- [] [] Check is in good condition (rocks in place and no permanent pools behind the structure).
- [] [] Has accumulated sediment been removed?.

#### 5. Rock Outlet Protection

#### Yes No NA

- [] [] [] Installed per plan.
- [] [] Installed concurrently with pipe installation.

#### Soil Stabilization

1. Topsoil and Spoil Stockpiles

#### Yes No NA

- [] [] [] Stockpiles are stabilized with vegetation and/or mulch.
- [] [] Sediment control is installed at the toe of the slope.

#### 2. Revegetation

#### Yes No NA

- [] [] [] Temporary seedings and mulch have been applied to idle areas.
- [] [] 4 inches minimum of topsoil has been applied under permanent seedings

#### Sediment Control Practices

1. Silt Fence and Linear Barriers

#### Yes No NA

- [] [] [] Installed on Contour, 10 feet from toe of slope (not across conveyance channels).
- [] [] Joints constructed by wrapping the two ends together for continuous support.
- [] [] Fabric buried 6 inches minimum.
- [] [] Posts are stable, fabric is tight and without rips or frayed areas.

Sediment accumulation is \_\_\_% of design capacity.

#### CONSTRUCTION DURATION INSPECTIONS

Page 4 of \_\_\_\_\_

#### Sediment Control Practices (continued)

2. Storm Drain Inlet Protection (Use for Stone & Block; Filter Fabric; Curb; or, Excavated; Filter Sock or Manufactured practices)

#### Yes No NA

- [] [] Installed concrete blocks lengthwise so open ends face outward, not upward.
- [] [] Placed wire screen between No. 3 crushed stone and concrete blocks.
- [] [] Drainage area is 1acre or less.
- [] [] [] Excavated area is 900 cubic feet.
- [] [] Excavated side slopes should be 2:1.
- [] [] 2" x 4" frame is constructed and structurally sound.
- [] [] Posts 3-foot maximum spacing between posts.
- [] [] Fabric is embedded 1 to 1.5 feet below ground and secured to frame/posts with staples at max 8-inch spacing.
- [] [] Posts are stable, fabric is tight and without rips or frayed areas.
- [] [] [] Manufactured insert fabric is free of tears and punctures.
- [] [] Filter Sock is not torn or flattened and fill material is contained within the mesh sock.

Sediment accumulation \_\_\_\_% of design capacity.

3. Temporary Sediment Trap

#### Yes No NA

- [] [] Outlet structure is constructed per the approved plan or drawing.
- [] [] Geotextile fabric has been placed beneath rock fill.
- [] [] Sediment trap slopes and disturbed areas are stabilized.

Sediment accumulation is \_\_\_% of design capacity.

4. Temporary Sediment Basin

#### Yes No NA

- [] [] Basin and outlet structure constructed per the approved plan.
- [] [] Basin side slopes are stabilized with seed/mulch.
- [] [] Drainage structure flushed and basin surface restored upon removal of sediment basin facility.
- [] [] Sediment basin dewatering pool is dewatering at appropriate rate.

Sediment accumulation is \_\_\_% of design capacity.

<u>Note</u>: Not all erosion and sediment control practices are included in this listing. Add additional pages to this list as required by site specific design. All practices shall be maintained in accordance with their respective standards.

Construction inspection checklists for post-development stormwater management practices can be found in Appendix F of the New York Stormwater Management Design Manual.

#### CONSTRUCTION DURATION INSPECTIONS

#### **b.** Modifications to the SWPPP (To be completed as described below)

The Operator shall amend the SWPPP whenever:

- 1. There is a significant change in design, construction, operation, or maintenance which may have a significant effect on the potential for the discharge of pollutants to the waters of the United States and which has not otherwise been addressed in the SWPPP; or
- 2. The SWPPP proves to be ineffective in:
  - a. Eliminating or significantly minimizing pollutants from sources identified in the SWPPP and as required by this permit; or
  - b. Achieving the general objectives of controlling pollutants in stormwater discharges from permitted construction activity; and
- 3. Additionally, the SWPPP shall be amended to identify any new contractor or subcontractor that will implement any measure of the SWPPP.

#### **Modification & Reason:**

## Kimley »Horn

## Appendix 8

NYS Dishwasher Detergent and Nutrient Runoff Law



Department of Environmental Conservation

## Lawn Fertilizer (NYS Nutrient Runoff Law)

## Look for the Zero!

Before buying lawn fertilizer, check the bag for a set of three numbers showing the percentage of nitrogen, phosphorus and potassium. Buy a bag with a **"0"** in the middle.

# Zero in the middle means phosphorus free and that means....

**Zero pollution -** Phosphorus is one of the leading causes of water pollution. Even if you live far from a water body, excess phosphorus from your lawn can wash off and pollute lakes and streams, harming fish and ruining boating and swimming. More than 100 water bodies in New York State cannot be used for drinking, fishing or swimming because they contain too much phosphorus.

**Zero waste -** Why pay for a chemical your lawn doesn't need? Generally, only newly established lawns or those with poor soil need phosphorus. Phosphorus applied to a lawn that doesn't need it won't be used and can cause water pollution.



**Zero hassle -** It's against the law to use phosphorus on lawns that don't need it. (New York State Environmental Conservation Law, article 17, title 21 and Agriculture and Markets Law § 146-g) Check local laws, too-some municipalities have stricter laws about selling and using lawn fertilizers.



An over-fertilized lawn can lead to a green lake! Excess phosphorus in waterbodies can cause algae overgrowth, including harmful algal blooms, with serious impacts to the environment and public health.

## When fertilizing your lawn...

Follow the requirements of the law. Do not:

- Use lawn fertilizer that contains phosphorus unless you are establishing a new lawn, or a soil test shows that your lawn does not have enough phosphorus.
- Apply any lawn fertilizer December 1 April 1.
- Apply fertilizer on sidewalks, driveways or other impervious surfaces. If fertilizer spills onto these surfaces, you MUST sweep it up to prevent it from washing into drains or waterways. Do not hose if off.
- Apply lawn fertilizer within 20 feet of any water body unless...
  - There is at least a 10-foot buffer of shrubs, trees or other plants between the area you are fertilizing and the water,
    - or
  - Fertilizer can be applied no closer than 3 feet from the water using a device with a spreader guard, deflector shield or drop spreader.

#### What should I see at the store?

Retailers who sell fertilizer must display phosphorus-containing fertilizer separately from phosphorus-free fertilizers and post a sign near the display. A sample sign is available in the "Important Links" section on the right-hand side of this page.

#### The law applies to:

- · Homeowners applying fertilizer themselves
- · Landscapers and lawn care professionals
- · Pesticide applicators
- · Retailers, distributors and manufacturers of lawn fertilizers
- Fertilizer/pesticide combination products (sometimes called "weed and feeds") when these products contain over 0.67% phosphorus.
- Organic phosphorus fertilizer (such as bone meal).

#### The law does not apply to:

- Use of products with 0.67 in the middle or lower
- Agricultural fertilizer or fertilizer for trees, shrubs or gardens
- Compost

#### Penalties:

- For an owner, owner's agent, or occupant of a household, the penalties are: issuance of a written warning with educational materials for a first violation; a fine of up to \$100 for a second violation; and fines up to \$250 for subsequent violations.
- The penalties for all others are: a fine up to \$500 for a first of violation; and fines up to \$1000 for subsequent offenses.

## Get a Soil Test

If you think your lawn might need extra phosphorus, test your soil. Tests cost \$10-\$20. There are several options:

• Have testing done through your local Cornell Cooperative Extension office.

- Find a commercial laboratory that tests soil.
- Use a home test kit. These tests tend to be less accurate and do not come with fertilizer recommendations.

## Fertilizer and Nitrogen on Long Island

Nitrogen runoff impacts both surface and groundwater quality on Long Island. Suffolk and Nassau counties have their own fertilizer laws to restrict nitrogen from fertilizer from reaching waterbodies. Visit the counties' websites for details about their respective regulations.

### **Dishwasher Detergent**

The Nutrient Runoff Law also includes provisions regarding the sale of dishwasher detergent:

- The sale of newly stocked, phosphorus-containing dishwasher detergents for household use and commercial use is prohibited in NYS.
- There is no change to the phosphorus limits for detergents used to clean dairy equipment or food processing equipment.

As retailers are no longer allowed to sell phosphorus-containing dish detergent in NYS, consumers do not need to take any steps to comply with this portion of the law.

## More about Lawn Fertilizer (NYS Nutrient Runoff Law):

Text of Nutrient Runoff Law - Environmental Conservation Law, article 17, title 21 and Agriculture and Markets Law § 146-g, effective January 2012

FAQ for Lawn Fertilizer - Additional information about the specifics of the Nutrient Runoff Law.

## Kimley »Horn

## Appendix 9 Water Quality Calculations



Kimley Horn of New York, P.C. 1 N Lexington Avenue, Suite 1575 White Plains, New York 10601

Calculated By: CJO Checked By:

#### CS-4 Unit Design

#### Compute Water Quality Volume (Drainage Area "PR-ROOF")

NYSDEC Required Water Quality Volume (90% Storm Event)

D.	1 5	- 00% Painfall Event Number from Figure #1			
Ρ.	1.5	- 90% Raimai Event Number nom Figure #			
Rv:	0.9500	= 0.05 + 0.009(I)			
1:	100	= Impervious	coverage p	ercentage	
l:	0.785	= impervious	area (in acı	res)	
A:	0.785	= Total Drainage Area (in acres)			
WQv:	<u>0.093</u>	= Req'd Water Quality Volume (in ac-ft)			
		= <u>(P)(Rv)(A)</u>			
		12			
	100% WQv =	0.093	ac-ft		
	Designed WQv =	0.093	ac-ft		

(a) Compute peak water quality discharge (Qwq):

	P = Qa = WQy / Area	1.5 inch			
	Qa =	<u>1.43</u> inch			
	CN = 1000 / [10 + 5 CN =	5P + 10Q <sub>a</sub> - 10(Q <sub>a</sub> <sup>2</sup> + <u><b>99.4</b></u>	· 1.25*Q <sub>a</sub> *P)^0.5	]	
From TR-55, Table 4-1	1:	la = 0.041	la / P =	0.027	
From TR-55, Exhibit 4	-111:	q <sub>u</sub> = 65	0 csm/in		
Q <sub>wq</sub> = (q <sub>u</sub> )(Site Area, ac/ 640 ac/ sq. mi)(Q <sub>a</sub> )					
Required	Q <sub>wq</sub> = 1.14	cfs			
			_		

#### 100-year by-pass Storm Discharge Rate

Q<sub>100</sub> = 6.8 cfs

Water Quality Unit Sizing

Proposed Q <sub>wq</sub> = (Use CS-4)	1.80	cfs	
By-Pass Consoity Flow			
Rate Q <sub>100</sub> =	> 10	cfs	

#### **CASCADE SEPARATOR DESIGN NOTES**



P

THE STANDARD CS-4 CONFIGURATION IS SHOWN. ALTERNATE CONFIGURATIONS ARE AVAILABLE AND ARE LISTED BELOW. SOME CONFIGURATIONS

SITE SPECIFIC
DATA REQUIREMENTS

STRUCTURE ID						
WATER QUALITY FLO						
PEAK FLOW RATE (cfs	; [L/s])					
RETURN PERIOD OF F	3)					
RIM ELEVATION						
PIPE DATA:	PIPE DATA: INVERT MATERIAL					
INLET PIPE 1						
INLET PIPE 2						
OUTLET PIPE						
NOTES / SPECIAL REQUIREMENTS:						

FOR SITE SPECIFIC DRAWINGS WITH DETAILED STRUCTURE DIMENSIONS AND WEIGHT, PLEASE CONTACT YOUR CONTECH ENGINEERED

CASCADE SEPARATOR WATER QUALITY STRUCTURE SHALL BE IN ACCORDANCE WITH ALL DESIGN DATA AND INFORMATION CONTAINED IN

CASCADE SEPARATOR STRUCTURE SHALL MEET AASHTO HS20 LOAD RATING, ASSUMING EARTH COVER OF 0' - 2' [610], AND GROUNDWATER ELEVATION AT, OR BELOW, THE OUTLET PIPE INVERT ELEVATION. ENGINEER OF RECORD TO CONFIRM ACTUAL GROUNDWATER ELEVATION.

CASCADE SEPARATOR STRUCTURE SHALL BE PRECAST CONCRETE CONFORMING TO ASTM C478 AND AASHTO LOAD FACTOR DESIGN

A. ANY SUB-BASE, BACKFILL DEPTH, AND/OR ANTI-FLOTATION PROVISIONS ARE SITE-SPECIFIC DESIGN CONSIDERATIONS AND SHALL BE

CONTRACTOR TO PROVIDE EQUIPMENT WITH SUFFICIENT LIFTING AND REACH CAPACITY TO LIFT AND SET THE CASCADE SEPARATOR

CONTRACTOR TO PROVIDE, INSTALL, AND GROUT INLET AND OUTLET PIPE(S). MATCH PIPE INVERTS WITH ELEVATIONS SHOWN. ALL PIPE

CONTRACTOR TO TAKE APPROPRIATE MEASURES TO ASSURE UNIT IS WATER TIGHT, HOLDING WATER TO FLOWLINE INVERT MINIMUM. IT IS

CS-4 CASCADE SEPARATOR STANDARD DETAIL



### State of New Jersey

DEPARTMENT OF ENVIRONMENTAL PROTECTION Bureau of Nonpoint Pollution Control Division of Water Quality 401-02B Post Office Box 420 Trenton, New Jersey 08625-0420 609-633-7021 Fax: 609-777-0432 http://www.state.nj.us/dep/dwg/bnpc home.htm

CATHERINE R. MCCABE Commissioner

May 18, 2020

Derek M. Berg Director – Stormwater Regulatory Management - East Contech Engineered Solutions LLC 71 US Route 1, Suite F Scarborough, ME 04074

Re: MTD Lab Certification Cascade Separator<sup>TM</sup> On-line Installation

#### **TSS Removal Rate 50%**

Dear Mr. Berg:

This revised certification letter supersedes the Department's prior certification dated October 1, 2019. This revision was completed to reflect Contech's enhanced fabrication capability to manufacture a smaller-size unit of its the Cascade Separator<sup>™</sup> Manufactured Treatment Device (MTD), while still meeting the scaling methodology as agreed upon by the manufacturers' working group on September 19, 2016. Based on this modification, Table A-1 of the New Jersey Corporation for Advanced Technology (NJCAT) Verification report located at <a href="http://www.njcat.org/uploads/newDocs/NJCATTechnologyVerificationFinal.pdf">http://www.njcat.org/uploads/newDocs/NJCATTechnologyVerificationFinal.pdf</a> has been revised to reflect this same updated model size and flow rate.

The Stormwater Management rules under N.J.A.C. 7:8-5.5(b) and 5.7(c) allow the use of manufactured treatment devices (MTDs) for compliance with the design and performance standards at N.J.A.C. 7:8-5 if the pollutant removal rates have been verified by the New Jersey Corporation for Advanced Technology (NJCAT) and have been certified by the New Jersey Department of Environmental Protection (NJDEP). Contech Engineered Solutions, LLC (Contech) has requested an MTD Laboratory Certification for the Cascade Separator<sup>TM</sup> stormwater treatment system.

The project falls under the "Procedure for Obtaining Verification of a Stormwater Manufactured Treatment Device from New Jersey Corporation for Advance Technology" dated January 25,

PHILIP D. MURPHY Governor

SHEILA Y. OLIVER Lt. Governor 2013. The applicable protocol is the "New Jersey Laboratory Testing Protocol to Assess Total Suspended Solids Removal by a Hydrodynamic Sedimentation Manufactured Treatment Device" dated January 25, 2013.

NJCAT verification documents submitted to the NJDEP indicate that the requirements of the aforementioned protocol have been met or exceeded. The NJCAT letter also included a recommended certification TSS removal rate and the required maintenance plan. The NJCAT Verification Report with the Verification Appendix (dated September 2019) for this device is published online at <u>http://www.njcat.org/verification-process/technology-verification-database.html.</u>

# The NJDEP certifies the use of the Cascade Separator<sup>™</sup> stormwater treatment system at a TSS removal rate of 50% when designed, operated, and maintained in accordance with the information provided in the Verification Appendix and the following conditions:

- 1. The maximum treatment flow rate (MTFR) for the manufactured treatment device (MTD) is calculated using the New Jersey Water Quality Design Storm (1.25 inches in 2 hrs) in N.J.A.C. 7:8-5.5.
- 2. The Cascade Separator<sup>TM</sup> shall be installed using the same configuration reviewed by NJCAT and shall be sized in accordance with the criteria specified in item 6 below.
- 3. This Cascade Separator<sup>™</sup> cannot be used in series with another MTD or a media filter (such as a sand filter) to achieve an enhanced removal rate for total suspended solids (TSS) removal under N.J.A.C. 7:8-5.5.
- 4. Additional design criteria for MTDs can be found in Chapter 9.6 of the New Jersey Stormwater Best Management Practices (NJ Stormwater BMP) Manual, which can be found online at <u>www.njstormwater.org</u>.
- 5. The maintenance plan for a site using this device shall incorporate, at a minimum, the maintenance requirements for the Cascade Separator<sup>TM</sup>. A copy of the maintenance plan is attached to this certification. However, it is recommended to review the maintenance website at <u>https://www.conteches.com/Portals/0/Documents/Maintenance%20Guides/Cascade-Maintenance%20Guide.pdf?ver=2018-11-05-093254-300</u>. for any changes to the maintenance requirements.
- 6. Sizing Requirement:

The example below demonstrates the sizing procedure for the Cascade Separator<sup>TM</sup>:

Example: A 0.25-acre impervious site is to be treated to 50% TSS removal using a Cascade Separator<sup>TM</sup>. The impervious site runoff (Q) based on the New Jersey Water Quality Design Storm was determined to be 0.79 cfs.

Maximum Treatment Flow Rate (MTFR) Evaluation:

The site runoff (Q) was based on the following:

time of concentration = 10 minutes i = 3.2 in/hr (page 5-8, Fig. 5-3 of the NJ Stormwater BMP Manual) c = 0.99 (runoff coefficient for impervious) Q = ciA = 0.99 x 3.2 x 0.25 = 0.79 cfs

Given the site runoff is 0.79 cfs and based on Table A-1 below, the Cascade Separator<sup>TM</sup> Model CS-3 with an MTFR of 1.02 cfs would be the smallest model approved that could be used for this site to remove 50% of the TSS from the impervious area without exceeding the MTFR.

The sizing table corresponding to the available system models is noted below. Additional specifications regarding each model can be found in the Verification Appendix under Table A-1.

Model	Manhole Diameter (ft)	MTFR (cfs)	50% Maximum Sediment Storage Area Volume (ft <sup>3</sup> )
CS-3	3	1.02	5.3
CS-4	4	1.80	9.4
CS-5	5	2.81	14.7
CS-6	6	4.05	21.2
CS-8	8	7.20	37.7
CS-10	10	11.3	58.9
CS-12	12	16.2	84.8

Table A-1 Cascade Separator<sup>™</sup> Models and Associated MTFRs

A detailed maintenance plan is mandatory for any project with a stormwater BMP subject to the Stormwater Management rules under N.J.A.C. 7:8. The plan must include all of the items identified in the Maintenance requirements section of the Stormwater Management rules under N.J.A.C. 7:8-5.8. Such items include, but are not limited to, the list of inspection and maintenance equipment and tools, specific corrective and preventative maintenance tasks, indication of problems in the system, and training of maintenance personnel. Additional information can be found in Chapter 8: Maintenance and Retrofit of Stormwater Management Measures.

If you have any questions regarding the above information, please contact Brian Salvo of my office at (609) 633-7021.

Sincerely,

Labiel Mahor

Gabriel Mahon, Chief Bureau of Nonpoint Pollution Control

Attachment: Maintenance Plan

cc: Chron File Richard Magee, NJCAT Jim Murphy, NJDEP-BNPC Vince Mazzei, NJDEP-DLUR Brian Salvo, NJDEP-BNPC



## Cascade Separator<sup>™</sup> Inspection and Maintenance Guide





#### Maintenance

The Cascade Separator<sup>™</sup> system should be inspected at regular intervals and maintained when necessary to ensure optimum performance. The rate at which the system collects sediment and debris will depend upon on-site activities and site pollutant characteristics. For example, unstable soils or heavy winter sanding will cause the sediment storage sump to fill more quickly but regular sweeping of paved surfaces will slow accumulation.

#### Inspection

Inspection is the key to effective maintenance and is easily performed. Pollutant transport and deposition may vary from year to year and regular inspections will help ensure that the system is cleaned out at the appropriate time. At a minimum, inspections should be performed twice per year (i.e. spring and fall). However, more frequent inspections may be necessary in climates where winter sanding operations may lead to rapid accumulations, or in equipment wash-down areas. Installations should also be inspected more frequently where excessive amounts of trash are expected.

A visual inspection should ascertain that the system components are in working order and that there are no blockages or obstructions in the inlet chamber, flumes or outlet channel. The inspection should also quantify the accumulation of hydrocarbons, trash and sediment in the system. Measuring pollutant accumulation can be done with a calibrated dipstick, tape measure or other measuring instrument. If absorbent material is used for enhanced removal of hydrocarbons, the level of discoloration of the sorbent material should also be identified during inspection. It is useful and often required as part of an operating permit to keep a record of each inspection. A simple form for doing so is provided in this Inspection and Maintenance Guide.

Access to the Cascade Separator unit is typically achieved through one manhole access cover. The opening allows for inspection and cleanout of the center chamber (cylinder) and sediment storage sump, as well as inspection of the inlet chamber and slanted skirt. For large units, multiple manhole covers allow access to the chambers and sump.

The Cascade Separator system should be cleaned before the level of sediment in the sump reaches the maximum sediment depth and/or when an appreciable level of hydrocarbons and trash has accumulated. If sorbent material is used, it must be replaced when significant discoloration has occurred. Performance may be impacted when maximum sediment storage capacity is exceeded. Contech recommends maintaining the system when sediment level reaches the 50% storage volume. The level of sediment is easily determined by measuring from finished grade down to the top of the sediment pile. To avoid underestimating the level of sediment in the chamber, the measuring device must be lowered to the top of the sediment pile carefully. Finer, silty particles at the top of the pile typically offer less resistance to the end of the rod than larger particles toward the bottom of the pile. Once this measurement is recorded, it should be compared to the as-built drawing for the unit to determine if the height of the sediment pile off the bottom of the sump floor exceeds 50% of the total height of sediment storage sump.

#### Cleaning

Cleaning of a Cascade Separator system should be done during dry weather conditions when no flow is entering the system. The use of a vacuum truck is generally the most effective and convenient method of removing pollutants from the system. Simply remove the manhole cover and insert the vacuum hose down through the center chamber and into the sump. The system should be completely drained down and the sump fully evacuated of sediment. The areas outside the center chamber and the slanted skirt should also be washed off if pollutant build-up exists in these areas.

In installations where the risk of petroleum spills is small, liquid contaminants may not accumulate as quickly as sediment. However, the system should be cleaned out immediately in the event of an oil or gasoline spill. Motor oil and other hydrocarbons that accumulate on a more routine basis should be removed when an appreciable layer has been captured. To remove these pollutants, it may be preferable to use absorbent pads since they are usually less expensive to dispose than the oil/water emulsion that may be created by vacuuming the oily layer. Trash and debris can be netted out to separate it from the other pollutants. Then the system should be power washed to ensure it is free of trash and debris.

Manhole covers should be securely seated following cleaning activities to prevent leakage of runoff into the system from above and to ensure proper safety precautions. Confined space entry procedures need to be followed if physical access is required. Disposal of all material removed from the Cascade Separator system must be done is accordance with local regulations. In many locations, disposal of evacuated sediments may be handled in the same manner as disposal of sediments removed from catch basins or deep sump manholes. Check your local regulations for specific requirements on disposal. If any components are damaged, replacement parts can be ordered from the manufacturer.



Cascade Separator Inspection & Maintenance Log						
Cascade Model:			Location:			
Date	Water Depth to Sediment <sup>1</sup>	Floatable Layer Thickness <sup>2</sup>	Describe Maintenance Performed	Maintenance Personnel	Comments	

1. The depth to sediment is determined by taking a measurement from the manhole opening to the top of the sediment pile. Once this measurement is recorded, it should be compared to the as-built drawing for the unit to determine if the height of the sediment pile off the bottom of the sump floor exceeds 50% of the total height of sediment storage sump. Note: to avoid underestimating the volume of sediment in the chamber, the measuring device must be carefully lowered to the top of the sediment pile.

2. For optimum performance, the system should be cleaned out when the floating hydrocarbon layer accumulates to an appreciable thickness. In the event of an oil spill, the system should be cleaned immediately.



A Cascade Separator unit can be easily cleaned in less than 30 minutes.



A vacuum truck excavates pollutants from the systems.

#### SUPPORT

- Drawings and specifications are available at www.ContechES.com.
- Site-specific design support is available from our engineers.
- ©2018 Contech Engineered Solutions LLC, a QUIKRETE Company

Contech Engineered Solutions LLC provides site solutions for the civil engineering industry. Contech's portfolio includes bridges, drainage, sanitary sewer, stormwater, and earth stabilization products. For information, visit www.ContechES.com or call 800.338.1122

NOTHING IN THIS CATALOG SHOULD BE CONSTRUED AS A WARRANTY. APPLICATIONS SUGGESTED HEREIN ARE DESCRIBED ONLY TO HELP READERS MAKE THEIR OWN EVALUATIONS AND DECISIONS, AND ARE NEITHER GUARANTEES NOR WARRANTIES OF SUITABILITY FOR ANY APPLICATION. CONTECH MAKES NO WARRANTY WHATSOEVER, EXPRESS OR IMPLIED, RELATED TO THE APPLICATIONS, MATERIALS, COATINGS, OR PRODUCTS DISCUSSED HEREIN. ALL IMPLIED WARRANTIES OF MERCHANTABILITY PAID ALL IMPLIED WARRANTIES OF FITNESS FOR ANY PARTICULAR PURPOSE ARE DISCLAIMED BY CONTECH. SEE CONTECH'S CONDITIONS OF SALE (AVAILABLE AT WWW.CONTECHES.COM/COS) FOR MORE INFORMATION.

